

IN THE UNITED STATES BANKRUPTCY COURT FOR THE EASTERN DISTRICT OF TEXAS TYLER DIVISION

In re:)	Chapter 11
O'BENCO IV, LP)	Case No. 19-60384 (BP)
)	
Debtor.)	

ORDER (A) AUTHORIZING AND APPROVING (I) THE ASSET PURCHASE AGREEMENT FOR THE KANSAS, WEST TEXAS, AND OTHER ASSETS; (II) THE SALE OF THE DEBTOR'S KANSAS, WEST TEXAS AND OTHER ASSETS FREE AND CLEAR OF ALL LIENS, CLAIMS, ENCUMBRANCES AND INTERESTS; AND (III) THE ASSUMPTION AND ASSIGNMENT OF CERTAIN EXECUTORY CONTRACTS AND UNEXPIRED LEASES; AND

(B) GRANTING RELATED RELIEF

[Relates to Doc. No. 60]

Upon the motion (the "Sale Motion")¹ of the above captioned debtor and debtor in possession (the "Debtor") in the above-captioned case (the "Chapter 11 Case") for entry of an order (this "Sale Order") (a) approving the sale (the "Sale") of all of the Debtor's oil and gas assets, free and clear of all liens, claims, encumbrances and interests of any kind to the Successful Bidder(s) and (b) authorizing the assumption and assignment of certain executory contracts and unexpired leases (the "Assigned Contracts") of the Debtor that are to be assumed and assigned to the Successful Bidder(s) in connection with the Sale; and the Court having entered an order approving, among other things, the Bidding Procedures (the "Bidding Procedures Order") [Dkt. No. 30], as well as an order amending the Bidding Procedures (the "Order Amending Bidding Procedures") [Dkt. No. 102]; and the Auction having been held in accordance with the Bidding

Capitalized terms not otherwise defined herein shall have the meanings ascribed to such terms in the Asset Purchase Agreement (as defined below) or, if not defined in the Asset Purchase Agreement, the Sale Motion.

² "Bidding Procedures" as used herein shall include the bidding procedures as amended by the Order Amending Bidding Procedures.

Procedures Order; and at the conclusion of the Auction, Pachira Oil and Gas, LLC (the "Buyer") having been chosen as the Successful Bidder for the Kansas, West Texas, and Other Assets;³ and upon the Buyer and Debtor having entered into that certain Asset Purchase Agreement, dated August 27, 2019 (together with all ancillary documents, as may be amended, modified or supplemented, the "Asset Purchase Agreement", attached hereto as Attachment 1); and the Court having conducted the Sale Hearing on August 30, 2019; and all parties in interest having been heard or having had the opportunity to be heard regarding the Asset Purchase Agreement; and it appearing that adequate and proper notice of the Sale Motion has been given and that no other or further notice need be given; and the Sale Hearing having been held to consider the relief requested in the Sale Motion; and upon the record of the Sale Hearing and all of the proceedings had before the Court; and the Court having found and determined that the relief sought in the Sale Motion is in the best interests of the Debtor, its estate, its creditors and all other parties in interest; and that the legal and factual bases set forth in the Sale Motion, the Declaration of Joshua Eaves and the testimony adduced at the Sale Hearing establish just cause for the relief granted herein; and after due deliberation and sufficient cause appearing therefor,

IT IS HEREBY FOUND AND DETERMINED THAT:4

- A. <u>Jurisdiction</u>. The Court has jurisdiction to hear and determine the Sale Motion and to grant the relief requested in the Sale Motion pursuant to 28 U.S.C. § 157(b)(1).
- B. <u>Venue</u>. Venue of this case and the Sale Motion in this district is proper under 28 U.S.C. § 1408. This is a core proceeding within the meaning of 28 U.S.C. § 157(b)(2).

The "Kansas, West Texas, and Other Assets" are the Oil and Gas Assets under the Asset Purchase Agreement and generally those oil and gas assets in Kansas, West Texas, and other locations that were not included as "Oil and Gas Assets" in the Stalking Horse APA, subject in all respects to the specific terms of the asset purchase agreement between Pachira and the Debtor attached hereto as Attachment 1.

Findings of fact shall be construed as conclusions of law and conclusions of law shall be construed as findings of fact to the fullest extent of the law. *See* Fed. R. Bankr. P. 7052.

- C. <u>Statutory Predicates</u>. The statutory and legal predicates for the relief requested in the Sale Motion are sections 105, 363 and 365 of title 11 of the United States Code, 11 U.S.C. §§101, <u>et seq.</u> (the "<u>Bankruptcy Code</u>"), Rules 2002, 6004, 6006, 9006 and 9014 of the Federal Rules of Bankruptcy Procedure (the "<u>Bankruptcy Rules</u>") and Bankruptcy Local Rule 2002-1.
- D. Notice. Notice of the proposed Sale of Kansas, West Texas, and Other Assets, Auction and Sale Hearing has been provided to: (i) counsel for the Official Committee of Unsecured Creditors appointed in this Chapter 11 Case; (ii) the United States Trustee for the Eastern District of Texas; (iii) all other parties known to the Debtor who have or may have asserted liens against any of the Kansas, West Texas, and Other Assets; (iv) the Debtor's twenty (20) largest unsecured creditors; (v) all parties that have requested notice pursuant to Bankruptcy Rule 2002; (vi) all parties known to have expressed an interest in a transaction with respect to all or a part of the Kansas, West Texas, and Other Assets; (vii) all parties on the Debtor's creditor matrix dated June 7, 2019; and (viii) all other parties on the Debtor's Master Service List (collectively, the "Notice Parties"), all in accordance with and as provided by the Bidding Procedures Order. Cure Notices have been provided to all counterparties to Assigned Contracts and publication notice of the Auction and the Sale was provided through the New York Times, in each case, in accordance with and as provided by the Bidding Procedures Order.
- E. <u>Notice Sufficient</u>. Based upon the affidavits of service previously filed with the Court and the evidence presented at the Sale Hearing, adequate and sufficient notice of the Sale Motion, Auction, Sale Hearing, the Sale, and the transactions contemplated thereby, including without limitation, the assumption and assignment of the Assigned Contracts to the Buyer, has been provided in accordance with the Bidding Procedures Order, sections 105(a), 363 and 365 of

the Bankruptcy Code, Bankruptcy Rules 2002, 6004, 6006 and 9006 and Local Bankruptcy Rule 2002-1. A reasonable opportunity to object and be heard with respect to the Sale, the Sale Motion and the relief requested therein, including the assumption and assignment of the Assigned Contracts to the Buyer and the amounts necessary under section 365(b) of the Bankruptcy Code to cure defaults thereunder, as such amounts have been transmitted pursuant to a written notice delivered to the applicable counterparty, has been afforded to all interested persons and entities, including the Notice Parties.

- F. <u>Kansas, West Texas, and Other Assets Property of the Estate</u>. The Kansas, West Texas, and Other Assets sought to be transferred and/or assigned by the Debtor to the Buyer pursuant to the Asset Purchase Agreement are property of the Debtor's estate and title thereto is vested in the Debtor's estate.
- G. <u>Sufficiency of Marketing</u>. The Debtor and its professionals marketed the Kansas, West Texas, and Other Assets and conducted the marketing and sale process as set forth in and in accordance with the Sale Motion and the Bidding Procedures. Based upon the record of this Chapter 11 Case, all creditors and other parties in interest and all prospective buyers have been afforded a reasonable and fair opportunity to bid for the Kansas, West Texas, and Other Assets. Under the circumstances, the Debtor and its professionals have adequately and appropriately marketed the Kansas, West Texas, and Other Assets.
- H. <u>Bidding Procedures</u>. The Bidding Procedures were substantively and procedurally fair to all parties and all potential bidders and afforded notice and a full, fair and reasonable opportunity for any person to make a higher or otherwise better offer to purchase the Kansas, West Texas, and Other Assets. The Debtor conducted the sale process (including the Auction) without collusion and in accordance with the Bidding Procedures.

- I. <u>Auction</u>. After the conclusion of the Auction held on August 21, 2019, the Debtor determined in a valid and sound exercise of its business judgment that the highest and best Qualifying Bid for the Kansas, West Texas, and Other Assets was that of the Buyer as contemplated in the Asset Purchase Agreement.
- J. <u>Corporate Authority</u>. Subject to the entry of this Sale Order, the Debtor:

 (i) has full power and authority to execute the Asset Purchase Agreement and all other documents contemplated thereby; (ii) has all of the power and authority necessary to consummate the transactions contemplated by the Asset Purchase Agreement (collectively, the "<u>Transactions</u>"), and (iii) has taken all company action necessary to authorize and approve the Asset Purchase Agreement and the sale of the Kansas, West Texas, and Other Assets, and any actions required to be performed by the Debtor in order to consummate the Transactions contemplated in the Asset Purchase Agreement. No consents or approvals, other than those expressly provided for in the Asset Purchase Agreement or this Sale Order, are required for the Debtor to consummate the Sale.
- K. Arm's-Length Sale and Buyer's Good Faith. The Asset Purchase Agreement was negotiated and is undertaken by the Debtor and the Buyer at arm's-length without collusion or fraud, and in good faith within the meaning of section 363(m) of the Bankruptcy Code. Neither the Buyer nor any of its affiliates, members, officers, directors, or shareholders is an "insider" of the Debtor as that term is defined by section 101(31) of the Bankruptcy Code. The Buyer (i) recognized that the Debtor was free to deal with any other party interested in acquiring the Kansas, West Texas, and Other Assets, (ii) complied with the Bidding Procedures Order and (iii) willingly subjected its bid to the competitive Bidding Procedures approved in the Bidding Procedures Order. All payments to be made by the Buyer and other agreements or arrangements entered into by the Buyer in connection with the Sale have been disclosed, the Buyer has not

violated section 363(n) of the Bankruptcy Code by any action or inaction, and no common identity of directors or controlling stockholders exists between the Buyer and the Debtor. As a result of the foregoing, the Buyer is a "good faith buyer" within the meaning of section 363(m) of the Bankruptcy Code, and as such, is entitled to all of the protections afforded thereby, including in the event this Sale Order or any portion thereof is reversed or modified on appeal, and otherwise has proceeded in good faith in all respects in connection with the Sale specifically and this Chapter 11 Case generally.

- L. <u>Sale Highest or Best Offer</u>. The offer submitted by the Buyer for the Kansas, West Texas, and Other Assets as reflected in the Asset Purchase Agreement, including the form and total consideration to be realized thereunder, is the highest and best offer for the Kansas, West Texas, and Other Assets. No other person or entity or group of persons or entities has offered to purchase the Kansas, West Texas, and Other Assets for an amount that would provide greater value to the Debtor than the Buyer, including through the reduction of claims against the estate. The Court's approval of the Sale Motion, the Asset Purchase Agreement, and the Transactions maximizes the Debtor's recovery for the Kansas, West Texas, and Other Assets, and, thus, is in the best interests of the Debtor and its estate, creditors and all other parties in interest.
- M. <u>No Fraudulent Transfer</u>. The Purchase Price constitutes reasonably equivalent value and fair consideration under the Bankruptcy Code, the Uniform Fraudulent Transfer Act, the Uniform Fraudulent Conveyance Act and any other applicable law, and may not be avoided under section 363(n) or any other section of the Bankruptcy Code. The Asset Purchase Agreement was not entered into, and the Sale is not being consummated, for the purpose of hindering, delaying or defrauding creditors of the Debtor under the Bankruptcy Code or under the

laws of the United States, any state, territory, possession thereof, or the District of Columbia, or any other applicable law. Neither the Debtor nor the Buyer has entered into the Asset Purchase Agreement or is consummating the Sale with any fraudulent or otherwise improper purpose.

N. <u>No Liability under Section 363(n)</u>. Neither the Debtor nor the Buyer engaged in any conduct that would cause or permit the Asset Purchase Agreement or the consummation of the Sale to be avoided, or costs or damages to be imposed, under section 363(n) of the Bankruptcy Code.

O. Transfer of Kansas, West Texas, and Other Assets Free and Clear. The Debtor is the sole and lawful owner of the Kansas, West Texas, and Other Assets. Subject to section 363(f) of the Bankruptcy Code, and except as otherwise provided in the Asset Purchase Agreement, the transfer of each of the Kansas, West Texas, and Other Assets to the Buyer will be, as of the Closing Date, a legal, valid, and effective transfer of the Kansas, West Texas, and Other Assets, which transfer vests or will vest the Buyer with all right, title, and interest of the Debtor to the Kansas, West Texas, and Other Assets free and clear of, among other things, (i) all Encumbrances, (ii) all debts arising under, relating to, or in connection with any act of the Debtor or claims (as that term is defined in section 101(5) of the Bankruptcy Code), liabilities, obligations, demands, guaranties, options, rights, contractual commitments, restrictions, interests and matters of any kind and nature, whether arising prior to or subsequent to the commencement of the Debtor's Chapter 11 Case, and whether imposed by agreement, understanding, law, equity or otherwise (including, without limitation, rights with respect to Claims (as defined below) and Encumbrances (x) that purport to give to any party a right of setoff or recoupment against, or a

⁵ As defined in the Asset Purchase Agreement to be "any charge, lien, claim, mortgage, lease, sublease, hypothecation, deed of trust, pledge, security interest, option, right of use or possession, right of first offer or first refusal, easement, servitude, restrictive covenant, encroachment, encumbrance, third party interest or other restriction or limitation of any kind."

right or option to effect any forfeiture, modification, profit sharing interest, right of first refusal, purchase or repurchase right or option, or termination of, the Debtor's or the Buyer's interests in the Kansas, West Texas, and Other Assets, or any similar rights, or (y) in respect of taxes, restrictions, rights of first refusal, charges or interests of any kind or nature, if any, including, without limitation, any restriction on use, voting, transfer, receipt of income or other exercise of any attributes of ownership) (collectively, as defined in this clause (ii), "Claims"), relating to, accruing or arising any time prior to or on the Closing Date, with the exception of Permitted Encumbrances⁶ that expressly survive the Closing pursuant to the Asset Purchase Agreement and Assumed Liabilities (collectively with subsections (i) and (ii) above, the "Interests").

P. Free and Clear Findings Required by Buyer. The Buyer would not have entered into the Asset Purchase Agreement and would not consummate the transactions contemplated thereby if the Sale of the Kansas, West Texas, and Other Assets to the Buyer were not free and clear of any and all Interests (other than Permitted Encumbrances or Assumed Liabilities) pursuant to section 363(f) of the Bankruptcy Code, or if the Buyer would, or in the future could, be liable for any of such Interests. Effective upon the Closing Date, the Buyer shall not be responsible for any Interests (other than Permitted Encumbrances or Assumed Liabilities), including in respect of the following: (i) labor or employment agreements; (ii) all mortgages, deeds of trust, and security interests; (iii) any intercompany loans and receivables between the Debtor and any non-Debtor affiliate; (iv) any welfare, compensation or other employee benefit plans, agreements, practices and programs, including, without limitation, any pension plan of the Debtor, any affiliate of any Debtor, or any member of the Debtor's "control group;" (v) any other employee, worker's compensation, occupational disease or unemployment or temporary disability

⁶ For the sake of clarity, Permitted Encumbrances includes any ad valorem tax liens for unpaid 2019 ad valorem taxes related to the Assets.

related claim, including, without limitation, claims that might otherwise arise under or pursuant to (a) the Employee Retirement Income Security Act of 1974, as amended, (b) the Fair Labor Standards Act, (c) Title VII of the Civil Rights Act of 1964, (d) the Federal Rehabilitation Act of 1973, (e) the National Labor Relations Act, (f) the Age Discrimination and Employee Act of 1967 and Age Discrimination in Employment Act, as amended, (g) the Americans with Disabilities Act of 1990, (h) the Consolidated Omnibus Budget Reconciliation Act of 1985, (i) state discrimination laws, (i) state unemployment compensation laws or any other similar state laws, (k) the Worker Adjustment and Retraining Notification Act, 29 U.S.C §§ 2101 et. seq., or (1) any other state or federal benefits or claims relating to any employment with the Debtors or any of their predecessors; (vi) Interests arising under any environmental, health and safety laws with respect to any assets owned or operated by Debtor or any corporate predecessor at any time prior to the Closing Date and any liabilities of the Debtor other than the Assumed Liabilities; (vii) any bulk sales or similar law; (viii) any tax statutes or ordinances, including, without limitation, the Internal Revenue Code of 1986, as amended; (ix) any and all Interests arising out of violations, or other non-compliance with any law(s), regulation(s), standard(s), guideline(s), enforcement order(s), or any other authority or requirement enforced by, or under the supervision of the Occupational Safety and Health Administration; (x) any theories of successor liability or causes of action related thereto; (xi) any unpaid pre-Effective Time royalties, non-operating working interest disbursements, joint interest billing obligations, or similar interests relating to any oil and gas leases included in the Kansas, West Texas, and Other Assets; and (xii) any rights or claims based on Excluded Contracts or Excluded Liabilities. A sale of the Kansas, West Texas, and Other Assets other than one free and clear of all Interests would yield substantially less value for the Debtor's estate, with less certainty, than the Sale as contemplated. Therefore, the Sale contemplated by the Asset Purchase

Agreement maximizes the Debtor's recovery on the Kansas, West Texas, and Other Assets, and, thus, is in the best interests of the Debtor and its estate, creditors and all other parties in interest.

- Q. <u>Satisfaction of Section 363(f) Standards</u>. The Debtor may sell the Kansas, West Texas, and Other Assets free and clear of all Interests (other than Permitted Encumbrances and Assumed Liabilities) because, with respect to each creditor or other person or entity asserting an Interest, one or more of the standards set forth in section 363(f)(l)-(5) of the Bankruptcy Code has been satisfied. Those holders of Interests who did not object (or who ultimately withdrew their objections, if any) to the Sale or the Sale Motion are deemed to have consented to the Sale Motion and Sale pursuant to section 363(f)(2) of the Bankruptcy Code. Those holders of Interests who did object fall within one or more of the other subsections of section 363(f) of the Bankruptcy Code.
- R. No Successor Liability. The Buyer is not holding itself out to the public as a continuation of the Debtor and is not an "insider" or "affiliate" of the Debtor, as those terms are defined in the Bankruptcy Code, and no common identity of incorporators, directors or stockholders exists now or has ever existed between the Buyer and the Debtor. The conveyance of the Kansas, West Texas, and Other Assets does not amount to a consolidation, merger or *de facto* merger of the Buyer and the Debtor and/or Debtor's estate, there is not substantial continuity between the Buyer and the Debtor, there is no continuity of enterprise between the Debtor and the Buyer, the Buyer is not a mere continuation of the Debtor or its estate, and the Buyer does not constitute a successor to the Debtor or its estate. To the fullest extent of the law, the Buyer's acquisition of the Kansas, West Texas, and Other Assets shall be free and clear of any "successor liability" claims of any nature whatsoever, whether known or unknown and whether asserted or unasserted as of the Closing. The Buyer's operations shall not be deemed a continuation of the

Debtor's business as a result of the acquisition of the Kansas, West Texas, and Other Assets. The Buyer would not have acquired the Kansas, West Texas, and Other Assets but for the foregoing protections against potential claims based upon "successor liability" theories.

- S. <u>Pre-Effective Time Transactions</u>. The liabilities of the Debtor, if any, for all royalty, working interest, overriding royalty interest, joint interest billing obligations, and other vendor obligations related to the Kansas, West Texas, and Other Assets and all other operating expenses accruing after the Petition Date but prior to the Effective Time shall not be affected by this order.
- T. <u>Post-Effective Time Transactions</u>. Subject to the terms of the Asset Purchase Agreement, the Buyer shall be responsible for all royalty, working interest, overriding royalty interest, joint interest billing obligations, and other vendor obligations related to the Kansas, West Texas, and Other Assets and all other operating expenses accruing from the Effective Time forward.
- U. <u>Assigned Contracts</u>. Each and every provision of the Assigned Contracts or applicable non-bankruptcy law that purports to prohibit, restrict, or condition, or could be construed as prohibiting, restricting, or conditioning assignment of any Assigned Contract has been satisfied or is otherwise unenforceable under section 365 of the Bankruptcy Code. There shall be no rent accelerations, assignment fees, increases or any other fees charged to the Debtor or the Buyer as a result of the assumption and/or assignment of the Assigned Contracts. All counterparties of the Assigned Contracts who did not or do not timely file an objection to the assumption and assignment of the Assigned Contract(s) to which they are counterparty are deemed to consent to the assumption by the Debtor of their respective Assigned Contract(s) and the assignment thereof to the Buyer, and the Buyer shall enjoy all of the rights and benefits under each

such Assigned Contract as of the applicable date of assumption and assignment without the necessity of obtaining such non-debtor party's written consent to the assumption or assignment thereof. Upon the assignment and sale to the Buyer, the Assigned Contracts shall be deemed valid and binding, in full force and effect in accordance with their terms, subject to the provisions of this Sale Order, and shall be assigned and transferred to the Buyer, notwithstanding any provision in the Assigned Contracts prohibiting or otherwise restricting assignment or transfer. The Debtor has demonstrated that it is an exercise of its sound business judgment to assume and assign the Assigned Contracts to the Buyer in connection with the consummation of the Sale of the Kansas, West Texas, and Other Assets, and the assumption and assignment of the Assigned Contracts is in the best interests of the Debtor, its estate and creditors, and other parties in interest. The Assigned Contracts being assigned to the Buyer are an integral part of the Sale of the Kansas, West Texas, and Other Assets and, accordingly, their assumption and assignment are reasonable and an enhancement to the value of the Debtor's estate.

V. <u>Cure/Adequate Assurance</u>. Pursuant to the Asset Purchase Agreement, the Cure Costs will be paid by Buyer as "Assumed Liabilities" under the Asset Purchase Agreement. The Buyer has demonstrated adequate assurance of future performance of all Assigned Contracts within the meaning of section 365 of the Bankruptcy Code, including its promise to perform the Debtor's obligations under the Assigned Contracts for periods on and after the Closing. The Cure Costs are deemed the amounts necessary to "cure" all "defaults" (each, within the meaning of section 365(b) of the Bankruptcy Code) under such Assigned Contracts. Any objections to the Cure Costs, to the extent not otherwise resolved, are hereby overruled. To the extent that any counterparty failed to timely object to its Cure Cost or the assignment of its Assigned Contract or to raise any other alleged default or breach of contract, such counterparty is deemed to have

consented to such Cure Cost and to the assignment of its respective Assigned Contract(s) to the Buyer and to have waived any other defaults or breaches. The payment of the Cure Costs as provided in the Asset Purchase Agreement is appropriate and is deemed to fully satisfy the Debtor's obligations under sections 365(b) and 365(f) of the Bankruptcy Code. Accordingly, all of the requirements of sections 365(b) and 365(f) of the Bankruptcy Code have been satisfied for the assumption by the Debtor, and the assignment by the Debtor to the Buyer, of each of the Assigned Contracts.

- W. <u>Kansas, West Texas, and Other Assets Assignable</u>. Any contractual provision or applicable non-bankruptcy law that purports to prohibit, restrict, or condition assignment of any of the Kansas, West Texas, and Other Assets has been satisfied or is otherwise unenforceable under section 365 of the Bankruptcy Code.
- X. Sale as Exercise of Business Judgment. Entry into and consummation of the Asset Purchase Agreement constitute the exercise by the Debtor of sound business judgment, and such acts are in the best interests of the Debtor, its estate and creditors, and all parties in interest. The Court finds that the Debtor has articulated good and sufficient business reasons justifying the Sale of the Kansas, West Texas, and Other Assets to the Buyer. Additionally: (i) the Asset Purchase Agreement constitutes the highest and best offer for the Kansas, West Texas, and Other Assets; (ii) the Asset Purchase Agreement and the closing thereon presents the best opportunity to realize the maximum value of the Kansas, West Texas, and Other Assets and avoid a decline and devaluation of the Kansas, West Texas, and Other Assets; (iii) there is risk of deterioration of the value of the Kansas, West Texas, and Other Assets if the Sale is not consummated promptly; and (iv) the Asset Purchase Agreement and the closing thereon will provide a greater recovery for the Debtor's creditors than would be provided by any other presently

available alternative. The Debtor has demonstrated compelling circumstances and a good, sufficient and sound business purpose and justification for the Sale prior to, and outside of, a plan of reorganization.

- Y. <u>Compelling Reasons for an Immediate Sale</u>. Good and sufficient reasons for approval of the Asset Purchase Agreement have been articulated by the Debtor. The Debtor has demonstrated compelling circumstances for the Sale outside: (a) the ordinary course of business, pursuant to section 363(b) of the Bankruptcy Code; and (b) a plan of reorganization, in that, among other things, the immediate consummation of the Sale to the Buyer is necessary and appropriate to preserve and to maximize the value of the Debtor's estate. To maximize the value of the Kansas, West Texas, and Other Assets and to preserve the viability of the businesses to which the Kansas, West Texas, and Other Assets relate, it is essential that the Sale occur promptly. Time is of the essence in consummating the Sale.
- Z. No Sub Rosa Plan. The Sale does not constitute a *sub rosa* chapter 11 plan. The Sale neither impermissibly restructures the rights of the Debtors' creditors nor impermissibly dictates the terms of a plan of reorganization for the Debtors.
- AA. <u>Final Order</u>. This Sale Order constitutes a final order within the meaning of 28 U.S.C. § 158(a). Notwithstanding Bankruptcy Rules 6004(h) and 6006(d), the Court expressly finds that there is no just reason for delay in the implementation of this Sale Order and expressly directs entry of judgment as set forth herein.

IT IS THEREFORE ORDERED, ADJUDGED AND DECREED THAT:

1. <u>Sale Motion Granted.</u> The relief requested in the Sale Motion is GRANTED and the Sale of the Kansas, West Texas, and Other Assets to Buyer is approved, all as set forth in this Sale Order.

- 2. <u>Objections Overruled.</u> All objections, if any, with regard to the relief sought in the Sale Motion that have not been withdrawn, waived, settled or otherwise dealt with as expressly provided herein or on the record at the Sale Hearing are hereby overruled on the merits, with prejudice.
- 3. Approval. Pursuant to sections 105, 363 and 365 of the Bankruptcy Code, the Asset Purchase Agreement, the assumption and assignment of the Assigned Contracts to the Buyer as of the Closing Date and the Sale of the Kansas, West Texas, and Other Assets and the other transactions contemplated thereby are hereby approved in all respects. Pursuant to sections 105, 363 and 365 of the Bankruptcy Code, the Debtor and its officers, agents, professionals and other representatives are authorized and directed to perform their obligations under, and comply with the terms of, the Asset Purchase Agreement, and to consummate, the Sale, including the sale, transfer and assignment of all of the Debtor's right, title and interest in the Kansas, West Texas, and Other Assets to the Buyer free and clear of any and all Interests (other than Permitted Encumbrances and the Assumed Liabilities) in accordance with the terms of the Asset Purchase Agreement and this Sale Order. Pursuant to sections 105, 363 and 365 of the Bankruptcy Code, the Debtor, the Buyer and each of their respective officers, directors, employees, professionals, agents and other representatives are each hereby authorized and directed to take any and all actions necessary or appropriate to: (a) consummate the Sale of the Kansas, West Texas, and Other Assets to the Buyer and the Closing of the Sale, pursuant to the Asset Purchase Agreement and this Sale Order, (b) assume and assign the Assigned Contracts to be assumed and assigned to the Buyer as of the Closing Date, and (c) perform, consummate, implement and close fully the Asset Purchase Agreement together with any and all additional instruments and documents that may be reasonably necessary or desirable to implement the Asset Purchase Agreement. The Debtor is hereby

authorized and directed to perform its respective covenants and undertakings as provided in the Asset Purchase Agreement prior to or after the Closing of the Sale without further order of the Court. The Buyer and the Debtor shall have no obligation to close the Sale except as is contemplated and provided for in the Asset Purchase Agreement.

4. Transfer Free and Clear. Pursuant to sections 105, 363 and 365 of the Bankruptcy Code, upon the Closing, neither the Buyer nor its respective successors and assigns shall have any liability for any Interest, except for Permitted Encumbrances and Assumed Liabilities, whether known or unknown as of the Closing Date, now existing or hereafter arising, whether fixed or contingent, whether as a successor, vicariously or otherwise, of any kind, nature or character whatsoever, including for any Interests arising under, without limitation: (i) any labor or employment agreements; (ii) all mortgages, deeds of trust and security interests; (iii) any intercompany loans and receivables between the Debtor and any non-Debtor affiliate; (iv) any welfare, compensation or other employee benefit plans, agreements, practices and programs, including, without limitation, any pension plan of any affiliate of any Debtor, or any member of the Debtor's "control group;" (v) any other employee, worker's compensation, occupational disease or unemployment or temporary disability related claim, including, without limitation, claims that might otherwise arise under or pursuant to (a) the Employee Retirement Income Security Act of 1974, as amended, (b) the Fair Labor Standards Act, (c) Title VII of the Civil Rights Act of 1964, (d) the Federal Rehabilitation Act of 1973, (e) the National Labor Relations Act, (f) the Age Discrimination and Employee Act of 1967 and Age Discrimination in Employment Act, as amended, (g) the Americans with Disabilities Act of 1990, (h) the Consolidated Omnibus Budget Reconciliation Act of 1985, (i) state discrimination laws, (j) state unemployment compensation laws or any other similar state laws, (k) the Worker Adjustment and

Retraining Notification Act, 29 U.S.C §§ 2101 et. seq., or (1) any other state or federal benefits or claims relating to any employment with the Debtors or any of their predecessors; (vi) Interests arising under any environmental, health and safety laws with respect to any assets owned or operated by Debtor or any corporate predecessor at any time prior to the Closing Date and any liabilities of the Debtor other than the Assumed Liabilities; (vii) any bulk sales or similar law; (viii) any tax statutes or ordinances, including, without limitation, the Internal Revenue Code of 1986, as amended; (ix) any and all Interests arising out of violations, or other non-compliance with any law(s), regulation(s), standard(s), guideline(s), enforcement order(s), or any other authority or requirement enforced by, or under the supervision of the Occupational Safety and Health Administration; (x) any theories of successor liability or causes of action related thereto; (xi) any unpaid pre-Effective Time royalties, non-operating working interest disbursements, joint interest billing obligations, or similar interests relating to any oil and gas leases included in the Kansas, West Texas, and Other Assets; and (xii) any rights or claims based on Excluded Contracts or Excluded Liabilities. Any Interests shall attach to the cash proceeds of the Transactions ultimately attributable to the Kansas, West Texas, and Other Assets in which such creditor alleges an Interest with the same validity, extent and priority as immediately prior to the Transactions, subject to any rights, claims and defenses that the Debtor and other parties in interest may possess with respect thereto.

5. <u>Preferential Purchase Rights</u>. Other than contractual rights of first refusal that have been waived by agreement and stipulation, any contractual rights of first refusal or similar contractual purchasing rights regarding the Debtor's Kansas, West Texas, and Other Assets are found to be impermissible restrictions on assignment pursuant to section 365(f) of the Bankruptcy Code and are therefore unenforceable in the context of the Debtor's proposed sale of the Kansas,

West Texas, and Other Assets. Nonetheless, assumption and assignment of any executory contract or unexpired lease by the Debtor to the Buyer shall not be free and clear of any contractual rights of first refusal or similar contractual purchasing rights regarding the Debtor's Kansas, West Texas, and Other Assets, and such rights shall apply to any subsequent transfer or sale as provided under applicable non-bankruptcy law.

- 6. <u>Surrender of Possession</u>. Any and all Kansas, West Texas, and Other Assets in the possession or control of any person or entity, including any vendor, supplier or employee of the Debtor, shall be transferred to the Buyer free and clear of all Interests, except for Permitted Encumbrances and Assumed Liabilities, and shall be delivered to the Buyer and deemed delivered at the time of Closing (or such other time as provided in the Asset Purchase Agreement).
- 7. <u>Valid Transfer</u>. Effective upon the Closing, the transfer to the Buyer of the Debtor's right, title and interest in the Kansas, West Texas, and Other Assets pursuant to the Asset Purchase Agreement shall be, and hereby is deemed to be, a legal, valid and effective transfer of the Debtor's right, title and interest in the Kansas, West Texas, and Other Assets, and vests with or will vest in the Buyer all right, title and interest of the Debtor in the Kansas, West Texas, and Other Assets, free and clear of all Interests (other than Permitted Encumbrances or Assumed Liabilities).
- 8. Exculpation and Release of Buyer. None of the Buyer or any of its members or their respective affiliates, successors, assigns and advisors shall have or incur any liability to, or be subject to any action by the Debtor, its estate or any of its predecessors, successors or assigns, arising out of the negotiation, due diligence in respect of, preparation, execution or delivery of the Asset Purchase Agreement, and the entry into and consummation of the Sale, other than (with respect to the Buyer only) the Assumed Liabilities being assumed by the Buyer.

- 9. Exculpation and Release of Debtor. Except for the enumerated rights and obligations of the Debtor and the Buyer under the Asset Purchase Agreement, the Buyer, to the extent permitted by law, is hereby deemed to have irrevocably and unconditionally released, remised, and forever discharged the Debtor and its affiliates, and all their past, present and future shareholders, partners, members, board of directors and/or supervisors, managers, officers, employees, agents, representatives and advisors from any and all suits, legal or administrative proceedings, claims, demands, damages, losses, costs, liabilities, interest or causes of action whatsoever at law or in equity, known or unknown, which the Buyer or its affiliates might now or subsequently may have, based on, relating to or arising out of the Asset Purchase Agreement, the transactions contemplated thereby, the ownership, use or operation of the Kansas, West Texas, and Other Assets or the condition, quality, status or nature of the Kansas, West Texas, and Other Assets, including rights to contribution under the Comprehensive Environmental Response, Compensation, and Liability Act of 1980, as amended, breaches of statutory or implied warranties, nuisance or other tort actions, rights to punitive damages, common law rights of contribution and rights under insurance maintained by the Debtor or any of its affiliates.
- 10. <u>Injunction</u>. Except as expressly provided in the Asset Purchase Agreement or this Sale Order, effective upon the Closing all persons and entities, including, but not limited to, all debt security holders, equity security holders, holders of preferential purchase rights, governmental, tax and regulatory authorities, lenders, vendors, suppliers, employees, trade creditors, litigation claimants and other persons holding Interests (other than Permitted Encumbrances or Assumed Liabilities) against or in the Debtor or the Debtor's interests in the Kansas, West Texas, and Other Assets (whether known or unknown, legal or equitable, matured or unmatured, contingent or noncontingent, liquidated or unliquidated, asserted or unasserted,

whether arising prior to or subsequent to the commencement of this Chapter 11 Case, whether imposed by agreement, understanding, law, equity or otherwise), shall be and hereby are forever barred, estopped and permanently enjoined from asserting, prosecuting or otherwise pursuing such Interests against the Buyer or any of its members or their respective affiliates, agents, advisors, representatives, officers, successors and assigns, the Kansas, West Texas, and Other Assets, or the interests of the Debtor or the Buyer in such Kansas, West Texas, and Other Assets, including, without limitation, taking any of the following actions with respect to an Interest (other than, with respect to the Buyer only, the Permitted Encumbrances and Assumed Liabilities): (a) commencing or continuing in any manner any action or other proceeding against such parties or Kansas, West Texas, and Other Assets; (b) enforcing, attaching, collecting or recovering in any manner any judgment, award, decree or order against such parties or Kansas, West Texas, and Other Assets; (c) creating, perfecting or enforcing any liens, claims, encumbrances or other interests against such parties or Kansas, West Texas, and Other Assets; (d) asserting a Claim as a setoff, right of subrogation or recoupment of any kind against any obligation due the Buyer or its affiliates, agents, advisors, representatives, officers, successors or assigns; or (e) commencing or continuing any action in any manner or place that does not comply, or is inconsistent, with the provisions of this Sale Order or the agreements or actions contemplated or taken in respect thereof. All persons are hereby enjoined from taking any action that would interfere with or adversely affect the Debtor's ability to transfer the Kansas, West Texas, and Other Assets in accordance with the terms of the Asset Purchase Agreement and this Sale Order. Following the Closing, no holder of an Interest (including as such term is used in section 363(f)) against the Debtor shall interfere with the Buyer's title to or use and enjoyment of the Kansas, West Texas, and Other Assets.

- Buyer in good faith and the Buyer is a good faith Buyer of the Kansas, West Texas, and Other Assets as that term is used in section 363(m) of the Bankruptcy Code, and accordingly, the reversal or modification on appeal of the authorization of the Sale provided herein shall neither affect the validity of this Sale nor the transfer of the Kansas, West Texas, and Other Assets to Buyer, free and clear of Interests, unless such authorization is duly stayed before the Closing pending such appeal. The Buyer is entitled to all of the protections afforded by section 363(m) of the Bankruptcy Code, and the Buyer otherwise has proceeded in good faith in all respects in connection with the Sale specifically and this Chapter 11 Case generally.
- 12. <u>No Bulk Sales</u>. No bulk sales law or any similar law of any state or other jurisdiction shall apply in any way to the transactions contemplated by the Asset Purchase Agreement, the Sale Motion and this Sale Order.
- 13. Fair and Equivalent Value. The consideration provided by the Buyer for the Kansas, West Texas, and Other Assets under the Asset Purchase Agreement shall be deemed for all purposes to constitute reasonably equivalent value and fair consideration under the Bankruptcy Code and any other applicable law, and the Sale may not be avoided, or costs or damages imposed or awarded under section 363(n) or any other provision of the Bankruptcy Code, the Uniform Fraudulent Transfer Act, the Uniform Fraudulent Conveyance Act or any other similar state laws.
- 14. <u>Transfer of Marketable Title</u>. Upon the Closing, this Sale Order shall be construed and shall constitute for any and all purposes a full and complete general assignment, conveyance and transfer of all of the Debtor's right, title and interest in the Kansas, West Texas, and Other Assets and/or a bill of sale transferring good and marketable title in such Kansas, West Texas, and

Other Assets to the Buyer at the Closing pursuant to the terms of the Asset Purchase Agreement, free and clear of all Interests (other than Permitted Encumbrances or Assumed Liabilities).

- 15. No Successor Liability. The consummation of the Sale does not amount to a consolidation, merger or de facto merger of the Buyer and the Debtor and/or its estate, there is not substantial continuity between the Buyer and the Debtor, there is no continuity of enterprise between the Debtor and the Buyer, the Buyer is not a mere continuation of the Debtor or its estate, and the Buyer does not constitute a successor to the Debtor or its estate. Upon the Closing, to the fullest extent of the law, the Buyer's acquisition of the Kansas, West Texas, and Other Assets shall be free and clear of any "successor liability" claims of any nature whatsoever, whether known or unknown and whether asserted or unasserted as of the time of the Closing. The Kansas, West Texas, and Other Assets shall not be subject to any Interests arising under or in connection with any Excluded Contract, Excluded Asset, or Excluded Liability. The Buyer's operations shall not be deemed a continuation of the Debtor's business as a result of the acquisition of the Kansas, West Texas, and Other Assets. The transfer of the Kansas, West Texas, and Other Assets to the Buyer, except as otherwise set forth in the Asset Purchase Agreement, does not, and will not, subject the Buyer to any liability whatsoever, with respect to operation of the Debtor's business prior to the Closing.
- 16. <u>Authorization to Assign</u>. Pursuant to section 365(f) of the Bankruptcy Code, and the Asset Purchase Agreement, and notwithstanding any provision of any contract governing the Kansas, West Texas, and Other Assets or any Assigned Contract to be assumed and assigned to the Buyer or applicable non-bankruptcy law that prohibits, restricts, or conditions the assignment of the Kansas, West Texas, and Other Assets or the Assigned Contracts, the Debtor is authorized to (a) assign, sell and transfer the Kansas, West Texas, and Other Assets to the Buyer and

- (b) assume and assign the Assigned Contracts to the Buyer, which assignments shall take place on and be effective as of the Closing Date or as otherwise provided by a separate order of this Court.
- a. There shall be no accelerations, assignment fees, increases, or any other fees charged to the Buyer or the Debtor as a result of the assumption and assignment of the Kansas, West Texas, and Other Assets and the Assigned Contracts.
- b. The Debtor has met all of the requirements of section 365(b) of the Bankruptcy Code for each of the Assigned Contracts to be assumed and assigned to the Buyer as of Closing. Notwithstanding the foregoing, unless required by the Buyer under the Asset Purchase Agreement, no Debtor shall be required by the Court to assume and assign any Assigned Contracts, and, if no such assumption and assignment occurs, no Cure Costs shall be due and no adequate assurance of future performance shall be required.
- c. The Debtor's assumption of the Assigned Contracts is subject to the consummation of the Sale of the Kansas, West Texas, and Other Assets to the Buyer. To the extent that an objection by a counterparty to any Assigned Contract, including an objection related to the applicable Cure Cost, is not resolved prior to the Closing, the Buyer may, without any further approval of the Court or notice to any party, elect to (i) not have the Debtor assume and assign such Assigned Contract to it or (ii) have the Debtor postpone the assumption of such Assigned Contract in accordance with section 2.5 of the Asset Purchase Agreement until the resolution of such objection; provided that the Debtor, the Buyer, and the relevant non-debtor counterparty under each Assigned Contract shall have authority to compromise, settle or otherwise resolve any objections to proposed Cure Costs without further order of the Bankruptcy Court, with any such agreed upon Cure Costs being paid to the appropriate counterparty as a condition subsequent to such assumption and assignment of the relevant Assigned Contract.

- d. The Debtor is authorized to execute and deliver to the Buyer such agreements, documents and other instruments as may facilitate or document the sale, assignment, transfer, conveyance and delivery of the Assigned Contracts to the Buyer.
- 17. Assigned Contracts. As of the Closing, subject to the provisions of this Sale Order, the Buyer shall succeed to the entirety of the Debtor's rights and obligations in the Assigned Contracts to be assumed and assigned to the Buyer first arising and attributable to the time period occurring on or after the Closing and shall have all rights thereunder. Each non-Debtor party to an Assigned Contract is hereby forever barred, estopped, and permanently enjoined from raising or asserting against the Debtor or the Buyer, or the property of either of them, any assignment fee, default, breach or claim of pecuniary loss, or condition to assignment, arising under or related to the Assigned Contracts, existing as of the date of the Sale Hearing, or arising by reason of the consummation of transactions contemplated by the Asset Purchase Agreement, including the Sale and the assumption and assignment of the Assigned Contracts. Any party that may have had the right to consent to the assignment of an Assigned Contract is deemed to have consented to such assignment for purposes of section 365(e)(2)(A)(ii) of the Bankruptcy Code if such party failed to object to the assumption and assignment of such Assigned Contract.
- a. Upon Closing, (i) all defaults (monetary and non-monetary) under the Assigned Contracts shall be deemed cured and satisfied in full through the payment of the Cure Costs, (ii) no other amounts will be owed by the Debtor, its estate or the Buyer with respect to amounts first arising or accruing during, or attributable or related to, the period before Closing with respect to the Assigned Contracts and (iii) any and all persons or entities shall be forever barred and estopped from asserting a claim against the Debtor, its estate, the Buyer, or the Kansas, West Texas, and Other Assets, that any additional amounts are due or that any defaults exist under

the Assigned Contracts that arose or accrued, or relate to or are attributable to the period before the Closing, whether declared or undeclared, or known or unknown. The Buyer's promise pursuant to the terms of the Asset Purchase Agreement to pay the Cure Costs and to perform the Debtor's obligations under the Assigned Contracts for the period on or after the Closing shall constitute adequate assurance of its future performance under the Assigned Contracts being assigned to it as of the Closing within the meaning of sections 365(b)(l)(C) and (f)(2)(B) of the Bankruptcy Code.

- b. Upon assumption of the Assigned Contracts, such Assigned Contracts shall be deemed valid and binding, in full force and effect in accordance with their terms, subject to the provisions of this Sale Order, and shall be assigned and transferred to the Buyer notwithstanding any provision in such Assigned Contracts or other restrictions prohibiting assignment or transfer. The assumption and assignment of the Assigned Contracts as authorized under this Sale Order will not take effect until the Closing. Furthermore, other than Assigned Contracts, no other contract shall be deemed assumed by the Debtor and assigned to the Buyer pursuant to section 365 of the Bankruptcy Code. The failure of the Debtors or the Buyer to enforce at any time one or more terms or conditions of any Assigned Contract shall not be a waiver of such terms or conditions, or of the Debtors' and the Buyer's rights to enforce every term and condition of such Assigned Contract.
- c. All counterparties to the Assigned Contracts to be assumed and assigned to Buyer as of the Closing shall cooperate and expeditiously execute and deliver, upon the reasonable request of the Buyer, and shall not charge the Debtor or the Buyer for, any instruments, applications, consents, or other documents which may be required or requested by any public or

quasi-public authority or other party or entity to effectuate the applicable transfers in connection with the Transactions.

- d. Notwithstanding the foregoing, in accordance with and pursuant to the terms and conditions of Section 2.5 of the Asset Purchase Agreement, the Debtor may, at the Buyer's direction, amend the list of Assigned Contracts and designate any Assigned Contract as an Excluded Contract without further Court order. If the list of Assigned Contracts attached to the Asset Purchase Agreement at Disclosure Schedule 5.4 is amended from the version attached to this Sale Order, then the Debtor shall file a final notice listing all Assigned Contracts as soon as practicable following such amendment; provided that the Designation Deadline for any Executory Contract with respect to which a dispute regarding a Cure Amount exists on such date shall be five (5) days after the resolution of such dispute.
- 18. Ad Valorem Taxes. For the avoidance of doubt, the ad valorem tax liens of Atlanta ISD, Harris County, Robertson County, and Shelby County for the 2019 tax year are hereby expressly retained against the Kansas, West Texas, and Other Assets until payment is made to fully satisfy the 2019 ad valorem taxes, and any penalties or interest which may ultimately accrue to those 2019 taxes if not timely paid.
- Agreement releases, nullifies, precludes, or enjoins the enforcement of any police or regulatory liability to a governmental unit that any entity would be subject to as the owner or operator of property after the date of entry of this Sale Order. Nothing in this Sale Order or the Asset Purchase Agreement authorizes the transfer or assignment of any governmental (a) license, (b) permit, (c) registration, (d) authorization or (e) approval, or the discontinuation of any obligation thereunder, without compliance with all applicable legal requirements and approvals under police or regulatory

law. Nothing in this Sale Order divests any tribunal of any jurisdiction it may have under police or regulatory law to interpret this Sale Order or to adjudicate any defense asserted under this Sale Order.

- 20. Release of Interests. Effective upon the Closing, this Sale Order: (a) is and shall be effective as a determination that all Interests (other than Permitted Encumbrances or Assumed Liabilities) of any kind or nature whatsoever existing as to the Kansas, West Texas, and Other Assets prior to the Closing have been unconditionally released, discharged and terminated, and that the conveyances described herein have been effected; (b) is and shall be binding upon and shall govern the acts of all entities, including all filing agents, filing officers, title agents, title companies, recorders of mortgages, recorders of deeds, registrars of deeds, administrative agencies or units, governmental departments or units, secretaries of state, federal, state and local officials and all other persons and entities who may be required by operation of law, the duties of their office, or contract, to accept, file, register or otherwise record or release any documents or instruments, or who may be required to report or insure any title or state of title in or to the Kansas, West Texas, and Other Assets conveyed to the Buyer, and all recorded Interests (other than Permitted Encumbrances or Assumed Liabilities) against the Kansas, West Texas, and Other Assets shall be deemed stricken from such entities' records, official and otherwise.
- 21. Approval to Release Interests. If any person or entity that has filed financing statements, mortgages, mechanic's liens or other documents or agreements evidencing Interests in, Liens on, or Claims against the Kansas, West Texas, and Other Assets shall not have delivered to the Debtor before the Closing, in proper form for filing and executed by the appropriate parties, the appropriate documentation with respect to the release of such Interests, the Debtor and the Buyer are hereby authorized to execute and file such statements, instruments, releases and other

documents on behalf of such person or entity with respect to the Kansas, West Texas, and Other Assets. The Buyer is hereby authorized to file, register or otherwise record a certified copy of this Sale Order, which, once filed, registered or otherwise recorded, shall constitute conclusive evidence of the release of all Interests against the Kansas, West Texas, and Other Assets other than the Permitted Encumbrances and the Assumed Liabilities. This Sale Order is deemed to be in recordable form sufficient to be placed in the filing or recording system of each and every federal, state or local government agency, department or office.

- 22. <u>Allocation of Kansas, West Texas, and Other Assets.</u> The Buyer is hereby authorized in connection with the consummation of the Sale to allocate the Kansas, West Texas, and Other Assets, including the Assigned Contracts, among its affiliates, designees, assigns, and/or successors, in a manner as it in its sole discretion deems appropriate, and to assign, lease, sublease, license, sublicense, transfer, or otherwise dispose of any of the Kansas, West Texas, and Other Assets, including the Assigned Contracts, to its affiliates, designees, assignees and/or successors with all of the rights and protections accorded to the Buyer under this Sale Order and the Asset Purchase Agreement with respect thereto. The Debtor shall cooperate with and take all actions reasonably requested by the Buyer to effectuate any of the foregoing.
- 23. Governmental Authorization to Effectuate Sale and Assignments. Each and every federal, state and governmental agency or department, and any other person or entity, is hereby authorized to accept any and all documents and instruments in connection with or necessary to consummate the transactions contemplated by the Asset Purchase Agreement. No governmental unit may revoke or suspend any lawful right, license, trademark or other permission relating to the use of the Kansas, West Texas, and Other Assets sold, transferred or conveyed to the Buyer on account of the filing or pendency of this Chapter 11 Case or the consummation of the Transactions.

For the avoidance of doubt, the Sale of the Kansas, West Texas, and Other Assets authorized herein shall be of full force and effect, regardless of whether the Debtor or any of their affiliates lack good standing in any jurisdiction in which such entity is formed or is authorized to transact business.

- 24. <u>Inconsistencies with Prior Orders, Pleadings or Agreements</u>. To the extent this Sale Order is inconsistent with any prior order or pleading with respect to the Motion in this Chapter 11 Case, the terms of this Sale Order shall govern. To the extent there is any inconsistency between the terms of this Sale Order and the terms of the Asset Purchase Agreement (including all ancillary documents executed in connection therewith), the terms of this Sale Order shall govern.
- 25. Binding Effect of Sale Order; Subsequent Orders. This Sale Order and the Asset Purchase Agreement shall be binding in all respects upon the Debtor, its estate, all creditors of, and holders of equity interests in, the Debtor, any holders of Interests in, against or on all or any portion of the Kansas, West Texas, and Other Assets (whether known or unknown), the Buyer and all successors and assigns of the Buyer, all successors and assigns to the Kansas, West Texas, and Other Assets and any trustee, examiner, "responsible person" or other fiduciary appointed in the Chapter 11 Cases or upon a conversion to chapter 7 under the Bankruptcy Code. Any such trustee, examiner, "responsible person" or other fiduciary appointed in the Chapter 11 Case or upon a conversion to chapter 7 under the Bankruptcy Code shall have full authority to take any action or execute any document on behalf of the Debtor as may be necessary to ensure the full performance of the Debtor's obligations under Section 2.7 of the Asset Purchase Agreement. The Asset Purchase Agreement shall not be subject to rejection or avoidance under any circumstances. If any order under section 1112 of the Bankruptcy Code is entered, such order shall provide (in accordance with sections 105 and 349 of the Bankruptcy Code) that this Sale Order, including the

rights granted to the Buyer hereunder, shall remain effective and, notwithstanding such conversion or dismissal, shall remain binding on parties in interest. This Sale Order shall not be modified by any chapter 11 plan confirmed in this Chapter 11 Case or any subsequent order(s) of this Court.

- 26. <u>Failure to Specify Provisions</u>. The failure specifically to include or make reference to any particular provisions of the Asset Purchase Agreement or any related ancillary document in this Sale Order shall not diminish or impair the effectiveness of such provision, it being the intent of the Court that the Asset Purchase Agreement and all related ancillary documents are authorized and approved in their entirety.
- 27. Retention of Jurisdiction. The Court retains jurisdiction with respect to all matters arising from or related to the implementation of this Sale Order, including, without limitation, the authority to: (i) interpret, implement and enforce the terms and provisions of this Sale Order (including the exculpation, release and injunctive provisions in this Sale Order) and the terms of the Asset Purchase Agreement, all amendments thereto and any waivers and consents thereunder; (ii) protect the Buyer, or the Kansas, West Texas, and Other Assets, from and against any Interests (other than Permitted Encumbrances or Assumed Liabilities); (iii) compel delivery of all Kansas, West Texas, and Other Assets to the Buyer; (iv) compel the Debtor and the Buyer to perform all of their respective obligations under the Asset Purchase Agreement; and (v) resolve any disputes arising under or related to the Asset Purchase Agreement or the Sale.
- Automatic Stay. The Buyer shall not be required to seek or obtain relief from the automatic stay under section 362 of the Bankruptcy Code to enforce any of its remedies under the Asset Purchase Agreement or any other sale-related document. The automatic stay imposed by section 362 of the Bankruptcy Code is modified solely to the extent necessary to implement the provisions of this Sale Order.

- 29. <u>No Material Modifications</u>. The Asset Purchase Agreement and any related agreements, documents or other instruments may be modified, amended, or supplemented through a written document signed by the parties thereto in accordance with the terms thereof without further order of the Court; <u>provided</u>, <u>however</u>, that any such modification, amendment or supplement is neither material nor materially changes the economic substance of the transactions contemplated hereby.
- 30. <u>Immediate Effect</u>. This Sale Order constitutes a final order within the meaning of 28 U.S.C. § 158(a). Notwithstanding any provision in the Bankruptcy Rules to the contrary, the Court expressly finds there is no reason for delay in the implementation of this Sale Order and, accordingly: (i) the terms of this Sale Order shall be immediately effective and enforceable upon its entry and the fourteen (14) day stay provided in Bankruptcy Rule 6004(h) and Bankruptcy Rule 6006(d) is hereby expressly waived and shall not apply; (ii) the Debtor is not subject to any stay in the implementation, enforcement or realization of the relief granted in this Sale Order; and (iii) the Debtor may, in its discretion and without further delay, take any action and perform any act authorized under this Sale Order.
- 31. <u>Provisions Non-Severable</u>. The provisions of this Sale Order are nonseverable and mutually dependent.
- 32. <u>Back-Up Bidder</u>. The combined bid of (a) Ritchie Exploration, Inc. for the Kansas Assets in the cash bid amount of \$2,451,000 and (b) O2 Energy, LLC for the West Texas and Other Assets in the cash bid amount of \$924,000 is hereby declared the Back-Up Bid for the Kansas, West Texas, and Other Assets, and Ritchie Exploration, Inc. and O2 Energy, LLC together are declared the Back-Up Bidder for the Kansas, West Texas, and Other Assets. Subject to, and in accordance with, the terms of this Sale Order, the Debtor is authorized to consummate, complete

and close the sale of the Kansas, West Texas, and Other Assets and the assumption and assignment of the Assigned Contracts to the Back-Up Bidder in the event that the sale to the Buyer as contemplated by the Asset Purchase Agreement does not close for any reason whatsoever, without further order of this Court.

33. <u>Satisfaction of Conditions Precedent</u>. Neither the Buyer nor the Debtor shall have an obligation to close the Transactions until all conditions precedent in the Asset Purchase Agreement to each of their respective obligations to close the Transactions have been met, satisfied, or waived in accordance with the terms of the Asset Purchase Agreement.

Signed on 09/03/2019

THE HONORABLE BILL PARKER

CHIEF UNITED STATES BANKRUPTCY JUDGE

Execution Version

ASSET PURCHASE AGREEMENT

DATED AS OF AUGUST 27, 2019,

BY AND BETWEEN

O'BENCO IV, LP,

AS SELLER,

PACHIRA OIL AND GAS, LLC,
AS BUYER

AND

TABLE OF CONTENTS

		<u>Page</u>
ART	ICLE 1 DEFINITIONS	2
1.1	Definitions	2
1.2	Other Definitions and Interpretive Matters	
ART	ICLE 2 PURCHASE AND SALE	14
2.1	Purchase and Sale	14
2.2	Excluded Assets	
2.3	Assumed Liabilities	
2.4	Excluded Liabilities	
2.5	Assigned Contracts and Cure Costs	
2.6	Assignment of Oil and Gas Assets Subject to Consent Requirements	
2.7	Further Assurances	
ART	ICLE 3 PURCHASE PRICE	20
3.1	Purchase Price	20
3.2	Good Faith Deposit	
ART	ICLE 4 CLOSING	21
4.1	Closing Date	21
4.2	Payment on the Closing Date	
4.3	Buyer's Deliveries	
4.4	Seller's Deliveries	
ART	ICLE 5 REPRESENTATIONS AND WARRANTIES OF SELLER	23
5.1	Organization and Good Standing	23
5.2	Authority; Validity; Consents	
5.3	No Conflict	
5.4	Material Assigned Contracts	
5.5	Permits	
5.6	Wells; Plug and Abandon Notice	
5.7	Imbalances	
5.8	AFEs	
5.9	Non-Consent Operations	
5.10	Hedging	
5.11	Preferential Purchase Rights	
5.12	Intellectual Property	
5.13	Taxes	
5.14	Legal Proceedings	
5.15	Sufficiency of Oil and Gas Assets	
5.16	Compliance with Laws	
5.17	Equipment	
5.18	Brokers or Finders	
5.19	Title Matters	
5.20	Knowledge Qualifier for Non-Operated Oil and Gas Assets	
5.21	Rankruntev	26

TABLE OF CONTENTS (continued)

		<u>Page</u>
ART	ICLE 6 REPRESENTATIONS AND WARRANTIES OF BUYER	27
6.1	Organization and Good Standing	27
6.2	Authority; Validity; Consents	
6.3	No Conflict	
6.4	Availability of Funds	
6.5	Litigation.	
6.6	Brokers or Finders	
6.7	Business Use, Bargaining Position, Representation	
6.8	Independent Investigation	
6.9	Bankruptcy	
ARTI	ICLE 7 ACTIONS PRIOR TO THE CLOSING DATE	29
7.1	Access and Reports	29
7.2	Operations Prior to the Closing Date	30
7.3	Reasonable Best Efforts	
7.4	Bankruptcy Court Approval	
7.5	Bankruptcy Filings	
7.6	Bidding Procedures.	
7.7	Back-Up Bidder	
7.8	Updates and Amendments of Exhibits and Schedules	
ARTI	ICLE 8 ADDITIONAL AGREEMENTS	34
8.1	Taxes	34
8.2	Allocation of Purchase Price	
8.3	Bulk Sales	
8.4	Payments Received	
8.5	Assigned Contracts: Adequate Assurance and Performance; Security Arrangements	
8.6	Post-Closing Books and Records and Personnel	
8.7	No Other Representations or Warranties; Disclaimers; Indemnification	
8.8	Casualty	
8.9	Preferential Purchase Rights	
8.10	Accounting Adjustments for Revenues and Expenses	
8.11	Initial Adjustment at Closing	
ARTI	ICLE 9 CONDITIONS PRECEDENT TO OBLIGATIONS OF BUYER TO CLOSE.	41
9.1	Accuracy of Representations	41
9.2	Seller's Performance	
9.3	No Order	
9.4	Seller's Deliveries	
9.5	Sale Order	
ARTI	ICLE 10 CONDITIONS PRECEDENT TO THE OBLIGATION OF SELLER TO CI	LOSE 42
10.1	Accuracy of Representations	42
10.2		

TABLE OF CONTENTS (continued)

		Page
10.3	Buyer's Performance	42
10.4	No Order	42
10.5	Buyer's Deliveries	42
ARTI	CLE 11 TERMINATION	43
11.1	Termination Events	43
11.2	Effect of Termination	44
11.3	Mutual Acknowledgment of Agreements	44
ARTI	CLE 12 TITLE AND ENVIRONMENTAL MATTERS	44
12.1	NORM	44
12.2	Sole Remedy; Waiver of Title and Environmental Matters	
ARTI	CLE 13 GENERAL PROVISIONS	45
13.1	Survival	45
13.2	Confidentiality	46
13.3	Public Announcements	46
13.4	Notices	46
13.5	Waiver, Waiver of Damages	47
13.6	Entire Agreement; Amendment	
13.7	Assignment	
13.8	Severability	
13.9	Expenses	
13.10	Time of Essence	
13.11	Governing Law; Consent to Jurisdiction and Venue; Jury Trial Waiver	
13.12	Counterparts	
13.13	Parties in Interest; No Third Party Beneficiaries	
13.14	Non-Recourse	
13.15	Disclosure Schedules; Materiality	49
13.16	WAIVER OF TEXAS DECEPTIVE TRADE PRACTICES – CONSUMER	
	PROTECTION ACT	
13.17	Injunctive Relief	50

SCHEDULES

Seismic Surveys
Certain excluded cash and cash equivalents
Excluded Leases and Excluded Mineral Interests
Excluded Contracts
Excluded Surface Interests
Assumed Accounts Payables
Executory Contracts and Cure Costs
Operations Prior to the Closing Date
Allocated Values
Tax Allocation
Security Arrangements

DISCLOSURE SCHEDULES

Disclosure Schedule 5.2	Required Consents
Disclosure Schedule 5.4	Material Assigned Contracts
Disclosure Schedule 5.6	Wells
Disclosure Schedule 5.7	Imbalances
Disclosure Schedule 5.8	AFEs
Disclosure Schedule 5.9	Non-Consent Operations
Disclosure Schedule 5.11	Preferential Purchase Rights
Disclosure Schedule 5.12	Intellectual Property
Disclosure Schedule 5.14	Legal Proceedings
Disclosure Schedule 5.15	Sufficiency of Oil and Gas Assets
Disclosure Schedule 5.16	Compliance with Laws
Disclosure Schedule 5.19	Title Matters

EXHIBITS

Exhibit A Exhibit B Exhibit C Exhibit D Exhibit E Exhibit F Exhibit G Exhibit H	Assigned Leases and Interests Wells Assigned Contracts Surface Interests Form of Deed Form of Assignment Bidding Procedures Form of Mineral Deed
Exhibit I	Equipment
EXHIBIT I	Equipment

ASSET PURCHASE AGREEMENT

THIS ASSET PURCHASE AGREEMENT (this "<u>Agreement</u>"), dated as of August 27, 2019 ("<u>Execution Date</u>"), is by and between O'Benco IV, LP, a Delaware limited partnership ("<u>Seller</u>"), and Pachira Oil and Gas, LLC, a Delaware limited liability company ("<u>Buyer</u>"). Capitalized terms used but not otherwise defined herein have the meanings set forth in <u>Article 1</u>. Seller and Buyer are sometimes referred to collectively herein as the "<u>Parties</u>" and individually as a "<u>Party</u>".

RECITALS

WHEREAS, Seller is engaged in the business of onshore oil and natural gas exploration, development and production in the United States of America, and owns, in varying proportions, certain oil and gas leases and associated assets more particularly described in <u>Section 2.1</u>;

WHEREAS, Seller is a debtor and debtor in possession under title 11 of the United States Code, 11 U.S.C. §§ 101, et seq. (the "Bankruptcy Code"), and filed a voluntary petition for relief under chapter 11 of the Bankruptcy Code on June 3, 2019 (the "Petition Date"), in the United States Bankruptcy Court for the Eastern District of Texas (the "Bankruptcy Court"), where Seller's bankruptcy case is being administered under Case No. 19-60384 (the "Bankruptcy Case");

WHEREAS, Seller desires to sell to Buyer all of the Oil and Gas Assets, and Buyer desires to purchase from Seller all of the Oil and Gas Assets and assume all of the Assumed Liabilities, upon the terms and conditions hereinafter set forth;

WHEREAS, the Parties intend to effectuate the transactions contemplated by this Agreement through a sale of the Oil and Gas Assets pursuant to Sections 105, 363 and 365 of the Bankruptcy Code; and

WHEREAS, the Seller's ability to consummate the transactions set forth in this Agreement is subject to, among other things, the entry of the Sale Order by the Bankruptcy Court.

NOW, THEREFORE, in consideration of the premises, the mutual promises herein made, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties agree as follows:

ARTICLE 1

DEFINITIONS

1.1 Definitions.

For purposes of this Agreement, the following terms have the meanings specified or referenced below.

- "Action" means any legal action, suit or arbitration, or any inquiry, proceeding or investigation, by or before any Governmental Authority.
 - "AFEs" has the meaning set forth in Section 5.8.
- "Affiliate" has the meaning set forth in Rule 12b-2 of the regulations promulgated under the Securities Act.
 - "Agreement" has the meaning set forth in the introductory paragraph.
 - "Allocated Value" has the meaning set forth in Section 8.2.
 - "Alternative Outside Date" means September 30, 2019.
 - "Alternative Transaction" means:
- (i) any merger, consolidation, share exchange or other similar transaction to which the Seller is a party that has the effect of transferring, directly or indirectly, all or a substantial portion of the Oil and Gas Assets;
- (ii) any direct or indirect sale or sales of all or a substantial portion of the Oil and Gas Assets; or
- (iii) any other transaction or transactions, including a plan of liquidation or reorganization, that transfers or vests ownership of, economic rights to, or benefits in all or a substantial portion of the Oil and Gas Assets, in all cases of subparts (i) (iii) of this definition, to any party other than Buyer.
 - "Asset Taxes" has the meaning set forth in Section 8.1(b).
 - "Assigned Contracts" has the meaning set forth in Section 2.1(b)(vi).
 - "Assigned Leases and Interests" has the meaning set forth in Section 2.1(b)(i).
 - "Assigned Surface Interests" has the meaning set forth in Section 2.1(b)(vii).
- "Assignment" means the Assignment and Bill of Sale substantially in the form attached hereto as **Exhibit F**.
- "<u>Assumed Accounts Payables</u>" means those accrued expenses and accounts payable by Seller with respect to those suppliers and service providers set forth on <u>Schedule 2.3(h)</u>.
 - "Assumed Liabilities" has the meaning set forth in Section 2.3.

- "Auction" has the meaning set forth in the Bidding Procedures.
- "Avoidance Actions" means any and all claims for relief of Seller under chapter 5 of the Bankruptcy Code.
- "<u>Back-Up Bidder</u>" means the bidder, determined by the Seller, to have made the second highest or next best offer for the Oil and Gas Assets at the Auction.
 - "Bankruptcy Case" has the meaning set forth in the recitals.
 - "Bankruptcy Code" means Title 11 of the United States Code, Section 101 et seq.
 - "Bankruptcy Court" has the meaning set forth in the recitals.
 - "Base Purchase Price" has the meaning set forth in Section 3.1.
- "Bidding Procedures" means bid procedures in substantially the form attached hereto as $\underline{\mathbf{Exhibit}}$ $\underline{\mathbf{G}}$, as approved and amended by the Bankruptcy Court.
- "Bidding Procedures Order" means the Order of the Bankruptcy Court dated June 7, 2019, Approving Bidding Procedures; Scheduling Bidding Deadline, Auction Date, and Sale Hearing Date; Approving Form and Notice Thereof, and any amendment thereto.
- "Business Day" means a day that is not a Saturday, a Sunday or any other day on which national banking institutions in Houston, Texas are open to the public for conducting business and are not required or authorized to close.
 - "Buyer" has the meaning set forth in the introductory paragraph.
- "Buyer Parties" means Buyer, its respective Affiliates and the former, current or future equity holders and Representatives of each of the foregoing.
- "Casualty Loss" means any loss, damage, or destruction of the Oil and Gas Assets that occurs during the period between the Execution Date and the Closing for any reason, including any act of God, fire, explosion, collision, earthquake, windstorm, flood, terrorism, or other casualty or condemnation taking under the right of eminent domain, but excluding any loss, damage, or destruction as a result of depreciation, ordinary wear and tear, and any change in condition of the Oil and Gas Assets for production of Hydrocarbons through normal depletion (which exclusion shall include the watering-out of any Well, collapsed casing, sand infiltration of any Well, or other reservoir changes relating to production issues).
 - "Closing" has the meaning set forth in Section 4.1.
 - "Closing Date" means the date and time as of which the Closing occurs as set forth in Section 4.1.
 - "Code" means the Internal Revenue Code of 1986, as amended.
 - "Confidentiality Agreement" has the meaning set forth in Section 13.2.
- "Contract" means any agreement, contract, obligation, promise or undertaking (in each case, whether written or oral), other than a Lease or any document constituting Seller's chain of title to the Oil and Gas Assets, that is legally binding.

"Contract Notice" has the meaning set forth in Section 2.5(e).

"Copyrights" means all United States and foreign copyright rights in any original works of authorship, whether registered or unregistered, including all copyright registrations and applications.

"Cure Costs" has the meaning set forth in Section 2.5.

"Defensible Title" means that beneficial title to the Properties as of the Effective Time and record title as of the Closing which, subject to any Permitted Encumbrances: (i) with respect to each Well set forth on **Exhibit B**, entitles Seller to not less than the Net Revenue Interest set forth in **Exhibit B** for such Well throughout the productive life thereof, except (A) decreases in connection with those operations in which Seller may elect after the date hereof to be a non-consenting co-owner, (B) decreases resulting from the establishment or amendment, after the date hereof, of pools or units, (C) decreases required to allow other Working Interest owners to make up past underproduction or pipelines to make up past under deliveries, and (D) as stated in **Exhibit B**; (ii) with respect to each Well set forth on **Exhibit B**, obligates Seller to bear a Working Interest for such Well that is not greater than the Working Interest set forth in **Exhibit B** for such Well without increase throughout the productive life of such Well, except (A) as stated in **Exhibit B** and (B) increases resulting from contribution requirements with respect to defaulting or non-consenting co-owners under applicable operating agreements or applicable Legal Requirements and increases that are accompanied by at least a proportionate increase in Seller's Net Revenue Interest for such Well; and (iii) is free and clear of all Encumbrances other than Permitted Encumbrances.

"<u>Designation Deadline</u>" means 5:00 p.m., prevailing Central Time, on the date that is five (5) Business Days prior to the Closing Date or such later date as Buyer and Seller shall mutually agree and as the Bankruptcy Court may authorize; *provided*, that the Designation Deadline for any Executory Contract with respect to which a dispute regarding a Cure Cost exists on such date shall be five (5) days after the date of the resolution of such dispute.

"<u>DTPA</u>" has the meaning set forth in <u>Section 13.16</u>.

"Effective Time" means 12:01 a.m., Central Standard Time, on June 1, 2019.

"Encumbrance" means any charge, lien, claim, mortgage, lease, sublease, hypothecation, deed of trust, pledge, security interest, option, right of use or possession, right of first offer or first refusal, easement, servitude, restrictive covenant, encroachment, encumbrance, third party interest or other restriction or limitation of any kind.

"Environmental Laws" means any federal, state, or local Legal Requirement relating to the prevention of pollution, remediation of contamination, protection of the environment and natural resources, and restoration of environmental quality, including the following federal statutes and the regulations promulgated thereunder: (i) the Comprehensive Environmental Response, Compensation and Liability Act of 1980, (ii) the Emergency Planning and Community Right-To-Know Act, (iii) the Resources Conservation and Recovery Act, (iv) the Clean Air Act, (v) the Clean Water Act, (vi) the Safe Drinking Water Act, (vii) the Oil Pollution Act and (viii) the Toxic Substances Control Act, as each of the foregoing has been amended and in effect on the Closing Date.

"Equipment" has the meaning set forth in Section 2.1(b)(iv).

"Excluded Assets" has the meaning set forth in Section 2.2.

"Excluded Contracts" means those Contracts described on Schedule 2.2(h).

- "Excluded Leases" means those Leases described on Schedule 2.2(g).
- "Excluded Liabilities" has the meaning set forth in Section 2.4.
- "Excluded Mineral Interests" means those Mineral Interests described on Schedule 2.2(g).
- "Excluded Surface Interests" means those Surface Interests described on Schedule 2.2(i).
- "Execution Date" has the meaning set forth in the introductory paragraph.
- "Executory Contract" means any executory Contract related to the Oil and Gas Assets to which Seller is a party.

"Final Order" means, as applicable, an order or judgment of the Bankruptcy Court or other court of competent jurisdiction, which has not been reversed, stayed, modified or amended, and as to which the time to appeal or seek certiorari has expired and no appeal or petition for certiorari has been timely taken, or as to which any appeal that has been taken or any petition for certiorari that has been or may be filed has been resolved by the highest court to which the order or judgment could be appealed or from which certiorari could be sought or the new trial, reargument or rehearing shall have been denied, resulted in no modification of such order or has otherwise been dismissed with prejudice; provided that the possibility that a motion under Rule 60 of the Federal Rules of Civil Procedure, as made applicable by Rule 9024 of the Federal Rules of Bankruptcy Procedure, may be filed relating to such order shall not cause such order to not be a Final Order. "Final Order" shall also include any order or judgment that has been appealed but for which a stay pending appeal has not been obtained, if the Bankruptcy Court, after notice and a hearing, concludes that the debtors are entitled to give effect to such order or judgment notwithstanding such appeal.

"Good Faith Deposit" has the meaning set forth in Section 3.2(a).

"Governmental Authority" means any United States federal, state or local or any foreign government, governmental authority or regulatory or administrative authority or any court, tribunal or judicial body having jurisdiction.

"Governmental Authorization" means any approval, consent, license, permit, waiver or other authorization issued, granted or otherwise made available by or under the authority of any Governmental Authority.

"Hard Consent" means a consent that, if Closing were to occur without having obtained such consent would (i) cause the assignment of the Oil and Gas Asset(s) affected thereby to Buyer to be void or voidable, (ii) cause the termination of a Lease, Assigned Contract or Assigned Surface Interest under the express terms thereof, or (iii) require the payment of a fee (unless such fee has been paid by Buyer); provided, however, that (x) consents and approvals from Governmental Authorities for the assignment of the Oil and Gas Assets to Buyer that are customarily obtained after such assignment of properties similar to the Oil and Gas Assets, (y) subject to subpart (i) of this paragraph, any consents and approvals which cannot be unreasonably withheld by the holder thereof, and (z) any consents that would be Hard Consents but which consent or approval has been granted in writing prior to the Closing Date, in each case shall not be "Hard Consents."

"Hazardous Substance" means petroleum and its byproducts, asbestos and any other material or substance which is defined as a "hazardous waste," "hazardous substance," "hazardous material," "restricted hazardous waste," "industrial waste," "solid waste," "contaminant," "pollutant," "toxic waste"

or "toxic substance" or otherwise regulated under any Environmental Law or for which Liability can be imposed under any Environmental Law.

"Hydrocarbons" means oil, gas, minerals, casinghead gas, coalbed methane, and other gaseous and liquid hydrocarbons, or any combination of the foregoing, sulfur extracted from hydrocarbons, and all other Lease substances.

"Imbalances" means over-production or under-production or over-deliveries or under-deliveries with respect to Hydrocarbons produced from or allocated to the Properties, regardless of whether such over-production or under-production or over-deliveries or under-deliveries arise at the wellhead, pipeline, gathering system, transportation system, processing plant, or other location, including any imbalances under gas balancing or similar agreements, imbalances under production handling agreements, imbalances under processing agreements, imbalances under the Assigned Leases and Interests, and imbalances under gathering or transportation agreements.

"Intellectual Property" means all intellectual property, including all Copyrights, Patents and Trademarks, owned, used or licensed by Seller and used or held for use exclusively in the ownership and operation of the Oil and Gas Assets, excluding any geological, geophysical and seismic data.

"Knowledge" means (i) with respect to the Seller, the actual knowledge (after reasonable inquiry) of any of the following individuals: William J. O'Brien IV, Mike Ameen and Gary Love, and (ii) with respect to Buyer, the actual knowledge (after reasonable inquiry) of any of the following individuals: Brad Morse, Daniel Silverman, and Ryan Sullivan.

"<u>Lease</u>" means any existing oil and gas lease, oil, gas and mineral lease, sublease, and other Hydrocarbon leasehold interest, and the leasehold estates created thereby, including carried interests, rights of recoupment, options, reversionary interests, convertible interests, and rights to reassignment.

"<u>Legal Requirement</u>" means any federal, state, provincial, local, municipal, foreign, international, multinational, or other administrative Order, constitution, law, ordinance, principle of common law, regulation, statute or treaty.

"<u>Liability</u>" means any debt, loss, claim (including "claim" as defined in the Bankruptcy Code), damage, demand, Tax, fine, judgment, penalty, liability, expense, commitment or obligation (including those arising out of any action, such as any settlement or compromise thereof or judgment or award therein), whether known or unknown, asserted or un-asserted, absolute or contingent, accrued or un-accrued, liquidated or unliquidated, or due or to become due, and whether in contract, tort, strict liability or otherwise, and whether or not resulting from third party claims.

"Material Adverse Effect" means any change, event, condition, circumstance or occurrence that individually or in the aggregate (taking into account all other such changes, events, conditions, circumstances or occurrences) has had, or would be reasonably likely to have, a material adverse change in or material adverse effect on the Oil and Gas Assets taken as a whole, but excluding:

(a) any change, event, condition, circumstance or occurrence to the extent that it results from or arises out of (i) the commencement or pendency of the Bankruptcy Case; (ii) the execution and delivery of this Agreement or the announcement thereof or consummation of the transactions contemplated hereby; (iii) changes in (or proposals to change) Legal Requirements, generally accepted accounting principles or other accounting regulations or principles; (iv) any Casualty Loss; or (v) any action contemplated by this Agreement or taken at the request of Buyer;

- (b) any change or effect generally applicable to (i) the industries and markets in which Seller operates or (ii) economic or political conditions or the securities or financial markets in any country or region;
- (c) any event that is not within the reasonable control of the Parties, including, without limitation, any outbreak or escalation of hostilities or war, any act of terrorism or any act of God, flood, drought, earthquake or other natural disaster;
 - (d) the departure of officers or directors of Seller after the Execution Date;
- (e) any objections in the Bankruptcy Court to (i) this Agreement and the other Transaction Documents and the transactions contemplated hereby and thereby, (ii) the reorganization of Seller and any related plan of reorganization or disclosure statement, (iii) the Bidding Procedures or the Sale Motion, or (iv) the assumption or rejection of any Material Assigned Contract;
- (f) any Order of the Bankruptcy Court or any actions or omissions of Seller in compliance therewith;
 - (g) any action taken by Seller at the request of, or with the consent of, Buyer; and
- (h) any of the matters disclosed on any Exhibit, Schedule or Disclosure Schedule to this Agreement.

"Material Assigned Contracts" means, to the extent that they relate to the Oil and Gas Assets, the following: (a) any Assigned Contract that can reasonably be expected to result in aggregate payments by or revenues to Seller or Buyer with respect to the Oil and Gas Assets of more than \$200,000.00 during the current or any subsequent fiscal year (based solely on the terms thereof and without regard to any expected increase in volumes or revenues); (b) Hydrocarbon purchase and sale, exchange, marketing, compression, gathering, transportation, processing, refining, or similar Assigned Contracts (in each case) to which Seller is a party (or to which any portion of the Oil and Gas Assets is subject) with respect to Hydrocarbons from the Oil and Gas Assets that are not terminable without penalty on ninety (90) days or less notice (including any Assigned Contract providing for acreage dedication or volumetric or monetary commitments or indemnification therefor or for dedication of future production); (c) any Assigned Contract binding upon Seller to sell, lease, farmout, or otherwise dispose of or encumber any interest in any of the Oil and Gas Assets after the date hereof, other than conventional rights of reassignment arising in connection with Seller's surrender or release of any of the Oil and Gas Assets (except where any such right of reassignment has been triggered prior to the Effective Time); (d) any Assigned Contracts that would obligate Buyer to drill additional wells or conduct other material development operations after the Closing; (e) any Assigned Contracts that constitute a non-competition agreement or any agreement that purports to restrict, limit, or prohibit the manner in which, or the locations in which, Seller conducts business, including areas of mutual interest; (f) any Assigned Contracts providing for any call upon, option to purchase, or similar rights with respect to the Oil and Gas Assets or to the production therefrom or the processing thereof, or is a dedication of production or otherwise requires production to be transported, processed, or sold in a particular fashion; (g) any Assigned Contract that constitutes a joint or unit operating agreement; (h) any Assigned Contract that constitutes a farmout agreement, partnership agreement, participation agreement, joint venture agreement, or similar Contract; and (i) any Assigned Contract between Seller and any Affiliate of Seller.

"Mineral Deed" means the mineral deed to be executed and delivered by the Parties at Closing pursuant to Section 4.3(d) and Section 4.4(a) substantially in the form attached hereto as Exhibit H.

"Mineral Interests" means all mineral fee interests, non-leasehold mineral rights, and mineral servitudes in which Seller owns an interest, including royalty interests, overriding royalty interests, net profit interests, production payments, and without limiting the foregoing, other rights of whatever nature, whether legal or equitable, whether vested or contingent.

"Net Cash Flow" has the meaning set forth in Section 8.10.

"Net Revenue Interest" means, (i) with respect to any Well set forth on **Exhibit B**, Seller's interest (expressed as a percentage or a decimal) in and to the Hydrocarbons produced and saved or sold from or allocated to such Well on **Exhibit B** from those formations in which such Well is currently producing, or with respect to a Well that is not currently producing, the last depth or formation at which it produced, or (ii) with respect to any Lease set forth on **Exhibit A**, Seller's interest (expressed as a percentage or a decimal) in and to the Hydrocarbons produced and saved or sold from or allocated to such Lease as to the lands and depths described for such Lease on **Exhibit A**, in each case of items (i) and (ii), after giving effect to all royalties, overriding royalties, nonparticipating royalties, net profits interests, production payments, carried interests, reversionary interests and other burdens on, measured by or payable out of production therefrom.

"NORM" means naturally occurring radioactive materials.

"O'Brien Resources" means O'Brien Resources, LLC, a Texas limited liability company, d/b/a O'Benco, Inc.

"Oil and Gas Assets" has the meaning set forth in Section 2.1(b).

"Order" means any award, writ, injunction, judgment, order or decree entered, issued, made, or rendered by any Governmental Authority.

"Outside Date" has the meaning set forth in Section 11.1(b)(iii).

"Party" or "Parties" has the meaning set forth in the introductory paragraph.

"<u>Patents</u>" means United States and foreign patents and patent applications, as well as any continuations, continuations-in-part, divisions, extensions, reexaminations, reissues, renewals and patent disclosures related thereto.

"Paying Party" has the meaning set forth in Section 8.1(c).

"Permits" means any governmental (whether federal, state or local) permits, licenses, authorizations, franchises, grants, easements, variances, exceptions, consents, certificates, approvals and related instruments or rights of any Governmental Authority or other third party, and any writ, judgment, decree, award, order, injunction or similar order, writ, ruling, directive or other requirement of any Governmental Authority (in each such case whether preliminary or final) required for the ownership, operation or use of the Properties or Equipment.

"Permitted Encumbrances" means any of the following: (a) any rights, obligations, or duties reserved to or vested in any municipality or other Governmental Authority to: (1) control or regulate any Oil and Gas Asset in any manner including all applicable Legal Requirements, (2) purchase, condemn, expropriate, or recapture any Oil and Gas Asset, (3) designate a purchaser of any Oil and Gas Asset, or (4) use any Oil and Gas Asset in any manner, except to the extent such use operates to reduce the applicable Net Revenue Interest in a Well below that shown in **Exhibit B**, or increase the applicable Working Interest

in a Well above that shown in Exhibit B, without a proportionate increase in the applicable Net Revenue Interest; (b) the terms and conditions of all options, servitudes, contracts for sale, purchase, exchange, refining or processing of Hydrocarbons, operating agreements, unitization, pooling, and communitization agreements, farmins, farmouts, area of mutual interest agreements, declarations, or orders, construction agreements, construction and operation agreements, participation agreements, shoot-to-earn agreements, exploration agreements, partnership agreements, processing agreements, plant agreements, pipeline, gathering, exchange, and transportation agreements, disposal agreements, permits, licenses, and any other agreements affecting the Oil and Gas Assets, including those set forth as Assigned Contracts on Exhibit C, but only to the extent that they do not, either individually or in the aggregate, (1) operate to reduce the applicable Net Revenue Interest in a Well below that shown in Exhibit B, or increase the applicable Working Interest in a Well above that shown in **Exhibit B** without a proportionate increase in the applicable Net Revenue Interest in such Well, or (2) except in the case of Assigned Contracts listed on Exhibit C adversely affect the ownership and/or operation of the affected Oil and Gas Assets (as currently used or owned) in any material respect; (c) any consent applicable to the transactions contemplated hereby; (d) easements, rights-of-way, servitudes, permits, surface leases, and other similar rights on, over, or in respect of any of the Oil and Gas Assets, as long as any such encumbrances, individually or in the aggregate, do not operate to reduce the applicable Net Revenue Interest in a Well below that shown in Exhibit B, or increase the applicable Working Interest in a Well above that shown in **Exhibit B**, without a proportionate increase in the applicable Net Revenue Interest or interfere in any material respect with Seller's use or operation of the Oil and Gas Assets (as currently used or operated) burdened thereby; (e) all royalties, overriding royalties, production payments, net profits interests, reversionary interests, carried interests, and other burdens with respect to a Well if the net cumulative effect of such burdens does not operate to reduce the Net Revenue Interest in such Well below that shown in Exhibit B, or increase the Working Interest in such Well above that shown in **Exhibit B** without a proportionate increase in the Net Revenue Interest in such Well; (f) defects or irregularities of title (i) as to which the relevant statute(s) of limitations or prescription would bar any attack or claim against Seller's title, (ii) arising out of lack of corporate authorization or a variation in corporate name, (iii) consisting of the failure to recite marital status or omissions of heirship proceedings in documents, (iv) resulting from lack of survey or failure to record releases of liens, production payments, or mortgages that have expired by their own terms or (v) based solely on lack of information in Seller's files or Records; (g) liens or other Encumbrances for Taxes not yet due and payable or that are being contested in good faith; (h) materialman's, mechanic's, repairman's, employee's, contractor's, operator's, and other similar liens or Encumbrances arising in the ordinary course of business for payments not yet delinquent that are inchoate and have not been perfected pursuant to law or that are contained in joint operating agreements or similar agreements covering the Oil and Gas Assets or if delinquent, that are being contested in good faith; (i) Imbalances; (j) plugging and surface restoration obligations; (k) all rights to consent or approval by, required notices to, filings with, or other actions by, Governmental Authorities in connection with the conveyance of the Assigned Leases and Interests, if the same are customarily sought and received after the Closing; (1) calls on Hydrocarbon production under any Assigned Contracts; (m) the terms and conditions of the Assigned Leases and Interests, including any depth limitations or similar limitations that may be set forth therein and any liens or security interests reserved in the Assigned Leases and Interests for royalty, bonus, or rental, or for compliance with the terms of the Assigned Leases and Interests, except to the extent such terms operate to reduce the applicable Net Revenue Interest in a Well below that shown in **Exhibit B**, or increase the applicable Working Interest in a Well above that shown in Exhibit B, without a proportionate increase in the applicable Net Revenue Interest or interfere in any material respect with Seller's use or operation of the Oil and Gas Assets (as currently used or operated); (n) liens, obligations, defects, irregularities, or other Encumbrances affecting the Oil and Gas Assets that would be waived by an ordinary prudent operator or company experienced in the acquisition or divestiture of producing properties; (o) mortgages on the lessor's interest under an Assigned Lease and Interest; (p) subject to Section 8.9, Preferential Purchase Rights; (q) any maintenance of uniform interest provisions (i) contained in any agreement applicable to the Oil and Gas Assets and Leases to the extent (A) compliance with such provision has been waived in writing by the parties to such agreement, or (B) Buyer is able to confirm (through review of the Records and communications with, and cooperation from, Seller) that such maintenance of uniform interest provision has been previously violated without the transferring party obtaining waivers, or without a non-transferring party under such agreement having filed a lawsuit or having asserted a claim for breach of such agreement, (ii) contained in any Lease to the extent the applicable lessor has waived compliance with such provisions in writing or breach of such provisions will not result in a suspension of material rights under such Lease, the right of the lessor to terminate such Lease or the termination of such Lease or (iii) if the violation of such provision would not give rise to the unwinding of the sale of the affected Oil and Gas Asset from the Seller to Buyer; (r) liens or trusts arising in connection with workers' compensation, unemployment insurance, or pension; (s) conventional rights of reassignment obligating the Seller to reassign its interest in any portion of the Assigned Leases and Interests to a third party, if such right is only triggered when Buyer expressly indicates its intention to release or abandon such interest prior to the expiration of the primary term or other termination of such interest; (t) decreases in the Net Revenue Interest of Seller (or Buyer as successor in interest to Seller) with respect to any Well below the Net Revenue Interest shown in Exhibit B with respect to such Well, (i) in connection with those operations in which Seller or its successors or assigns may from and after the date of this Agreement, elect in accordance with the terms hereof to be a non-consenting co-owner, (ii) resulting from the establishment or amendment from and after the date of this Agreement of drilling and spacing units, pooling orders or pooled units, unitization agreements, communitization agreements, or other units, or (iii) required to allow other working interest owners to make up past underproduction or pipelines to make up past under deliveries; (u) increases in the Working Interest of Seller (or Buyer as successor in interest to Seller) with respect to any Well above that shown in Exhibit B for such Well, resulting from contribution requirements with respect to defaulting co-owners under applicable operating agreements; and (v) any Encumbrances that will be released by the Sale Order.

"Person" means any individual, corporation (including any non-profit corporation), partnership, limited liability company, joint venture, estate, trust, association, organization or other entity or Governmental Authority.

"Petition Date" has the meaning set forth in the recitals.

"Phase I Environmental Site Assessment" means a Phase I environmental property assessment of the Oil and Gas Assets that satisfies the basic assessment requirements set forth under the current ASTM International Standard Practice for Environmental Site Assessments (Designation E1527-13).

"<u>Preferential Purchase Right</u>" means any right or agreement that enables any Person to purchase or acquire any Oil and Gas Asset or any interest therein or portion thereof as a result of or in connection with the execution or delivery of this Agreement or the consummation of the transactions contemplated hereby.

"Preliminary Settlement Statement" means that certain statement provided by Seller to Buyer pursuant to Section 8.11, as amended (if applicable) by mutual agreement prior to Closing, setting forth those initial adjustments to the Base Purchase Price made at Closing.

"Proceeding" means any Action, arbitration, audit, hearing, investigation, litigation, or suit (whether civil, criminal, administrative or investigative) commenced, brought, conducted, or heard by or before, or otherwise involving, any Governmental Authority.

"Properties" has the meaning set forth in Section 2.1(b)(ii).

"Property Costs" means all operating expenses (including costs of insurance, rentals, shut-in payments, other payments called for by the Leases, title examination and curative actions, and overhead costs) and capital expenditures (including bonuses, broker fees, and other Lease acquisition costs, costs of

drilling and completing wells, and costs of acquiring equipment) incurred in the ownership and operation of the Oil and Gas Assets in the ordinary course of business, but excluding (without limitation) liabilities, losses, costs, expenses, and damages attributable to:

- (i) claims, investigations, administrative proceedings, arbitration or litigation directly or indirectly arising out of or resulting from actual or claimed personal injury, illness, or death; property damages; environmental damage or contamination; other torts; private rights of action given under any Legal Requirement; or violation of any Legal Requirement;
- (ii) obligations to plug and/or abandon wells, dismantle, decommission or remove facilities;
- (iii) obligations to remediate any contamination of groundwater, surface water, soil, sediments, or Equipment;
 - (iv) title and environmental claims (including claims that Leases have terminated);
- (v) claims of improper calculation or payment of royalties (including overriding royalties and other burdens on production);
 - (vi) gas balancing and other production balancing obligations;
 - (vii) Casualty Loss;
 - (viii) Taxes; and
- (ix) any claims for indemnification, contribution, or reimbursement from any third Person with respect to liabilities, losses, costs, and expenses of the type described in preceding clauses (i) through (viii), whether such claims are made pursuant to contract or otherwise.
 - "Purchase Price" has the meaning set forth in Section 3.1.
 - "Records" has the meaning set forth in Section 2.1(b)(ix).
 - "Reimbursing Party" has the meaning set forth in Section 8.1(c).
- "Release" means any past or present spilling, leaking, pumping, pouring, emitting, emptying, discharging, injecting, escaping, leaching, dumping or disposing of a Hazardous Substance into the environment (including the abandonment or discharging of barrels, containers and other closed receptacles containing any Hazardous Substance).
- "Representative" means, with respect to a particular Person, any director, officer, employee, agent, consultant, advisor or other representative of such Person, including legal counsel, accountants and financial advisors.
 - "Sale Hearing" has the meaning set forth in the Bidding Procedures.
- "Sale Motion" means the Debtor's Motion for (A) Entry of an Order (I) Approving Bidding Procedures; (II) Scheduling Bidding Deadline, Auction Date, and Sale Hearing Date; (III) Approving Form and Notice Thereof; (B) Entry of an Order After the Sale Hearing (I) Authorizing the Debtor to Sell Its Assets; and (II) Authorizing the Debtor to Assume and Assign Certain Executory Contracts and Unexpired Leases; and (C) Granting Related Relief, filed on June 21, 2019.

"Sale Order" means the Order of the Bankruptcy Court (I) Authorizing and Approving the Debtor to Sell Its Assets, (II) Authorizing the Debtor to Assume and Assign Certain Executory Contracts and Unexpired Leases and (III) Granting Related Relief, a proposed version of which Order is to be filed by Seller with the Bankruptcy Court as soon as reasonably practicable prior to the hearing on the Sale Motion.

"Securities Act" means the Securities Act of 1933, as amended.

"Security Arrangements" has the meaning set forth in Section 8.5(c).

"Seller" has the meaning set forth in the introductory paragraph.

"Seller Parties" means Seller, their respective Affiliates and the former, current or future equity holders and Representatives of each of the foregoing.

"Seller's Obligations" has the meaning set forth in Section 8.5(c).

"Straddle Period" has the meaning set forth in Section 8.1(b).

"Successful Bid" has the meaning set forth in the Bidding Procedures.

"Successful Bidder" has the meaning set forth in the Bidding Procedures.

"Surface Interests" means all surface fee interests, surface leases, subsurface leases, rights-of-way, water rights, licenses and easements applicable to, or used or held in connection with the ownership, operation, maintenance or repair of, or the production, gathering, treatment, processing, storing, sale or disposal of Hydrocarbons or produced water from, the Properties, together with all surface fee interests in the lands covered by the Assigned Leases and Interests, excluding any surface use agreements and other surface or subsurface rights agreements.

"Tax" or "Taxes" means (i) all taxes, assessments, fees and other charges of any kind whatsoever imposed by any Governmental Authority, including any federal, state, local and/or foreign income tax, surtax, remittance tax, presumptive tax, net worth tax, special contribution tax, production tax, value added tax, withholding tax, gross receipts tax, windfall profits tax, profits tax, ad valorem tax, personal property tax, real property tax, sales tax, goods and services tax, service tax, transfer tax, use tax, excise tax, premium tax, stamp tax, motor vehicle tax, entertainment tax, insurance tax, capital stock tax, franchise tax, occupation tax, payroll tax, employment tax, unemployment tax, disability tax, alternative or add-on minimum tax and estimated tax, and (ii) any interest, fine, penalty or additions to tax imposed by a Governmental Authority in connection with any item described in clause (i).

"<u>Tax Allocation</u>" has the meaning set forth in <u>Section 8.2</u>.

"<u>Tax Return</u>" means any return, declaration, report, claim for refund, information return or other document (including any related or supporting estimates, elections, schedules, statements, or information) filed or required to be filed in connection with the determination, assessment or collection of any Tax or the administration of any laws, regulations or administrative requirements relating to any Tax.

"<u>Trademarks</u>" means United States, state and foreign trademarks, service marks, logos, slogans, trade dress and trade names, Internet domain names and any other similar designations of source of goods or services, whether registered or unregistered, and registrations and pending applications to register the foregoing, and all goodwill related to or symbolized by the foregoing.

"<u>Transaction Documents</u>" means this Agreement and any other agreements, instruments or documents entered into pursuant to this Agreement including, but not limited to executed originals of Forms P-4's and T-4's as promulgated by the Texas Railroad Commission.

"Transfer Taxes" has the meaning set forth in Section 8.1(a).

"Wells" has the meaning set forth in Section 2.1(b)(ii).

"Working Interest" means, with respect to any Well, the interest (expressed as a percentage or a decimal) that is burdened with the obligation to bear and pay costs and expenses of maintenance, development and operations for that Well from those formations in which such Well is currently producing, or with respect to a Well that is not currently producing, the last depth or formation at which it produced, in each case, without regard to the effect of any royalties, overriding royalties, nonparticipating royalties, net profits interests, production payments, carried interests, reversionary interests and other burdens on, measured by or payable out of production therefrom.

1.2 Other Definitions and Interpretive Matters.

(a) Unless otherwise expressly provided, for purposes of this Agreement, the following rules of interpretation shall apply:

<u>Calculation of Time Period</u>. When calculating the period of time before which, within which or following which any act is to be done or step taken pursuant to this Agreement, the date that is the reference date in calculating such period shall be excluded. If the last day of such period (or if any other date specified in this Agreement for giving any notice or taking any action) is a day other than a Business Day, then the period (or date) in question shall end on (or be deemed to be) the next succeeding Business Day.

Dollars. Any reference in this Agreement to \$ means United States dollars.

<u>Exhibits/Schedules/Disclosure Schedules</u>. All Exhibits, Schedules and Disclosure Schedules attached or annexed hereto or referred to herein are hereby incorporated in and made a part of this Agreement as if set forth in full herein. Any capitalized terms used in any Exhibit, Schedule or Disclosure Schedule but not otherwise defined therein shall be defined as set forth in this Agreement.

Gender and Number. Any reference in this Agreement to gender includes all genders, and words imparting only the singular number include the plural and vice versa.

<u>Headings</u>. The provision of a table of contents, the division of this Agreement into Articles, Sections and other subdivisions and the insertion of headings are for convenience of reference only and shall not affect or be utilized in the construction or interpretation of this Agreement. All references in this Agreement to any "<u>Section</u>" or "<u>Article</u>" are to the corresponding Section or Article of this Agreement unless otherwise specified.

<u>Herein</u>. Words such as "<u>herein</u>," "<u>hereof</u>" and "<u>hereunder</u>" refer to this Agreement as a whole and not merely to a subdivision in which such words appear, unless the context otherwise requires.

<u>Including</u>. The word "<u>including</u>" or any variation thereof means "<u>including</u>, <u>without limitation</u>," and shall not be construed to limit any general statement that it follows to the specific or similar items or matters immediately following it.

(b) No Strict Construction. Buyer, on the one hand, and Seller, on the other hand, participated jointly in the negotiation and drafting of this Agreement, and, in the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as jointly drafted by Buyer, on the one hand, and Seller, on the other hand, and no presumption or burden of proof shall arise favoring or disfavoring any Party by virtue of the authorship of any provision of this Agreement. Without limitation as to the foregoing, no rule of strict construction construing ambiguities against the draftsperson shall be applied against any Person with respect to this Agreement.

ARTICLE 2

PURCHASE AND SALE

2.1 Purchase and Sale.

- (a) Upon the terms and subject to the conditions of this Agreement and the Sale Order, on the Closing Date, Seller shall sell, transfer, assign, convey and deliver, or cause to be sold, transferred, assigned, conveyed and delivered, to Buyer, and Buyer shall purchase from Seller, free and clear of all Encumbrances (other than Permitted Encumbrances), the Oil and Gas Assets.
- (b) The "Oil and Gas Assets" shall include all right, title and interest of Seller in, to or under the following (after excluding any Excluded Assets):
- (i) all Leases (other than the Excluded Leases) and Mineral Interests (other than the Excluded Mineral Interests), including (1) the Leases and Mineral Interests described on **Exhibit A** attached hereto, and (2) those Leases and Mineral Interests located in or under (y) the lands covered by the Leases and/or Mineral Interests described on **Exhibit A** attached hereto, and (z) the Leases and lands included in any units with which the Leases, the Mineral Interests, or the lands covered thereby may have been pooled, unitized or communitized (collectively, the "Assigned Leases and Interests");
- (ii) the oil, gas, water, disposal, observation or injection wells located on the Assigned Leases and Interests, whether producing, non-producing, shut-in, or temporarily abandoned, including, without limitation, those described on **Exhibit B** (collectively, the "Wells", and together with the Assigned Leases and Interests, the "Properties");
- (iii) all Hydrocarbons produced from or attributable to the Properties in storage owned by Seller above custody transfer point at the Closing Date or produced on and after the Closing Date, and all proceeds attributable thereto;
- (iv) all equipment, machinery, fixtures and other tangible personal property and improvements located on, used or held for use or obtained in connection with the operation of the Properties, including that equipment, machinery, fixtures, and other tangible personal property described on **Exhibit I** attached hereto (collectively, the "Equipment");
 - (v) all gathering and processing systems related to the Properties;
- (vi) all Contracts (other than the Excluded Contracts) including, operating agreements, exploration agreements, development agreements, seismic licenses (but only to the extent transferable without material restriction (including a material restriction against assignment without prior consent)), balancing agreements, farmout agreements, service agreements, and other contracts, agreements and instruments, including the Contracts described on **Exhibit C** attached hereto, as such Exhibit may be

subsequently amended by Buyer pursuant to <u>Section 2.5</u>, insofar as they relate to any other Oil and Gas Asset (collectively, the "<u>Assigned Contracts</u>");

- (vii) all Surface Interests (other than Excluded Surface Interests), including those described on **Exhibit D** attached hereto, insofar as they relate to any other Oil and Gas Asset (collectively, the "Assigned Surface Interests");
- (viii) except with respect to the Excluded Assets and the Excluded Liabilities, all claims, refunds, abatements, variances, allocations, causes of action, claims for relief, choses in action, rights of recovery, audit rights, rights of set-off, rights of indemnity, contribution or recoupment, counterclaims, cross-claims and defenses of Seller to the extent related to the Oil and Gas Assets and arising or relating to events occurring from and after the Closing Date or related to the Assumed Liabilities;
- (ix) all information, books, databases, files, records and data (whether in written or electronic format) relating directly to any Oil and Gas Asset and to any Assumed Liabilities except for (A) records that Seller is prohibited from disclosing or transferring under any third party agreement or applicable law, (B) information entitled to legal privilege, including attorney work product and attorney-client communications (except for title opinions, which shall be included in the Records), (C) economic projections and records of offers from or negotiations with, Buyer or third parties with respect to any proposed transfer of any of the Oil and Gas Assets and economic analyses associated therewith (collectively, subject to such exclusions, the "Records"), which Records shall include all lease records, well records, well files, title records (including abstracts of title, title opinions and memoranda, and title curative documents), correspondence (including email correspondence within the past five (5) years, to the extent related to the Oil and Gas Assets, with any third-party field land agent tasked with acquiring Leases on behalf of Seller), production records; and all other data (but only to the extent transferable without material restriction (including a material restriction against assignment without prior consent)), and any other proprietary data which Seller has the right to obtain (either without the payment of money or delivery of other consideration or unduly burdensome effort or, upon Buyer's written election, at Buyer's expense) and relating to the ownership, operation, development, maintenance or repair of, or the production, gathering, treatment, processing, storing, sale or disposal of Hydrocarbons or produced water from, the other Oil and Gas Assets; provided, that if any Records can only be assigned to Buyer with a fee or penalty, Buyer shall bear responsibility for such fee or penalty;
- (x) all trade credits, accounts receivable, note receivables, take or pay amounts receivable, and other receivables attributable to the other Oil and Gas Assets, with respect to any period of time on and after the Closing Date;
- (xi) to the extent that they may be assigned, all Permits that are primarily used in connection with the ownership or operation of the Oil and Gas Assets; and
- (xii) to the extent transferable without payment of any fee (unless Buyer agrees in writing to bear all such fees), all geological, geophysical, or seismic data relating to the Oil and Gas Assets (but excluding Seller's or Seller's Affiliates' interpretation of such geological, geophysical or seismic data and any such data that is subject to licensing requirements and/or other restrictions on transfers), including those surveys described on <u>Schedule 2.1(b)(xii)</u>.

2.2 Excluded Assets.

Notwithstanding the foregoing, the Oil and Gas Assets shall not include, and there is excepted, reserved and excluded from the transaction contemplated hereby, the following (collectively, the "<u>Excluded Assets</u>"):

- (a) the Purchase Price delivered to Seller pursuant to this Agreement;
- (b) all cash and cash equivalents, including checks, commercial paper, treasury bills, certificates of deposit, bank accounts and other bank deposits as of the Closing Date, in each case, excluding any item described in <u>Schedule 2.2(b)</u>;
- (c) all trade credits, accounts receivable, note receivables, take or pay amounts receivable, and other receivables attributable to the Oil and Gas Assets with respect to any period of time prior to the Closing Date or attributable to matters constituting Excluded Liabilities;
- (d) all Hydrocarbons produced from or attributable to the Properties prior to the Closing Date, other than Hydrocarbons produced from or attributable to the Properties in storage owned by Seller above custody transfer points at the Closing Date, and all proceeds attributable thereto;
 - (e) all minute books, stock ledgers, corporate seals and stock certificates of Seller;
- (f) all (i) corporate, financial, Tax and legal records of Seller that relates to Seller's business generally (excepting the same to the extent relating to the Assumed Liabilities and the Oil and Gas Assets) and (ii) books, records and files that relate to any Excluded Assets;
 - (g) all Excluded Leases and all Excluded Mineral Interests;
- (h) all Excluded Contracts (including, to the extent they do not relate to the Assumed Liabilities, all contracts of insurance and all claims, rights and interests of Seller or any Affiliate of Seller (i) under any policy or agreement of insurance or indemnity agreement, or (ii) to any insurance or condemnation proceeds or awards arising, in each case, from acts, omissions or events, or damage to, or destruction of, an Oil and Gas Asset prior to the Closing Date);
 - (i) all Excluded Surface Interests;
- (j) all rights to any refunds or credits of Taxes (or other related costs or expenses) that are borne by or the responsibility of Seller or attributable to any Tax asset of Seller;
- (k) any refunds due to Seller by a third party for any overpayment of rentals, royalties, excess royalty interests or production payments attributable to the Oil and Gas Assets with respect to any period of time prior to the Closing Date;
 - (1) subject to <u>Section 8.8</u>, all insurance policies and rights to proceeds thereof;
- (m) all prepayments, good faith and other bid deposits submitted by any third party under the terms of the Bidding Procedures Order;
- (n) all claims, refunds, abatements, variances, allocations, causes of action, claims for relief, choses in action, rights of recovery, audit rights, rights of set-off, rights of indemnity, contribution or recoupment, counter-claims, cross-claims and defenses of Seller other than those constituting Oil and Gas Assets or Assumed Liabilities;
- (o) all rights, claims or causes of action by or in the right of Seller against any current or former director or officer of Seller;
 - (p) the Avoidance Actions;

- (q) all claims and causes of action of Seller (i) arising from acts, omissions, or events, or damage to or destruction of property occurring prior to the Closing Date, or (ii) affecting any of the other Excluded Assets:
- (r) any documents or records withheld or not transferred pursuant to the exceptions set forth in Section 2.1(b)(ix), including all documents and instruments of Seller and its Affiliates that are protected by attorney-client privilege;
 - (s) any and all claims for relief of Seller under chapter 5 of the Bankruptcy Code; and
- (t) any rights, claims or causes of action of Seller under this Agreement or any other Transaction Document.

2.3 Assumed Liabilities.

Upon the terms and subject to the conditions of this Agreement, on the Closing Date, Buyer shall assume and agree to discharge, when due (in accordance with their respective terms and subject to the respective conditions thereof), only the following Liabilities (collectively, the "Assumed Liabilities") and no others:

- (a) <u>Assigned Contracts</u>. All of the Seller's Liabilities under the Assigned Contracts, to the extent such Liabilities are attributable to periods from and after the Closing Date.
- (b) <u>Properties</u>. All of the Seller's Liabilities under the Properties (including all plugging and abandonment obligations, and all Property Costs (other than any Excluded Liabilities)), to the extent such Liabilities are attributable to periods from and after the Closing Date.
- (c) <u>Assigned Surface Interests</u>. All of the Seller's Liabilities under the Assigned Surface Interests, to the extent such Liabilities are attributable to periods from and after the Closing Date.
 - (d) Cure Costs. All Cure Costs.
- (e) <u>Imbalances</u>. All over-production and under-delivery Imbalances with respect to the Properties, including those arising as of the Closing Date, as disclosed on <u>Disclosure Schedule 5.7</u>.
- (f) <u>Buyer Taxes</u>. Asset Taxes, to the extent that they are the responsibility of Buyer pursuant to <u>Section 8.1(b)</u>.
 - (g) Transfer Taxes. All Transfer Taxes.
 - (h) Assumed Accounts Payables. All Assumed Accounts Payables.
- (i) Other Oil and Gas Assets. To the extent not already described in Sections 2.3(a) through (h) above, all Liabilities arising from, related to, or associated with the Oil and Gas Assets, to the extent such Liabilities are attributable to periods from and after the Closing Date.

The assumption by Buyer of the Assumed Liabilities shall not, in any way, enlarge the rights of any third parties relating thereto.

2.4 Excluded Liabilities.

Notwithstanding any provision in this Agreement to the contrary, Buyer shall not assume and shall not be obligated to assume or be obliged to pay, perform or otherwise discharge any Liability of Seller, and Seller shall be solely and exclusively liable with respect to all Liabilities of Seller, other than the Assumed Liabilities (such Liabilities other than Assumed Liabilities, collectively, the "Excluded Liabilities"). For purposes of clarity, and without limitation of the generality of the foregoing, the Excluded Liabilities shall include, without limitation, each of the following Liabilities of Seller, other than the Assumed Liabilities:

- (a) all indebtedness for borrowed money of Seller;
- (b) all guarantees of third party obligations by Seller and reimbursement obligations to guarantors of Seller's obligations or under letters of credit;
- (c) all accrued expenses and accounts payables, other than the Assumed Accounts Payables;
- (d) Asset Taxes, to the extent that they are the responsibility of Seller pursuant to Section 8.1(b);
 - (e) those Actions and Proceedings set forth on <u>Disclosure Schedule 5.14</u>;
- (f) all Liabilities of Seller to any owner or former owner of capital stock or warrants, or holder of indebtedness for borrowed money;
 - (g) drafts or checks outstanding at the Closing; and
 - (h) any claims to the extent related to the Excluded Assets.

2.5 Assigned Contracts and Cure Costs.

- (a) At the Closing (or, if the applicable Assigned Contract or Assigned Lease and Interest is not assigned to Buyer at Closing pursuant to the terms of Section 2.6 or Section 8.8, then at the time of the assignment of such Assigned Contract or Assigned Lease and Interest to Buyer, if ever), Buyer shall pay, pursuant to Section 365 of the Bankruptcy Code and the Sale Order, any and all cure and reinstatement costs or expenses that are required to be paid under sections 365(b)(1)(A) and (B) of the Bankruptcy Code in connection with the assumption and assignment of the Assigned Contracts and Assigned Leases and Interests (the "Cure Costs"). For the avoidance of doubt, the payment of any Cure Costs called for by this Agreement shall be made by Buyer in cash at Closing and shall be in addition to the Purchase Price but in no event shall Buyer be required to make any payment of Cure Costs for, and shall not assume any Liabilities with respect to, any Contract that is not an Assigned Contract.
- (b) Schedule 2.5(b) sets forth each Executory Contract and Seller's good faith estimate of the amount of the Cure Costs payable in respect of each such Executory Contract (and if no Cure Cost is estimated to be payable in respect of any particular Executory Contract, the amount of such Cure Cost designated for such Contract shall be "\$0.00").
- (c) Ten (10) Business Days after Seller has made available to Buyer correct and complete copies of all Executory Contracts listed on <u>Schedule 2.5(b)</u>, and subject to Buyer's rights under <u>Section 2.5(e)</u> below to subsequently amend such designations, Buyer will deliver to Seller (i) an <u>Exhibit</u> <u>C</u>, to be determined in Buyer's sole discretion, which shall be a schedule of the Executory Contracts to be

assumed by Seller and assigned to Buyer (as Assigned Contracts) at the Closing, and (ii) a <u>Schedule 2.2(h)</u>, to be determined in Buyer's sole discretion, which shall be a schedule of the Executory Contracts to be rejected by Seller (as Excluded Contracts). The Seller shall provide sufficient notice under the Bankruptcy Code and local rules of the Bankruptcy Court to all counterparties to the Assigned Contracts of their assumption or rejection and, with respect to the Assigned Contracts to be assumed, also provide a schedule of Cure Costs.

- (d) On the date that is no later than five (5) days prior to the Sale Hearing Date, the Seller shall deliver a written notice in a form reasonably acceptable to Buyer of the proposed assignments of the Executory Contracts and the proposed Cure Costs for each Executory Contract to all non-debtor parties of Executory Contracts, which notice shall notify each non-debtor party of (i) the proposed Cure Cost for such Executory Contract and (ii) an objection deadline for such non-debtor party to object to the proposed Cure Cost.
- (e) At any time prior to the Designation Deadline, Buyer shall have the right, which may be exercised in Buyer's sole discretion, to provide written notice to Seller (each such notice, a "Contract Notice") of Buyer's election to designate any Executory Contract (including any Contract that is an Assigned Contract immediately before such designation) (1) as an Excluded Contract, and upon such designation such Contract shall constitute an Excluded Contract and, if applicable, shall cease to constitute an Assigned Contract or (2) to the extent not already rejected, as an Assigned Contract and upon such designation such Contract shall constitute an Assigned Contract and shall cease to constitute an Excluded Contract. If, at any time after the Designation Deadline, the Cure Costs fixed by the Bankruptcy Court for any Assigned Contract are (i) greater than the amount set forth on the initial schedule of Cure Costs filed by Seller with the Bankruptcy Court prior to the Designation Deadline and (ii) are not consented to by Buyer, then Buyer shall be permitted, no later than two (2) Business Days after entry of an order by the Bankruptcy Court setting such Cure Costs, to provide Seller a Contract Notice of Buyer's election to revoke its designation of any such Contract as an Assigned Contract and thereupon such Contract shall be deemed to be an Excluded Contract for all purposes of this Agreement. The Parties shall amend Schedule 2.2(h) and Exhibit C to reflect changes made pursuant to this Section 2.5(e).
- (f) If Buyer exercises its rights in <u>Section 2.5(e)</u> above to designate a Contract, including a Contract that was an Assigned Contract immediately before such designation, as an Excluded Contract, there shall be no reduction in the Purchase Price as a result of such designation or change in designation.

2.6 Assignment of Oil and Gas Assets Subject to Consent Requirements.

(a) Within three (3) Business Days after the Sale Order is entered, Seller shall send letters seeking any consent to assignment (including Hard Consents) or approval rights applicable to the transactions contemplated hereby. Subject to Section 2.6(b), this Agreement shall not constitute an agreement to assign or transfer and shall not effectuate the assignment or transfer of any Oil and Gas Asset if such Oil and Gas Asset is subject to an un-obtained Hard Consent. Seller and Buyer shall use their commercially reasonable efforts to obtain all such Hard Consents; provided, however, neither Seller nor Buyer will be obligated to pay any consideration or initiate any Proceedings to obtain any such Hard Consent. Subject to Section 2.6(b), each Oil and Gas Asset affected by a Hard Consent that is not obtained prior to Closing shall, at the option of Buyer, be held back from the Oil and Gas Assets conveyed at Closing (without any adjustment to the Purchase Price), subject to the following provisions of this Section 2.6(a). If any of the Oil and Gas Assets is not conveyed at Closing due to failure to obtain a Hard Consent, then (1) Seller shall, following the Closing, at Buyer's sole expense and subject to any approval of the Bankruptcy Court that may be required, reasonably cooperate with Buyer to provide to Buyer the benefits thereof in some other manner (including the exercise of the rights of Seller thereunder or the subcontracting,

sub licensing, or sub-leasing thereof) and (2) such Oil and Gas Asset shall be conveyed to Buyer within ten (10) Business Days after such Hard Consent has been obtained, waived or otherwise satisfied or after Buyer otherwise so requests, whichever occurs earlier. Except for Hard Consents, if any consents to the assignment of any Oil and Gas Asset are not obtained prior to Closing, such Oil and Gas Asset shall nevertheless be sold and conveyed to Buyer at the Closing. Notwithstanding the foregoing, nothing in this Section 2.6 shall (x) require Seller to make any expenditure or incur any obligation on its own or on behalf of Buyer, or (y) prohibit Seller from ceasing operations or winding up its affairs following the Closing.

(b) The provisions of Section 2.6(a) shall only apply with respect to those Oil and Gas Assets (i) that are subject to any consent to assignment (including Hard Consents) or approval rights applicable to the transactions contemplated hereby and (ii) which cannot be acquired by Buyer free and clear of such consent to assignment or approval right pursuant to the Bidding Procedures Order, the Sale Order or the Bankruptcy Code.

2.7 Further Assurances.

The Parties agree to (a) furnish upon request to each other such further information, (b) execute, acknowledge and deliver to each other such other documents, and (c) do such other acts and things, all as the other Party may reasonably request for the purpose of carrying out the intent of this Agreement and the Transaction Documents; *provided*, that Seller's covenants and obligations after Closing shall be limited to the extent Seller has employees and resources available for such purposes and *provided*, *further* that nothing in this Section 2.7 or anywhere else in this Agreement shall prohibit Seller from ceasing operations or winding up its affairs following the Closing.

ARTICLE 3

PURCHASE PRICE

3.1 Purchase Price.

The purchase price for the purchase, sale, assignment and conveyance of Seller's right, title and interest in, to and under the Oil and Gas Assets shall consist of the following (collectively, the "Base Purchase Price"):

- (a) cash in an amount equal to Three Million Four Hundred Thousand Dollars (\$3,400,000.00); and
 - (b) the assumption of the Assumed Liabilities.

Notwithstanding the foregoing, the Base Purchase Price shall be adjusted as provided in <u>Section 8.10</u> hereof (as adjusted, the "<u>Purchase Price</u>"). The cash components of the Purchase Price shall be delivered by Buyer as set forth in <u>Section 4.2</u>.

3.2 Good Faith Deposit.

(a) As required by the terms of the Bidding Procedures Order, Buyer has submitted to Seller via a wire transfer of immediately available funds (the "Good Faith Deposit") an amount equal to ten percent (10%) of the cash portion of the Base Purchase Price. Within two (2) Business Days after the Sale Order has been entered by the Bankruptcy Court, the Good Faith Deposit shall be held and disbursed by Seller in accordance with the terms of this Agreement.

- (b) If the Closing of the sale of the Oil and Gas Assets by Seller to Buyer pursuant to this Agreement is not consummated because of a breach of this Agreement by Buyer resulting in termination pursuant to Section 11.1(c)(i), then, in such event, Seller shall have the right to receive the Good Faith Deposit, free of any claims by Buyer thereto, as liquidated damages due to the difficulty of ascertaining actual damages with any certainty and, to the fullest extent permitted by applicable Legal Requirements, the Good Faith Deposit shall be Seller's sole and exclusive remedy with respect to such breach by Buyer notwithstanding any provision to the contrary contained in Article 11 or elsewhere in this Agreement.
- (c) If this Agreement is terminated for any reason other than a breach of this Agreement by Buyer resulting in termination pursuant to Section 11.1(c)(i), Buyer shall be reimbursed the Good Faith Deposit, together with any interest or other earnings thereon. The Parties shall cause the reimbursement of the Good Faith Deposit to be delivered to Buyer within five (5) Business Days of such termination, free of any claims by Seller with respect thereto.

ARTICLE 4

CLOSING

4.1 <u>Closing Date</u>.

Upon the terms and subject to the conditions hereof, the closing of the sale of the Oil and Gas Assets and the assumption of the Assumed Liabilities contemplated hereby (the "Closing") shall take place at 10:00 a.m., Central Standard Time, at the offices of Bracewell LLP, 711 Louisiana Street, Suite 2300, Houston, Texas 77002, on the date that the Parties mutually agree upon in writing that is (a) September 30, 2019, (b) three (3) Business Days following the date on which the conditions set forth in Article 9 and Article 10 have been satisfied or, if permissible, waived, excepting the conditions which by their nature are to be satisfied at the Closing, but subject to the satisfaction or, if permissible, waiver of such conditions, or (c) such other date prior to September 30, 2019. The date and time at which the Closing actually occurs is hereinafter referred to as the "Closing Date."

4.2 Payment on the Closing Date.

Subject to satisfaction or, if permissible, waiver of the conditions set forth in <u>Article 9</u> and <u>Article 10</u>, at the Closing, Buyer shall pay, or cause to be paid, the cash component of the Purchase Price, *less* the amount of the Good Faith Deposit, by wire transfer of immediately available funds to an account specified in writing by the Seller prior to the Closing Date.

4.3 <u>Buyer's Deliveries</u>.

At the Closing, Buyer shall deliver or cause to be delivered to Seller (or such other Persons where so designated):

- (a) the cash consideration referenced in <u>Section 3.1(a)</u> to the Seller in accordance with Section 4.2;
 - (b) the Cure Costs in accordance with Section 2.5(a);
- (c) the deed, duly executed by Buyer, substantially in the form attached hereto as **Exhibit E**;

- (d) the Assignment, the Mineral Deed, and each other Transaction Document to which Buyer is a party, duly executed (and acknowledged, where applicable) by Buyer, including the Assignment, letters-in-lieu of transfer orders to be prepared by Buyer, change of operator forms to be prepared by Seller, any applicable change of operator notices required under applicable operating agreements, and any other applicable forms and declarations required by federal and state agencies relative to Buyer's assumption of operations and plugging and abandonment Liabilities with respect to all of the Oil and Gas Assets;
- (e) the certificates of Buyer to be received by Seller pursuant to $\underline{\text{Sections } 10.1}$ and $\underline{10.3}$;
- (f) a certificate duly executed by an authorized officer of Buyer, dated as of the Closing, (i) attaching and certifying on behalf of Buyer complete and correct copies of the resolutions of the Board of Directors or other equivalent governing body of Buyer authorizing the execution, delivery, and performance by Buyer of this Agreement and the transactions contemplated hereby, which resolutions or consent shall be dated prior to the date of this Agreement, and (ii) certifying on behalf of Buyer the incumbency of each officer of Buyer executing this Agreement or any document delivered in connection with the Closing;
 - (g) a counterpart of the Preliminary Settlement Statement executed by Buyer;
- (h) evidence (including evidence of satisfaction of all applicable bonding or insurance requirements) as Seller may reasonably request demonstrating that Buyer (or its designees) is qualified with the applicable Governmental Authorities and pursuant to any applicable operating agreement to succeed Seller as the owners and, where applicable, the operator of the Oil and Gas Assets; and
- (i) such other assignments and other good and sufficient instruments of assumption and transfer, in form reasonably satisfactory to Seller, as Seller may reasonably request to transfer and assign the Assumed Liabilities to Buyer.

4.4 Seller's Deliveries.

At the Closing, Seller shall deliver to Buyer:

- (a) the Assignment, the Mineral Deed, and each other Transaction Document to which Seller is a party (including any applicable change of operator forms), duly executed by the Seller;
- (b) the deed, duly executed by Seller, substantially in the form attached hereto as $\mathbf{E}\mathbf{x}\mathbf{h}\mathbf{i}\mathbf{b}\mathbf{i}\mathbf{t}$;
 - (c) a certified copy of the Sale Order;
 - (d) the certificates of Seller to be received by Buyer pursuant to <u>Sections 9.1</u> and <u>9.2</u>;
- (e) a certificate duly executed by an authorized officer of Seller, dated as of the Closing, (i) attaching and certifying on behalf of Seller complete and correct copies of the resolutions or unanimous consent of the Board of Directors, managers, partners, or other equivalent governing body of Seller authorizing the execution, delivery, and performance by Seller of this Agreement and the transactions contemplated hereby, which resolutions or consent shall be dated prior to the date of this Agreement, and (ii) certifying on behalf of Seller the incumbency of each officer of Seller executing this Agreement or any document delivered in connection with the Closing;

- (f) a non-foreign affidavit by Seller dated as of the Closing Date, sworn under penalty of perjury and in form and substance required under the Treasury Regulations issued pursuant to Code §1445, stating that Seller is not a "foreign person" as defined therein;
- (g) duly executed and acknowledged releases of all mortgages, deeds of trust, and security agreements created by Seller that encumber the Oil and Gas Assets and releases of any related financing statements;
 - (h) a counterpart of the Preliminary Settlement Statement executed by Seller;
- (i) such other bills of sale, deeds, endorsements, assignments and other good and sufficient instruments of conveyance and transfer, in form reasonably satisfactory to Buyer, as Buyer may reasonably request to vest in Buyer all the right, title and interest of Seller in, to or under any or all the Oil and Gas Assets; and
- (j) subject to <u>Section 7.3(d)</u>, executed originals of Texas Railroad Commission Form P-4 naming Buyer (or its designee) as Operator of the Wells as of Closing Date and Form T-4s naming Buyer (or its designee) as operator of the pipelines that comprise a portion of the Oil and Gas Assets.

ARTICLE 5

REPRESENTATIONS AND WARRANTIES OF SELLER

Except as disclosed in the Disclosure Schedules attached hereto, Seller represents and warrants the following to Buyer:

5.1 Organization and Good Standing.

Seller is an entity duly organized, validly existing and in good standing under the laws of the jurisdiction of its organization. Seller has the requisite partnership power and authority to own or lease and to operate and use its properties and to carry on its business as now conducted. Seller is duly qualified or licensed to do business and is in good standing in each jurisdiction where the character of its business or the nature of its properties makes such qualification or licensing necessary, except for such failures to be so qualified or licensed or in good standing as would not, individually or in the aggregate, have a Material Adverse Effect.

5.2 <u>Authority; Validity; Consents.</u>

Seller has, subject to requisite Bankruptcy Court approval, the requisite partnership power and authority necessary to enter into and perform its obligations under this Agreement and the other Transaction Documents to which Seller is a party and to consummate the transactions contemplated hereby and thereby, and, subject to requisite Bankruptcy Court approval, the execution, delivery and performance of this Agreement and such other Transaction Documents by Seller and the consummation by Seller of the transactions contemplated herein and therein have been duly and validly authorized by all requisite partnership action. This Agreement has been duly and validly executed and delivered by Seller and each other Transaction Document required to be executed and delivered by Seller at the Closing will be duly and validly executed and delivered by Seller at the Closing. Subject to requisite Bankruptcy Court approval, this Agreement and the other Transaction Documents which Seller is party thereto constitute the legal, valid and binding obligations of Seller, enforceable against Seller in accordance with their respective terms, except as such enforceability is limited by general principles of equity. Subject to requisite Bankruptcy Court approval, to Seller's Knowledge, except (a) for entry of the Sale Order, (b) for notices, filings and

consents required in connection with the Bankruptcy Case and (c) for the notices, filings and consents set forth on <u>Disclosure Schedule 5.2</u>, Seller is not required to give any notice to, make any filing with or obtain any consent from any Person (including any Governmental Authority) in connection with the execution and delivery of this Agreement and the other Transaction Documents or the consummation or performance of any of the transactions contemplated hereby and thereby, except as would not, individually or in the aggregate, have a Material Adverse Effect.

5.3 No Conflict.

When the consents and other actions described in <u>Section 5.2</u>, including requisite Bankruptcy Court approval, have been obtained and taken, to Seller's Knowledge, the execution and delivery of this Agreement and the other Transaction Documents and the consummation of the transactions provided for herein and therein will not result in the breach of any of the terms and provisions of, or constitute a default under, or conflict with, or cause any acceleration of any obligation of Seller under (a) any agreement, indenture, or other instrument to which Seller is bound, (b) the partnership agreement, bylaws or other governing documents of Seller, (c) any Order or (d) any Legal Requirement, except as would not, individually or in the aggregate, have a Material Adverse Effect.

5.4 <u>Material Assigned Contracts</u>.

To Seller's Knowledge, (a) <u>Disclosure Schedule 5.4</u> lists all Material Assigned Contracts in effect as of the Effective Time, to which Seller is a party or by which its interests in the Oil and Gas Assets are bound, (b) all Material Assigned Contracts are in full force and effect, except (i) as such enforceability may be limited by bankruptcy, insolvency, reorganization, moratorium or similar laws now or hereafter in effect relating to creditors' rights generally or general principles of equity and (ii) as would not, individually or in the aggregate, have a Material Adverse Effect, and (c) except as may result from the Bankruptcy Case, no default or breach (or event that, with notice or lapse of time, or both, would become a default or breach) of any such Material Assigned Contracts has occurred or is continuing on the part of Seller.

5.5 Permits.

To Seller's Knowledge, as of the Effective Time, all necessary Permits with regard to the ownership or operation of the Oil and Gas Assets have been obtained and maintained in effect and no violations exist in respect of such Permits, except for such non-compliance, failure to obtain or maintain, and such facts, conditions or circumstances, the existence of which would not constitute a Material Adverse Effect.

5.6 Wells; Plug and Abandon Notice.

To Seller's Knowledge, except as set forth on <u>Disclosure Schedule 5.6</u>, there are no Wells (a) in respect of which Seller has received an order from any Governmental Authority requiring that such Wells be plugged and abandoned, or (b) that are neither in use for purposes of production or injection, nor suspended or temporarily abandoned in accordance with applicable Legal Requirements, that are required to be plugged and abandoned in accordance with applicable Legal Requirements but have not been plugged and abandoned.

5.7 <u>Imbalances</u>.

To Seller's Knowledge, all Imbalances relating to the Oil and Gas Assets are reflected in <u>Disclosure Schedule 5.7</u> as of the date stated therein.

5.8 AFEs.

<u>Disclosure Schedule 5.8</u> contains a list, true and correct as of the date set forth therein, of all material authorities for expenditures (collectively, "<u>AFEs</u>") for capital expenditures with respect to the Oil and Gas Assets that have been proposed by any Person having authority to do so (including internal AFEs of Seller not delivered to third parties) or other commitment to make expenditures in respect of the ownership or operation of the Oil and Gas Assets.

5.9 Non-Consent Operations.

Except as set forth on <u>Disclosure Schedule 5.9</u>, to Seller's Knowledge, no operations are being conducted or have been conducted on the Oil and Gas Assets with respect to which Seller has elected to be a non-consenting party under the applicable operating agreement and with respect to which all of Seller's rights have not yet reverted to it.

5.10 <u>Hedging</u>.

There are no futures, options, swaps or other derivatives with respect to the sale of Hydrocarbons from the Oil and Gas Assets that are or will be binding on the Oil and Gas Assets at any time after the Closing Date.

5.11 Preferential Purchase Rights.

To Seller's Knowledge, <u>Disclosure Schedule 5.11</u> lists all Preferential Purchase Rights applicable to the transactions contemplated by this Agreement.

5.12 Intellectual Property.

To Seller's Knowledge, except as set forth on <u>Disclosure Schedule 5.12</u>, Seller does not own any Intellectual Property related to or used in connection with the ownership or operation of the Oil and Gas Assets.

5.13 Taxes.

Seller has (a) timely filed or caused to be filed on its behalf all material Tax Returns required to be filed by it in connection with the Asset Taxes on or prior to the date hereof (taking into account any valid extension of the due date for filing) and (b) timely paid all material Asset Taxes shown as due and payable on such Tax Returns. No material examination of any such Tax Return required to be filed by Seller in connection with the Asset Taxes is currently in progress by any Governmental Authority. With respect to any such Tax Returns required to be filed by Seller in connection with the Asset Taxes, no material adjustment has been proposed in writing for the last five (5) fiscal years by any Governmental Authority.

5.14 <u>Legal Proceedings</u>.

Except for the Bankruptcy Case and as set forth on <u>Disclosure Schedule 5.14</u>, there is no Proceeding or Order pending, outstanding or, to Seller's Knowledge, threatened against Seller that seeks to restrain or prohibit or otherwise challenge the consummation, legality or validity of the transactions contemplated hereby or that would have, individually or in the aggregate, a Material Adverse Effect.

5.15 <u>Sufficiency of Oil and Gas Assets.</u>

Except as set forth on <u>Disclosure Schedule 5.15</u>, to Seller's Knowledge, the Oil and Gas Assets, together with Seller's agreements under this Agreement and the other Transaction Documents, will, as of the Closing Date, constitute all of the material assets and rights necessary for Buyer to own the Properties substantially as presently owned by Seller.

5.16 Compliance with Laws.

Except as set forth on <u>Disclosure Schedule 5.16</u> and as may result from the Bankruptcy Case, from and after the date that is twelve (12) months prior to the date of execution of this Agreement (a) Seller has not received written notice from any Governmental Authorities alleging a violation of Environmental Laws relating to the Properties, and to Seller's Knowledge, no such notice is pending, and (b) to Seller's Knowledge, except where lack of compliance would not have a Material Adverse Effect, the Properties are in compliance with all Legal Requirements other than Environmental Laws.

5.17 Equipment.

To Seller's Knowledge, except as would not have a Material Adverse Effect, all of the Equipment is in a state of reasonable repair (ordinary wear and tear excepted) so as to be suitable for the purposes of which such equipment was constructed, obtained or is currently being used in all material respects.

5.18 Brokers or Finders.

Seller has not incurred any obligation or liability, contingent or otherwise, for brokerage or finders' fees or agents' commissions or other similar payment in connection with this Agreement, the other Transaction Documents or the transactions contemplated hereby or thereby for which Buyer is or will become liable, and Seller shall indemnify and hold harmless Buyer from any claims with respect to any such fees or commissions, in each case except to the extent that such fees, commissions and other similar payments constitute Assumed Liabilities.

5.19 <u>Title Matters</u>.

Except as set forth on <u>Disclosure Schedule 5.19</u>, to Seller's Knowledge, Seller holds Defensible Title to each of the Oil and Gas Assets and, subject to the entry of the Sale Order, at the Closing, Buyer shall be vested with Defensible Title to the Oil and Gas Assets.

5.20 <u>Knowledge Qualifier for Non-Operated Oil and Gas Assets.</u>

To the extent that Seller has made any representations or warranties in this <u>Article 5</u> in connection with matters relating to Oil and Gas Assets for which Seller is not the operator, each and every such representation and warranty shall be deemed to be qualified by the phrase "To Seller's Knowledge".

5.21 <u>Bankruptcy</u>.

From the Petition Date through the date that this Agreement is executed, Seller has been at all times in its Bankruptcy Case a debtor in possession pursuant to Section 1107 of the Bankruptcy Code, and no trustee or examiner has been appointed in the Bankruptcy Case.

ARTICLE 6

REPRESENTATIONS AND WARRANTIES OF BUYER

Buyer represents and warrants to Seller as follows:

6.1 <u>Organization and Good Standing.</u>

Buyer is a limited liability company, duly organized, validly existing and in good standing under the laws of the State of Delaware. Buyer has the requisite power and authority to own or lease and to operate and use its Properties and to carry on its business as now conducted.

6.2 Authority; Validity; Consents.

Buyer has the requisite power and authority necessary to enter into and perform its obligations under this Agreement and the other Transaction Documents to which it is a party and to consummate the transactions contemplated hereby and thereby. The execution, delivery and performance of this Agreement by Buyer and the consummation by Buyer of the transactions contemplated herein have been duly and validly authorized by all requisite limited liability company or corporate actions in respect thereof. This Agreement has been duly and validly executed and delivered by Buyer and each other Transaction Document to which Buyer is a Party will be duly and validly executed and delivered by Buyer, as applicable, at the Closing. This Agreement and the other Transaction Documents to which Buyer is a party constitute the legal, valid and binding obligation of Buyer, enforceable against Buyer in accordance with their respective terms, except in each case as such enforceability is limited by bankruptcy, insolvency, reorganization, moratorium or similar laws now or hereafter in effect relating to creditors' rights generally or general principles of equity. Buyer is not or will not be required to give any notice to or obtain any consent from any Person in connection with the execution and delivery of this Agreement and the other Transaction Documents to which it is a Party or the consummation or performance of any of the transactions contemplated hereby or thereby, except for such notices, filings and consents, the failure of which to provide, make or obtain, would not, individually or in the aggregate, affect Buyer's ability to perform its obligations under this Agreement or any other Transaction Documents or to consummate the transactions contemplated hereby or thereby.

6.3 No Conflict.

When the consents and other actions described in <u>Section 6.2</u> have been obtained and taken, the execution and delivery of this Agreement and the other Transaction Documents and the consummation of the transactions provided for herein and therein will not result in the breach of any of the terms and provisions of, or constitute a default under, or conflict with, or cause any acceleration of any obligation of Buyer under (a) any agreement, indenture, or other instrument to which it is bound, (b) the partnership agreement, bylaws or other governing documents of Buyer, as applicable, (c) any Order or (d) any Legal Requirement.

6.4 <u>Availability of Funds</u>.

As of the Closing, Buyer will have sufficient cash in immediately available funds (without giving effect to any unfunded financing, regardless of whether any such financing is committed) to pay the Purchase Price, all costs, fees and expenses to be paid by Buyer that are necessary to consummate the transactions contemplated by this Agreement and the other Transaction Documents, and assume the Assumed Liabilities.

6.5 Litigation.

There are no Proceedings pending or, to the Knowledge of Buyer, threatened, that would affect Buyer's ability to perform its obligations under this Agreement or any other Transaction Documents or to consummate the transactions contemplated hereby or thereby.

6.6 Brokers or Finders.

Neither Buyer nor any Person acting on behalf of Buyer has paid or become obligated to pay any fee or commission to any broker, finder, investment banker, agent or intermediary for or on account of the transactions contemplated by this Agreement for which Seller is or will become liable, and Buyer shall hold harmless and indemnify Seller from any claims with respect to any such fees or commissions.

6.7 Business Use, Bargaining Position, Representation.

Buyer is purchasing the Oil and Gas Assets for commercial or business use and has knowledge and experience in financial and business matters that enables it to evaluate the merits and the risks of a transaction such as this. Buyer is not in a significantly disparate bargaining position with Seller and is represented by legal counsel.

6.8 <u>Independent Investigation</u>.

Buyer is (or its advisors are) experienced and knowledgeable in the oil and gas business and aware of the risks of that business. Buyer acknowledges and affirms that (a) it has completed such independent investigation, verification, analysis, and evaluation of the Oil and Gas Assets and has made all such reviews and inspections of the Oil and Gas Assets as it has deemed necessary or appropriate to enter into this Agreement, and (b) prior to Closing, it will have completed its independent investigation, verification, analysis, and evaluation of the Oil and Gas Assets and made all such reviews and inspections of the Oil and Gas Assets as it deems necessary or appropriate to consummate the transactions contemplated hereby. Except for the representations and warranties expressly made by Seller in Article 5 of this Agreement, the Assignment, the Mineral Deed, the deed to be delivered to Buyer pursuant to Section 4.4(b) of this Agreement, or the certificate to be delivered to Buyer pursuant to Section 4.4(d) of this Agreement, Buyer acknowledges that there are no representations or warranties, express or implied, as to the financial condition, Liabilities, operations, business, or prospects of the Oil and Gas Assets and that, in making its decision to enter into this Agreement and to consummate the transactions contemplated hereby, Buyer has relied solely upon its own independent investigation, verification, analysis, and evaluation. Buyer understands and acknowledges that neither the United States Securities and Exchange Commission nor any federal, state, or foreign agency has passed upon the Oil and Gas Assets or made any finding or determination as to the fairness of an investment in the Oil and Gas Assets or the accuracy or adequacy of the disclosures made to Buyer, and, except as set forth in Article 11, Buyer is not entitled to cancel, terminate or revoke this Agreement.

6.9 Bankruptcy.

There are no bankruptcy, reorganization or receivership proceedings pending, being contemplated by or, to Buyer's Knowledge, threatened in writing against Buyer or any Affiliate of Buyer. Buyer is not (and will not be upon consummation of the transactions contemplated hereby) insolvent.

ARTICLE 7

ACTIONS PRIOR TO THE CLOSING DATE

7.1 Access and Reports.

- (a) Subject to applicable Legal Requirements, upon receipt of written notice from Buyer of any such activities no less than two (2) Business Days in advance, Seller shall (and shall cause its Affiliates to) afford Buyer's authorized Representatives reasonable access, during normal business hours until the date that is five (5) Business Days prior to the scheduled Closing Date, to its employees, properties, books, Contracts and Records, and, during such period, Seller shall furnish promptly to Buyer all information concerning the Oil and Gas Assets as may reasonably be requested; *provided, however*, such access shall not interfere with the ordinary conduct of business or the operation of the Oil and Gas Assets and, unless Seller provides written consent otherwise, at all times during such access, Buyer's authorized Representatives shall be accompanied by at least one Representative of Seller. All requests for information made pursuant to this Section 7.1 shall be submitted in accordance with Section 13.4. All such information shall be governed by the terms of the Confidentiality Agreement. No investigation pursuant to this Section 7.1 or by Buyer or its Representatives at any time prior to or following the date hereof shall affect or be deemed to modify any representation or warranty made by the Seller herein.
- From and after the execution of this Agreement until the date that is five (5) Business Days prior to the scheduled Closing Date, Buyer shall have the right, at its sole cost, risk, liability, and expense, to conduct a Phase I Environmental Site Assessment of the Oil and Gas Assets. During the Seller's regular hours of business and after providing the Seller with written notice of any such activities no less than two (2) Business Days in advance (which written notice shall include the written permission of the operator (if other than a Seller) and any other third party whose permission is legally required, which Seller shall reasonably cooperate with Buyer in securing), Buyer and its authorized Representatives shall be permitted to enter upon the Oil and Gas Assets, inspect the same, review all of Seller's files and records (other than those for which Seller has an attorney-client privilege) relating to the Oil and Gas Assets, and generally conduct visual, non-invasive tests, examinations, and investigations; provided, however, that such entry shall not interfere with the ordinary conduct of business or operation of the Oil and Gas Assets and, unless Seller provides written consent otherwise, at all times during such entry, Buyer's authorized Representatives shall be accompanied by at least one Representative of Seller. No sampling or other invasive inspections of the Oil and Gas Assets may be conducted without the Seller's prior written consent, which shall not be unreasonably withheld. The withholding of consent to sampling or any other invasive inspection by the Seller shall be deemed reasonable if (without limitation) the need for such sampling or invasive testing was not indicated from a visual inspection or, based on the Phase I Environmental Assessment. Seller will have the right, which it may exercise at its sole discretion, to (i) observe such investigation, and (ii) promptly receive a copy of all results, analyses, reports, and reviews, except for such information for which Buyer has an attorney-client privilege. All information obtained or reviewed by Buyer shall be maintained confidential by Buyer and shall be governed by the terms of the Confidentiality Agreement.
- (c) This Section 7.1 shall not require Seller to permit any access to, or to disclose (i) any information that, in the reasonable, good faith judgment (after consultation with counsel, which may be in-house counsel) of the Seller, is reasonably likely to result in any violation of any Legal Requirement or any Contract to which the Seller is a party or cause any privilege (including attorney-client privilege) that Seller would be entitled to assert to be undermined with respect to such information and such undermining of such privilege could in the Seller's good faith judgment (after consultation with counsel, which may be in-house counsel) adversely affect in any material respect Seller's position in any pending or, what the Seller believes in good faith (after consultation with counsel, which may be in-house counsel)

could be, future litigation or (ii) if the Seller, on the one hand, and Buyer or any of its Affiliates, on the other hand, are adverse parties in a litigation, any information that is reasonably pertinent thereto; *provided*, that, in the case of clause (i), the Parties shall cooperate in seeking to find a way to allow disclosure of such information to the extent doing so (A) would not (in the good faith belief of the Seller (after consultation with counsel, which may be in-house counsel)) be reasonably likely to result in the violation of any such Legal Requirement or Contract or be reasonably likely to cause such privilege to be undermined with respect to such information or (B) could reasonably (in the good faith belief of the Seller (after consultation with counsel, which may be in-house counsel)) be managed through the use of customary "clean-room" arrangements pursuant to which non-employee Representatives of Buyer could be provided access to such information.

- (d) The information provided pursuant to this <u>Section 7.1</u> shall be used solely for the purpose of the transactions contemplated by this Agreement, and such information shall be kept confidential by Buyer and Seller in accordance with, and Buyer and Seller shall otherwise abide by and be subject to the terms and conditions of, the Confidentiality Agreement.
- (e) BUYER HEREBY AGREES TO DEFEND, RELEASE, INDEMNIFY AND HOLD HARMLESS SELLER PARTIES FROM AND AGAINST ANY AND ALL LIABILITIES THAT MAY BE ASSERTED AGAINST ANY SELLER PARTY, BASED UPON INJURY TO PERSON, INCLUDING DEATH, OR TO PROPERTY, ARISING IN ANY MANNER WHATSOEVER FROM ANY INSPECTION BY ANY BUYER PARTY OF THE OIL AND GAS ASSETS AND ACCESS BY ANY BUYER PARTY TO THE OIL AND GAS ASSETS PRIOR TO THE CLOSING DATE, WHETHER OR NOT BASED UPON STRICT LIABILITY OR CAUSED BY THE SOLE OR CONCURRENT NEGLIGENCE (WHETHER ACTIVE OR PASSIVE) OF ANY SELLER PARTY, UNLESS SUCH INJURY WAS OCCASIONED SOLELY BY THE GROSS NEGLIGENCE OR WILLFUL MISCONDUCT OF A SELLER PARTY.

7.2 Operations Prior to the Closing Date.

Seller covenants and agrees that, except (v) as expressly contemplated by this Agreement, (w) as disclosed in <u>Schedule 7.2</u>, (x) with the prior written consent of Buyer (which consent shall not be unreasonably withheld, conditioned or delayed) or (y) as otherwise required by Legal Requirements, after the Effective Time and prior to the Closing Date:

(a) Seller shall:

- (i) pay or cause to be paid all Property Costs with respect to the Oil and Gas Assets for which it receives joint interest billings from the applicable operator of such Oil and Gas Assets;
- (ii) maintain its books, accounts and records in accordance with past custom and practice; and
- (iii) use commercially reasonable efforts, taking into account Seller's status as debtors in possession and as a non-operator, to maintain its relationships with and preserve the goodwill of its key service providers.

(b) Seller shall not:

(i) exercise any election to abandon any Oil and Gas Asset (except any abandonment of Leases to the extent any such Leases terminate pursuant to their terms);

- (ii) propose or agree to participate in any single operation with respect to the Wells or Assigned Leases and Interests with an anticipated cost in excess of Fifty Thousand Dollars (\$50,000), except for emergency operations, operations scheduled under the AFEs, or operations required by any Governmental Authority;
- (iii) terminate, cancel, or materially amend or modify any Assigned Contract or Assigned Lease and Interest;
- (iv) sell, lease, encumber, or otherwise dispose of all or any portion of any Oil and Gas Assets, except sales of Hydrocarbons in the ordinary course of business; or
- (v) enter into any agreement or commitment to take any action prohibited by this Section 7.2(b).

Notwithstanding the foregoing provisions of this <u>Section 7.2</u>, in the event of an emergency, Seller may take such action as reasonably necessary and shall notify Buyer of such action promptly thereafter. Buyer acknowledges that Seller owns undivided interests in the Oil and Gas Assets and is not the operator of the Oil and Gas Assets. The acts or omissions of third Persons (including the applicable operators of the Oil and Gas Assets) who are not controlled Affiliates of Seller shall not constitute a violation of the provisions of this <u>Section 7.2</u>, nor shall any action required by a vote of working interest owners constitute such a violation so long as the Seller has voted its interests in a manner consistent with the provisions of this <u>Section 7.2</u>.

7.3 Reasonable Best Efforts.

- (a) Seller, on the one hand, and Buyer, on the other hand, shall use reasonable best efforts to take, or cause to be taken, all actions, and to do, or cause to be done, and to assist and cooperate with the other in doing, all things necessary, proper or advisable to consummate and make effective, in the most expeditious manner practicable, the transactions contemplated hereby, including using reasonable best efforts to accomplish the following: (i) the taking of all reasonable acts necessary to cause the conditions precedent set forth in <u>Article 9</u> and <u>Article 10</u> to be satisfied, (ii) the obtaining, at the earliest practicable date, of all necessary Governmental Authorizations and the making of all necessary registrations, declarations and filings (including registrations, declarations and filings with Governmental Authorities, if any) and the taking of all reasonable steps as may be necessary to avoid any Proceeding by any Governmental Authority, and (iii) the execution or delivery of any additional instruments necessary to consummate the transactions contemplated hereby and to fully carry out the purposes of this Agreement.
- (b) To the extent required by applicable Legal Requirement, Seller, on the one hand, and Buyer, on the other hand, (i) shall promptly inform each other of any communication from any Governmental Authority concerning this Agreement, the transactions contemplated hereby, and any filing, notification or request for approval and (ii) shall permit the other to review in advance any proposed written or material oral communication or information submitted to any such Governmental Authority in response thereto. In addition, none of Parties shall agree to participate in any meeting with any Governmental Authority in respect of any filings, investigation or other inquiry with respect to this Agreement or the transactions contemplated hereby, unless such Party consults with the other Parties in advance and, to the extent permitted by any such Governmental Authority, gives the other Parties the opportunity to attend and participate in such meeting, in each case to the maximum extent practicable. Subject to any restrictions under applicable laws, rules or regulations, each of Buyer, on the one hand, and Seller, on the other hand, shall furnish the other with copies of all correspondence, filings and communications (and memoranda setting forth the substance thereof) between it and its Affiliates and their respective Representatives on the one hand, and the Governmental Authority or members of its staff on the other hand, with respect to this

Agreement, the transactions contemplated hereby (excluding documents and communications which are subject to preexisting confidentiality agreements or to the attorney-client privilege or work product doctrine) or any such filing, notification or request for approval. Each Party shall also furnish the other Party with such necessary information and assistance as such other Party and its Affiliates may reasonably request in connection with their preparation of necessary filings, registration or submissions of information to the Governmental Authority in connection with this Agreement, the transactions contemplated hereby and any such filing, notification or request for approval.

- (c) Subject to the terms and conditions of this Agreement, Buyer shall take any and all steps necessary to avoid or eliminate any impediments under any applicable antitrust, competition or trade regulation laws, rules or regulations that may be asserted by any Governmental Authority with respect to the transactions contemplated hereby so as to enable the Closing to occur as soon as reasonably possible, including, without limitation, proposing, negotiating, committing to and effecting, by consent decree or otherwise, the sale, divestiture or disposition of such assets or businesses of Buyer or any of its controlled Affiliates as may be required in order to avoid the entry, or to effect the dissolution, of any injunction, temporary restraining order or other order in any suit or proceeding, which would otherwise have the effect of preventing, delaying or restricting the consummation of the transactions contemplated in this Agreement.
- (d) The Parties hereby acknowledge and agree that O'Brien Resources holds record and/or beneficial title to, and is a party to certain contracts relating to, certain Oil and Gas Assets, on behalf of Seller. Prior to Closing, O'Brien Resources covenants and agrees to use commercially reasonable efforts to assign such Oil and Gas Assets and such contracts to Seller. O'Brien Resources further covenants and agrees to relinquish operatorship of those certain Oil and Gas Assets to which Seller owns greater than a Fifty Percent (50%) Working Interest in and to and, to the extent that Seller owns less than a Fifty Percent (50%) Working Interest in and to such Oil and Gas Assets, O'Brien Resources covenants and agrees to use its reasonable best efforts to obtain all ballots and authorizations required for Buyer (or its designee) to be named as operator of record and to transfer operatorship to Buyer (or its designee), for no additional consideration. Buyer acknowledges that Seller cannot and does not covenant or warrant that Buyer (or its designee) shall become successor operator of such Oil and Gas Assets since the Oil and Gas Assets or portions thereof may be subject to operating or other agreements that control the appointment of a successor operator.
- (e) Upon the earlier of (i) payment of all proceeds of production to Seller related to the Oil and Gas Assets which is attributable to the time period between the Effective Time and Closing or (ii) ninety (90) days after the Closing Date, O'Brien Resources shall deliver letters-in-lieu of transfer or division orders to Buyer. To the extent that O'Brien Resources receives revenue from purchasers of production that is attributable to post-Closing time periods, O'Brien Resources shall remit such revenue to which Buyer is entitled within five (5) Business Days after receiving such revenue; *provided*, that Buyer shall pay all expenses attributable to such revenue with respect to post-Closing time periods. To the extent that Buyer receives revenue from purchasers of production that is attributable to pre-Closing time periods, Buyer shall remit to O'Brien Resources such revenue to which Seller and/or O'Brien Resources is entitled within five (5) Business Days after receiving such revenue; *provided*, that O'Brien Resources shall pay all expenses attributable to such revenue with respect to pre-Closing time periods.

7.4 <u>Bankruptcy Court Approval</u>.

(a) Seller and Buyer acknowledge that this Agreement and the sale of the Oil and Gas Assets and the assumption and assignment of the Assigned Contracts and Assigned Leases and Interests are subject to Bankruptcy Court approval. Seller and Buyer acknowledge that (i) to obtain such approval, Seller must demonstrate that it has taken reasonable steps to obtain the highest and otherwise best offer possible for the Oil and Gas Assets, and that such demonstration shall include giving notice of the

transactions contemplated by this Agreement to creditors and other interested parties as ordered by the Bankruptcy Court, and (ii) Buyer must provide adequate assurance of future performance under the to-be-assigned Leases and Executory Contracts.

- (b) In the event an appeal is taken or a stay pending appeal is requested, from the Sale Order, Seller shall promptly notify Buyer of such appeal or stay request and shall provide to Buyer promptly a copy of the related notice of appeal or order of stay. Seller shall also provide Buyer with written notice of any motion or application filed in connection with any appeal from either of such orders.
- (c) From and after the date of execution of this Agreement and prior to the Closing or the termination of this Agreement in accordance with <u>Section 11.1</u>, Seller shall not take any action which is intended to (or is reasonably likely to), or fail to take any action the intent (or the reasonably likely result) of which failure to act is to, result in the reversal, voiding, modification or staying of the Sale Order or this Agreement.

7.5 Bankruptcy Filings.

- Seller shall file the form of the Sale Order, which shall be reasonably acceptable to Buyer, for entry with the Bankruptcy Court prior to the Sale Hearing Date. Subject to Section 7.4, Seller shall pursue diligently the entry of the Sale Order. Buyer agrees that it shall promptly take such actions as are reasonably requested by Seller to assist in obtaining entry of the Sale Order and a finding of adequate assurance of future performance by Buyer of the Assigned Contracts and Assigned Leases and Interests, including furnishing affidavits or other documents or information for filing with the Bankruptcy Court for the purposes, among others, of providing necessary assurances of performance by Buyer under this Agreement and demonstrating that Buyer is a "good faith" purchaser under Section 363(m) of the Bankruptcy Code. In the event that the entry of the Sale Order is appealed or a stay pending appeal is sought, Seller shall oppose the appeal or the stay pending appeal and seek the dismissal of any appeal (including a petition for certiorari, motion for rehearing, re-argument, reconsideration or revocation). Notwithstanding the foregoing, any resulting changes to this Agreement or any related document or resulting material changes to the Sale Order shall be subject to the approval of Buyer in its reasonable discretion. Seller shall (i) provide Buyer with drafts of any and all other pleadings and proposed orders to be filed or submitted in connection with this Agreement and the transactions contemplated hereby, and such pleadings and proposed orders shall be in form and substance reasonably acceptable to Buyer and (ii) make reasonable efforts to consult and cooperate with Buyer regarding any discovery taken in connection with seeking entry of the Sale Order (including any depositions).
- (b) Seller acknowledges and agrees, and the Sale Order shall provide that, on the Closing Date and concurrently with the Closing, all then existing or thereafter arising liens, claims, and encumbrances (other than Permitted Encumbrances and Assumed Liabilities) of, against or created by Seller or its bankruptcy estate, shall be fully released from and with respect to the Oil and Gas Assets, which shall be transferred to Buyer free and clear of all such liens, claims, and encumbrances (other than Permitted Encumbrances and Assumed Liabilities).

7.6 Bidding Procedures.

Buyer agrees and acknowledges that Seller, including through its Representatives, is and may continue soliciting inquiries, proposals, or offers from third parties for all or any part of the Oil and Gas Assets, as contemplated by the Bidding Procedures and the Bid Protection Order.

7.7 <u>Back-Up Bidder</u>.

If an Auction is conducted and Seller does not choose Buyer as the Successful Bidder, but instead chooses Buyer as the Back-Up Bidder, Buyer will be the Back-Up Bidder. If Buyer is chosen as the Back-Up Bidder, Buyer will be required to keep its bid to consummate the transactions contemplated by this Agreement on the terms and conditions set forth in this Agreement (as may be amended with Buyer's written consent prior to or at the Auction) open and irrevocable until the closing of the sale of the Oil and Gas Assets to the Successful Bidder. If the Successful Bid with the Successful Bidder is terminated prior to closing, Buyer will be deemed to be the Successful Bidder and will forthwith consummate the transactions contemplated by this Agreement on the terms and conditions set forth in this Agreement (as the same may be amended with Buyer's written consent prior to or at the Auction).

7.8 Updates and Amendments of Exhibits and Schedules.

Until the fifth (5th) Business Day before Closing, Seller shall have the right to amend, modify and/or supplement Exhibit A, Exhibit B and Exhibit C (subject to Section 2.5), Schedule 2.2(g) and Schedule 2.2(h), in each case, as applicable, in order to reflect (i) any additional Contracts or Leases taken by Seller, (ii) the deletion of any Contracts or Leases from any such Exhibit or Schedule, or (iii) the addition of Exhibits or Schedules that are responsive to the representations and warranties contained herein but for which an Exhibit or Schedule is not contemplated as of the execution of this Agreement. For all purposes of this Agreement, including for purposes of termination pursuant to Article 11 and for determining whether the conditions set forth in Article 11 have been fulfilled, the Exhibits and Schedules to Seller's representations and warranties contained in this Agreement shall be deemed to include only that information contained therein on the date that this Agreement is executed and shall be deemed to exclude all information contained in any addition, supplement or amendment thereto.

ARTICLE 8

ADDITIONAL AGREEMENTS

8.1 Taxes.

- (a) Any transfer, documentary, sales, use, stamp, registration and other such Taxes, and all conveyance fees, recording charges and other fees and charges (including any penalties and interest) incurred in connection with the consummation of the transactions contemplated by this Agreement (collectively, the "<u>Transfer Taxes</u>") shall be borne by Buyer. Seller and Buyer shall use commercially reasonable efforts and cooperate in good faith to exempt the sale and transfer of the Oil and Gas Assets from any Transfer Taxes, including under Section 1146(a) of the Bankruptcy Code. Buyer will, at its own expense, file all necessary Tax Returns and other documentation with respect to all Transfer Taxes, and, if required by applicable law, rule or regulation, the Parties will, and will cause their Affiliates to, join in the execution of any such Tax Returns and other documentation.
- (b) Seller shall retain responsibility for, and shall bear and pay, all ad valorem, property, excise, severance, production or similar Taxes based upon operation or ownership of the Oil and Gas Assets or the production of Hydrocarbons or the receipt of proceeds therefrom (but excluding, for the avoidance of doubt, income, franchise and similar Taxes and Transfer Taxes) (collectively, the "Asset Taxes") for (i) any period ending prior to the Effective Time and (ii) the portion of any Straddle Period ending prior to the Effective Time. For purposes of allocation between the Parties of Asset Taxes assessed with respect to the Oil and Gas Assets that are payable with respect to any tax periods beginning before and ending after the Effective Time ("Straddle Periods"), the portion of any such Taxes that are attributable to the portion of the Straddle Period that ends prior to the Effective Time shall (A) in the case of such Asset

Taxes that are based upon or related to income or receipts or imposed on a transactional basis such as severance or production taxes, be allocated based on revenues from sales occurring before the Effective Time (which shall be Seller's responsibility) and after the Effective Time (which shall be Buyer's responsibility); and (B) in the case of other Asset Taxes, be allocated pro rata per day between the period prior to the Effective Time (which shall be Seller's responsibility) and the period after the Effective Time (which shall be Buyer's responsibility). For purposes of clause (A) and (B) of the preceding sentence, any exemption, deduction, credit or other item that is calculated on an annual basis shall be allocated pro rata per day between the period ending prior to the Effective Time and the period beginning after the Effective Time. At the Closing, Asset Taxes for the applicable Straddle Period shall be prorated in accordance with the foregoing provisions based on the Asset Tax assessment for such Straddle Period, if available, or if otherwise, based on the Asset Taxes paid during the preceding Tax period. With respect to any not yet delinquent Asset Taxes relating to a Tax year ending after the Closing Date, Buyer will assume responsibility for the actual payment of all such Asset Taxes to the applicable Governmental Authority. With respect to any Asset Taxes relating to a Straddle Period or pre-Closing Tax Period that are delinquent as of the Closing Date, the amount of which is known and not subject to dispute, Buyer shall pay the delinquent amount of such Taxes directly to the applicable Governmental Authority at the Closing.

- (c) Seller, on the one hand, or Buyer, on the other hand, as the case may be (the "Reimbursing Party"), shall provide reimbursement for any Tax paid by the other (the "Paying Party") all or a portion of which is the responsibility of the Reimbursing Party in accordance with the terms of this Section 8.1, except to the extent an adjustment to the Base Purchase Price was made pursuant to Section 8.10(b) to compensate the Paying Party for such Tax. Within a reasonable time prior to the payment of any such Tax, the Paying Party shall give notice to the Reimbursing Party of the Tax payable and the Paying Party's and Reimbursing Party's respective Liability therefor, although failure to do so will not relieve the Reimbursing Party from its Liability hereunder except to the extent the Reimbursing Party is prejudiced thereby. Any amounts which may become payable from Seller to Buyer pursuant to this Section 8.1 shall constitute a super priority administrative expense of Seller under Section 364(c)(1) of the Bankruptcy Code with priority over any and all administrative expenses of the kind specified in Sections 503(b) or 507(b) of the Bankruptcy Code.
- (d) From and after Closing Buyer and Seller agree to furnish or cause to be furnished to each other, upon request, as promptly as practicable, such information and assistance relating to the Oil and Gas Assets (including access to books and records and Tax Returns and related working papers dated before Closing) as is reasonably necessary for the filing of all Tax Returns, the making of any election relating to Taxes, the preparation for any audit by any taxing authority, the prosecution or defense of any claims, suit or proceeding relating to any Tax, and the claiming by Buyer of any federal, state or local business tax credits or incentives that Buyer may qualify for in any of the jurisdictions in which any of the Oil and Gas Assets are located; *provided*, *however*, that neither Buyer nor Seller shall be required to disclose the contents of its income Tax Returns to any Person. Any expenses incurred in furnishing such information or assistance pursuant to this Section 8.1(d) shall be borne by the Party requesting it.

8.2 Allocation of Purchase Price.

The Base Purchase Price shall be allocated among the Oil and Gas Assets as set forth in <u>Schedule 8.2(a)</u> hereto, and the portion of the Purchase Price allocated to each Oil and Gas Asset is referred to herein as the "<u>Allocated Value</u>" of such Oil and Gas Asset. Seller and Buyer agree to be bound by the Allocated Values set forth in <u>Schedule 8.2(a)</u> and <u>Schedule 8.2(b)</u> for purposes of this Agreement. <u>Schedule 8.2(b)</u> sets forth the allocation of the Base Purchase Price and any liabilities assumed by Buyer under this Agreement that are treated as consideration for Tax purposes among the Oil and Gas Assets in accordance with Code Section 1060, and the regulations thereunder (the "<u>Tax Allocation</u>"). Seller and Buyer agree that the Tax Allocation shall be prepared in a manner consistent with the Allocated Values, as set forth on

Schedule 8.2(a). Seller and Buyer each agree to report, and to cause their respective Affiliates to report, the federal, state and local income and other Tax consequences of the transactions contemplated herein, and in particular to report the information required by Section 1060(b) of the Code, and to jointly prepare Form 8594 (Asset Acquisition Statement under Section 1060 of the Code) as promptly as possible following the Closing Date and in a manner consistent with the Tax Allocation as revised to take into account subsequent adjustments to the Base Purchase Price, including any adjustments pursuant to the Agreement to determine the Purchase Price, and shall not take any position inconsistent therewith upon examination of any Tax Return, in any refund claim, in any litigation, investigation, or otherwise, unless required to do so by any Legal Requirement after notice to and discussions with the other Party, or with such other Party's prior consent; *provided, however*, that nothing contained herein shall prevent Buyer or Seller from settling any proposed deficiency or adjustment by any Governmental Authority based upon or arising out of the Tax Allocation, and neither Buyer nor Seller shall be required to litigate before any court any proposed deficiency or adjustment by any Governmental Authority challenging the Tax Allocation.

8.3 Bulk Sales.

Buyer and Seller hereby waive compliance with all "bulk sales," "bulk transfer" and similar Legal Requirements that may otherwise be applicable with respect to the sale and transfer of any or all of the Oil and Gas Assets to Buyer.

8.4 Payments Received.

Seller, on the one hand, and Buyer, on the other hand, each agree that, after the Closing, each will hold and will promptly transfer and deliver to the other, from time to time as and when received by them, any cash, checks with appropriate endorsements (using their best efforts not to convert such checks into cash) or other property that they may receive on or after the Closing which properly belongs to the other and will account to the other for all such receipts.

8.5 Assigned Contracts: Adequate Assurance and Performance; Security Arrangements.

- (a) With respect to each Assigned Contract, Buyer shall provide adequate assurance as required under the Bankruptcy Code of the future performance by Buyer of each such Assigned Contract. Buyer and Seller agree that they will promptly take all actions reasonably required to assist in obtaining a Bankruptcy Court finding that there has been an adequate demonstration of adequate assurance of future performance under the Assigned Contracts, such as furnishing timely requested and factually accurate affidavits, non-confidential financial information and other documents or information for filing with the Bankruptcy Court and making Buyer's and Seller's employees and Representatives available to testify before the Bankruptcy Court.
- (b) Buyer shall pay, perform or satisfy the Assumed Liabilities from time to time and as such Assumed Liabilities become due and payable or are required to be performed or satisfied in accordance with their respective terms.
- (c) Without limiting the provisions of Section 8.5(a), Buyer acknowledges that various bonds, surety bonds, letters of credit, guarantees, and/or cash deposits, including those set forth on Schedule 8.5(c) (collectively the "Security Arrangements") have been provided by Seller and/or its Affiliates to secure the payment and/or performance of certain of Seller's Obligations related to the Oil and Gas Assets. Buyer acknowledges that Seller has no duty to maintain any Security Arrangements after the Closing. To the extent Seller and/or any of its Affiliates have any obligations pursuant to any Security Arrangement or have pledged or otherwise provided any property that secures any such Security Arrangement (collectively, the "Seller's Obligations"), Buyer shall take such actions, during the Straddle Period, as are necessary to

cause the Seller's Obligations arising under the Security Arrangements set forth on Schedule 8.5(c) (and such Security Arrangements) to be released and terminated, and any of Seller's property pledged or otherwise provided to secure such Security Arrangements returned to the Seller, concurrent with the Closing. As to those Security Arrangements not listed on Schedule 8.5(c), Buyer shall take such actions as are necessary to cause the Seller's Obligations arising under such Security Arrangements (and such Security Arrangements) to be released and terminated, and any of Seller's property pledged or otherwise provided to secure such Security Arrangements returned to the Seller, within thirty (30) days following Seller's notifying Buyer (or if earlier, Buyer's otherwise becoming aware) of such Security Arrangement, and Buyer shall reimburse Seller, within ten (10) days following Seller's demand therefor, the aggregate amount of Seller's Obligations that are paid or performed by Seller under such Security Arrangements following the Closing.

8.6 Post-Closing Books and Records and Personnel.

For five (5) years after the Closing Date (or such longer period as may be required by any Governmental Authority or ongoing claim), (a) Buyer shall not dispose of or destroy any of the Records received by Buyer as Oil and Gas Assets and (b) Buyer shall allow Seller (including, for clarity, any trust established under a chapter 11 plan of Seller or any other successors of Seller) and any of its directors, officers, employees, counsel, representatives, accountants and auditors reasonable access during normal business hours, at Seller's (or such Person's) sole expense and upon reasonable advance notice, to all employees and files of Buyer and their respective controlled Affiliates and any Records included in the Oil and Gas Assets for purposes relating to the Bankruptcy Case, the wind-down of the operations of Seller, the functions of any such trusts or successors, or other reasonable business purposes, and Seller (including any such trust or successors) and such directors, officers, employees, counsel, representatives, accountants and auditors shall have the right to make copies of any such files, books, records and other materials. Until the closing of the Bankruptcy Case or the liquidation and winding up of Seller's estates, Seller shall preserve and keep the Records and, at Buyer's sole expense, shall make such Records, records, and Seller's personnel available to Buyer as may be reasonably required by Buyer in connection with, among other things, any insurance claims by, Proceedings, Actions or Tax audits against, or governmental investigations of, Buyer or any of its Affiliates or in order to enable Buyer to comply with its obligations under this Agreement and each other Transaction Document. In the event any Party desires to destroy any such Records during or after the time during which they must be maintained pursuant to this Section 8.6, such Party shall first give ninety (90) days prior written notice to the other Party and any such other Party shall have the right at its option and expense, upon prior written notice given within such ninety (90) day period to the Party desiring to destroy such Records or records, to take possession of the applicable Records within one hundred and eighty (180) days after the date of such notice, or such shorter period as the liquidation and winding up of Seller's estates shall permit.

8.7 No Other Representations or Warranties; Disclaimers; Indemnification.

(a) NOTWITHSTANDING ANYTHING TO THE CONTRARY CONTAINED IN THIS AGREEMENT, EXCEPT AS AND TO THE EXTENT EXPRESSLY SET FORTH IN THIS AGREEMENT AND IN THE TRANSACTION DOCUMENTS. SELLER MAKES REPRESENTATIONS OR WARRANTIES WHATSOEVER, AND DISCLAIMS ALL LIABILITY AND RESPONSIBILITY REPRESENTATION, WARRANTY, **STATEMENT** FOR ANY INFORMATION MADE OR COMMUNICATED (ORALLY OR IN WRITING) TO BUYER (INCLUDING ANY OPINION, INFORMATION, OR ADVICE THAT MAY HAVE BEEN PROVIDED TO BUYER BY ANY AFFILIATE OR REPRESENTATIVE OF SELLER OR BY ANY INVESTMENT BANK OR INVESTMENT BANKING FIRM, ANY PETROLEUM ENGINEER OR ENGINEERING FIRM, SELLER'S COUNSEL, OR ANY OTHER AGENT, CONSULTANT, OR REPRESENTATIVE OF SELLER). WITHOUT LIMITING THE GENERALITY OF THE FOREGOING, EXCEPT AS AND TO THE EXTENT EXPRESSLY SET FORTH IN THIS AGREEMENT AND IN THE TRANSACTION DOCUMENTS, SELLER EXPRESSLY DISCLAIMS AND NEGATES ANY REPRESENTATION OR WARRANTY, EXPRESS, IMPLIED, AT COMMON LAW, BY STATUTE, OR OTHERWISE, RELATING TO (A) THE TITLE TO ANY OF THE OIL AND GAS ASSETS, (B) THE CONDITION OF THE OIL AND GAS ASSETS (INCLUDING ANY IMPLIED OR EXPRESS WARRANTY OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, OR CONFORMITY TO MODELS OR SAMPLES OF MATERIALS), IT BEING DISTINCTLY UNDERSTOOD THAT THE OIL AND GAS ASSETS ARE BEING SOLD "AS IS," "WHERE IS," AND "WITH ALL FAULTS AS TO ALL MATTERS," (C) ANY INFRINGEMENT BY SELLER OF ANY PATENT OR PROPRIETARY RIGHT OF ANY THIRD PARTY, (D) ANY INFORMATION, DATA, OR OTHER MATERIALS (WHETHER WRITTEN OR ORAL) FURNISHED TO BUYER BY OR ON BEHALF OF SELLER (INCLUDING WITHOUT LIMITATION, IN RESPECT OF ANY SEISMIC DATA, THE EXISTENCE OR EXTENT OF HYDROCARBONS OR THE MINERAL RESERVES, THE RECOVERABILITY OF SUCH RESERVES, ANY PRODUCT PRICING ASSUMPTIONS, AND THE ABILITY TO SELL HYDROCARBON PRODUCTION AFTER THE CLOSING), AND (E) THE ENVIRONMENTAL CONDITION AND OTHER CONDITION OF THE OIL AND GAS ASSETS AND ANY POTENTIAL LIABILITY ARISING FROM OR RELATED TO THE OIL AND GAS ASSETS.

- (b) Buyer acknowledges and affirms that it has made, or will have made as of the Closing, its own independent investigation, analysis, and evaluation of the transactions contemplated hereby and the Oil and Gas Assets (including Buyer's own estimate and appraisal of the extent and value of Seller's Hydrocarbon reserves attributable to the Oil and Gas Assets and an independent assessment and appraisal of the environmental risks associated with the acquisition of the Oil and Gas Assets). Buyer acknowledges that in entering into this Agreement, it has relied only on the aforementioned investigation and the express representations and warranties of Seller contained in this Agreement and the Transaction Documents. Buyer hereby irrevocably covenants to refrain from, directly or indirectly, asserting any claim, or commencing, instituting, or causing to be commenced, any Proceeding of any kind against Seller or its Affiliates, alleging facts contrary to the foregoing acknowledgment and affirmation.
- (c) From and after Closing, O'Brien Resources shall be liable to and shall indemnify, defend, and hold harmless, Buyer and its Affiliates from and against all Liability incurred or suffered by Buyer caused by, arising out of, or resulting from claims by royalty owners for amounts held in suspense and related to the time period prior to the Closing Date.

8.8 <u>Casualty</u>.

- (a) If, after the date of execution of this Agreement and prior to the Closing, a material part of the Oil and Gas Assets suffers a Casualty Loss or if a material part of the Oil and Gas Assets is taken in condemnation or under the right of eminent domain or if proceedings for such purposes are pending or threatened, Seller shall promptly give Buyer written notice of such occurrence, including reasonable particulars with respect thereto, and this Agreement shall remain in full force and effect notwithstanding any such destruction, taking, proceeding, or threat.
- (b) With regard to a Casualty Loss or condemnation occurring after the date of execution of this Agreement and prior to the Closing, subject to Seller's consent, which shall not be unreasonably withheld, conditioned, or delayed, Buyer may elect to exclude the affected Oil and Gas Assets from this Agreement, without any adjustment to the Purchase Price, and the affected Oil and Gas Assets shall be treated as Excluded Assets for all purposes under this Agreement.
- (c) Unless the Oil and Gas Assets affected by a Casualty Loss are excluded pursuant to Section 8.8(b), (i) at the Closing, the Oil and Gas Assets affected by a Casualty Loss or condemnation

shall be included in the Closing and Buyer shall pay the full Allocated Value therefor, subject to any applicable adjustments under this Agreement, and (ii) Buyer's recourse with respect to a condemnation or Casualty Loss shall be limited to the proceeds of Seller's applicable insurance coverage actually recovered by Seller in respect thereof or other sums paid to Seller by third parties (or an assignment of claims related thereto), which proceeds or other sums shall be payable to Buyer only upon or after the Closing of the transactions contemplated hereby. Seller shall have no other liability or responsibility to Buyer (and there shall not be any adjustment to the Purchase Price) with respect to a condemnation or Casualty Loss, EVEN IF SUCH CASUALTY LOSS SHALL HAVE RESULTED FROM OR SHALL HAVE ARISEN OUT OF THE SOLE OR CONCURRENT NEGLIGENCE, FAULT, VIOLATION OF A LEGAL REQUIREMENT, OR WILLFUL MISCONDUCT OF SELLER OR SELLER PARTIES.

(d) Unless the affected Oil and Gas Assets are excluded pursuant to Section 8.8(b), (i) no insurance or condemnation proceeds shall be committed or applied by Seller to repair, restore or replace a lost, damaged, destroyed or taken portion of the Oil and Gas Assets if the cost to repair, restore or replace a lost, damaged, destroyed or taken portion of the Oil and Gas Assets is projected to exceed \$200,000.00, (ii) to the extent such proceeds are not committed or applied by any Seller prior to the Closing Date in accordance with this Section 8.8(d), Seller shall, at the Closing, pay to Buyer all sums paid to Seller by reason of such loss, damage, destruction or taking, less any reasonable costs and expenses incurred by Seller in collecting such proceeds, (iii) in addition and to the extent such proceeds have not been committed or applied by Seller in accordance with this Section 8.8(d), in such repair, restoration, or replacement, Seller shall transfer to Buyer, at the Closing, without recourse against Seller, all of the right, title, and interest of Seller in and to any unpaid insurance or condemnation proceeds arising out of such loss, damage, destruction or taking, less any reasonable costs and expenses incurred by Seller in collecting such proceeds, and (iv) any such funds that have been committed by Seller for repair, restoration or replacement as aforesaid shall be paid by Seller for such purposes or, at Seller's option, delivered to Buyer upon Seller's receipt from Buyer of adequate assurance and indemnity that Seller shall incur no liability or expense as a result of such commitment.

8.9 Preferential Purchase Rights.

- (a) Within two (2) Business Days after the Sale Order is entered, Seller shall deliver to each such holder of a Preferential Purchase Right a notice that is in material compliance with the contractual provisions applicable thereto, offering to sell to each such holder the applicable Oil and Gas Assets subject to such Preferential Purchase Right in accordance with the terms of this Agreement and in exchange for an amount equal to the Allocated Value of such Oil and Gas Asset, adjusted as set forth in this Agreement, or, alternatively, seeking such holder's consent to the assignment of the applicable Oil and Gas Assets to Buyer; it being understood and agreed by the Parties that Seller shall not be obligated to make any payments or undertake obligations in connection with the obtaining of such consents. If, following the Sale Order, Buyer discovers a Preferential Purchase Right that is not set forth on Disclosure Schedule 5.11, Buyer may, at its option, provide a written notice to Seller of such Preferential Purchase Right, and Seller shall deliver a written notice as described in the immediately preceding sentence to the holder of such Preferential Purchase Right.
- (b) If, as of the Closing, a holder of a Preferential Purchase Right has elected to exercise its Preferential Purchase Right with respect to the Oil and Gas Assets to which its Preferential Purchase Right applies, then (i) each of the Oil and Gas Assets affected by such Preferential Purchase Right shall be held back from the Oil and Gas Assets conveyed at Closing, without reduction to the Purchase Price, (ii) such Oil and Gas Assets shall thereafter be Excluded Assets (except as otherwise provided in the following sentence), and (iii) at Closing or the consummation of the purchase and sale of such Oil and Gas Assets with the holder of such Preferential Purchase Right, whichever is later, if Seller receives any proceeds generated from the exercise of such Preferential Purchase Right, Seller shall pay (or shall cause

such holder to pay) to Buyer all such proceeds. If for any reason after Closing, the purchase and sale of such Oil and Gas Assets is not or cannot be consummated with the holder of such Preferential Purchase Right and Seller is permitted after Closing (including, without limitation, pursuant to the Sale Order) to transfer such Oil and Gas Assets to Buyer pursuant to the terms of such Preferential Purchase Right, then the Oil and Gas Assets affected by such Preferential Purchase Right will be conveyed to Buyer at a subsequent Closing to be held within ten (10) Business Days thereafter, at which subsequent closing Seller shall assign, transfer and convey to Buyer, and Buyer shall acquire and accept from Seller, such Oil and Gas Assets pursuant to the terms of this Agreement. Notwithstanding the foregoing, nothing in this Section 8.9(b) shall prevent Seller from ceasing operations or winding up its affairs following the Closing.

- (c) All Oil and Gas Assets for which any applicable Preferential Purchase Right (i) has been waived, (ii) as to which the period to exercise the applicable Preferential Purchase Right has expired without the exercise thereof, or (iii) has not been exercised or waived and the time for exercising such Preferential Purchase Right has not expired, in each case, prior to Closing shall be transferred to Buyer at Closing pursuant to the provisions of this Agreement. With respect to any Preferential Purchase Right described in Section 8.9(c)(iii) above, in the event the holder thereof exercises such Preferential Purchase Right after Closing to consummate the purchase of the Oil and Gas Asset(s) to which such Preferential Purchase Right applies, Buyer shall comply with the terms of such Preferential Purchase Right and shall be entitled to, and such holder shall pay to Buyer, all of the proceeds generated from the exercise of such Preferential Purchase Right.
- (d) The provisions of this <u>Section 8.9</u> shall only apply with respect to those Oil and Gas Assets that are subject to Preferential Purchase Rights and that cannot be acquired by Buyer free and clear of such Preferential Purchase Rights pursuant to the Sale Order or the Bankruptcy Code.

8.10 Accounting Adjustments for Revenues and Expenses.

- (a) The Base Purchase Price shall be adjusted to reflect the net cash flow of the Oil and Gas Assets from the Effective Time until and through the Closing Date. Based on an examination of records maintained and provided by Seller, the net cash flow of the Oil and Gas Assets is hereby agreed to be One Hundred Thousand Dollars (\$100,000.00) (the "Net Cash Flow") per calendar month and accordingly, the Base Purchase Price shall be adjusted downward as such adjustment is adjusted pursuant to Section 8.11. In the event that Closing occurs between August 31, 2019 and the Outside Date, the Net Cash Flow shall be calculated as Three Thousand Three Hundred Thirty-Three and 33/100 Dollars (\$3,333.33) per day and the Base Purchase Price shall be further reduced on a day to day basis until Closing occurs.
- (b) The Base Purchase Price also shall be (i) increased by the amount of Asset Taxes allocable to Buyer pursuant to <u>Section 8.1(b)</u> but paid or otherwise economically borne by Seller and (ii) decreased by the amount of Asset Taxes allocable to Seller pursuant to <u>Section 8.1(b)</u> but paid or otherwise economically borne by Buyer.

8.11 Initial Adjustment at Closing.

At least five (5) days before the Closing Date, Seller shall provide to Buyer a statement showing its computations, calculated in good faith, of the amount of the adjustments provided for in Section 8.10 above. Buyer and Seller shall attempt to agree upon such adjustments prior to Closing, provided, that if agreement is not reached, Seller's computation prepared in good faith shall be used at Closing. If the amount of adjustments so determined which would result in a credit to Buyer exceed the amount of adjustments so determined which would result in a credit to Seller, Buyer shall receive a Base Purchase Price reduction at

Closing for the amount of such excess, and if the converse is true, then the amount to be paid by Buyer to Seller at Closing shall be increased by the amount of such excess.

ARTICLE 9

CONDITIONS PRECEDENT TO OBLIGATIONS OF BUYER TO CLOSE

The obligations of Buyer to consummate the transactions contemplated by this Agreement are subject to the satisfaction or waiver, at or prior to the Closing, of each of the following conditions:

9.1 <u>Accuracy of Representations.</u>

The representations and warranties of Seller set forth in this Agreement shall be true and correct in all material respects on and as of the Closing Date with the same effect as though such representations and warranties had been made on and as of the Closing (*provided*, *that* representations and warranties which are confined to a specified date need only be true and correct as of such date); *provided*, *however*, that in the event of one or more breaches of or inaccuracies in the representations and warranties of Seller set forth in this Agreement, the condition set forth in this Section 9.1 shall be deemed satisfied unless the effect of all such breaches of and/or inaccuracies in such representations and warranties taken together results in a Material Adverse Effect; *provided*, *further*, *however*, notwithstanding the preceding parts of this Section 9.1 the representations and warranties of Seller set forth in Sections 5.1, 5.2, 5.3, 5.13, and 5.18 shall be true and correct in all respects. Buyer shall have received a certificate of Seller to such effect signed by a duly authorized officer of Seller. If Buyer determines that there has been a breach or inaccuracy of any of Seller's representations and warranties, it shall provide Seller with notice of such breach or inaccuracy as promptly as reasonably practicable so that Seller may attempt to cure such breach or inaccuracy on or before the Closing Date.

9.2 Seller's Performance.

The covenants and agreements that Seller is required to perform or to comply with pursuant to this Agreement at or prior to the Closing shall have been duly performed and complied with in all material respects (except that those covenants and agreements that are qualified as to materiality or Material Adverse Effect or similar expressions shall have been duly performed and complied with in all respects), and Buyer shall have received a certificate of Seller to such effect signed by a duly authorized officer thereof.

9.3 No Order.

No Governmental Authority shall have enacted, issued, promulgated or entered any Order which is in effect and has the effect of making illegal or otherwise prohibiting the consummation of the transactions contemplated by this Agreement or could cause any of such transactions to be rescinded following the Closing.

9.4 Seller's Deliveries.

Each of the deliveries required to be made to Buyer pursuant to <u>Section 4.4</u> shall have been so delivered or Seller shall be ready, willing and able to make such deliveries (or cause such deliveries to be made).

9.5 Sale Order.

The Bankruptcy Court shall have entered the Sale Order and the Sale Order shall be in full force and effect and shall not be subject to a stay pending appeal and such Sale Order is a Final Order.

ARTICLE 10

CONDITIONS PRECEDENT TO THE OBLIGATION OF SELLER TO CLOSE

Seller's obligation to consummate the transactions contemplated by this Agreement is subject to the satisfaction or waiver, at or prior to the Closing, of each of the following conditions:

10.1 Accuracy of Representations.

The representations and warranties of Buyer set forth in this Agreement shall be true and correct in all material respects (except that those representations and warranties which are qualified as to materiality or similar expressions shall be true and correct in all respects) as of the Closing Date with the same effect as though such representations and warranties had been made on and as of the Closing Date (*provided*, that representations and warranties which are confined to a specified date need only be true and correct as of such date), and Seller shall have received a certificate of Buyer to such effect signed by a duly authorized officer of Buyer.

10.2 Sale Order in Effect.

The Bankruptcy Court shall have entered the Sale Order and the Sale Order shall be in full force and effect and shall not be subject to a stay pending appeal.

10.3 Buyer's Performance.

The covenants and agreements that Buyer is required to perform or to comply with pursuant to this Agreement at or prior to the Closing shall have been performed and complied with in all material respects (except that those covenants and agreements that are qualified as to materiality or similar expressions shall have been duly performed and complied with in all respects), and Seller shall have received a certificate of Buyer to such effect signed by a duly authorized officer of Buyer.

10.4 No Order.

No Governmental Authority shall have enacted, issued, promulgated or entered any Order which is in effect and which has the effect of making illegal or otherwise prohibiting the consummation of the transactions contemplated by this Agreement or could cause any of such transactions to be rescinded following the Closing.

10.5 <u>Buyer's Deliveries</u>.

Each of the deliveries required to be made to Seller pursuant to <u>Section 4.3</u> shall have been so delivered or Buyer shall be ready, willing and able to make such deliveries (or cause such deliveries to be made).

ARTICLE 11

TERMINATION

11.1 Termination Events.

Anything contained in this Agreement to the contrary notwithstanding, this Agreement may be terminated at any time prior to the Closing:

- (a) by either Seller or Buyer:
- (i) if a Governmental Authority issues a final, non-appealable ruling or Order prohibiting the transactions contemplated hereby where such ruling or Order was not requested, encouraged or supported by any of Seller or Buyer or either of their Affiliates;
 - (ii) by mutual written consent of Seller and Buyer; and
- (iii) if the Bankruptcy Court enters an Order dismissing, or converting into cases under chapter 7 of the Bankruptcy Code, any of the cases commenced by Seller under chapter 11 of the Bankruptcy Code and comprising part of the Bankruptcy Case.
 - (b) by Buyer:
- (i) in the event of any breach by Seller of any of Seller's agreements, covenants, representations or warranties contained herein and such breach (A) would result in the failure of a condition set forth in Section 9.1 or Section 9.2 to be satisfied, and (B) has not been cured within fifteen (15) Business Days after the giving of written notice thereof by Buyer to Seller (which notice shall specify the nature of such breach and be given as promptly as practicable); provided, however, that Buyer shall not be permitted to terminate this Agreement pursuant to this Section 11.1(b) if Buyer is itself in material breach of any of its representations, warranties, covenants or agreements contained herein (or in breach at all with respect to those representations, warranties, covenants and agreements that are qualified as to materiality or similar expressions);
- (ii) in the event that Seller enters into and consummates an Alternative Transaction;
- (iii) if the Closing shall not have occurred by 5:00 p.m. local time in Houston, Texas on September 30, 2019 (the "Outside Date"), or such later date as agreed to in writing by Seller and Buyer; provided, however, that Buyer shall not be permitted to terminate this Agreement pursuant to this Section 11.1(b)(iii) if Buyer is itself in material breach of any of its representations, warranties, covenants or agreements contained herein (or in breach at all with respect to those representations, warranties, covenants and agreements that are qualified as to materiality or similar expressions); provided, further, that in the event Buyer is determined to be the Back-Up Bidder with respect to the Oil and Gas Assets, Buyer shall have no right to terminate this Agreement until the earlier of: (x) the closing of the Successful Bid with the Successful Bidder; or, (y) if the Successful Bid with the Successful Bidder is terminated prior to closing and Buyer is then deemed to be the Successful Bidder, the Alternative Outside Date; and
- (iv) if a breach of Seller's representation and warranty in <u>Section 5.19</u> affects more than ten (10%) of the Base Purchase Price based on the aggregate Allocated Value of the Wells, as set forth on **Exhibit B**, affected by such breach.

(c) by Seller:

- (i) in the event of any breach by Buyer of any of Buyer's agreements, covenants, representations or warranties contained herein and such breach (A) would result in the failure of a condition set forth in Section 10.1 or Section 10.3 to be satisfied and (B) has not been cured within fifteen (15) Business Days after the giving of written notice thereof by Seller to Buyer (which notice shall specify the nature of such breach and be given as promptly as practicable); provided, however, that Seller shall not be permitted to terminate this Agreement pursuant to this Section 11.1(c) if Seller is itself in material breach of any of its representations, warranties, covenants or agreements contained herein (or in breach at all with respect to those representations, warranties, covenants and agreements that are qualified as to materiality or Material Adverse Effect or similar expressions);
- (ii) in the event that Seller enters into and consummates an Alternative Transaction; and
- (iii) if the Closing shall not have occurred by the Outside Date, or such later date as agreed to in writing by Seller and Buyer; *provided*, *however*, that Seller shall not be permitted to terminate this Agreement pursuant to this Section 11.1(c)(iii) if Seller is itself in material breach of any of its representations, warranties, covenants or agreements contained herein (or in breach at all with respect to those representations, warranties, covenants and agreements that are qualified as to materiality or Material Adverse Effect or similar expressions).

11.2 Effect of Termination.

Subject to Section 3.2, in the event of termination of this Agreement by Buyer or Seller pursuant to this Article 11, this Agreement shall become null and void and have no effect, and all rights and obligations of the Parties under this Agreement shall terminate without any Liability of any Party to any other Party (other than as expressly provided herein); *provided*, that the provisions of Sections 7.1(c) (Access and Reports; Confidentiality), 13.9 (Expenses), and 13.11 (Governing Law, Consent to Jurisdiction and Venue; Jury Trial Waiver), and this Section 11.2 (and, to the extent applicable to the interpretation or enforcement of such provisions, Article 11 shall expressly survive the termination of this Agreement).

11.3 Mutual Acknowledgment of Agreements.

Each Party acknowledges that the agreements contained in <u>Section 11.2</u> are an integral part of the transactions contemplated by this Agreement, that without these agreements such Party would not have entered into this Agreement.

ARTICLE 12

TITLE AND ENVIRONMENTAL MATTERS

12.1 NORM.

BUYER ACKNOWLEDGES THAT THE OIL AND GAS ASSETS HAVE BEEN USED FOR EXPLORATION, DEVELOPMENT AND PRODUCTION OF HYDROCARBONS AND WATER AND THAT THERE MAY BE PETROLEUM, PRODUCED WATER, WASTES OR OTHER MATERIALS LOCATED ON, UNDER OR ASSOCIATED WITH THE OIL AND GAS ASSETS. EQUIPMENT AND SITES INCLUDED IN THE OIL AND GAS ASSETS MAY CONTAIN NORM. NORM MAY AFFIX OR ATTACH ITSELF TO THE INSIDE OF WELLS, MATERIALS AND EQUIPMENT AS SCALE, OR IN OTHER FORMS; THE WELLS, MATERIALS AND

EQUIPMENT LOCATED ON OR INCLUDED IN THE OIL AND GAS ASSETS MAY CONTAIN NORM AND OTHER WASTES OR HAZARDOUS SUBSTANCES/ AND HAZARDOUS MATERIALS; AND NORM CONTAINING MATERIAL AND OTHER WASTES OR HAZARDOUS SUBSTANCES OR HAZARDOUS MATERIALS MAY HAVE BEEN BURIED, COME IN CONTACT WITH THE SOIL OR OTHERWISE BEEN DISPOSED OF ON OR AROUND THE OIL AND GAS ASSETS. SPECIAL PROCEDURES MAY BE REQUIRED FOR THE REMEDIATION, REMOVAL, TRANSPORTATION OR DISPOSAL OF WASTES, ASBESTOS, HAZARDOUS SUBSTANCES OR HAZARDOUS MATERIALS, INCLUDING HYDROGEN SULFIDE GAS AND NORM FROM THE OIL AND GAS ASSETS. FROM AND AFTER THE CLOSING, BUYER SHALL ASSUME RESPONSIBILITY FOR THE CONTROL, STORAGE, HANDLING, TRANSPORTING AND DISPOSING OF OR DISCHARGE OF ALL MATERIALS, SUBSTANCES AND WASTES FROM THE OIL AND GAS ASSETS (INCLUDING PRODUCED WATER, HYDROGEN SULFIDE GAS, DRILLING FLUIDS, NORM AND OTHER WASTES), PRESENT AFTER THE EFFECTIVE TIME, IN A SAFE AND PRUDENT MANNER AND IN ACCORDANCE WITH ALL APPLICABLE ENVIRONMENTAL LAWS.

12.2 <u>Sole Remedy; Waiver of Title and Environmental Matters.</u>

Buyer hereby acknowledges and agrees that Buyer's sole and exclusive remedy for any (i) title defect, deficiency or other title matter, including any Encumbrance, with respect to any Oil and Gas Assets, or otherwise and (ii) circumstance with respect to the Oil and Gas Assets relating to Environmental Laws, the Release of materials or Hazardous Substances into the environment or protection of the environment or public health, in each case, shall be, if applicable, the rights of Buyer to terminate this Agreement pursuant to Section 11.1(a) or Section 11.1(b), and (y) Buyer hereby expressly waives any and all rights or remedies with respect to any defect, deficiency or other title matter, with respect to the Oil and Gas Assets (subject to Buyer's right to receive the Oil and Gas Assets free and clear of all Encumbrances (other than Permitted Encumbrances) upon the terms and subject to the conditions of the Sale Order, all as contemplated by Section 2.1) and (z) BUYER ALSO WAIVES ANY RIGHTS AGAINST SELLER WITH RESPECT TO ANY CIRCUMSTANCE WITH RESPECT TO THE OIL AND GAS ASSETS RELATING TO ENVIRONMENTAL LAWS, THE RELEASE OF MATERIALS OR HAZARDOUS SUBSTANCES INTO THE ENVIRONMENT OR PROTECTION OF THE ENVIRONMENT OR PUBLIC HEALTH.

ARTICLE 13

GENERAL PROVISIONS

13.1 Survival.

All covenants and agreements contained in this Agreement which by their terms are to be performed in whole or in part, or which prohibit actions, subsequent to the Closing shall, solely to the extent such covenants and agreements are to be performed, or prohibit actions, subsequent to the Closing, survive the Closing in accordance with their terms. Subject to the following sentence, all other covenants and agreements contained in this Agreement, and all representations and warranties contained in this Agreement or in any certificated deliveries hereunder, shall not survive the Closing and shall thereupon terminate, including any Actions for damages in respect of any breach thereof. Notwithstanding anything to the contrary, the indemnity obligations set forth in Section 5.18, Section 6.6, Section 7.1(e), and Section 8.7(c), and the representation and warranties set forth in Section 5.18 and Section 6.6 shall survive indefinitely. Notwithstanding the foregoing, nothing in this Section 13.1 shall prohibit Seller from ceasing operations or winding up its affairs following Closing.

13.2 <u>Confidentiality</u>.

The Parties agree that the confidentiality agreement entered into by them and their Affiliates, dated July 11, 2019 (the "Confidentiality Agreement"), shall continue in full force and effect notwithstanding the execution and delivery by the Parties of this Agreement; *provided*, *however*, that (a) disclosure of matters that become a matter of public record as a result of the Bankruptcy Case and the filings related thereto shall not constitute a breach of such Confidentiality Agreement, and (b) disclosures permitted under this Agreement shall not constitute a breach of such Confidentiality Agreement.

13.3 Public Announcements.

Unless otherwise required by applicable Legal Requirement or by obligations of Buyer or Seller or their respective Affiliates pursuant to any listing agreement with or rules of any securities exchange or with respect to disclosure of matters that become a matter of public record as a result of the Bankruptcy Case and the filings related thereto, Buyer, on the one hand, and Seller, on the other hand, shall consult with each other before issuing any press release or otherwise making any public statement with respect to this Agreement, the transactions contemplated hereby or the activities and operations of the other and shall not issue any such release or make any such statement without the prior written consent of the other (such consent not to be unreasonably withheld or delayed).

13.4 Notices.

All notices (including any notice of termination pursuant to <u>Section 11.1</u> of this Agreement), consents, waivers and other communications under this Agreement must be in writing and shall be deemed to have been duly given when (a) delivered by hand (with written confirmation of receipt), (b) sent by email (with read receipt requested, with the receiving Party being obligated to respond affirmatively to any read receipt requests delivered by the other Party), (c) received by the addressee, if sent by a delivery service (prepaid, receipt requested) or (d) received by the addressee, if sent by registered or certified mail (postage prepaid, return receipt requested), in each case to the appropriate addresses and representatives (if applicable) set forth below (or to such other addresses and representatives as a Party may designate by notice to the other Parties):

(i) If to Seller, then to:

O'Benco IV, LP 425 Ashley Ridge Blvd., Suite 300 P.O. Box 6149 Shreveport, Louisiana 71136-6149 Attn: William J. O'Brien IV Telephone: 318-865-8568

Facsimile: 318-861-5173

E-mail: wobrien@obrienenergyco.com

with a copy (which shall not constitute notice) to:

Bracewell LLP 711 Louisiana Street, Suite 2300 Houston, Texas 77002 Attn: Trey Wood III

E-mail: Trey. Wood@bracewell.com

(ii) If to Buyer:

Pachira Oil and Gas, LLC 410 17th Street Suite 1110 Denver, Colorado 80202 Attn: Brad Morse

with a copy (which shall not constitute notice) to:

Davis Graham & Stubbs LLP 1550 17th Street Suite 500 Denver, Colorado 80202 Attn: Sam Niebrugge

13.5 Waiver, Waiver of Damages.

Neither the failure nor any delay by any Party in exercising any right, power or privilege under this Agreement or the documents referred to in this Agreement shall operate as a waiver of such right, power or privilege, and no single or partial exercise of any such right, power or privilege shall preclude any other or further exercise of such right, power or privilege or the exercise of any other right, power or privilege. To the maximum extent permitted by applicable law, (a) no waiver that may be given by a Party shall be applicable except in the specific instance for which it is given, and (b) no notice to or demand on one Party shall be deemed to be a waiver of any right of the Party giving such notice or demand to take further action without notice or demand. NOTWITHSTANDING ANYTHING TO THE CONTRARY CONTAINED HEREIN, NO PARTY SHALL BE LIABLE TO THE OTHER FOR SPECIAL, INDIRECT, EXEMPLARY OR PUNITIVE DAMAGES ARISING OUT OF, ASSOCIATED WITH, OR RELATING TO THIS AGREEMENT (INCLUDING LOSS OF PROFIT OR BUSINESS INTERRUPTIONS, HOWEVER SAME MAY BE CAUSED) AND THE PARTIES HEREBY WAIVE ALL CLAIMS FOR ANY SUCH DAMAGES.

13.6 Entire Agreement; Amendment.

This Agreement (including the Schedules and the Exhibits) and the other Transaction Documents supersede all prior agreements between Buyer, on the one hand, and Seller, on the other hand, with respect to its subject matter and constitute a complete and exclusive statement of the terms of the agreements between Buyer, on the one hand, and Seller, on the other hand, with respect to their subject matter. This Agreement may not be amended except by a written agreement executed by all of the Parties.

13.7 <u>Assignment</u>.

This Agreement, and the rights, interests and obligations hereunder, shall not be assigned by any Party by operation of law or otherwise without the express written consent of the other Parties (which consent may be granted or withheld in the sole discretion of such other Party); *provided*, *however*, that Buyer shall be permitted, upon prior notice to Seller, to assign all or part of its rights and/or obligations hereunder to one or more of its Affiliates, and each such assignee shall be deemed to be a Buyer for all purposes under the Transaction Documents to the extent of the rights and obligations so assigned to such assignee. No assignment of any obligations hereunder shall relieve the Parties hereto of any such obligations. Upon any such permitted assignment, the references in this Agreement to the Seller or Buyer shall also apply to any such assignee unless the context otherwise requires.

13.8 <u>Severability</u>.

The provisions of this Agreement shall be deemed severable, and the invalidity or unenforceability of any provision shall not affect the validity or enforceability of the other provisions hereof. If any provision of this Agreement, or the application thereof to any Person or any circumstance, is invalid or unenforceable, (a) a suitable and equitable provision shall be substituted therefor in order to carry out, so far as may be valid and enforceable, the intent and purpose of such invalid or unenforceable provision and (b) the remainder of this Agreement and the application of such provision to other Persons or circumstances shall not be affected by such invalidity or unenforceability.

13.9 Expenses.

Whether or not the transactions contemplated by this Agreement are consummated, the Parties shall bear their own respective expenses (including all compensation and expenses of counsel, financial advisors, consultants, actuaries and independent accountants) incurred in connection with this Agreement and the transactions contemplated hereby; *provided*, *however*, that, subject to <u>Section 8.1(a)</u>, Transfer Taxes shall be borne by Buyer.

13.10 Time of Essence.

Time shall be of the essence with respect to all time periods and notice periods set forth in this Agreement.

13.11 Governing Law; Consent to Jurisdiction and Venue; Jury Trial Waiver.

- (a) EXCEPT TO THE EXTENT THE MANDATORY PROVISIONS OF THE BANKRUPTCY CODE APPLY, THIS AGREEMENT SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF TEXAS APPLICABLE TO CONTRACTS MADE AND TO BE PERFORMED ENTIRELY IN SUCH STATE WITHOUT REGARD TO PRINCIPLES OF CONFLICTS OR CHOICE OF LAWS OR ANY OTHER LAW THAT WOULD MAKE THE LAWS OF ANY OTHER JURISDICTION OTHER THAN THE STATE OF TEXAS APPLICABLE HERETO.
- Court, (i) the Bankruptcy Court shall retain exclusive jurisdiction to enforce the terms of this Agreement and to decide any claims or disputes which may arise or result from, or be connected with, this Agreement, any breach or default hereunder, or the transactions contemplated hereby and (ii) any and all claims relating to the foregoing shall be filed and maintained only in the Bankruptcy Court, and the Parties hereby consent and submit to the exclusive jurisdiction and venue of the Bankruptcy Court and irrevocably waive the defense of an inconvenient forum to the maintenance of any such Action or Proceeding; *provided*, *however*, that, if the Bankruptcy Case is closed, all Actions and Proceedings arising out of or relating to this Agreement shall be heard and determined in a Texas federal court sitting in Houston, Texas, and the Parties hereby irrevocably submit to the exclusive jurisdiction and venue of such courts in any such Action or Proceeding and irrevocably waive the defense of an inconvenient forum to the maintenance of any such Action or Proceeding. The Parties consent to service of process by mail (in accordance with Section 13.4) or any other manner permitted by Legal Requirement.
- (c) EACH PARTY HEREBY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LEGAL REQUIREMENTS, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN ANY PROCEEDING DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS AGREEMENT, THE OTHER TRANSACTION DOCUMENTS OR ANY OTHER AGREEMENT

CONTEMPLATED HERETO AND THERETO OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY (WHETHER BASED ON CONTRACT, TORT OR ANY OTHER THEORY). EACH PARTY (A) CERTIFIES THAT NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER AND (B) ACKNOWLEDGES THAT IT AND THE OTHER PARTIES HAVE BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION.

13.12 <u>Counterparts</u>.

This Agreement and any amendment hereto may be executed in one or more counterparts and by different Parties in separate counterparts, each of which shall be deemed to be an original of this Agreement or such amendment and all of which, when taken together, shall constitute one and the same agreement. Delivery of an executed counterpart of a signature page to this Agreement or any amendment hereto by facsimile or other electronic means (including portable document format sent via email) shall be as effective as delivery of a manually executed counterpart of this Agreement or such amendment, as applicable.

13.13 Parties in Interest; No Third Party Beneficiaries.

This Agreement shall inure to the benefit of and be binding upon the Parties and their respective successors and permitted assigns. This Agreement is for the sole benefit of the Parties and their permitted assigns, and except as expressly provided in <u>Section 7.1(e)</u>, nothing herein, express or implied, is intended to or shall confer upon any other Person any legal or equitable benefit, claim, cause of action, remedy or right of any kind.

13.14 Non-Recourse.

Any claim or cause of action based upon, arising out of, or related to this Agreement or any agreement, document or instrument contemplated hereby may only be brought against the Parties hereto, and then only with respect to the specific obligations set forth herein or therein. Other than the Parties hereto, no other party shall have any liability or obligation for any of the representations, warranties, covenants, agreements, obligations or Liabilities of any party under this Agreement, the other Transaction Documents, or any other agreement contemplated hereto and thereto or of or for any Proceeding based on, in respect of, or by reason of, the transactions contemplated hereby or thereby (including the breach, termination or failure to consummate such transactions), in each case whether based on contract, tort, fraud, strict liability, other Legal Requirements or otherwise and whether by piercing the corporate veil, by a claim by or on behalf of a Party hereto or another Person or otherwise.

13.15 Disclosure Schedules; Materiality.

The inclusion of any matter in any Disclosure Schedule shall be deemed to be an inclusion in all other Disclosure Schedules to the extent that such disclosure is sufficient to identify the matter to which such disclosure is responsive and reasonably apparent on its face, but inclusion therein shall not be deemed to constitute an admission, or otherwise imply, that any such matter is material or creates a measure for materiality for purposes of this Agreement. The disclosure of any particular fact or item in any Disclosure Schedule shall not be deemed an admission as to whether the fact or item is "material" or would constitute a "Material Adverse Effect."

13.16 <u>WAIVER OF TEXAS DECEPTIVE TRADE PRACTICES – CONSUMER PROTECTION ACT.</u>

BUYER EXPRESSLY ACKNOWLEDGES AND RECOGNIZES THAT THE PRICE FOR WHICH SELLER HAS AGREED TO SELL THE OIL AND GAS ASSETS AND PERFORM ITS OBLIGATIONS UNDER THE TERMS OF THIS AGREEMENT HAS BEEN PREDICATED UPON THE INAPPLICABILITY OF THE TEXAS DECEPTIVE TRADE PRACTICES - CONSUMER PROTECTION ACT, TEX. BUS. & COM. CODE § 17.41 ET SEQ. (THE "DTPA"), OR ANY SIMILAR STATUTE, AND THE WAIVER OF THE DTPA, AND ANY SIMILAR STATUTE, BY BUYER, SET FORTH IN THIS SECTION 13.16. BUYER'S RIGHTS AND REMEDIES WITH RESPECT TO THIS TRANSACTION AND WITH RESPECT TO ALL ACTS OR PRACTICES OF SELLER, PAST, PRESENT, OR FUTURE, IN CONNECTION WITH THIS TRANSACTION SHALL BE GOVERNED BY LEGAL PRINCIPLES OTHER THAN THE DTPA, OR ANY SIMILAR STATUTE OF ANY JURISDICTION THAT MAY BE APPLICABLE TO THE TRANSACTIONS CONTEMPLATED HEREBY. BUYER HEREBY UNCONDITIONALLY AND IRREVOCABLY WAIVES THE APPLICABILITY OF THE DTPA, OR ANY SIMILAR STATUTE, TO THIS TRANSACTION AND ANY AND ALL RIGHTS, DUTIES, OR REMEDIES THAT MIGHT BE IMPOSED BY THE DTPA, OR ANY SIMILAR STATUTE.

13.17 <u>Injunctive Relief.</u>

- (a) Except as provided in Section 3.2(b), the Parties agree that irreparable damage would occur in the event that any of the provisions of this Agreement or the other Transaction Documents (as applicable) were not performed in accordance with their specific terms or were otherwise breached, and that damages at law may be an inadequate remedy for the breach of any of the covenants, promises and agreements contained in this Agreement or the other Transaction Documents (as applicable), and, accordingly, any Party shall be entitled to injunctive relief to prevent any such breach, and to specifically enforce specifically the terms and provisions of this Agreement or the other Transaction Documents (as applicable), including without limitation specific performance of such covenants, promises or agreements or an order enjoining a Party from any threatened, or from the continuation of any actual, breach of the covenants, promises or agreements contained in this Agreement or the other Transaction Documents (as applicable). The rights set forth in this Section 13.17 shall be in addition to any other rights which a Party may have at law or in equity pursuant to this Agreement or the other Transaction Documents (as applicable).
- (b) The Parties hereby agree not to raise any objections to the availability of the equitable remedy of specific performance to prevent or restrain breaches of this Agreement or the other Transaction Documents (as applicable) by Buyer or Seller, as applicable, and to specifically enforce the terms and provisions of this Agreement or the other Transaction Documents (as applicable) to prevent breaches or threatened breaches of, or to enforce compliance with, the respective covenants and obligations of Buyer or Seller, as applicable, under this Agreement or the other Transaction Documents (as applicable) all in accordance with the terms of this Section 13.17.
- (c) Notwithstanding the foregoing, nothing in this <u>Section 13.17</u> shall prevent Seller from ceasing operations or winding up its affairs following the Closing.

[Signature pages follow.]

IN WITNESS WHEREOF, the Parties have caused this Agreement to be executed and delivered by their duly authorized representatives, all as of the day and year first above written.

SELLER:
O'BENCO IV, LP
By: Name: Title:
ely with respect to the covenants pursuant to <u>Section 7.3(d)</u> tion provided in <u>Section 8.7(c)</u> of this Agreement, O'Brienent as of the date first above written.
SELLER:
O'BRIEN RESOURCES, LLC
By:
Name:Title:
1100.

BUYER:

PACHIRA OIL AND GAS, LLC

By: Name:

Brac

Title:

Manager

EXHIBIT A

ASSIGNED LEASES AND INTERESTS

See attached.

EXHIBIT A ASSIGNED LEASES AND INTERESTS

LEASE#	DATE	LESSOR	LESSEE	BOOK-PAGE	STATE	COUNTY
42.419.14200.0001	02-15-1999	Marcelle Cobb	Pinkston Energy Corporation	0855-417	Texas	Shelby
42.419.14200.0002	02-15-1999	Marcelle Cobb	Pinkston Energy Corporation	0855-420	Texas	Shelby
42.419.14200.0003	08-08-2000	Betty J. Adkison Indiv and as AIF for Lena Williams widow of H.T. Williams	Pinkston Energy Corporation	0899-013	Texas	Shelby
42.419.14200.0005.01	12-05-2001	Pam Rouse Lovett, Co-Executor of the Estate of O.R. Rouse, deceased	Pinkston Energy Corporation	0928-847	Texas	Shelby
42.419.14200.0005.02	12-05-2001	Laverne Billie Wagner, et vir	Pinkston Energy Corporation	0928-850	Texas	Shelby
42.419.14200.0006.02	07-28-1998	Billy Wilson	Pinkston Energy Corporation	0845-867	Texas	Shelby
42.419.14200.0006.03	07-28-1998	Dorothy Brown	Pinkston Energy Corporation	0845-861	Texas	Shelby
42.419.14200.0007.01	12-17-2001	E. Colene Clark	Pinkston Energy Corporation	0930-237	Texas	Shelby
42.419.14200.0007.02	12-01-2001	Inez Ingram	Pinkston Energy Corporation	0928-843	Texas	Shelby
42.419.14200.0008-02	09-08-2000	E. Colene Clark	Pinkston Energy Corporation	0899-017	Texas	Shelby
42.419.14200.0009.01	11-23-2001	Eugene Crawford	Pinkston Energy Corporation	0928-868	Texas	Shelby
42.419.14200.0009.02	11-23-2001	Wanda Horne	Pinkston Energy Corporation	0928-872	Texas	Shelby
42.419.14200.0009.03	11-23-2001	Billy Wilson	Pinkston Energy Corporation	0928-874	Texas	Shelby
42.419.14200.0009.04	11-23-2001	Golda Graves	Pinkston Energy Corporation	0928-876	Texas	Shelby
42.419.14200.0009.05	01-30-2002	Donald Harris	Pinkston Energy Corporation	0935-491	Texas	Shelby
42.419.14200.0009.06	01-30-2002	Judy Harris	Pinkston Energy Corporation	0935-493	Texas	Shelby
42.419.14200.0009.07		Clara E. Grace	Pinkston Energy Corporation	0929-612	Texas	Shelby
42.419.14200.0009.08	08-15-1997	Haskell Williams	Pinkston Energy Corporation	0824-481	Texas	Shelby
42.419.14200.0009.09	11-30-2001		Pinkston Energy Corporation	0929-853	Texas	Shelby
42.419.14200.0010		Mac Samford et ux	Pinkston Energy Corporation	0872-790	Texas	Shelby
42.419.14200.0011	06-18-2001		Pinkston Energy Corporation	0918-934	Texas	Shelby
42.419.14200.0012	06-15-2001	· · · · · · · · · · · · · · · · · · ·		0915-867	Texas	Shelby
42.419.14200.0012	00-13-2001	and as Trustees of the Wesley J. Samford also known as the W.J Samford and Rachel V.	Filiston Energy Corporation	0913-007	rexas	Sileiby
42.419.14200.0014.02	06-26-2001	Edith Franks	Pinkston Energy Corporation	0916-341	Texas	Shelby
42.419.14200.0014.03	06-21-2001	Dorothy Brown	Pinkston Energy Corporation	0917-039	Texas	Shelby
42.419.14200.0014.04	06-21-2001	Billy Wilson	Pinkston Energy Corporation	0917-407	Texas	Shelby
42.419.14200.0015	10-19-2001	Ada Crawford Gorton	Pinkston Energy Corporation	0925-008	Texas	Shelby
42.419.14200.0016	12-22-2001	Billy Ray Sanders et ux	Pinkston Energy Corporation	0929-902	Texas	Shelby
42.419.14200.0017	04-25-2000	Betty J. Adkison, Indiv and as AIF for Lena Williams Widow of H.T. Williams	Pinkston Energy Corporation	0899-010	Texas	Shelby
42.419.14200.0018	06-12-2001	Mac Samford et ux	Pinkston Energy Corporation	0915-870	Texas	Shelby
42.419.14200.0019	06-15-2001	Jimy Dane Samford et ux	Pinkston Energy Corporation	0917-036	Texas	Shelby
42.419.14200.0020.01	12-12-2001	Michael Powdrill	Pinkston Energy Corporation	0930-391	Texas	Shelby
42.419.14200.0020.02	12-12-2001	Richard Powdrill et ux	Pinkston Energy Corporation	0930-391	Texas	Shelby
42.419.14200.0020.03	12-12-2001	Verlon Ray Powdrill	Pinkston Energy Corporation	0928-057	Texas	Shelby
42.419.14200.0020.04	09-20-2001	Joshua W. Kellum	Pinkston Energy Corporation	0925-002	Texas	Shelby
42.419.14200.0021	09-20-2001	James W. Kellum	Pinkston Energy Corporation	0925-005	Texas	Shelby
42.419.14200.0022	10-09-2001		Pinkston Energy Corporation	0925-011	Texas	Shelby
42.419.14200.0023.01	11-19-2001	Charlie Bushley	Pinkston Energy Corporation	0928-865	Texas	Shelby
42.419.14200.0023.02		Louis Bushley	Pinkston Energy Corporation	0928-859	Texas	Shelby
42.419.14200.0023.03		Beth Bushley	Pinkston Energy Corporation	0928-862	Texas	Shelby
42.419.14200.0024		Robert Powdrilll et al	Pinkston Energy Corporation	0925-328	Texas	Shelby
	3. 00 2001		etc L.io.gy Corporation	3323 020	. 5,100	y

1

A2.419.14/20.00/25.01 11-23-20/01 Carolyn Bischarum	LEASE#	DATE	LESSOR	LESSEE	BOOK-PAGE	STATE	COUNTY
42.416.14200.0025.03 11-25.2011 Tammy Lee Williams	42.419.14200.0025.01	11-23-2001	Carolyln Beckham	Pinkston Energy Corporation	0935-489	Texas	Shelby
42.419.1420.0005.04 11-23-2001 Alma Jean Bedwell	42.419.14200.0025.02	11-23-2001	Susan Moore	Pinkston Energy Corporation	0935-752	Texas	Shelby
A2.419.14220.00002	42.419.14200.0025.03	11-23-2001	Tommy Lee Williams	Pinkston Energy Corporation	0928-878	Texas	Shelby
22.419.14220.0003.01 11-25-1988 Robert J. Criffin, Jr. of Lix	42.419.14200.0025.04	11-23-2001	Alma Jean Bridwell	Pinkston Energy Corporation	0928-870	Texas	Shelby
42.419.14/20.0003.02 07-15-2002 Densen Bulson Piniston Energy Corporation 0947-156 Texas Shelby	42.419.14220.0002	08-06-2002	Bobbie Lynn Carlton et al	Pinkston Energy Corporation	0948-643	Texas	Shelby
24.19.14/20.0003.03 07.15.2002 Paul Bibb Piniston Energy Corporation 0947-156 Texas Sinetry 142.419.14/220.0003.04 06.29.2002 tener Cook Piniston Energy Corporation 0947-156 Texas Sinetry 142.419.14/220.0003.05 07.11.2005 Sohert Johnson Townso Piniston Energy Corporation 0901-140 Texas Sinetry 142.419.14/220.0004.01 11.28.2004 Clare E Graco Piniston Energy Corporation 0901-140 Texas Sinetry 142.419.14/220.0006.00 11.28.2004 Clare E Graco Piniston Energy Corporation 0909-240 Texas Sinetry 142.419.14/220.0006.01 11.28.2004 Billy B. Wilson Piniston Energy Corporation 0909-240 Texas Sinetry 142.419.14/220.0005 11.28.2004 Clare E Graco Piniston Energy Corporation 1007-856 Texas Sinetry 142.419.14/220.0005 11.28.2004 Clare E Graco Piniston Energy Corporation 1007-856 Texas Sinetry 142.419.14/220.0005 11.28.2004 Clare E Graco Piniston Energy Corporation 1007-856 Texas Sinetry 142.419.14/220.0005 11.28.2008 Clare Ed Graco Piniston Energy Corporation 1007-856 Texas Sinetry 142.419.14/220.0005 11.28.2008 Clare Ed Graco Piniston Energy Corporation 1007-856 Texas Sinetry 142.419.14/220.0000 05-27-2008 Oberty Jean Bissis Official Resources, LLC 2871-167 Texas Nacogodote 142.447.14/280.0001 07-28-2008 Clare Ed Graco Piniston Energy Corporation 142.447.14/280.0005 07-23-2008 Clare Ed Graco Piniston Energy Corporation 142.447.14/280.0005 07-23-2008 Clare Ed Graco Piniston Energy Corporation 142.447.14/280.0006 11.24-2008 Clare Ed Graco Piniston Energy Co	42.419.14220.0003.01	11-25-1998	Robert J. Griffin, Jr. et ux	Pinkston Energy Corporation	0850-346	Texas	Shelby
42.419.14220.0003.04 06-29-2002 treme Cook Pinkston Energy Corporation 0947-154 Texas Shelby 42.419.14220.0003.05 011-12005 Scherie Johnson Townes Pinkston Energy Corporation 0901-040 Texas Shelby 42.419.14220.0004.02 06-10-2004 Billy B. Wilson Pinkston Energy Corporation 0901-040 Texas Shelby 42.419.14220.0005.02 06-10-2004 Billy B. Wilson Pinkston Energy Corporation 0901-040 Texas Shelby 42.419.14220.0005.02 11-28-2004 Billy B. Wilson Pinkston Energy Corporation 1007-685 Texas Shelby 42.419.14220.0005.02 11-28-2004 Billy B. Wilson Pinkston Energy Corporation 1007-685 Texas Shelby 42.419.14220.0005.02 11-28-2004 Billy B. Wilson Pinkston Energy Corporation 1005-205 Texas Shelby 42.419.14220.0005.02 11-28-2004 Billy B. Wilson Pinkston Energy Corporation 1005-205 Texas Shelby 42.419.14220.0005.02 04-28-2006 Canada Canada Pinkston Energy Corporation 1005-205 Texas Shelby 42.419.14220.0005 07-23-2008 Derothy Join Shale O'Brian Resources, ILC 2891-167 Texas Nacogdoche 42.347.14280.0003 01-28-2008 Cales as Eties O'Brian Resources, ILC 2891-167 Texas Nacogdoche 42.347.14280.0003 01-28-2008 Cales as Eties O'Brian Resources, ILC 2904-171 Texas Nacogdoche 42.347.14280.0000 11-24-2008 Genald Douglas Whitton O'Brian Resources, ILC 3003-178 Texas Nacogdoche 42.347.14280.0000 11-24-2008 Bilby Gene Whittion O'Brian Resources, ILC 3003-178 Texas Nacogdoche 42.347.14280.0001 11-24-2008 Bilby Gene Whittion O'Brian Resources, ILC 3003-179 Texas Nacogdoche 42.347.14280.0011 11-24-2008 Marier Division O'Brian Resources, ILC 3003-170 Texas Nacogdoche 42.347.14280.0011 11-24-2008 Marier Division O'Brian Resources, ILC 3003-161 Texas Nacogdoche 42.347.14280.0011 11-24-2008 Marier Division O'Brian Resources, ILC 3003-161 Texas Nacogdoche 42.347.14280.0011 11-24-2008 Marier Divisio	42.419.14220.0003.02	07-15-2002	Doreen Bulson	Pinkston Energy Corporation	0947-156	Texas	Shelby
42.419.14220.0003.05 01-11-2005 Scherie Johnson Townes Pinkston Energy Corporation 1010-068 Texas Shelby 42.419.14220.0004.01 11-28-2004 Clara E. Grace Pinkston Energy Corporation 0991-040 Texas Shelby 42.419.14220.0005.01 11-28-2004 Billy B. Wilson Pinkston Energy Corporation 0991-040 Texas Shelby 42.419.14220.0005.01 11-28-2004 Billy B. Wilson Pinkston Energy Corporation 0902-290 Texas Shelby 42.419.14220.0005.01 11-28-2004 Billy B. Wilson Pinkston Energy Corporation 0902-290 Texas Shelby 42.419.14220.0005.02 11-28-2004 Clara E. Grace Pinkston Energy Corporation 1005-205 Texas Shelby 42.419.14220.0005.02 11-28-2004 Clara E. Grace Pinkston Energy Corporation 1005-205 Texas Shelby 42.419.14220.0005.02 11-28-2008 Johnnie Rull Goodson, Indiv & as Ex of the Est O'Brien Resources, LLC 2831-271 Texas Nacogdothe 42.417.14280.0003 07-28-2008 Dorothy Jean Staine O'Brien Resources, LLC 2871-167 Texas Nacogdothe 42.417.14280.0003 01-28-2008 Clara Little O'Brien Resources, LLC 2871-167 Texas Nacogdothe 42.417.14280.0003 01-28-2008 Odessa S. Fowdrill O'Brien Resources, LLC 1833-268 Texas Nacogdothe 42.417.14280.0004 01-28-2008 Clara Little O'Brien Resources, LLC 1833-265 Texas Nacogdothe 42.417.14280.0005 07-28-2008 John C. Williamson O'Brien Resources, LLC 3003-181 Texas Nacogdothe 42.417.14280.0000 11-28-2008 Gerald Douglas Willton O'Brien Resources, LLC 3003-181 Texas Nacogdothe 42.417.14280.0001 11-28-2008 Gerald Douglas Willton O'Brien Resources, LLC 3003-181 Texas Nacogdothe 42.417.14280.0001 11-28-2008 Jamet Willton O'Brien Resources, LLC 3003-172 Texas Nacogdothe 42.417.14280.0001 11-28-2008 Jamet Willton O'Brien Resources, LLC 3003-173 Texas Nacogdothe 42.417.14280.0001 11-28-2008 Gloria Drummord O'Brien Resources, LLC 3003-180 Texas Nacogdothe 42.417.14280.0001 11-28-2008 Marion O'Brien Resources, LLC 3003-180 Texas Nacogdothe 42.417.14280.0001 11-28-2008 Marion O'Brien Resources, LLC 3003-180 Texas Nacogdothe 42.417.14280.0001 11-28-2008 Marion O'Brien Resources, LLC 3003-180 Texas Nacogdothe 42.417.14280.0001 1	42.419.14220.0003.03	07-15-2002	Paul Bibb	Pinkston Energy Corporation	0947-158	Texas	Shelby
42.419.4220.0004.02	42.419.14220.0003.04	06-29-2002	Irene Cook	Pinkston Energy Corporation	0947-154	Texas	Shelby
42.419.14220.0001.02 06-10-2004 Billy B. Wilson Pinkston Energy Corporation 0962-290 Texas Shelby 42.419.14220.0005.01 11-28-2004 Billy B. Wilson Pinkston Energy Corporation 1007-595 Texas Shelby 42.419.14220.0005.02 11-28-2004 Billy B. Wilson Pinkston Energy Corporation 1007-595 Texas Shelby 42.419.14220.0005.02 11-28-2004 Clara E. Grace Pinkston Energy Corporation 1007-595 Texas Shelby 42.497.14220.0002 09-27-2008 Denote Run Goodson, Indiv & as Ex of the Est O'Brien Resources, LLC 2833-271 Texas Nacogdoche of Alfred C. Goodson, Deceased V2.497.14220.0003 09-27-2008 Denote Value State Value V	42.419.14220.0003.05	01-11-2005	Scherie Johnson Townes	Pinkston Energy Corporation	1010-908	Texas	Shelby
42.419.44220.0005.01 11-28-2004 Billy B. Wilson Pinkston Energy Corporation 1007-885 Texas Shelby 42.419.14220.0005.02 1128-2004 Clara E. Grace Pinkston Energy Corporation 1005-205 Texas Shelby 42.347.14280.0001 04-08-2008 Johnnie Ruth Goodson, Indiv & as Ex of the Est O'Brien Resources, LLC 2871-167 Texas Nacogdothe 42.347.14280.0002 05-77-2008 Dorothy Jean Staine O'Brien Resources, LLC 2871-167 Texas Nacogdothe 42.347.14280.0003 01-28-2008 Clara Little O'Brien Resources, LLC 1833-268 Texas Nacogdothe 42.347.14280.0004 01-24-2008 Odessas S. Powdrill O'Brien Resources, LLC 2904-171 Texas Nacogdothe 42.347.14280.0005 07-23-2008 John C. Williamson O'Brien Resources, LLC 2904-171 Texas Nacogdothe 42.347.14280.0006 11-24-2008 Genald Douglas Whitton O'Brien Resources, LLC 2003-181 Texas Nacogdothe 42.347.14280.0006 11-24-2008 Boebly Gene Whitton, Sr. O'Brien Resources, LLC 2003-181 Texas Nacogdothe 42.347.14280.0009 12-01-2008 William F. Winkler Davis O'Brien Resources, LLC 2003-172 Texas Nacogdothe 42.347.14280.0009 12-01-2008 Janet Winkler Davis O'Brien Resources, LLC 2003-172 Texas Nacogdothe 42.347.14280.0009 12-01-2008 Janet Winkler Davis O'Brien Resources, LLC 2003-172 Texas Nacogdothe 42.347.14280.0010 11-24-2008 Janet Winkler Davis O'Brien Resources, LLC 2003-172 Texas Nacogdothe 42.347.14280.0011 11-24-2008 Janet Winkler Davis O'Brien Resources, LLC 2003-165 Texas Nacogdothe 42.347.14280.0012 12-03-2008 Giora Drummond O'Brien Resources, LLC 2003-165 Texas Nacogdothe 42.347.14280.0012 12-03-2008 Ramela A. Martin O'Brien Resources, LLC 2003-165 Texas Nacogdothe 42.347.14280.0013 12-01-2008 Martin O'R Bacon O'Brien Resources, LLC 2003-165 Texas Nacogdothe 42.347.14280.0016 11-24-2008 Almothor Almothor O'Brien Resources, LLC 2003-165 Texas Nacogdothe 42.347.1428	42.419.14220.0004.01	11-28-2004	Clara E. Grace	Pinkston Energy Corporation	0991-040	Texas	Shelby
42.491,14280,0005 12 11-28-2004 Claria E, Grace Pinkston Energy Corporation 1005-205 Texas Shebty 42.347,14280,0001 04-08-2008 Johnnin Ruth Goodson, Indivi & set Ex of the Est O'Brien Resources, LLC 2833-271 Texas Nacogdoche of Alfred C, Goodson, Decased 42.347,14280,0002 05-27-2008 Doroffly Jean Staine O'Brien Resources, LLC 2871-167 Texas Nacogdoche 42.347,14280,0003 01-28-2008 Clara Little O'Brien Resources, LLC 1833-268 Texas Nacogdoche 42.347,14280,0005 07-23-2008 John C, Williamson O'Brien Resources, LLC 1833-265 Texas Nacogdoche 42.347,14280,0006 17-24-2006 Ceriad Douglas Whitton O'Brien Resources, LLC 2004-171 Texas Nacogdoche 42.347,14280,0006 17-24-2008 Bobby Gene Whitton, Sr. O'Brien Resources, LLC 3003-178 Texas Nacogdoche 42.347,14280,0006 17-24-2008 William F, Winkfer, Jr. O'Brien Resources, LLC 3003-178 Texas Nacogdoche 42.347,14280,0009 12-01-2008 Janet Wrinkfer Davis O'Brien Resources, LLC 3003-175 Texas Nacogdoche 42.347,14280,0009 12-01-2008 Janet Wrinkfer Davis O'Brien Resources, LLC 3003-176 Texas Nacogdoche 42.347,14280,00010 17-24-2008 Borby Gene Whitton O'Brien Resources, LLC 3003-176 Texas Nacogdoche 42.347,14280,00010 17-24-2008 Jenny L. Wrinkfer Davis O'Brien Resources, LLC 3003-160 Texas Nacogdoche 42.347,14280,0011 12-17-2008 Giria Drummond O'Brien Resources, LLC 3003-160 Texas Nacogdoche 42.347,14280,0011 12-17-2008 Marino for Bacon O'Brien Resources, LLC 3003-150 Texas Nacogdoche 42.347,14280,0011 12-12-2008 Marino for Bacon O'Brien Resources, LLC 3003-150 Texas Nacogdoche 42.347,14280,0011 12-12-2008 Marino for Bacon O'Brien Resources, LLC 3003-150 Texas Nacogdoche 42.347,14280,0011 12-12-2008 Marino for Bacon O'Brien Resources, LLC 3003-150 Texas Nacogdoche 42.347,14280,0011 12-12-2008 Marino for Bacon O'Brien Resources, LLC 3003-160 Texas Nacogdoche 42.347,14280,000	42.419.14220.0004.02	06-10-2004	Billy B. Wilson	Pinkston Energy Corporation	0992-299	Texas	Shelby
42.347.14280.0001 04-08-2008 Johnnie Ruth Goodson, Indiv & as Ex of the Est O'Brien Resources, LLC 2833-271 Texas Nacogdoche 42.347.14280.0002 05-27-2008 Dorothy Jean Staine O'Brien Resources, LLC 2871-167 Texas Nacogdoche 42.347.14280.0003 01-28-2008 Citar a Little O'Brien Resources, LLC 1833-268 Texas Nacogdoche 42.347.14280.0009 01-24-2008 Odessas S. Powdrill O'Brien Resources, LLC 1833-265 Texas Nacogdoche 42.347.14280.0009 01-24-2008 Odessas S. Powdrill O'Brien Resources, LLC 2904-171 Texas Nacogdoche 42.347.14280.0006 11-24-2008 Genel Douglas Writton O'Brien Resources, LLC 3003-181 Texas Nacogdoche 42.347.14280.0006 11-24-2008 Genel Douglas Writton O'Brien Resources, LLC 3003-175 Texas Nacogdoche 42.347.14280.0007 11-24-2008 Bobby Gene Whitton, Sr. O'Brien Resources, LLC 3003-175 Texas Nacogdoche 42.347.14280.0009 12-01-2008 Janet Wrinder Davis O'Brien Resources, LLC 3003-175 Texas Nacogdoche 42.347.14280.0009 12-01-2008 Janet Wrinder Davis O'Brien Resources, LLC 3003-175 Texas Nacogdoche 42.347.14280.0010 11-24-2008 Grace P. Whitton O'Brien Resources, LLC 3003-175 Texas Nacogdoche 42.347.14280.0011 12-17-2008 Gloria Drummond O'Brien Resources, LLC 3003-150 Texas Nacogdoche 42.347.14280.0011 12-17-2008 Gloria Drummond O'Brien Resources, LLC 3003-150 Texas Nacogdoche 42.347.14280.0011 12-17-2008 Marin O'R Bason O'Brien Resources, LLC 3003-150 Texas Nacogdoche 42.347.14280.0011 12-17-2008 Marin O'R Bason O'Brien Resources, LLC 3003-150 Texas Nacogdoche 42.347.14280.0015 11-19-2008 Kerry K Writton O'Brien Resources, LLC 3003-151 Texas Nacogdoche 42.347.14280.0016 11-24-2008 John Million O'R Bason O'Brien Resources, LLC 3003-151 Texas Nacogdoche 42.347.14280.0016 11-24-2008 Marin O'R Bason O'Brien Resources, LLC 3003-151 Texas Nacogdoche 42.347.14280.0016 11-24-2008 Marin O'R Bason O'Brien Resources, LLC 3003-151 Texas Nacogdoche 42.347.14280.0016 11-24-2008 Marin O'R Bason O'Brien Resources, LLC 3003-157 Texas Nacogdoche 42.347.14280.0016 11-24-2008 John Million O'Brien Resources, LLC 3003-157 Texas Nacogdoc	42.419.14220.0005.01	11-28-2004	Billy B. Wilson	Pinkston Energy Corporation	1007-585	Texas	Shelby
of Alfré C. Goodson, Decessed 42.347.14280.0002 05-27-2008 Dorothy Jean Staine O'Brien Resources, LLC 2871-167 Texas Nacogdoche 42.347.14280.0003 01:28-2008 Clara Liftle O'Brien Resources, LLC 1833-265 Texas Nacogdoche 42.347.14280.0005 07-23-2008 John C. Williamson O'Brien Resources, LLC 2904-171 Texas Nacogdoche 42.347.14280.0006 11:24-2008 Gerald Douglas Whitton O'Brien Resources, LLC 3003-181 Texas Nacogdoche 42.347.14280.0006 11:24-2008 Gerald Douglas Whitton O'Brien Resources, LLC 3003-181 Texas Nacogdoche 42.347.14280.0006 11:24-2008 Gerald Douglas Whitton O'Brien Resources, LLC 3003-181 Texas Nacogdoche 42.347.14280.0006 12-01-2008 William F. Winkler, Jr. O'Brien Resources, LLC 3003-181 Texas Nacogdoche 42.347.14280.0009 12-01-2008 Jamet Winkler Davis O'Brien Resources, LLC 3003-172 Texas Nacogdoche 42.347.14280.0001 11:24-2008 Jarry L. Whitton O'Brien Resources, LLC 3003-172 Texas Nacogdoche 42.347.14280.0010 11:24-2008 Jarry L. Whitton O'Brien Resources, LLC 3003-172 Texas Nacogdoche 42.347.14280.0011 12-01-2008 Jarry L. Whitton O'Brien Resources, LLC 3003-166 Texas Nacogdoche 42.347.14280.0011 12-01-2008 Marion O'R Bacon O'Brien Resources, LLC 3003-166 Texas Nacogdoche 42.347.14280.0011 12-01-2008 Marion O'R Bacon O'Brien Resources, LLC 3003-165 Texas Nacogdoche 42.347.14280.0011 12-01-2008 Marion O'R Bacon O'Brien Resources, LLC 3003-163 Texas Nacogdoche 42.347.14280.0015 11-01-2008 Marion O'R Bacon O'Brien Resources, LLC 3003-163 Texas Nacogdoche 42.347.14280.0015 11-24-2008 Allow M. Whitton O'Brien Resources, LLC 3003-160 Texas Nacogdoche 42.347.14280.0016 11-22-2008 Sally G. Maitlingly O'Brien Resources, LLC 3003-167 Texas Nacogdoche 42.347.14280.0019 12-15-2008 William C. Rhoades O'Brien Resources, LLC 3003-190 Texas Nacogdoche 42.347.14280.0021 12-15-2008 William C. Rhoades O'Brien Resources, LLC 3003-193 Texas Nacogdoche 42.347.14280.0021 12-23-2008 Sally Am O'Brien Resources, LLC 3003-193 Texas Nacogdoche 42.347.14280.0022 09-01-0209 John Thompson et ux O'Brien Resources, LLC 3003-193 Texas N	42.419.14220.0005.02	11-28-2004	Clara E. Grace	Pinkston Energy Corporation	1005-205	Texas	Shelby
42:347.14280.0003 01:28-2008 Clara Little O'Brien Resources, LLC 1833-268 Texas Nacogdoche 42:347.14280.0004 01:24-2008 Cdessa S. Powdrill O'Brien Resources, LLC 2804-171 Texas Nacogdoche 42:347.14280.0005 07:23-2008 John C. Williamson O'Brien Resources, LLC 2804-171 Texas Nacogdoche 42:347.14280.0007 11:24-2008 Gerald Douglas Whitton O'Brien Resources, LLC 3003-181 Texas Nacogdoche 42:347.14280.0007 11:24-2008 Bobby Gene Whitton, Sr. O'Brien Resources, LLC 3003-187 Texas Nacogdoche 42:347.14280.0008 12:01-2008 William F. Winkler, Jr. O'Brien Resources, LLC 3003-175 Texas Nacogdoche 42:347.14280.0009 12:01-2008 Janet Winkler Davis O'Brien Resources, LLC 3003-175 Texas Nacogdoche 42:347.14280.0010 11:24-2008 Janet Winkler Davis O'Brien Resources, LLC 3003-180 Texas Nacogdoche 42:347.14280.0011 12:17-2008 Gloria Drummond O'Brien Resources, LLC 3003-189 Texas Nacogdoche 42:347.14280.0011 12:17-2008 Gloria Drummond O'Brien Resources, LLC 3003-180 Texas Nacogdoche 42:347.14280.0012 12:03-2008 Grace P. Whitton et al O'Brien Resources, LLC 3003-180 Texas Nacogdoche 42:347.14280.0013 12:01-2008 Marion Orr Bacon O'Brien Resources, LLC 3003-183 Texas Nacogdoche 42:347.14280.0014 12:01-2008 Marion Orr Bacon O'Brien Resources, LLC 3003-183 Texas Nacogdoche 42:347.14280.0015 11:19-2008 Kerry K. Whitton O'Brien Resources, LLC 3003-183 Texas Nacogdoche 42:347.14280.0016 11-24-2008 Jalice M. Whitton O'Brien Resources, LLC 3003-183 Texas Nacogdoche 42:347.14280.0016 11-24-2008 Sandra Lee O'rr Marrufo O'Brien Resources, LLC 3003-184 Texas Nacogdoche 42:347.14280.0016 11-24-2008 Sandra Lee O'rr Marrufo O'Brien Resources, LLC 3003-187 Texas Nacogdoche 42:347.14280.0019 12:15-2008 William C.Rhoades O'Brien Resources, LLC 3003-190 Texas Nacogdoche 42:347.14280.0020 12:15-2008 William C.Rhoades O'Brien Resources, LLC 3003-190 Texas Nacogdoche 42:347.14280.0021 12:23-2008 Sandra Lee O'rr Marrufo O'Brien Resources, LLC 3003-190 Texas Nacogdoche 42:347.14280.0021 12:25-2008 Unima Whitsoid O'Brien Resources, LLC 3003-190 Texas Nacogdoche	42.347.14280.0001	04-08-2008		O'Brien Resources, LLC	2833-271	Texas	Nacogdoche
42 347.14280.0004	42.347.14280.0002	05-27-2008	Dorothy Jean Staine	O'Brien Resources, LLC	2871-167	Texas	Nacogdoche
42 347.14280.0005 07-23-2008 John C. Williamson O'Brien Resources, LLC 2904-171 Texas Nacogdoche 42 347.14280.0006 11-24-2008 Gerald Douglas Whitton O'Brien Resources, LLC 3003-181 Texas Nacogdoche 42 347.14280.0007 11-24-2008 Bobby Gene Whitton, Sr. O'Brien Resources, LLC 3003-175 Texas Nacogdoche 42 347.14280.0008 12-01-2008 William F. Winkler, Jr. O'Brien Resources, LLC 3003-175 Texas Nacogdoche 42 347.14280.0009 12-01-2008 Jamet Winkler Davis O'Brien Resources, LLC 3003-175 Texas Nacogdoche 42 347.14280.0001 11-24-2008 Jamet Winkler Davis O'Brien Resources, LLC 3003-175 Texas Nacogdoche 42 347.14280.0011 12-17-2008 Gloria Drummord O'Brien Resources, LLC 3003-160 Texas Nacogdoche 42 347.14280.0011 12-17-2008 Gloria Drummord O'Brien Resources, LLC 3003-160 Texas Nacogdoche 42 347.14280.0011 12-17-2008 Gloria Drummord O'Brien Resources, LLC 3003-160 Texas Nacogdoche 42 347.14280.0011 12-03-2008 Grace P. Whitton et al O'Brien Resources, LLC 3003-150 Texas Nacogdoche 42 347.14280.0011 12-03-2008 Marion O'R Bacon O'Brien Resources, LLC 3003-150 Texas Nacogdoche 42 347.14280.0011 12-03-2008 Bamela A. Martin O'Brien Resources, LLC 3003-163 Texas Nacogdoche 42 347.14280.0016 11-24-2008 Marion O'R Martin O'Brien Resources, LLC 3003-163 Texas Nacogdoche 42 347.14280.0016 11-24-2008 Bally M. Whitton O'Brien Resources, LLC 3003-184 Texas Nacogdoche 42 347.14280.0016 11-24-2008 William C. Rhoades O'Brien Resources, LLC 3003-187 Texas Nacogdoche 42 347.14280.0019 12-25-2008 Sally Am O'R O'Brien Resources, LLC 3003-190 Texas Nacogdoche 42 347.14280.0021 12-25-2008 Sally Am O'R O'Brien Resources, LLC 3003-190 Texas Nacogdoche 42 347.14280.0021 12-23-2008 Sally Am O'R O'Brien Resources, LLC 3003-190 Texas Nacogdoche 42 347.14280.0021 12-23-2008 Sally Am O'R O'Brien Resources, LLC 3003-190 Texas Nacogdoche 42 347.14280.0022 01-19-2009 John Thompson et ux O'Brien Resources, LLC 3003-190 Texas Nacogdoche 42 347.14280.0022 01-19-2009 John Thompson et ux O'Brien Resources, LLC 3003-202 Texas Nacogdoche 42 347.14280.0023 11-24-200	42.347.14280.0003	01-28-2008	Clara Little	O'Brien Resources, LLC	1833-268	Texas	Nacogdoche
42 347.14280.0006 11-24-2008 Gerald Douglas Whitton O'Brien Resources, LLC 3003-181 Texas Nacogdoche 42 347.14280.0007 11-24-2008 Bobby Gene Whitton, Sr. O'Brien Resources, LLC 3003-176 Texas Nacogdoche 42 347.14280.0008 12-01-2008 Julian F. Winkler, Jr. O'Brien Resources, LLC 3003-175 Texas Nacogdoche 42 347.14280.0009 12-01-2008 Jamet Winkler Davis O'Brien Resources, LLC 3003-175 Texas Nacogdoche 42 347.14280.0010 11-24-2008 Jamet Winkler Davis O'Brien Resources, LLC 3003-169 Texas Nacogdoche 42 347.14280.0011 12-01-2008 Grace P. Whitton et al O'Brien Resources, LLC 3003-160 Texas Nacogdoche 42 347.14280.0013 12-01-2008 Marion O'R Bacon O'Brien Resources, LLC 3003-163 Texas Nacogdoche 42 347.14280.0014 11-20-2008 Alery K. Whitton O'Brien Resources, LLC 3003-160 Texas Nacogdoche 42 347.14280.0015 11-19-2008 Rerry K. Whitton O'Brien Resources, LLC	42.347.14280.0004	01-24-2008	Odessa S. Powdrill	O'Brien Resources, LLC	1833-265	Texas	Nacogdoche
42:347.14280.0007	42.347.14280.0005	07-23-2008	John C. Williamson	O'Brien Resources, LLC	2904-171	Texas	Nacogdoche
42:347.14280.0008 12-01-2008 William F, Winkler, Jr. O'Brien Resources, LLC 3003-175 Texas Nacogdoche 42:347.14280.0009 12-01-2008 Jamet Winkler Davis O'Brien Resources, LLC 3003-172 Texas Nacogdoche 42:347.14280.0010 11-24-2008 Jerry L. Whitton O'Brien Resources, LLC 3003-169 Texas Nacogdoche 42:347.14280.0011 12-17-2008 Gloria Drummond O'Brien Resources, LLC 3003-165 Texas Nacogdoche 42:347.14280.0012 12-03-2008 Grace P. Whitton et al O'Brien Resources, LLC 3003-153 Texas Nacogdoche 42:347.14280.0013 12-01-2008 Barion Orr Bacon O'Brien Resources, LLC 3003-153 Texas Nacogdoche 42:347.14280.0014 12-01-2008 Famela A. Martin O'Brien Resources, LLC 3003-163 Texas Nacogdoche 42:347.14280.0015 11-19-2008 Kerry K. Whitton O'Brien Resources, LLC 3003-167 Texas Nacogdoche 42:347.14280.0012 11-24-2008 Alice M. Whitton O'Brien Resources, LLC 3003-19	42.347.14280.0006	11-24-2008	Gerald Douglas Whitton	O'Brien Resources, LLC	3003-181	Texas	Nacogdoche
42:347.14280.0009 12-01-2008 Janet Winkler Davis O'Brien Resources, LLC 3003-172 Texas Nacogdoche 42:347.14280.0010 11-24-2008 Jerry L. Whitton O'Brien Resources, LLC 3003-169 Texas Nacogdoche 42:347.14280.0011 12-17-2008 Gloria Drummond O'Brien Resources, LLC 3003-160 Texas Nacogdoche 42:347.14280.0012 12-03-2008 Marion Orr Bacon O'Brien Resources, LLC 3003-153 Texas Nacogdoche 42:347.14280.0014 12-01-2008 Marion Orr Bacon O'Brien Resources, LLC 3003-163 Texas Nacogdoche 42:347.14280.0015 11-19-2008 Kerry K. Whitton O'Brien Resources, LLC 3003-160 Texas Nacogdoche 42:347.14280.0016 11-24-2008 Alice M. Whitton O'Brien Resources, LLC 3003-160 Texas Nacogdoche 42:347.14280.0017 01-05-2009 Sally G. Mattingly O'Brien Resources, LLC 3003-167 Texas Nacogdoche 42:347.14280.0018 12-23-2008 Sandra Lee Orr Marrufo O'Brien Resources, LLC 3003-187 <td>42.347.14280.0007</td> <td>11-24-2008</td> <td>Bobby Gene Whitton, Sr.</td> <td>O'Brien Resources, LLC</td> <td>3003-178</td> <td>Texas</td> <td>Nacogdoche</td>	42.347.14280.0007	11-24-2008	Bobby Gene Whitton, Sr.	O'Brien Resources, LLC	3003-178	Texas	Nacogdoche
42.347.14280.0010 11-24-2008 Jerry L. Whitton O'Brien Resources, LLC 3003-169 Texas Nacogdoche 42.347.14280.0011 12-17-2008 Gloria Drummond O'Brien Resources, LLC 3003-166 Texas Nacogdoche 42.347.14280.0012 12-03-2008 Grace P. Whitton et al O'Brien Resources, LLC 3003-150 Texas Nacogdoche 42.347.14280.0013 12-01-2008 Marion Orr Bacon O'Brien Resources, LLC 3003-153 Texas Nacogdoche 42.347.14280.0014 12-01-2008 Pamela A. Martin O'Brien Resources, LLC 3003-160 Texas Nacogdoche 42.347.14280.0016 11-24-2008 Alice M. Whitton O'Brien Resources, LLC 3003-167 Texas Nacogdoche 42.347.14280.0017 01-05-2009 Sally G. Mattingly O'Brien Resources, LLC 3003-187 Texas Nacogdoche 42.347.14280.0019 12-15-2008 William C.Rhoades O'Brien Resources, LLC 3003-187 Texas Nacogdoche 42.347.14280.0020 11-24-2008 Donna Michele W. McLallen O'Brien Resources, LLC 3003-	42.347.14280.0008	12-01-2008	William F. Winkler, Jr.	O'Brien Resources, LLC	3003-175	Texas	Nacogdoche
42.347.14280.0011 12-17-2008 Gloria Drummond O'Brien Resources, LLC 3003-166 Texas Nacogdoche 42.347.14280.0012 12-03-2008 Grace P. Whitton et al O'Brien Resources, LLC 3003-150 Texas Nacogdoche 42.347.14280.0013 12-01-2008 Marion Orr Bacon O'Brien Resources, LLC 3003-153 Texas Nacogdoche 42.347.14280.0014 12-01-2008 Pamela A. Martin O'Brien Resources, LLC 3003-160 Texas Nacogdoche 42.347.14280.0015 11-19-2008 Kerry K. Whitton O'Brien Resources, LLC 3003-160 Texas Nacogdoche 42.347.14280.0016 11-24-2008 Alice M. Whitton O'Brien Resources, LLC 3003-187 Texas Nacogdoche 42.347.14280.0017 01-05-2009 Sally G. Mattingly O'Brien Resources, LLC 3003-187 Texas Nacogdoche 42.347.14280.0019 12-15-2008 William C.Rhoades O'Brien Resources, LLC 3003-190 Texas Nacogdoche 42.347.14280.0020 11-24-2008 Donna Michele W. McLallen O'Brien Resources, LLC 3003-	42.347.14280.0009	12-01-2008	Janet Winkler Davis	O'Brien Resources, LLC	3003-172	Texas	Nacogdoche
42.347.14280.0012 12.03-2008 Grace P. Whitton et al O'Brien Resources, LLC 3003-150 Texas Nacogdoche 42.347.14280.0013 12.01-2008 Marion Orr Bacon O'Brien Resources, LLC 3003-153 Texas Nacogdoche 42.347.14280.0014 12.01-2008 Pamela A. Martin O'Brien Resources, LLC 3003-160 Texas Nacogdoche 42.347.14280.0015 11-19-2008 Kerry K. Whitton O'Brien Resources, LLC 3003-160 Texas Nacogdoche 42.347.14280.0016 11-24-2008 Alice M. Whitton O'Brien Resources, LLC 3003-157 Texas Nacogdoche 42.347.14280.0017 01-05-2009 Sally G. Mattingly O'Brien Resources, LLC 3003-184 Texas Nacogdoche 42.347.14280.0018 12-23-2008 Sandra Lee Orr Marrufo O'Brien Resources, LLC 3003-187 Texas Nacogdoche 42.347.14280.0020 11-24-2008 Donna Michele W. McLallen O'Brien Resources, LLC 3003-193 Texas Nacogdoche 42.347.14280.0021 12-23-2008 Sally Ann Orr O'Brien Resources, LLC 30	42.347.14280.0010	11-24-2008	Jerry L. Whitton	O'Brien Resources, LLC	3003-169	Texas	Nacogdoche
42.347.14280.0013 12-01-2008 Marion Orr Bacon O'Brien Resources, LLC 3003-153 Texas Nacogdoche 42.347.14280.0014 12-01-2008 Pamela A. Martin O'Brien Resources, LLC 3003-163 Texas Nacogdoche 42.347.14280.0015 11-19-2008 Kerry K. Whitton O'Brien Resources, LLC 3003-160 Texas Nacogdoche 42.347.14280.0016 11-24-2008 Alice M. Whitton O'Brien Resources, LLC 3003-157 Texas Nacogdoche 42.347.14280.0017 01-05-2009 Sally G. Mattingly O'Brien Resources, LLC 3003-184 Texas Nacogdoche 42.347.14280.0018 12-23-2008 Sandra Lee Orr Marrufo O'Brien Resources, LLC 3003-190 Texas Nacogdoche 42.347.14280.0020 11-24-2008 Donna Michele W. McLallen O'Brien Resources, LLC 3003-193 Texas Nacogdoche 42.347.14280.0021 12-23-2008 Sally Ann Orr O'Brien Resources, LLC 3003-193 Texas Nacogdoche 42.347.14280.0023 11-24-2008 William Paul Whitton O'Brien Resources, LLC 3003	42.347.14280.0011	12-17-2008	Gloria Drummond	O'Brien Resources, LLC	3003-166	Texas	Nacogdoche
42.347.14280.0014 12-01-2008 Pamela A. Martin O'Brien Resources, LLC 3003-163 Texas Nacogdoche 42.347.14280.0015 11-19-2008 Kerry K. Whitton O'Brien Resources, LLC 3003-160 Texas Nacogdoche 42.347.14280.0016 11-24-2008 Alice M. Whitton O'Brien Resources, LLC 3003-157 Texas Nacogdoche 42.347.14280.0017 01-05-2009 Sally G. Mattingly O'Brien Resources, LLC 3003-184 Texas Nacogdoche 42.347.14280.0018 12-23-2008 Sandra Lee Orr Marrufo O'Brien Resources, LLC 3003-197 Texas Nacogdoche 42.347.14280.0021 12-23-2008 Sally Ann Orr O'Brien Resources, LLC 3003-193 Texas Nacogdoche 42.347.14280.0021 12-23-2008 Sally Ann Orr O'Brien Resources, LLC 3003-196 Texas Nacogdoche 42.347.14280.0022 09-01-2008 John Thompson et ux O'Brien Resources, LLC 3003-199 Texas Nacogdoche 42.347.14280.0022 11-24-2008 William Paul Whitton O'Brien Resources, LLC 3003-199	42.347.14280.0012	12-03-2008	Grace P. Whitton et al	O'Brien Resources, LLC	3003-150	Texas	Nacogdoche
42.347.14280.0015 11-19-2008 Kerry K. Whitton O'Brien Resources, LLC 3003-160 Texas Nacogdoche 42.347.14280.0016 11-24-2008 Alice M. Whitton O'Brien Resources, LLC 3003-157 Texas Nacogdoche 42.347.14280.0017 01-05-2009 Sally G. Mattingly O'Brien Resources, LLC 3003-184 Texas Nacogdoche 42.347.14280.0018 12-23-2008 Sandra Lee Orr Marrufo O'Brien Resources, LLC 3003-187 Texas Nacogdoche 42.347.14280.0019 12-15-2008 William C.Rhoades O'Brien Resources, LLC 3003-190 Texas Nacogdoche 42.347.14280.0020 11-24-2008 Donna Michele W. McLallen O'Brien Resources, LLC 3003-196 Texas Nacogdoche 42.347.14280.0021 12-23-2008 Sally Ann Orr O'Brien Resources, LLC 3003-196 Texas Nacogdoche 42.347.14280.0022 12-20-208 John Thompson et ux O'Brien Resources, LLC 3003-199 Texas Nacogdoche 42.347.14280.0023 11-24-2008 William Paul Whitton O'Brien Resources, LLC 3	42.347.14280.0013	12-01-2008	Marion Orr Bacon	O'Brien Resources, LLC	3003-153	Texas	Nacogdoche
42.347.14280.0016 11-24-2008 Alice M. Whitton O'Brien Resources, LLC 3003-157 Texas Nacogdoche 42.347.14280.0017 01-05-2009 Sally G. Mattingly O'Brien Resources, LLC 3003-184 Texas Nacogdoche 42.347.14280.0018 12-23-2008 Sandra Lee Orr Marrufo O'Brien Resources, LLC 3003-187 Texas Nacogdoche 42.347.14280.0019 12-15-2008 William C.Rhoades O'Brien Resources, LLC 3003-190 Texas Nacogdoche 42.347.14280.0020 11-24-2008 Donna Michele W. McLallen O'Brien Resources, LLC 3003-193 Texas Nacogdoche 42.347.14280.0021 12-23-2008 Sally Ann Orr O'Brien Resources, LLC 3003-196 Texas Nacogdoche 42.347.14280.0022 09-01-2008 John Thompson et ux O'Brien Resources, LLC 2922-170 Texas Nacogdoche 42.347.14280.0023 11-24-2008 William Paul Whitton O'Brien Resources, LLC 3003-199 Texas Nacogdoche 42.347.14280.0024 01-19-2009 Diane Whiteside O'Brien Resources, LLC 3	42.347.14280.0014	12-01-2008	Pamela A. Martin	O'Brien Resources, LLC	3003-163	Texas	Nacogdoche
42.347.14280.0017 01-05-2009 Sally G. Mattingly O'Brien Resources, LLC 3003-184 Texas Nacogdoche 42.347.14280.0018 12-23-2008 Sandra Lee Orr Marrufo O'Brien Resources, LLC 3003-187 Texas Nacogdoche 42.347.14280.0019 12-15-2008 William C.Rhoades O'Brien Resources, LLC 3003-190 Texas Nacogdoche 42.347.14280.0020 11-24-2008 Donna Michele W. McLallen O'Brien Resources, LLC 3003-193 Texas Nacogdoche 42.347.14280.0021 12-23-2008 Sally Ann Orr O'Brien Resources, LLC 3003-196 Texas Nacogdoche 42.347.14280.0022 09-01-2008 John Thompson et ux O'Brien Resources, LLC 2922-170 Texas Nacogdoche 42.347.14280.0023 11-24-2008 William Paul Whitton O'Brien Resources, LLC 3003-199 Texas Nacogdoche 42.347.14280.0024 01-19-2009 Diane Whiteside O'Brien Resources, LLC 3003-205 Texas Nacogdoche 42.347.14280.0025 01-19-2009 Terri Lynn Jensen O'Brien Resources, LLC	42.347.14280.0015	11-19-2008	Kerry K. Whitton	O'Brien Resources, LLC	3003-160	Texas	Nacogdoche
42.347.14280.0018 12-23-2008 Sandra Lee Orr Marrufo O'Brien Resources, LLC 3003-187 Texas Nacogdoche 42.347.14280.0019 12-15-2008 William C.Rhoades O'Brien Resources, LLC 3003-190 Texas Nacogdoche 42.347.14280.0020 11-24-2008 Donna Michele W. McLallen O'Brien Resources, LLC 3003-193 Texas Nacogdoche 42.347.14280.0021 12-23-2008 Sally Ann Orr O'Brien Resources, LLC 3003-196 Texas Nacogdoche 42.347.14280.0022 09-01-2008 John Thompson et ux O'Brien Resources, LLC 2922-170 Texas Nacogdoche 42.347.14280.0023 11-24-2008 William Paul Whitton O'Brien Resources, LLC 3003-199 Texas Nacogdoche 42.347.14280.0024 01-19-2009 Diane Whiteside O'Brien Resources, LLC 3003-205 Texas Nacogdoche 42.347.14280.0025 01-19-2009 Terri Lynn Jensen O'Brien Resources, LLC 3003-208 Texas Nacogdoche 42.347.14280.0027 01-16-2009 Bettie B. Griffin, widow O'Brien Resources, LLC	42.347.14280.0016	11-24-2008	Alice M. Whitton	O'Brien Resources, LLC	3003-157	Texas	Nacogdoche
42.347.14280.0019 12-15-2008 William C.Rhoades O'Brien Resources, LLC 3003-190 Texas Nacogdoche 42.347.14280.0020 11-24-2008 Donna Michele W. McLallen O'Brien Resources, LLC 3003-193 Texas Nacogdoche 42.347.14280.0021 12-23-2008 Sally Ann Orr O'Brien Resources, LLC 3003-196 Texas Nacogdoche 42.347.14280.0022 09-01-2008 John Thompson et ux O'Brien Resources, LLC 2922-170 Texas Nacogdoche 42.347.14280.0023 11-24-2008 William Paul Whitton O'Brien Resources, LLC 3003-199 Texas Nacogdoche 42.347.14280.0024 01-19-2009 Diane Whiteside O'Brien Resources, LLC 3003-205 Texas Nacogdoche 42.347.14280.0025 01-19-2009 Terri Lynn Jensen O'Brien Resources, LLC 3003-208 Texas Nacogdoche 42.347.14280.0026 01-19-2009 Terri Lynn Jensen O'Brien Resources, LLC 3003-208 Texas Nacogdoche 42.347.14280.0027 01-16-2009 Bettie B. Griffin, widow O'Brien Resources, LLC <td< td=""><td>42.347.14280.0017</td><td>01-05-2009</td><td>Sally G. Mattingly</td><td>O'Brien Resources, LLC</td><td>3003-184</td><td>Texas</td><td>Nacogdoche</td></td<>	42.347.14280.0017	01-05-2009	Sally G. Mattingly	O'Brien Resources, LLC	3003-184	Texas	Nacogdoche
42.347.14280.0020 11-24-2008 Donna Michele W. McLallen O'Brien Resources, LLC 3003-193 Texas Nacogdoche 42.347.14280.0021 12-23-2008 Sally Ann Orr O'Brien Resources, LLC 3003-196 Texas Nacogdoche 42.347.14280.0022 09-01-2008 John Thompson et ux O'Brien Resources, LLC 2922-170 Texas Nacogdoche 42.347.14280.0023 11-24-2008 William Paul Whitton O'Brien Resources, LLC 3003-199 Texas Nacogdoche 42.347.14280.0024 01-19-2009 Diane Whiteside O'Brien Resources, LLC 3003-205 Texas Nacogdoche 42.347.14280.0025 01-19-2009 Carol Whiteside O'Brien Resources, LLC 3003-202 Texas Nacogdoche 42.347.14280.0026 01-19-2009 Terri Lynn Jensen O'Brien Resources, LLC 3003-208 Texas Nacogdoche 42.347.14280.0027 01-16-2009 Bettie B. Griffin, widow O'Brien Resources, LLC 3024-218 Texas Nacogdoche 42.347.14280.0028 12-10-2008 Marian B. Whiteside Lattin O'Brien Resources, LLC	42.347.14280.0018	12-23-2008	Sandra Lee Orr Marrufo	O'Brien Resources, LLC	3003-187	Texas	Nacogdoche
42.347.14280.0021 12-23-2008 Sally Ann Orr O'Brien Resources, LLC 3003-196 Texas Nacogdoche 42.347.14280.0022 09-01-2008 John Thompson et ux O'Brien Resources, LLC 2922-170 Texas Nacogdoche 42.347.14280.0023 11-24-2008 William Paul Whitton O'Brien Resources, LLC 3003-199 Texas Nacogdoche 42.347.14280.0024 01-19-2009 Diane Whiteside O'Brien Resources, LLC 3003-205 Texas Nacogdoche 42.347.14280.0025 01-19-2009 Carol Whiteside O'Brien Resources, LLC 3003-202 Texas Nacogdoche 42.347.14280.0026 01-19-2009 Terri Lynn Jensen O'Brien Resources, LLC 3003-208 Texas Nacogdoche 42.347.14280.0027 01-16-2009 Bettie B. Griffin, widow O'Brien Resources, LLC 2995-218 Texas Nacogdoche 42.347.14280.0028 12-10-2008 Marian B. Whiteside Lattin O'Brien Resources, LLC 3024-172 Texas Nacogdoche 42.347.14280.0030 02-15-2008 Robert James Rhoades O'Brien Resources, LLC <	42.347.14280.0019	12-15-2008	William C.Rhoades	O'Brien Resources, LLC	3003-190	Texas	Nacogdoche
42.347.14280.0022 09-01-2008 John Thompson et ux O'Brien Resources, LLC 2922-170 Texas Nacogdoche 42.347.14280.0023 11-24-2008 William Paul Whitton O'Brien Resources, LLC 3003-199 Texas Nacogdoche 42.347.14280.0024 01-19-2009 Diane Whiteside O'Brien Resources, LLC 3003-205 Texas Nacogdoche 42.347.14280.0025 01-19-2009 Carol Whiteside O'Brien Resources, LLC 3003-202 Texas Nacogdoche 42.347.14280.0026 01-19-2009 Terri Lynn Jensen O'Brien Resources, LLC 3003-208 Texas Nacogdoche 42.347.14280.0027 01-16-2009 Bettie B. Griffin, widow O'Brien Resources, LLC 2995-218 Texas Nacogdoche 42.347.14280.0028 12-10-2008 Marian B. Whiteside Lattin O'Brien Resources, LLC 3024-172 Texas Nacogdoche 42.347.14280.0030 02-11-2009 Patsy Ann Sturrock O'Brien Resources, LLC 3024-217 Texas Nacogdoche 42.347.14280.0031 02-23-2012 Lewin O'Neil Samford O'Brien Resources, LLC	42.347.14280.0020	11-24-2008	Donna Michele W. McLallen	O'Brien Resources, LLC	3003-193	Texas	Nacogdoche
42.347.14280.0023 11-24-2008 William Paul Whitton O'Brien Resources, LLC 3003-199 Texas Nacogdoche 42.347.14280.0024 01-19-2009 Diane Whiteside O'Brien Resources, LLC 3003-205 Texas Nacogdoche 42.347.14280.0025 01-19-2009 Carol Whiteside O'Brien Resources, LLC 3003-202 Texas Nacogdoche 42.347.14280.0026 01-19-2009 Terri Lynn Jensen O'Brien Resources, LLC 3003-208 Texas Nacogdoche 42.347.14280.0027 01-16-2009 Bettie B. Griffin, widow O'Brien Resources, LLC 2995-218 Texas Nacogdoche 42.347.14280.0028 12-10-2008 Marian B. Whiteside Lattin O'Brien Resources, LLC 3024-172 Texas Nacogdoche 42.347.14280.0029 12-15-2008 Robert James Rhoades O'Brien Resources, LLC 3024-172 Texas Nacogdoche 42.347.14280.0030 02-11-2009 Patsy Ann Sturrock O'Brien Resources, LLC 3024-217 Texas Nacogdoche 42.347.14280.0031 02-23-2012 Lewin O'Neil Samford O'Brien Resources, LLC 3024-208 Texas Nacogdoche 42.347.14280.	42.347.14280.0021	12-23-2008	Sally Ann Orr	O'Brien Resources, LLC	3003-196	Texas	Nacogdoche
42.347.14280.0024 01-19-2009 Diane Whiteside O'Brien Resources, LLC 3003-205 Texas Nacogdoche 42.347.14280.0025 01-19-2009 Carol Whiteside O'Brien Resources, LLC 3003-202 Texas Nacogdoche 42.347.14280.0026 01-19-2009 Terri Lynn Jensen O'Brien Resources, LLC 3003-208 Texas Nacogdoche 42.347.14280.0027 01-16-2009 Bettie B. Griffin, widow O'Brien Resources, LLC 2995-218 Texas Nacogdoche 42.347.14280.0028 12-10-2008 Marian B. Whiteside Lattin O'Brien Resources, LLC 3024-172 Texas Nacogdoche 42.347.14280.0029 12-15-2008 Robert James Rhoades O'Brien Resources, LLC 3024-169 Texas Nacogdoche 42.347.14280.0030 02-11-2009 Patsy Ann Sturrock O'Brien Resources, LLC 3024-217 Texas Nacogdoche 42.347.14280.0031 02-23-2012 Lewin O'Neil Samford O'Brien Resources, LLC 3024-201 Texas Nacogdoche 42.347.14280.0033 02-19-2009 J. W. Samford O'Brien Resources, LLC <t< td=""><td>42.347.14280.0022</td><td>09-01-2008</td><td>John Thompson et ux</td><td>O'Brien Resources, LLC</td><td>2922-170</td><td>Texas</td><td>Nacogdoche</td></t<>	42.347.14280.0022	09-01-2008	John Thompson et ux	O'Brien Resources, LLC	2922-170	Texas	Nacogdoche
42.347.14280.0025 01-19-2009 Carol Whiteside O'Brien Resources, LLC 3003-202 Texas Nacogdoche 42.347.14280.0026 01-19-2009 Terri Lynn Jensen O'Brien Resources, LLC 3003-208 Texas Nacogdoche 42.347.14280.0027 01-16-2009 Bettie B. Griffin, widow O'Brien Resources, LLC 2995-218 Texas Nacogdoche 42.347.14280.0028 12-10-2008 Marian B. Whiteside Lattin O'Brien Resources, LLC 3024-172 Texas Nacogdoche 42.347.14280.0029 12-15-2008 Robert James Rhoades O'Brien Resources, LLC 3024-169 Texas Nacogdoche 42.347.14280.0030 02-11-2009 Patsy Ann Sturrock O'Brien Resources, LLC 3024-217 Texas Nacogdoche 42.347.14280.0031 02-23-2012 Lewin O'Neil Samford, et al O'Brien Resources, LLC 3024-211 Texas Nacogdoche 42.347.14280.0032 02-19-2009 Mac Samford O'Brien Resources, LLC 3024-208 Texas Nacogdoche 42.347.14280.0033 02-19-2009 J. W. Samford O'Brien Resources, LLC	42.347.14280.0023	11-24-2008	William Paul Whitton	O'Brien Resources, LLC	3003-199	Texas	Nacogdoche
42.347.14280.0026 01-19-2009 Terri Lynn Jensen O'Brien Resources, LLC 3003-208 Texas Nacogdoche 42.347.14280.0027 01-16-2009 Bettie B. Griffin, widow O'Brien Resources, LLC 2995-218 Texas Nacogdoche 42.347.14280.0028 12-10-2008 Marian B. Whiteside Lattin O'Brien Resources, LLC 3024-172 Texas Nacogdoche 42.347.14280.0029 12-15-2008 Robert James Rhoades O'Brien Resources, LLC 3024-169 Texas Nacogdoche 42.347.14280.0030 02-11-2009 Patsy Ann Sturrock O'Brien Resources, LLC 3024-217 Texas Nacogdoche 42.347.14280.0031 02-23-2012 Lewin O'Neil Samford, et al O'Brien Resources, LLC 3024-211 Texas Nacogdoche 42.347.14280.0032 02-19-2009 Mac Samford O'Brien Resources, LLC 3024-208 Texas Nacogdoche 42.347.14280.0033 02-19-2009 J. W. Samford O'Brien Resources, LLC 3024-205 Texas Nacogdoche 42.347.14280.0034 02-19-2009 Jimmy D. Samford O'Brien Resources, LLC	42.347.14280.0024	01-19-2009	Diane Whiteside	O'Brien Resources, LLC	3003-205	Texas	Nacogdoche
42.347.14280.0027 01-16-2009 Bettie B. Griffin, widow O'Brien Resources, LLC 2995-218 Texas Nacogdoche 42.347.14280.0028 12-10-2008 Marian B. Whiteside Lattin O'Brien Resources, LLC 3024-172 Texas Nacogdoche 42.347.14280.0029 12-15-2008 Robert James Rhoades O'Brien Resources, LLC 3024-169 Texas Nacogdoche 42.347.14280.0030 02-11-2009 Patsy Ann Sturrock O'Brien Resources, LLC 3024-217 Texas Nacogdoche 42.347.14280.0031 02-23-2012 Lewin O'Neil Samford, et al O'Brien Resources, LLC 3024-211 Texas Nacogdoche 42.347.14280.0032 02-19-2009 Mac Samford O'Brien Resources, LLC 3024-208 Texas Nacogdoche 42.347.14280.0033 02-19-2009 J. W. Samford O'Brien Resources, LLC 3024-205 Texas Nacogdoche 42.347.14280.0034 02-19-2009 Jimmy D. Samford O'Brien Resources, LLC 3024-202 Texas Nacogdoche	42.347.14280.0025	01-19-2009	Carol Whiteside	O'Brien Resources, LLC	3003-202	Texas	Nacogdoche
42.347.14280.0027 01-16-2009 Bettie B. Griffin, widow O'Brien Resources, LLC 2995-218 Texas Nacogdoche 42.347.14280.0028 12-10-2008 Marian B. Whiteside Lattin O'Brien Resources, LLC 3024-172 Texas Nacogdoche 42.347.14280.0029 12-15-2008 Robert James Rhoades O'Brien Resources, LLC 3024-169 Texas Nacogdoche 42.347.14280.0030 02-11-2009 Patsy Ann Sturrock O'Brien Resources, LLC 3024-217 Texas Nacogdoche 42.347.14280.0031 02-23-2012 Lewin O'Neil Samford, et al O'Brien Resources, LLC 3024-211 Texas Nacogdoche 42.347.14280.0032 02-19-2009 Mac Samford O'Brien Resources, LLC 3024-208 Texas Nacogdoche 42.347.14280.0033 02-19-2009 J. W. Samford O'Brien Resources, LLC 3024-205 Texas Nacogdoche 42.347.14280.0034 02-19-2009 Jimmy D. Samford O'Brien Resources, LLC 3024-202 Texas Nacogdoche	42.347.14280.0026	01-19-2009	Terri Lynn Jensen	O'Brien Resources, LLC	3003-208	Texas	Nacogdoche
42.347.14280.0028 12-10-2008 Marian B. Whiteside Lattin O'Brien Resources, LLC 3024-172 Texas Nacogdoche 42.347.14280.0029 12-15-2008 Robert James Rhoades O'Brien Resources, LLC 3024-169 Texas Nacogdoche 42.347.14280.0030 02-11-2009 Patsy Ann Sturrock O'Brien Resources, LLC 3024-217 Texas Nacogdoche 42.347.14280.0031 02-23-2012 Lewin O'Neil Samford, et al O'Brien Resources, LLC 3024-211 Texas Nacogdoche 42.347.14280.0032 02-19-2009 Mac Samford O'Brien Resources, LLC 3024-208 Texas Nacogdoche 42.347.14280.0033 02-19-2009 J. W. Samford O'Brien Resources, LLC 3024-205 Texas Nacogdoche 42.347.14280.0034 02-19-2009 Jimmy D. Samford O'Brien Resources, LLC 3024-202 Texas Nacogdoche	42.347.14280.0027	01-16-2009	Bettie B. Griffin, widow	O'Brien Resources, LLC	2995-218	Texas	
42.347.14280.0029 12-15-2008 Robert James Rhoades O'Brien Resources, LLC 3024-169 Texas Nacogdoche 42.347.14280.0030 02-11-2009 Patsy Ann Sturrock O'Brien Resources, LLC 3024-217 Texas Nacogdoche 42.347.14280.0031 02-23-2012 Lewin O'Neil Samford, et al O'Brien Resources, LLC 3024-211 Texas Nacogdoche 42.347.14280.0032 02-19-2009 Mac Samford O'Brien Resources, LLC 3024-208 Texas Nacogdoche 42.347.14280.0033 02-19-2009 J. W. Samford O'Brien Resources, LLC 3024-205 Texas Nacogdoche 42.347.14280.0034 02-19-2009 Jimmy D. Samford O'Brien Resources, LLC 3024-202 Texas Nacogdoche	42.347.14280.0028	12-10-2008	Marian B. Whiteside Lattin	O'Brien Resources, LLC	3024-172	Texas	
42.347.14280.0031 02-23-2012 Lewin O'Neil Samford, et al O'Brien Resources, LLC 3024-211 Texas Nacogdoche 42.347.14280.0032 02-19-2009 Mac Samford O'Brien Resources, LLC 3024-208 Texas Nacogdoche 42.347.14280.0033 02-19-2009 J. W. Samford O'Brien Resources, LLC 3024-205 Texas Nacogdoche 42.347.14280.0034 02-19-2009 Jimmy D. Samford O'Brien Resources, LLC 3024-202 Texas Nacogdoche	42.347.14280.0029	12-15-2008	Robert James Rhoades	O'Brien Resources, LLC	3024-169	Texas	Nacogdoche
42.347.14280.0031 02-23-2012 Lewin O'Neil Samford, et al O'Brien Resources, LLC 3024-211 Texas Nacogdoche 42.347.14280.0032 02-19-2009 Mac Samford O'Brien Resources, LLC 3024-208 Texas Nacogdoche 42.347.14280.0033 02-19-2009 J. W. Samford O'Brien Resources, LLC 3024-205 Texas Nacogdoche 42.347.14280.0034 02-19-2009 Jimmy D. Samford O'Brien Resources, LLC 3024-202 Texas Nacogdoche	42.347.14280.0030	02-11-2009	Patsy Ann Sturrock	O'Brien Resources, LLC	3024-217	Texas	Nacogdoche
42.347.14280.0032 02-19-2009 Mac Samford O'Brien Resources, LLC 3024-208 Texas Nacogdoche 42.347.14280.0033 02-19-2009 J. W. Samford O'Brien Resources, LLC 3024-205 Texas Nacogdoche 42.347.14280.0034 02-19-2009 Jimmy D. Samford O'Brien Resources, LLC 3024-202 Texas Nacogdoche	42.347.14280.0031			O'Brien Resources, LLC	3024-211	Texas	Nacogdoche
42.347.14280.0034 02-19-2009 Jimmy D. Samford O'Brien Resources, LLC 3024-202 Texas Nacogdoche	42.347.14280.0032	02-19-2009	Mac Samford	O'Brien Resources, LLC	3024-208	Texas	Nacogdoche
-	42.347.14280.0033	02-19-2009	J. W. Samford	O'Brien Resources, LLC	3024-205	Texas	Nacogdoche
	42.347.14280.0034	02-19-2009	Jimmy D. Samford	O'Brien Resources, LLC	3024-202	Texas	Nacogdoche
	-	02-19-2009	Kay Samford Sanders	O'Brien Resources, LLC	3024-199	Texas	Nacogdoche

LEASE#	DATE	LESSOR	LESSEE	BOOK-PAGE	STATE	COUNTY
42.347.14280.0036	02-17-2009	Paatsy Ann Samford Sturrock et al	O'Brien Resources, LLC	3024-196	Texas	Nacogdoche
42.347.14280.0037	01-23-2009	Sandra Turner	O'Brien Resources, LLC	3024-214	Texas	Nacogdoche
42.347.14280.0038	02-24-2009	Tracy Lee Turner	O'Brien Resources, LLC	3024-193	Texas	Nacogdoche
42.347.14280.0039	02-24-2009	Troy Scott Turner	O'Brien Resources, LLC	3024-190	Texas	Nacogdoche
42.347.14280.0040	01-19-2012	Kathleen Whiteside Ratliff	W. Stockman Ext	3024-175	Texas	Nacogdoche
42.347.14280.0041	01-19-2012	Julie Whiteside Watson	W. Stockman Ext	3024-178	Texas	Nacogdoche
42.347.14280.0042	03-11-2009	C. D. Mills et ux	O'Brien Resources, LLC	3024-220	Texas	Nacogdoche
42.347.14280.0043	01-19-2009	Brenda Whiteside	O'Brien Resources, LLC	3024-181	Texas	Nacogdoche
42.347.14280.0044	01-19-2009	Darcie Whiteside	O'Brien Resources, LLC	3024-184	Texas	Nacogdoche
42.347.14280.0045	01-19-2009	James B. Whiteside III	O'Brien Resources, LLC	3024-187	Texas	Nacogdoche
42.347.14280.0046	03-23-2009	Dave Patterson	O'Brien Resources, LLC	3038-192	Texas	Nacogdoche
42.347.14280.0047	03-23-2009	John Mark Patterson	O'Brien Resources, LLC	3038-193	Texas	Nacogdoche
42.347.14280.0048	03-23-2009	Davene Patterson Woods	O'Brien Resources, LLC	3038-194	Texas	Nacogdoche
42.347.14280.0049	11-03-2009	Danny P. Windham et ux	O'Brien Resources, LLC	3255-171	Texas	Nacogdoche
42.347.14280.0050	03-21-2008	Jean Caryl and Ronda Taylor	Devon Energy Production	1931-269	Texas	Nacogdoche
42.347.14280.0051	03-21-2008	Jean Caryl and Laurie Newman	Devon Energy Production	2838-134	Texas	Nacogdoche
42.347.14280.0052	03-08-2008	David W. Martin	Devon Energy Production	2838-150	Texas	Nacogdoche
42.347.14280.0053	09-03-2008	Bobby Dale Adkison	Devon Energy Production	2921-265	Texas	Nacogdoche
42.347.14280.0054	03-08-2008	Sarah Elizabeth Adkison	Devon Energy Production	2838-158	Texas	Nacogdoche
42.347.14280.0055	04-01-2008	Johnnie Ruth Goodson	Devon Energy Production	2838-117	Texas	Nacogdoche
42.347.14280.0056	04-01-2008	Danny Lee Crowley et ux	Devon Energy Production	2838-121	Texas	Nacogdoche
42.347.14280.0057	04-06-2008	Frieda Green	Devon Energy Production	2838-125	Texas	Nacogdoche
42.347.14280.0058	09-15-2008	J. Reade Dawson et ux	Devon Energy Production	2976-038	Texas	Nacogdoche
42.347.14280.0059	04-06-2008	Royce Edwin McIntyre	Devon Energy Production	2838-146	Texas	Nacogdoche
42.347.14280.0061	06-30-2009	Deirdre Kay Shaw Garner	Rowoil, Inc.	3088-060	Texas	Nacogdoche
42.415.15700.0008	09-13-2011	Kisk Oil Company	Figure 3 Working Interest,	321-722	Texas	Borden
42.415.15700.0018	08-29-2012	Gary Welsh, Deceased	O'Brien Resources, LLC	333-312	Texas	Borden
42.415.15700.0111	08-23-2012	Bettie Lou Hardin	O'Brien Resources, LLC	332-451	Texas	Borden
42.415.15700.0112	08-17-2012	John Glenn Pogue	O'Brien Resources, LLC	332-452	Texas	Borden
42.415.15700.0113	08-15-2012	Tony and Madonna Phillips	O'Brien Resources, LLC	332-453	Texas	Borden
42.415.15700.0114	08-17-2012		O'Brien Resources, LLC	332-438	Texas	Borden
42.415.15700.0115		Jane Curtis	O'Brien Resources, LLC	332-439	Texas	Borden
42.415.15700.0116		R. D. Lewis and Carol Sue Lewis, husb. and wife	O'Brien Resources, LLC	332-442	Texas	Borden
42.415.15700.0117	08-29-2012	Cary Welsh	O'Brien Resources, LLC	332-443	Texas	Borden
42.415.15700.0118	08-29-2012	Mike Welsh	O'Brien Resources, LLC	332-444	Texas	Borden
42.415.15700.0119	08-29-2012	Tommy Welsh	O'Brien Resources, LLC	332-455	Texas	Borden
42.415.15700.0120	08-23-2012	Wanda Joy Adams	O'Brien Resources, LLC	332-456	Texas	Borden
42.415.15700.0121	08-29-2012	Theresa Avera	O'Brien Resources, LLC	332-445	Texas	Borden
42.415.15700.0122		Judy Collier	O'Brien Resources, LLC	332-446	Texas	Borden
42.415.15700.0123		Michael Collier	O'Brien Resources, LLC	332-447	Texas	Borden
42.415.15700.0124		Dan W. Shipman	O'Brien Resources, LLC	332-448	Texas	Borden
42.415.15700.0125		Billye Fern Long	O'Brien Resources, LLC	332-449	Texas	Borden
42.415.15700.0126		Mozelle Sledge	O'Brien Resources, LLC	332-450	Texas	Borden
42.415.15700.0127		Keith Collier	O'Brien Resources, LLC	332-430	Texas	Borden
42.415.15700.0127		Jane Wilson	O'Brien Resources, LLC	332-304	Texas	Borden
42.415.15700.0128		Jo Ann Collett	O'Brien Resources, LLC	332-440	Texas	Borden
-			·			
42.415.15700.0130		Bobbye Lou Roberts	O'Brien Resources, LLC	332-454	Texas	Borden
42.415.15700.0131		Mary Emma Shipman	O'Brien Resources, LLC	332-441	Texas	Borden
42.415.15700.0132	08-29-2012	Tanya Tarpley	O'Brien Resources, LLC	332-458	Texas	Borden

LEASE#	DATE	LESSOR	LESSEE	BOOK-PAGE	STATE	COUNTY
42.415.15700.0133	08-29-2012	Barry Welsh	O'Brien Resources, LLC	333-311	Texas	Borden
42.415.15700.0134	12-26-2011	Don A. Jones et ux	DYM Energy, Inc.	325-306	Texas	Borden
42.415.15700.0134.01	12-26-2011	World News Radio	O'Brien Resources, LLC	325-306	Texas	Borden
42.415.15700.0134.02	12-26-2011	Christian Care Centers, Inc.	O'Brien Resources, LLC	325-306	Texas	Borden
42.415.15700.0134.03	12-26-2011	Abilene Christian University	O'Brien Resources, LLC	325-306	Texas	Borden
42.415.15700.0134.04	12-26-2011	World News Radio	O'Brien Resources, LLC	325-306	Texas	Borden
42.415.15700.0134.05	12-26-2011	Abilene Christian University	O'Brien Resources, LLC	325-306	Texas	Borden
42.415.15700.0134.06	12-26-2011	Christian Care Centers, Inc.	O'Brien Resources, LLC	325-306	Texas	Borden
42.415.15700.0137	01-31-2014	Joyce Herridge as Receiver for Frieda Mills and Patricia Downs	O'Brien Resources, LLC	342-358	Texas	Borden
42.415.15700.0138	05-20-2011	Don A. Jones and Nelva R. Jones	Winne Land & Minerals, Inc.	320-659	Texas	Borden
42.415.15700.0138.01	12-26-2011	World News Radio	O'Brien Resources, LLC	320-659	Texas	Borden
42.415.15700.0138.02	12-26-2011	Christian Care Centers, Inc.	O'Brien Resources, LLC	320-659	Texas	Borden
42.415.15700.0138.03	12-26-2011	Abilene Christian University	O'Brien Resources, LLC	320-659	Texas	Borden
42.415.15700.0160	06-30-2014	Don Wood	O'Brien Resources, LLC	346-364	Texas	Borden
42.415.15700.0161	06-30-2014	Danny Wood	O'Brien Resources, LLC	346-366	Texas	Borden
42.415.15700.0162	06-30-2014	Richard Wood	O'Brien Resources, LLC	346-362	Texas	Borden

LEASE#	DATE	LESSOR	LESSEE	BOOK-PAGE	STATE COUNTY
03.027.24400.0001	11-22-2017	Taylor F. Moore, et ux	O'Brien Resources, LLC	2017-004548	Arkansa Columbia
03.027.24400.0002	11-22-2017	Loy B. and Peggy H. Moore Family Interests LLC	O'Brien Resources, LLC	2017-004549	Arkansa Columbia
03.027.24400.0003	11-22-2017	Haynesville Merchantile Company	O'Brien Resources, LLC	2017-004550	Arkansa Columbia
03.027.24400.0004	11-22-2017	W. Harlan Beene, III	O'Brien Resources, LLC	2018-000144	Arkansa Columbia
03.027.24400.0006	07-12-2017	Michael Tarpley Garvin, et ux	O'Brien Resources, LLC	2017-002846	Arkansa Columbia
03.027.24400.0007	03-20-2017	Billy Gene Gray	O'Brien Resources, LLC	2017-004188	Arkansa Columbia
03.027.24400.0008	09-23-2016	Camp Properties, LLC	O'Brien Resources, LLC	2017-000641	Arkansa Columbia
03.027.24400.0009	09-23-2016	Randel H. Kirkland, III	O'Brien Resources, LLC	2017-000643	Arkansa Columbia
03.027.24400.0010	01-23-2018	Susan Darcy Kirkland	O'Brien Resources, LLC	2018-000214	Arkansa Columbia
03.027.24400.0011	11-01-2017	Billie Sunshine Waller Allen	O'Brien Resources, LLC	2017-004541	Arkansa Columbia
03.027.24400.0012	01-23-2018	Candice Stewart Robinson, et vir	O'Brien Resources, LLC	2018-000214	Arkansa Columbia
03.027.24400.0013	03-20-2017	Russell Todd Taylor, et ux	O'Brien Resources, LLC	2017-002036	Arkansa Columbia
03.027.24400.0017	11-22-2017	Randle T. Moore III, et ux	O'Brien Resources, LLC	2017-004551	Arkansa Columbia
03.027.24400.0018	01-23-2018	Pamela Stewart Owens	O'Brien Resources, LLC	2018-000214	Arkansa Columbia
03.027.24400.0019	11-22-2017	Ruby Ann McCraw Stokes, et vir	O'Brien Resources, LLC	2017-004552	Arkansa Columbia
03.027.24400.0020	09-23-2016	Richard H. Camp, et ux	O'Brien Resources, LLC	710-994	Arkansa Columbia
03.027.24400.0022	09-23-2016	Edwin Charles Sincox, Sr. Test. Trust and Lebois Wadlington Sincox	O'Brien Resources, LLC	710-986	Arkansa Columbia
03.027.24400.0023	03-20-2017	Flave W. Peters, et ux	O'Brien Resources, LLC	2017-001671	Arkansa Columbia
03.027.24400.0024	09-23-2016	Ryan Owens	O'Brien Resources, LLC	2017-000330	Arkansa Columbia
03.027.24400.0025	04-14-2017	Michael Bird Camp, et ux	O'Brien Resources, LLC	2017-001753	Arkansa Columbia
03.027.24400.0026	06-08-2017	Lisa M. Busch	O'Brien Resources, LLC	2017-002434	Arkansa Columbia
03.027.24400.0027	01-23-2018	Byron Durant Sincox	O'Brien Resources, LLC	2018-000214	Arkansa Columbia
03.027.24400.0028	09-23-2016	Pamela Paige Robinson Vlcek, et ux	O'Brien Resources, LLC	710-987	Arkansa Columbia
03.027.24400.0029	10-06-2016	Brandi Owens Eckols, et vir	O'Brien Resources, LLC	703-989	Arkansa Columbia
03.027.24400.0030	04-01-2017	Lewark, LLC	O'Brien Resources, LLC	2017-001670	Arkansa Columbia
03.027.24400.0031	04-01-2017	Nancy Camp Evans Irrevocable Family Trust	O'Brien Resources, LLC	2017-001669	Arkansa Columbia
03.027.24400.0032	04-01-2017	Lewis Louisiana Property, LLC	O'Brien Resources, LLC	2017-001668	Arkansa Columbia
03.027.24400.0033	05-24-2017	Amy Evans McKinney, et ux	O'Brien Resources, LLC	2017-003244	Arkansa Columbia
03.027.24400.0034	04-13-2017	Angela Evans Mills, et vir	O'Brien Resources, LLC	2017-001659	Arkansa Columbia
03.027.24400.0036	11-01-2017	Billie Sunshine Waller Allen	O'Brien Resources, LLC	2017-004542	Arkansa Columbia
03.027.24400.0037	04-14-2017	BenMar Trust	O'Brien Resources, LLC	2017-001755	Arkansa Columbia
03.027.24400.0073	01-23-2018	Angela Woodward Pryor	O'Brien Resources, LLC	2018-000214	Arkansa Columbia
03.027.24400.0082	11-21-2017	Weyerhaeuser Company ATT: Pamela J. Reed	O'Brien Resources, LLC	2017-004539	Arkansa Columbia
03.027.24400.0088	01-23-2018	Linda Rushton Selman Family Trust	O'Brien Resources, LLC	2018-000214	Arkansa Columbia
03.027.24400.0091	03-20-2017	Dana W. Davis, et vir	O'Brien Resources, LLC	2017-001335	Arkansa Columbia
03.027.24400.0092	09-23-2016	C. H. G. Minerals, Inc.	O'Brien Resources, LLC	2017-000331	Arkansa Columbia
03.027.24400.0093	06-26-2017	Dorothy T. Woodward	O'Brien Resources, LLC		Arkansa Columbia
03.027.24400.0094	06-08-2017	Amanda Dillard Shufeldt, et vir	O'Brien Resources, LLC	2017-004543	Arkansa Columbia
03.027.24400.0103	04-01-2017	Fred A. Hutcheson, Jr., et ux	O'Brien Resources, LLC	2017-003248	Arkansa Columbia
03.027.24400.0104	04-01-2017	Gordon S. Foster, et ux	O'Brien Resources, LLC	2017-001334	Arkansa Columbia

LEASE#	DATE	LESSOR	LESSEE	BOOK-PAGE	STATE	COUNTY
03.027.24400.0105	09-23-2016	Jean Lee Derby aka Jean Rogers Lee, Ind & as Trustee of the J W Lee Rev Trust dated 01/17/00.	O'Brien Resources, LLC	710-993	Arkansa	Columbia
03.027.24400.0108	04-01-2017	The Yvonne S. Allen Rev Trust	O'Brien Resources, LLC	2017-001333	Arkansa	Columbia
03.027.24400.0109	04-01-2017	Sam T. Allen III Rev Trust	O'Brien Resources, LLC	2017-001332	Arkansa	Columbia
03.027.24400.0110	04-14-2017	James T. Taylor, et ux	O'Brien Resources, LLC	2017-001658	Arkansa	Columbia
03.027.24400.0113	05-24-2017	Susan Rogers West	O'Brien Resources, LLC	2017-002124	Arkansa	Columbia
03.027.24400.0114	04-01-2017	Claude H. Wilson, et ux	O'Brien Resources, LLC	2017-001661	Arkansa	Columbia
03.027.24400.0115	09-23-2016	John William Peters, et ux Joanna R. Peters	O'Brien Resources, LLC	710-990	Arkansa	Columbia
03.027.24400.0116		Parrie Jane Carroll Family Trust	O'Brien Resources, LLC	2018-000145	Arkansa	Columbia
03.027.24400.0118	04-01-2017	Dan Joslin, et ux	O'Brien Resources, LLC	2017-001756	Arkansa	Columbia
03.027.24400.0137	01-23-2018	Richard Gardner, et ux	O'Brien Resources, LLC	2018-000214	Arkansa	Columbia
03.027.24400.0150	01-23-2018	Larry N. McNatt, et ux	O'Brien Resources, LLC	2018-000214	Arkansa	Columbia
03.027.24400.0152	06-09-2017	J. S. Beebe, Jr. Oil Properties, LLC	O'Brien Resources, LLC	2017-002290	Arkansa	Columbia
03.027.24400.0154		David Dudney	O'Brien Resources, LLC	2017-000332	Arkansa	Columbia
03.027.24400.0154a	09-26-2017	Martha Gray Taylor McCalman Heirs	O'Brien Resources, LLC	2018-000214	Arkansa	Columbia
03.027.24400.0154b		Vora J. Tippy	O'Brien Resources, LLC	2018-000214		Columbia
03.027.24400.0154c		Todd Bearden Dillard	O'Brien Resources, LLC	2018-000214		Columbia
03.027.24400.0154f		Jimmy E. Speer, et ux	O'Brien Resources, LLC	2018-000214		Columbia
03.027.24400.0154g		Rusty Brantly Gilbert, et ux	O'Brien Resources, LLC	2018-000214		Columbia
03.027.24400.0154i		Robert G. Ryan	O'Brien Resources, LLC	2018-000214		Columbia
03.027.24400.0154k		Vivian Carrington	O'Brien Resources, LLC	2018-000214		Columbia
03.027.24400.01541		Angela Carrington	O'Brien Resources, LLC	2018-000214		Columbia
03.027.24400.0154m		Virginia Pierce Jackson	O'Brien Resources, LLC	2018-000214		Columbia
03.027.24400.0154n		Brandy Michele Pierce Doty	O'Brien Resources, LLC	2018-000214		Columbia
03.027.24400.01540		Tommy L Silvertooth, Jr., et ux	O'Brien Resources, LLC	2018-000214		Columbia
03.027.24400.0154p		Tina Silvertooth Newsom, et vir	O'Brien Resources, LLC	2018-000214		Columbia
03.027.24400.0154p		Charles Marion Camp, et ux	O'Brien Resources, LLC	2017-001665		Columbia
03.027.24400.0172		The Virginia Pharr Williams Family Trust dated September 26, 2014	, -	2017-001865		Columbia
03.027.24400.0173	05-25-2017	Bonnie J. Pharr	O'Brien Resources, LLC	2017-002132	Arkansa	Columbia
03.027.24400.0174		Sherry L. Pharr	O'Brien Resources, LLC	2017-002436	Arkansa	Columbia
03.027.24400.0175a	03-02-2017	Kristine Pharr	O'Brien Resources, LLC	2017-001331	Arkansa	Columbia
03.027.24400.0176	05-24-2017	Julie Pharr Sullivan, et vir	O'Brien Resources, LLC	2017-002435	Arkansa	Columbia
03.027.24400.0177		Jennifer Pharr	O'Brien Resources, LLC	2017-001675		Columbia
03.027.24400.0179		Cynthia Allen Outlaw, et vir	O'Brien Resources, LLC	2017-002034		Columbia
03.027.24400.0180		Steven D. Allen, et ux	O'Brien Resources, LLC	2018-000146		Columbia
03.027.24400.0181		John Lee Smith	O'Brien Resources, LLC	2017-000329		Columbia
03.027.24400.0182		Paul Hayes Jameson, et ux	Atlanta Oil Company	704-113		Columbia
03.027.24400.0184		Patrick D. Cupp, et ux	O'Brien Resources, LLC	2017-002757		Columbia
03.027.24400.0190		Gray Hayes McCraw, Jr.	O'Brien Resources, LLC	2017-002757		Columbia
03.027.24400.0190		Haynesville Merchantile Company	O'Brien Resources, LLC	2017-001664		Columbia
03.027.24400.0191		Ruby Ann McCraw Stokes, et vir	O'Brien Resources, LLC	2017-004553		Columbia
03.027.24400.0193		Loy B. and Peggy H. Moore Family Interests LLC	O'Brien Resources, LLC	2017-004555		Columbia
03.027.24400.0194	11-22-2017	Randle T. Moore III, et ux	O'Brien Resources, LLC	2017-004556	Arkansa	Columbia
03.027.24400.0195		Taylor F. Moore, et ux	O'Brien Resources, LLC	2017-004547		Columbia
03.027.24400.0196		W. Harlan Beene, III	O'Brien Resources, LLC	2018-000148		Columbia
03.027.24400.0197		Camp Properties, LLC	O'Brien Resources, LLC	2017-002123		Columbia
03.027.24400.0198		Michael Bird Camp, et ux	O'Brien Resources, LLC	2017-002123		Columbia
	- 1-7-2011	Site Samp, or un		2017 001704	nai iod	- >.41111010

					-
LEASE#	DATE	LESSOR	LESSEE	BOOK-PAGE	STATE COUNTY
03.027.24400.0199	04-14-2017	Richard H. Camp, et ux	O'Brien Resources, LLC	2017-001657	Arkansa Columbia
03.027.24400.0200	04-14-2017	Nancy Camp Evans Irrevocable Family Trust	O'Brien Resources, LLC	2017-001663	Arkansa Columbia
03.027.24400.0201	04-14-2017	Angela Evans Mills, et vir	O'Brien Resources, LLC	2017-001660	Arkansa Columbia
03.027.24400.0202	09-23-2016	Barbara S. Taylor	O'Brien Resources, LLC	710-996	Arkansa Columbia
03.027.24400.0203	04-14-2017	Russell Todd Taylor, et ux	O'Brien Resources, LLC	2017-002035	Arkansa Columbia
03.027.24400.0204	04-01-2017	Lewark, LLC	O'Brien Resources, LLC	2017-001666	Arkansa Columbia
03.027.24400.0206	09-23-2016	William Patrick Sincox, et ux	O'Brien Resources, LLC	710-985	Arkansa Columbia
03.027.24400.0207	04-01-2017	Lewis Louisiana Property, LLC	O'Brien Resources, LLC	2017-001667	Arkansa Columbia
03.027.24400.0208	05-19-2017	Randel H. Kirkland, III	O'Brien Resources, LLC	2017-002758	Arkansa Columbia
03.027.24400.0209	01-23-2018	Susan Darcy Kirkland	O'Brien Resources, LLC	2018-000214	Arkansa Columbia
03.027.24400.0210	01-23-2018	Candice Stewart Robinson, et vir	O'Brien Resources, LLC	2018-000214	Arkansa Columbia
03.027.24400.0211	05-24-2017	Pamela Stewart Owens	O'Brien Resources, LLC	2017-003247	Arkansa Columbia
03.027.24400.0214	06-08-2017	Lisa M. Busch	O'Brien Resources, LLC	2017-003245	Arkansa Columbia
03.027.24400.0215		Edwin Charles Sincox, Jr., et ux	O'Brien Resources, LLC	2017-001673	Arkansa Columbia
03.027.24400.0216		Amy Evans McKinney, et ux	O'Brien Resources, LLC		Arkansa Columbia
03.027.24400.0217		Patricia Pearl Waller Garvin	O'Brien Resources, LLC	709-609	Arkansa Columbia
03.027.24400.0217		Michael Tarpley Garvin, et ux	O'Brien Resources, LLC	2017-002845	Arkansa Columbia
03.027.24400.0219		Barbara Agnes Waller Gray, et vir	O'Brien Resources, LLC	2018-000214	Arkansa Columbia
03.027.24400.0220		Saralynn Dudney Beasley	O'Brien Resources, LLC	2017-001662	Arkansa Columbia
03.027.24400.0222		Jo Brenda Lee Wilbern, et vir	O'Brien Resources, LLC	2017-001002	Arkansa Columbia
03.027.24400.0224		Mark A. Chaffin, et ux	O'Brien Resources, LLC	2017-001202	
		·	,		Arkansa Columbia
03.027.24400.0229		Alexander Properties, LLP	O'Brien Resources, LLC	710-984	Arkansa Columbia
03.027.24400.0232		Reeves Land and Timber Company	O'Brien Resources, LLC	2017-001672	Arkansa Columbia
03.027.24400.0234		Barbara Sue Gean Ricks, et vir	O'Brien Resources, LLC	2018-000214	Arkansa Columbia
03.027.24400.0235		William Todd Evans, et ux	O'Brien Resources, LLC	2018-000214	Arkansa Columbia
03.027.24400.0245		James C. Bailey, et ux	O'Brien Resources, LLC	2017-000912	Arkansa Columbia
03.027.24400.0249		Mary Lynn Baucum Whitley	O'Brien Resources, LLC	2017-000916	Arkansa Columbia
03.027.24400.0250		Dora Kennedy	O'Brien Resources, LLC	2017-000915	Arkansa Columbia
03.027.24400.0251		Elizabeth Moss, et vir	O'Brien Resources, LLC	2017-000942	Arkansa Columbia
03.027.24400.0252		Haley D. Humphrey	O'Brien Resources, LLC	2017-000943	Arkansa Columbia
03.027.24400.0253	08-03-2016	Samuel and Deborah Millican Joint Revocable Tr dated 12-15-2014, et al	O'Brien Resources, LLC	2017-000944	Arkansa Columbia
03.027.24400.0255	08-03-2016	Charles Ray Owen, et ux	O'Brien Resources, LLC	2017-000991	Arkansa Columbia
03.027.24400.0256	09-13-2016	Thomas Samuel Jameson, et ux	O'Brien Resources, LLC	710-991	Arkansa Columbia
03.027.24400.0257	09-13-2016	Thomas H. French, et ux	O'Brien Resources, LLC	710-992	Arkansa Columbia
03.027.24400.0258	09-08-2016	Carey Michael Hughes	O'Brien Resources, LLC	2017-000989	Arkansa Columbia
03.027.24400.0259	08-03-2016	Velma I. Baucum	O'Brien Resources, LLC	2017-000945	Arkansa Columbia
03.027.24400.0261	08-30-2016	Nancy Susan McAnulty	O'Brien Resources, LLC	2017-000988	Arkansa Columbia
03.027.24400.0262	08-09-2016	Nancy Baucum Kerley, et ux	O'Brien Resources, LLC	2017-000959	Arkansa Columbia
03.027.24400.0263	08-03-2016	Don Camp Baucum, et ux	O'Brien Resources, LLC	2017-000951	Arkansa Columbia
03.027.24400.0264	08-03-2016	Laurel Ann Leveritt	O'Brien Resources, LLC	2017-000954	Arkansa Columbia
03.027.24400.0265	08-09-2016	Kathy B. Cobb, et vir	O'Brien Resources, LLC	2017-000994	Arkansa Columbia
03.027.24400.0266	08-09-2016	Karen Ann McKnight	O'Brien Resources, LLC	2017-000958	Arkansa Columbia
03.027.24400.0268	08-03-2016	Jennifer Leveritt	O'Brien Resources, LLC	2017-000955	Arkansa Columbia
03.027.24400.0269	08-09-2016	Douglas D. Baucum, et ux	O'Brien Resources, LLC	2017-000960	Arkansa Columbia
03.027.24400.0270	08-09-2016	Wanda Lee Rothwell	O'Brien Resources, LLC	2017-000952	Arkansa Columbia
03.027.24400.0271	08-09-2016	Sara Baucum Loudermilk, et vir	O'Brien Resources, LLC	2017-000984	Arkansa Columbia
03.027.24400.0272		Richard H. Sale, et ux	O'Brien Resources, LLC	2017-000957	Arkansa Columbia
		•	, -		*

LEASE#	DATE	LESSOR	LESSEE	BOOK-PAGE	STATE COUNTY
03.027.24400.0273	08-09-2016	Randy Merritt, et ux	O'Brien Resources, LLC	2017-000956	Arkansa Columbia
03.027.24400.0274	09-08-2016	Phyllis Anne Temple, et vir	O'Brien Resources, LLC	2017-000987	Arkansa Columbia
03.027.24400.0275	08-03-2016	Penny Hughes Jones, et ux	O'Brien Resources, LLC	2017-000992	Arkansa Columbia
03.027.24400.0276	09-08-2016	Dana Hughes Evans, et vir	O'Brien Resources, LLC	2017-000993	Arkansa Columbia
03.027.24400.0277	08-30-2016	Gene E. McKnight Jr., et ux	O'Brien Resources, LLC	2017-000990	Arkansa Columbia
03.027.24400.0278	08-03-2016	Bennie Dee Hughes, et ux	O'Brien Resources, LLC	2017-000953	Arkansa Columbia
03.027.24400.0279	08-03-2016	LaBois Wadlington Sincox, individually and as Trustee of the Edwin Charles Sincox, Sr. Testemenatry Tr	O'Brien Resources, LLC	2017-000985	Arkansa Columbia
03.027.24400.0280	08-25-2016	John W. Stevens, et ux	O'Brien Resources, LLC	710-995	Arkansa Columbia
03.027.24400.0281	08-25-2016	Steve Carrington, et ux	O'Brien Resources, LLC	710-910	Arkansa Columbia
03.027.24400.0282	09-23-2016	Gray Hayes McCraw, Jr.	O'Brien Resources, LLC	710-0983	Arkansa Columbia
03.027.24400.0283	09-23-2016	Edwin Charles Sincox, Jr., et ux	O'Brien Resources, LLC	710-988	Arkansa Columbia
03.027.24400.0286	10-06-2016	Glenva D. Hines	O'Brien Resources, LLC	2017-000963	Arkansa Columbia
03.027.24400.0287	12-06-2016	Paul David Miller, Jr.	O'Brien Resources, LLC	2017-000937	Arkansa Columbia
03.027.24400.0288	09-28-2016	Frank Taylor Speer, et ux	O'Brien Resources, LLC	2017-000979	Arkansa Columbia
03.027.24400.0289	09-28-2016	Gary Gardner, et ux	O'Brien Resources, LLC	2017-000961	Arkansa Columbia
03.027.24400.0290		Gary Lee Speer Sr., et ux	O'Brien Resources, LLC	2017-000981	Arkansa Columbia
03.027.24400.0291		Peggy Jean Wilson Pease	O'Brien Resources, LLC	2017-000924	Arkansa Columbia
03.027.24400.0292	09-28-2016		O'Brien Resources, LLC	2017-000931	Arkansa Columbia
03.027.24400.0293		Carole Denise Williams	O'Brien Resources, LLC	2017-000964	Arkansa Columbia
03.027.24400.0294		Sherry L. Pharr	O'Brien Resources, LLC	2017-000923	Arkansa Columbia
03.027.24400.0295		Wade P. Pitcock	O'Brien Resources, LLC	2017-000971	Arkansa Columbia
03.027.24400.0296		Steve Hines	O'Brien Resources, LLC	2017-000920	Arkansa Columbia
03.027.24400.0297		Connie F. Speer Remedies, et vir	O'Brien Resources, LLC	2017-000969	Arkansa Columbia
03.027.24400.0298		Jerry Don Hines, et ux	O'Brien Resources, LLC	2017-000966	Arkansa Columbia
03.027.24400.0299		Stacy Malone Scolley, et ux	O'Brien Resources, LLC	2017-000926	Arkansa Columbia
03.027.24400.0300		Linda Joyce Firment, et ux	O'Brien Resources, LLC	2017-000920	Arkansa Columbia
03.027.24400.0300		Jackie Lynn McCann, et vir	O'Brien Resources, LLC	2017-000929	Arkansa Columbia
03.027.24400.0301		Brenda Tuberville	O'Brien Resources, LLC	2017-000921	Arkansa Columbia
03.027.24400.0302		Carl Hines, et ux	O'Brien Resources, LLC	2017-000962	Arkansa Columbia
03.027.24400.0303		James Edwin Gurnick, et ux	O'Brien Resources, LLC	2017-000973	Arkansa Columbia Arkansa Columbia
			<u> </u>		
03.027.24400.0305		Nell M. Hines Smith	O'Brien Resources, LLC	2017-000970	Arkansa Columbia
03.027.24400.0306		Ohorlos Marion Comp. et uv	O'Brien Resources, LLC	2017-000974	Arkansa Columbia
03.027.24400.0307		Charles Marion Camp, et ux	O'Brien Resources, LLC	2017-000983	Arkansa Columbia
03.027.24400.0308		Bonnie Speer Williams, et ux	O'Brien Resources, LLC	2017-000939	Arkansa Columbia
03.027.24400.0309		Robert L. Taylor	O'Brien Resources, LLC	2017-000968	Arkansa Columbia
03.027.24400.0310		Phillip T. McKenzie	O'Brien Resources, LLC	2017-000922	Arkansa Columbia
03.027.24400.0311		Patsy Speer	O'Brien Resources, LLC	2017-000925	Arkansa Columbia
03.027.24400.0312		Oneta F. Camp	O'Brien Resources, LLC	2017-000977	Arkansa Columbia
03.027.24400.0313		Michelle Camp Nicholas	O'Brien Resources, LLC	2017-000972	Arkansa Columbia
03.027.24400.0314		Laurie Wynn Jacob, et ux	O'Brien Resources, LLC	2017-000899	Arkansa Columbia
03.027.24400.0315	09-28-2016	Linda M. Wynn	O'Brien Resources, LLC	2017-000978	Arkansa Columbia
03.027.24400.0316	09-28-2016	Linda Wynn Blakely	O'Brien Resources, LLC	2017-000976	Arkansa Columbia
03.027.24400.0317	10-06-2016	Linda Sue Harper	O'Brien Resources, LLC	2017-000980	Arkansa Columbia
03.027.24400.0318	09-28-2016	James H. Watson, et ux	O'Brien Resources, LLC	2017-000975	Arkansa Columbia
03.027.24400.0319	08-09-2016	Bobbie Jo Sale Stinson	O'Brien Resources, LLC	2017-000936	Arkansa Columbia
03.027.24400.0320	09-28-2016	Frank T. Watson, et ux	O'Brien Resources, LLC	2017-000935	Arkansa Columbia
03.027.24400.0321	09-28-2016	Ellis Ann McKenzie Lay, et vir	O'Brien Resources, LLC	2017-000934	Arkansa Columbia

LEASE#	DATE	LESSOR	LESSEE	BOOK-PAGE	STATE COUNTY
03.027.24400.0322	09-28-2016	Michael D. Wynn	O'Brien Resources, LLC	2017-000933	Arkansa Columbia
03.027.24400.0323	09-28-2016	Tammy Lynette Camp	O'Brien Resources, LLC	2017-000938	Arkansa Columbia
03.027.24400.0324	09-28-2016	Thomas Watson, et ux	O'Brien Resources, LLC	2017-000982	Arkansa Columbia
03.027.24400.0325	09-28-2016	Sharon Kay Doss McKinney, et vir	O'Brien Resources, LLC	2017-000930	Arkansa Columbia
03.027.24400.0326	10-06-2016	Sharon Speer Bounds, et vir	O'Brien Resources, LLC	2017-000930	Arkansa Columbia
03.027.24400.0327	09-26-2016	Scott Edward Speer, et ux	O'Brien Resources, LLC	2017-000930	Arkansa Columbia
03.027.24400.0328	09-23-2016	Kellie Speer Schulte, et vir	O'Brien Resources, LLC	2017-001201	Arkansa Columbia
03.027.24400.0329	10-06-2016	Carolyn Hines Woodard, et vir	O'Brien Resources, LLC	2017-000904	Arkansa Columbia
03.027.24400.0330	11-04-2016	Casey Tilton, et ux	O'Brien Resources, LLC	2017-000901	Arkansa Columbia
03.027.24400.0331	09-28-2016	Kellie Speer Schulte, et vir	O'Brien Resources, LLC	2017-000902	Arkansa Columbia
03.027.24400.0332	10-06-2016	Kevin Hines, et ux	O'Brien Resources, LLC	2017-000903	Arkansa Columbia
03.027.24400.0333	03-08-2017	Margaret Ann Hughes Mohr, et vir	O'Brien Resources, LLC	2017-000911	Arkansa Columbia
03.027.24400.0334	03-02-2017	Shirley Hughes Thomas, et vir	O'Brien Resources, LLC	2017-000910	Arkansa Columbia
03.027.24400.0335	01-28-2017	Calvin Glen Nelson, et ux	O'Brien Resources, LLC	2017-000940	Arkansa Columbia
03.027.24400.0336	11-04-2016	The Austin Cheatham Irrevocable Trust Dated 01/24/2014	Marlin Exploration, LLC	2017-000900	Arkansa Columbia
03.027.24400.0337	09-28-2016	Barbara Duane Doss Lewis	O'Brien Resources, LLC	2017-000927	Arkansa Columbia
03.027.24400.0338	08-03-2016	Diddlebugs, LLC, et al	O'Brien Resources, LLC	2017-001412	Arkansa Columbia
03.027.24400.0339	03-08-2017	Sue Hughes Chandler	O'Brien Resources, LLC	2017-001414	Arkansa Columbia
03.027.24400.0340	01-25-2017	Harper W. Baucum, II, et ux	O'Brien Resources, LLC	2017-001420	Arkansa Columbia
03.027.24400.0341	01-25-2017	Bennie Joe Baucum, Jr.	O'Brien Resources, LLC	2017-001419	Arkansa Columbia
03.027.24400.0342	04-05-2017	Jennifer Pharr	O'Brien Resources, LLC	2017-001677	Arkansa Columbia
03.027.24400.0343	03-28-2017	Lewark, LLC	O'Brien Resources, LLC	2017-001418	Arkansa Columbia
03.027.24400.0344	04-10-2017	Tri-State Land & Minerals, L.L.C.	O'Brien Resources, LLC	2017-001417	Arkansa Columbia
03.027.24400.0345	03-17-2017	Charles Doak Victor Boreing	O'Brien Resources, LLC	2017-001413	Arkansa Columbia
03.027.24400.0346	04-06-2017	Joe D. Woodward Revocable Trust dated March 16, 2005	O'Brien Resources, LLC	2017-001416	Arkansa Columbia
03.027.24400.0347	04-05-2017	Kristine Pharr	O'Brien Resources, LLC	2017-001676	Arkansa Columbia
03.027.24400.0348	09-28-2016	Sheila Wynn Still, et vir	O'Brien Resources, LLC	2017-001415	Arkansa Columbia
03.027.24400.0349	04-05-2017	The Virginia Pharr Williams Family Trust dated September 26, 2014	O'Brien Resources, LLC	2017-001674	Arkansa Columbia
03.027.24400.0350	03-28-2017	Wanda Lee Rothwell	O'Brien Resources, LLC	2017-001452	Arkansa Columbia
03.027.24400.0351	09-09-2015	Lewark, LLC	Marlin Exploration, LLC	710-050	Arkansa Columbia
03.027.24400.0354	09-09-2015	Edwin Charles Sincox, Jr., et ux	Marlin Exploration, LLC	710-052	Arkansa Columbia
03.027.24400.0355	09-09-2015	Byron Durant SIncox	Marlin Exploration, LLC	710-051	Arkansa Columbia
03.027.24400.0364	03-20-2017	Tammy Lynette Camp	O'Brien Resources, LLC	2017-002033	Arkansa Columbia
03.027.24400.0365	05-11-2017	Ella Sue Lutjen	O'Brien Resources, LLC	2017-001844	Arkansa Columbia
03.027.24400.0366	05-11-2017	Benny Emmert, et ux	O'Brien Resources, LLC	2017-002305	Arkansa Columbia
03.027.24400.0367	05-25-2017	Bonnie J. Pharr	O'Brien Resources, LLC	2017-002125	Arkansa Columbia
03.027.24400.0368	04-05-2017	Douglas T. Pharr	O'Brien Resources, LLC	2017-002126	Arkansa Columbia
03.027.24400.0369	04-05-2017	The Three Oaks Revocable Living Trust dated August 13, 2011	O'Brien Resources, LLC	2017-002127	Arkansa Columbia
03.027.24400.0370	04-05-2017	Edward H. Pharr	O'Brien Resources, LLC	2017-002128	Arkansa Columbia
03.027.24400.0371	04-05-2017	Sherry L. Pharr	O'Brien Resources, LLC	2017-002129	Arkansa Columbia
03.027.24400.0372	05-01-2017	Betty Jane Baucum Burnham, et vir	O'Brien Resources, LLC	2017-001845	Arkansa Columbia
03.027.24400.0373	09-28-2016	Robert Floyd Doss, et ux	O'Brien Resources, LLC	2017-002291	Arkansa Columbia
03.027.24400.0374	06-01-2017	James L. Cheatham, et ux	O'Brien Resources, LLC	2017-002304	Arkansa Columbia
03.027.24400.0375	05-02-2017	Dinah Harmon Hughes, et ux	O'Brien Resources, LLC	2017-001847	Arkansa Columbia
03.027.24400.0376	05-23-2017	David Ralph Hines, et ux	O'Brien Resources, LLC	2017-002292	Arkansa Columbia
03.027.24400.0377	05-08-2017	Gloria Thomas	O'Brien Resources, LLC	2017-001853	Arkansa Columbia
-					

					· · · · · · · · · · · · · · · · · · ·
LEASE#	DATE	LESSOR	LESSEE	BOOK-PAGE	STATE COUNTY
03.027.24400.0378	05-02-2017	Julie Ann Harmon	O'Brien Resources, LLC	2017-001848	Arkansa Columbia
03.027.24400.0379	05-03-2017	Jane E. Taylor Cananza, et vir	O'Brien Resources, LLC	2017-001849	Arkansa Columbia
03.027.24400.0380	05-03-2017	Jeff Thomas	O'Brien Resources, LLC	2017-002306	Arkansa Columbia
03.027.24400.0381	05-01-2017	Lisa Cay Gilbert Farmer, et vir	O'Brien Resources, LLC	2017-001850	Arkansa Columbia
03.027.24400.0382	05-02-2017	Larry N. McNatt, et ux	O'Brien Resources, LLC	2017-001846	Arkansa Columbia
03.027.24400.0383	05-23-2017	Miriam Margaret Baucum Godwin	O'Brien Resources, LLC	2017-002288	Arkansa Columbia
03.027.24400.0384	05-03-2017	Robert Clayton Baucum, et ux	O'Brien Resources, LLC	2017-002307	Arkansa Columbia
03.027.24400.0385	05-03-2017	Mary Virginia Taylor	O'Brien Resources, LLC	2017-001852	Arkansa Columbia
03.027.24400.0386	05-08-2017	Cody Thomas, et ux	O'Brien Resources, LLC	2017-001851	Arkansa Columbia
03.027.24400.0387	06-01-2017	Loretta Kay Cheatham	O'Brien Resources, LLC	2017-002302	Arkansa Columbia
03.027.24400.0388	06-01-2017	Barbara C. Tate-Logan, et vir	O'Brien Resources, LLC	2017-002303	Arkansa Columbia
03.027.24400.0389	05-23-2017	Gwendolyn M. Baucum	O'Brien Resources, LLC	2017-002298	Arkansa Columbia
03.027.24400.0390	06-01-2017	Windell L. Cheatham, et ux	O'Brien Resources, LLC	2017-002299	Arkansa Columbia
03.027.24400.0391	06-01-2017	Larry L. Cheatham, et ux	O'Brien Resources, LLC	2017-002297	Arkansa Columbia
03.027.24400.0392	06-01-2017	Diane Cheatham Nielsen	O'Brien Resources, LLC	2017-002300	Arkansa Columbia
03.027.24400.0393	06-01-2017	William L. Cheatham, et ux	O'Brien Resources, LLC	2017-002301	Arkansa Columbia
03.027.24400.0394	06-01-2017	Brenda Cheatham Nelson, et ux	O'Brien Resources, LLC	2017-002296	Arkansa Columbia
03.027.24400.0395	06-01-2017	Chris Cheatham, et ux	O'Brien Resources, LLC	2017-002295	Arkansa Columbia
03.027.24400.0396	03-24-2017	Jonette Mitchell Richardson	O'Brien Resources, LLC	2017-002289	Arkansa Columbia
03.027.24400.0397	06-01-2017	Cheatham Properties, LLC	O'Brien Resources, LLC	2017-002309	Arkansa Columbia
03.027.24400.0398	06-01-2017	Kevin K. Cheatham, et ux	O'Brien Resources, LLC	2017-002308	Arkansa Columbia
03.027.24400.0400	04-05-2017	Julie Pharr Sullivan, et vir	O'Brien Resources, LLC	2017-002437	Arkansa Columbia
03.027.24400.0402	06-13-2017	Helen Susan Diane Bussey	O'Brien Resources, LLC	2017-002759	Arkansa Columbia
03.027.24400.0403	05-24-2017	Edward H. Pharr	O'Brien Resources, LLC	2017-002130	Arkansa Columbia
03.027.24400.0404	05-24-2017	Douglas T. Pharr	O'Brien Resources, LLC	2017-002131	Arkansa Columbia
03.027.24400.0406	04-22-2017	Kenneth Wayne Crawford, et ux	O'Brien Resources, LLC	2017-002294	Arkansa Columbia
03.027.24400.0407	10-06-2016	Robert Wayne Speer	O'Brien Resources, LLC	2017-000932	Arkansa Columbia
03.027.24400.0408	08-09-2016	Virginia White Baucum Walton	O'Brien Resources, LLC	2017-000941	Arkansa Columbia
03.027.24400.0409	08-03-2017	Barbara S. Taylor	O'Brien Resources, LLC	2017-003246	Arkansa Columbia
03.027.24400.0410	08-03-2016	Becky Munn	O'Brien Resources, LLC	2017-000946	Arkansa Columbia
03.027.24400.0411	08-03-2016	Debra J. Shelton, et vir	O'Brien Resources, LLC	2017-000950	Arkansa Columbia
03.027.24400.0412	08-03-2016	Gary L. Edwards, et ux	O'Brien Resources, LLC	2017-000948	Arkansa Columbia
03.027.24400.0413		Mark A. Taylor, et ux	O'Brien Resources, LLC	2017-000947	Arkansa Columbia
03.027.24400.0414		Steven R. Taylor, et ux	O'Brien Resources, LLC	2017-000949	Arkansa Columbia
03.027.24400.0415		Steven D. Allen, et ux	O'Brien Resources, LLC	2018-000147	Arkansa Columbia
03.027.24400.0416		Donald C. Wynn Sr., et ux	O'Brien Resources, LLC	2017-004540	Arkansa Columbia
03.027.24400.0417		Brock William Hines	O'Brien Resources, LLC	2017-004545	Arkansa Columbia
03.027.24400.0418	11-08-2017		O'Brien Resources, LLC	2017-004544	Arkansa Columbia
03.027.24400.0419		Dr. Timothy G. Boreing	O'Brien Resources, LLC	2017-004546	Arkansa Columbia
03.027.24400.0420		Shelda Hines Shackelford, et vir	O'Brien Resources, LLC	2018-000149	Arkansa Columbia
03.027.24400.0420		Gary L. Hines, et ux	O'Brien Resources, LLC	2018-000150	Arkansa Columbia
03.027.24400.0422		David Lawrence Rogers	O'Brien Resources, LLC	2018-000151	Arkansa Columbia
03.027.24400.0423		Richard Gardner, et ux	O'Brien Resources, LLC	2018-000131	Arkansa Columbia
AMHM-001		Ava Margaret H. Morgan	O'Brien Resources, LLC	2018-000214	Arkansa Columbia
AP.01	05-24-2017		O'Brien Resources, LLC	2010-000214	Arkansa Columbia
ASG-001			<u> </u>	2018-000214	Arkansa Columbia
		Andy Scott Gilbert, et ux	O'Brien Resources, LLC		
BEHJ-001		Billy E. Hines, Jr.	O'Brien Resources, LLC	2018-000214	Arkansa Columbia
CAO-001	01-23-2018	Cynthia Allen Outlaw, et vir	O'Brien Resources, LLC	2018-000214	Arkansa Columbia

LEASE#	DATE	LESSOR	LESSEE	BOOK-PAGE	STATE COUNTY
CB-001	01-23-2018	Cynthia Birdsong, et vir	O'Brien Resources, LLC	2018-000214	Arkansa Columbia
DLTK-001	01-23-2018	Dee Layne Thomas Kubicki, et vir	O'Brien Resources, LLC	2018-000214	Arkansa Columbia
JES-A-001	01-23-2018	Jimmy E. Speer, et ux	O'Brien Resources, LLC	2018-000214	Arkansa Columbia
JHD-001	01-23-2018	James Horace Doss	O'Brien Resources, LLC	2018-000214	Arkansa Columbia
JLS-A-001	01-23-2018	John Lee Smith	O'Brien Resources, LLC	2018-000214	Arkansa Columbia
JWAJ-001	01-23-2018	Jerry Wayne Allen, Jr., et ux	O'Brien Resources, LLC	2018-000214	Arkansa Columbia
MAWW-001	01-23-2018	Molly Ann Wilson Willis	O'Brien Resources, LLC	2018-000214	Arkansa Columbia
Open.03	09-26-2017	Billy E. Hines, Jr.	O'Brien Resources, LLC		Arkansa Columbia
Open.38	01-23-2018	Nancy Elizabeth Bussey	O'Brien Resources, LLC	2018-000214	Arkansa Columbia
RBG-001	01-23-2018	Rusty Brantly Gilbert, et ux	O'Brien Resources, LLC	2018-000214	Arkansa Columbia
RDS-A-001	01-23-2018	Ritchie D. Smith, et ux	O'Brien Resources, LLC	2018-000214	Arkansa Columbia
RR-001	01-23-2018	Robert G. Ryan	O'Brien Resources, LLC	2018-000214	Arkansa Columbia
SS-A-001	01-23-2018	Stephen Speer	O'Brien Resources, LLC	2018-000214	Arkansa Columbia
TLS-A-001	01-23-2018	Tommy L Silvertooth, Jr., et ux	O'Brien Resources, LLC	2018-000214	Arkansa Columbia
TS-001	01-23-2018	Timothy S. Speer	O'Brien Resources, LLC	2018-000214	Arkansa Columbia
TSN-001	01-23-2018	Tina Silvertooth Newsom, et vir	O'Brien Resources, LLC	2018-000214	Arkansa Columbia
VLH-001	01-23-2018	Vickie L. Hines	O'Brien Resources, LLC	2018-000214	Arkansa Columbia
WCS-001	01-23-2018	William C. Stacy, et ux	O'Brien Resources, LLC	2018-000214	Arkansa Columbia
15.165.18640.0007	07-05-2016	Joseph Menzak and Elizabeth Menzak Revocable Trust dated december 10, 2010	O'Brien Resources, LLC	168-784	Kansas Rush
15.165.18640.0008	09-01-2012	David A. Vondracek and Cindy Vondracek	KANEX, L.C.	162-356	Kansas Rush
15.165.18640.0011	07-28-2016	James A. Jecha and Carole Jo Jecha	O'Brien Resources, LLC	168-795	Kansas Rush
15.165.18660.0004	09-19-2016	Joseph Menzak and Elizabeth Menzak Revocable Trust, dated 12-10-2010	O'Brien Resources, LLC	169-103	Kansas Rush
15.101.18110.0008	03-28-2005	E.R. Redburn, Jr et ux	D. D. Morgen	110-163	Kansas Lane
15.101.18110.0012	03-03-2005	Gene W. Boomhower et ux	D. D. Morgen	110-157	Kansas Lane
15.101.18110.0014	03-03-2005	Gene W. Boomhower et ux	D. D. Morgen	110-158	Kansas Lane
15.101.18110.0023	04-12-2005	Alice L. Riley	D. D. Morgen	111-120	Kansas Lane
15.101.18140.0003	11-14-2005	Karl E. Jennison et al	D. D. Morgen	112-165	Kansas Lane
15.101.18140.0006	01-21-2011	Jennison Ranch	D. D. Morgen	134-043	Kansas Lane
15.101.18140.0016	01-24-2006	First Baptist Church Ellinwood	D. D. Morgen	114-045	Kansas Lane
15.101.18140.0019	01-31-2014	Margaret J Harper	Paramount Land, Inc.	148-036	Kansas Lane
15.101.18140.0035	02-11-2011	John A. Snider et al	Kanex L.C.	134-005	Kansas Lane
15.101.18140.0036	02-15-2011	Sharp Seed Farms L.L.C.	Kanex, L.C.	134-071	Kansas Lane
15.101.18140.0038	02-15-2011	Betty Sharp	Kanex, L.C.	134-069	Kansas Lane
15.101.18140.0039	02-15-2011	Brenda Jean Sharp	Kanex, L.C.	134-070	Kansas Lane
15.101.18140.0041	08-19-2013	Daniel Sharp and LaFonda Sharp	Paramount Land, Inc.	146-216	Kansas Lane
15.165.18620.0007	04-25-2017	Wayne W. Schope and Janice M. Schope	O'Brien Resources, LLC	170-510	Kansas Rush
15.165.18620.0008		Gloria A. Ochs and Vincent W. Ochs	O'Brien Resources, LLC	170-512	Kansas Rush
15.165.18700.0004		Wayne W. Schope et ux	O'Brien Resources, LLC	170-510	Kansas Rush
15.165.18700.0005		Gloria A. Ochs et vir	O'Brien Resources, LLC	170-512	Kansas Rush
15.165.18700.0008		Marsha K. Moeder Trust	O'Brien Resources, LLC	168-420	Kansas Rush
15.165.18700.0019		William L. Besperat	Paramount Land, Inc.	165-942	Kansas Rush
15.165.18700.0025		Joseph Menzak and Elizabeth Menzak Revocable Trust, dated 12-10-2010	O'Brien Resources, LLC	169-101	Kansas Rush
15.165.18700.0031	09-01-2012	Anthony C. Vondracek et al	KANEX, L.C.	162-333	Kansas Rush
15.165.18700.0033		Anthony C. Vondracek et al	Paramount Land, Inc.	166-086	Kansas Rush
15.165.18700.0036		James A. Jecha et ux	O'Brien Resources, LLC	168-795	Kansas Rush
15.165.18700.0042		Bernard Juno and Beverly Juno	O'Brien Resources, LLC	167-178	Kansas Rush
10.100.10700.0042	07-00-2015	Demard June and Devery June	O Dileit Nesoulces, LLC	107-170	Manaa Musii

LEASE#	DATE	LESSOR	LESSEE	BOOK-PAGE		COUNTY
15.063.18800.0003		Terry L. Swart et ux	Red Oak Energy, Inc.	169-1004	Kansas	
15.063.18800.0006		Prather Farms INC	Red Oak Energy, Inc.	169-796	Kansas	
15.063.18800.0019		Robert A. McElfatrick et ux	O'Brien Resources, LLC	172-072		_
15.063.18800.0023	05-04-2012	Consuela A. Townsend	Paramount Land, Inc.	179-775	Kansas	
15.063.18800.0024		Nazareth Convent and Academy Corporation	Paramount Land, Inc.	179-907		Gove
15.063.18800.0026	05-10-2012	Ervin A. Penner et ux	Paramount Land, Inc.	180-039	Kansas	Gove
15.063.18800.0027	04-23-2012	Texas MSJ Investments LLC	Paramount Land, Inc.	179-907	Kansas	Gove
15.063.18800.0028	05-02-2012	C. J. Steckel Trust #1	Paramount Land, Inc.	180-245	Kansas	Gove
15.063.18800.0029	04-23-2012	Gale H. Schlageck et ux	Paramount Land, Inc.	180-240	Kansas	Gove
15.063.18800.0030	05-02-2012	Irene M. Steckel Trust #1	Paramount Land, Inc.	180-242	Kansas	Gove
15.063.18800.0031	04-27-2012	Holt Royalty, LLC	Paramount Land, Inc.	179-905	Kansas	Gove
17.013.36200.0001	07-21-1989	ELBIN LAMAR GREER	Equity Oil Company	567-236	Louisian	BIENVILLE
17.013.36200.0002	10-05-1989	Premier Bank (Bank One) Trustee for Charles Bryan Trust	Equity Oil Company		Louisian	BIENVILLE
17.013.36200.0003	08-03-1989	LOTTIE JO GREER WARD	Equity Oil Company	567-228	Louisian	BIENVILLE
17.013.36200.0004	08-07-1989	CALEB JOSEPH GREER	Equity Oil Company	567-224	Louisian	BIENVILLE
17.013.36200.0005	03-01-2003	BILLY D. BRYAN	O'Brien Energy Company	0956-330	Louisian	BIENVILLE
17.013.36200.0006	08-03-1989	JOAN RUTH WILLIAMS KING	Equity Oil Company	567-216	Louisian	BIENVILLE
17.013.36200.0007	07-21-1989	GARY L. GREER ET AL	Equity Oil Company	570-010	Louisian	BIENVILLE
17.013.36200.0008	08-16-1989	BILLY D. BRYAN ET UX	Equity Oil Company	567-212	Louisian	BIENVILLE
17.013.36200.0009	03-04-2003	CELIA ANNE JONES	O'Brien Energy Company	0956-333	Louisian	BIENVILLE
17.013.36200.0010	03-06-2003	BETTYE JEANETTE JONE KRACK	O'Brien Energy Company	0956-342	Louisian	BIENVILLE
17.013.36200.0011	03-05-2003	JANIS LOUISE JONES GRIGSBY	O'Brien Energy Company	0956-336	Louisian	BIENVILLE
17.013.36200.0012	03-11-2003	CINDY LOU CLOYD	O'Brien Energy Company	0956-345	Louisian	BIENVILLE
17.013.36200.0013	08-11-1989	DEWEY STOVALL	Equity Oil Company	0567-249	Louisian	BIENVILLE
17.013.36200.0014	08-11-1989	BETTY JEAN MCGEE			Louisian	BIENVILLE
17.013.36200.0015	01-15-1990	BETTE JEANETTE JONES KRACK	Equity Oil Company	589-098	Louisian	BIENVILLE
17.013.36200.0016	01-31-1990	CELIA ANN JONES	Equity Oil Company	589-080	Louisian	BIENVILLE
17.013.36200.0017	02-08-1990	JANIS J. GRIGSBY	Equity Oil Company	589-0068	Louisian	BIENVILLE
17.013.36200.0018	02-08-1990	LUCILLE B. JONES, WIDOW	Equity Oil Company	589-074	Louisian	BIENVILLE
17.013.36200.0019	01-15-1990	ANNA JONES SISSON TEJADA	Equity Oil Company	0589-086	Louisian	BIENVILLE
17.013.36200.0020	06-13-1997	DANIEL VICTOR READ	Equity Oil Company	0797-014	Louisian	BIENVILLE
17.013.36200.0021	07-20-1989	GROVER A . JONES ET UX	Equity Oil Company	567-202	Louisian	BIENVILLE
17.013.36200.0022	06-13-1997	CINDY LOU CLOYD	Equity Oil Company	0797-019	Louisian	BIENVILLE
17.013.36200.0023	03-06-2003	LAUREEN KENDALL KRACK	O'Brien Energy Company	0956-339	Louisian	BIENVILLE
17.013.36200.0024	03-11-2003	DANIEL VICTOR READ	O'Brien Energy Company	0956-348	Louisian	BIENVILLE
17.013.36200.0026	04-15-1989	HOLSTUN C. FLEMING ET UX	Equity Oil Company		Louisian	BIENVILLE
17.013.36200.0027		MS. ANNIE LOU JONES ET UX	<u> </u>			BIENVILLE
17.013.36200.0028		JOHN D. BUTLER	OHIO OIL CO	0231-051		BIENVILLE
17.013.36200.0029	05-01-1964	R. A. REED	RUTLEDGE H. DEAS, JR.	0260-164	Louisian	BIENVILLE
17.013.36200.0030	01-02-1965	E. L. REED			Louisian	BIENVILLE
17.013.36200.0031		WIMBERLY MOORE ET AL	Sun Oil Company	0253-187		BIENVILLE
17.013.36200.0032		H. W. Perritt	Sun Oil Company	0240-117		BIENVILLE
17.013.36200.0033		A.G. NUSSER	Sun Oil Company	0240-116		BIENVILLE
17.013.36200.0034		MS. TOM J. MOFFITT ET AL	Sun Oil Company	0240-115		BIENVILLE
17.013.36200.0035		EBERT D. HOLSTUN	Sun Oil Company	0240-168		BIENVILLE
17.013.36200.0036		HARRY HOLSTUN				BIENVILLE
17.013.36200.0037		JOHN D. WILLIAMS, JR.	Equity Oil Company	567-220		BIENVILLE
17.010.00200.0037	00-00-1909	OOTHY D. WILLIAMO, UIV.	Equity On Company	001-220	Louisidii	PILIAVILLE

LEASE#	DATE	LESSOR	LESSEE	BOOK-PAGE	STATE COUNTY
23.005.42002.0001	09-24-1958	Norma B. Roberts	H.G. Anderson	0031-348	Mississi Amite

LEASE#	DATE	LESSOR	LESSEE	BOOK-PAGE	STATE	COUNTY
42.349.16000.0001	11-23-2011	Fredereka Faye Johnson	O'Brien Resources, LLC	00001331	Texas	Navarro
42.349.16000.0007	11-23-2011	Billy Doyle Murphree	O'Brien Resources, LLC	00001324	Texas	Navarro
42.349.16000.0009	11-23-2011	Bonnie Jeanne Hocutt	O'Brien Resources, LLC	00001325	Texas	Navarro
42.349.16000.0011	12-13-2011	Wayne Gatlin	O'Brien Resources, LLC	00001328	Texas	Navarro
42.349.16000.0012	12-13-2011	Amy Gatlin Heard Caviness	O'Brien Resources, LLC	00001327	Texas	Navarro
42.349.16000.0013	12-07-2011	John G. Jackson et ux	O'Brien Resources, LLC	00001326	Texas	Navarro
42.349.16000.0014	12-12-2011	Robert L. Boyd et ux	O'Brien Resources, LLC	00001329	Texas	Navarro
42.349.16000.0016	12-16-2011	Donald Jock et ux	O'Brien Resources, LLC	00001315	Texas	Navarro
42.349.16000.0022	12-05-2011	Geneva Penney	O'Brien Resources, LLC	00001401	Texas	Navarro
42.349.16000.0038	01-17-2012	Linda Boyd Sparkman	O'Brien Resources, LLC	00001350	Texas	Navarro
42.349.16000.0039	01-17-2012	Carol Boyd Stephenson	O'Brien Resources, LLC	00001349	Texas	Navarro
42.349.16000.0040	01-10-2012	Floyd Dell Nutt et ux	O'Brien Resources, LLC	00001352	Texas	Navarro
42.349.16000.0043	01-10-2012	Marsylene Jenkins	O'Brien Resources, LLC	00001351	Texas	Navarro
42.349.16000.0044	01-10-2012	Mary Sue Brown	O'Brien Resources, LLC	00001347	Texas	Navarro
42.349.16000.0047	12-15-2011	Wallis M. Boyd et ux	O'Brien Resources, LLC	00001343	Texas	Navarro
42.349.16000.0048	01-05-2012	Austin Walker Boyd	O'Brien Resources, LLC	00001342	Texas	Navarro
42.349.16000.0050	01-05-2012	James Jarrell Boyd	O'Brien Resources, LLC	00001289	Texas	Navarro
42.349.16000.0052	01-17-2012	Betty Boyd Storms	O'Brien Resources, LLC	00001339	Texas	Navarro
42.349.16000.0054	01-23-2012	Ellen Nutt, Life Tennant, & Kristy Nutt Ind & Remainderman	O'Brien Resources, LLC	00001338	Texas	Navarro
42.349.16000.0055	01-10-2012	Linda Nutt Lee	O'Brien Resources, LLC	00001337	Texas	Navarro
42.349.16000.0056	01-10-2012	Ruby Lee Radney	O'Brien Resources, LLC	00001336	Texas	Navarro
42.349.16000.0065	01-03-2012	Billy Wayne Chandler et ux	O'Brien Resources, LLC	00001298	Texas	Navarro
42.349.16000.0080	01-10-2012	Ivie Ray Nutt et ux	O'Brien Resources, LLC	00002275	Texas	Navarro
42.349.16000.0094	02-15-2012	Brice R. Bonner IV	O'Brien Resources, LLC	00001322	Texas	Navarro
42.349.16000.0095	02-15-2012	Kristen Bonner Jennings	O'Brien Resources, LLC	00001323	Texas	Navarro
42.349.16000.0188	01-31-2011	ANN LIVELY GATLIN	Team Invest Inc.	00002614	Texas	Navarro
42.349.16000.0268	02-16-2012	D'Ann Gatlin Carrington	O'Brien Resources, LLC	00001266	Texas	Navarro
42.349.16000.0279	03-05-2012	Suzanne Slatkin	O'Brien Resources, LLC	00002320	Texas	Navarro
42.349.16000.0286	03-05-2012	Larry R. Norman et ux Revocable Living Trust	O'Brien Resources, LLC	00003494	Texas	Navarro
42.349.16000.0305	09-22-2011	Percy Travis Fullwood, II	O'Brien Resources, LLC	00007874	Texas	Navarro
42.349.16000.0307	03-05-2012	Beth Norman Hyatt	O'Brien Resources, LLC	00002322	Texas	Navarro
42.349.16000.0352	03-13-2012	John T. May et ux	O'Brien Resources, LLC	00002337	Texas	Navarro
42.349.16000.0365	05-08-2012	William R. Dorsett	O'Brien Resources, LLC	00005373	Texas	Navarro
42.349.16000.0370	03-30-2012	Doris Jean Eaton	O'Brien Resources, LLC	00003905	Texas	Navarro
42.349.16000.0371	05-08-2012	Rosemary Moore	O'Brien Resources, LLC	00005371	Texas	Navarro
42.349.16000.0439	06-07-2012	Nicky Dale Herndon	O'Brien Resources, LLC	00005375	Texas	Navarro
42.349.16000.0451	06-07-2012	James A. Wood	O'Brien Resources, LLC	00006756	Texas	Navarro
42.349.16000.0452	06-05-2012	Lori A. Dohanich	O'Brien Resources, LLC	00006757	Texas	Navarro
42.349.16000.0478	07-10-2012	Susan Wood	O'Brien Resources, LLC	00006782	Texas	Navarro
42.349.16000.0531	08-14-2012	Paul Wood	O'Brien Resources, LLC	00008261	Texas	Navarro
42.349.16000.0562		Jean Elizabeth Wilson	O'Brien Resources, LLC	00008272	Texas	Navarro
42.349.16000.0571		Chad Bingham and wife, Amy	O'Brien Resources, LLC	00010147	Texas	Navarro
		J =,,	,			

Book 0180 Page 354

East Nancy Field, Clarke County, Mississippi:

LEASES:

Book 0190 Page 99

- 1. Oil, Gas and Mineral Lease dated April 15, 1959, by and between Jessie Allen and wife, Ardee Allen, as Lessors, and Southwest Gas Producing Company, Inc., as Lessee, recorded in Book 11 at Page 345 of the Oil & Gas Records of Clarke County, Mississippi.
- 2. Oil, Gas and Mineral Lease dated May 20, 1963, by and between Ralph V. St. John, as Lessor, and Placid Oil Company, Inc., as Lessee, recorded in Book 20 at Page 590 of the Oil & Gas Records of Clarke County, Mississippi.
- 3. Lease dated January 1, 1964, by and between Masonite Corporation, as Lessor, and Shell Oil Company, Inc., as Lessee, recorded in Book 26 at Page 39 of the Oil & Gas Records of Clarke County, Mississippi.
- 4. Oil, Gas and Mineral Lease dated September 1, 1967, by and between R. S. Vaught, as Lessor, and Roger A. Mateer, as Lessee, recorded in Book 34 at Page 283 of the Oil & Gas Records of Clarke County, Mississippi.
- 5. Oil, Gas and Mineral Lease dated September 12, 1967, by and between Helen Irene Boyd, as Lessor, and W. Baldwin Lloyd, as Lessee, recorded in Book 34 at Page 340 of the Oil & Gas Records of Clarke County, Mississippi.
- 6. Oil, Gas and Mineral Lease dated September 25, 1967, by and between Roger A. Mateer, as Lessor, and W. Baldwin Lloyd, as Lessee, recorded in Book 34 at Page 391 of the Oil & Gas Records of Clarke County, Mississippi.
- 7. Oil, Gas and Mineral Lease dated September 5, 1967, by and between Jennie G. King and Nella J. Hand, as Lessors, and Robert Mosbacher, as Lessee, recorded in Book 34 at Page 501 of the Oil & Gas Records of Clarke County, Mississippi.
- 8. Oil, Gas and Mineral Lease dated September 13, 1967, by and between S. H. Lovitt, as Lessor, and Robert Mosbacher, as Lesso, recorded in Book 35 at Page 81 of the Oil & Gas Records of Clarke County, Mississippi.
- 9. Oil, Gas and Mineral Lease dated September 5, 1967, by and between Jent P. Mitchell, Jr. and wife Margaret A. Mitchell, as Lessors, and Robert Mosbacher, as Lessee, recorded in Book 35 at Page 218 of the Oil & Gas Records of Clarke County, Mississippi.
- 10. Oil, Gas and Mineral Lease dated December 14, 1967, by and between Roger A. Mateer, as Lessor, and W. Baldwin Lloyd, as Lessee, recorded in Book 36 at Page 628 of the Oil & Gas Records of Clarke County, Mississippi.
- 11. Oil, Gas and Mineral Lease dated March 15, 1968, by and between Laurel A. Hicks, Sr., as Lessor, and Victor P. Smith, as

Lesses, recorded in Book 40 at Page 150 of the Oil Bookal Received 55 of Clarke County, Mississippi.

- 12. Oil, Gas and Mineral Lease dated March 15, 1968, by and between Laurel A. Hicks, Sr., as Lessor, and Victor P. Smith, as Lessee, recorded in Book 45 at Page 277 of the Oil & Gas Records of Clarka County, Mississippi.
- 13. Oil, Gas and Mineral Lease dated March 15, 1968, by and between Laurel A. Hicks, Sr., as Lessor, and Victor P. Smith, as Lessee, recorded in Book 40 at Page 152 of the Oil & Gas Records of Clarke County, Mississippi.
- 14. Oil, Gas and Mineral Lease dated March 15, 1968, by and between Laurel A. Hicks, Sr., as Lessor, and Victor P. Smith, as Lessee, recorded in Book 45 at Page 279 of the Oil & Gas Recorde of Clarke County, Mississippi.
- 15. Oil, Gas and Mineral Lease dated March 15, 1968, by and between Mrs. Dorothy Phillips, as Lessor, and Victor P. Smith, as Lessee, recorded in Book 40 at Page 154 of the Oil & Gas Records of Clarke County, Mississippi.
- 16. Oil, Gas and Mineral Lease dated March 15, 1968, by and between Mrs. Dorothy Phillips, as Lessor, and Victor P. Smith, as Lessee, recorded in Book 45 at Page 275 of the Oil & Gas Records of Clarke County, Mississippi.
- 17. Oil, Gas and Mineral Lease dated April 4, 1968, by and between Effie Watts Heard, as Lessor, and Southwest Gas Producing Company, Inc., as Lessee, recorded in Book 43 at Page 361 of the Oil & Gas Records of Clarke County, Mississippi.
- 18. Oil, Gas and Mineral Lease dated April 4, 1968, by and between Jeff Watts, as Lessor, and Southwest Gas Producing Company, Inc., as Lessee, recorded in Book 43 at Page 363 of the Oil & Gas Records of Clarke County, Mississippi.
- 19. Oil, Gas and Mineral Lease dated July 18, 1968, by and between Jack R. Creel, as Lessor, and Karl E. Weigl, as Lessor recorded in Book 45 at Page 9 of the Oil & Gas Records of Clarke County, Mississippi.
- 20. Oil, Gas and Mineral Lease dated July 3, 1968, by and between Ralph C. Brashier, et al, as Lessors, and Getty Oil Company, as Lessee, recorded in Book 45 at Page 118 of the Oil & Gas Records of Clarke County, Mississippi.
- 21. Oil, Gas and Mineral Lease dated May 27, 1968, by and between J. D. Haynes, as Lessor, and Getty Oil Company, as Lessee, recorded in Book 47 at Page 56 of the Oil & Gas Records of Clarke County, Mississippi.
- 22. Oil, Gas and Mineral Lease dated May 27, 1968, by and between J. D. Haynes, as Lessor, and Getty Oil Company, as Lessee, recorded in Book 65 at Page 578 of the Oil & Gas Records of Clarke County, Mississippi.
- 23. Oil, Gas and Mineral Lease dated October 11, 1968, by and between Wanda Sue Phillips, as Lessor, and Victor P. Smith, as Lessee, recorded in Book 47 at Page 145 of the Oil & Gas Records of Clarke County, Mississippi.
- . 24. Oil, Gas and Mineral Lease dated October 11, 1968, by and between Robert Del Phillips, as Lessor, and Victor P. Smith, as Lessee, recorded in Book 47 at Page 147 of the Oil & Gas Record of Clarke County, Mississippi.
 - 25. Oil and Gas Lease dated September 24, 1970, by and between Quash Parker, as Lessor, and Shell Oil Company, as Lessoe,

- recorded in Book 66 at Page 150 of the Oil & Gas Records of \checkmark Clarks County, Mississippi.
- 26. Oil and Gas Lease dated September 24, 1970, by and between Donald E. Thomas, as Lessor, and Shell Oil Company, as Lessee, recorded in Book 66 at Page 123 of the Oil & Gas Records of Clarke County, Mississippi.
- 27. Oil and Gas Lease dated September 24, 1970, by and between James A. Thomas, as Lessor, and Shell Oil Company, as Lease, recorded in Book 66 at Page 120 of the Oil & Gas Records of Clarke County, Mississippi.
- 28. Oil and Gas Lease dated September 23, 1970, by and between M. P. McCellan, Sr., as Lessor, and Shell Oil Company, as Lessee, recorded in Book 66 at Page 147 of the Oil & Gas Records of Clarke County, Mississippi.
- 29. Oil and Gas Lease dated September 21, 1970, by and between Jule McGill, as Lessor, and Shell Oil Company, as Lessoe, recorded in Book 66 at Page 144 of the Oil & Gas Bacords Page 101
- 30. Oil and Gas Lease dated September 24, 1970, by and between B. E. Flowers, et al, as Lessors, and Shell Oil Company, as Lessee, recorded in Book 66 at Page 141 of the Oil & Gas Records of Clarke County, Mississippi.
- 31. Oil and Gas Lease dated September 29, 1970, by and between Ethel P. Touchstone, as Lessor, and Shell Oil Company, as Lessoe, recorded in Book 66 at Page 117 of the Oil & Gas Records of Clarke County, Mississippi.
- 32. Oil and Gas Lease dated September 24, 1970, by and between Blis G. Travis, as Lessor, and Shell Oil Company, as Lessoe, recorded in Book 66 at Page 114 of the Oil & Gas Records of Clarke County, Mississippi.
- 33. Oil and Gas Lease dated October 1, 1970, by and between Mary R. Cooley, et al, as Lessors, and Shell Oil Company, as Lessee, recorded in Book 66 at Page 138 of the Oil & Gas Recorde of Clarke County, Mississippi.
- 34. Oil and Gas Lease dated September 24, 1970, by and between Nell Travis Collins, as Lessor, and Shell Oil Company, as Lessee, recorded in Book 66 at Page 135 of the Oil & Gas Records of Clarke County, Mississippi.
- 35. Oil and Gas Lease dated October 26, 1970, by and between Frank Heard, as Lessor, and Shell Oil Company, as Lessee, recorded in Book 66 at Page 90 of the Oil & Gas Records of Clarke County, Mississippi.
- 36. Oil and Gas Lease dated February 22, 1971, by and between the Masonite Corporation, as Lessor, and Shell Oil Company, as Lessee, recorded in Book 68 at Page 577 of the Oil & Gas Records of Clarke County, Mississippi.
- 37. Oil and Gas Lease dated October 26, 1970, by and between Lillie M. H. Brown, as Lessor, and Shell Oil Company, as Lessee recorded in Book 66 at Page 132 of the Oil & Gas Records of Clarke County, Mississippi.
- 38. Oil and Gas Lease dated October 30, 1970, by and between Rance Thomas, as Lessor, and Shell Oil Company, as Lessee, recorded in Book 66 at Page 111 of the Oil & Gas Records of Clarke County, Mississippi.
- 39. Oil and Gas Lease dated October 26, 1970, by and between Rllen Heard, as Lessor, and Shell Oil Company, as Lessee, recorded in Book 66 at Page 102 of the Oil & Gas Records of Clarke County, Mississippi.

- 40. Oil and Gas Lease dated October 27, 1970, by and between Lillie Mae Thomas, as Lessor, and Shell Oil Company, as Lessee, recorded in Book 66 at Page 108 of the Oil & Gas Records of Clarke County, Mississippi.
- 41. Oil and Gas Lease dated October 28, 1970, by and between Jessie Mae Moody, as Lessor, and Shell Oil Company, as Lessee, recorded in Book 66 at Page 129 of the Oil & Gas Records of Clarke County, Mississippi.
- 42. Oil and Gas Lease dated October 31, 1970, by and between Betty J. Mitchell, as Lessor, and Shell Oil Company, as Lessee, recorded in Book 66 at Page 126 of the Oil & Gas Records of Clarke County, Mississippi.
- 43. Oil and Gas Lease dated October 26, 1970, by and between Dan Heard, as Lessor, and Shell Oil Company, as Lessee, recorded in Book 66 at Page 99 of the Oil & Gas Records of Clarke County, Mississippi.
- 44. Oil and Gas Lease dated October 31, 1970, by and between John Thomas, as Lessor, and Shell Oil Company, as Lessee, recorded in Book 66 at Page 105 of the Oil & Gas Records of Clarke County, Mississippi.
- 45. Oil and Gas Lease dated October 26, 1970, by and between Frank Heard, as Lessor, and Shell Oil Company, as Lessee, recorded in Book 66 at Page 96 of the Oil & Gas Records of Clarke County, Mississippi.
- 46. Oil and Gas Lease dated October 31, 1970, by and between Edna Reasonober, as Lessor, and Shell Oil Company, as Lessee, recorded in Book 66 at Page 458 of the Oil & Gas Records of Clarke County, Mississippi.
- 47. Oil and Gas Lease dated October 26, 1970, by and between Luke Heard, as Lessor, and Shell Oil Company, as Lessee, recorded in Book 66 at Page 93 of the Oil & Gas Records of Clarke County, Mississippi.
- 48. Oil and Gas Lease dated October 31, 1970, by and between Vincent Thomas, as Lessor, and Shell Oil Company, as Lessee, recorded in Book 66 at Page 455 of the Oil & Gas Records of Clarke County, Mississippi.
- 49. Oil and Gas Lease dated December 7, 1970, by and between Charles Thomas, as Lessor, and Shell Oil Company, as Lessee, recorded in Book 66 at Page 482 of the Oil & Gas Records of Clarke County, Mississippi.
- 50. Oil and Gas Lease dated December 29, 1970, by and between Robert Thomas, as Lessor, and Shell Oil Company, as Lessee, recorded in Book 67 at Page 16 of the Oil & Gas Records of Clarke County, Mississippi.

Book 0180 Page 357

Book 0190 Page *102

EXHIBIT B

WELLS

Operated vs Nonoperated	Operator	Prop Num	Property Name	Well Status	API County/ Paris	sh St ^{Joi} r	nt Interest Billing Decimal	Total O'BENCO IV Revenue + minerals (Roy + ORRI + REV WI)
Operated	O'BENCO, Inc.	15709	GAYLAN JANE COLLIER SWD	SWD	42-033-31623 Borden	TX	1.00000000	-
Operated	O'BENCO, Inc.	15702	JONES 519#1	producing	42-033-32301 Borden	TX	0.89623921	0.66707957
Operated	O'BENCO, Inc.	15705	JONES 519-2H	producing	42-033-32340 Borden	TX	0.89623921	0.66707957
Operated	O'BENCO, Inc.	14281	LITTLE #1	Shut-in	42-347-32992 Nacogdoches	TX	0.81062922	0.63097312
Operated	O'BENCO, Inc.	14282	LITTLE #2 (West Gas Unit)	Shut-in	42-347-33043 Nacogdoches	TX	0.80933637	0.55405591
Operated	O'BENCO, Inc.	14283	LITTLE #3 (East Gas Unit)	producing	42-347-33055 Nacogdoches	TX	0.80933637	0.62996678
Operated	O'BENCO, Inc.	16001	BONNER #1-H	producing	42-349-34987 Navarro	TX	0.97975978	0.76225468
Operated	O'BENCO, Inc.	16003	FINLEY#1	Shut-in	42-349-34998 Navarro	TX	0.99496315	0.75693022
Operated	O'BENCO, Inc.	16011	BRICE BONNER	producing	42-349-35005 Navarro	TX	0.97590080	0.75388337
Operated	O'BENCO, Inc.	14201	SAMFORD #1	Shut-in	42-419-30822 Shelby	TX	0.27562500	0.20576808
Operated	O'BENCO, Inc.	14202	SAMFORD #2	Shut-in	42-419-30848 Shelby	TX	0.28436041	0.21228131
Operated	O'BENCO, Inc.	14221	GRIFFIN #1	Shut-in	42-419-30997 Shelby	TX	0.58706783	0.43935690
Operated	O'BENCO, Inc.	14203	SAMFORD #3	Shut-in	42-419-31056 Shelby	TX	0.27250284	0.20448827
Operated	O'BENCO, Inc.	14222	GRIFFIN #2	Shut-in	42-419-31068 Shelby	TX	0.56855883	0.42641912
Operated	O'BENCO, Inc.	14223	GRIFFIN #3	producing	42-419-31091 Shelby	TX	0.58855883	0.44201912
Operated	O'BENCO, Inc.	14224	GRIFFIN #4	producing	42-419-31124 Shelby	TX	0.56855883	0.42641912
Operated	O'BENCO, Inc.	14204	SAMFORD #4	producing	42-419-31125 Shelby	TX	0.27250284	0.20448827
Operated	O'BENCO, Inc.	14225	GRIFFIN #5	producing	42-419-31211 Shelby	TX	0.65758583	0.49318937
Operated	O'BENCO, Inc.	14226	GRIFFIN #6	Shut-in	42-419-31457 Shelby	TX	0.82775285	0.61343771
Operated	O'Brien Resources LLC	18803	SWART 21-1	producing	15-063-22107 Gove	KS	0.92110169	0.73515428
Operated	O'Brien Resources LLC	18802	PRATHER FARMS 22-1	producing	15-063-22108 Gove	KS	0.87131298	0.70903094
Operated	O'Brien Resources LLC	18806	MSJ-PENNER 28 #1	producing	15-063-22184 Gove	KS	0.94601212	0.75503593
Operated Operated	O'Brien Resources LLC O'Brien Resources LLC	18141 18143	JENNISON 1-1 COLT 45 FBC #2-1	Shut-in	15-101-22099 Lane 15-101-22117 Lane	KS KS	0.63888888	0.49993055 0.49993055
		18143 18116		producing Shut-in				0.49993055 0.66657407
Operated	O'Brien Resources LLC O'Brien Resources LLC		BOOMHOWER #36-1	Shut-in SWD	15-101-22151 Lane	KS	0.85185185	
Operated Operated		18146	JENNISON 1-2 SWD		15-101-22203 Lane	KS	0.8888889	0.69555555
Operated	O'Brien Resources LLC O'Brien Resources LLC	18148 18228	SNIDER 35-1 JENNISON 36 #1	Shut-in producing	15-101-22289 Lane 15-101-22311 Lane	KS KS	0.92000000 0.92000000	0.71990000 0.71990000
Operated	O'Brien Resources LLC	18229	SHARP SEED 2#1	Shut-in	15-101-22314 Lane	KS	0.80500000	0.62965000
Operated	O'Brien Resources LLC	18117	BOOMHOWER 36 #2	Shut-in	15-101-22343 Lane	KS	0.85185185	0.66657407
Operated	O'Brien Resources LLC	18230	SNIDER-SHARP SEED 35#1	producina	15-101-22445 Lane	KS	0.86250000	0.67652344
Operated	O'Brien Resources LLC	18173	BOOMHOWER 36#3	producing	15-101-22464 Lane	KS	0.85185185	0.66657407
Operated	O'Brien Resources LLC	18174	REDBURN 1-1	producing	15-101-22468 Lane	KS	0.73250000	0.57318125
Operated	O'Brien Resources LLC	18234	COLT 45 FBCE 2#4	producing	15-101-22478 Lane	KS	0.66758081	0.52233665
Operated	O'Brien Resources LLC	18175	BOOMHOWER 36 #4	producing	15-101-22481 Lane	KS	0.85185185	0.66657407
Operated	O'Brien Resources LLC	18235	SHARP SEED 2#2	producing	15-101-22493 Lane	KS	0.80500000	0.62965000
Operated	O'Brien Resources LLC	18156	HARPER 35#1	producing	15-101-22516 Lane	KS	0.92000000	0.71990000
Operated	O'Brien Resources LLC	18703	VONDRACEK 4#1	producing	15-165-22088 Rush	KS	0.88000000	0.68860000
Operated	O'Brien Resources LLC	18707	MOEDER TRUST 3 No 1	producing	15-165-22130 Rush	KS	0.88000000	0.71610000
Operated	O'Brien Resources LLC	18714	VONDRACEK ET AL 4 #2	producing	15-165-22133 Rush	KS	0.88000000	0.71610000
Operated	O'Brien Resources LLC	18715	VONDRACEK ET AL 34 #1	producing	15-165-22136 Rush	KS	0.88000000	0.71610000
Operated	O'Brien Resources LLC	18717	BESPERAT 33 #1	producing	15-165-22139 Rush	KS	0.88000000	0.71610000
Operated	O'Brien Resources LLC	18709	MOEDER 3 #2	Shut-in?	15-165-22143 Rush	KS	0.88000000	0.71610000
Operated	O'Brien Resources LLC	36207	CLOYD 29 #1	producing	17-013-21108 Bienville	LA	0.21639593	0.15193833
Operated	O'Brien Energy Company	42105	MASONITE 17-12	Shut-in	23-023-20108 Clarke	MS	0.97592500	-
Operated	O'Brien Energy Company	42106	MASONITE 17-13	Shut-in	23-023-20035 Clarke	MS	0.97592500	-
Operated	O'Brien Energy Company	42109	REDDOCK 17-15	SWD	23-023-20078 Clarke	MS	0.62938120	-
Operated	O'Brien Energy Company	42110	JESSIE ALLEN 20-3	Shut-in	23-023-20049 Clarke	MS	1.00000000	-
Operated	O'Brien Energy Company	42103	JESSIE ALLEN 17 14 #1	producing	23-023-20068 Clarke	MS	0.63826036	0.49676301
Nonop	Smith Operating and Management	36206	McCULLIN FAMILY TRUST etal #1	producing	17-013-21528 Bienville	LA	0.04860010	0.03532769
Nonop	Franks/Brammer	36101	ALLEN EST. NO. 1 ALT	producing	17-015- Bossier	LA	0.09389424	0.07302473
Nonop	Franks/Brammer	36102	ALLEN EST. NO. 2 ALT	Shut-in	17-015- Bossier	LA	0.81455920	0.06339613
Nonop	Comstock Oil & Gas, Lp	11308	SPIVEY#1	Shut-in	17-027-21567 Union	LA	0.06185350	0.04408805
Nonop	Indigo Minerals LLC	11340	WILL DRILL NORA G JONES	ORLLC ORRI	17-027-22169 Claiborne	LA	-	-
Nonop	PetroChem Operating Co, Inc	11311	SPIVEY 17-1	producing	17-027-22334 Claiborne	LA	0.05332189	0.03465923
Nonop	PetroChem Operating Co, Inc	11312	HATTER 17-1	producing	17-027-22344 Claiborne	LA	0.02857038	0.01857075
Nonop	PetroChem Operating Co, Inc	11313	BAILEY 17-1	producing	17-027-22469 Claiborne	LA	0.05332189	0.03465923
Nonop	TMR Exploration	11481	MAX DREYFUS ETAL 106 #1	500% non-consent penalty	17-077-20616 Pointe Coupee	LA	0.12500000	0.09375000
Nonop	Stroud Petroleum Inc.	11271 43301	PAYNE #1 MASONITE 15 15 #1	producing Shut-in	17-111-25591 Union 23-023-20276 Clarke	LA MS	0.02995000 0.00343600	0.02405983
Nonop	Signal LLC Signal LLC	43301 43330	MASONITE 15 15 #1 NANCY RII FY	Shut-in Shut-in	23-023-20276 Clarke 23-077-20114 Lawrence	MS MS	0.00343600 0.00029297	0.00021094
Nonop Nonop	Signal LLC Tellus Operating Group LLC	43330 43104	NANCY RILEY LANE SHANKS UNIT #1	Shut-in producing	23-077-20114 Lawrence 23-129-20039 Smith	MS MS	0.00029297	0.00021094 0.04204637
Nonop	Ranken Energy	19184	BRAUM, W. H. #1-7	producing	25-129-20039 Smith 35-087-22013 McClain	OK		0.04204637
. volop	. Camon Energy	19104	5.5 (GW), 11.11.#1-/	producing	33-007-22013 MICOIAIII	OK	•	-

EXHIBIT C ASSIGNED CONTRACTS

EXHIBIT C ASSIGNED CONTRACTS

Contract	Party 1	Party 2	Signed	Effective	State	Counties	Prospect
Participation Agreement	O'BENCO IV, LP	Marlin Exploration, LLC	29-Jul-2015	20-Jul-2015	AR	Columbia	Pisgah
JOA	Getty Oil Company	Mosbacher et al	15-Aug-1968	15-Aug-1968	MS	Clarke	East Nancy
					IΔ	Bienville	West Arcadia

EXHIBIT D

ASSIGNED SURFACE INTEREST

None.

EXHIBIT E

FORM OF DEED

Exhibit Version

EXHIBIT E

FORM OF DEED

DEED

THE STATE OF TEXAS

COUNTY OF [____]

KNOW ALL MEN BY THESE PRESENTS: That O'BENCO IV, LP, a Delaware limited partnership ("Grantor"), for and in consideration of the sum of TEN AND NO/100 (\$10.00) DOLLARS and other valuable consideration to the undersigned paid by the grantee herein named, the receipt of which is hereby acknowledged, has GRANTED, SOLD and CONVEYED and by these presents does GRANT, SELL, AND CONVEY unto PACHIRA OIL AND GAS, LLC, a Delaware limited liability company ("Grantee"), all of Grantor's right, title and interest in and to the real property described in Exhibit A attached hereto (the "Property"), also, all and singular, the rights and appurtenances pertaining thereto, together with all of Grantor's right, title and interest thereunder. TO HAVE AND TO HOLD the Property and premises unto the said Grantee, their heirs and assigns forever, together with all and singular the rights and appurtenances thereto and anywise belonging, subject to all matters of record and the matters herein stated.

This Deed is made without warranty of title of any kind whatsoever, express, implied or statutory, and without recourse, even as to the return of any consideration, but with full substitution and subrogation of Grantee, and all persons claiming by, through and under Grantee, to the extent assignable, in and to all covenants and warranties by Grantor's predecessors in title and with full subrogation of all rights accruing under the statutes of limitation or prescription under the laws of the State of Texas. Any covenants or warranties implied by statute or law by the use of the word "grants," "sells," "transfers" or "conveys" or other similar words in this Deed are hereby expressly retained, disclaimed, waived and negated. It is the intent of Grantor to convey particular rights in

land itself as specifically described in this Deed and not just a chance of title. Consequently, this instrument is a conveyance without warranties, as provided herein, and not a quitclaim.

Grantor disclaims all liability and responsibility for any representation, warranty, statements or communications (orally or in writing) to any person (including, any information contained in any opinion, information or advice that may have been provided to any such person by any officer, director, stockholder, partner, employee, agent, consultant, representative or contractor of Grantor, its affiliates or any engineer or engineering firm, or other agent, consultant or representative) wherever and however made with respect to the transactions contemplated hereby. BY ACCEPTING THIS DEED GRANTEE ACKNOWLEDGES THAT GRANTOR HAS MADE NO, AND GRANTOR HEREBY EXPRESSLY DISCLAIMS AND NEGATES, AND GRANTEE HEREBY EXPRESSLY WAIVES, ANY REPRESENTATION OR WARRANTY, EXPRESS, IMPLIED, AT COMMON LAW, BY STATUTE OR OTHERWISE RELATING TO (a) PRODUCTION RATES, RECOMPLETION OPPORTUNITIES, DECLINE RATES, OR THE QUALITY, QUANTITY OR VOLUME OF THE RESERVES OF MINERALS, IF ANY, ATTRIBUTABLE TO THE OIL AND GAS ASSETS, (b) THE ACCURACY, COMPLETENESS OR MATERIALITY OF ANY INFORMATION, DATA OR OTHER MATERIALS (WRITTEN OR ORAL) NOW, HERETOFORE OR HEREAFTER FURNISHED TO GRANTEE BY OR ON BEHALF OF GRANTOR, OR (c) THE ENVIRONMENTAL CONDITION OF THE OIL AND GAS ASSETS. NOTWITHSTANDING ANYTHING TO THE CONTRARY IN THIS DEED, GRANTOR EXPRESSLY DISCLAIMS AND NEGATES, AND BY ACCEPTING THIS DEED GRANTEE HEREBY WAIVES, (i) ANY IMPLIED OR EXPRESS WARRANTY OF MERCHANTABILITY, (ii) ANY IMPLIED OR EXPRESS WARRANTY OF FITNESS FOR A PARTICULAR PURPOSE, (iii) ANY

IMPLIED OR EXPRESS WARRANTY OF CONFORMITY TO MODELS OR SAMPLES OF MATERIALS, (iv) ANY IMPLIED OR EXPRESS WARRANTY OF FREEDOM FROM DEFECTS, WHETHER KNOWN OR UNKNOWN, (v) ANY AND ALL IMPLIED WARRANTIES EXISTING UNDER APPLICABLE LAW, AND (vi) ANY IMPLIED OR EXPRESS WARRANTY REGARDING ENVIRONMENTAL LAWS, THE RELEASE OF MATERIALS INTO THE ENVIRONMENT, OR PROTECTION OF THE ENVIRONMENT OR HEALTH, IT BEING THE EXPRESS INTENTION OF GRANTEE AND GRANTOR THAT THE OIL AND GAS ASSETS SHALL BE CONVEYED TO GRANTEE, AND GRANTEE SHALL ACCEPT SAME, AS IS, WHERE IS, WITH ALL FAULTS AND IN THEIR PRESENT CONDITION AND STATE OF REPAIR AND GRANTEE REPRESENTS TO GRANTOR THAT GRANTEE HAS MADE OR CAUSED TO BE MADE SUCH INSPECTIONS WITH RESPECT TO SUCH OIL AND GAS ASSETS AS GRANTEE DEEMS APPROPRIATE. GRANTOR AND GRANTEE AGREE THAT, TO THE EXTENT REQUIRED BY APPLICABLE LAW TO BE EFFECTIVE, THE DISCLAIMERS OF CERTAIN WARRANTIES CONTAINED IN THIS SECTION ARE "CONSPICUOUS" DISCLAIMERS FOR THE PURPOSES OF ANY APPLICABLE LAW, RULE OR ORDER.

[Signature pages follow]

EXECUTED this	day of	, 2019	
		GRANTOR O'BENCO IV, LP	
		By:	
THE STATE OF TEXAS	S §		
COUNTY OF []	§		
BEFORE ME	, the undersig	igned authority, on this day personally	appeared
	, known to me	ne to be the person whose name is subscribed to the	he foregoing
instrument, as [] of O'BENCO	O IV, LP, and acknowledged to me that he executed	the same for
the purposes and conside	eration therein exp	pressed, in the capacity therein stated, and as the act	t and deed o
said partnership.			
GIVEN UNDER	MY HAND AND	D SEAL OF OFFICE this theday of	, 2019.
Notary Public in and for the State of Texas			

EXECUTED this	day of	, 2019	
		GRANTEE PACHIRA OIL AND GAS, I	LC
		By:	
THE STATE OF []		
COUNTY OF []	§ §		
BEFORE ME,	the undersign	ed authority, on this da	y personally appeared
	, known to me	to be the person whose name is	subscribed to the foregoing
instrument, as [_] of PACHIRA	OIL AND GAS, LLC, and acl	knowledged to me that he
executed the same for the	purposes and consi	ideration therein expressed, in the	capacity therein stated, and
as the act and deed of said	company.		
GIVEN UNDER	MY HAND AND S	SEAL OF OFFICE this the	day of, 2019.
Notary Public in and for the State of			

(SIGNATURE PAGE TO DEED)

THE PROPERTY

EXHIBIT F

FORM OF ASSIGNMENT

EXHIBIT F

FORM OF ASSIGNMENT, BILL OF SALE AND CONVEYANCE

ASSIGNMENT, BILL OF SALE AND CONVEYANCE

THIS ASSIGNMENT, BILL OF SALE AND CONVEYANCE (this "Assignment"), effective as of 12:01 a.m., Central Standard Time on June 1, 2019 (the "Effective Time"), is by and between O'Benco IV, LP, a Delaware limited partnership ("Assignor"), and Pachira Oil and Gas, LLC, a Delaware Limited Liability Company ("Assignee"). Assignor and Assignee are sometimes referred to collectively herein as the "Parties" and individually as a "Party".

WHEREAS, Assignor is a debtor and debtor in possession under title 11 of the United States Code, 11 U.S.C. §§ 101, et seq. (the "Bankruptcy Code"), and filed a voluntary petition for relief under chapter 11 of the Bankruptcy Code on June 3, 2019 (the "Petition Date") in the United States Bankruptcy Court for the Eastern District of Texas (the "Bankruptcy Court"), where Assignor's bankruptcy case is being administered under Case No. 19-60384 (the "Bankruptcy Case");

WHEREAS, Assignor and Assignee entered into that certain Asset Purchase Agreement dated [●], 2019 (the "APA"), providing for the sale by Assignor of, and the purchase by Assignee of, the Oil and Gas Assets (as such term is defined below), and the terms and provisions of the APA shall not be deemed to have merged herein. In the event that any provision of this Assignment is construed to conflict with any provision of the APA, the provisions of the APA shall be deemed controlling to the extent of such conflict;

WHEREAS, references herein to terms which are defined separately herein or in APA shall have the meaning prescribed to such term herein or in the APA, as applicable, even if the term (as printed in this Assignment) is printed in ALL CAPITALIZED text, in each case, unless the context expressly requires otherwise:

WHEREAS, on [●], 2019, the Bankruptcy Court entered the Sale Order authorizing the Assignor to sell the Oil and Gas Assets to Assignee pursuant to the terms and conditions of the APA; and

NOW, THEREFORE, in consideration of the premises, the mutual promises herein made, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties agree as follows:

ARTICLE 1

ASSIGNMENT

1.1 Assignment.

(a) Assignor hereby GRANTS, SELLS, TRANSFERS, ASSIGNS, CONVEYS AND DELIVERS to Assignee all of the Oil and Gas Assets.

(b) The "Oil and Gas Assets" are all right, title and interest of Assignor in, to and under the following, but LESS AND EXCEPT the Excluded Assets:

(i) all Leases (other than the Excluded Leases), including (1) the Leases described on Exhibit A attached hereto, and (2) those Leases located in or under (y) the lands covered by the Leases described on Exhibit A attached hereto, and (z) the Leases and lands included in any units with which the Leases or the lands covered thereby may have been pooled, unitized or communitized (collectively, the "Assigned Leases and Interests");

(ii) the oil, gas, water, disposal, observation or injection wells located on the Assigned Leases and Interests, whether producing, non-producing, shut-in, or temporarily abandoned, including, without limitation, those described on <u>Exhibit B</u> (collectively, the "<u>Wells</u>", and together with the Assigned Leases and Interests, the "<u>Properties</u>");

(iii) all Hydrocarbons produced from or attributable to the Properties in storage owned by Assignor above custody transfer point at the Closing Date or produced on and after the Closing Date, and all proceeds attributable thereto;

(iv) all equipment, machinery, fixtures and other tangible personal property and improvements located on, used or held for use or obtained in connection with the operation of the Properties (collectively, the "Equipment");

(v) all gathering and processing systems related to the Properties;

(vi) all Contracts (other than the Excluded Contracts) including, operating agreements, exploration agreements, development agreements, seismic licenses (but only to the extent transferable without material restriction (including a material restriction against assignment without prior consent)), balancing agreements, farmout agreements, service agreements, and other contracts, agreements and instruments, including the Contracts described on Exhibit C attached hereto, as such Exhibit may be subsequently amended by Assignee pursuant to Section 2.5 of the APA, insofar as they relate to any other Oil and Gas Asset (collectively, the "Assigned Contracts");

(vii) all Surface Interests (other than Excluded Surface Interests), including those described on <u>Exhibit D</u> attached hereto, insofar as they relate to any other Oil and Gas Asset (collectively, the "<u>Assigned Surface Interests</u>");

(viii) except with respect to the Excluded Assets and the Excluded Liabilities, all claims, refunds, abatements, variances, allocations, causes of action, claims for relief, choses in action, rights of recovery, audit rights, rights of set-off, rights of indemnity, contribution or recoupment, counterclaims, cross-claims and defenses of Assignor to the extent related to the Oil and Gas Assets and arising or relating to events occurring from and after the Closing Date or related to the Assumed Liabilities;

(ix) all information, books, databases, files, records and data (whether in written or electronic format) relating directly to any Oil and Gas Asset and to any Assumed Liabilities except for (A) records that Assignor is prohibited from disclosing or transferring under any third party agreement or applicable law, (B) information entitled to legal privilege, including attorney work product and attorney-client communications (except for title opinions, which shall be included in the Records), (C) economic projections and records of offers from or negotiations with, Assignee or third parties with respect to any proposed transfer of any of the Oil and Gas Assets and economic analyses associated therewith (collectively, subject to such exclusions, the "Records"), which Records shall include all lease records, well records, well files, title records (including abstracts of title, title opinions and memoranda, and title curative

2

¹ NTD: Exhibit A to this Assignment should list all of the Assigned Leases, but mineral interests are being assigned via the mineral deed, so mineral interests can be removed from Exhibit A to this Assignment.

documents), correspondence (including email correspondence within the past five (5) years, to the extent related to the Oil and Gas Assets, with any third-party field land agent tasked with acquiring Leases on behalf of Assignor), production records; and all other data (but only to the extent transferable without material restriction (including a material restriction against assignment without prior consent)), and any other proprietary data which Assignor has the right to obtain (either without the payment of money or delivery of other consideration or unduly burdensome effort or, upon Assignee's written election, at Assignee's expense) and relating to the ownership, operation, development, maintenance or repair of, or the production, gathering, treatment, processing, storing, sale or disposal of Hydrocarbons or produced water from, the other Oil and Gas Assets; *provided*, that if any Records can only be assigned to Assignee with a fee or penalty, Assignee shall bear responsibility for such fee or penalty;

- (x) all trade credits, accounts receivable, note receivables, take or pay amounts receivable, and other receivables attributable to the other Oil and Gas Assets, with respect to any period of time on and after the Closing Date;
- (xi) to the extent that they may be assigned, all Permits that are primarily used in connection with the ownership or operation of the Oil and Gas Assets; and
- (xii) to the extent transferable without payment of any fee (unless Assignee agrees in writing to bear all such fees), all geological, geophysical, or seismic data relating to the Oil and Gas Assets (but excluding Assignor's or Assignor's Affiliates' interpretation of such geological, geophysical or seismic data and any such data that is subject to licensing requirements and/or other restrictions on transfers), including those surveys described on Schedule 2.1(b)(xii) attached hereto.
- 1.2 <u>Excluded Assets</u>. Notwithstanding the foregoing, the Oil and Gas Assets shall not include, and there is expected, reserved and excluded from the transaction contemplated hereby the following (collectively, the "<u>Excluded Assets</u>"):
 - (a) the Purchase Price delivered to Assignor pursuant to this Assignment;
- (b) all cash and cash equivalents, including checks, commercial paper, treasury bills, certificates of deposit, bank accounts and other bank deposits as of the Closing Date, in each case, excluding any item described in <u>Schedule 2.2(b)</u> attached hereto;
- (c) all trade credits, accounts receivable, note receivables, take or pay amounts receivable, and other receivables attributable to the Oil and Gas Assets with respect to any period of time prior to the Closing Date or attributable to matters constituting Excluded Liabilities;
- (d) all Hydrocarbons produced from or attributable to the Properties prior to the Closing Date, other than Hydrocarbons produced from or attributable to the Properties in storage owned by Assignor above custody transfer point at the Closing Date, and all proceeds attributable thereto;
- (e) all minute books, stock ledgers, corporate seals and stock certificates of Assignor;
- (f) all (i) corporate, financial, Tax and legal records of Assignor that relates to Assignor's business generally (excepting the same to the extent relating to the Assumed Liabilities and the Oil and Gas Assets) and (ii) books, records and files that relate to any Excluded Assets;
 - (g) all Excluded Leases and all Excluded Mineral Interests;

- (h) all Excluded Contracts;
- (i) all Excluded Surface Interests;
- (j) all rights to any refunds or credits of Taxes (or other related costs or expenses) that are borne by or the responsibility of Assignor or attributable to any Tax asset of Assignor;
- (k) any refunds due to Assignor by a third party for any overpayment of rentals, royalties, excess royalty interests or production payments attributable to the Oil and Gas Assets with respect to any period of time prior to the Closing Date;
- (l) subject to <u>Section 8.8</u> of the APA, all insurance policies and rights to proceeds thereof;
- (m) all prepayments, good faith and other bid deposits submitted by any third party under the terms of the Bidding Procedures Order;
- (n) all claims, refunds, abatements, variances, allocations, causes of action, claims for relief, choses in action, rights of recovery, audit rights, rights of set-off, rights of indemnity, contribution or recoupment, counter-claims, cross-claims and defenses of Assignor other than those constituting Oil and Gas Assets;
- (o) all rights, claims or causes of action by or in the right of Assignor against any current or former director or officer of Assignor;
 - (p) the Avoidance Actions;
- (q) all claims and causes of action of Assignor (i) arising from acts, omissions, or events, or damage to or destruction of property occurring prior to the Closing Date, or (ii) affecting any of the other Excluded Assets;
- (r) any documents or records withheld or not transferred pursuant to the exceptions set forth in Section 2.1(b)(ix) of the APA, including all documents and instruments of Assignor and its Affiliates that are protected by attorney-client privilege;
- (s) any and all claims for relief of Assignor under chapter 5 of the Bankruptcy Code; and
- (t) any rights, claims or causes of action of Assignor under this Assignment or any other Transaction Document.

ARTICLE 2

ADDITIONAL MATTERS

2.1 <u>Disclaimers</u>. NOTWITHSTANDING ANYTHING TO THE CONTRARY CONTAINED IN THIS ASSIGNMENT, EXCEPT AS AND TO THE EXTENT EXPRESSLY SET FORTH IN THIS ASSIGNMENT AND IN THE TRANSACTION DOCUMENTS, ASSIGNOR MAKES

NO REPRESENTATIONS OR WARRANTIES WHATSOEVER, AND DISCLAIMS ALL LIABILITY AND RESPONSIBILITY FOR ANY REPRESENTATION, WARRANTY, STATEMENT OR INFORMATION MADE OR COMMUNICATED (ORALLY OR IN WRITING) TO ASSIGNEE (INCLUDING ANY OPINION, INFORMATION, OR ADVICE THAT MAY HAVE BEEN PROVIDED TO ASSIGNEE BY ANY AFFILIATE OR REPRESENTATIVE OF ASSIGNOR OR BY ANY INVESTMENT BANK OR INVESTMENT BANKING FIRM, ANY PETROLEUM ENGINEER OR ENGINEERING FIRM, ASSIGNOR'S COUNSEL, OR ANY OTHER AGENT, CONSULTANT, OR REPRESENTATIVE OF ASSIGNOR). WITHOUT LIMITING THE GENERALITY OF THE FOREGOING, EXCEPT AS AND TO THE EXTENT EXPRESSLY SET FORTH IN THIS ASSIGNMENT AND IN THE TRANSACTION DOCUMENTS, ASSIGNOR EXPRESSLY DISCLAIMS AND NEGATES ANY REPRESENTATION OR WARRANTY, EXPRESS, IMPLIED, AT COMMON LAW, BY STATUTE, OR OTHERWISE, RELATING TO (A) THE TITLE TO ANY OF THE OIL AND GAS ASSETS, (B) THE CONDITION OF THE OIL AND GAS ASSETS (INCLUDING ANY IMPLIED OR EXPRESS WARRANTY OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, OR CONFORMITY TO MODELS OR SAMPLES OF MATERIALS), IT BEING DISTINCTLY UNDERSTOOD THAT THE OIL AND GAS ASSETS ARE BEING SOLD "AS IS," "WHERE IS," AND "WITH ALL FAULTS AS TO ALL MATTERS," (C) ANY INFRINGEMENT BY ASSIGNOR OF ANY PATENT OR PROPRIETARY RIGHT OF ANY THIRD PARTY, (D) ANY INFORMATION, DATA, OR OTHER MATERIALS (WHETHER WRITTEN OR ORAL) FURNISHED TO ASSIGNEE BY OR ON BEHALF OF ASSIGNOR (INCLUDING WITHOUT LIMITATION, IN RESPECT OF ANY SEISMIC DATA, THE EXISTENCE OR EXTENT OF HYDROCARBONS OR THE MINERAL RESERVES, THE RECOVERABILITY OF SUCH RESERVES, ANY PRODUCT PRICING ASSUMPTIONS, AND THE ABILITY TO SELL HYDROCARBON PRODUCTION AFTER THE CLOSING), AND (E) THE ENVIRONMENTAL CONDITION AND OTHER CONDITION OF THE OIL AND GAS ASSETS AND ANY POTENTIAL LIABILITY ARISING FROM OR RELATED TO THE OIL AND GAS ASSETS.

2.2 <u>Further Assurances</u>. The Parties agree to (a) furnish upon request to each other such further information, (b) execute, acknowledge and deliver to each other such other documents, and (c) do such other acts and things, all as the other Party may reasonably request for the purpose of carrying out the intent of this Assignment and the Transaction Documents; *provided*, that Assignor's covenants and obligations after Closing shall be limited to the extent Assignor has employees and resources available for such purposes and *provided*, *further*, that nothing in this <u>Section 2.2</u> or anywhere else in this Assignment shall prohibit Assignor from ceasing operations or winding up its affairs following the Closing.

ARTICLE 3

MISCELLANEOUS

- 3.1 <u>Amendment</u>. This Assignment may not be amended except by a written agreement executed by all of the Parties.
- 3.2 <u>Parties in Interest; No Third Party Beneficiaries</u>. This Assignment shall inure to the benefit of and be binding upon the Parties and their respective successors and permitted assigns. This Assignment is for the sole benefit of the Parties and their permitted assigns, and except as expressly provided in <u>Section 7.1(e)</u> of the APA, nothing herein, express or implied, is intended to or shall confer upon any other Person any legal or equitable benefit, claim, cause of action, remedy or right of any kind.
- 3.3 <u>Severability</u>. The provisions of this Assignment shall be deemed severable, and the invalidity or unenforceability of any provision shall not affect the validity or enforceability of the other provisions hereof. If any provision of this Assignment, or the application thereof to any Person or any

circumstance, is invalid or unenforceable, (a) a suitable and equitable provision shall be substituted therefor in order to carry out, so far as may be valid and enforceable, the intent and purpose of such invalid or unenforceable provision and (b) the remainder of this Assignment and the application of such provision to other Persons or circumstances shall not be affected by such invalidity or unenforceability.

3.4 <u>Relationship of the Parties</u>. This Assignment does not create and it is not the purpose or intention of the Parties to create any partnership, mining partnership, joint venture, general partnership, or other partnership relationship and none will be inferred, and nothing in this Assignment will be construed to establish a fiduciary relationship between the Parties for any purpose.

3.5 Governing Law; Consent to Jurisdiction and Venue; Jury Trial Waiver.

(a) EXCEPT TO THE EXTENT THE MANDATORY PROVISIONS OF THE BANKRUPTCY CODE APPLY, THIS ASSIGNMENT SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF TEXAS APPLICABLE TO CONTRACTS MADE AND TO BE PERFORMED ENTIRELY IN SUCH STATE WITHOUT REGARD TO PRINCIPLES OF CONFLICTS OR CHOICE OF LAWS OR ANY OTHER LAW THAT WOULD MAKE THE LAWS OF ANY OTHER JURISDICTION OTHER THAN THE STATE OF TEXAS APPLICABLE HERETO.

(b) Without limitation of any Party's right to appeal any Order of the Bankruptcy Court, (i) the Bankruptcy Court shall retain exclusive jurisdiction to enforce the terms of this Assignment and to decide any claims or disputes which may arise or result from, or be connected with, this Assignment, any breach or default hereunder, or the transactions contemplated hereby and (ii) any and all claims relating to the foregoing shall be filed and maintained only in the Bankruptcy Court, and the Parties hereby consent and submit to the exclusive jurisdiction and venue of the Bankruptcy Court and irrevocably waive the defense of an inconvenient forum to the maintenance of any such Action or Proceeding; *provided*, *however*, that, if the Bankruptcy Case is closed, all Actions and Proceedings arising out of or relating to this Assignment shall be heard and determined in a Texas federal court sitting in Houston, Texas, and the Parties hereby irrevocably submit to the exclusive jurisdiction and venue of such courts in any such Action or Proceeding and irrevocably waive the defense of an inconvenient forum to the maintenance of any such Action or Proceeding. The Parties consent to service of process by mail (in accordance with Section 13.4 of the APA) or any other manner permitted by Legal Requirement.

(c) EACH PARTY HEREBY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LEGAL REQUIREMENTS, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN ANY PROCEEDING DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS ASSIGNMENT, THE OTHER TRANSACTION DOCUMENTS OR ANY OTHER AGREEMENT CONTEMPLATED HERETO AND THERETO OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY (WHETHER BASED ON CONTRACT, TORT OR ANY OTHER THEORY). EACH PARTY (A) CERTIFIES THAT NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER AND (B) ACKNOWLEDGES THAT IT AND THE OTHER PARTIES HAVE BEEN INDUCED TO ENTER INTO THIS ASSIGNMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION

3.6 No Strict Construction. Assignee, on the one hand, and Assignor, on the other hand, participated jointly in the negotiation and drafting of this Assignment, and, in the event an ambiguity or question of intent or interpretation arises, this Assignment shall be construed as jointly drafted by Assignee, on the one hand, and Assignor, on the other hand, and no presumption or burden of proof shall arise favoring or disfavoring any Party by virtue of the authorship of any provision of this Assignment.

Without limitation as to the foregoing, no rule of strict construction construing ambiguities against the draftsperson shall be applied against any Person with respect to this Assignment.

- 3.7 <u>Exhibits</u>. The Exhibits attached hereto are hereby made part hereof and incorporated herein by this reference. Unless provided otherwise, all recording references in such Exhibits are to the appropriate records of the counties in which the applicable Oil and Gas Assets are located.
- 3.8 Counterparts. This Assignment and any amendment hereto may be executed in one or more counterparts and by different Parties in separate counterparts, each of which shall be deemed to be an original of this Assignment or such amendment and all of which, when taken together, shall constitute one and the same agreement. Delivery of an executed counterpart of a signature page to this Assignment or any amendment hereto by facsimile or other electronic means (including portable document format sent via email) shall be as effective as delivery of a manually executed counterpart of this Assignment or such amendment, as applicable.

[Signature pages follow.]

The authorized representatives of Assignor and Assignee execute this Assignment on the dates set forth in their respective acknowledgements hereto, but this Assignment shall be effective as of the Effective Time as a conveyance of the Oil and Gas Assets, and for the other purposes contemplated by the APA.

ASSIGNOR:	
O'BENCO IV, LP	
By: Name: Title:	

ASSIGNEE:	
PACHIRA OIL AND GAS, LLC	
By: Name: Title:	

ASSIGNOR'S ACKNOWLEDGMENT

STATE OF TEXAS	§		
	§		
COUNTY OF	§		
DEFORE LEE	1 0		2010
BEFORE ME, on the			, 2019, personally appeared
, known or	r proved to me to be the_		of O'Benco IV, LP, a Delaware
limited partnership, and ackn	owledged to me that he/s	she executed this ins	trument on behalf of said limited
partnership.			
por mersing.			
		37	1.0 1.0 0.0
		Notary Public in	and for the State of Texas

ASSIGNEE'S ACKNOWLEDGMENT

STATE OF []	
\$ COUNTY OF []	
BEFORE ME, on theday of	, 2019, personally appeared
	of Pachira Oil and Gas, LLC, a
Delaware Limited Liability Company, and acknowled	ged to me that he/she executed this instrument on
behalf of said Limited Liability Company.	
	N. (D. 11: ' 1 C. (1 C.
	Notary Public in and for the State of

EXHIBIT A

Assigned Leases

EXHIBIT B

Wells

EXHIBIT C

Assigned Contracts

EXHIBIT D

Surface Interests

SCHEDULE 2.1(b)(xii)

Seismic Surveys

SCHEDULE 2.2(b)

Certain Excluded Cash and Cash Equivalents

SCHEDULE 2.2(g)

Excluded Leases and Excluded Mineral Interests

SCHEDULE 2.2(h)

Excluded Contracts

SCHEDULE 2.2(i)

Excluded Surface Interests

EXHIBIT G

BIDDING PROCEDURES

O'BENCO IV, LP – Amended Bidding Procedures

Set forth below are the bid procedures (the "<u>Bidding Procedures</u>") to be employed by O'BENCO IV, LP (the "<u>Debtor</u>") in connection with the sale of the Debtor's Assets, free and clear of liens and interests, pursuant to and in accordance with the terms and conditions specified therein.

On June 7, 2019, the United States Bankruptcy Court for the Eastern District of Texas (the "Bankruptcy Court") entered an order which, among other things (a) approved these Bidding Procedures, (b) approved the form and manner of the sale notice and bidding procedures notice, \emptyset scheduled a Sale Hearing date, and (d) approved procedures for the assumption and, if necessary, assignment of executory contracts and unexpired leases (the "Bidding Procedures Order"). The Bankruptcy Court amended these procedures pursuant to an order entered on [•].

The Debtor has now selected Valence Operating Company as the "Stalking Horse Bidder" under these Bidding Procedures, and the parties have negotiated the Asset Purchase Agreement dated as of July 23, 2019, by and between O'BENCO IV, LP as Seller, and Valence Operating Company, as Buyer (the "Stalking Horse APA").

1. Key Dates

These Bidding Procedures provide interested parties with the opportunity to complete diligence, to submit competing bids for all or a portion of the Assets, and to participate in an auction to be conducted by the Debtor (the "Auction").

The key dates for the sale process are as follows:

July 22, 2019	Cure Objection Deadline
July 22, 2019	Sale Objection Deadline
August 14, 2019	Bid Deadline
August 21, 2019	Auction
August 23, 2019	Auction Objection Deadline
[TBD]	Sale Hearing

2. Property to be Sold

The Debtor shall offer for sale (the "<u>Sale</u>") all of its oil and gas assets, including but not limited to, interests in oil and gas leases, related inventory and equipment, and various personal property (such assets collectively referred to herein as the "<u>Assets</u>"). The Debtor may consider bids for all or less than all of the Assets.

¹ Capitalized terms used herein and not otherwise defined shall have the meanings set forth in the Debtor's Motion for (A) Entry of an Order (I) Approving Bidding Procedures; (II) Scheduling Bidding Deadline, Auction Date, and Sale Hearing Date; (III) Approving Form and Notice thereof; (B) Entry of an Order After the Sale Hearing (I) Authorizing the Debtor to Sell Its Assets; and (II) Authorizing the Debtor to Assume and Assign Certain Executory Contracts and Unexpired Leases; and (C) Granting Related Relief (Docket No. 3).

3. Participation Requirements

Any person seeking to conduct due diligence on the Assets and participate in the bidding must become a "Diligence Party." As a prerequisite to becoming a Diligence Party, any interested party: (a) must deliver an executed confidentiality agreement in form and substance reasonably acceptable to the Debtor in consultation with the Agent;² and (b) must be able, as determined by the Debtor in consultation with its financial advisor and the Agent, to consummate a transaction based upon a Qualifying Bid (as defined below). The Debtor, in consultation with the Agent, reserves the right (i) at any time to require any Diligence Party to provide additional evidence of its ability to consummate a transaction based upon a Qualifying Bid (as defined below), and (ii) to exclude any such Diligence Party from participating further in the auction process as a result of its inability to satisfy such further requirements. The confidentiality agreements executed by Diligence Parties shall inure to the benefit of the Successful Bidder following the closing of the Sale transaction.

4. Due Diligence

The Debtor, with the assistance if its financial advisor, Houlihan Lokey ("HL"), shall afford any Diligence Party the time and opportunity to conduct reasonable due diligence. This will include, without limitation, an opportunity for Diligence Parties to inspect the Assets and receive copies of, among other things, all engineering reports regarding reserves, all production reports, all deeds and leases for oil and gas rights and all title opinions and information. The due diligence period shall extend through and include the Bid Deadline (as defined below). The Debtor and its representatives shall not be obligated to furnish any due diligence information after the Bid Deadline. Diligence inquiries should be made to:

Houlihan Lokey
1001 Fannin Street, Suite 4650
Houston, TX 77002
Attn: Jerry Eumont
JEumont@HL.com
Attn: Joshua Eaves
JEaves@HL.com
(832) 319-5119

For the avoidance of doubt, the Agent and its advisors shall be provided prompt access to all executed Confidentiality Agreements and due diligence materials, including without limitation any management presentations, on-site inspections, and any other information provided to any Diligence Party that was not previously made available to the Agent. The Debtor and HL shall consult with the Agent and its advisors regarding materials produced in connection with the Sale, including without limitation the form of confidentiality agreement and any marketing materials or information memoranda or "teasers." In addition, HL shall (a) hold weekly Sale update conferences with the Agent and its financial advisors and (b) provide weekly reporting to the

² For the avoidance of doubt, parties who have already executed a confidentiality agreement with the Debtor shall not be required to execute a new confidentiality agreement provided that the existing confidentiality agreement is in form and substance acceptable to the Debtor in consultation with the Agent.

Agent's financial advisors, including without limitation (i) a written log containing a summary of HL's correspondence (phone, email, etc.) with potential interested parties, including Diligence Parties, (ii) a written summary of marketing activity including parties contacted and current status (e.g., parties under Confidentiality Agreements, indications of interest/letters of intent received, parties scheduled for management/advisor presentations, site visits, etc.), and (iii) a written log of diligence activities and status of fulfilling requests.

5. Bid Requirements

The Debtor encourages Qualified Bidders that wish to bid on less than all the Assets to submit a bid(s) on the relevant assets. The Debtor, in consultation with the Agent, may make any modifications to the Bidding Procedures, including the Bid requirements described immediately below, as necessary to accommodate a piecemeal bid they determine is in the best interests of the estate.

To be deemed a "<u>Qualifying Bid</u>," a bid must be received from a Diligence Party by a date no later than the Bid Deadline (as defined below) that satisfies each of the following conditions:

- i. <u>Purchase Price</u>. Each bid must identify the cash consideration to be paid for relevant Assets ("Purchase Price");
- ii. Executed Agreement. Each bid must be accompanied by (a) a duly executed asset purchase agreement (the "Modified APA"); and (B) a marked Modified APA reflecting any variations from the Stalking Horse APA executed by the Stalking Horse Bidder;
- iii. Irrevocable. Each bid must expressly provide that (a) the offer reflected in the bid is irrevocable until the conclusion of the Auction, provided that if such party is designated as the Successful Bidder or the Back-up Bidder (each, as defined below) then such offer shall remain open and irrevocable until the closing of the Sale or the Outside Back-up Date as described below and (b) if such party is selected as the Successful Bidder, then such party will take all commercially reasonable steps to consummate the transaction set forth in the Modified APA within 15 days of entry of the Sale Order and satisfaction of the closing conditions (if any) set forth in the Modified APA;
- iv. <u>Bound to Bid Procedure Terms</u>. Each bid must state that the bidder agrees to be bound by the terms of the Bidding Procedures Order and the Bidding Procedures;
- v. <u>Proof of Financial Ability to Perform</u>. Each bid must contain such financial and other information that allows the Debtor and HL to make a reasonable determination, in consultation with the Agent, as to the Qualifying Bidder's financial and other capabilities to consummate the transactions contemplated by the Modified APA, including, without limitation, such financial and other information setting forth adequate assurance of future performance. Without limiting the foregoing, such information must

- include current financial statements or similar financial information certified to be true and correct as of the date thereof, proof of financing commitments, if needed to close the transaction, contact information for verification of such information, including any financing sources, and any other information reasonably requested by the Debtor necessary to demonstrate that such party has the ability to close the transaction;
- vi. <u>Designation of Contracts and Leases</u>. Each bid must identify with particularity each and every executory contract and unexpired lease, the assumption and, as applicable, assignment of which is a condition to closing; *provided* that the Modified APA may allow for the Qualifying Bidder to identify additional contracts prior to the closing or at such other date as agreed by the Debtor in consultation with the Agent;
- vii. <u>Regulatory and Third Party Approval</u>. Each bid must identify each regulatory and third-party approval required for the bidder to consummate the Sale, if any, and the time period within which the bidder expects to receive such regulatory and third-party approvals;
- viii. <u>Disclosure of Identity of Bidder</u>. Each bid must fully disclose the identity of each entity that will be bidding in the Auction or otherwise participating in connection with such bid, and the complete terms of any such participation;
 - ix. Contingencies. Each bid shall (i) not contain any financing contingencies of any kind; (ii) not contain any conditions to closing other than the conditions precedent identified in the Modified APA; (iii) not contain any due diligence contingency; and (iv) contain evidence that the bidder has received debt and/or equity funding commitments or has financial resources readily available sufficient in the aggregate to consummate the Sale, which evidence is reasonably satisfactory to the Debtor in consultation with the Agent;
 - x. <u>Authorizations</u>. Each bid must include written evidence of authorization and approval from the bidder's board of directors (or comparable governing body) with respect to the submission, execution, delivery and performance of the Modified APA that is acceptable to the Debtor in consultation with the Agent;
- xi. No Break-Up Fees. Other than the Stalking Horse APA, bid may not request or entitle the bidder to any break-up fee, expense reimbursement, termination fee or similar type of payment or bid protection; and
- xii. <u>Good Faith Deposit</u>. Each bid must be accompanied by a purchase deposit equal to or greater than 10% of the total Purchase Price.

xiii. Minimum Bid. Each bid must result in a value to the Debtor's estate which, in the Debtor's reasonable judgment, is more than the aggregate of the value of the sum of (a) the Break-Up Fee and the Expense Reimbursement (both as defined in the Stalking Horse APA), (b) the cash purchase price to be paid under the Stalking Horse APA (\$25,500,000), and (c) \$250,000, resulting in a minimum aggregate amount of \$26,514,000.

A competing bid meeting the above requirements shall constitute a "Qualifying Bid" and the party submitting such bid shall constitute a "Qualifying Bidder." The Debtor, in consultation with the Agent, shall make a determination regarding whether a bid is a Qualifying Bid and shall notify bidders whether their bids have been determined to be qualified by no later than 4:00 p.m. (Prevailing Central time) on August 15, 2019. For the avoidance of doubt, the Agent (or its designee), on behalf of the Secured Lenders, shall be deemed a Qualifying Bidder for purposes of these Bidding Procedures, and any credit bid complying with the provisions set forth in the section entitled "Credit Bidding" herein shall be deemed a Qualifying Bid for all purposes in connection with the Bidding Procedures, the Auction and the Sale. Furthermore, the Stalking Horse Bidder is deemed a Qualified Bidder and the Stalking Horse APA constitutes a Qualifying Bid for all purposes, any other provision of these Bidding Procedures notwithstanding.

The Debtor, in consultation with the Agent, reserves the right (i) at any time to require any bidder previously determined to be a Qualified Bidder (other than the Stalking Horse Bidder) to provide additional evidence of its ability to consummate a transaction based upon a Qualifying Bid in order to remain a Qualified Bidder, and (ii) to exclude any bidder from participating further in the auction process as a result of its inability to satisfy such further requirements to remain a Qualified Bidder.

Persons who collectively are referred to as a "Qualifying Bidder" need not be affiliates and need not act in concert with one another, and the Debtor, in consultation with the Agent, may aggregate separate bids from unaffiliated persons to create one "bid" from a "Qualifying Bidder"; provided, however, that all bidders shall remain subject to the provisions of Bankruptcy Code section 363(n) regarding collusive bidding. Nothing contained herein shall prevent the consideration of bids for less than all Assets, and the Debtor, with the consent of the Agent, may make any modifications to the Bidding Procedures as necessary to accommodate such a bid as they determine is in the best interests of the estate.

6.No Qualifying Bids

If no Qualifying Bids other than the Qualifying Bid submitted by the Stalking Horse Bidder are submitted by the Bid Deadline, the Debtor shall not hold an Auction and instead shall request at the Sale Hearing that the Bankruptcy Court approve the Stalking Horse APA with the Stalking Horse Bidder and declare the Stalking Horse Bidder and the Stalking Horse APA to be the Prevailing Bid.

7.Bid Deadline

A Qualifying Bidder that desires to make a bid shall deliver a written or electronic copy of its bid to: (i) counsel to the Debtor, Bracewell LLP, 711 Louisiana Street, Suite 2300, Houston,

Texas 77002 (Attn: William A. (Trey) Wood III, email: Trey.Wood@bracewell.com and Jason G. Cohen, email: Jason.Cohen@bracewell.com) and (ii) counsel to the Agent, Sidley Austin LLP, 1000 Louisiana Street, Suite 5900, Houston, TX 77002 (Attn: Brian Minyard, email: bminyard@sidley.com and Dennis Twomey, email: dtwomey@sidley.com) (collectively, the "Notice Parties") so as to be received by a date no later than 4:00 p.m. (Prevailing Central Time) on August 14, 2019 (the "Bid Deadline").

The Bid Deadline may be changed without further order of the Bankruptcy Court upon the agreement of the Debtor and the Agent.

8. Evaluation of Qualifying Bids

At least 48 hours prior to the Auction, the Debtor shall determine, in its reasonable judgment after consultation with the Debtor's financial and legal advisors and the Agent (unless the Agent has given a Credit Bid Notification (as defined below)), which Qualifying Bid (including aggregated Qualifying Bids if appropriated as provided herein) is or are likely to result in the highest and best value to the Debtor's estate, and shall provide a copy of that Qualifying Bid or Qualifying Bids to all Qualifying Bidders prior to the beginning of the Auction. The Debtor may discuss or clarify the terms of the Qualifying Bids with the Qualifying Bidders prior to such date, but no modifications may be made to the Qualifying Bids after the Bid Deadline without the Debtor's prior written consent, in consultation with the Agent.

9. Credit Bidding

Only the Agent (or its designee), on behalf of the Secured Lenders, and the Stalking Horse Bidder, shall be permitted to submit a credit bid at the Auction. The Agent will notify the Debtor of its intention to potentially credit bid no later than the Bid Deadline (the "Credit Bid Notification"). To the extent the Agent indicates in the Credit Bid Notification that it does intend to potentially credit bid at the Auction, it shall not receive the consent/consultation rights described above in the "Evaluation of Qualifying Bids" section or below in the "Auction" section; provided, however, that in the event the Agent subsequently notifies the Debtor that the Agent is no longer credit bidding, then the Agent shall again receive the consent/consultation rights described herein.

The Agent (or its designee) shall be permitted to credit bid up to the total allowed amount of its secured claim outstanding at the time of the Auction under that certain Credit Agreement dated January 10, 2017, as amended from time to time, and related documents (the "Secured Lender Claim"). In the event the Agent (or its designee) submits a credit bid at the Auction, it shall provide the Debtor with a Modified APA at the Auction, but will not be required to meet any of the other requirements set forth in the "Bid Requirements" section (including, but not limited to, the

requirement to submit a deposit, and no portion of any credit bid shall constitute or be deemed a deposit for any purpose).

The Stalking Horse Bidder shall be permitted to credit bid some or all of its Break-Up Fee and Expense Reimbursement under the Stalking Horse APA at the time of the Auction.

10. Insider Participation

To the extent any insider of the Debtor or any of its affiliates, or any person or entity controlled by any insider of the Debtor or its affiliates, chooses to submit a bid at the Bid Deadline, and such bid is deemed a Qualifying Bid, such insider and any person associated with such insider shall be expressly prohibited from participating in any manner in the "Evaluation of Qualifying Bids" and "Auction" (other than in his or her capacity as a Qualified Bidder) sections as described herein.

11. Auction

In the event that the Debtor timely receives two or more Qualifying Bids, the Debtor shall conduct an auction (the "Auction") starting at 10:00 a.m. on August 21, 2019 (the "Auction Date") at the law offices of Bracewell LLP, 711 Louisiana Street, Suite 2300, Houston, Texas 77002. In the event that only one Qualifying Bid is timely received and the Agent has elected not to credit bid, the Debtor shall declare that bid to be the Successful Bid and the Auction shall be canceled. The Auction shall be governed by the following procedures:

- i.Only Qualifying Bidders (including the Agent or its designee to the extent it makes a credit bid) who have made a Qualifying Bid shall be entitled to make any subsequent bids at the Auction;
- ii. Each Qualifying Bidder shall be required to confirm that it has not engaged in any collusion with respect to the bidding or the sale;
- iii. Qualifying Bidders shall appear in person at the Auction, or through a duly authorized representative;
- iv. Bidding shall commence at the amount of the highest Qualifying Bid as determined pursuant to Paragraph (g) above;
- v. Each Qualifying Bidder may then submit successive bids in increments of at least \$250,000.00 higher than the bid at which the Auction commenced and then continue in minimum increments of at least \$250,000.00 higher than the previous bid; provided that the Debtor, in consultation with the Agent, shall retain the right to modify the bid increment requirements at the Auction;
- vi. The Auction will be conducted openly and shall be transcribed or recorded, and the Qualifying Bidders will be informed of the terms of the previous bid;

- vii. Qualifying Bidders shall have the right to submit additional bids and make additional modifications to the Modified APA at the Auction, provided that any such modifications to the Modified APA on an aggregate basis and viewed in whole, shall not be less favorable to the Debtor's estate, as determined by the Debtor in consultation with the Agent, than the terms of the highest and best Qualifying Bid at that time;
- viii. The Auction shall continue until there is only one bid that the Debtor determines in consultation with the Agent, subject to Court approval, is the highest or best from among the Qualifying Bids submitted at the Auction (the "Successful Bid"). In making this decision, the Debtor shall consider, without limitation, the amount of the Purchase Price or other amounts to be paid to the Debtor, the form of consideration being offered, the likelihood of the bidder's ability to close a transaction and the timing thereof, the number, type and nature of any changes to the Form APA requested by each bidder, and the net benefit to the Debtor's estate. The bidder or bidders submitting such Successful Bid shall become the "Successful Bidder," and shall have such rights and responsibilities of the purchaser, as set forth in the applicable Modified APA;
 - ix. The Debtor will determine in consultation with the Agent which Qualifying Bid, if any, is the next highest and/or best Qualifying Bid to the Successful Bid, and will designate such Qualifying Bid as a back-up bid ("Back-up Bid") to be consummated in the event the Successful Bid fails to consummate the contemplated sale transaction. There may be more than one Back-up Bid if aggregated Qualifying Bids are determined to be Back-up Bids. A Qualifying Bidder that submitted a Qualifying Bid that is designated as a Back-up Bid is a "Back-up Bidder." Each Back-up Bid shall remain open and binding until the Outside Back-up Date;
 - x.At the Sale Hearing, the Debtor shall present the Successful Bid and Back-up Bid to the Court for approval. The Debtor's presentation of the announced Successful Bid to the Court for approval does not constitute the Debtor's acceptance of such Successful Bid. The Debtor shall have accepted such Successful Bid only when approved by the Court;
- xi. Within fifteen (15) calendar days after the entry of the Sale Order, each Successful Bidder and the Debtor (i) shall complete and execute all agreements, contracts, instruments and other documents evidencing and containing the terms and conditions upon which

such Successful Bid was made, and (ii) shall close the sale transaction; and

xii. The Debtor in consultation with the Agent may adopt such other rules for the Auction (including rules that may depart from those set forth herein) that they anticipate will result in the highest or otherwise best value for the Debtor's estate and that are not inconsistent with any applicable Bankruptcy Court order, provided that any changed or additional rules of the Auction are not materially inconsistent with the Bidding Procedures and are communicated to all participants at or prior to the Auction.

12. Sale Hearing

The Successful Bid will be subject to approval by the Bankruptcy Court. The Debtor requests that the hearing to approve the Successful Bid (the "Sale Hearing") take place on [TBD] __.m. (Prevailing Central Time), unless adjourned or continued by the Debtor. At such time, the Debtor will seek the entry of an order of the Bankruptcy Court approving and authorizing the Sale to the Successful Bidder on the terms and conditions of the Successful Bid. The Successful Bidder shall appear at the Sale Hearing and be prepared to testify in support of the Successful Bid and the Successful Bidder's ability to close in a timely manner and provide adequate assurance of its future performance under any and all contracts and leases to be assumed and/or assigned as part of the proposed Sale.

13. Return of Deposits

Except as provided herein, all deposits shall be returned to each bidder not selected by the Debtor as a Successful Bidder no later than five (5) business days following the conclusion of the Auction, except for any Back-up Bidder (as defined above) whose deposit will be returned by no later than the third (3rd) business day after the earlier to occur of: (a) the closing of the sale transaction with the relevant Successful Bidder or (b) the Outside Back-up Date.

14. Failure to Consummate Purchase

The Back-up Bidder shall keep its bid open and irrevocable until the earlier of 5:00 p.m. (Prevailing Central time) on (i) September 30, 2019 (the "Outside Back-up Date") or (ii) the closing of the Sale transaction with the Successful Bidder. Following the Sale Hearing, if any Successful Bidder fails to consummate an approved Sale because of (a) the failure of a condition precedent beyond the control of either the Debtor or the Successful Bidder or (b) a breach or failure to perform on the part of such Successful Bidder, the Back-up Bidder will be deemed to be the new Successful Bidder, and the Debtor will be authorized to consummate the Sale with the Back-up Bidder without further order of the Bankruptcy Court. In the case of (b) above, the defaulting Successful Bidder's deposit shall be forfeited to the Debtor as the Debtor's sole remedy with respect to the defaulting Successful Bidder.

15. Reservation of Rights; Deadline Extension

The Debtor reserves its rights, in the exercise of its fiduciary obligations and, with the consent of the <u>Agent</u> (not to be unreasonably withheld), to modify these Bidding Procedures or impose at, or prior to, the Auction, additional customary terms and conditions on the Sale of the Assets, including, without limitation, extending the deadlines set forth in the Auction procedures, modifying bidding increments, adjourning the Auction and/or adjourning the Sale Hearing in open court without further notice, withdrawing from the Auction the Assets at any time prior to or during the Auction or canceling the Auction in the Debtor's business judgment.

16. Emergency Hearing

The Debtor and the Agent agree that in the event of any dispute regarding the Sale process outlined herein, (i) either party may seek an emergency hearing requesting that the Court resolve such dispute and (ii) the other party shall consent to such request for an emergency hearing.

17. Sale As Is/Where Is

The Assets sold pursuant to the Bidding Procedures shall be sold free and clear of all liens claims and encumbrances as permitted by section 363(f) of the Bankruptcy Code other than any assumed liabilities as set forth in the applicable APA and conveyed at closing in their then-present condition, "AS IS, WHERE IS, WITH ALL FAULTS, AND WITHOUT ANY WARRANTY WHATSOEVER, EXPRESS OR IMPLIED", except to the extent set forth in the definitive agreement for the Successful Bid, if applicable; *provided* that any such sale shall be subject to the rights of any DIP Lender or Prepetition Lender or other party in interest to object to the extent otherwise set forth herein.

EXHIBIT H

FORM OF MINERAL DEED

EXHIBIT H

FORM OF MINERAL DEED

MINERAL DEED

THIS MINERAL DEED (this "<u>Deed</u>"), dated as of [●], 2019, but effective as of 12:01 a.m., Central Standard Time, on June 1, 2019 (the "<u>Effective Time</u>"), is by and between O'Benco IV, LP, a Delaware limited partnership ("<u>Grantor</u>"), and Pachira Oil and Gas, LLC, a Delaware Limited Liability Company ("<u>Grantee</u>"). Grantor and Grantee are sometimes referred to collectively herein as the "<u>Parties</u>" and individually as a "<u>Party</u>".

WHEREAS, Grantor is a debtor and debtor in possession under title 11 of the United States Code, 11 U.S.C. §§ 101, et seq. (the "Bankruptcy Code"), and filed a voluntary petition for relief under chapter 11 of the Bankruptcy Code on June 3, 2019 (the "Petition Date") in the United States Bankruptcy Court for the Eastern District of Texas (the "Bankruptcy Court"), where Grantor's bankruptcy case is being administered under Case No. 19-60384 (the "Bankruptcy Case");

WHEREAS, Grantor and Grantee entered into that certain Asset Purchase Agreement dated [•], 2019 (the "APA"), providing for the sale by Grantor of, and the purchase by Grantee of, the Oil and Gas Assets (including the Mineral Interests), and the terms and provisions of the APA shall not be deemed to have merged herein. In the event that any provision of this Deed is construed to conflict with any provision of the APA, the provisions of the APA shall be deemed controlling to the extent of such conflict;

WHEREAS, references herein to terms which are defined separately herein or in APA shall have the meaning prescribed to such term herein or in the APA, as applicable, even if the term (as printed in this Deed) is printed in ALL CAPITALIZED text, in each case, unless the context expressly requires otherwise;

WHEREAS, on [●], 2019, the Bankruptcy Court entered the Sale Order authorizing the Grantor to sell the Oil and Gas Assets (including the Mineral Interests) to Grantee pursuant to the terms and conditions of the APA; and

NOW, THEREFORE, in consideration of the premises, the mutual promises herein made, and other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the Parties agree as follows:

ARTICLE 1

CONVEYANCE

1.1 Granting Clause. Grantor hereby GRANTS, SELLS, TRANSFERS, ASSIGNS, CONVEYS AND DELIVERS to Grantee, free and clear of all Encumbrances and Liabilities (other than Permitted Encumbrances and Assumed Liabilities), all of Grantor's right, title and interest in and to all mineral fee interests, non-leasehold mineral rights, operating rights, and mineral servitudes in which Grantor owns an interest, including bonus, executive rights, royalty interests, overriding royalty interests, net profit interests, production payments, and without limiting the foregoing, other rights of whatever nature, whether legal or equitable, whether vested, contingent or reversionary, in oil, gas and other minerals in, on, under or that may be produced from the lands described in or covered by the instruments creating the Oil and Gas Assets ("Mineral Interests"), including the Mineral Interests described on Exhibit A attached hereto, but excluding the excluded Mineral Interests on Exhibit B attached hereto. This Deed is

¹ NTD: Exhibit B should disclose the Excluded Mineral Interests that are listed on Schedule 2.2(g) to the APA.

made subject to (i) any rights now existing to any lessees or assigns under any valid and subsisting oil and gas lease heretofore executed and now of legal record in the jurisdictions in which the Mineral Interests are located; it being understood and agreed that Grantee shall have the right to receive, and enjoy all bonuses, rents, royalties and other benefits which may accrue under the terms of said lease(s) insofar as it covers any Mineral Interests, precisely as if the Grantee herein had been the lessor under said lease(s), and (ii) any and all restrictions, covenants, conditions, easements, prescriptive rights, reservations, rights-of-way, and all other matters of record in the county in which the Mineral Interests are located. Grantors to convey, and this Deed hereby conveys, to Grantee the Mineral Interests, regardless of errors in description, any incorrect or misspelled names, or any transcribed or incorrect recording references.

ARTICLE 2

ADDITIONAL MATTERS

2.1 Disclaimers. NOTWITHSTANDING ANYTHING TO THE CONTRARY CONTAINED IN THIS DEED, EXCEPT AS AND TO THE EXTENT EXPRESSLY SET FORTH IN THIS DEED AND IN THE TRANSACTION DOCUMENTS, GRANTOR MAKES NO REPRESENTATIONS OR WARRANTIES WHATSOEVER, AND DISCLAIMS ALL LIABILITY AND REPRESENTATION, WARRANTY, RESPONSIBILITY FOR ANY STATEMENT INFORMATION MADE OR COMMUNICATED (ORALLY OR IN WRITING) TO GRANTEE (INCLUDING ANY OPINION, INFORMATION, OR ADVICE THAT MAY HAVE BEEN PROVIDED TO GRANTEE BY ANY AFFILIATE OR REPRESENTATIVE OF GRANTOR OR BY ANY INVESTMENT BANK OR INVESTMENT BANKING FIRM, ANY PETROLEUM ENGINEER OR ENGINEERING FIRM, GRANTOR'S COUNSEL, OR ANY OTHER AGENT, CONSULTANT, OR REPRESENTATIVE OF GRANTOR). WITHOUT LIMITING THE GENERALITY OF THE FOREGOING, EXCEPT AS AND TO THE EXTENT EXPRESSLY SET FORTH IN THIS DEED AND IN THE TRANSACTION DOCUMENTS, GRANTOR EXPRESSLY DISCLAIMS AND NEGATES ANY REPRESENTATION OR WARRANTY, EXPRESS, IMPLIED, AT COMMON LAW, BY STATUTE, OR OTHERWISE, RELATING TO (A) THE TITLE TO ANY OF THE MINERAL INTERESTS, (B) THE CONDITION OF THE MINERAL INTERESTS (INCLUDING ANY IMPLIED OR EXPRESS WARRANTY OF MERCHANTABILITY, FITNESS FOR A PARTICULAR PURPOSE, OR CONFORMITY TO MODELS OR SAMPLES OF MATERIALS), IT BEING DISTINCTLY UNDERSTOOD THAT THE MINERAL INTERESTS ARE BEING SOLD "AS IS," "WHERE IS," AND "WITH ALL FAULTS AS TO ALL MATTERS," (C) ANY INFRINGEMENT BY GRANTOR OF ANY PATENT OR PROPRIETARY RIGHT OF ANY THIRD PARTY, (D) ANY INFORMATION, DATA, OR OTHER MATERIALS (WHETHER WRITTEN OR ORAL) FURNISHED TO GRANTEE BY OR ON BEHALF OF GRANTOR (INCLUDING WITHOUT LIMITATION, IN RESPECT OF ANY SEISMIC DATA, THE EXISTENCE OR EXTENT OF HYDROCARBONS OR THE MINERAL RESERVES, THE RECOVERABILITY OF SUCH RESERVES, ANY PRODUCT PRICING ASSUMPTIONS, AND THE ABILITY TO SELL HYDROCARBON PRODUCTION AFTER THE CLOSING), AND (E) THE ENVIRONMENTAL CONDITION AND OTHER CONDITION OF THE MINERAL INTERESTS AND ANY POTENTIAL LIABILITY ARISING FROM OR RELATED TO THE MINERAL INTERESTS.

2.2 <u>Further Assurances</u>. The Parties agree to (a) furnish upon request to each other such further information, (b) execute, acknowledge and deliver to each other such other documents, and (c) do such other acts and things, all as the other Party may reasonably request for the purpose of carrying out the intent of this Deed and the Transaction Documents; *provided*, that Grantor's covenants and obligations after Closing shall be limited to the extent Grantor has employees and resources available for such purposes

and *provided*, *further*, that nothing in this <u>Section 2.2</u> or anywhere else in this Deed shall prohibit Grantor from ceasing operations or winding up its affairs following the Closing.

ARTICLE 3

MISCELLANEOUS

- 3.1 <u>Amendment</u>. This Deed may not be amended except by a written agreement executed by all of the Parties.
- 3.2 <u>Parties in Interest; No Third Party Beneficiaries</u>. This Deed shall inure to the benefit of and be binding upon the Parties and their respective successors and permitted assigns. This Deed is for the sole benefit of the Parties and their permitted assigns, and except as expressly provided in <u>Section 7.1(e)</u> of the APA, nothing herein, express or implied, is intended to or shall confer upon any other Person any legal or equitable benefit, claim, cause of action, remedy or right of any kind.
- 3.3 <u>Relationship of the Parties</u>. This Deed does not create and it is not the purpose or intention of the Parties to create any partnership, mining partnership, joint venture, general partnership, or other partnership relationship and none will be inferred, and nothing in this Deed will be construed to establish a fiduciary relationship between the Parties for any purpose.
- 3.4 Severability. The provisions of this Deed shall be deemed severable, and the invalidity or unenforceability of any provision shall not affect the validity or enforceability of the other provisions hereof. If any provision of this Deed, or the application thereof to any Person or any circumstance, is invalid or unenforceable, (a) a suitable and equitable provision shall be substituted therefor in order to carry out, so far as may be valid and enforceable, the intent and purpose of such invalid or unenforceable provision and (b) the remainder of this Deed and the application of such provision to other Persons or circumstances shall not be affected by such invalidity or unenforceability.
 - 3.5 Governing Law; Consent to Jurisdiction and Venue; Jury Trial Waiver.
- (a) EXCEPT TO THE EXTENT THE MANDATORY PROVISIONS OF THE BANKRUPTCY CODE APPLY, THIS DEED SHALL BE GOVERNED BY, AND CONSTRUED IN ACCORDANCE WITH, THE LAWS OF THE STATE OF TEXAS APPLICABLE TO CONTRACTS MADE AND TO BE PERFORMED ENTIRELY IN SUCH STATE WITHOUT REGARD TO PRINCIPLES OF CONFLICTS OR CHOICE OF LAWS OR ANY OTHER LAW THAT WOULD MAKE THE LAWS OF ANY OTHER JURISDICTION OTHER THAN THE STATE OF TEXAS APPLICABLE HERETO.
- (b) Without limitation of any Party's right to appeal any Order of the Bankruptcy Court, (i) the Bankruptcy Court shall retain exclusive jurisdiction to enforce the terms of this Deed and to decide any claims or disputes which may arise or result from, or be connected with, this Deed, any breach or default hereunder, or the transactions contemplated hereby and (ii) any and all claims relating to the foregoing shall be filed and maintained only in the Bankruptcy Court, and the Parties hereby consent and submit to the exclusive jurisdiction and venue of the Bankruptcy Court and irrevocably waive the defense of an inconvenient forum to the maintenance of any such Action or Proceeding; *provided*, *however*, that, if the Bankruptcy Case is closed, all Actions and Proceedings arising out of or relating to this Deed shall be heard and determined in a Texas federal court sitting in Houston, Texas, and the Parties hereby irrevocably submit to the exclusive jurisdiction and venue of such courts in any such Action or Proceeding and irrevocably waive the defense of an inconvenient forum to the maintenance of any such Action or Proceeding. The Parties consent to service of process by mail (in accordance with Section 13.4 of the APA) or any other manner permitted by Legal Requirement.

- (c) EACH PARTY HEREBY WAIVES, TO THE FULLEST EXTENT PERMITTED BY APPLICABLE LEGAL REQUIREMENTS, ANY RIGHT IT MAY HAVE TO A TRIAL BY JURY IN ANY PROCEEDING DIRECTLY OR INDIRECTLY ARISING OUT OF OR RELATING TO THIS DEED, THE OTHER TRANSACTION DOCUMENTS OR ANY OTHER AGREEMENT CONTEMPLATED HERETO AND THERETO OR THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY (WHETHER BASED ON CONTRACT, TORT OR ANY OTHER THEORY). EACH PARTY (A) CERTIFIES THAT NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER AND (B) ACKNOWLEDGES THAT IT AND THE OTHER PARTIES HAVE BEEN INDUCED TO ENTER INTO THIS DEED BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION
- 3.6 No Strict Construction. Grantee, on the one hand, and Grantor, on the other hand, participated jointly in the negotiation and drafting of this Deed, and, in the event an ambiguity or question of intent or interpretation arises, this Deed shall be construed as jointly drafted by Grantee, on the one hand, and Grantor, on the other hand, and no presumption or burden of proof shall arise favoring or disfavoring any Party by virtue of the authorship of any provision of this Deed. Without limitation as to the foregoing, no rule of strict construction construing ambiguities against the draftsperson shall be applied against any Person with respect to this Deed.
- 3.7 <u>Exhibits</u>. The Exhibits attached hereto are hereby made part hereof and incorporated herein by this reference. Unless provided otherwise, all recording references in such Exhibits are to the appropriate records of the counties in which the applicable Mineral Interests are located.
- 3.8 <u>Counterparts</u>. This Deed and any amendment hereto may be executed in one or more counterparts and by different Parties in separate counterparts, each of which shall be deemed to be an original of this Deed or such amendment and all of which, when taken together, shall constitute one and the same agreement. Delivery of an executed counterpart of a signature page to this Deed or any amendment hereto by facsimile or other electronic means (including portable document format sent via email) shall be as effective as delivery of a manually executed counterpart of this Deed or such amendment, as applicable.

[Signature pages follow.]

The authorized representatives of Grantor and Grantee execute this Deed on the date first set forth above in their respective acknowledgements hereto, but this Deed shall be effective as of the Effective Time as a conveyance of the Mineral Interests, and for the other purposes contemplated by the APA.

GRANTOR:	
O'BENCO IV, LP	
By:Name: Title:	

GRANTEE:
PACHIRA OIL AND GAS, LLC
By:Name:
Title:

GRANTOR'S ACKNOWLEDGMENT

STATE OF TEXAS	§		
	§		
COUNTY OF	§		
BEFORE ME, on the			, 20, personally appeared of O'Benco IV, LP, a Delaware
			astrument on behalf of said limited
		Notary Public i	n and for the State of Texas

GRANTEE'S ACKNOWLEDGMENT

STATE OF §	
<u> </u>	
COUNTY OF §	
BEFORE ME, on theday of	, 20_, personally appeared
, known or proved to me to be the	
	rledged to me that he/she executed this instrument on
behalf of said Limited Liability Company.	
	Notary Public in and for the State of

Exhibit A to Mineral Deed

Mineral Interests

Exhibit B to Mineral Deed

Excluded Mineral Interests

EXHIBIT I

Equipment

Account Description	SubAccount	Well Name	Description
PROCESS EQUIP-SEP,MTR,TNK,DEHY	14225	GRIFFIN #5	KELLY PRODUCTION SERVICE - SEPARATOR FOR GRIFFIN #5 1/24
PROCESS EQUIP-SEP,MTR,TNK,DEHY	14225	GRIFFIN#5	GARDNER CONSULTANTS, INC - METER RUN & METERS FOR GRIFFIN #5 1/31
PROCESS EQUIP-SEP,MTR,TNK,DEHY	14225	GRIFFIN#5	CHALLENGER PROCESS SYSTEM - 2 - 300 BARRELS STEEL TANKS, 1 - COATED, WALKWAY, STAIRS, AND FIREWALL CROSSOVER 8/14
PROCESS EQUIP-SEP,MTR,TNK,DEHY	14225	GRIFFIN#5	WEATHERFORD ENGINEERED CHEMISTRY - SOLAR CHEMICAL PUMP FOR H2S CHEMICAL 7/11
PROCESS EQUIP-SEP,MTR,TNK,DEHY	14226	GRIFFIN#6	KELLY PRODUCTION SERVICE - 24" X 10", 1440# W.P., 3-PHASE VERTICAL SEPARATOR & VOLUME POT 8/28
PROCESS EQUIP-SEP,MTR,TNK,DEHY	14226	GRIFFIN #6	GARDNER CONSULTANTS, INC - 3" SKID MOUNTED METER RUN, EFM, (3 PEN) CHART RECORDER, METER HOUSE & SOLAR PANEL 9/7
PROCESS EQUIP-SEP,MTR,TNK,DEHY	14226	GRIFFIN#6	CHALLENGER PROCESS SYSTEM - (1) 300 BBL STEEL TANK, (1) 300 BBL STEEL TANK COATED, STAIRS, WALKWAY & FIREWALL CROSSOVER 9/10
PROCESS EQUIP-SEP,MTR,TNK,DEHY	14281	LITTLE #1	CHALLENGER PROCESS SYSTEM - (2) 300 BBLS STEEL TANKS (1) COATED, STAIRWAY, WALKWAY, FIREWALL CROSSOVER 4/2
PROCESS EQUIP-SEP,MTR,TNK,DEHY	14281	LITTLE #1	KELLY PRODUCTION SERVICE - 24" X 10", 1440# W.P. 3-PHASE VERTICAL SEPARATOR W/VOLUME POT 4/29
PROCESS EQUIP-SEP,MTR,TNK,DEHY	14281	LITTLE #1	GARDNER CONSULTANTS, INC - (1) TOTAL FLOW, (1) 3 PEN RECORDER, 3" METER RUN, SOLAR PANEL 5/14
PROCESS EQUIP-SEP,MTR,TNK,DEHY	14282	LITTLE #2	CHALLENGER PROCESS SYSTEM - (1) 300 BBL COATED STEEL TANK 4/2
PROCESS EQUIP-SEP,MTR,TNK,DEHY	14282	LITTLE #2	KELLY PRODUCTION SERVICE - 24" X 10" 1440# PSI, 3-PHASE VERTICAL SEPARATOR & VOLUME POT 5/29
PROCESS EQUIP-SEP,MTR,TNK,DEHY	14282	LITTLE #2	GARDNER CONSULTANTS, INC - 3" SKID MOUNTED METER RUN, EFM, (3 PEN) CHART RECORDING, METER HOUSE & SOLAR PANEL 9/15
PROCESS EQUIP-SEP,MTR,TNK,DEHY	14282	LITTLE #2	LONG INDUSTRIES, INC (2) 300 BBL STEEL TANKS & STAIRS & WALKWAY & FIREWALL CROSS OVER 4/29
PROCESS EQUIP-SEP,MTR,TNK,DEHY	14283	LITTLE #3	GARDNER CONSULTANTS, INC - 3 PEN RECORDER, SOLAR PANEL, METER HOUSE, 3" SKID MOUNTED METER RUN WIEFM 9/15
PROCESS EQUIP-SEP,MTR,TNK,DEHY	14283	LITTLE #3	KELLY PRODUCTION SERVICE - 24" X 10" 1440# W.P. 3-PHASE VERTICAL SEPARATOR & VOLUME POT STAGE II 9/21
PROCESS EQUIP-SEP,MTR,TNK,DEHY	14283	LITTLE #3	BELL SUPPLY CO - BATTERY FOR METER CLOCKS 9/28
PROCESS EQUIP-SEP,MTR,TNK,DEHY	14283	LITTLE #3	GARDNER CONSULTANTS, INC - GAS SAMPLES & ANALYSIS 4/15
PROCESS EQUIP-SEP,MTR,TNK,DEHY	14289	W. STOCKMAN GATHERING SYSTEM	TEJAS PRODUCTION SERVICE, - COALESCING FILTER, B-TEX UNIT, DEHY TOWER & REBOILER, HORIZONTAL SEPARATOR 6/12
PROCESS EQUIP-SEP,MTR,TNK,DEHY	14289	W. STOCKMAN GATHERING SYSTEM	J-W MEASUREMENT COMPANY - 6" METER RUN (SR) & 3 PEN RECORDER, EFM & METER HOUSE 6/30
PROCESS EQUIP-SEP,MTR,TNK,DEHY	14289	W. STOCKMAN GATHERING SYSTEM	CHALLENGER PROCESS SYSTEM - (2) 210 BBLS. STEEL COATED TANKS, STAIRWAY, WALKWAY & FIREWALL CROSSOVER 7/10
PROCESS EQUIP-SEP,MTR,TNK,DEHY	14289	W. STOCKMAN GATHERING SYSTEM	J-W MEASUREMENT COMPANY - ORIFICE PLATE FLANGE FOR METER RUN 4/12
PROCESS EQUIP-SEP,MTR,TNK,DEHY	14289	W. STOCKMAN GATHERING SYSTEM	GARDNER CONSULTANTS, INC - (2) PEN RECORDER & CONNECTION TO HOOK IT UP , GAS ANALYSIS 4/15
PROCESS EQUIP-SEP,MTR,TNK,DEHY	16001	BONNER #1-H	WEATHERFORD ARTIFICIAL LIFT - RECALCULATE VESSEL FOR JET PUMP 11/14

Account Description	SubAccount	Well Name	Description
PROCESS EQUIP-SEP,MTR,TNK,DEHY	16001	BONNER #1-H	LONG INDUSTRIES, INC (3) 300 BBL STEEL TANKS, (2) 300 BBL FIBERGLASS TANKS, (1) 400 BBL FIBERGLASS TANK, STAIRS, CATWALK 12/20
PROCESS EQUIP-SEP,MTR,TNK,DEHY	16001	BONNER #1-H	LONG INDUSTRIES, INC HEATER TREATER 12/23
PROCESS EQUIP-SEP,MTR,TNK,DEHY	16001	BONNER #1-H	NATIONAL OILWELL VARCO, L.P HIGH TEMP WIRE FOR FLARE 1/3
PROCESS EQUIP-SEP,MTR,TNK,DEHY	16001	BONNER #1-H	LEGACY MEASUREMENT SOLUTIONS, INC 3 PEN CHART RECORDER & CONNECTIONS TO HOOK IT UP 1/7
PROCESS EQUIP-SEP,MTR,TNK,DEHY	16001	BONNER #1-H	SABINE PIPE, INC PIPE TO BUILD DRIP POT 1/10
PROCESS EQUIP-SEP,MTR,TNK,DEHY	16001	BONNER #1-H	QUITMAN TANK SOLUTIONS LLC - 400 BBL STEEL POWER OIL TANK 1/15
PROCESS EQUIP-SEP,MTR,TNK,DEHY	16001	BONNER #1-H	J&P HOLDINGS - 40' STORAGE CONTAINER
PROCESS EQUIP-SEP,MTR,TNK,DEHY	16001	BONNER #1-H	J & B ENERGY SERVICE - NEW 5 HP ELECTRIC MOTOR & CENTRIFUGAL PUMP FOR POWER OIL CHARGE PUMP 1/28
PROCESS EQUIP-SEP,MTR,TNK,DEHY	16001	BONNER #1-H	LEGACY MEASUREMENT SOLUTIONS, INC MANIFOLD FOR GAS METER 1/28
PROCESS EQUIP-SEP,MTR,TNK,DEHY	16001	BONNER #1-H	MASTER PUMPS & POWER - (1) 5 HP ELECTRIC MOTOR FOR POWER OIL CHG. PUMP 1/31
PROCESS EQUIP-SEP,MTR,TNK,DEHY	16001	BONNER #1-H	MATERIAL TRANSFER 08/30/17 - 400 BBL STEEL POWER OIL TANK SN #369 (QTY 1) FROM BONNER #1H TO GERALD LAMBRIGHT #1D SWD
PROCESS EQUIP-SEP,MTR,TNK,DEHY	16003	FINLEY #1	WEATHERFORD ARTIFICIAL LIFT - RECYCLE VESSEL 11/25
PROCESS EQUIP-SEP,MTR,TNK,DEHY	16003	FINLEY #1	MT 12/2/13 - 1-10'x24", 3 PHASE 1,440 PSI HORIZONTAL SEP. SER. #3B179-001 FROM THE HERMAN MENTZEL #1H TO THE FINLEY #1H
PROCESS EQUIP-SEP,MTR,TNK,DEHY	16003	FINLEY #1	LONG INDUSTRIES, INC 6 X 20 75# HEATER TREATER 12/5
PROCESS EQUIP-SEP,MTR,TNK,DEHY	16003	FINLEY #1	LONG INDUSTRIES, INC (3) NEW STEEL 300 BBL TANKS, (2) 300 BBL FIBERGLASS TANKS, (1) 400 BBL FIBERGLASS TANK, STAIRS & CATWALK 12/5
PROCESS EQUIP-SEP,MTR,TNK,DEHY	16003	FINLEY #1	GLOBAL VESSEL & TANK, LLC - 24' X 7'-6" 2PH VERTICAL SEPARATOR, BASE BLOCK & ACCESSORIES 1/24
PROCESS EQUIP-SEP,MTR,TNK,DEHY	16003	FINLEY #1	2/7/14 MT - 1 42° X 10' SKID MOUNTED RECYCLE VESSEL 245 PSI. S/N 49.13-2 FROM FINLEY #1H TO JONES 519 #1H
PROCESS EQUIP-SEP,MTR,TNK,DEHY	16011	BRICE BONNER #1	GLOBAL VESSEL & TANK, LLC - HEATER TREATER 5/29
PROCESS EQUIP-SEP,MTR,TNK,DEHY	16011	BRICE BONNER #1	LEGACY MEASUREMENT SOLUTIONS, INC 3 PEN CHART RECORDER, CONNECTIONS & MATERIAL TO SET & HOOKUP 6/2
PROCESS EQUIP-SEP,MTR,TNK,DEHY	15702	JONES 519 #1	2/7/14 MT - 1 42° X 10' SKID MOUNTED RECYCLE VESSEL 245 PSI. S/N 49.13-2 FROM FINLEY #1H TO JONES 519 #1H
PROCESS EQUIP-SEP,MTR,TNK,DEHY	15702	JONES 519 #1	GLOBAL VESSEL & TANK, LLC - 6' X 20' HEATER TREATER, (2) 400 BBL STEEL TANK COATED, (2) 400 BBL STEEL TANKS, (2) FIREWALL CROSS-OVER, STAIRS AND WALKWAY & FREIGH
PROCESS EQUIP-SEP,MTR,TNK,DEHY	15702	JONES 519 #1	GLOBAL VESSEL & TANK, LLC - 6' X 15' HORIZONTAL FWKO 5/7
PROCESS EQUIP-SEP,MTR,TNK,DEHY	15702	JONES 519 #1	IRA PUMP & SUPPLY CO., INC LMI CHEMICAL PUMP 8/7
PROCESS EQUIP-SEP,MTR,TNK,DEHY	15702	JONES 519 #1	MATERIAL TRANSFER 05/24/2019 - 6'x15' HORIZONTAL NON-CODE, 125#WP, 2 PHASE MECHANICAL, FREEWATER KNOCKOUT, INTERNALLY COAT COMPLETE W/ NOVAPLATE UHS - SN: LV5411 #1H TO HOLLIDAY #1H

Account Description	SubAccount	Well Name	Description	
PROCESS EQUIP-SEP,MTR,TNK,DEHY	15702	JONES 519 #1	PUMPING UNITS AND ALL EQUIPMENT IN THE FIELD YARD UTILIZED FOR THE FIELD IN WHICH THIS WELL IS LOCATED	
PROCESS EQUIP-SEP,MTR,TNK,DEHY	15705	JONES 519-2H	PUMPING UNITS AND ALL EQUIPMENT IN THE FIELD YARD UTILIZED FOR THE FIELD IN WHICH THIS WELL IS LOCATED	
PROCESS EQUIP-SEP,MTR,TNK,DEHY	15705	JONES 519-2H	GLOBAL VESSEL & TANK, LLC - TREATER, TANKS & WALKWAY 8/27	
PROCESS EQUIP-SEP,MTR,TNK,DEHY	15705	JONES 519-2H	LEGACY MEASUREMENT SOLUTIONS, INC 2 PEN CHART RECORDER, CONECTIONS & MATERIAL TO SET AND HOOK UP 9/4	
PROCESS EQUIP-SEP,MTR,TNK,DEHY	15705	JONES 519-2H	IRA PUMP & SUPPLY CO., INC CHEMICAL PUMP 10/27	
PROCESS EQUIP-SEP,MTR,TNK,DEHY	15709	GAYLAN JANE COLLIER 519 SWD	IRA PUMP & SUPPLY CO., INC 8" CHART RECORDER & ACCESSORIES 7/15	
PROCESS EQUIP-SEP,MTR,TNK,DEHY	15709	GAYLAN JANE COLLIER 519 SWD	IRA PUMP & SUPPLY CO., INC LMI CHEMICAL PUMP 8/8	
PROCESS EQUIP-SEP,MTR,TNK,DEHY	15709	GAYLAN JANE COLLIER 519 SWD	PUMPING UNITS AND ALL EQUIPMENT IN THE FIELD YARD UTILIZED FOR THE FIELD IN WHICH THIS WELL IS LOCATED	
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18116	BOOMHOWER #36-1	HESS SERVICES, INC FIBERGLASS GUNBARREL, (1) 12' X 10' STEEL PRODUCTION TANK, (1) 12' X 10' FIBERGLASS TANK & FITTINGS 2/24	
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18116	BOOMHOWER #36-1	MT# 18115 03-26-09 (1) 12'x10' HESS WELDED STEEL PROD TANK & 15' EXTENSION FR INV TO BOOMHOWER 36-1 ON 3-26-09 (RE: HESS SERVICES INV# 27467)	
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18117	BOOMHOWER #36-2	MT 3/14/12 SET STAIRS, CATWALK & WALKOVER FROM YOST #16-1 TO BOOMHOWER #36-2	
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18117	BOOMHOWER #36-2	MT 3/14/12 210 BBL FIBERGLASS GUN BARREL S/N 3309 FROM YOST #16-1 TO BOOMHOWER #36-2	
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18117	BOOMHOWER #36-2	MT 3/14/12 200 BBL KBK FIBERGLASS S/W TANK S/N 3344 FROM YOST #16-1 TO BOOMHOWER #36-2	
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18117	BOOMHOWER #36-2	MT 3/14/12 200 BBL STEEL STOCK TANKS S/N 13164 & 13253 FROM YOST #16-1 TO BOOMHOWER #36-2	
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18117	BOOMHOWER #36-2	SUNRISE OILFIELD SUPPLY - CHEMICAL PUMP & CONNECTION TO HOOKUP 3/19	
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18141	JENNISON 1-1	GABEL LEASE SERVICE, INC 3-STEEL TANKS 1-FG TANK; GUNBARREL 6/20	
PROCESS EQUIP-SEP,MTR,TNK,DEHY	36207	CLOYD 29 #1	ALL EQUIPMENT	
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18143	COLT 45 FBC #2-1	HESS SERVICES, INC TANKS & GUN BARREL 9/30	
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18143	COLT 45 FBC #2-1	HESS SERVICES, INC 3 WELDED STEEL PROD. TANKS 9/9	
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18143	COLT 45 FBC #2-1	HESS SERVICES, INC 18' WALKWAY, 15' EXTENSION, STAIRWAY FOR 10' HIGH TANK 11/7	
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18146	JENNISON 1-2 SWD	HESS SERVICES, INC GUN BARREL, PROD. TANK, SW TANK W/STAIRS 1/26	
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18146	JENNISON 1-2 SWD	MT 11/18/11 200 BBL HESS STEEL TANK S/N 11220 FROM JENNISON 1-2 TO JENNISON 36#1	
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18146	JENNISON 1-2 SWD	GABEL LEASE SERVICE, INC INSTALL E-PUMP 10/15	
PROCESS EQUIP-SEP,MTR,TNK,DEHY	42103	JESSIE ALLEN 17 14 #1	ALL EQUIPMENT	

Account Description	SubAccount	Well Name	Description
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18148	SNIDER 35-1	HESS SERVICES, INC (1) 10' X 15' FG. GUN BARREL, (2) 12' X 10' 200 BBL. STEEL STOCK TANKS, (1) 12' X 10' 200 BBL. FG. SW TANK & STAIRS & WALKWAY 6/14
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18148	SNIDER 35-1	JACAM CHEMICALS, LLC - 65 GALLON UNLINED STEEL DRUM 9/27/11
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18148	SNIDER 35-1	JACAM CHEMICALS, LLC - BLUE PULSAFEEDER PUMP 10/12/11
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18173	BOOMHOWER 36 #3	PFEIFER DOZER & WELL SVC, LLC - 200 BBL FIBERGLASS GUNBARREL, 200 BBL. FIBERGLASS WATER TANK, 200 BBL. STEEL STOCK TANKS 11/8
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18173	BOOMHOWER 36 #3	PFEIFER DOZER & WELL SVC, LLC - 200 BBL FIBERGLASS GUNBARREL, 200 BBL. FIBERGLASS WATER TANK, 200 BBL. STEEL STOCK TANKS 11/8
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18173	BOOMHOWER 36 #3	PFEIFER DOZER & WELL SVC, LLC - 200 BBL FIBERGLASS GUNBARREL, 200 BBL. FIBERGLASS WATER TANK, 200 BBL. STEEL STOCK TANKS 11/8
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18174	REDBURN 1-1	PFEIFER DOZER & WELL SVC, LLC - 200 BBL. FIBERGLASS GUNBARREL, 200 BBL. FIBERGLASS WATER TANKS & 200 BBL. STEEL TANKS 11/27
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18175	BOOMHOWER 36 #4	PFEIFER DOZER & WELL SVC, LLC - 10 X 12 CLOSED TOP SW FIBERGLASS TANK 2/19
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18228	JENNISON 36#1	MT 11/18/11 200 BBL HESS STEEL TANK S/N 13037 FROM YOST 16-1 TO JENNISON 36#1
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18228	JENNISON 36#1	MT 11/18/11 200 BBL HESS STEEL TANK S/N 11220 FROM JENNISON 1-2 TO JENNISON 36#1
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18228	JENNISON 36#1	MT 11/18/11 200 BBL KBK FIBERGLASS S/W TANK S/N 58031 FROM GOOSSEN 11-1 TO JENNISON 36#1
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18228	JENNISON 36#1	JACAM CHEMICALS, LLC - CHEMICAL PUMP 1/20
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18229	SHARP SEED 2 #1	NORTHWEST T&T ELECTRIC, INC PARTS TO HOOKUP SHARP SEED 2 #1 TO ELECTRIC 11/18
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18229	SHARP SEED 2 #1	NORTHWEST T&T ELECTRIC, INC LABOR & PARTS TO HOOKUP ELECTRIC TO WELL 11/18
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18229	SHARP SEED 2 #1	MT 11/18/11 210 BBL SUNRISE FIBERGLASS GUN BARREL SIN 59772 FROM GOOSSEN 11-1 TO SHARP SEED 2#1
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18229	SHARP SEED 2 #1	MT 11/18/11 200 BBL HESS STEEL STOCK TANKS S/N 22169 & 22168 FROM GOOSSEN 11-1 TO SHARP SEED 2#1
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18229	SHARP SEED 2 #1	MT 11/18/11 200 BBL KBK FIB 59800 FROM GOOSSEN 11-1 TO SHARP SEED 2#1
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18230	SNIDER-SHARP SEED 35 #1	PFEIFER DOZER & WELL SVC, LLC - 200 BBL STOCK TANKS, 8 X 15 GUNBARREL & 200 BBL WATER TANK 8/9
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18234	COLT 45 FBCE 2 #4	PFEIFER DOZER & WELL SVC, LLC - FIBERGLASS GUNBARREL, WATER TANKS & STEEL STOCK TANK 1/27
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18703	VONDRACEK 4 #1	PFEIFER DOZER & WELL SVC, LLC - 200 BBL. GUN BARREL, 200 BBL. SW TANK & 200 BBL. STOCK TANK 9/19
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18703	VONDRACEK 4 #1	PFEIFER DOZER & WELL SVC, LLC - 200 BBL. STOCK TANK & 15' EXT. WALK 10/2
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18707	MOEDER TRUST 3 NO. 1	MCDONALD TANK & EQUIPMENT CO. INC (2) 12' X 10' 200 BBL. FIBERGLASS STOCK TANKS & 16' WALKWAY 7/26
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18707	MOEDER TRUST 3 NO. 1	MT 7/27/16 - 200 BBL 12' X 10' HESS FIBERGLASS TANKS, 10'X15' HESS FIBERGLASS GUN BARREL, 200 BBL 12'X10' HESS FIBERGLASS SW TANK, STAIRS & WALKWAY FROM SCHULTZ #25-1
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18707	MOEDER TRUST 3 NO. 1	DOUBLE F PUMP & SUPPLY - CHEMICAL PUMP (BEAM) & CONNECTIONS 9/21

Account Description	SubAccount	Well Name	Description
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18709	MOEDER 3 #2	MATERIAL TRANSFER 02/23/17 - (2) 200 BBL 12'X10' STEEL TANKS SER. # 1375 & 17377, (1) 10'X15' FIBERGLASS GUN BARREL SER. #17537, (1) 12'X10' FIBERGLASS WATER TANK, STAIRS & WALK KOB 12 #1 TO MOEDER 3 #2
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18714	VONDRACEK ETAL 4 #2	MCDONALD TANK & EQUIPMENT CO. INC (2) 12' X 10' 200 BBL. FIBERGLASS STOCK TANKS SERIAL # M-16887 SERIAL # M-1079 9/20
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18714	VONDRACEK ETAL 4 #2	MATERIAL TRANSFER 9/13/16 - 200 BBL 12' X 10' HESS STEEL TANKS (QTY 2), 10' X 15' HESS FIBERGLASS GUN BARREL (QTY 1), 200 BBL 12' X 10' HESS FIBERGLASS SW TANK (QTY 1), STAIRS & FROM YOST #9-1 TO VONDRACEK ETAL 4 #2
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18714	VONDRACEK ETAL 4 #2	MCDONALD TANK & EQUIPMENT CO. INC NEW 10' X 15' FIBERGLASS GUNBARREL, (3) 12' X 10' FIBERGLASS STOCK TANKS, WALKWAY, STAIRS 10/6
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18714	VONDRACEK ETAL 4 #2	MCDONALD TANK & EQUIPMENT CO. INC (2) NEW 12' X 10' FIBERGLASS STOCK TANKS, STAIRS AND WALKWAY 10/7
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18717	BESPERAT 33#1	MCDONALD TANK & EQUIPMENT CO. INC (1) FIBERGLASS 10' X 15' GUNBARREL, (1) FIBERGLASS 12' X 10' SW TANK, (2) FIBERGLASS 12' X 10' STOCK TANKS, WALKWAY & STAIRS
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18802	PRATHER FARMS 22-1	PFEIFER DOZER & WELL SVC, LLC - GUNBARREL, WATER TANK & STOCK TANK & WALK 6/12
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18803	SWART 21 #1	PFEIFER DOZER & WELL SVC, LLC - GUNBARREL, WATER TANK, STOCK TANK & WALK 6/28
PROCESS EQUIP-SEP,MTR,TNK,DEHY	18806	MSJ-PENNER 28-1	PFEIFER DOZER & WELL SVC, LLC - (1) 10' X 15' FIBERGLASS GUNBARREL, (1) 10' X 12' FIBERGLASS WATER TANK, (2) STEEL STOCK TANKS & STAIRS & CATWALK 4/8

SCHEDULE 2.1(b)(xii) SEISMIC SURVEYS

Schedule 2.1(b)(xii) Seismic Surveys

SURVEY NAME	STATE	COUNTY	AREA (SQMI)	SURVEY NAME	STATE	COUNTY	AREA (SQMI)
Timber Ridge	Kansas	Cheyenne	29	Fluvanna	Texas	Borden	13
Cheyenne	Kansas	Cheyenne	23	West Arcadia	Louisiana	Bienville	3.3
Aunt Gertie	Kansas	Thomas	5			Total:	16.3
Berado	Kansas	Thomas	2				
Old Timer	Kansas	Thomas	6				
STS	Kansas	Thomas	3				
West Gove	Kansas	Gove	13				
Plum Creek	Kansas	Gove	8				
Coberly	Kansas	Gove	5				
Ellis 12-20	Kansas	Ellis	20				
Ellis 13-19	Kansas	Ellis	13				
Matilda	Kansas	Russell	4				
Moore	Kansas	Rush	17				
Helen Ready	Kansas	Pawnee	8				
Colt 45	Kansas	Lane	9				
Kobe	Kansas	Lane	8				
Alloy Never-Ending	Kansas	Lane	10				
King James	Kansas	Lane	3				
BoomBoom	Kansas	Lane	15				
Xmen South	Kansas	Lane	15				
Reischneider	Kansas	Lane	8				
Firecracker	Kansas	Lane	1				
Uncle Remus	Kansas	Lane	7				
Lane 20-28	Kansas	Lane	32				
		Total:	264				

SCHEDULE 2.2(b)

CERTAIN EXCLUDED CASH AND EQUIVALENTS

None.

SCHEDULE 2.2(g)

EXCLUDED LEASES AND EXCLUDED MINERAL INTERESTS

Lease #	Date	Lessor	Lessee	Book-Page	State	County
42.395.14700.0001	6/1/2005	Devon Energy Production Company, L. P.	Enduring Resouces, LLC	0909-258	Texas	Robertson
42.395.14700.0002	6/1/2005	Devon Energy Production Company, L. P.	Enduring Resouces, LLC	0909-264	Texas	Robertson
42.395.14700.0004	12/1/2007	Devon Energy Production Company, L. P.	O'Brien Resources, LLC	1017-647	Texas	Robertson
42.395.14700.0005	12/1/2007	Devon Energy Production Company, L. P.	O'Brien Resources, LLC	1017-654	Texas	Robertson
42.395.14700.0006	12/1/2007	Devon Energy Production Company, L. P.	O'Brien Resources, LLC	1017-660	Texas	Robertson
42.395.14700.0007	1/10/2003	TXU Mining Company LP and TXU Generation Company LP	Devon Energy Production	0832-088	Texas	Robertson
42.395.14700.0008	6/18/2007	TXU Mineral Development Company I, LP	O'Brien Energy Company	0996-299	Texas	Robertson
42.395.14700.0030	1/25/2005	Joe T. Tullos, Sr. et ux	Prime Resources, LP	0918-788	Texas	Robertson
42.395.14700.0045	2/11/2009	Sue Dee Pirkle	O'Brien Resources, LLC	1066-024	Texas	Robertson
42.395.14700.0058	8/20/2008	Merideth Wells Blair	O'Brien Resources, LLC	1059-283	Texas	Robertson
42.395.14700.0059	6/23/2007	Harry E. Ezell	O'Brien Energy Company	0999-406	Texas	Robertson
42.395.14700.0060	5/6/2008	Elizabeth Fowler	O'Brien Resources, LLC	1051-345	Texas	Robertson
42.395.14700.0061	6/9/2011	Ann Yarbrough Jones	O'Brien Resources, LLC	1142-223	Texas	Robertson
42.395.14700.0062	6/9/2011	Frank Markham Jr., et ux	O'Brien Resources, LLC	1142-220	Texas	Robertson
42.395.14700.0063	12/5/2013	Virginia Wells	O'Brien Resources, LLC	1229-024	Texas	Robertson
42.395.14700.0064	4/7/2008	Howard Wookey (Mary Puffer Estate)	O'Brien Resources, LLC	1043-326	Texas	Robertson
42.395.14700.0065	5/17/2006	Lora Bilbro	Sarah Exploration Company	0955-601	Texas	Robertson
42.395.14700.0067	8/24/2005	Lora Bilbro	Sarah Exploration Company	0933-417	Texas	Robertson
42.395.14700.0068	5/17/2006	Larry Scott Heggie	Sarah Exploration Company	0959-0506	Texas	Robertson
42.395.14700.0069	8/24/2005	William Dean Heggie	Sarah Exploration Company	0943-0542	Texas	Robertson
42.395.14700.0073	9/16/2005	Robert Watson	Sarah Exploration Company	0993-0404	Texas	Robertson
42.395.14700.0075	4/5/2007	Patricia Ann Lamont	O'Brien Energy Company	0985-671	Texas	Robertson
42.395.14700.0077	10/28/2005	James W. Jackson, Jr. et ux	Sarah Exploration Company	0933-496	Texas	Robertson
42.395.14700.0078	5/7/2008	Victor Wooten, et ux	O'Brien Resources, LLC	1050-073	Texas	Robertson
42.395.14700.0079	2/11/2009	Paula Dee Willis	O'Brien Resources, LLC	1066-027	Texas	Robertson
42.395.14700.0080	5/17/2006	Larry Scott Heggie	Sarah Exploration Company	0959-0506	Texas	Robertson
42.395.14700.0081	4/8/2008	William Dean Heggie	O'Brien Resources, LLC	1043-278	Texas	Robertson
42.395.14700.0082	4/8/2008	Willian Inez Heggie	O'Brien Resources, LLC	1043-273	Texas	Robertson
42.395.14700.0083	4/19/2007	Robert Watson	O'Brien Energy Company	0992-025	Texas	Robertson
42.395.14700.0085	10/15/2005	Lynda Jo Corn Bear	Sarah Exploration Company	0933-0411	Texas	Robertson
42.395.14700.0086	10/15/2005	E. D. Corn Jr.	Sarah Exploration Company	0933-0447	Texas	Robertson
42.395.14700.0087	10/15/2005	Holly Ann John	Sarah Exploration Company	0933-0500	Texas	Robertson
42.395.14700.0093	5/14/2006	Lynda Jo Corn Bear	Sarah Exploration Company	0959-481	Texas	Robertson
42.395.14700.0094	2/29/2008	Frank V. Bernobich	O'Brien Resources, LLC	1045-769	Texas	Robertson
42.395.14700.0095	5/14/2006	E. D. Corn Jr.	Sarah Exploration Company	0955-598	Texas	Robertson
42.395.14700.0096	5/15/2006	James W. Jackson, Jr. et ux	Sarah Exploration Company	0955-615	Texas	Robertson
42.395.14700.0097	5/14/2006	Holly Ann John	Sarah Exploration Company	0967-440	Texas	Robertson
42.395.14700.0098	5/29/2008	Allen Cay Brantner	O'Brien Resources, LLC	1043-281	Texas	Robertson
42.395.14700.0099	5/29/2008	Ronald Dean Brantner	O'Brien Resources, LLC	1043-285	Texas	Robertson
42.395.14700.0100	6/6/2008	Betty Jo Kyle	O'Brien Resources, LLC	1043-305	Texas	Robertson
42.395.14700.0101	5/22/2008	Katherine Sherrouse	O'Brien Resources, LLC	1043-283	Texas	Robertson
42.395.14700.0102	6/6/2008	Dale Wiley	O'Brien Resources, LLC	1043-301	Texas	Robertson
42.395.14700.0103	6/6/2008	Kenneth W. Wiley	O'Brien Resources, LLC	1043-303	Texas	Robertson
42.395.14700.0104	9/15/2008	James S. Ingram	O'Brien Resources, LLC	1055-256	Texas	Robertson
42.395.14700.0105	6/3/2008	Jimmy Lynn Truett	O'Brien Resources, LLC	1047-604	Texas	Robertson
42.395.14700.0106	1/30/2009	Deborah Allard Wilson	Eagle Oil & Gas Co.	1067-076	Texas	Robertson
42.395.14700.0107	2/23/2009	Jimmy Lynn Truett	O'Brien Resources, LLC	1067-504	Texas	Robertson
42.395.14700.0108	6/30/2005	Esmeralda Wiley	Prime Resources, LP	0933-750	Texas	Robertson
42.395.14700.0109	11/29/2005	Vivian Brown Testamentary Trust	Prime Resources, LP	0955-621	Texas	Robertson
42.395.14700.0110	2/22/2008	Dan Kyle et ux	O'Brien Resources, LLC	1029-662	Texas	Robertson
42.395.14700.0111	2/22/2008	Dan Kyle, Trustee	O'Brien Resources, LLC	1029-661	Texas	Robertson
42.395.14700.0112	2/22/2008	Daryl E. Kyle et ux	O'Brien Resources, LLC	1029-663	Texas	Robertson
42.395.14700.0113	2/21/2008	Douglas Kyle	O'Brien Resources, LLC	1060-386	Texas	Robertson
42.395.14700.0114	2/22/2008	Nell McKinney	O'Brien Resources, LLC	1029-660	Texas	Robertson
42.395.14700.0115	2/8/2008	Donnie Pauline Miller	O'Brien Resources, LLC	1029-664	Texas	Robertson
42.395.14700.0116	2/8/2008	Dennis Wayne Watson	O'Brien Resources, LLC	1029-658	Texas	Robertson
42.395.14700.0117	2/8/2008	Vernon Ray Watson	O'Brien Resources, LLC	1029-666	Texas	Robertson
42.395.14700.0118	3/1/2008	Candace Ann Simmons Anderson	O'Brien Resources, LLC	1054-418	Texas	Robertson
42.395.14700.0119	2/11/2008	Frank Wesley Grayson, Successor Trustee of Vivian Brown Testamentary Trust	O'Brien Resources, LLC	1024-493	Texas	Robertson
42.395.14700.0120	3/1/2008	Gene Julie Simmons	O'Brien Resources, LLC	1054-409	Texas	Robertson
42.395.14700.0121	3/1/2008	Ross Crafford Simmons	O'Brien Resources, LLC	1054-412	Texas	Robertson
42.395.14700.0122	3/1/2008	Cornelia Ann Loreant Starkey Trust	O'Brien Resources, LLC	1054-415	Texas	Robertson
42.395.14700.0123	3/1/2008	Johnny R. Starkey	O'Brien Resources, LLC	1055-253	Texas	Robertson
42.395.14700.0124		Myron Koehler	O'Brien Energy Company	1043-201	Texas	Robertson
42.395.14700.0125		Donnie Pauline Miller aka Polly Miller	O'Brien Resources, LLC	1038-409	Texas	Robertson
	J					

	1038-408 0988-221 0957-414	Texas	Robertson
24.00.1.00.00.00.00.00.00.00.00.00.00.00.0		Tayaa	1 1000110011
4 285 EM FRO 2019 ST HOUSED BY AND AN EXPORTED Stant Replacetor Company 4 285 EM FRO 2019 5 150000 By Anne St March B	0957-414	Texas	Robertson
4385-547001701 11400000 ESC Morros, LP 4235-547001701 6170000 Alle Michael Hollywin Michael path Esc Records, LP 4235-547001701 6170000 Alle Michael Hollywin Michael path Esc Records, LP 4235-547001701 6170000 Alle Michael Hollywin Michael path Esc Records, LP 4235-547001701 6170000 Michael Hollywin Michael path Esc Records, LP 4235-547001701 6170000 Michael Hollywin Michael path Esc Records, LP 4235-547001701 6170000 Michael Hollywin Michael path Esc Records, LP 4235-547001701 6170000 Michael Hollywin Michael path Esc Records, LP 4235-547001701 6170000 Michael Hollywin Michael path Esc Records, LP 4235-547001701 6170000 Michael Hollywin Michael path Esc Records, LP 4235-547001701 6170000 Michael Hollywin Michael path Esc Records, LP 4235-547001701 6170000 Michael Hollywin Michael path Esc Records, LP 4235-547001701 6170000 Michael Hollywin Michael path Esc Records, LP 4235-5470001701		Texas	Robertson
42295 M1701210 5170000 Ballow Engingsym Michael Samb Explacation Company 42305 M1701210 677000 Activity Michael Prime Recounce, LP 42305 M1701210 627000 Activity Michael Samb Explacation Company 42305 M1701210 627000 Michael Michael Company 42305 M1701210 627000 Michael Michael Company 42305 M1701210 627000 Michael Michael Michael Michael Company 42305 M1701210 627000 Michael Michael Michael Michael Defendences 42305 M1701210 8170000 Michael Michael Michael Defendences 42305 M1701210 8170000 8170000 Michael Michael Defendences Defendences 42305 M1701210 8170000 8170000 Michael Michael Michael 42305 M1701210 8170000 Michael Michael Michael Michael 42305 M1701210 8170000 Michael Michael Michael Michael 42305 M1701210 8170000	0955-618	Texas	Robertson
423314 MERODICE 677000 Ablas Vateria Loon Prime Resource, LP 423304 MERODICES 422001 Machine Microsoft Stanta Education Company 423304 MERODICES 422001 March Education Company Others Energy Company 423304 MERODICES 422001 Machine Microsoft Others Energy Company 423304 MERODICES 430000 Flow Microsoft Prime Resource, LP 423304 MERODICES 430000 Machine Deliveration of Company 423304 MERODICES 430000 Machine De	1091-304	Texas	Robertson
42.30.5 (4.200.15) 6.700.00 Joseph V. Lame Plane Replication Company 42.30.5 (4.200.15) 4.900.00 College Lynning Teach Decay College Lynning College Lynning 42.30.5 (4.200.15) 4.900.00 College Lynning College Lynning College Lynning 42.30.5 (4.200.15) 4.900.00 College Lynning College Lynning College Lynning 42.30.5 (4.200.15) 4.900.00 College Lynning College Lynning College Lynning 42.30.5 (4.200.15) 4.900.00 College Lynning College Lynning College Lynning 42.30.5 (4.200.15) 4.900.00 College Lynning College Lynning College Lynning 42.30.5 (4.200.15) 4.900.00 College Lynning College Lynning College Lynning 42.30.5 (4.200.15) 4.900.00 College Lynning College Lynning College Lynning 42.30.5 (4.200.15) 4.900.00 College Lynning College Lynning College Lynning 42.30.5 (4.200.15) 4.900.00 College Lynning College Lynning College Lynning College Lynning 42.30.5 (4.200.15) <td>0955-595</td> <td>Texas</td> <td>Robertson</td>	0955-595	Texas	Robertson
429816 (1700) 1354 49/2015 Month Epiralism (2000) ASS (1700) 1355 49/2015 Month Epiralism (2000) 42.306 (1700) 1355 49/2017 Month Epiralism (2000) Office Except Company 42.306 (1700) 1356 49/2017 Silve Treeton, Jr. Office Except Company 42.306 (1700) 1369 19/2000 Month Except Company 42.306 (1700) 1369 19/2000 Month Except Company 42.306 (1700) 1369 19/2000 Month Except Company 42.306 (1700) 1361 21/1000 Month Except C	0918-510	Texas	Robertson
42305 14700.035 43007 Mary Trevino Slouny Office Energy Company 42365 14700.035 43007 Assert Takes, it Office Energy Company 42365 14700.035 430070 Marce Takes, it Office Energy Company 42365 14700.036 1870000 Myror Koceler Home Resources, LP 42365 14700.036 1870000 Month Control Seasth Expression Prints Resources, LP 42365 14700.036 1870000 Miror Koceler Seasth Expression Office Resources, LC 42365 14700.031 191000 Say F. McDonald at all Say F. McDonald at all Seasth Expression Company 42365 14700.031 192000 Say F. McDonald at all Say F. McDonald at all Seasth Expression Company 42365 14700.035 192000 Say F. McDonald at all Seasth Expression Company Seasth Expression Company 42365 14700.035 192000 Say F. McDonald at all Seasth Expression Company Seasth Expression Company 42365 14700.035 192000 All Seasth Expression Company Seasth Expression Company Seasth Expression Company 42365 14700.035 192000 Al	0918-516	Texas	Robertson
4.2.36.1470.0134 4.00000 Root Treview. Jr. O'Bleen Energy Company 4.2.36.1470.0137 4.00000 Myrob. Service Long. Phine Resources, LP 4.2.36.1470.0139 5110000 Myrob. Myrob. Content. Phine Resources, LP 4.2.36.1470.0141 2110000 Myrob. Myrob. Mochanic Seath Englished Seath Englished 4.2.36.1470.0141 2110000 Myrob. Mary F. McChond et al. Seath Englished Seath Englished 4.2.36.1470.0141 2110000 Myrob. Testing of Belle Myrob. Seath Englished Seath Englished 4.2.36.1470.0141 2110000 Myrob. Myrob. Myrob. Seath Englished Seath Englished 4.2.36.1470.0140 2110000 Myrob. Myrob. Myrob. Seath Englished Seath Englished 4.2.36.1470.0140 210000 Myrob.	0967-434	Texas	Robertson
43.05.1.470.01372 40.0000 Free Transport Company 42.05.1.470.01380 67.0000 Myrox Konfere is us 42.05.1.470.01390 57.00000 Mirch Company 42.05.1.470.01390 57.00000 Mirch Company 42.05.1.470.01390 10.00000 Mirch Company 42.05.1.470.01391 10.00000 Mirch Company 42.05.1.470.01391 10.00000 Mirch Company 42.05.1.470.01391 40.00000 Mirch Company 42.05.1.470.01391 40.00000 Mirch Company 42.05.1.470.01391 40.000000 Mirch Company 42.05.1.470.01391 40.0000000 Mirch Company 42.05.1.470.01391 40.00000000000000000000000000000000000	0992-004	Texas	Robertson
42.356.1470.01348 67 162000 Myron Norderte ut Prime Resources, LP 42.356.1470.0140 67 162000 Myron Norderte ut Prime Resources, LP 42.356.1470.0141 21 102000 Myron Norderte ut Globel Production Company 42.356.1470.0144 40 102000 Microbio Machine Company 42.356.1470.0144 40 102000 The Edited of Eval Media Groups 42.356.1470.0144 60 102000 The Edited of Eval Media Groups 42.356.1470.0144 60 102000 Description Company 42.356.1470.0144 60 102000 Control Specimen Search Epiporation Company 42.356.1470.0144 60 102000 Control My Never Search Epiporation Company 42.356.1470.0146 40 102000 Control My Never Search Epiporation Company 42.356.1470.0146 40 102000 Resource LP LP Association Company 42.356.1470.0147 40 102000 Resource LP LP LP Association Company 42.356.1470.0148 40 102000 Resource LP	0992-001	Texas	Robertson
42.356.1470.01191 S18.0000 Revent John Resource LUP 42.356.1470.01194 S18.0000 Revent John Resident of to Great Fill (Schoolsfor Company) 42.356.1470.0114 10.10000 Mary F. McChonal et all 42.356.1470.0114 81.00000 Mary F. McChonal et all 42.356.1470.0114 81.00000 Richel Signature Steach Exploration Company 42.356.1470.0114 81.00000 Received Signature Steach Exploration Company 42.356.1470.0116 81.00000 Received Signature Steach Exploration Company 42.356.1470.0116 91.000000 Received Signature Steach Exploration Company 42.356.1470.0116 91.00000 Received Signature Steach Exploration Company 42.356.1470.0116 91.0000000 Interest Signature Steach Exploration Company 42.356.1470.0116 91.000000000000000000000000000000000000	0992-007	Texas	Robertson
42.35s.1470.014 81 10000 Martinez Des Factores Chife Personance, LLC 42.35s.1470.0144 20 10000 Martinez Des Factores Chife Personance, LLC 42.35s.1470.0140 20 20000 The Claim of Company Chife Medical and an all call control Chife Personance, LLC 42.35s.1470.0140 81 10000 Rectange of Status of Company Chife Status of Company Chife Status of Statu	0943-450	Texas	Robertson
42.356.1470.01441 21.9159 Mary F. Morbard Sort R. Explaination Company 42.356.1470.01441 31.9050007 The Estate of Cew Mark Shoquetien Cemeration Company 42.356.1470.01441 81.900007 18.100000 Sand Explaination Company 42.356.1470.01441 81.900007 18.100000 Sand Explaination Company 42.356.1470.01441 81.900007 18.100000 Cemeration Company 42.356.1470.01441 81.900007 18.100000 Cemeration Company 42.356.1470.01441 81.900007 Reliability Company Cemeration Company 42.356.1470.01441 9000000 Inchesting Company Cemeration Company 42.356.1470.01441 9000000 Inchesting Company Cemeration Company 42.356.1470.01441 9000000 Inchesting Company Cemeration Company 42.356.1470.01451 9000000 Inchesting Company Cemeration Company 42.356.1470.01451 9000000 Inchesting Company Cemeration Company 42.356.1470.01451 90000000 Inchesting Company Cemeration Company 42.356.1470.01451 9000000000	0943-444	Texas	Robertson
42.39s.1470.0144 18/2005	0933-408	Texas	Robertson
42.306.14700.0144 378/02007 The Extator of Eva Mas Sinequien Scrain Expertation Company 14.2306.14700.0144 878/0200 Scrain Expertation Company 14.2306.14700.0148 878/0200 Scrain Expertation Company 14.2306.14700.0148 878/0200 Scrain Expertation Company 14.2306.14700.0148 878/0200 Control State Expertation Company 14.2306.14700.0148 952/0200 Recompany 14.2306.1	1066-030	Texas	Robertson
42.396.1470.0144 81802005 Related Signanovic Samh Exploration Company 42.396.1470.0145 81812005 Robert Signanovic Samh Exploration Company 42.396.1470.0147 418207 Mill Exaltery Officer Energy Company 42.396.1470.0147 418207 Robin Exploration Company 42.396.1470.0149 9872000 Revisible Energy Company 42.396.1470.0150 61870.000 Investigation Company 42.396.1470.0150 61870.000 Investigation Company 42.396.1470.0151 32020000 Horne Many Marked Samh Exploration Company 42.396.1470.0154 4180.000 Horne Resources, LLC 42.396.1470.0156 4180.000 Horne Resources, LLC 42.396.1470.0157 8170.000 Mill Resources, LLC 42.396.1470.0158 8170.000 Horne Resources, LLC 42.396.1470.0159 8170.000 Robert Exploration Company 42.396.1470.0159 8170.000 Mill Resources, LLC 42.396.1470.0159 8170.000 Mill Resources, LLC 42.396.1470.0159 8170.0000 Horne Resources, LLC	0933-531	Texas	Robertson
42.386.1470.0146 8182005 Polesty Weener Sanh Exploration Company 42.386.1470.0148 818205 Control Weener Sanh Exploration Company 42.386.1470.0148 4122007 Member May Company College Exploration Company 42.386.1470.0148 952008 Member Modern et us Sanh Exploration Company 42.386.1470.0158 952008 Member May Company College Exploration Company 42.386.1470.0158 952000 Vernor May Law Containe Officer Resources, LD 42.386.1470.0158 952000 Vernor New Washen et us Sanh Exploration Company 42.386.1470.0158 972000 Vernor New Washen et us Pime Resources, LD 42.386.1470.0158 972000 Charles Resources, LD Officer Resources, LD 42.386.1470.0159 972000 Charles Resources, LD Officer Resources, LD 42.386.1470.0158 972000 Charles Resources, LD Officer Resources, LD 42.386.1470.0159 982000 Corres Murit Corres Murit Officer Resources, LD 42.386.1470.0161 1911.010 Resources, LD Officer Resources, LD <tr< td=""><td>0989-686</td><td>Texas</td><td>Robertson</td></tr<>	0989-686	Texas	Robertson
42.366.1470.0144 81/82.007 Chrith Easterly Christ Direct Company 42.366.1470.0147 41/22.007 Rulin Easterly Christ Direct Company 42.366.1470.0148 9.85/200 Kenneth McLeman et ux Scrah Exploration Company 42.366.1470.0149 0.95/200 Kenneth McLeman et ux Scrah Exploration Company 42.366.1470.0150 0.95/200 Scrah Exploration Company 42.366.1470.0151 0.92/2000 Homes Andersources, LLC 42.366.1470.0154 4.91/200 Robert L, White et ux Offering Resources, LLC 42.366.1470.0156 0.91/200 Robert L, White et ux Offering Resources, LLC 42.366.1470.0157 0.91/200 Robert L, White et ux Offering Resources, LLC 42.366.1470.0157 0.91/200 Aller Robert L, White et ux Offering Resources, LLC 42.366.1470.0158 0.91/200 Robert L, White et ux Offering Resources, LLC 42.366.1470.0158 0.91/200 Robert L, White et ux Offering Resources, LLC 42.366.1470.0158 0.91/200 Robert L, White et ux Offering Resources, LLC 42.366.1470.0158 0.	0933-573	Texas	Robertson
4.2396.14700.0147 4.102007 Ruth Easterly Officien Energy Company 4.2396.14700.0148 3.020007 Kein Jefford, et ut. Officien Energy Company 4.2396.14700.0150 0.1160008 James D. Marren Officien Energy Company 4.2396.14700.0152 3.020007 Very Lame Graham Officien Energy Company 4.2396.14700.0153 3.0220007 Very Lame Graham Officien Energy Company 4.2396.14700.0154 4.182007 Sobert L. White et ux Prime Resources, LP 4.2396.14700.0155 3.192007 Osyg Monthead, et ux Officien Energy Company 4.2396.14700.0156 3.012000 Overlace Regulated to United at ux Officien Resources, LP 4.2396.14700.0157 3.012000 Overlace Regulated to United Annual Hoggie Hall Officien Resources, LLC 4.2396.14700.0158 3.022000 Very Define Resources, LLC Officien Resources, LLC 4.2396.14700.0169 3.022000 Overlace Regulated Fision Officien Resources, LLC 4.2396.14700.0161 0.0112007 Round March Statement Annual Hoggie Hall Officien Resources, LLC 4.2396.14700.0161 0.0112007 Ro	0933-576	Texas	Robertson
42.995.1470.01146 3/92.027 Revin Jeffcoat, et us Certin Ferrity Company 42.995.1470.0149 95/2006 Inmesh McLerrane tu x Sanné Exploration Company 42.995.1470.0152 91/2000 Jernach McLerrane tu x O'Brien Ferrity Company 42.995.1470.0152 32/2000 Jernach James J. Marren O'Brien Ferrity Company 42.995.1470.0152 91/2000 Perrity Wilston et ux O'Brien Ferrity Company 42.395.1470.0156 91/2000 Outs Morehand, et ux O'Brien Ferrity Company 42.395.1470.0157 81/2000 Outs Morehand, et ux O'Brien Ferrity Company 42.395.1470.0158 81/2000 Orothy Frances Gamer O'Brien Resources, LLC 42.395.1470.0159 81/2000 O'Brien Resources, LLC 42.395.1470.0159 81/2000 O'Brien Resources, LLC 42.395.1470.0150 81/2000 O'Brien Resources, LLC 42.395.1470.0151 81/2000 O'Brien Resources, LLC 42.395.1470.0152 81/2000 O'Brien Resources, LLC 42.395.1470.0156 81/2000 D'Brien Resources, LLC 42.395.1470.0157 81/2000	0933-620	Texas	Robertson
42.995.1470.01940 95/2006 Kernels McLeram et u. Sarah Exploration Company 42.995.1470.01952 91/2008 James J. Minren O'Brien Resources, LLC 42.995.1470.01952 3/202007 James J. Minren O'Brien Resources, LLC 42.995.1470.01953 5/202008 Verron Ray Watson et ux Sarah Exploration Company 42.995.1470.01954 418,0005 Nobel M. Willie et ux O'Brien Resources, LP 42.295.1470.01956 318,0007 Dougl Morenhead, et ux O'Brien Resources, LLC 42.395.1470.01956 81/2000 Carrians Pay Morenhead et ux O'Brien Resources, LLC 42.395.1470.01957 82/2000 Gordin Frances Garner O'Brien Resources, LLC 42.395.1470.01961 81/2000 Cleman Jones, Trustee O'Brien Resources, LLC 42.395.1470.0197 8/2000 Gerien Jones, Trustee O'Brien Resources, LLC 42.395.1470.01981 101/142007 Berny Varian and Sharnon Varian O'Brien Resources, LLC 42.395.1470.01982 101/142007 Berny Leckhold L Willie et ux O'Brien Resources, LLC 42.395.1470.01981 101/142007 Brien Resources, LLC	0985-675	Texas	Robertson
42.955.4700.0150 6/18/200 James J. Marrien O'Brien Resources, LLC 42.956.14700.0153 5/22/200 Horry Lane Garbarin Carrien Garbarin Carr	0984-398	Texas	Robertson
42.395.1470.01512 3/20/207 Jerry Laine Grisham Client Energy Company 42.395.1470.01541 4/182.025 Vernon Ray Watson et ux Sarah Exploration Company 42.395.1470.01540 4/182.025 Robert L. White et ux O'Brinn Energy Company 42.395.1470.01550 3/192.007 Dougl Morehead et ux O'Brinn Energy Company 42.395.1470.01570 8/12208 Hima Rub Hegge Hall O'Brinn Resources, LLC 42.395.1470.01570 8/12208 Hima Rub Hegge Hall O'Brinn Resources, LLC 42.395.1470.01581 8/22080 Chorth Jones, Trustee O'Brinn Resources, LLC 42.395.1470.01591 8/22081 Chorne Mart O'Brinn Resources, LLC 42.395.1470.01510 10/11/207 Bruy Yann and Samon Vann O'Brinn Resources, LLC 42.395.1470.01511 10/11/207 Romald L. White et ux O'Brinn Resources, LLC 42.395.1470.0151 6/10/2080 Cardischerior Fields O'Brinn Resources, LLC 42.395.1470.0152 6/10/2080 Cardischerior Fields O'Brinn Resources, LLC 42.395.1470.0152 6/22005 Carlyllams Sheckelford C'Brinn Resources, LLC	0967-446	Texas	Robertson
42.395.1470.0155 522200 Veron Ray Watson et ux Sarah Exploration Company 42.395.1470.0154 4182005 Robert L White et ux Offeren Resources, LP 42.395.1470.0155 319200 Charles Ray Morehead et ux Offeren Resources, LLC 42.395.1470.0157 822005 Ama Ruth Heggie Hall Offeren Resources, LLC 42.395.1470.0158 822000 Dorothy Farances Garner Offeren Resources, LLC 42.395.1470.0159 822000 Dorothy Farances Garner Offeren Resources, LLC 42.395.1470.0159 822000 Loren Muff Offeren Resources, LLC 42.395.1470.0161 1011/2007 Roma Kuth Heggie Hall Offeren Resources, LLC 42.395.1470.0161 1011/2007 Roma Kuth Heggie Hall Offeren Resources, LLC 42.395.1470.0161 1011/2007 Roma Kuth L White et ux Offeren Resources, LLC 42.395.1470.0161 61/42007 Roma Kuth Hall Offeren Resources, LLC 42.395.1470.0163 62/2009 Sooth Assanceford Offeren Resources, LLC 42.395.1470.0165 63/2000 Sooth Assanceford Offeren Resources, LLC	1051-346	Texas	Robertson
42.395.14700.0156 4/18/2005 Robert L White et ux Offern Resources, LP 42.395.14700.0156 3/19/2007 Doug Morehead, et ux Offern Energy Company 42.395.14700.0156 8/12/200 Alria Skay Morehead et ux Offern Resources, LLC 42.395.14700.0157 8/22/200 Alma Ruh Hegpi Hall Offerin Resources, LLC 42.395.14700.0158 8/22/200 Glein Lose, Trustee Offerin Resources, LLC 42.395.14700.0150 8/22/200 Glein Lose, Trustee Offerin Resources, LLC 42.395.14700.0160 8/22/200 Glein Lose, Trustee Offerin Resources, LLC 42.395.14700.0161 10/11/2007 Barry Varin and Sharnon Varin Offerin Resources, LLC 42.395.14700.0161 10/11/2007 Barry Varin and Sharnon Varin Offerin Resources, LLC 42.395.14700.0161 10/11/2007 Rosal Sharkelford Offerin Resources, LLC 42.395.14700.0162 6/3/2009 Sozi Alan Shackelford Offerin Resources, LLC 42.395.14700.0163 6/3/2009 Sozi Alan Shackelford Offerin Resources, LLC 42.395.14700.0163 10/3/2009 Sozial Alan Shackelford Of	0984-401	Texas	Robertson
42.395.14700.0155 31/92007 Oblighe Resources, LLC 42.395.14700.0156 81/2008 Charles Ray Morehead et ux O'Brien Resources, LLC 42.395.14700.0158 82/2008 Alma Ruhl Heggle Hall O'Brien Resources, LLC 42.395.14700.0158 82/2008 Gordin Jones, Tustee O'Brien Resources, LLC 42.395.14700.0161 82/2008 Lern Jones, Tustee O'Brien Resources, LLC 42.395.14700.0161 82/2008 Lern Jones, Tustee O'Brien Resources, LLC 42.395.14700.0161 1011/2007 Berry Vann and Sharnon Varin O'Brien Resources, LLC 42.395.14700.0161 1011/2007 Berry Vann and Sharnon Varin O'Brien Resources, LLC 42.395.14700.0181 64/2009 Jona Shackeford Fields O'Brien Resources, LLC 42.395.14700.0181 64/2009 Jona Shackeford Fields O'Brien Resources, LLC 42.395.14700.0181 63/2009 Soot Alain Shackeford O'Brien Resources, LLC 42.395.14700.0181 63/2009 Soot Alain Shackeford O'Brien Resources, LLC 42.395.14700.0182 63/2009 Villiam Berry Jackson Jr. O'Brien Resources, LLC <t< td=""><td>0955-586</td><td>Texas</td><td>Robertson</td></t<>	0955-586	Texas	Robertson
42.395.1470.01516 8 I1/2008 Charles Ray Morehead et ux O'Brien Resources, LLC 42.395.1470.01517 8 J2/2008 Charles Ray Morehead et ux O'Brien Resources, LLC 42.395.1470.01518 8 J2/2008 Chord Transcer C'Brien Resources, LLC 42.395.1470.01519 8 J2/2008 Cloren Muth C'Brien Resources, LLC 42.395.1470.0161 1011/2007 Barry Anna and Shannon Varn O'Brien Resources, LLC 42.395.1470.0161 1011/2007 Ronald L White et ux O'Brien Energy Company 42.395.1470.0161 1014/2007 Ronald L White et ux O'Brien Energy Company 42.395.1470.0161 6 J4/200 Gons Shackeflord O'Brien Resources, LLC 42.395.1470.0161 6 J4/200 Scott Man Shackeflord O'Brien Resources, LLC 42.395.1470.0181 6 J4/200 Scott Man Shackeflord O'Brien Resources, LLC 42.395.1470.0181 6 J4/2000 Stophen Jyms Shackeflord O'Brien Resources, LLC 42.395.1470.0181 6 J4/2001 William Berry Jackson Jr. O'Brien Resources, LLC 42.395.1470.0182 10 J4/2007 William Berry Jackson Jr. O'Brien Re	0919-067	Texas	Robertson
42.395.14700.0157 8/2/2008 Mana Ruth Heggie Half O'Brien Resources, LLC 42.395.14700.0158 8/2/2008 Corolly Frances Garner O'Brien Resources, LLC 42.395.14700.0169 8/2/2008 Corolly Frances Garner O'Brien Resources, LLC 42.395.14700.0161 8/2/2008 Lorene Murff O'Brien Resources, LLC 42.395.14700.0161 10/14/2007 Barry Varn and Shannon Varn O'Brien Resources, LLC 42.395.14700.0161 6/4/2008 Josabcelford Fields O'Brien Resources, LLC 42.395.14700.0181 6/4/2009 Josabcelford Fields O'Brien Resources, LLC 42.395.14700.0183 6/2/2009 Gary Hilman Shackelford O'Brien Resources, LLC 42.395.14700.0184 6/3/2009 Soott Alan Shackelford O'Brien Resources, LLC 42.395.14700.0185 6/3/2009 Soott Alan Shackelford O'Brien Resources, LLC 42.395.14700.0184 10/2/2007 William Berry Jackson Jr. O'Brien Resources, LLC 42.395.14700.0185 10/2/2007 William Berry Jackson Jr. O'Brien Resources, LLC 42.395.14700.0186 10/12/2007 William Berry Jackson Jr. <t< td=""><td>0985-668</td><td>Texas</td><td>Robertson</td></t<>	0985-668	Texas	Robertson
42 395.14700.0158 8/2/2008 Glenn Jones, Trustee OBrien Resources, LLC 42 395.14700.0160 8/2/2008 Glenn Jones, Trustee OBrien Resources, LLC 42 395.14700.0161 8/1/2008 Barry Jone Murt OBrien Resources, LLC 42 395.14700.0161 10/11/2007 Bornal Machinery Company OBrien Energy Company 42 395.14700.0181 6/4/2009 OBrien Resources, LLC OBrien Resources, LLC 42 395.14700.0181 6/4/2009 Jone Shackelford Fields OBrien Resources, LLC 42 395.14700.0182 6/3/2009 Seliz Eizabeth Kerr OBrien Resources, LLC 42 395.14700.0183 6/3/2009 Selot Alan Shackelford OBrien Resources, LLC 42 395.14700.0184 6/3/2009 Slephen Lynn Shackelford OBrien Resources, LLC 42 395.14700.0185 6/3/2009 Slephen Lynn Shackelford OBrien Resources, LLC 42 395.14700.0186 10/2/2019 William Berry Jackson, Jr. OBrien Resources, LLC 42 395.14700.0189 10/14/2007 Splice Feergy Company OBrien Resources, LLC 42 395.14700.0180 10/16/2005 Splice Fee Workeland, Widow Wiremainderman to Charle	1047-608	Texas	Robertson
42 395.14700.0199 8/2/2008 Glen Jones, Trustele O'Brien Resources, LLC 42 395.14700.0161 10/11/2007 Briry Varn and Shannon Varn O'Brien Energy Company 42 395.14700.0161 10/11/2007 Ronald L. White et ux O'Brien Energy Company 42 395.14700.0162 10/14/2007 Ronald L. White et ux O'Brien Resources, LLC 42 395.14700.0181 6/3/2009 Jean Skackelford Fields O'Brien Resources, LLC 42 395.14700.0182 6/3/2009 Sery Hillman Shackelford O'Brien Resources, LLC 42 395.14700.0183 6/22/2009 Stophen Lyrn Shackelford O'Brien Resources, LLC 42 395.14700.0184 6/3/2009 Stophen Lyrn Shackelford O'Brien Resources, LLC 42 395.14700.0185 6/3/2009 Stophen Lyrn Shackelford O'Brien Resources, LLC 42 395.14700.0186 10/1/2009 Villiam Berry Jackson Jr. O'Brien Resources, LLC 42 395.14700.0189 10/14/2007 William Berry Jackson, Jr. O'Brien Resources, LLC 42 395.14700.0191 10/16/2005 Silly Faye Morehead, widow w/remainderman to Charles Ray Morehead O'Brien Resources, LLC 42 395.14700.0192	1063-382	Texas	Robertson
42 395 14700 0160 8 8/2/2008 Lorene Murff O'Brien Resources, LLC 42 395 14700 0161 10 11/12/07 Barry Vann and Shannon Vann O'Brien Energy Company 42 395 14700 0169 10 1/14/2007 Ronald L. White et ux O'Brien Energy Company 42 395 14700 0181 6/4/2009 Joan Shackelford Fields O'Brien Resources, LLC 42 395 14700 0181 6/3/2009 Leslie Elizabeth Kerr O'Brien Resources, LLC 42 395 14700 0184 6/23/2009 Gary Hillman Shackelford O'Brien Resources, LLC 42 395 14700 0185 6/3/2009 Soot Hand Shackelford O'Brien Resources, LLC 42 395 14700 0186 12 //2019 William Berry Jackson Jr. O'Brien Resources, LLC 42 395 14700 0188 10 //24/207 William Berry Jackson Jr. O'Brien Resources, LLC 42 395 14700 0189 10 //24/207 Brien Resources, LLC O'Brien Resources, LLC 42 395 14700 0189 10 //24/207 Brien Resources, LLC O'Brien Resources, LLC 42 395 14700 0191 10 //24/208 Sybe Faye Morehead, widow w/remainderman to Charles Ray Morehead O'Brien Resources, LLC 42 395 14700 0192 10 //2	1063-370	Texas	Robertson
42 395 14700 0161 10/11/2007 Barry Varn and Shannon Varn O'Brien Energy Company 42 395 14700 0181 6/4/2009 Jonal L. White et ux O'Brien Energy Company 42 395 14700 0181 6/4/2009 Joan Shackelford Fields O'Brien Resources, LLC 42 395 14700 0182 6/3/2009 Edis Elizabeth Kerr O'Brien Resources, LLC 42 395 14700 0184 6/3/2009 Scott Alan Shackelford O'Brien Resources, LLC 42 395 14700 0184 6/3/2009 Scott Alan Shackelford O'Brien Resources, LLC 42 395 14700 0185 6/3/2009 Stephen Lyrn Shackelford O'Brien Resources, LLC 42 395 14700 0186 12/7/2010 William Berry Jackson Jr. O'Brien Resources, LLC 42 395 14700 0189 10/24/2007 William Berry Jackson, Jr. O'Brien Resources, LLC 42 395 14700 0189 10/16/2008 Sple Faye Morehead, widow w/remainderman to Charles Ray Morehead O'Brien Resources, LLC 42 395 14700 0191 8/8/2008 Billy W. Bodiford O'Brien Resources, LLC 42 395 14700 0194 12/13/2007 Mary Beth While O'Brien Resources, LLC 42 395 14700 0195 10/14/2007	1059-261	Texas	Robertson
42 395 14700 0169 10/14/2007 Rodl L White et ux O'Brien Energy Company 42 395 14700 0181 6/4/209 Joan Shackelford Fields O'Brien Resources, LLC 42 395 14700 0183 6/3/209 Lesile Elizabeth Kerr O'Brien Resources, LLC 42 395 14700 0183 6/22/209 Gott Alan Shackelford O'Brien Resources, LLC 42 395 14700 0184 6/3/209 Stoht Alan Shackelford O'Brien Resources, LLC 42 395 14700 0186 6/3/209 Stoht Alan Shackelford O'Brien Resources, LLC 42 395 14700 0186 6/3/209 Stephen Lynn Shackelford O'Brien Resources, LLC 42 395 14700 0186 1/27/201 William Berry Jackson Jr. O'Brien Resources, LLC 42 395 14700 0188 1/27/201 Billiam Berry Jackson Jr. O'Brien Resources, LLC 42 395 14700 0189 7/30/2014 Bernadine Zajac Kublak O'Brien Resources, LLC 42 395 14700 0190 8/6/2008 Billy W. Bodiford O'Brien Resources, LLC 42 395 14700 0191 8/6/2008 Lay J. Berth White O'Brien Resources, LLC 42 395 14700 0192 8/6/2009 Aury Beth White O'Brien Resources	1059-263	Texas	Robertson
42.395.14700.0181 6/4/2009 Joan Shackelford Fields O'Brien Resources, LLC 42.395.14700.0182 6/3/2009 Leslie Elizabeth Kerr O'Brien Resources, LLC 42.395.14700.0183 6/2/2009 Sort Alan Shackelford O'Brien Resources, LLC 42.395.14700.0184 6/3/2009 Stott Alan Shackelford O'Brien Resources, LLC 42.395.14700.0185 6/3/2009 Stephen Lynn Shackelford O'Brien Resources, LLC 42.395.14700.0186 12/7/2010 William Berry Jackson Jr. O'Brien Resources, LLC 42.395.14700.0188 10/24/2007 William Berry Jackson Jr. O'Brien Resources, LLC 42.395.14700.0189 7/30/2014 Bernadine Zajac Kubliak O'Brien Resources, LLC 42.395.14700.0190 10/16/2008 Sibly E-yes Morehead, widow w/remainderman to Charles Ray Morehead O'Brien Resources, LLC 42.395.14700.0191 8/6/2008 Liny J. Bodiford O'Brien Resources, LLC 42.395.14700.0194 12/13/2007 May Beth White O'Brien Resources, LLC 42.395.14700.0195 10/14/2007 R. H. Leonard White O'Brien Resources, LLC 42.395.14700.0196 3/25/2008	1020-716	Texas	Robertson
42.395.14700.0182 6/3/2009 Leslie Elizabeth Kerr O'Brien Resources, LLC 42.395.14700.0183 6/22/2009 Gory Hillman Shackelford O'Brien Resources, LLC 42.395.14700.0184 6/23/2009 Sott Alan Shackelford O'Brien Resources, LLC 42.395.14700.0185 6/3/2009 Stephen Lynn Shackelford O'Brien Resources, LLC 42.395.14700.0186 12/7/2010 William Berry Jackson Jr. O'Brien Resources, LLC 42.395.14700.0188 10/24/2007 William Berry Jackson, Jr. O'Brien Resources, LLC 42.395.14700.0190 10/16/2008 Syble Faye Morehead, widow w/remainderman to Charles Ray Morehead O'Brien Resources, LLC 42.395.14700.0191 8/6/2008 Billy W. Bodiford O'Brien Resources, LLC 42.395.14700.0192 8/6/2008 Billy W. Bodiford O'Brien Resources, LLC 42.395.14700.0193 12/13/2007 Aurry J. Bodiford O'Brien Resources, LLC 42.395.14700.0194 12/13/2007 Aurry J. Bodiford O'Brien Resources, LLC 42.395.14700.0195 10/14/2007 R. L. Leonard White O'Brien Resources, LLC 42.395.14700.0196 3/25/200	1015-429	Texas	Robertson
42.395.14700.0183 6/22/209 Gary Hillman Shackelford O'Brien Resources, LLC 42.395.14700.0184 6/23/209 Scott Alan Shackelford O'Brien Resources, LLC 42.395.14700.0186 6/3/2009 Stephen Lynn Shackelford O'Brien Resources, LLC 42.395.14700.0186 12/7/2010 William Berry Jackson Jr. O'Brien Resources, LLC 42.395.14700.0188 7/30/2014 Bernadine Zajac Kubiak O'Brien Resources, LLC 42.395.14700.0190 10/16/200 Syble Faye Morehead, widow w/remainderman to Charles Ray Morehead O'Brien Resources, LLC 42.395.14700.0191 8/6/2008 Billy W. Bodiford O'Brien Resources, LLC 42.395.14700.0192 8/6/2008 Larry J. Bodiford O'Brien Resources, LLC 42.395.14700.0193 10/14/2007 Mary Beth White O'Brien Resources, LLC 42.395.14700.0194 12/13/2007 R. L. Leonard White O'Brien Resources, LLC 42.395.14700.0200 7/29/200 Archie Lee Ramsey O'Brien Resources, LLC 42.395.14700.0202 9/5/2014 Rex Sidney Jackson O'Brien Resources, LLC 42.395.14700.0203 9/5/2014 Rex Sidney Jac	1082-513	Texas	Robertson
42.395.14700.0184 6/23/2009 Scott Alan Shackelford O'Brien Resources, LLC 42.395.14700.0185 6/3/2009 Stephen Lynn Shackelford O'Brien Resources, LLC 42.395.14700.0186 12/7/201 William Berry Jackson Jr. O'Brien Resources, LLC 42.395.14700.0188 10/24/2007 William Berry Jackson, Jr. O'Brien Resources, LLC 42.395.14700.0189 7/30/2014 Bernadine Zajac Kubiak O'Brien Resources, LLC 42.395.14700.0190 10/16/2008 Sibly W. Bodiford O'Brien Resources, LLC 42.395.14700.0191 8/6/2008 Larry J. Bodiford O'Brien Resources, LLC 42.395.14700.0192 8/6/2008 Larry J. Bodiford O'Brien Resources, LLC 42.395.14700.0193 10/14/200 May Beth White O'Brien Resources, LLC 42.395.14700.0194 12/13/200 May Beth White O'Brien Resources, LLC 42.395.14700.0200 7/29/2008 Archie Lee Ramsey O'Brien Resources, LLC 42.395.14700.0202 9/5/2014 Rex Sidney Jackson O'Brien Resources, LLC 42.395.14700.0203 10/10/2014 Candace Brantner Johnson et vir O'Brien Resou	1078-757	Texas	Robertson
42.395.14700.0185 6/3/2009 Stephen Lynn Shackelford O'Brien Resources, LLC 42.395.14700.0186 12/7/2010 William Berry Jackson Jr. O'Brien Resources, LLC 42.395.14700.0188 10/24/2007 William Berry Jackson, Jr. O'Brien Energy Company 42.395.14700.0189 7/30/2014 Bernadine Zajac Kubiak O'Brien Resources, LLC 42.395.14700.0190 10/16/2008 Syble Faye Morehead, widow w/remainderman to Charles Ray Morehead O'Brien Resources, LLC 42.395.14700.0191 8/6/2008 Billy W. Bodiford O'Brien Resources, LLC 42.395.14700.0192 8/6/2008 Larry J. Bodiford O'Brien Resources, LLC 42.395.14700.0194 12/13/2007 Mary Beth White O'Brien Resources, LLC 42.395.14700.0195 10/14/2007 R. H. Leonard White O'Brien Resources, LLC 42.395.14700.0196 3/25/2008 Robert L. White et ux O'Brien Resources, LLC 42.395.14700.0200 7/29/2008 Archie Lee Ramsey O'Brien Resources, LLC 42.395.14700.0202 9/5/2014 Rex Sidney Jackson O'Brien Resources, LLC 42.395.14700.0203 10/10/2014 Candace Brantner Johnson et vir O'Brien Resources, LLC	1078-761	Texas	Robertson
42.395.14700.0186 12/7/2010 William Berry Jackson Jr. O'Brien Resources, LLC 42.395.14700.0189 7/30/2014 Bernadine Zajac Kubiak O'Brien Resources, LLC 42.395.14700.0190 10/16/2008 Syble Faye Morehead, widow w/remainderman to Charles Ray Morehead O'Brien Resources, LLC 42.395.14700.0191 8/6/2008 Billy W. Bodiford O'Brien Resources, LLC 42.395.14700.0192 8/6/2008 Larry J. Bodiford O'Brien Resources, LLC 42.395.14700.0194 12/13/2007 Mary Beth White O'Brien Resources, LLC 42.395.14700.0195 10/14/2007 R. H. Leonard White O'Brien Resources, LLC 42.395.14700.0196 3/25/2008 Robert L. White et ux O'Brien Resources, LLC 42.395.14700.0200 7/29/2008 Archie Lee Ramsey O'Brien Resources, LLC 42.395.14700.0202 9/5/2014 Rex Sidney Jackson O'Brien Resources, LLC 42.395.14700.0203 10/10/2014 Candace Brantner Johnson et vir O'Brien Resources, LLC 42.395.14700.0204 9/5/2014 Lean Ellen Hunter O'Brien Resources, LLC	1078-759	Texas	Robertson
42.395.14700.0188 10/24/2007 William Berry Jackson, Jr. O'Brien Energy Company 42.395.14700.0189 7/30/2014 Bernadine Zajac Kubiak O'Brien Resources, LLC 42.395.14700.0190 10/16/2008 Syble Faye Morehead, widow w/remainderman to Charles Ray Morehead O'Brien Resources, LLC 42.395.14700.0191 8/6/2008 Billy W. Bodiford O'Brien Resources, LLC 42.395.14700.0192 8/6/2008 Larry J. Bodiford O'Brien Resources, LLC 42.395.14700.0194 12/13/2007 Mary Beth White O'Brien Resources, LLC 42.395.14700.0195 10/14/2007 R. H. Leonard White O'Brien Resources, LLC 42.395.14700.0200 7/29/2008 Robert L. White et ux O'Brien Resources, LLC 42.395.14700.0202 9/5/2014 Rex Sidney Jackson O'Brien Resources, LLC 42.395.14700.0203 10/10/2014 Candace Brantner Johnson et vir O'Brien Resources, LLC 42.395.14700.0204 9/5/204 Jean Ellen Hunter O'Brien Resources, LLC	1078-768	Texas	Robertson
42.395.14700.0189 7/30/2014 Bernadine Zajac Kubiak O'Brien Resources, LLC 42.395.14700.0190 10/16/2008 Syble Faye Morehead, widow w/remainderman to Charles Ray Morehead O'Brien Resources, LLC 42.395.14700.0191 8/6/2008 Billy W. Bodiford O'Brien Resources, LLC 42.395.14700.0192 8/6/2008 Larry J. Bodiford O'Brien Resources, LLC 42.395.14700.0194 12/13/2007 Mary Beth White O'Brien Resources, LLC 42.395.14700.0195 10/14/2007 R. H. Leonard White O'Brien Resources, LLC 42.395.14700.0196 3/25/2008 Robert L. White et ux O'Brien Resources, LLC 42.395.14700.0200 7/29/2008 Archie Lee Ramsey O'Brien Resources, LLC 42.395.14700.0202 9/5/2014 Rex Sidney Jackson O'Brien Resources, LLC 42.395.14700.0203 10/10/2014 Candace Brantner Johnson et vir O'Brien Resources, LLC 42.395.14700.0204 9/5/2014 Jean Ellen Hunter O'Brien Resources, LLC	1124-731	Texas	Robertson
42.395.14700.0190 10/16/2008 Syble Faye Morehead, widow w/remainderman to Charles Ray Morehead O'Brien Resources, LLC 42.395.14700.0191 8/6/2008 Billy W. Bodiford O'Brien Resources, LLC 42.395.14700.0192 8/6/2008 Larry J. Bodiford O'Brien Resources, LLC 42.395.14700.0194 12/13/2007 Mary Beth White O'Brien Resources, LLC 42.395.14700.0195 10/14/2007 R. H. Leonard White O'Brien Resources, LLC 42.395.14700.0200 7/29/2008 Robert L. White et ux O'Brien Resources, LLC 42.395.14700.0200 7/29/2008 Archie Lee Ramsey O'Brien Resources, LLC 42.395.14700.0202 9/5/2014 Rex Sidney Jackson O'Brien Resources, LLC 42.395.14700.0203 10/10/2014 Candace Branther Johnson et vir O'Brien Resources, LLC 42.395.14700.0204 9/5/2014 Lean Ellen Hunter O'Brien Resources, LLC	1038-192	Texas	Robertson
42.395.14700.0191 8/6/2008 Billy W. Bodiford O'Brien Resources, LLC 42.395.14700.0192 8/6/2008 Larry J. Bodiford O'Brien Resources, LLC 42.395.14700.0194 12/13/2007 Mary Beth White O'Brien Resources, LLC 42.395.14700.0195 10/14/2007 R. H. Leonard White O'Brien Energy Company 42.395.14700.0196 3/25/2008 Robert L. White et ux O'Brien Resources, LLC 42.395.14700.0200 7/29/2008 Archie Lee Ramsey O'Brien Resources, LLC 42.395.14700.0202 9/5/2014 Rex Sidney Jackson O'Brien Resources, LLC 42.395.14700.0203 10/10/2014 Candace Branther Johnson et vir O'Brien Resources, LLC 42.395.14700.0204 9/5/2014 Jean Ellen Hunter O'Brien Resources, LLC	1254-500	Texas	Robertson
42.395.14700.0192 8/6/2008 Larry J. Bodiford O'Brien Resources, LLC 42.395.14700.0194 12/13/2007 Mary Beth White O'Brien Resources, LLC 42.395.14700.0195 10/14/2007 R. H. Leonard White O'Brien Energy Company 42.395.14700.0196 3/25/2008 Robert L. White et ux O'Brien Resources, LLC 42.395.14700.0200 7/29/2008 Archie Lee Ramsey O'Brien Resources, LLC 42.395.14700.0202 9/5/2014 Rex Sidney Jackson O'Brien Resources, LLC 42.395.14700.0203 10/10/2014 Candace Branther Johnson et vir O'Brien Resources, LLC 42.395.14700.0204 9/5/2014 Jean Ellen Hunter O'Brien Resources, LLC	1055-249	Texas	Robertson
42.395.14700.0194 12/13/2007 Mary Beth White O'Brien Resources, LLC 42.395.14700.0195 10/14/2007 R. H. Leonard White O'Brien Energy Company 42.395.14700.0196 3/25/2008 Robert L. White et ux O'Brien Resources, LLC 42.395.14700.0200 7/29/2008 Archie Lee Ramsey O'Brien Resources, LLC 42.395.14700.0202 9/5/2014 Rex Sidney Jackson O'Brien Resources, LLC 42.395.14700.0203 10/10/2014 Candace Branther Johnson et vir O'Brien Resources, LLC 42.395.14700.0204 9/5/2014 Jean Ellen Hunter O'Brien Resources, LLC	1051-332	Texas	Robertson
42.395.14700.0195 10/14/2007 R. H. Leonard White O'Brien Energy Company 42.395.14700.0196 3/25/2008 Robert L. White et ux O'Brien Resources, LLC 42.395.14700.0200 7/29/2008 Archie Lee Ramsey O'Brien Resources, LLC 42.395.14700.0202 9/5/2014 Rex Sidney Jackson O'Brien Resources, LLC 42.395.14700.0203 10/10/2014 Candace Branther Johnson et vir O'Brien Resources, LLC 42.395.14700.0204 9/5/2014 Jean Ellen Hunter O'Brien Resources, LLC	1051-334	Texas	Robertson
42.395.14700.0196 3/25/2008 Robert L. White et ux O'Brien Resources, LLC 42.395.14700.0200 7/29/2008 Archie Lee Ramsey O'Brien Resources, LLC 42.395.14700.0202 9/5/2014 Rex Sidney Jackson O'Brien Resources, LLC 42.395.14700.0203 10/10/2014 Candace Brantner Johnson et vir O'Brien Resources, LLC 42.395.14700.0204 9/5/2014 Jean Ellen Hunter O'Brien Resources, LLC	1024-474	Texas	Robertson
42.395.14700.0200 7/29/2008 Archie Lee Ramsey O'Brien Resources, LLC 42.395.14700.0202 9/5/2014 Rex Sidney Jackson O'Brien Resources, LLC 42.395.14700.0203 10/10/2014 Candace Brantner Johnson et vir O'Brien Resources, LLC 42.395.14700.0204 9/5/2014 Jean Ellen Hunter O'Brien Resources, LLC	1015-427	Texas	Robertson
42.395.14700.0202 9/5/2014 Rex Sidney Jackson O'Brien Resources, LLC 42.395.14700.0203 10/10/2014 Candace Brantner Johnson et vir O'Brien Resources, LLC 42.395.14700.0204 9/5/2014 Jean Ellen Hunter O'Brien Resources, LLC	1034-429	Texas	Robertson
42.395.14700.0203 10/10/2014 Candace Brantner Johnson et vir O'Brien Resources, LLC 42.395.14700.0204 9/5/2014 Jean Ellen Hunter O'Brien Resources, LLC	1051-342	Texas	Robertson
42.395.14700.0204 9/5/2014 Jean Ellen Hunter O'Brien Resources, LLC	1254-479	Texas	Robertson
	1254-485	Texas	Robertson
42.395.14700.0205 9/22/2014 Charles Ray Morehead O'Brien Resources, LLC	1254-482	Texas	Robertson
	1254-497	Texas	Robertson
42.395.14700.0207 10/23/2008 Jean Ann McKinley O'Brien Resources, LLC	1055-654	Texas	Robertson
42.395.14700.0208 11/7/2008 Charles Ray Morehead et ux O'Brien Resources, LLC	1057-476	Texas	Robertson
42.395.14700.0211 11/14/2007 Ethel Bodiford et al O'Brien Energy Company	1020-717	Texas	Robertson
42.395.14700.0213 7/1/2008 Nancy Severs O'Brien Resources, LLC	1055-231	Texas	Robertson
42.395.14700.0214 10/19/2011 Randall W. Abel O'Brien Resources, LLC	1152-304	Texas	Robertson
42.395.14700.0215 3/25/2008 Susan Gafford O'Brien Resources, LLC	1038-182	Texas	Robertson
42.395.14700.0216 3/25/2008 Michael F. Jund O'Brien Resources, LLC	1038-153	Texas	Robertson
42.395.14700.0217 5/6/2008 Lucy Pfluger O'Brien Resources, LLC	1047-588	Texas	Robertson
42.395.14700.0218 5/6/2008 Mary Pfluger O'Brien Resources, LLC	1047-586	Texas	Robertson
42.395.14700.0219 3/28/2008 Dorothy Pitman O'Brien Resources, LLC	1038-184	Texas	Robertson

Lease #	Date	Lessor	Lessee	Book-Page	State_	County
42.395.14700.0220	5/1/2008	Waldine Pfluger Rust	O'Brien Resources, LLC	1043-307	Texas	Robertson
42.395.14700.0224	12/31/2008	Shirley McKinley Scott	O'Brien Resources, LLC	1064-781	Texas	Robertson
42.395.14700.0237	6/24/2008	Donald Brannon	O'Brien Resources, LLC	1043-243	Texas	Robertson
42.395.14700.0260	8/14/2008	Glenn E. Herzog	O'Brien Resources, LLC	1051-350	Texas	Robertson
42.395.14700.0263	6/14/2006	Denyse K. Du Puy	Sarah Exploration Company	0959-484	Texas	Robertson
42.395.14700.0264	6/14/2006	Suzanne K. Du Puy	Sarah Exploration Company	0959-487	Texas	Robertson
42.395.14700.0265	4/9/2008	George Markham Jr.	O'Brien Resources, LLC	1034-419	Texas	Robertson
42.395.14700.0266	5/7/2008	Brenda Minces	O'Brien Resources, LLC	1043-287	Texas	Robertson
42.395.14700.0267	5/15/2008	Candace Wall et vir	O'Brien Resources, LLC	1038-410	Texas	Robertson
42.395.14700.0268	3/28/2008	Frances Muzyka Gerik	O'Brien Resources, LLC	1034-425	Texas	Robertson
42.395.14700.0269	3/28/2008	Jerry Wayne Kocian	O'Brien Resources, LLC Sarah Exploration Company	1034-427	Texas	Robertson
42.395.14700.0270 42.395.14700.0271	8/15/2005 9/16/2005		Sarah Exploration Company	0933-438 0942-641	Texas	Robertson Robertson
42.395.14700.0271		Patricia Cole	Sarah Exploration Company	0959-474	Texas	Robertson
42.395.14700.0272	9/1/2005	Judy Engelberg	Sarah Exploration Company	0933-465	Texas	Robertson
42.395.14700.0274	2/15/2008	Arnold Glick	O'Brien Resources, LLC	1034-393	Texas	Robertson
42.395.14700.0275	2/15/2008		O'Brien Resources, LLC	1034-397	Texas	Robertson
42.395.14700.0276		Roger Glick	O'Brien Resources, LLC	1034-389	Texas	Robertson
42.395.14700.0277	10/11/2005	Sue Houghton	Sarah Exploration Company	0942-680	Texas	Robertson
42.395.14700.0278	4/15/2008	Jane R. Lancaster	O'Brien Resources, LLC	1043-331	Texas	Robertson
42.395.14700.0279	9/16/2005	Susan Cole Ortner	Sarah Exploration Company	0943-618	Texas	Robertson
42.395.14700.0280	11/4/2005	Petroleum Enterprises, Inc.	Sarah Exploration Company	0933-389	Texas	Robertson
42.395.14700.0281	4/14/2008	James I. Riddle III	O'Brien Resources, LLC	1038-149	Texas	Robertson
42.395.14700.0282	8/15/2005	Sue S. Schaefer	Sarah Exploration Company	0933-561	Texas	Robertson
42.395.14700.0283	9/1/2005	Margaret Ann Solomon	Sarah Exploration Company	0942-647	Texas	Robertson
42.395.14700.0284	9/1/2005	Mary L. Solomon	Sarah Exploration Company	0933-567	Texas	Robertson
42.395.14700.0285	8/15/2005		Sarah Exploration Company	0933-579	Texas	Robertson
42.395.14700.0286	3/28/2008	Frances Marks Travis Family Trust	O'Brien Resources, LLC	1038-168	Texas	Robertson
42.395.14700.0287	8/4/2005	Mary G. Watson	Sarah Exploration Company	0933-616	Texas	Robertson
42.395.14700.0288 42.395.14700.0289	4/30/2007 3/5/2007	Peggy Aileen Wolf Victoria Bird	O'Brien Energy Company O'Brien Energy Company	0989-692 0988-234	Texas	Robertson Robertson
42.395.14700.0299	4/17/2007		O'Brien Energy Company	0989-704	Texas	Robertson
42.395.14700.0291		Beverly F. Merritt	Prime Resources, LP	0918-556	Texas	Robertson
42.395.14700.0292	4/27/2005		Prime Resources, LP	0918-614	Texas	Robertson
42.395.14700.0293		Henry Spindle, Jr.	Prime Resources, LP	0943-475	Texas	Robertson
42.395.14700.0294	4/27/2005		Prime Resources, LP	0943-480	Texas	Robertson
42.395.14700.0295	3/2/2007	Robert W. Spindle, et ux	O'Brien Energy Company	1011-758	Texas	Robertson
42.395.14700.0296	8/11/2008	Richard L. Brunette	O'Brien Resources, LLC	1047-590	Texas	Robertson
42.395.14700.0297	6/3/2008	Laverne Dean Blackwell	O'Brien Resources, LLC	1051-352	Texas	Robertson
42.395.14700.0298	7/13/2007	Geneva Williams et vir	O'Brien Energy Company	0999-409	Texas	Robertson
42.395.14700.0299	6/10/2008	Charles Danny Cottingham	Eagle Oil & Gas Co.	1039-446	Texas	Robertson
42.395.14700.0300	6/14/2007	Marie Abraham et al	O'Brien Energy Company	0995-527	Texas	Robertson
42.395.14700.0301	6/10/2008	Leila Patricia Cottingham Thorne	Eagle Oil & Gas Co.	1041-005	Texas	Robertson
42.395.14700.0309	5/7/2008	Heards Prairie Baptist Church	O'Brien Resources, LLC	1043-247	Texas	Robertson
42.395.14700.0310	3/17/2003	Joe L. Muzyka	Devon Energy Production	0834-737	Texas	Robertson
42.395.14700.0311	3/17/2003		Devon Energy Production	0832-791	Texas	Robertson
42.395.14700.0312	11/4/2003	Sophie Muzyka Krysinski	Devon Energy Production	0854-327	Texas	Robertson
42.395.14700.0313 42.395.14700.0314	2/27/2003		Devon Energy Production Sarah Exploration Company	0854-208 0933-432	Texas Texas	Robertson Robertson
42.395.14700.0314	2/27/2003	Carol Davis Brown Margaret Ann Christofferson	Sarah Exploration Company Devon Energy Production	0933-432	Texas	Robertson
42.395.14700.0316		Nora Rita Cook	Sarah Exploration Company	0933-444	Texas	Robertson
42.395.14700.0317		Opal Cronin	Sarah Exploration Company	0933-450	Texas	Robertson
42.395.14700.0318		Daughters of Charity of St. Vincent DePaul of Indiana, Inc.	Sarah Exploration Company	0941-130	Texas	Robertson
42.395.14700.0319		Mary Ann Davis Lee	Sarah Exploration Company	0933-513	Texas	Robertson
42.395.14700.0320		John S. Lucey et al	Sarah Exploration Company	0959-509	Texas	Robertson
42.395.14700.0321	10/13/2005	Judy Davis Masi	Sarah Exploration Company	0933-522	Texas	Robertson
42.395.14700.0322	10/7/2005	James T. McNamara	Sarah Exploration Company	0933-537	Texas	Robertson
42.395.14700.0323	2/27/2003	Bridget Mosley	Devon Energy Production	0835-678	Texas	Robertson
42.395.14700.0324	8/14/2005	Mary Ott	Sarah Exploration Company	0933-552	Texas	Robertson
42.395.14700.0325	8/13/2005	Doris Gwin Townsend Estate	Sarah Exploration Company	0933-459	Texas	Robertson
42.395.14700.0326	11/10/2003	Betty L. Baird	Devon Energy Production	0854-672	Texas	Robertson
42.395.14700.0327		Leocadia Czajkowski	Devon Energy Production	0829-619	Texas	Robertson
42.395.14700.0328		Paul Raymond Hopcus et ux	Devon Energy Production	0854-678	Texas	Robertson
42.395.14700.0329		Patricia Hopcus-Jorski	Devon Energy Production	0854-675	Texas	Robertson
42.395.14700.0330	3/17/2003	Blake Smith, Jr. et ux	Devon Energy Production	0835-687	Texas	Robertson

Lease #	Date	Lessor	Lessee	Book-Page	State	County
42.395.14700.0331	8/6/2005	T. W. Whaley, Jr.	Sarah Exploration Company	0933-631	Texas	Robertson
42.395.14700.0332	2/12/2008	Daniel Muzyka	O'Brien Resources, LLC	1029-648	Texas	Robertson
42.395.14700.0333	2/12/2008	Kie Muzyka	O'Brien Resources, LLC	1029-644	Texas	Robertson
42.395.14700.0334	2/12/2008	Louis T. Muzyka	O'Brien Resources, LLC	1029-646	Texas	Robertson
42.395.14700.0335	2/12/2008	Raymond Muzyka	O'Brien Resources, LLC	1029-651	Texas	Robertson
42.395.14700.0336	2/13/2008	Charles R. Yezak	O'Brien Resources, LLC	1029-653	Texas	Robertson
42.395.14700.0337	2/13/2008	Leon D. Yezak Trust	O'Brien Resources, LLC	1034-431	Texas	Robertson
42.395.14700.0338	2/11/2008	Billy Joe Muzyka et ux (now deceased)	O'Brien Resources, LLC	1029-655	Texas	Robertson
42.395.14700.0339	2/21/2003	Jerry Wayne Kocian	Devon Energy Production	0832-092	Texas	Robertson
42.395.14700.0340	6/20/2006	Eula Mae Allen	Sarah Exploration Company	0959-478	Texas	Robertson
42.395.14700.0341	6/28/2006	Mary Margaret Bletsch	Sarah Exploration Company	0967-425	Texas	Robertson
42.395.14700.0342	7/28/2006	Michael J. Bletsch	Sarah Exploration Company	0967-443	Texas	Robertson
42.395.14700.0343	7/28/2006	Lemuel Buddy Evans	Sarah Exploration Company	0967-443	Texas	Robertson
42.395.14700.0344	6/26/2006	Cody L. McKenney	Sarah Exploration Company	0959-499	Texas	Robertson
42.395.14700.0345	6/26/2006	Gary A. McKenney	Sarah Exploration Company	0957-408	Texas	Robertson
42.395.14700.0346	6/26/2006	Shane A. McKenney	Sarah Exploration Company	0959-502	Texas	Robertson
42.395.14700.0347	7/12/2005	Albert Ruiz	Prime Resources, LP	0933-710	Texas	Robertson
42.395.14700.0348	6/28/2006	Nancy Bletsch Skinner	Sarah Exploration Company	0999-038	Texas	Robertson
42.395.14700.0349	2/1/2005	Charles S. Stilson et ux	Prime Resources, LP	0918-742	Texas	Robertson
42.395.14700.0350	8/24/2005	James S. Thompson	Prime Resources, LP	0933-732	Texas	Robertson
42.395.14700.0351	8/24/2005	John H. Thompson	Prime Resources, LP	0933-737	Texas	Robertson
42.395.14700.0352	6/26/2006	Sunni McKenney West	Sarah Exploration Company	0957-399	Texas	Robertson
42.395.14700.0353	9/24/2007	Sue Funderburk	O'Brien Energy Company	1008-086	Texas	Robertson
42.395.14700.0354	9/24/2007	Bill R. Swingler, et ux	O'Brien Energy Company	1008-344	Texas	Robertson
42.395.14700.0355	7/21/2007	The Charles Z. Renfroe and Nella J. Renfroe Revocable Living Trust	O'Brien Energy Company	0999-400	Texas	Robertson
42.395.14700.0356	2/5/2008	Hubert Brantner, dealing in his sole and separate property	O'Brien Resources, LLC	1024-497	Texas	Robertson
42.395.14700.0357	5/15/2007	Candace Elizabeth Wall	O'Brien Energy Company	0992-028	Texas	Robertson
42.395.14700.0358	5/21/2007	Danny Oldham et ux	O'Brien Energy Company	0992-032	Texas	Robertson
42.395.14700.0359	6/30/2007	Evadell Hines	O'Brien Energy Company	0995-532	Texas	Robertson
42.395.14700.0360	6/2/2007	Frankie Lee Wallace	O'Brien Energy Company	0992-010	Texas	Robertson
42.395.14700.0361	6/30/2007	William W. Wallace	O'Brien Energy Company	0995-535	Texas	Robertson
42.395.14700.0365	8/18/2005	Lloyd M. Frey	Sarah Exploration Company	0933-471	Texas	Robertson
42.395.14700.0366	12/5/2005	J. D. Minerals	Sarah Exploration Company	0945-656	Texas	Robertson
42.395.14700.0367	11/20/2005	Shirley Gouge	Sarah Exploration Company	0945-635	Texas	Robertson
42.395.14700.0368	11/20/2005	Daniel Wasik	Sarah Exploration Company	0933-401	Texas	Robertson
42.395.14700.0369	10/1/2005	Murlene G. Swingler Trust	Sarah Exploration Company	0933-549	Texas	Robertson
42.395.14700.0370	1/5/2008	Suzanne Swingler Anthony aka Suize Anthony	O'Brien Resources, LLC	1020-724	Texas	Robertson
42.395.14700.0371	1/5/2008	Jimmy Marvin Swingler	O'Brien Resources, LLC	1018-253	Texas	Robertson
42.395.14700.0372	1/5/2008	Shea Graham Tyra	O'Brien Resources, LLC	1020-728	Texas	Robertson
42.395.14700.0373	10/8/2011	Carl Owen Dyer, Jr.	O'Brien Resources, LLC	1154-083	Texas	Robertson
42.395.14700.0374	5/12/2008	Cecil B. McLeod	O'Brien Resources, LLC	1043-245	Texas	Robertson
42.395.14700.0375	6/7/2005	Mike Tolleson	Sarah Exploration Company	0933-591	Texas	Robertson
42.395.14700.0376		Lona Beth Miller Family Trust	Sarah Exploration Company	0933-540	Texas	Robertson
42.395.14700.0377		Shirley Ann Shackelford	Sarah Exploration Company	0933-564	Texas	Robertson
42.395.14700.0378	9/14/2005	J. V. Ware, Jr. (now deceased)	Sarah Exploration Company	0933-613	Texas	Robertson
42.395.14700.0379		Lona Beth Miller Family Trust	O'Brien Resources, LLC	1057-483	Texas	Robertson
42.395.14700.0380		Shirley Ann Shackelford	O'Brien Resources, LLC	1059-265	Texas	Robertson
42.395.14700.0381		J.V. Ware, Jr. (now deceased)	O'Brien Resources, LLC	1057-480	Texas	Robertson
42.395.14700.0382		Ruby Oldham	O'Brien Energy Company	0985-678	Texas	Robertson
42.395.14700.0383		Mary Basham Melcer	O'Brien Energy Company	0985-682	Texas	Robertson
42.395.14700.0384	5/8/2007	Suzanne Swingler Anthony aka Suize Anthony	O'Brien Energy Company	0992-022	Texas	Robertson
42.395.14700.0385	5/1/2007	Ray Green	O'Brien Energy Company	0988-224	Texas	Robertson
42.395.14700.0386	5/3/2007	Katherine Ann Miller	O'Brien Energy Company	0988-228	Texas	Robertson
42.395.14700.0387	5/1/2007	Aubrey Wayne Cude	O'Brien Energy Company	0988-231	Texas	Robertson
42.395.14700.0388	5/1/2007	Billy J. Green	O'Brien Energy Company	0988-231	Texas	Robertson
42.395.14700.0390		Kenneth Bell et ux	O'Brien Energy Company	1018-255	Texas	Robertson
42.395.14700.0391		William Eugene Campbell	Sarah Exploration Company	0942-656	Texas	Robertson
42.395.14700.0392			Sarah Exploration Company	0955-608	Texas	Robertson
42.395.14700.0393		Laurie L. Conaway	Sarah Exploration Company	0955-612	Texas	Robertson
42.395.14700.0394	1/16/2006	John Paul Cooper	Sarah Exploration Company	0942-659	Texas	Robertson
42.395.14700.0395	12/28/2005	Ramona Conaway Cooper	Sarah Exploration Company	0942-662	Texas	Robertson
42.395.14700.0396	1/16/2006	Donna L. Fyffe	Sarah Exploration Company	0942-668	Texas	Robertson
42.395.14700.0397	12/16/2005	Bailey J. McDaniel, Deceased	Sarah Exploration Company	0942-684	Texas	Robertson
42.395.14700.0398	12/17/2005	Joseph T. McDaniel	Sarah Exploration Company	0942-687	Texas	Robertson
42.395.14700.0399	12/26/2005	Kerry L McDaniel	Sarah Exploration Company	0942-650	Texas	Robertson

Lease #	Date	Lessor	Lessee	Book-Page	State	County
42.395.14700.0400	12/17/2005	Lamar McDaniel, Jr.	Sarah Exploration Company	0942-690	Texas	Robertson
42.395.14700.0401	12/26/2005	Vernon C. McDaniel	Sarah Exploration Company	0942-693	Texas	Robertson
42.395.14700.0402	1/14/2006	Victor S. McDaniel	Sarah Exploration Company	0942-638	Texas	Robertson
42.395.14700.0403	12/28/2005	Sharon Conaway Neff	Sarah Exploration Company	0942-696	Texas	Robertson
42.395.14700.0404	12/27/2005	Richard Porter	Sarah Exploration Company	0942-699	Texas	Robertson
42.395.14700.0408	11/12/2007	Mary Jo Reynolds	O'Brien Energy Company	1015-423	Texas	Robertson
42.395.14700.0409		Linda McDaniel Ringwald	Sarah Exploration Company	0942-702	Texas	Robertson
42.395.14700.0410		Kathleen D. Robb	Sarah Exploration Company	0945-644	Texas	Robertson
42.395.14700.0411		William E. Wegener	Sarah Exploration Company	0945-638	Texas	Robertson
42.395.14700.0412		Ann McDaniel Young	Sarah Exploration Company	0942-705	Texas	Robertson
42.395.14700.0413		Allen Jack Cameron Allen Wade Cameron	Sarah Exploration Company Sarah Exploration Company	0942-644	Texas	Robertson
42.395.14700.0414 42.395.14700.0415		George Edward Cameron	Sarah Exploration Company	0955-604 0942-653	Texas	Robertson Robertson
42.395.14700.0416		Susan Jean Cameron Chiachi	Sarah Exploration Company	0955-575	Texas	Robertson
42.395.14700.0417		Gloria Cameron Davis	Sarah Exploration Company	0942-665	Texas	Robertson
42.395.14700.0418		Mary Jo Reynolds	O'Brien Resources, LLC	1060-387	Texas	Robertson
42.395.14700.0419	10/31/2007	Joe Bailey et ux	O'Brien Energy Company	1018-267	Texas	Robertson
42.395.14700.0420	10/31/2007	Jim Bailey et ux	O'Brien Energy Company	1018-266	Texas	Robertson
42.395.14700.0421	11/24/2008	Mobil Producing Texas and New Mexico, Inc.	O'Brien Resources, LLC	1061-755	Texas	Robertson
42.395.14700.0422	5/15/2007	Texinia Corporation, R. J. Bromell, President	O'Brien Energy Company	0990-645	Texas	Robertson
42.395.14700.0424	2/27/2008	Joe B. Camp, widower of Roberta G. Camp	O'Brien Resources, LLC	1028-167	Texas	Robertson
42.395.14700.0425	2/18/2005	Frances Trojacek et vir	Prime Resources, LP	0918-777	Texas	Robertson
42.395.14700.0426	8/16/2007	Evelyn Irene White et al	Daystar Resources, LTD	0999-576	Texas	Robertson
42.395.14700.0427		Sidney K. Herrington	Sarah Exploration Company	0967-428	Texas	Robertson
42.395.14700.0428		Cornell Hickman	Sarah Exploration Company	0974-067	Texas	Robertson
42.395.14700.0429		Chris Smitherman, Receiver for the Unknown Heirs/Devisees of Isaac Jackson	O'Brien Energy Company	1012-719	Texas	Robertson
42.395.14700.0430	8/29/2007	Chris Smitherman, Receiver for the Unknown Heirs/Devisees of Rubie Mae Whitehead Oliver	O'Brien Energy Company	1012-716	Texas	Robertson
42.395.14700.0431 42.395.14700.0432		H. Maurice Pollock Estate	Sarah Exploration Company	0967-431 0984-408	Texas	Robertson Robertson
42.395.14700.0432		Maribeth Rhodes Allen Tony S. Boykin	O'Brien Energy Company O'Brien Energy Company	0984-420	Texas	Robertson
42.395.14700.0434		Cheryl A. Brown	O'Brien Energy Company	0984-386	Texas	Robertson
42.395.14700.0435		Mary Sue Burchfield	O'Brien Energy Company	0984-392	Texas	Robertson
42.395.14700.0436		Vonda Cravey	O'Brien Energy Company	0984-414	Texas	Robertson
42.395.14700.0437		Christine N. Jackson	O'Brien Energy Company	0984-405	Texas	Robertson
42.395.14700.0438	3/22/2007	Elizabeth Winn Jackson	O'Brien Energy Company	0984-389	Texas	Robertson
42.395.14700.0439	3/22/2007	Brian Jones	O'Brien Energy Company	0984-432	Texas	Robertson
42.395.14700.0440	3/22/2007	Mary Ann Jones	O'Brien Energy Company	0984-426	Texas	Robertson
42.395.14700.0441	3/22/2007	Ray Matthews	O'Brien Energy Company	0984-417	Texas	Robertson
42.395.14700.0442	3/22/2007	Ann C. McFarland	O'Brien Energy Company	0984-435	Texas	Robertson
42.395.14700.0443	3/22/2007	Lorene Murff	O'Brien Energy Company	0984-423	Texas	Robertson
42.395.14700.0444		Stan Murff	O'Brien Energy Company	0984-383	Texas	Robertson
42.395.14700.0445		Cecil T. Rhodes	O'Brien Energy Company	0984-441	Texas	Robertson
42.395.14700.0446		John Robert Rhodes	O'Brien Energy Company	0984-438	Texas	Robertson
42.395.14700.0447		Sandra Diane Rhodes	O'Brien Energy Company	0984-444	Texas	Robertson
42.395.14700.0448		Royce Shaw	O'Brien Energy Company O'Brien Energy Company	0984-411	Texas	Robertson
42.395.14700.0449 42.395.14700.0450		Kimberly L. Sisco Larry Swanson	O'Brien Energy Company	0984-447 0984-380	Texas	Robertson Robertson
42.395.14700.0450		Elizabeth Ann Trammell	O'Brien Energy Company	0984-395	Texas	Robertson
42.395.14700.0451	3/22/2007		O'Brien Energy Company	0984-429	Texas	Robertson
42.395.14700.0453		Janie Trojacek Estate	Prime Resources, LP	0943-426	Texas	Robertson
42.395.14700.0454		Betty L. Baird	O'Brien Energy Company	0999-397	Texas	Robertson
42.395.14700.0455		Leocadia Czajkowski	Sarah Exploration Company	0933-453	Texas	Robertson
42.395.14700.0456	7/24/2007	Patricia Hopcus-Jorski	O'Brien Energy Company	0999-394	Texas	Robertson
42.395.14700.0457	4/23/2007	Robert Pat Smith, Sr.	O'Brien Energy Company	1024-490	Texas	Robertson
42.395.14700.0458	9/30/2005	T. W. Whaley, Jr.	Sarah Exploration Company	0933-627	Texas	Robertson
42.395.14700.0459	4/18/2005	Antonette Koscielniak	Prime Resources, LP	0918-421	Texas	Robertson
42.395.14700.0460	9/13/2005	Jan Stockard Cato	Sarah Exploration Company	0933-377	Texas	Robertson
42.395.14700.0461	8/23/2005	Jane R. Lancaster	Sarah Exploration Company	0933-510	Texas	Robertson
42.395.14700.0462	8/23/2005	James I. Riddle, III	Sarah Exploration Company	0933-558	Texas	Robertson
42.395.14700.0463	9/13/2005	Sue S. Schaefer	Sarah Exploration Company	0933-392	Texas	Robertson
42.395.14700.0464		Walter A. Stockard, Jr.	Sarah Exploration Company	0933-395	Texas	Robertson
42.395.14700.0465		Frances Marks Travis et al	O'Brien Resources, LLC	1043-315	Texas	Robertson
42.395.14700.0466		Loyce Majkazak	Prime Resources, LP	0918-528	Texas	Robertson
42.395.14700.0467		Anabeth Goleman Boedecker	Prime Resources, LP	0918-142	Texas	Robertson
42.395.14700.0468	6/14/2005	Alan Jackson	Prime Resources, LP	0918-386	Texas	Robertson

Lease #	Date	Lessor	Lessee	Book-Page	State	County
42.395.14700.0469	5/26/2005	Winona Perkins Martin	Prime Resources, LP		Texas	Robertson
42.395.14700.0470	5/25/2005	Clyde Perkins	Prime Resources, LP	0918-635	Texas	Robertson
42.395.14700.0471	5/26/2005	Louise R. Perkins	Prime Resources, LP	0918-640	Texas	Robertson
42.395.14700.0472	5/26/2005	Betty Perkins Wells	Prime Resources, LP	0919-043	Texas	Robertson
42.395.14700.0494	2/28/2008	Bessie K. Bishop Perry, et al	O'Brien Resources, LLC	1034-401	Texas	Robertson
42.395.14700.0495	3/7/2008	James L. Hailey et ux	O'Brien Resources, LLC	1029-673	Texas	Robertson
42.395.14700.0496	3/7/2008	Robert L. Hailey et ux	O'Brien Resources, LLC	1029-675	Texas	Robertson
42.395.14700.0498	2/27/2008	The Estate of Michael Muse et al	O'Brien Resources, LLC	1029-670	Texas	Robertson
42.395.14700.0499	3/15/2005	Michael Muse et ux	Prime Resources, LP	0918-592	Texas	Robertson
42.395.14700.0504	3/17/2005	Cynthia Slominski Cappadona	Prime Resources, LP	0918-164	Texas	Robertson
42.395.14700.0505	3/17/2005	Nancy Slominski Galinski	Prime Resources, LP	0918-287	Texas	Robertson
42.395.14700.0506	3/28/2005	Louise Muzyka	Prime Resources, LP	0918-603	Texas	Robertson
42.395.14700.0507	3/7/2008	Mable L. London	O'Brien Resources, LLC	1029-671	Texas	Robertson
42.395.14700.0508	3/30/2005	Pauline Slominski Reade	Prime Resources, LP	0918-650	Texas	Robertson
42.395.14700.0509	3/23/2005	Darrell W. Trojacek et ux	Prime Resources, LP	0918-772	Texas	Robertson
42.395.14700.0510	8/15/2005	Vincent Slominski	Prime Resources, LP	0943-470	Texas	Robertson
42.395.14700.0511	4/21/2008	Don Buckner	O'Brien Resources, LLC	1034-435	Texas	Robertson
42.395.14700.0512	2/5/2008	Darrell W. Trojacek et ux	O'Brien Resources, LLC	1029-669	Texas	Robertson
42.395.14700.0513	4/7/2005	Richard J. Schwartz et ux	Prime Resources, LP	0918-696	Texas	Robertson
42.395.14700.0514	4/7/2005	Elaine Anderson	Prime Resources, LP	0918-066	Texas	Robertson
42.395.14700.0515	4/7/2005	Charles A. Ballew	Prime Resources, LP	0918-090	Texas	Robertson
42.395.14700.0516	4/7/2005	Sam Ballew	Prime Resources, LP	0918-095	Texas	Robertson
42.395.14700.0517	4/7/2005	Terry Ballew	Prime Resources, LP	0918-100	Texas	Robertson
42.395.14700.0518	4/7/2005	Frances Edge	Prime Resources, LP	0918-264	Texas	Robertson
42.395.14700.0519	4/7/2005	Nancy Lenz	Prime Resources, LP	0918-488	Texas	Robertson
42.395.14700.0520	5/24/2005	Evelyn Rudek	Prime Resources, LP	0918-668	Texas	Robertson
42.395.14700.0522	2/18/2009	James W. Jackson, Jr. et ux	O'Brien Resources, LLC	1067-498	Texas	Robertson
42.395.14700.0527	5/12/2005	David E. Krolczyk	Prime Resources, LP	0918-433	Texas	Robertson
42.395.14700.0528	5/2/2005	Leon V. Krolczyk	Prime Resources, LP	0918-438	Texas	Robertson
42.395.14700.0529	5/3/2005	Martha S. Graham	Prime Resources, LP	0918-300	Texas	Robertson
42.395.14700.0530	5/3/2005	Patricia S. Williams	Prime Resources, LP	0919-115	Texas	Robertson
42.395.14700.056	1/8/2008	Kenneth W. Wiley	O'Brien Resources, LLC	1028-0180	Texas	Robertson
42.395.14700.0567	5/26/2005	Peggy S. Dingley	Prime Resources, LP	0918-237	Texas	Robertson
42.395.14700.0568	5/26/2005	Sandra K. Howell	Prime Resources, LP	0918-376	Texas	Robertson
42.395.14700.0569	5/26/2005	Linda Patranella	Prime Resources, LP	0918-630	Texas	Robertson
42.395.14700.0614	4/24/2007	Michael Wayne Axtell, et ux	O'Brien Energy Company	0985-665	Texas	Robertson
42.395.14700.0615	5/23/2005	Herman A. Mentzel et ux (now deceased)	Prime Resources, LP	0918-551	Texas	Robertson
42.395.14700.0624	1/8/2018	Kristopher Paige Gostecnik	O'Brien Resources, LLC	1337-0601	Texas	Robertson
42.395.14700.0628	2/18/2009	Victor R. Wooten, et ux	O'Brien Resources, LLC	1067-617	Texas	Robertson
42.395.14700.0635	5/22/2009	White Rock Cemetery Assn & The White Rock Community	O'Brien Resources, LLC	1075-330	Texas	Robertson
42.395.14700.0644	5/20/2009	Bremond ISD of Robertson County	O'Brien Resources, LLC	1078-777	Texas	Robertson
42.395.14700.0646	5/19/2009	Freddie Ray Zach	O'Brien Resources, LLC	1078-764	Texas	Robertson
42.395.14700.0647	6/5/2009	Jerry Crow	O'Brien Resources, LLC	1078-770	Texas	Robertson
42.395.14700.0648	7/14/2009	Jerald Wayne Carter	O'Brien Resources, LLC	1082-547	Texas	Robertson
42.395.14700.0649	7/14/2009	Mary Louise Carter	O'Brien Resources, LLC	1082-551	Texas	Robertson
42.395.14700.0650	7/14/2009	Danny Jean Carter	O'Brien Resources, LLC	1082-543	Texas	Robertson
42.395.14700.0651	7/14/2009	Ronald Charles Gillam	O'Brien Resources, LLC	1082-555	Texas	Robertson
42.395.14700.0652	7/14/2009	Scott A. Jarvis	O'Brien Resources, LLC	1082-559	Texas	Robertson
42.395.14700.0653	7/14/2009	Margaret Louise Davis	O'Brien Resources, LLC	1082-563	Texas	Robertson
42.395.14700.0654	7/14/2009	Mary Gillam McDonald	O'Brien Resources, LLC	1082-571	Texas	Robertson
42.395.14700.0655	7/14/2009	Harry R. Maxwell	O'Brien Resources, LLC	1082-525	Texas	Robertson
42.395.14700.0656	7/14/2009	Kenneth Duane Carter	O'Brien Resources, LLC	1082-567	Texas	Robertson
42.395.14700.0657	7/14/2009	James Ray Maxwell	O'Brien Resources, LLC	1082-529	Texas	Robertson
42.395.14700.0658	7/14/2009	Stephanie D. Dowdell	O'Brien Resources, LLC	1082-537	Texas	Robertson
42.395.14700.0660	7/14/2009	Diana Jarvis Carrington	O'Brien Resources, LLC	1082-533	Texas	Robertson
42.395.14700.0661	8/3/2009	William Friday Heirs through Receiver Kent Schuster	O'Brien Resources, LLC	1084-691	Texas	Robertson
42.395.14700.0662	8/3/2009	Mollie Friday Heirs through Receiver Kent Schuster	O'Brien Resources, LLC	1084-695	Texas	Robertson
42.395.14700.0668	8/4/2009	PEC Minerals, LP	O'Brien Resources, LLC	1086-789	Texas	Robertson
42.395.14700.0676	1/18/2010	Martha Kierbow Grove, Life Estate	O'Brien Resources, LLC	1098-495	Texas	Robertson
42.395.14700.0687		Denyse K. Du Puy	O'Brien Resources, LLC	1104-663	Texas	Robertson
		Suzanne K. Du Puy	O'Brien Resources, LLC	1104-666	Texas	Robertson
42.395.14700.0688						
42.395.14700.0688 42.395.14700.0699	5/19/2009	Larry Scott Heggie	O'Brien Resources, LLC	1082-585	Texas	Robertson
			O'Brien Resources, LLC O'Brien Resources, LLC		Texas Texas	Robertson Robertson
42.395.14700.0699	8/15/2010	Larry Scott Heggie Devon Energy Production Company, L. P. Elsie Mae Wallace, Individually and as Life Tenant				

September Sept	Lease #	Date	Lessor	Lessee	Book-Page	State	County
CR MITCH (1988) CR MITCH (42.395.14700.0703	10/10/2017	Elsie Mae Wallace, Individually and as Life Tenant	O'Brien Resources, LLC	1333-212	Texas	Robertson
28 MIN DIR MIN MAN MAN MAN MAN MAN MAN MAN MAN MAN MA	42.395.14700.0704	10/10/2017	Elsie Mae Wallace, Individually and as Life Tenant	O'Brien Resources, LLC	1333-216	Texas	Robertson
CASES MINISTERS Controllers	42.395.14700.0705	10/10/2017	Elsie Mae Wallace, Individually and as Life Tenant	O'Brien Resources, LLC	1333-220	Texas	Robertson
CRA MICHINED CASIS MICHINED CHARTON CHARTON <td>42.395.14700.0706</td> <td>8/26/2010</td> <td>Elsie Mae Wallace, Life Estate</td> <td>O'Brien Resources, LLC</td> <td>1114-764</td> <td>Texas</td> <td>Robertson</td>	42.395.14700.0706	8/26/2010	Elsie Mae Wallace, Life Estate	O'Brien Resources, LLC	1114-764	Texas	Robertson
CBM MORDING CSM MORDING Cyber More State (1987) Cyber More More More More More More More Mo	42.395.14700.0708	7/16/2010	Ray Edward Hunter	O'Brien Resources, LLC	1114-023	Texas	Robertson
ABASE MIREDAM SQUARD Open Sentence 111-10 140-10	42.395.14700.0709	7/16/2010	Sheri Jo Liscano	O'Brien Resources, LLC	1114-019	Texas	Robertson
CORD MINISTER CORD MINISTER Comment (Company) C	42.395.14700.0710	7/16/2010	Eddy Truett	O'Brien Resources, LLC	1114-027	Texas	Robertson
2000 SI MORNING 2001 MI SARD ARTHORITY 2001 Min Sarporty 6 Morning 2000 SI MORNING 2001 Min Sarporty 6 Morning	42.395.14700.0713	1/7/2011	Mary Brunette	O'Brien Resources, LLC	1124-472	Texas	Robertson
1968 1969	42.395.14700.0714	3/3/2011	Bruno Zajac	O'Brien Resources, LLC	1130-744	Texas	Robertson
USA MIRTURE STATEMENT MERITARE MERITARE STATEMENT MERITARE STATEME	42.395.14700.0715	3/3/2011	Bernadine Zajac Kubiak	O'Brien Resources, LLC	1130-741	Texas	Robertson
September Sept	42.395.14700.0717	3/15/2011	Helen Joyce Jackson et al	O'Brien Resources, LLC	1133-007	Texas	Robertson
CRAIN FROM TWO STATES 1871 MIN Journal Black of all Controlled States 1871 MIN JOURNAL STATES	42.395.14700.0719	5/13/2011	Luminant Mineral Development Company, LLC	O'Brien Resources, LLC	1135-681	Texas	Robertson
Jack 1981 (1981) 18 (19	42.395.14700.0722	3/11/2011	The Estate of Sheam Moody, Jr.	O'Brien Resources, LLC	1133-472	Texas	Robertson
USB STREETING CRIBOTIL InventMent Decignant Concava LLT Class American Other Research 1 (1922) Laws Assertation 24 SS FREETING 25 SS FREETING Class Lead William Class FREETING Class Lead William 11472 Tax Readed 24 SS FREETING SS FREETING Class Lead Point Egiste Product Egiste Product Conference and Joseph Peach Andread Office Readed Lead 1182-21 Tax Readed 24 SS FREETING SS FREETING SS FREETING Class Edit Readed Class Edit Readed 1182-21 Tax Readed 24 SS FREETING SS FREETING SS FREETING Class Edit Readed 1182-22 Tax Readed 24 SS FREETING SS FREETING SS FREETING ASK TAX 1182-22 Tax Readed 24 SS FREETING SS FREETING SS FREETING ASK TAX 1182-22 Tax Readed 24 SS FREETING SS FREETING SS FREETING ASK TAX ASK TAX Tax Readed 24 SS FREETING SS FREETING SS FREETING ASK TAX ASK TAX Tax	42.395.14700.0723	3/11/2011	Moody National Bank et al	O'Brien Resources, LLC	1133-468	Texas	Robertson
Close Montalination Office Management (Institute) 114-001 Face of Management (Institute) 114-001 Face of Management (Institute) Control (Institute) 114-001 Face of Management (Institute) 114-	42.395.14700.0724	3/11/2011	The Estate of Frances Newman Moody	O'Brien Resources, LLC	1133-470	Texas	Robertson
42.35 MR 1987 1982	42.395.14700.0726	6/28/2011	Luminant Mineral Development Company LLC	O'Brien Resources, LLC	1139-067	Texas	Robertson
Campa Camp	42.395.14700.0729	6/22/2011	James Ray White	O'Brien Resources, LLC	1147-267	Texas	Robertson
23.30 Min 2007 Min School Sign Min School School Min School School Min Schoo	42.395.14700.0730	6/22/2011	Stephen Lee White	O'Brien Resources, LLC	1147-270	Texas	Robertson
4.2. M. S. M. 2007. LG 9.00 Month State State Color 150 9.00 Month State St	42.395.14700.0731	10/13/2011	O'BENCO III, LP	O'Brien Resources, LLC	1148-142	Texas	Robertson
42.25.6.1.1.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0	42.395.14700.0735	8/5/2011	Estate of Delmar Eugene Pieskee, Co- Administrators Galen Peiskee and Jeanne Peisek Andreski	O'Brien Resources, LLC	1162-331	Texas	Robertson
22.000.000.000.000.000.000.000.000.000.	42.395.14700.0738	9/4/2012	Jerry William Markham	O'Brien Resources, LLC	1186-712	Texas	Robertson
2005 MITTATION 1000 FIRED STATE STATE AND STATE STATE AND STATE STATE STATE AND STATE	42.395.14700.0739	9/4/2012	Victoria Shay Markham Sills	O'Brien Resources, LLC	1186-708	Texas	Robertson
200 SEA FRODORTY 201 MICRODITY 10 MICRODITY 10 MICRODITY 70 MICRO	42.395.14700.0745	12/8/2004	Elizabeth A. Fowler	Sendero Resources, Inc.	893-421	Texas	Robertson
22.25.16.100/10084 511.0000 Ann. Y. Dreve Robertom 22.25.16.1000.0017 12.0000 Virgina E. Wilds Robertom Robertom Robertom 22.35.16.1000.00175 10.00000 Wilds France Abstraction Robertom Robertom 23.35.16.1000.00175 10.00000 President Robustomania Robertom Robertom Robertom Robertom 23.35.16.1000.00175 10.00000 President Robustomania Robertom Robertom Robertom 23.35.16.1000.00175 20.0000 D. A. Bottom, Jr. et al. Robertom Robertom Robertom 23.35.16.1000.00175 20.0000 Deel R. Bibby et al. Robertom Robertom Robertom 23.35.16.1000.00175 20.0000 Robertom Robertom Robertom Robertom Robertom 23.35.16.1000.00175 20.0000 Robertom Robertom Robertom Robertom Robertom 23.35.16.1000.00175 70.0000 Robertom Robertom Robertom Robertom Robertom 23.35.16.1000.00175	42.395.14700.0746	12/10/2004	Dr. Harry Eugene Ezell, Jr. as Trustee of the Dr. Harry Eugene Ezell Jr. Trust joined by wife Elizabeth M. Ezell	Sendero Resources, Inc.	893-418	Texas	Robertson
42.20.5.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.	42.395.14700.0747	5/11/2001	Merideth Wells Blair	4-JW, LTD	785-616	Texas	Robertson
42 20 5 1 470 070 70 1 502 700 70 1 502 700 70 1 100	42.395.14700.0748	5/11/2001	Ann Y. Jones	4-JW, LTD	785-616	Texas	Robertson
42 20 5 1470 0 70 15 16 710 0 0 15 Fank Dames Machinary Let us Fank Deciden Tributed Sandam Rizozansa, Inc. 85 0 10 Total Robistion 42 20 5 1470 0 70 15 32 20 20 15 D. A Brother Machinary Services Principates Inc. 85 0 10 Total Robistion 42 20 5 1470 0 70 15 32 20 20 15 D. A Brother Machinary Services Principates Inc. 85 0 10 Total Robistion 42 20 5 1470 0 70 15 32 20 20 15 D. A Brother Machinary Services Principates Inc. 85 0 1 Total Total Total 70 0 10 Total 70 0 10 10 0 10 Total Total 70 0 10 Total 70 0 10 10 0 10	42.395.14700.0749	2/10/2005	Virginia E. Wells	Sendero Resources, Inc.	893-424	Texas	Robertson
42.2005 L. 1900.000000000000000000000000000000000	42.395.14700.0750	1/27/2005	Howard Wookey, Trustee of the Mary E. Puffer Trust	Sendero Resources, Inc.	893-427	Texas	Robertson
42 28 51 1770 0775 29 200 D. A. Bolton, Jr. et u.X Rose of Passarian, Inc. 88 - 40 Total As (Anthol) 14 24 28 64 1470 0775 28 20 20 20 20 20 20 20 20 20 20 20 20 20	42.395.14700.0751	1/21/2005	Frank James Markham Jr. et ux	Sendero Resources, Inc.	898-680	Texas	Robertson
42.256.170.0754 39.5000 Belac Others et six 68.64 of Delanics, inc. 68.64 of Team Roberton 42.356.170.0755 26000 Demis Elizary et un 68.04 of Team 68.04 of Team Roberton 42.356.170.0757 325000 Vince Otheran 68.04 of Team 80.04 of Team 80.0	42.395.14700.0752	3/4/2005	Frank Destefano, Trustee of the 1982 Destefano Trust	Sendero Resources, Inc.	893-377	Texas	Robertson
42.396.1470.0736 28.0000 Certific Ribby et ut Sendron Rissurces, in. 68.14 15 Town Roberts 42.396.1470.0736 28.0000 Albodo Zuberloy et ot Sendron Rissurces, in. 69.14 15 Town Roberts 42.396.1470.0736 7.0000 March Declaration Sendron Rissurces, in. 69.84 70 Control 42.396.1470.0736 7.0000 March Thomas Sendron Rissurces, in. 69.17 10 Control 42.396.1470.0736 67.0000 March Thomas Control Residency Residency Resources, in. 69.17 10 Town Residency 42.396.1470.0736 11/12200 Mary Separation Residency Resources, in. 69.11 10 Residency Residency Resources, in. 69.11 10 Residency	42.395.14700.0753	2/3/2005	D. A. Bolton, Jr. et ux	Sendero Resources, Inc.	893-405	Texas	Robertson
42.356.1470.0776 28/2005 Williams Description Sendior Resources, Inc. 68/3-15 Town Position 42.356.1470.0757 32/2005 Williams Demande Sendior Resources, Inc. 68/20-75 Town Position 42.356.1470.0759 78/2005 Call Oldhium Sendior Resources, Inc. 68/20-75 Town Robertison 42.356.1470.0759 11/2007 Position Call Oldhium Sendior Resources, Inc. 69/21-72 Town Robertison 42.356.1470.0750 11/2007 Mill July Berpoids Sendior Resources, Inc. 69/24-72 Town Robertison 42.356.1470.0752 11/2009 July Berpoids Sendior Resources, Inc. 69/3-42 Robertison 42.356.1470.0754 11/2007 Mary Berpoids Sendior Resources, Inc. 69/3-42 Town Robertison 42.356.1470.0759 20/2007 Nomine E. Wiley et al. 11/2004 Sendior Resources, Inc. 69/3-42 Town Robertison 42.356.1470.0759 20/2007 Nomine E. Wiley et al. 11/2004 Sendior Resources, Inc. 69/3-42	42.395.14700.0754	3/25/2005	Bates Oldham et ux	Sendero Resources, Inc.	896-403	Texas	Robertson
42.256.1470.0757 32.2000 Murine Thomas Sendior Resource, Inc. 696.75 Towas Roberton 42.256.1470.0758 78/2000 Murine Thomas Sendior Resource, Inc. 690.72 Towas Roberton 42.256.1470.0759 78/2000 Cold Orbithm Sendior Resource, Inc. 690.72 Towas Roberton 42.256.1470.0751 11/122007 May De Rymolis Sendior Resource, Inc. 691.10 Towas Roberton 42.256.1470.0753 82.2000 Haybey M. Branch Indiv & sa AF For James O. Branch Bendior Resource, Inc. 695.42 Towas Roberton 42.256.1470.0753 82.2000 Haybey M. Branch Indiv & sa AF For James O. Branch Sendior Resource, Inc. 695.42 Towas Roberton 42.256.1470.0754 82.2000 M. Gold Towas American Resource, Inc. 695.42 Towas Roberton 42.256.1470.0756 21.2000 M. Collection American Resource, Inc. 695.42 Towas Roberton 42.256.1470.0756 21.2000 M. Collection American Resource, Inc. 696.40 Towas	42.395.14700.0755	2/8/2005	Derrill R. Bibby et ux	Sendero Resources, Inc.	893-412	Texas	Robertson
Age 14700.0758 78,2005 78,20	42.395.14700.0756	2/8/2005	Alfredo Zabroky et ux	Sendero Resources, Inc.	893-415	Texas	Robertson
42 356 14700.0759 78 2000 Cel Cliffum Sendor Resources, Inc. 996-723 Flows Robertson 42 356 14700.0750 61/12/007 Sendor Resources, Inc. 991-19 Tosa Robertson 42 356 14700.0752 11/12/2001 Sendor Resources, Inc. 991-24 Tosa Robertson 42 356 14700.0752 11/12/2001 Sendor Resources, Inc. 993-24 Tosa Robertson 42 356 14700.0753 86/2000 Gender Resources, Inc. 995-241 Tosa Robertson 42 356 14700.0756 62 12007 Normia E. Wiley et al Sendor Resources, Inc. 995-341 Tosa Robertson 42 356 14700.0756 24/2005 TOSA Sendor Resources, Inc. 995-341 Tosa Robertson 42 356 14700.0756 24/2005 TOSA Sendor Resources, Inc. 995-341 Tosa Robertson 42 356 14700.0756 24/11/2005 Lenor Resources, Inc. 995-341 Tosa Robertson 42 356 14700.0756 21/12005 Lenor Robertson Sendor Resources, Inc. 995-341 Tosa </td <td>42.395.14700.0757</td> <td>3/25/2005</td> <td>Wayne Oldham</td> <td>Sendero Resources, Inc.</td> <td>898-677</td> <td>Texas</td> <td>Robertson</td>	42.395.14700.0757	3/25/2005	Wayne Oldham	Sendero Resources, Inc.	898-677	Texas	Robertson
42.365.1470.0761 61/12070 File-Minerals, Inc. Senders Resources, Inc. 105-125 Robertson 42.366.1470.0761 111/12070 August Ju Regnordes Senders Resources, Inc. 105-125 Tosas Robertson 42.366.1470.0762 117/2006 Jurky Ju Regnordes Jack G. Garrett Senders Resources, Inc. 695-442 Tosas Robertson 42.366.1470.0762 82/2000 Hayly M. Branch Indiv & as Alf- For James O. Branch Senders Resources, Inc. 696-422 Tosas Robertson 42.366.1470.0767 82/2000 Rocege H. Nelson Jr. et ux Senders Resources, Inc. 696-402 Tosas Robertson 42.366.1470.0767 10/21007 Namia E. Wiley et al Senders Resources, Inc. 698-402 Tosas Robertson 42.366.1470.0768 21/1200 Hurrison August July August July Senders Resources, Inc. 698-402 Tosas Robertson 42.366.1470.0776 21/12005 Lurison August G. Garrett Senders Resources, Inc. 695-452 Tosas Robertson 42.366.1470.0777 3470.075	42.395.14700.0758	7/8/2005		Sendero Resources, Inc.	960-720	Texas	Robertson
42.356.1470.0781 11/12/2077 Many JA Reprolish Sendero Resources, Inc. 1015-423 Robertson 42.356.1470.0782 147/2075 Jack G. Garrett Sendero Resources, Inc. 893-344 Texas Robertson 42.356.1470.0784 82/2000 Hayley M. Branch Indiv's as AlF for James O. Branch Sendero Resources, Inc. 959-422 Tixas Robertson 42.356.1470.0784 62/2000 George H. Nelson I., et al. A. Discources Sendero Resources, Inc. 959-411 Toxas Robertson 42.356.1470.0786 62/2007 Nomina E. Wiley et al Sendero Resources, Inc. 883-100 Texas Robertson 42.356.1470.07870 10/20000 Marie S. Jett et al Sendero Resources, Inc. 883-400 Texas Robertson 42.356.1470.0788 21/11/2005 Lonora Reneau Sendero Resources, Inc. 883-400 Texas Robertson 42.356.1470.0779 14/2005 Lonora Reneau Sendero Resources, Inc. 883-400 Texas Robertson 42.356.1470.0779 14/2005 Lonora Reneau Sendero Resources, Inc. 88	42.395.14700.0759		Carl Oldham		960-723		
2.355.1470.0762 1/7/2005 Jack G. Garrett 1/7/2005 Jack G. Garrett 2.355.1470.0763 18/2006 Holyby M. Branch Indix & sa Al F for James O. Branch 42.355.1470.0763 8/2006 Holyby M. Branch Indix & sa Al F for James O. Branch 42.355.1470.0764 8/2001 George H. Nelion Jr. et ux		6/1/2007	FSB-Minerals, Inc.				Robertson
42.355,14700,0763 8/2/2001 George H. Melson Jr. et ux George H. Melson Jr. et ux Robertson 42.356,14700,0764 6/2/2007 George H. Melson Jr. et ux Sendero Resources, Inc. 799-066 Texas Robertson 42.356,14700,0765 6/2/2007 Naomis E. Wey et al George H. Melson Jr. et ux Sendero Resources, Inc. 995-40 Texas Robertson 42.356,14700,0766 2/2/2007 Naomis E. Wey et al George H. Melson Sendero Resources, Inc. 893-40 Texas Robertson 42.356,14700,0768 2/1/2005 Lenora Reneau Sendero Resources, Inc. 893-40 Texas Robertson 42.356,14700,0769 2/1/2005 Juli G. Mel Greensage Kertson Sendero Resources, Inc. 893-40 Texas Robertson 42.356,14700,0774 1/1/2005 Kertson George Resources, Inc. 995-40 Texas Robertson 42.356,14700,0775 2/1/2005 XLU Generation Co., LP Sendero Resources, Inc. 995-40 Texas Robertson 42.356,14700,0775 2/1/2005 XLU Generation Co., LP Sendero Resources, Inc.	42.395.14700.0761	11/12/2007				Texas	Robertson
42.395.14700.07764 8/29/201 George H. Nelson Jr. et ux Robertson 42.395.14700.0786 8/21/207 Naomia E. Wiley et al Sendero Resources, Inc. 965-341 Texas Robertson 42.395.14700.0786 2/4/2005 TXU Generation Co., LP Sendero Resources, Inc. 983-400 Texas Robertson 42.395.14700.0786 2/4/2005 TXU Generation Co., LP Sendero Resources, Inc. 883-100 Texas Robertson 42.395.14700.0788 2/11/2005 Lenora Reneau Sendero Resources, Inc. 895-449 Texas Robertson 42.395.14700.0789 2/11/2005 No Nel Greenage Sendero Resources, Inc. 895-409 Texas Robertson 42.395.14700.0773 1/12/2005 Kirstopher Palge Gesteronk Robertson Sendero Resources, Inc. 089-403 Texas Robertson 42.395.14700.0771 3/12/2005 Kirstopher Palge Gesteronk XIV Generation Co., LP Sendero Resources, Inc. 089-403 Texas Robertson 42.395.14700.0773 2/12/2005 Kull Generation Co., LP Sendero Resources, Inc. 089-4023	42.395.14700.0762	1/7/2005	Jack G. Garrett	Sendero Resources, Inc.	893-384	Texas	Robertson
42.355.14700.7765 6/21/207 Namina E, Wiley et al Sendero Resources, Inc. 995-341 Texas Robertson 42.356.14700.0767 10/29/2004 Marris S, Leit et al 42.356.14700.776 Sendero Resources, Inc. 683-430 Texas Robertson 42.356.14700.0767 10/29/2004 Marris S, Leit et al 42.356.14700.776 Sendero Resources, Inc. 885-400 Texas Robertson 42.356.14700.0769 2/11/2005 Long Generals Sendero Resources, Inc. 895-435 Texas Robertson 42.356.14700.0770 42/25005 Kristopher Paige Gostecnik Sendero Resources, Inc. 895-450 Texas Robertson 42.356.14700.0771 3/9/2005 VXIU Generation Co., LP Sendero Resources, Inc. 995-456 Texas Robertson 42.356.14700.0773 2/1/2005 Just Co., Le Sendero Resources, Inc. 996-450 Texas Robertson 42.356.14700.0773 1/1/2005 Just Co., Le Just Co., Le Sendero Resources, Inc. 997-497 Texas Robertson 42.356.14700.0775 1/1/2000	42.395.14700.0763	8/2/2006	Hayley M. Branch Indiv & as AIF for James O. Branch	Sendero Resources, Inc.	959-422	Texas	Robertson
42.395.14700.0776 2/4/2005 TXU Generation Co., LP Sendero Resources, Inc. 0893-430 Texas Robertson 42.395.14700.0767 10/29/2004 Marie S. Jett et al Sendero Resources, Inc. 883-100 Texas Robertson 42.395.14700.0768 2/11/2005 Lerora Reneau Sendero Resources, Inc. 895-490 Texas Robertson 42.395.14700.0770 4/25/2005 Kristopher Paige Gosteralk Sendero Resources, Inc. 895-495 Texas Robertson 42.395.14700.0771 3/9/2005 Kristopher Paige Gosteralk Sendero Resources, Inc. 896-400 Texas Robertson 42.395.14700.0771 3/9/2005 TXU Generation Co., LP Sendero Resources, Inc. 896-400 Texas Robertson 42.395.14700.0773 2/1/2005 TXU Generation Co., LP Sendero Resources, Inc. 896-400 Texas Robertson 42.395.14700.0773 1/1/2005 Jack G. Garrett Sendero Resources, Inc. 896-400 Texas Robertson 42.395.14700.0774 1/1/2006 Paarl McDonald Sendero Resources, Inc. 970-433		8/29/2001	George H. Nelson Jr. et ux	Sendero Resources, Inc.	790-066	Texas	Robertson
42 395 14700 0767 10/29/2004 Marie S. Jett et al Sendero Resources, Inc. 883-100 Texas Robertson 42 395 14700 0768 2/11/2005 Leonar Reneau Sendero Resources, Inc. 895-409 Texas Robertson 42 395 14700 0770 2/11/2005 Jonate (Freneau) Sendero Resources, Inc. 895-450 Texas Robertson 42 395 14700 0770 4/25005 Kirisopher Paige Gostecnik Sendero Resources, Inc. 905-456 Texas Robertson 42 395 14700 0770 3/92005 TXU Generation Co., LP Sendero Resources, Inc. 0894-003 Texas Robertson 42 395 14700 0772 2/1/2005 Axi Generation Co., LP Sendero Resources, Inc. 0892-023 Texas Robertson 42 395 14700 0773 2/1/2005 Dan Kyle Dean Kyle Sendero Resources, Inc. 970-437 Texas Robertson 42 395 14700 0775 11/8/2006 Dean Kyle Dean Kyle Sendero Resources, Inc. 970-437 Texas Robertson 42 395 14700 0775 11/8/2006 Gene Julia Simmons Elison XTO Energy,						Texas	
42.395.1470.0768 2/11/2005 Lenora Reneau Sendero Resources, Inc. 895.409 Texas Robertson 42.395.14700.0770 4/25/2005 Kridsprote Palge Gostacnik Sendero Resources, Inc. 995.456 Texas Robertson 42.395.14700.0771 3/30205 XTU Generation Co., LP Sendero Resources, Inc. 0894.002 Texas Robertson 42.395.14700.0773 2/1/2005 XTU Generation Co., LP Sendero Resources, Inc. 0892.02 Texas Robertson 42.395.14700.0773 2/1/2005 Acade Generation Co., LP Sendero Resources, Inc. 896.400 Texas Robertson 42.395.14700.0773 2/1/2005 July Generation Co., LP Sendero Resources, Inc. 896.400 Texas Robertson 42.395.14700.0773 1/16/2005 Dean Kyle Sendero Resources, Inc. 970.431 Texas Robertson 42.395.14700.0775 1/16/2005 Dean McDonald Texas Robertson 42.395.14700.0776 1/19/2006 Gendero Ann Simmons Anderson XTO Energy, Inc. 971.055 Texas Robertson <tr< td=""><td>42.395.14700.0766</td><td></td><td></td><td></td><td></td><td></td><td></td></tr<>	42.395.14700.0766						
42.956.14700.0779 2/11/2005 Jo Nell Greensage Sendero Resources, Inc. 893-435 Texas Robertson 42.395.14700.0770 4/25/2005 Kristopher Paige Gostecnik Sendero Resources, Inc. 905-456 Texas Robertson 42.395.14700.0771 3/9/2005 TXU Generation Co., LP Sendero Resources, Inc. 0894-030 Texas Robertson 42.395.14700.0773 2/1/2005 Jack G. Garrett Sendero Resources, Inc. 889-406 Texas Robertson 42.395.14700.0773 1/1/2006 Jack G. Garrett Sendero Resources, Inc. 897-437 Texas Robertson 42.395.14700.0774 1/1/2006 Dean Kyle Sendero Resources, Inc. 970-437 Texas Robertson 42.395.14700.0775 1/1/2006 Gearl McDonald Sendero Resources, Inc. 970-437 Texas Robertson 42.395.14700.0776 1/19/2006 Gearl McDonald Sendero Resources, Inc. 971-057 Texas Robertson 42.395.14700.0778 1/19/2006 Comelia Ann L Starkey, Truste XTO Energy, Inc. 971-053 Texas </td <td>42.395.14700.0767</td> <td></td> <td></td> <td>Sendero Resources, Inc.</td> <td>883-100</td> <td>Texas</td> <td>Robertson</td>	42.395.14700.0767			Sendero Resources, Inc.	883-100	Texas	Robertson
42.395.14700.0770 4/25/2005 Kristopher Paige Gostecnik Sendero Resources, Inc. 0894-030 Texas Robertson 42.395.14700.0772 2/1/2005 TXU Generation Co., LP Sendero Resources, Inc. 0892-023 Texas Robertson 42.395.14700.0773 2/1/2005 JAK G. Garrett Sendero Resources, Inc. 896-406 Texas Robertson 42.395.14700.0774 1/16/2006 Dack KJ. Garrett Sendero Resources, Inc. 970-437 Texas Robertson 42.395.14700.0775 1/16/2006 Peart McDonald Sendero Resources, Inc. 970-437 Texas Robertson 42.395.14700.0776 1/19/2006 Peart McDonald Sendero Resources, Inc. 970-433 Texas Robertson 42.395.14700.0776 1/19/2006 Gene Julia Simmons Elison XTO Energy, Inc. 971-059 Texas Robertson 42.395.14700.0778 1/19/2006 Candace Ann Simmons Anderson XTO Energy, Inc. 971-053 Texas Robertson 42.395.14700.0778 1/19/2006 Cornella Ann L. Starkey, Trustee XTO Energy, Inc. 2006443	42.395.14700.0768	2/11/2005	Lenora Reneau		895-409	Texas	Robertson
2.395.14700.0771 3/9/2005 TXU Generation Co., LP Sendero Resources, Inc. 0.894-030 Texas Robertson degree Resources, Inc. 0.892-023 Texas Robertson degree Resources, Inc. 0.892-023 Texas Robertson degree Resources, Inc. 0.892-023 Texas Robertson degree Resources, Inc. 0.896-406 Texas Rober	42.395.14700.0769	2/11/2005	Jo Nell Greensage	Sendero Resources, Inc.	893-435	Texas	Robertson
42.395.14700.0772 2/1/2005 TXU Generation CO., LP Sender Resources, Inc. 889-023 Texas Robertson 42.395.14700.0773 2/1/2005 Jack G. Garrett Sender Resources, Inc. 896-406 Texas Robertson 42.395.14700.0774 11/6/2006 Dean Kyle Sender Resources, Inc. 970-437 Texas Robertson 42.395.14700.0775 11/6/2006 Dean McDonald Sender Resources, Inc. 970-433 Texas Robertson 42.395.14700.0776 11/9/2006 Gene Julia Simmons Ellison XTO Energy, Inc. 971-055 Texas Robertson 42.395.14700.0777 11/9/2006 Cornella Ann L. Starkey, Trustee XTO Energy, Inc. 971-055 Texas Robertson 42.395.14700.0778 11/9/2006 Cornella Ann L. Starkey, Trustee XTO Energy, Inc. 2006543 Texas Robertson 42.395.14700.0778 1/2/2014 Cornelia Ann L. Starkey Trustee XTO Energy, Inc. 2006543 Texas Robertson 42.395.14700.0782 1/2/2014 Cornelia Ann L. Starkey Trustee Abbreviolation	42.395.14700.0770	4/25/2005			905-456	Texas	Robertson
42.395.14700.0773 2/1/2005 Jack G. Garrett Robertson 42.395.14700.0774 11/6/2006 Dean Kyle Sendero Resources, Inc. 970-437 Texas Robertson 42.395.14700.0775 11/6/2006 Pearl McDonald Sendero Resources, Inc. 971-059 Texas Robertson 42.395.14700.0776 11/9/2006 Gene Julia Simmons Ellison XTO Energy, Inc. 971-059 Texas Robertson 42.395.14700.0777 11/9/2006 Comelia Ann L. Starkey, Trustee XTO Energy, Inc. 971-050 Texas Robertson 42.395.14700.0778 11/9/2006 Antonio Robies XTO Energy, Inc. 20065431 Texas Robertson 42.395.14700.0780 8/2/2006 Antonio Robies XTO Energy, Inc. 20065448 Texas Robertson 42.395.14700.0781 1/2/2014 Cornelia Ann Loreant Starkey Trustee XTO Energy, Inc. 20065448 Texas Robertson 42.395.14700.0782 1/2/2014 Cornelia Ann Loreant Starkey Trust/Cornelia Ann Loreant Starkey, Trustee O'Brien Resources, LLC 1230-303 Texas Robertson	42.395.14700.0771	3/9/2005	TXU Generation Co., LP	Sendero Resources, Inc.	0894-030	Texas	Robertson
42.395.14700.0774 11/6/2006 Dean Kyle Sendero Resources, Inc. 970-437 Texas Robertson 42.395.14700.0775 11/6/2006 Pearl McDonald Sendero Resources, Inc. 971-059 Texas Robertson 42.395.14700.0776 11/9/2006 Gene Julia Simmons Ellison XTO Energy, Inc. 971-059 Texas Robertson 42.395.14700.0777 11/9/2006 Candace Ann Simmons Anderson XTO Energy, Inc. 971-050 Texas Robertson 42.395.14700.0778 11/9/2006 Comella Ann L. Starkey, Trustee XTO Energy, Inc. 971-063 Texas Robertson 42.395.14700.0780 8/2/2006 Antonio Robles XTO Energy, Inc. 20065431 Texas Robertson 42.395.14700.0781 1/2/1201 Comelia Ann Loreant Starkey Trust/Cornelia Ann Loreant Starkey, Trustee XTO Energy, Inc. 20065448 Texas Robertson 42.395.14700.0782 1/21/2014 Comelia Ann Loreant Starkey Trust/Cornelia Ann Loreant Starkey, Trustee O'Brien Resources, LLC 1230-303 Texas Robertson 42.395.14700.0783 1/21/2014 Johnny R. St	42.395.14700.0772	2/1/2005	TXU Generation CO., LP	Sendero Resources, Inc.	0892-023	Texas	Robertson
42.395.14700.0775 11/6/2008 Pearl McDonald Sendero Resources, Inc. 970-433 Texas Robertson 42.395.14700.0776 11/9/2006 Gene Julia Simmons Ellison XTO Energy, Inc. 971-059 Texas Robertson 42.395.14700.0777 11/9/2006 Candace Ann Simmons Anderson XTO Energy, Inc. 971-063 Texas Robertson 42.395.14700.0778 11/9/2006 Comella Ann L. Starkey, Trustee XTO Energy, Inc. 20065431 Texas Robertson 42.395.14700.0780 8/2/2006 Antonio Robles XTO Energy, Inc. 20065448 Texas Robertson 42.395.14700.0782 1/21/2014 Cornelia Ann Loreant Starkey Trust/Cornelia Ann Loreant Starkey, Trustee XTO Energy, Inc. 20065448 Texas Robertson 42.395.14700.0782 1/21/2014 Cornelia Ann Loreant Starkey Trust/Cornelia Ann Loreant Starkey, Trustee O'Brien Resources, LLC 1230-309 Texas Robertson 42.395.14700.0783 1/21/2014 Candace Ann Simmons Anderson O'Brien Resources, LLC 1230-303 Texas Robertson 42.395.14700.0786 1/21/2014	42.395.14700.0773	2/1/2005	Jack G. Garrett	Sendero Resources, Inc.	896-406	Texas	Robertson
42.395.14700.0776 11/9/2006 Gene Julia Simmons Ellison XTO Energy, Inc. 971-055 Texas Robertson 42.395.14700.0777 11/9/2006 Candace Ann Simmons Anderson XTO Energy, Inc. 971-055 Texas Robertson 42.395.14700.0778 11/9/2006 Comella Ann L. Starkey, Trustee XTO Energy, Inc. 20065431 Texas Robertson 42.395.14700.0780 8/2/2006 Antonio Robles XTO Energy, Inc. 20065448 Texas Robertson 42.395.14700.0782 1/21/2014 Cornelia Ann Loreant Starkey Trust/Cornelia Ann Loreant Starkey, Trustee O'Brien Resources, LLC 1230-309 Texas Robertson 42.395.14700.0783 1/21/2014 Candace Ann Simmons Anderson O'Brien Resources, LLC 1230-303 Texas Robertson 42.395.14700.0784 1/21/2014 Johnny R. Starkey O'Brien Resources, LLC 1230-301 Texas Robertson 42.395.14700.0785 1/21/2014 Circle M. Ranch Ltd. c/o Clara Ruth Martin, General Partner O'Brien Resources, LLC 1232-767 Texas Robertson 42.395.14700.0795 9/9/201	42.395.14700.0774	11/6/2006	Dean Kyle	Sendero Resources, Inc.	970-437	Texas	Robertson
42.395.14700.0777 11/9/2006 Candace Ann Simmons Anderson XTO Energy Inc. 971-055 Texas Robertson 42.395.14700.0778 11/9/2006 Comella Ann L. Starkey, Trustee XTO Energy, Inc. 20065431 Texas Robertson 42.395.14700.0780 8/2/2006 Antonio Robles XTO Energy, Inc. 20065448 Texas Robertson 42.395.14700.0780 8/2/2006 Tony Gonzales et ux XTO Energy, Inc. 20065448 Texas Robertson 42.395.14700.0782 1/21/2014 Cornelia Ann Loreant Starkey Trust/Cornelia Ann Loreant Starkey, Trustee O'Brien Resources, LLC 1230-309 Texas Robertson 42.395.14700.0783 1/21/2014 Candace Ann Simmons Anderson O'Brien Resources, LLC 1230-303 Texas Robertson 42.395.14700.0784 1/21/2014 Johnny R. Starkey O'Brien Resources, LLC 1230-312 Texas Robertson 42.395.14700.0785 1/21/2014 Circle M. Ranch Ltd. c/o Clara Ruth Martin, General Partner O'Brien Resources, LLC 1233-315 Texas Robertson 42.395.14700.0795 9/9/2014 Donna Reed Donna Reed O'Brien Resources, LLC 1254-491	42.395.14700.0775	11/6/2006	Pearl McDonald	Sendero Resources, Inc.	970-433	Texas	Robertson
42.395.14700.0778 11/9/2006 Comella Ann L. Starkey, Trustee XTO Energy, Inc. 971-063 Texas Robertson 42.395.14700.0779 8/2/2006 Antonio Robles XTO Energy, Inc. 20065431 Texas Robertson 42.395.14700.0780 8/2/2006 Tony Gonzales et ux XTO Energy, Inc. 20065448 Texas Robertson 42.395.14700.0782 1/21/2014 Cornelia Ann Loreant Starkey Trust/Cornelia Ann Loreant Starkey, Trustee O'Brien Resources, LLC 1230-303 Texas Robertson 42.395.14700.0783 1/21/2014 Candace Ann Simmons Anderson O'Brien Resources, LLC 1230-303 Texas Robertson 42.395.14700.0784 1/21/2014 Johnny R. Starkey O'Brien Resources, LLC 1230-303 Texas Robertson 42.395.14700.0786 1/21/2014 Circle M. Ranch Ltd. c/o Clara Ruth Martin, General Partner O'Brien Resources, LLC 1232-767 Texas Robertson 42.395.14700.07986 1/21/2014 Rose Crafford Simmons O'Brien Resources, LLC 1233-315 Texas Robertson 42.395.14700.07986 1/21/2014 <td< td=""><td>42.395.14700.0776</td><td>11/9/2006</td><td>Gene Julia Simmons Ellison</td><td>XTO Energy, Inc.</td><td>971-059</td><td>Texas</td><td>Robertson</td></td<>	42.395.14700.0776	11/9/2006	Gene Julia Simmons Ellison	XTO Energy, Inc.	971-059	Texas	Robertson
42.395.14700.0779 8/2/2006 Antonio Robles XTO Energy, Inc. 20065431 Texas Robertson 42.395.14700.0780 8/2/2006 Tony Gonzales et ux XTO Energy, Inc. 20065448 Texas Robertson 42.395.14700.0782 1/21/2014 Cornelia Ann Loreant Starkey Trust/Cornelia Ann Loreant Starkey, Trustee O'Brien Resources, LLC 1230-309 Texas Robertson 42.395.14700.0783 1/21/2014 Candace Ann Simmons Anderson O'Brien Resources, LLC 1230-303 Texas Robertson 42.395.14700.0784 1/21/2014 Johnny R. Starkey O'Brien Resources, LLC 1230-312 Texas Robertson 42.395.14700.0786 1/21/2014 Circle M. Ranch Ltd. c/o Clara Ruth Martin, General Partner O'Brien Resources, LLC 1232-767 Texas Robertson 42.395.14700.0786 1/21/2014 Rose Crafford Simmons O'Brien Resources, LLC 1233-315 Texas Robertson 42.395.14700.0798 1/21/2014 Donna Reed Donna Reed O'Brien Resources, LLC 1254-491 Texas Robertson	42.395.14700.0777	11/9/2006	Candace Ann Simmons Anderson	XTO Energy Inc.	971-055	Texas	Robertson
42.395.14700.0780 8/2/2006 Tony Gonzales et ux XTO Energy, Inc. 20065448 Texas Robertson 42.395.14700.0782 1/21/2014 Cornelia Ann Loreant Starkey Trust/Cornelia Ann Loreant Starkey, Trustee O'Brien Resources, LLC 1230-309 Texas Robertson 42.395.14700.0783 1/21/2014 Candace Ann Simmons Anderson O'Brien Resources, LLC 1230-3012 Texas Robertson 42.395.14700.0784 1/21/2014 Johnny R. Starkey Circle M. Ranch Ltd. c/o Clara Ruth Martin, General Partner O'Brien Resources, LLC 1230-312 Texas Robertson 42.395.14700.0786 1/21/2014 Rose Crafford Simmons O'Brien Resources, LLC 1233-315 Texas Robertson 42.395.14700.0786 1/21/2014 Rose Crafford Simmons O'Brien Resources, LLC 1233-315 Texas Robertson 42.395.14700.0798 9/9/2014 Donna Reed O'Brien Resources, LLC 1254-491 Texas Robertson	42.395.14700.0778	11/9/2006	Comella Ann L. Starkey, Trustee	XTO Energy, Inc.	971-063	Texas	Robertson
42.395.14700.0782 1/21/2014 Cornelia Ann Loreant Starkey Trust/Cornelia Ann Loreant Starkey, Trustee O'Brien Resources, LLC 1230-309 Texas Robertson 42.395.14700.0783 1/21/2014 Candace Ann Simmons Anderson O'Brien Resources, LLC 1230-303 Texas Robertson 42.395.14700.0784 1/21/2014 Johnny R. Starkey O'Brien Resources, LLC 1230-312 Texas Robertson 42.395.14700.0785 1/21/2014 Circle M. Ranch Ltd. c/o Clara Ruth Martin, General Partner O'Brien Resources, LLC 1232-767 Texas Robertson 42.395.14700.0786 1/21/2014 Rose Crafford Simmons O'Brien Resources, LLC 1233-315 Texas Robertson 42.395.14700.0795 9/9/2014 Donna Reed O'Brien Resources, LLC 1254-491 Texas Robertson	42.395.14700.0779	8/2/2006	Antonio Robles	XTO Energy, Inc.	20065431	Texas	Robertson
42.395.14700.0783 1/2/12014 Candace Ann Simmons Anderson O'Brien Resources, LLC 1230-303 Texas Robertson 42.395.14700.0784 1/2/12014 Johnny R. Starkey O'Brien Resources, LLC 1230-312 Texas Robertson 42.395.14700.0785 1/2/12014 Circle M. Ranch Ltd. c/o Clara Ruth Martin, General Partner O'Brien Resources, LLC 1232-767 Texas Robertson 42.395.14700.0786 1/2/1/2014 Ross Crafford Simmons O'Brien Resources, LLC 1233-315 Texas Robertson 42.395.14700.0795 9/9/2014 Donna Reed O'Brien Resources, LLC 1254-491 Texas Robertson	42.395.14700.0780	8/2/2006	Tony Gonzales et ux	XTO Energy, Inc.	20065448	Texas	Robertson
42.395.14700.0784 1/21/2014 Johnny R. Starkey O'Brien Resources, LLC 1230-312 Texas Robertson 42.395.14700.0785 1/21/2014 Circle M. Ranch Ltd. c/o Clara Ruth Martin, General Partner O'Brien Resources, LLC 1232-767 Texas Robertson 42.395.14700.0786 1/21/2014 Ross Crafford Simmons O'Brien Resources, LLC 1233-315 Texas Robertson 42.395.14700.0795 9/9/2014 Donna Reed O'Brien Resources, LLC 1254-491 Texas Robertson	42.395.14700.0782	1/21/2014	Cornelia Ann Loreant Starkey Trust/Cornelia Ann Loreant Starkey, Trustee	O'Brien Resources, LLC	1230-309	Texas	Robertson
42.395.14700.0785 1/21/2014 Circle M. Ranch Ltd. c/o Clara Ruth Martin, General Partner O'Brien Resources, LLC 1232-767 Texas Robertson 42.395.14700.0786 1/21/2014 Ross Crafford Simmons O'Brien Resources, LLC 1233-315 Texas Robertson 42.395.14700.0795 9/9/2014 Donna Reed O'Brien Resources, LLC 1254-491 Texas Robertson	42.395.14700.0783	1/21/2014	Candace Ann Simmons Anderson	O'Brien Resources, LLC	1230-303	Texas	Robertson
42.395.14700.0786 1/21/2014 Ross Crafford Simmons O'Brien Resources, LLC 1233-315 Texas Robertson 42.395.14700.0795 9/9/2014 Donna Reed O'Brien Resources, LLC 1254-491 Texas Robertson	42.395.14700.0784	1/21/2014	Johnny R. Starkey	O'Brien Resources, LLC	1230-312	Texas	Robertson
42.395.14700.0795 9/9/2014 Donna Reed O'Brien Resources, LLC 1254-491 Texas Robertson	42.395.14700.0785	1/21/2014	Circle M. Ranch Ltd. c/o Clara Ruth Martin, General Partner	O'Brien Resources, LLC	1232-767	Texas	Robertson
	42.395.14700.0786	1/21/2014	Ross Crafford Simmons	O'Brien Resources, LLC	1233-315	Texas	Robertson
42.395.14700.0797 9/9/2014 Gary L. Freeman O'Brien Resources, LLC 1254-488 Texas Robertson	42.395.14700.0795	9/9/2014	Donna Reed	O'Brien Resources, LLC	1254-491	Texas	Robertson
	42.395.14700.0797	9/9/2014	Gary L. Freeman	O'Brien Resources, LLC	1254-488	Texas	Robertson

Lease #	Date	Lessor	Lessee	Book-Page	State	County
42.395.14700.0798	6/28/2011	Lucy Pfluger	O'Brien Resources, LLC	1143-107	Texas	Robertson
42.395.14700.0800	8/12/2015	Shannon Vann	O'Brien Resources, LLC	Pending	Texas	Robertson
1001-504	5/26/1983	Harry S. Phillips	Texana Exploration, Inc.	1001-504	Texas	Anderson
2422-0393	7/8/2014	Judith Ann Fisher a/k/a Judith Ann Hitchcock Mitchell	TRI-C RESOURCES, LLC	2422-0393	Texas	Anderson
2422-0431	7/1/2014	Onia Battle a/k/a Onia Mae Brown Battle	PETRO-LAND GROUP, INC.	2422-0431	Texas	Anderson
2422-0432	7/1/2014	Sandra Jo Hill	PETRO-LAND GROUP, INC.	2422-0432	Texas	Anderson
2422-393	7/8/2014	Judith Ann Fisher a/k/a Judith Ann Hitchcock Mitchell	TRI-C RESOURCES, LLC	2422-393	Texas	Anderson
2424-0033	7/10/2014	Barbara Paderick	TRI-C RESOURCES, LLC	2424-0033	Texas	Anderson
2424-0034	7/14/2014	Theo B. Johnson	TRI-C RESOURCES, LLC	2424-0034	Texas	Anderson
2424-0035	7/15/2014	Charles Bednar	TRI-C RESOURCES, LLC	2424-0035	Texas	Anderson
2424-0053	7/15/2014	Anna Sexton Henry a/k/a Anna Glenn	TRI-C RESOURCES, LLC	2424-0053	Texas	Anderson
2424-0056	7/20/2014	K. R. & N. J. Magouirk Living Trust dated April 9, 2013 by Karl Rex Magouirk and Nancy J. Magouirk	TRI-C RESOURCES, LLC	2424-0056	Texas	Anderson
2424-0058	7/20/2014	Gary Leon Henry	TRI-C RESOURCES, LLC	2424-0058	Texas	Anderson
2424-0060	7/20/2014	Shelby Lynn Patterson	TRI-C RESOURCES, LLC	2424-0060	Texas	Anderson
2424-0073	7/23/2014	Peggy Lynn O'Neil Clair	TRI-C RESOURCES, LLC	2424-0073	Texas	Anderson
2424-0080	7/1/2014	Nancy Kownslar Wheeler	TRI-C RESOURCES, LLC	2424-0080	Texas	Anderson
2424-0082	7/1/2014	Donna Davis Ganster	PETRO-LAND GROUP, INC.	2424-0082	Texas	Anderson
2424-0448	7/20/2014	Jodi Lynn Trout	TRI-C RESOURCES, LLC	2424-0448	Texas	Anderson
2424-0450	7/20/2014	John Jacob Henry	TRI-C RESOURCES, LLC	2424-0450	Texas	Anderson
2424-0454	7/20/2014	Dana Lynn Henry	TRI-C RESOURCES, LLC	2424-0454	Texas	Anderson
2424-0471	7/8/2014	Tom K. Hitchcock	TRI-C RESOURCES, LLC	2424-0471	Texas	Anderson
2424-0472	7/1/2014	Three Boys-Jo Nelson Trust	PETRO-LAND GROUP, INC.	2424-0472	Texas	Anderson
2425-0507	7/23/2014	Charles Wesley Black III	TRI-C RESOURCES, LLC	2425-0507	Texas	Anderson
2425-0513	7/30/2014	Mary A. Richert	TRI-C RESOURCES, LLC	2425-0513	Texas	Anderson
2426-0139	7/12/2014	Fonda W. Campbell	TRI-C RESOURCES, LLC	2426-0139	Texas	Anderson
2426-0140	7/15/2014	Barbara Ellen Maynard, as sole and separate property	TRI-C RESOURCES, LLC	2426-0140	Texas	Anderson
2426-0144	7/2/2014	Michael L. Holmes	TRI-C RESOURCES, LLC	2426-0144	Texas	Anderson
2429-0152	8/20/2014	Robert Wayne Thornton	TRI-C RESOURCES, LLC	2429-0152	Texas	Anderson
2430-0680	8/5/2014	Robert Wayne Thornton	TRI-C RESOURCES, LLC	2430-0680	Texas	Anderson
2430-0683	8/5/2014	Carol Ann Thornton	TRI-C RESOURCES, LLC	2430-0683	Texas	Anderson
2430-0684	7/29/2014	Page D. Burney Martin a/k/a Page D. Burney	TRI-C RESOURCES, LLC	2430-0684	Texas	Anderson
2434-0045	8/20/2014	Carol Ann Thornton	TRI-C RESOURCES, LLC	2434-0045	Texas	Anderson
2434-0081	9/12/2014	Amanda Faith Stampley	TRI-C RESOURCES, LLC	2434-0081	Texas	Anderson
2434-0083	8/21/2014	Mae Lee-Jean Wardell	TRI-C RESOURCES, LLC	2434-0083	Texas	Anderson
2434-0086	9/12/2014	Toni Angelia Soape	TRI-C RESOURCES, LLC	2434-0086	Texas	Anderson
2434-0093	9/9/2014	Glenda Elaine Carr	TRI-C RESOURCES, LLC	2434-0093	Texas	Anderson
2435-0591	9/12/2014	Laura April Waldron	TRI-C RESOURCES, LLC	2435-0591	Texas	Anderson
2435-0593	9/9/2014	Estate of Jonathan Ellis Williams, by Leigh Anne Williams Hale, Independent Executrix	TRI-C RESOURCES, LLC	2435-0593	Texas	Anderson
2435-0595	9/16/2014	Michael Giles	TRI-C RESOURCES, LLC	2435-0595	Texas	Anderson
2435-0596	9/15/2014	Novella L. Oldham	TRI-C RESOURCES, LLC	2435-0596	Texas	Anderson
2435-0597	9/13/2014	Thomas Bryce Oldham	TRI-C RESOURCES, LLC	2435-0597	Texas	Anderson
2435-0602	9/10/2014	Dick Bednar	TRI-C RESOURCES, LLC	2435-0602	Texas	Anderson
2436-0224		Matthew L. Hodges and Krissie L. Hodges	TRI-C RESOURCES, LLC	2436-0224	Texas	Anderson
2437-0689		Lana Lisa Schitoskey	TRI-C RESOURCES, LLC	2437-0689	Texas	Anderson
2437-0723		Betty Carolyn Fitzgerald Head	TRI-C RESOURCES, LLC	2437-0723	Texas	Anderson
2439-0025		Don D. Dupus	TRI-C RESOURCES, LLC	2439-0025	Texas	Anderson
2439-0026		Charles Wesley Black, III	TRI-C RESOURCES, LLC	2439-0026	Texas	Anderson
2440-0343		Peggy Marie Larson	TRI-C RESOURCES, LLC	2440-0343	Texas	Anderson
2440-0356		Charles Aaron Oldham, Jr.	TRI-C RESOURCES, LLC	2440-0356	Texas	Anderson
2440-0365		Peggy Lynn O'Neil Clair	TRI-C RESOURCES, LLC	2440-0365	Texas	Anderson
2441-0669		Wanda Jean Carter Kolkhorst	TRI-C RESOURCES, LLC	2441-0669	Texas	Anderson
2441-0687		The Estate of Virginia C. Crawford, Dec'd, by Jesse Carroll Crawford, Individually & as Co- Executor and Sun Trust Bank, Co-Executor By	TRI-C RESOURCES, LLC	2441-0687	Texas	Anderson
2443-0205		Lana Wardell	TRI-C RESOURCES, LLC	2443-0205	Texas	Anderson
2443-0217		Brenda Jo Kelly	TRI-C RESOURCES, LLC	2443-0217	Texas	Anderson
2443-0363		Arnold Ray Poppell	TRI-C RESOURCES, LLC	2443-0363	Texas	Anderson
2443-0371		Joseph E. Williams	TRI-C RESOURCES, LLC	2443-0371	Texas	Anderson
2443-0373		Arnold R. Poppell and wife, Mary Ann Poppell	TRI-C RESOURCES, LLC	2443-0373	Texas	Anderson
2447-0715		Robert Wayne Thornton	TRI-C RESOURCES, LLC	2447-0715	Texas	Anderson
2447-0717		Bennett Witten Poppell	TRI-C RESOURCES, LLC	2447-0717	Texas	Anderson
2447-0734		Mary Martha Herrington Conner, dealing in his sole and separate property	TRI-C RESOURCES, LLC	2447-0734	Texas	Anderson
2447-0736		Paul Carlton Herrington	TRI-C RESOURCES, LLC	2447-0736	Texas	Anderson
2447-0737		David Edwin Herrington	TRI-C RESOURCES, LLC	2447-0737	Texas	Anderson
2447-0749		Carol Ann Thornton	TRI-C RESOURCES, LLC	2447-0749	Texas	Anderson
2447-0755		Rose Marie O'Neil	TRI-C RESOURCES, LLC	2447-0755	Texas	Anderson
2447-0757	12/8/2014	Janice Minter	TRI-C RESOURCES, LLC	2447-0757	Texas	Anderson

Lease #	Date	Lessor	Lessee	Book-Page S	tate County
2447-0759	12/4/2014	Chris Price	TRI-C RESOURCES, LLC		exas Anderson
2447-0765	11/25/2014	Harold David Poppell	TRI-C RESOURCES, LLC	2447-0765 Te	exas Anderson
2447-0767		James Harland Price, as sole and separate property	TRI-C RESOURCES, LLC		exas Anderson
2450-0097		Janet Prince	TRI-C RESOURCES, LLC		exas Anderson
2450-0101	12/22/2014	Roy A. Simpson	TRI-C RESOURCES, LLC	2450-0101 Te	exas Anderson
2450-0103	12/22/2014	Karen Ann Morse Thompson	TRI-C RESOURCES, LLC	2450-0103 Te	exas Anderson
2452-0139	1/17/2015	Charles Leroy Wharton	TRI-C RESOURCES, LLC	2452-0139 Te	exas Anderson
2452-0140	1/19/2015	Jamie Stunkard a/k/a Jamie Lynn Miskell	TRI-C RESOURCES, LLC	2452-0140 Te	exas Anderson
2452-0141		Justin Hollis Miskell	TRI-C RESOURCES, LLC	2452-0141 Te	exas Anderson
2452-0142	1/15/2015	Billy Ralph Jenkins a/k/a Bill Ralph (Bill) Jenkins	TRI-C RESOURCES, LLC	2452-0142 Te	exas Anderson
2452-0147	1/22/2015	Larry Legard	TRI-C RESOURCES, LLC	2452-0147 Te	exas Anderson
2452-0149	12/11/2014	Katie Jeanette Price Pierce, a/k/a Katie Jeanette Price Pierce as sole and separate property	TRI-C RESOURCES, LLC	2452-0149 Te	exas Anderson
2452-0153	11/11/2014	Marsha Pickle Ceniceros	TRI-C RESOURCES, LLC	2452-0153 Te	exas Anderson
2452-0158	12/22/2014	Sidney A. Morse, Jr.	TRI-C RESOURCES, LLC	2452-0158 Te	exas Anderson
2452-0160	1/16/2015	Mark Price	TRI-C RESOURCES, LLC	2452-0160 Te	exas Anderson
2456-0120	2/11/2015	Laura W. Kirby Schoppe	TRI-C RESOURCES, LLC	2456-0120 Te	exas Anderson
2456-0121	2/13/2015	Kurt A. Kinnie	TRI-C RESOURCES, LLC	2456-0121 Te	exas Anderson
2456-0123	1/30/2015	Karen Nell Klein	TRI-C RESOURCES, LLC	2456-0123 Te	exas Anderson
2456-0124	1/30/2015	David Hunter Fitzgerald, as his sole and separate property	TRI-C RESOURCES, LLC	2456-0124 Te	exas Anderson
2456-0126		Rita K. Clark	TRI-C RESOURCES, LLC	2456-0126 Te	exas Anderson
2456-0135	1/20/2015	Stacy Christina Jeanes a/k/a Stacy Christina Miskell	TRI-C RESOURCES, LLC	2456-0135 Te	exas Anderson
2456-0138	1/17/2015	Deann Buffington, sole and separate property	TRI-C RESOURCES, LLC	2456-0138 Te	exas Anderson
2456-0139		Dolores J. Williams	TRI-C RESOURCES, LLC	2456-0139 Te	exas Anderson
2456-0140	1/27/2015	Karen R. Sauceda	TRI-C RESOURCES, LLC	2456-0140 Te	exas Anderson
2456-0141	1/15/2015	Harry A. Jenkins and wife, Carletta Jenkins	TRI-C RESOURCES, LLC	2456-0141 Te	exas Anderson
2456-0142	1/15/2015	Harry A. Jenkins, A/K/A Harry Albert Jenkins, as his sole and separate property	TRI-C RESOURCES, LLC	2456-0142 Te	exas Anderson
2456-0144	1/22/2015	Jerry LeGard, as his sole and separate property	TRI-C RESOURCES, LLC	2456-0144 Te	exas Anderson
2456-0146	2/6/2015	Olivia L. Paulino	TRI-C RESOURCES, LLC	2456-0146 Te	exas Anderson
2456-0148	2/5/2015	Roy C. Price and Evaline Price Family Living Trust, by Evaline Price, Trustee	TRI-C RESOURCES, LLC	2456-0148 Te	exas Anderson
2456-0150	1/28/2015	Marvalee K. May, as her sole and separate property	TRI-C RESOURCES, LLC	2456-0150 Te	exas Anderson
2456-0155	2/5/2015	Charles L Price, as sole and separate property	TRI-C RESOURCES, LLC	2456-0155 Te	exas Anderson
2456-0157	1/30/2015	James Holmes Beto Unit #1723901	TRI-C RESOURCES, LLC	2456-0157 Te	exas Anderson
2456-0159	2/3/2015	Andrew J. Overton, A/K/A Andrew J. Overton, III	TRI-C RESOURCES, LLC	2456-0159 Te	exas Anderson
2456-0160	1/29/2015	Harvey Stahl, sole and separate property	TRI-C RESOURCES, LLC	2456-0160 Te	exas Anderson
2456-0165	2/11/2015	Martha C. Kirby Cappo	TRI-C RESOURCES, LLC	2456-0165 Te	exas Anderson
2456-0166	2/11/2015	Frank M. Kirby 1995 Trust, by Martha C. Kirby Capo and Laura W. Kirby Schoppe,	TRI-C RESOURCES, LLC	2456-0166 Te	exas Anderson
2456-0168	2/11/2015	Mark Price a/k/a Mark Glynn Price	TRI-C RESOURCES, LLC	2456-0168 Te	exas Anderson
2456-0178	2/11/2015	Jennifer Fitzgerald, as her sole and separate property	TRI-C RESOURCES, LLC	2456-0178 Te	exas Anderson
2456-0180	2/11/2015	Carol Fitzgerald	TRI-C RESOURCES, LLC	2456-0178 Te	exas Anderson
2456-0182	2/2/2015	James Curtis Fitzgerald, A/K/A James Curtis "Bo" Fitzgerald, as his sole and separate property	TRI-C RESOURCES, LLC	2456-0182 Te	exas Anderson
2456-0184	2/11/2015	Jerry Fitzgerald, A/K/A Jared A. Fitzgerald, as sole and separate property	TRI-C RESOURCES, LLC	2456-0184 Te	exas Anderson
2456-0186	1/30/2015	Jimmy Joe Fitzgerald, as his sole and separate property	TRI-C RESOURCES, LLC	2456-0186 Te	exas Anderson
2456-0193	2/5/2015	Jesse L. Price, as sole and separate property	TRI-C RESOURCES, LLC	2456-0193 Te	exas Anderson
2456-0196	2/24/2015	Christopher Eric Gann	TRI-C RESOURCES, LLC	2456-0196 Te	exas Anderson
2456-0199	2/17/2015	Clifford Parks Gann, III a/k/a Trey Gann	TRI-C RESOURCES, LLC	2456-0199 Te	exas Anderson
2456-149	1/28/2015	Richard O. May, a/k/a Rick May, as his sole and separate property	TRI-C RESOURCES, LLC	2456-149 Te	exas Anderson
2458-0783	2/3/2015	Peggy Jean Holmes	TRI-C RESOURCES, LLC	2458-0783 Te	exas Anderson
2458-0797	2/11/2015	Robert Allen Fitzgerald Jr., sole and separate property	TRI-C RESOURCES, LLC	2458-0797 Te	exas Anderson
2459-0001	2/18/2015	Rebecca Dacus a/k/a Rebecca Lynn "Becky" Fitzgerald Dacus	TRI-C RESOURCES, LLC	2459-0001 Te	exas Anderson
2459-0003	3/2/2015	Gladys Fitzgerald	TRI-C RESOURCES, LLC	2459-0003 Te	exas Anderson
2459-0006	2/16/2015	David Clyde Richardson a/k/a David C. Richardson	TRI-C RESOURCES, LLC	2459-0006 Te	exas Anderson
2459-0010	3/4/2015	Henry William Schitoskey, sole and separate property	TRI-C RESOURCES, LLC	2459-0010 Te	exas Anderson
2459-0012	3/5/2015	Charles Preston Schitoskey	TRI-C RESOURCES, LLC	2459-0012 Te	exas Anderson
2459-0014		Joy Lee Anderson	TRI-C RESOURCES, LLC		exas Anderson
2459-0016		Tina Marie Oxford	TRI-C RESOURCES, LLC		exas Anderson
2459-0018		Linda Hurtado	TRI-C RESOURCES, LLC		exas Anderson
2459-0024		Clifford Parks Gann, III alk/a Trey Gann	TRI-C RESOURCES, LLC		exas Anderson
2460-0047		Stephen E. Thorpe	TRI-C RESOURCES, LLC		exas Anderson
2460-0050		Patricia Schitoskey, A/K/A Patricia Ann Schitoskey Crowell, as sole and separate property	TRI-C RESOURCES, LLC		exas Anderson
2460-0052		Mary Media Miles	TRI-C RESOURCES, LLC		exas Anderson
2460-0052		Gary W. Thorpe	TRI-C RESOURCES, LLC		exas Anderson
2460-0056		Harry G. Thorpe	TRI-C RESOURCES, LLC		exas Anderson exas Anderson
2460-0056		Gerald Lee Stewart	TRI-C RESOURCES, LLC		exas Anderson
2460-0066		Paula Kathleen Stewart Merritt	TRI-C RESOURCES, LLC		exas Anderson
2463-0071	3/26/2015	Michael Charles Pettiette	TRI-C RESOURCES, LLC	2463-0071 Te	exas Anderson

Lease #	Date	Lessor	Lessee	Book-Page	State_	County
2463-0076	3/17/2015		TRI-C RESOURCES, LLC	2463-0076	Texas	Anderson
2463-0078	3/13/2015	Janet Schitoskey, A/K/A Janet Lynn Schitoskey Johnson, as sole and separate property	TRI-C RESOURCES, LLC	2463-0078	Texas	Anderson
2463-0081	3/24/2015	Essie Thornton Henry	TRI-C RESOURCES, LLC	2463-0081	Texas	Anderson
2463-0084	3/2/2015	Carol Fitzgerald	TRI-C RESOURCES, LLC	2463-0084	Texas	Anderson
2463-0086	1/27/2015	Gayla J. Tobey	TRI-C RESOURCES, LLC	2463-0086	Texas	Anderson
2463-0089	3/12/2015	David Clyde Richardson a/k/a David C. Richardson	TRI-C RESOURCES, LLC	2463-0089	Texas	Anderson
2463-0090	3/12/2015	Steven Bradley Richardson	TRI-C RESOURCES, LLC	2463-0090	Texas	Anderson
2463-0093	3/17/2015	Mary Ellen Butler a/k/a Mary Ellen Price Butler	TRI-C RESOURCES, LLC	2463-0093	Texas	Anderson
2463-0096		Kimberly Shae Richardson	TRI-C RESOURCES, LLC	2463-0096	Texas	Anderson
2463-0100	3/23/2015		TRI-C RESOURCES, LLC	2463-0100	Texas	Anderson
2463-0101	3/2/2015	Annette Boudreaux a/k/a Annette Carswell, sole and separate property	TRI-C RESOURCES, LLC	2463-0101	Texas	Anderson
2463-0104	3/2/2015	Stephanie Lynn Parrish	TRI-C RESOURCES, LLC	2463-0104	Texas	Anderson
2463-0116 2463-0118	3/31/2015 3/5/2015		TRI-C RESOURCES, LLC TRI-C RESOURCES, LLC	2463-0116 2463-0118	Texas	Anderson Anderson
2463-0121	3/18/2015	Hilda Faye Schitoskey Mark Glynn Price	TRI-C RESOURCES, LLC	2463-0116	Texas	Anderson
2463-0122	3/19/2015		TRI-C RESOURCES, LLC	2463-0122	Texas	Anderson
2463-0124	3/30/2015	·	TRI-C RESOURCES, LLC	2463-0124	Texas	Anderson
2463-0126		Lameda Jo Poorman Burton	TRI-C RESOURCES, LLC	2463-0126	Texas	Anderson
2463-0135	1/30/2015	Charlie Holmes	TRI-C RESOURCES, LLC	2463-0135	Texas	Anderson
2463-0138	4/8/2015	Richard D. Dupree, a/k/a Richard B. Dupree	TRI-C RESOURCES, LLC	2463-0138	Texas	Anderson
2465-0054	4/2/2015	Roger Bonney	TRI-C RESOURCES, LLC	2465-0054	Texas	Anderson
2465-0056	4/2/2015	Roger Bonney, Trustee, Estate of Orrin H. Bonney, Trust #B	TRI-C RESOURCES, LLC	2465-0056	Texas	Anderson
2465-0058	4/14/2015	John Walter Ekey	TRI-C RESOURCES, LLC	2465-0058	Texas	Anderson
2465-0060	3/31/2015	Clifford Parks Gann, III a/k/a Trey Gann	TRI-C RESOURCES, LLC	2465-0060	Texas	Anderson
2465-0063	4/15/2015	Ida C. Gray, by and through her Agent and Attorney-in-Fact, June Nave	TRI-C RESOURCES, LLC	2465-0063	Texas	Anderson
2465-0065	3/31/2015	David C. Richardson	TRI-C RESOURCES, LLC	2465-0065	Texas	Anderson
2465-0069	4/22/2015	Partin and Partin Family Partnership, Ltd.	TRI-C RESOURCES, LLC	2465-0069	Texas	Anderson
2465-0075	1/30/2015	Mark Holmes	TRI-C RESOURCES, LLC	2465-0075	Texas	Anderson
2465-0078	4/7/2015	Gary Richard Williams a/k/a Richard Williams	TRI-C RESOURCES, LLC	2465-0078	Texas	Anderson
2465-0081	3/13/2015	·	TRI-C RESOURCES, LLC	2465-0081	Texas	Anderson
2467-0065		Lana Kay Wilbanks	TRI-C RESOURCES, LLC	2467-0065	Texas	Anderson
2467-0074		Harry Albert Jenkins	TRI-C RESOURCES, LLC	2467-0074	Texas	Anderson
2467-0075		Ms. Saville Inman	TRI-C RESOURCES, LLC	2467-0075	Texas	Anderson
2467-0077 2467-0081		Lucretia Saville Weems Nelda Yvonne Bristow	TRI-C RESOURCES, LLC TRI-C RESOURCES, LLC	2467-0077 2467-0081	Texas Texas	Anderson Anderson
2468-0527		Harry Albert Jenkins	TRI-C RESOURCES, LLC	2468-0527	Texas	Anderson
2468-0528	3/31/2015		TRI-C RESOURCES, LLC	2468-0528	Texas	Anderson
2468-0532		Billy Ralph Jenkins a/k/a Bill Ralph (Bill) Jenkins	TRI-C RESOURCES, LLC	2468-0532	Texas	Anderson
2470-0054		Lloyd Miller, Jr.	TRI-C RESOURCES, LLC	2470-0054	Texas	Anderson
2470-0055		Mary Miller	TRI-C RESOURCES, LLC	2470-0055	Texas	Anderson
2470-0067	5/12/2015	Billy Ralph Jenkins a/k/a Bill Ralph (Bill) Jenkins	TRI-C RESOURCES, LLC	2470-0067	Texas	Anderson
2470-0074	3/31/2015	Kimberly Shae Richardson	TRI-C RESOURCES, LLC	2470-0074	Texas	Anderson
2470-0075	3/31/2015	Steven Bradley Richardson	TRI-C RESOURCES, LLC	2470-0075	Texas	Anderson
2475-0194	7/8/2015	Billy Ralph Jenkins a/k/a Bill Ralph (Bill) Jenkins	TRI-C RESOURCES, LLC	2475-0194	Texas	Anderson
2475-0197	3/19/2015	J. Frank Shiflet and wife, Kathryn T. Shiflet	TRI-C RESOURCES, LLC	2475-0197	Texas	Anderson
2475-0202	7/8/2015	Charles Ashley Jenkins, Jr.	TRI-C RESOURCES, LLC	2475-0202	Texas	Anderson
2475-0204	6/26/2015	Patrick McIver	TRI-C RESOURCES, LLC	2475-0204	Texas	Anderson
2480-0247	7/21/2015	BBDASUKA, LLC	TRI-C RESOURCES, LLC	2480-0247	Texas	Anderson
2480-0249		Bratten Family Ranch, LLC	TRI-C RESOURCES, LLC	2480-0249	Texas	Anderson
2480-0253		Timothy W. Williams	TRI-C RESOURCES, LLC	2480-0253	Texas	Anderson
42.001.25000.0026		Huey Ray Wilbanks	O'Brien Resources, LLC	2537-0236	Texas	Anderson
42.001.25000.0027		Rebecca Anne Wilbanks	O'Brien Resources, LLC	2537-0233	Texas	Anderson
42.001.25000.0029 42.001.25000.0067	8/8/2013	Rose Washington Alan W. Mims	PETRO-LAND GROUP, INC. O'Brien Resources, LLC	2373-0032 2537-0185	Texas	Anderson Anderson
42.001.25000.0067		Alan W. Mims Tony Mims, aka Anthony G. Mims, Dealing in his sole and separate property	O'Brien Resources, LLC	2537-0189	Texas	Anderson
42.001.25000.0069		Angela K. Hart, dealing in her sole and separate property	O'Brien Resources, LLC	2537-0109	Texas	Anderson
42.001.25000.0070		Timothy E. West, and wife Juanima West, aka Charlotte J. West	O'Brien Resources, LLC	2537-0175	Texas	Anderson
42.001.25000.0071		Charlotte J. West aka Juanima West	O'Brien Resources, LLC	2537-0167	Texas	Anderson
42.001.25000.0072		Daniel Monroe Matthews, Dealing in his sole and separate property	O'Brien Resources, LLC	2537-0171	Texas	Anderson
42.001.25000.0073		Tina M. Moore, dealing in her sole and separate property	O'Brien Resources, LLC	2537-0195	Texas	Anderson
42.001.25000.0076		LaMerle Boyd	O'Brien Resources, LLC	2537-0181	Texas	Anderson
42.001.25000.0077	10/17/2016	Gerald W. Lasater	O'Brien Resources, LLC	2551-0281	Texas	Anderson
42.001.25000.0078	10/14/2016	Jeff Stringfield	O'Brien Resources, LLC	2551-0294	Texas	Anderson
42.001.25000.0079	9/26/2016	Edwina W. Huddleston fka Edwina Matthews Wall	O'Brien Resources, LLC	2537-0164	Texas	Anderson
42.001.25000.0080	10/17/2016	Robbie Mims aka Robert Mims, dealing in his sole and separate property	O'Brien Resources, LLC	2551-0283	Texas	Anderson

Lease #	Date	Lessor	Lessee	Book-Page	State	County
42.001.25000.0095	11/22/2014	Nan Hubert Tiller, a/k/a Nan Hubert Daugherty	TRI-C RESOURCES, LLC	2443-0365	Texas	Anderson
42.001.25000.0096	11/25/2014	Patsy Jean Williams	TRI-C RESOURCES, LLC	2443-0375	Texas	Anderson
42.001.25000.0097	11/25/2014	Wanda L. Sanders	TRI-C RESOURCES, LLC	2443-0377	Texas	Anderson
42.001.25000.0099	11/25/2014	Dolores Walls Lambright, Deceased	TRI-C RESOURCES, LLC	2447-0735	Texas	Anderson
42.001.25000.0100	12/16/2014	Maxine Coleman Walls	TRI-C RESOURCES, LLC	2450-0113	Texas	Anderson
42.001.25000.0101	12/17/2014	Sharon Lee Walls Rightmer	TRI-C RESOURCES, LLC	2450-0114	Texas	Anderson
42.001.25000.0102	12/17/2014	Becky Rayne Walls a/k/a Becky Walls Kittleman	TRI-C RESOURCES, LLC	2450-0115	Texas	Anderson
42.001.25000.0103	12/17/2014	Dewey Ray Walls, Jr.	TRI-C RESOURCES, LLC	2450-0116	Texas	Anderson
42.001.25000.0104	7/20/2016	Lucinda Hanks Presley	O'Brien Resources, LLC	2530-0696	Texas	Anderson
42.001.25000.0105	7/20/2016	Jackson Hanks	O'Brien Resources, LLC	2530-0698	Texas	Anderson
42.001.25000.0106	7/20/2016	The C.W. Hanks Jr. Childrens Trust, by Jackson R. Hanks, Trustee	O'Brien Resources, LLC	2530-0697	Texas	Anderson
42.001.25000.0107	8/12/2016	Ronald Mark Hill	O'Brien Resources, LLC	2537-0242	Texas	Anderson
42.001.25000.0108	8/12/2016	Michael Charles Pettiette	O'Brien Resources, LLC	2537-245	Texas	Anderson
42.001.25000.0109	8/12/2016	Diane Hill Duncan	O'Brien Resources, LLC	2537-0227	Texas	Anderson
42.001.25000.0139	1/20/2017	Clifford E. Bomer	O'Brien Resources, LLC	2556-0391	Texas	Anderson
42.001.25000.0140	9/26/2016	Bobby A. Huddleston and Edwina W. Huddleston, Trustees for the B.A. & E.W. Huddleston Living Trust, dated July 26, 2012	O'Brien Resources, LLC	2537-0161	Texas	Anderson
42.001.25000.0141		Jackie Dwight Hill	O'Brien Resources, LLC	2537-239	Texas	Anderson
42.001.25000.0142		Helen Hartz Montgomery Freund Test. Trust	O'Brien Resources, LLC	2537-222	Texas	Anderson
42.001.25000.0143		Joe Ladd Faulk	O'Brien Resources, LLC	2537-205	Texas	Anderson
42.001.25000.0144		Harold Eugene Hughes, Jr.	O'Brien Resources, LLC	2537-218	Texas	Anderson
42.001.25000.0168		Floyd H. Davis	PETRO-LAND GROUP, INC.	2393-0381	Texas	Anderson
42.001.25000.0169		Robert K. Smith Trust, Raymond B. Keating, III Trustee	PETRO-LAND GROUP, INC.	2403-0457	Texas	Anderson
42.001.25000.0170		Virginia Howe Smith Royalty Oil and Gas Trust, Susan Lorigan Trustee c/o Raymond B. Keating, III	PETRO-LAND GROUP, INC.	2403-0456 2403-0458	Texas	Anderson
42.001.25000.0171 42.001.25000.0172		Louise H. Davis Irrevocable Trust, by William Villas Hanks, Trustee William Noble Smith, Jr. 1984 Trust, by Raymond B. Keating, III, Trustee	PETRO-LAND GROUP, INC. PETRO-LAND GROUP, INC.	2403-0459	Texas	Anderson Anderson
		William Noble Smith, Jr. 1994 Trust, by Kaymond B. Keating, III, Trustee Jean E. Smith			Texas	Anderson
42.001.25000.0173 42.001.25000.0178		Jean E. Smith	PETRO-LAND GROUP, INC. PETRO-LAND GROUP, INC.	2412-0147 2403-0464	Texas	Anderson
42.001.25000.0179		Gregory C. Smith	PETRO-LAND GROUP, INC.	2403-0493	Texas	Anderson
42.001.25000.0179		Kathryn Quinn Reid	PETRO-LAND GROUP, INC.	2393-0384	Texas	Anderson
42.001.25000.0182		Carol Quinn Culp	PETRO-LAND GROUP, INC.	2393-0386	Texas	Anderson
42.001.25000.0183		Scott Finlay Sommer	PETRO-LAND GROUP, INC.	2393-0388	Texas	Anderson
42.001.25000.0190		Mary Ann Allen	O'Brien Resources, LLC	2551-0092	Texas	Anderson
42.001.25000.0196		Melba Carol Pugh	O'Brien Resources, LLC	2540-0085	Texas	Anderson
42.001.25000.0197		Jimmy Darrell Allen	O'Brien Resources, LLC	2540-0082	Texas	Anderson
42.001.25000.0198		Marilyn Richardson Holmes	O'Brien Resources, LLC	2540-0094	Texas	Anderson
42.001.25000.0199		Lana C. Caperton	O'Brien Resources, LLC	2551-0116	Texas	Anderson
42.001.25000.0200	9/29/2016	Jennifer Carol Rummel	O'Brien Resources, LLC	2540-0088	Texas	Anderson
42.001.25000.0201	9/29/2016	Jerry Lynn Pugh, II	O'Brien Resources, LLC	2540-0091	Texas	Anderson
42.001.25000.0202	10/25/2016	Quinclyn L. Roland	O'Brien Resources, LLC	2551-0124	Texas	Anderson
42.001.25000.0203	9/20/2016	Beverly G. Skelton	O'Brien Resources, LLC	2551-0120	Texas	Anderson
42.001.25000.0204	10/20/2016	William H. Faulk & Mary Lou Faulk, indiv. & as Co-Trustees of the Faulk Family Trust	O'Brien Resources, LLC	2551-0287	Texas	Anderson
42.001.25000.0205	9/16/2016	Clifton Graham Williams, Jr.	O'Brien Resources, LLC	2551-0289	Texas	Anderson
42.001.25000.0206	9/16/2016	Clifton G. Williams, Jr.	O'Brien Resources, LLC	2544-0696	Texas	Anderson
42.001.25000.0207	9/16/2016	Jimmy C. Williams	O'Brien Resources, LLC	2544-0690	Texas	Anderson
42.001.25000.0208	10/31/2016	Jimmy C. Williams	O'Brien Resources, LLC	2544-0693	Texas	Anderson
42.001.25000.0209	10/14/2016	Jason Stringfield	O'Brien Resources, LLC	2551-0292	Texas	Anderson
42.001.25000.0210	10/14/2016	David Stringfield	O'Brien Resources, LLC	2551-0290	Texas	Anderson
42.001.25000.0211	9/22/2016	John H. Hansen	O'Brien Resources, LLC	2551-0296	Texas	Anderson
42.001.25000.0212	9/20/2016	Mark Winfred Hughes	O'Brien Resources, LLC	2551-0128	Texas	Anderson
42.001.25000.0213	11/3/2016	Boles Children's Home, Inc.	O'Brien Resources, LLC	2544-0681	Texas	Anderson
42.001.25000.0214	9/16/2016	Ida F. Williams	O'Brien Resources, LLC	2544-0684	Texas	Anderson
42.001.25000.0215		Ida F. Williams	O'Brien Resources, LLC	2544-0687	Texas	Anderson
42.001.25000.0217		Beverly Grace Skelton	PETRO-LAND GROUP, INC.	2399-0505	Texas	Anderson
42.001.25000.0218		Lana C. Caperton	PETRO-LAND GROUP, INC.	2393-0407	Texas	Anderson
42.001.25000.0219	1/3/2014	Quinclyn L. Roland	PETRO-LAND GROUP, INC.	2399-0501	Texas	Anderson
42.001.25000.0220	1/3/2014	Winfred E. Adams	PETRO-LAND GROUP, INC.	2393-0420	Texas	Anderson
42.001.25000.0223	1/6/2014	Harold Eugene Hughes, Jr.	PETRO-LAND GROUP, INC.	2399-0572	Texas	Anderson
42.001.25000.0224	1/6/2014	Malcolm Adams Hughes	PETRO-LAND GROUP, INC.	2399-0564	Texas	Anderson
42.001.25000.0225	1/6/2014	Mark Winfred Hughes	PETRO-LAND GROUP, INC.	2399-0565	Texas	Anderson
42.001.25000.0226	1/6/2014	Randy Leroy Adams	PETRO-LAND GROUP, INC.	2393-0415	Texas	Anderson
42.001.25000.0227	1/6/2014	Thomas William Hughes	PETRO-LAND GROUP, INC.	2399-0521	Texas	Anderson
42.001.25000.0228	1/8/2014	Johney Lee Williams	PETRO-LAND GROUP, INC.	2399-0545	Texas	Anderson
42.001.25000.0236		Edgar Murphy Douglas	PETRO-LAND GROUP, INC.	2399-0499	Texas	Anderson
42.001.25000.0247		Kirk Anthony Kownslar	O'Brien Resources, LLC	2544-0699	Texas	Anderson
42.001.25000.0248	11/10/2016	Sandra Grace Sursa	O'Brien Resources, LLC	2551-0132	rexas	Anderson

Lease #	Date Lessor	Lessee	Book-Page St	ate County
42.001.25000.0249	10/31/2016 Kirk Anthony Kownslar	O'Brien Resources, LLC	2544-0702 Te:	xas Anderson
42.001.25000.0250	11/16/2016 Thomas William Hughes	O'Brien Resources, LLC	2551-0112 Te:	xas Anderson
42.001.25000.0251	11/29/2016 Randy D. Tinsley and Marjean Tinsley Trust	O'Brien Resources, LLC	2551-0108 Te:	xas Anderson
42.001.25000.0252	11/29/2016 THE KNOX AND MADELINE SCOTT 2017 LIVING TRUST by Knox O. Scott , Jr. and Madeline Q. Scott as Co-Trustees	O'Brien Resources, LLC	2551-0136 Te:	xas Anderson
42.001.25000.0253	11/29/2016 William L. Scott	O'Brien Resources, LLC	2551-0142 Te:	xas Anderson
42.001.25000.0254	1/26/2017 Jerry Lynn Pugh, II	O'Brien Resources, LLC	2551-0098 Te:	xas Anderson
42.001.25000.0255	1/26/2017 Melba Carol Pugh	O'Brien Resources, LLC	2551-0095 Te:	xas Anderson
42.001.25000.0256	1/26/2017 Jennifer Carol Rummel	O'Brien Resources, LLC	2551-0101 Te:	xas Anderson
42.001.25000.0266	2/6/2014 Sandra Grace Sursa	PETRO-LAND GROUP, INC.	2399-0548 Te:	xas Anderson
42.001.25000.0267	2/7/2014 Petroleum Unlimited Family Limited Partnership	PETRO-LAND GROUP, INC.	2403-0454 Te:	xas Anderson
42.001.25000.0277	11/16/2016 Malcolm Adams Hughes	O'Brien Resources, LLC	2551-0104 Te:	xas Anderson
42.001.25000.0278	1/4/2017 Billy Goebel Byars Trust U/W/O B.G. Byars, dec., et al	O'Brien Resources, LLC	2551-0090 Te:	xas Anderson
42.001.25000.0279	1/19/2017 Cole O'Neill Ltd.	O'Brien Resources, LLC	2552-0237 Te:	xas Anderson
42.001.25000.0280	1/19/2017 Frances Royali Alexander	O'Brien Resources, LLC	2552-0233 Te:	xas Anderson
42.001.25000.0281	1/20/2017 Frances Royall Alexander	O'Brien Resources, LLC	2552-0231 Te:	xas Anderson
42.001.25000.0282	1/19/2017 Rebecca Royall	O'Brien Resources, LLC	2552-0225 Te:	xas Anderson
42.001.25000.0283	1/20/2017 Rebecca Royall	O'Brien Resources, LLC	2552-0223 Te:	xas Anderson
42.001.25000.0284	1/23/2017 Royall Company Shareholders Trust	O'Brien Resources, LLC	2552-0221 Te	xas Anderson
42.001.25000.0285	1/23/2017 Royall Fixed Investment Trust - 2000	O'Brien Resources, LLC	2552-0219 Te:	xas Anderson
42.001.25000.0286	1/19/2017 Thomas M. Colley Residual Trust	O'Brien Resources, LLC	2552-0235 Te:	xas Anderson
42.001.25000.0287	1/19/2017 Tucker B. Royall	O'Brien Resources, LLC	2552-0229 Te:	xas Anderson
42.001.25000.0288	1/20/2017 Tucker B. Royall	O'Brien Resources, LLC	2552-0227 Te:	xas Anderson
42.001.25000.0289	1/18/2017 Winfred E. Adams	O'Brien Resources, LLC	2556-0422 Te:	xas Anderson
42.001.25000.0298	3/19/2014 Georgi Davis Duwe	PETRO-LAND GROUP, INC.	2412-0153 Te:	xas Anderson
42.001.25000.0301	3/19/2014 Priscilla Davis Gravely	PETRO-LAND GROUP, INC.	2412-0174 Te:	xas Anderson
42.001.25000.0309	1/20/2017 Clifford E. Bomer	O'Brien Resources, LLC	2556-0380 Te:	xas Anderson
42.001.25000.0324	2/8/2017 Charles H. Ham Revocable Living Trust	O'Brien Resources, LLC	2564-0022 Te:	xas Anderson
42.001.25000.0325	1/1/2017 Glenn L. Brinkman and Carolyn L. Brinkman	O'Brien Resources, LLC	2564-0026 Te:	xas Anderson
42.001.25000.0326	1/24/2017 Richard A. Kelly	O'Brien Resources, LLC	2564-0019 Te:	xas Anderson
42.001.25000.0327	1/19/2017 Randy Leroy Adams	O'Brien Resources, LLC	2560-338 Te	xas Anderson
42.001.25000.0328	2/23/2017 Joe Marvin Wardell	O'Brien Resources, LLC	2654-0013 Te:	xas Anderson
42.001.25000.0329	2/23/2017 John P. Wardell	O'Brien Resources, LLC	2564-0010 Te:	xas Anderson
42.001.25000.0330	2/23/2017 Mary Wardell Ballard	O'Brien Resources, LLC	2564-0007 Te:	
42.001.25000.0331	2/7/2017 Kay L. Ocnaschek	O'Brien Resources, LLC	2564-0016 Te:	xas Anderson
42.001.25000.0332	2/14/2017 Mary Jo Wardell	O'Brien Resources, LLC	2564-0034 Te:	xas Anderson
42.001.25000.0333	3/10/2017 Orval Herrington	O'Brien Resources, LLC	2561-0524 Te:	xas Anderson
42.001.25000.0334	1/24/2017 Dorothy Ann Fowler	O'Brien Resources, LLC		xas Anderson
42.001.25000.0335	3/1/2017 Sharon Dowd Family Trust	O'Brien Resources, LLC		xas Anderson
42.001.25000.0336	3/9/2017 Donna G. Sierks	O'Brien Resources, LLC	2561-0522 Te:	
42.001.25000.0341	4/12/2017 Margaret Blake Hall	O'Brien Resources, LLC	2576-0193 Te:	xas Anderson
42.001.25000.0344	11/22/2016 Christy Lee Lambright	O'Brien Resources, LLC	2564-0030 Te:	
42.001.25000.0345	3/22/2017 Kimberly Kay Richardson Benson	O'Brien Resources, LLC	2563-0790 Te:	
42.001.25000.0346	3/22/2017 Paul R. Benson	O'Brien Resources, LLC	2563-0795 Te:	
42.001.25000.0347	2/23/2017 Mark Wardell	O'Brien Resources, LLC	2564-0004 Te:	
42.001.25000.0349	3/1/2017 Glenna Odom	O'Brien Resources, LLC		xas Anderson
42.001.25000.0350	3/1/2017 Laurie Rich	O'Brien Resources, LLC		xas Anderson
42.001.25000.0351	4/14/2017 Nelda Joyce Spence	O'Brien Resources, LLC		xas Anderson
42.001.25000.0352	4/7/2017 Rose Washington	O'Brien Resources, LLC	2576-0343 Te:	
42.001.25000.0353	4/14/2017 Lola W. Magee	O'Brien Resources, LLC	2576-0339 Te:	
42.001.25000.0354	4/14/2017 Charles R. Grasty	O'Brien Resources, LLC		xas Anderson
42.001.25000.0355	4/19/2017 Frieda Elliott Osborne	O'Brien Resources, LLC		xas Anderson
42.001.25000.0356	4/19/2017 Cynthia Davis Wright	O'Brien Resources, LLC		xas Anderson
42.001.25000.0357	4/14/2017 George W. Vannoy	O'Brien Resources, LLC	2576-0325 Te:	
42.001.25000.0358	4/14/2017 Carol A. Harrison	O'Brien Resources, LLC	2576-0313 Te:	
42.001.25000.0360	3/22/2017 Christel Renate Laughlin	O'Brien Resources, LLC		xas Anderson
42.001.25000.0361	4/14/2017 Patti Lynn Means	O'Brien Resources, LLC		xas Anderson
42.001.25000.0361	4/19/2017 James G. Thomas	O'Brien Resources, LLC		xas Anderson
42.001.25000.0362	4/28/2017 Edward Lowell DeLoach, Jr.	O'Brien Resources, LLC		xas Anderson
42.001.25000.0363		O'Brien Resources, LLC	2576-0288 Te:	
	4/14/2017 Gary D. Thomas			
42.001.25000.0365	4/14/2017 Homer A. Vannoy	O'Brien Resources, LLC		xas Anderson
42.001.25000.0366	4/14/2017 James Larry Thomas	O'Brien Resources, LLC		xas Anderson
42.001.25000.0367	4/19/2017 Jenni Davis Bridges	O'Brien Resources, LLC		xas Anderson
42.001.25000.0368	4/14/2017 Sharon Sue Givens	O'Brien Resources, LLC		xas Anderson
42.001.25000.0369	4/27/2017 Vicki Ann Champagne	O'Brien Resources, LLC	2576-0284 Te:	
42.001.25000.0370	3/20/2017 Margaret Jordan Royalties, Inc.	O'Brien Resources, LLC	2561-0526 Te:	xas Anderson

Lease #	Date	Lessor	Lessee	Book-Page	State	County
42.001.25000.0371	4/26/2017	Annie I. Wars	O'Brien Resources, LLC	2576-0272	Texas	Anderson
42.001.25000.0372	4/19/2017	Charles C. Thomas	O'Brien Resources, LLC	2576-0244	Texas	Anderson
42.001.25000.0373	4/28/2017	Cynthia Straus Bell	O'Brien Resources, LLC	2576-0240	Texas	Anderson
42.001.25000.0374	4/14/2017	Doris N. Vannoy	O'Brien Resources, LLC	2576-0248	Texas	Anderson
42.001.25000.0375	4/28/2017	Linda Elise Sherrill	O'Brien Resources, LLC	2576-0276	Texas	Anderson
42.001.25000.0376	4/14/2017	Roy Lee Vannoy	O'Brien Resources, LLC	2576-0255	Texas	Anderson
42.001.25000.0377	3/1/2017	Solnick Family Trust	O'Brien Resources, LLC	2576-0252	Texas	Anderson
42.001.25000.0378	4/28/2017	William Boyd Manning	O'Brien Resources, LLC	2576-0259	Texas	Anderson
42.001.25000.0379	4/28/2017	Virgil Allen Cox, Jr.	O'Brien Resources, LLC	2576-0208	Texas	Anderson
42.001.25000.0380	4/14/2017	Judy Willhite Elliff	O'Brien Resources, LLC	2576-0212	Texas	Anderson
42.001.25000.0381	4/14/2017	Clayton L. Grasty	O'Brien Resources, LLC	2576-0216	Texas	Anderson
42.001.25000.0382	4/26/2017	Kenneth Hogg	O'Brien Resources, LLC	2576-0236	Texas	Anderson
42.001.25000.0383	4/26/2017	Loretta Faye Hogg, FKA Loretta Faye Hobbs	O'Brien Resources, LLC	2576-0232	Texas	Anderson
42.001.25000.0384	4/14/2017	Leon A. Willhite	O'Brien Resources, LLC	2576-0220	Texas	Anderson
42.001.25000.0385	4/5/2017	Mary Alice Adamson	O'Brien Resources, LLC	2576-0228	Texas	Anderson
42.001.25000.0386	4/14/2017	William Lee Bishop, Sr.	O'Brien Resources, LLC	2576-0224	Texas	Anderson
42.001.25000.0387	7/9/2014	Donna Jackson Rhymes	TRI-C RESOURCES, LLC	2424-0088	Texas	Anderson
42.001.25000.0388	7/15/2014	Paula E. Trigg	TRI-C RESOURCES, LLC	2424-0476	Texas	Anderson
42.001.25000.0389	7/15/2014	Michael G. Bernard	TRI-C RESOURCES, LLC	2425-0514	Texas	Anderson
42.001.25000.0390	7/21/2014	Martha Etta Adams Gandy	TRI-C RESOURCES, LLC	2427-0573	Texas	Anderson
42.001.25000.0391	7/23/2014		TRI-C RESOURCES, LLC	2424-0654	Texas	Anderson
42.001.25000.0392	7/24/2014	Jackie Dwight Hill	TRI-C RESOURCES, LLC	2424-0415	Texas	Anderson
42.001.25000.0393		Ronald Mark Hill	TRI-C RESOURCES, LLC	2424-0418	Texas	Anderson
42.001.25000.0394		Diane Hill Duncan	TRI-C RESOURCES, LLC	2424-0421	Texas	Anderson
42.001.25000.0395		David H. McLellan	TRI-C RESOURCES, LLC	2424-0426	Texas	Anderson
42.001.25000.0396		Estate of Geraldine Robertson, by Owen Lee Robertson, Individually and as Independent Executor	TRI-C RESOURCES, LLC	2424-0447	Texas	Anderson
42.001.25000.0397	6/20/2017	Jamie Pettiette Rhone	O'Brien Resources, LLC	2576-0184	Texas	Anderson
42.001.25000.0398	5/12/2017		O'Brien Resources, LLC	2576-0198	Texas	Anderson
42.001.25000.0399		Robbie L. Davis, Jr.	O'Brien Resources, LLC	2576-0347	Texas	Anderson
42.001.25000.0400		Kerrie Lynn Laza	O'Brien Resources, LLC	2587-0411	Texas	Anderson
42.001.25000.0401	6/23/2017		O'Brien Resources, LLC	2587-0409	Texas	Anderson
42.001.25000.0402 42.001.25000.0403		Annie Lou Moy	O'Brien Resources, LLC O'Brien Resources, LLC	2576-0351 2576-0181	Texas	Anderson Anderson
42.001.25000.0403		Nan Beth Johnson McReynolds Michael James Johnson	O'Brien Resources, LLC	2577-0268	Texas	Anderson
42.001.25000.0405	6/29/2017		O'Brien Resources, LLC	2576-0178	Texas	Anderson
42.001.25000.0407		Dorothy Conard Newsom	O'Brien Resources, LLC	2010 0110	Texas	Anderson
42.001.25000.0408	6/23/2017	Randal Douglas Honeycutt	O'Brien Resources, LLC	2587-0408	Texas	Anderson
42.001.25000.0409	9/8/2014	Charlotte P. Maxwell	TRI-C RESOURCES, LLC	2435-0589	Texas	Anderson
42.001.25000.0410		Bonnie G. Baiamonte	TRI-C RESOURCES, LLC	2437-0726	Texas	Anderson
42.001.25000.0411		Jeanne Aillet	TRI-C RESOURCES, LLC	2437-0724	Texas	Anderson
42.001.25000.0412	6/29/2017	Felesha Ann Howell	O'Brien Resources, LLC	2578-0512	Texas	Anderson
42.001.25000.0413	7/19/2017	Stephen Paul Johnson	O'Brien Resources, LLC	2578-0509	Texas	Anderson
42.001.25000.0414	7/25/2017	Donn W. Conard	O'Brien Resources, LLC	257-0506	Texas	Anderson
42.001.25000.0415	4/19/2017	Lecil Conard Johnson	O'Brien Resources, LLC	2584-0559	Texas	Anderson
42.001.25000.0416	6/1/2017	Jayme Payne	O'Brien Resources, LLC	2584-0555	Texas	Anderson
42.001.25000.0417	6/16/2017	Linda Cormack	O'Brien Resources, LLC	2584-0552	Texas	Anderson
42.001.25000.0418	7/27/2017	Nancy M. Bryant	O'Brien Resources, LLC	2578-0503	Texas	Anderson
42.001.25000.0419	4/19/2017	Shirley Ann Summerall	O'Brien Resources, LLC	2583-0030	Texas	Anderson
42.001.25000.0420	7/25/2017	Annie E. Isabell	O'Brien Resources, LLC	2579-0859	Texas	Anderson
42.001.25000.0421	7/25/2017	Donald Ray Thomas	O'Brien Resources, LLC	2579-0862	Texas	Anderson
42.001.25000.0422	7/25/2017	David G. Thomas	O'Brien Resources, LLC	2579-0856	Texas	Anderson
42.001.25000.0423	6/29/2017	Hal Johnson	O'Brien Resources, LLC	2579-0865	Texas	Anderson
42.001.25000.0424	8/4/2017	James Roger Myers	O'Brien Resources, LLC	2579-0868	Texas	Anderson
42.001.25000.0425	6/1/2017	Kady Payne	O'Brien Resources, LLC	2583-0026	Texas	Anderson
42.001.25000.0426	9/5/2017	Judy Newman	O'Brien Resources, LLC	583-0034	Texas	Anderson
42.001.25000.0427	10/14/2014	Donna Jackson Rhymes	TRI-C RESOURCES, LLC	2439-0023	Texas	Anderson
42.001.25000.0428	11/3/2014	Kathryn T. Cathey	TRI-C RESOURCES, LLC	2447-0742	Texas	Anderson
42.001.25000.0429	7/27/2017	Joe R. Taylor and wife, Evelyn Taylor	O'Brien Resources, LLC	2588-0337	Texas	Anderson
42.001.25000.0430	12/11/2014	Carla Dee Borders	TRI-C RESOURCES, LLC	2456-0132	Texas	Anderson
42.001.25000.0431	12/1/2014	Anadarko E&P Onshore LLC	TRI-C RESOURCES, LLC	2450-0094	Texas	Anderson
42.001.25000.0432	11/26/2013	Sandra Marie Cone Cole, aka Sandra Marie Cone	PETRO-LAND GROUP, INC.	2393-0364	Texas	Anderson
42.001.25000.0433	6/28/2014	Thomas Mark Cooper alk/a T. Mark Cooper and Barbara Cooper	PETRO-LAND GROUP, INC.	2422-0001	Texas	Anderson
42.001.25000.0434		Elizabeth Jean Henry Cosper	PETRO-LAND GROUP, INC.	2359-0219	Texas	Anderson
42.001.25000.0436	3/19/2015		TRI-C RESOURCES, LLC	2475-0190	Texas	Anderson
42.001.25000.0437	7/9/2014	Carol Fitzgerald	TRI-C RESOURCES, LLC	2429-0159	Texas	Anderson

Lease #	Date	Lessor	Lessee	Book-Page State	County
42.001.25000.0438	11/12/2012	Darrell R. Henry	PETRO-LAND GROUP, INC.	2359-0394 Texas	Anderson
42.001.25000.0439	11/9/2012	David Lee Henry	PETRO-LAND GROUP, INC.	2359-0359 Texas	Anderson
42.001.25000.0440	9/30/2015	David Lee Henry	TRI-C RESOURCES, LLC	2489-0701 Texas	Anderson
42.001.25000.0441	11/9/2012	Grady A. Henry	PETRO-LAND GROUP, INC.	2359-0400 Texas	Anderson
42.001.25000.0442	11/12/2012	James Larry Henry	PETRO-LAND GROUP, INC.	2359-0402 Texas	Anderson
42.001.25000.0443	11/9/2012	Joseph Dwayne Henry	PETRO-LAND GROUP, INC.	2359-0403 Texas	Anderson
42.001.25000.0444	11/12/2012	Mary Patricia Henry	PETRO-LAND GROUP, INC.	2359-0404 Texas	Anderson
42.001.25000.0445	11/9/2012	Michael Heath Henry	PETRO-LAND GROUP, INC.	2359-0405 Texas	Anderson
42.001.25000.0446	12/9/2013	Curtis Louise Hoehn Lancer	PETRO-LAND GROUP, INC.	2393-0399 Texas	Anderson
42.001.25000.0447	11/26/2013	Laurin M. Linson	PETRO-LAND GROUP, INC.	2393-0374 Texas	Anderson
42.001.25000.0448	12/16/2014	Mary Ann Cathey O'Neill, a/k/a Mary Ann O'Neill	TRI-C RESOURCES, LLC	2447-0761 Texas	Anderson
42.001.25000.0449	12/9/2013	Mary Charlotte Hoehn Payne	PETRO-LAND GROUP, INC.	2412-0181 Texas	Anderson
42.001.25000.0450	11/29/2012	Dwight L. Phifer	Northwest Oil & Gas	2329-0126 Texas	Anderson
42.001.25000.0451	6/25/2014	Roger G. Schloo	PETRO-LAND GROUP, INC.	2422-0448 Texas	Anderson
42.001.25000.0452	8/15/2014	Joycelyn Kay Shearin	TRI-C RESOURCES, LLC	2440-0363 Texas	Anderson
42.001.25000.0453	1/21/2014	Annette Verner Sherman	PETRO-LAND GROUP, INC.	2399-0551 Texas	Anderson
42.001.25000.0454	11/13/2012	Nelda Jean Smith, fka Nelda Jean Hussey	PETRO-LAND GROUP, INC.	2360-0220 Texas	Anderson
42.001.25000.0455	12/12/2012	Kay Henry Steel	PETRO-LAND GROUP, INC.	2360-0239 Texas	Anderson
42.001.25000.0456	3/19/2015	The Bishop Quinn Foundation	TRI-C RESOURCES, LLC	2473-0211 Texas	Anderson
2452-0144	1/22/2015	James Harland Price	TRI-C RESOURCES, LLC	2452-0144 Texas	Anderson
2452-0146	1/22/2015	Katie Jeanette Price Pierce, a/k/a Katie Jeanette Price Pierce as sole and separate property	TRI-C RESOURCES, LLC	2452-0146 Texas	Anderson
42.213.16200.0029	7/8/2014	Bonnie Imogene Lackey by Susan E. Sallee AIF	O'Brien Resources, LLC	2015-00000875 Texas	Henderson
42.213.16200.0034	7/8/2014	Beverly Louise Beck	O'Brien Resources, LLC	2015-00000909 Texas	Henderson
42.213.16200.0036	7/1/2014	Betty Joyce Brotherton	O'Brien Resources, LLC	2015-00000906 Texas	Henderson
42.213.16200.0042	7/3/2014	Herschel D. Costlow	O'Brien Resources, LLC	2015-00000905 Texas	Henderson
42.213.16200.0043	7/3/2014	George M. Costlow	O'Brien Resources, LLC	2015-00000904 Texas	Henderson
42.213.16200.0048	7/15/2014	The Arabella Wofford Family Mineral Tr by Thomas W. Craddock, Trustee	O'Brien Resources, LLC	2015-00000890 Texas	Henderson
42.213.16200.0049	7/3/2014	Rebecca Anne Wilbanks	O'Brien Resources, LLC	2015-00000895 Texas	Henderson
42.213.16200.0050	7/7/2014	Melinda Easterwood	O'Brien Resources, LLC	2015-00000893 Texas	Henderson
42.213.16200.0053	7/3/2014	Huey Ray Wilbanks	O'Brien Energy Company	2015-00000892 Texas	Henderson
42.213.16200.0057	7/9/2014	Rocky Bachus	O'Brien Resources, LLC	2015-00000885 Texas	Henderson
42.213.16200.0057	7/9/2014	Rocky Bachus	O'Brien Resources, LLC	2015-00000885 Texas	Henderson
42.213.16200.0062	7/22/2014	Sherri Posey	O'Brien Resources, LLC	2015-00000871 Texas	Henderson
42.213.16200.0070	7/14/2014	Nowell E. Loop AGT & AIF for Michael Finley Robertson	O'Brien Resources, LLC	2015-00000862 Texas	Henderson
42.213.16200.0073	7/14/2014	Nowell Loop for Gertrude J. Smither, Deceased	O'Brien Resources, LLC	2015-00000861 Texas	Henderson
42.213.16200.0074	7/17/2014	Winnell Cason Herrington	O'Brien Resources, LLC	2015-00000873 Texas	Henderson
42.213.16200.0075	7/14/2014	Marshall Groce	O'Brien Resources, LLC	2015-00000868 Texas	Henderson
42.213.16200.0076	7/15/2014	T. B. Wofford Royalty, LLC by Emily Fredricks Wofford as Partner	O'Brien Resources, LLC	2015-00000879 Texas	Henderson
42.213.16200.0077	7/22/2014	James Posey	O'Brien Resources, LLC	2015-00000854 Texas	Henderson
42.213.16200.0078	7/17/2014	Emma V. Morton	O'Brien Resources, LLC	2015-00000855 Texas	Henderson
42.213.16200.0081	7/17/2014	Clarence M. Morton	O'Brien Resources, LLC	2015-00000856 Texas	Henderson
42.213.16200.0082	7/17/2014	Carolyn Morton Walker	O'Brien Resources, LLC	2015-00000859 Texas	Henderson
42.213.16200.0083	8/4/2014	Larry Travis Costlow	O'Brien Resources, LLC	2015-00000852 Texas	Henderson
42.213.16200.0085	7/14/2014	Lawrence E. Barclay, Jr.Deceased by Nowell Loop Agt & AIF	O'Brien Resources, LLC	2015-00000883 Texas	Henderson
42.213.16200.0086	8/7/2014	Roy Lee Dosser	O'Brien Resources, LLC	2015-00000847 Texas	Henderson
42.213.16200.0088	7/22/2014	Theresa Lynn Scarber	O'Brien Resources, LLC	2015-00000846 Texas	Henderson
42.213.16200.0102	8/7/2014	Teresa Dosser Johnson	O'Brien Resources, LLC	2015-00000848 Texas	Henderson
42.213.16200.0109	8/8/2014	Robert Earl Harris	O'Brien Resources, LLC	2015-00000837 Texas	Henderson
42.213.16200.0112	8/4/2014	Diane Woods	O'Brien Resources, LLC	2015-00000840 Texas	Henderson
42.213.16200.0113	6/23/2014	Phyllis Warren Onley	O'Brien Resources, LLC	2015-00000841 Texas	Henderson
42.213.16200.0115	8/4/2014	Damon C. Woods III	O'Brien Resources, LLC	2015-00000835 Texas	Henderson
42.213.16200.0127	8/15/2014	Tim Ezell	O'Brien Resources, LLC	2015-00000833 Texas	Henderson
42.213.16200.0127	8/15/2014	Tim Ezell	O'Brien Resources, LLC	2015-00000833 Texas	Henderson
42.213.16200.0129	9/9/2014	Jennie Nachlas by Martin Nachlas AIF	O'Brien Resources, LLC	2015-00000830 Texas	Henderson
42.213.16200.0138	9/15/2014	Walter Cunningham Windsor	O'Brien Resources, LLC	2015-00000825 Texas	Henderson
42.213.16200.0142	10/9/2014	David Ferrell Davis	O'Brien Resources, LLC	2015-00000807 Texas	Henderson
42.213.16200.0143	10/9/2014	Walter C. Windsor	O'Brien Resources, LLC	2015-00000804 Texas	Henderson
42.213.16200.0144	10/9/2014	Rentier, Inc.	O'Brien Resources, LLC	2015-00000806 Texas	Henderson
42.213.16200.0146	10/23/2014	Gary Wayne Dennis	O'Brien Resources, LLC	2015-00000801 Texas	Henderson
42.213.16200.0148	10/23/2014	Wesley Don Dennis	O'Brien Resources, LLC	2016-00012972 Texas	Henderson
42.213.16200.0151	9/23/2014	Susan Dennis Maddock	O'Brien Resources, LLC	2016-00012973 Texas	Henderson
42.213.16200.0153	9/25/2014	Madelyn Long Dockery (Donated interest to Holy Angels of Shreveport, LA)	O'Brien Resources, LLC	2015-00000796 Texas	Henderson
42.213.16200.0159	10/30/2014	Kenneth T. Wright	O'Brien Resources, LLC	2015-00000789 Texas	Henderson
42.213.16200.0161	10/13/2014	Gary Lynn Bauhof	O'Brien Resources, LLC	2015-00000786 Texas	Henderson
42.213.16200.0164	11/16/2014	John Lingo Schoellkopf, Alan C. Schoellkopf, Wilson Schoellkopf, Jr.	O'Brien Resources, LLC	2015-00010670 Texas	Henderson

Lease #	Date	Lessor	Lessee	Book-Page State	County
42.213.16200.0169	7/21/2014	Windsor Interests, Ltd.	O'Brien Resources, LLC	2014-00014699 Texas	Henderson
42.213.16200.0175	4/25/2017	Brenda Lott	O'Brien Resources, LLC	2017-00010906 Texas	Henderson
42.213.16200.0176	6/2/2017	Billy T. Elliott	O'Brien Resources, LLC	2017-00017561 Texas	Henderson
42.213.16200.0177	8/2/2017	Lori M. Trim	O'Brien Resources, LLC	2017-00017562 Texas	Henderson
42.213.16200.0178	8/14/2017	Nicole R. DeBusk	O'Brien Resources, LLC	2017-00017558 Texas	Henderson
42.213.16200.0179	8/14/2017	Michael G. LaRue et ux	O'Brien Resources, LLC	2017-00017557 Texas	Henderson
42.213.15550.0001	8/15/2016	J. W. Wylie, Jr. and Joe Ann Wylie	O'Brien Resources, LLC	2016-00016256 Texas	Henderson
42.213.15550.0002	8/23/2016	Shirley Ann Stringfield Hooks	O'Brien Resources, LLC	2016-00016255 Texas	Henderson
42.213.15550.0003	8/25/2016	Stephanie Rusk Jefcoat	O'Brien Resources, LLC	2016-00016254 Texas	Henderson
42.213.15550.0004	10/24/2016	Joseph Percy Coats	O'Brien Resources, LLC	2017-00001486 Texas	Henderson
42.213.15550.0005		Evelyn Mathis Ferguson	O'Brien Resources, LLC	2017-00004139 Texas	Henderson
42.213.15550.0006		Patsy Johnson aka Patsy Ruth Johnson	O'Brien Resources, LLC	2017-00004142 Texas	Henderson
42.213.15550.0007		Mack Parmer Stringfield	O' Brien Resources, LLC	2017-00004138 Texas	Henderson
42.213.15550.0007		Mack Parmer Stringfield	O' Brien Resources, LLC	2017-00004138 Texas	Henderson
42.213.15550.0008		James Patrick Ferguson, Jr.	O' Brien Resources, LLC	2017-00004140 Texas	Henderson
42.213.15550.0009		Ment Bentley Turbiville	O' Brien Resources, LLC	2017-00004141 Texas	Henderson
42.213.15550.0010		The Parrott Family Trust	O'Brien Resources, LLC O'Brien Resources, LLC	2017-00004137 Texas	Henderson
42.213.15550.0011 42.213.15550.0012		Jadin Dublin Hollyne Clair Dublin	O'Brien Resources, LLC	2017-00006630 Texas 2017-00006629 Texas	Henderson Henderson
42.213.15550.0012			O'Brien Resources, LLC	2017-00006635 Texas	Henderson
42.213.15550.0013		Raschelle Browning Loudenslager Sarah Jones Roye and Dave Roye	O'Brien Resources, LLC	2017-00006636 Texas	Henderson
42.213.15550.0014		Nancy Jones Huber and Gregory E. Huber	O'Brien Resources, LLC	2017-00006634 Texas	Henderson
42.213.15550.0016		Shirley Ruth Farish	O'Brien Resources, LLC	2017-00006633 Texas	Henderson
42.213.15550.0017		John C. Hallum	O'Brien Resources, LLC	2017-00006632 Texas	Henderson
42.213.15550.0018	3/8/2017	Jimmie C. Johnson	O'Brien Resources, LLC	2017-00006631 Texas	Henderson
42.213.15550.0019		Adrian Jones Cheek and Neal Cheek	O'Brien Resources, LLC	2017-00006626 Texas	Henderson
42.213.15550.0020		Est of Margaret "Emmy" Jones Deceased by Anita Beth Carlberg	O'Brien Resources, LLC	2017-00006627 Texas	Henderson
42.213.15550.0021		Eddie C. Lehr	O'Brien Resources, LLC	2017-00006628 Texas	Henderson
42.213.15550.0022		Martha Kate Jones	O'Brien Resources, LLC	2017-00006625 Texas	Henderson
42.213.15550.0023	4/7/2017	The Patricia Louise Clements Trust	O'Brien Resources, LLC	2017-00007229 Texas	Henderson
42.213.15550.0024	6/19/2017	Jon William Pugh	O'Brien Resources, LLC	2017-00010912 Texas	Henderson
42.213.15550.0025	6/21/2017	Elaine Wagoner Asmussen	O'Brien Resources, LLC	2017-00010914 Texas	Henderson
42.213.15550.0026	6/19/2017	Jerry Lynn Pugh	O'Brien Resources, LLC	2017-00010911 Texas	Henderson
42.213.15550.0027	6/16/2017	Marjorie Owens	O'Brien Resources, LLC	2017-0001908 Texas	Henderson
42.213.15550.0028	6/21/2017	Peggy Wagoner Hall	O'Brien Resources, LLC	2017-00010915 Texas	Henderson
42.213.15550.0029	6/16/2017	Stella Floy Pace	O'Brien Resources, LLC	2017-00010910 Texas	Henderson
42.213.15550.0030	6/21/2017	William Loyd Pace	O'Brien Resources, LLC	2017-00010909 Texas	Henderson
42.213.15550.0031		Joy Pugh Burnett	O'Brien Resources, LLC	2017-00010913 Texas	Henderson
42.213.15550.0032		Michael Rife	O'Brien Resources, LLC	2017-00010916 Texas	Henderson
42.213.15550.0033		Robin Rife Engle	O'Brien Resources, LLC	2017-00010917 Texas	Henderson
42.349.16000.0001		Fredereka Faye Johnson	O'Brien Resources, LLC	1331 Texas	Navarro
42.349.16000.0007		Billy Doyle Murphree	O'Brien Resources, LLC	1324 Texas	Navarro
42.349.16000.0009		Bonnie Jeanne Hocutt	O'Brien Resources, LLC	1325 Texas	Navarro
42.349.16000.0011		Wayne Gattin	O'Brien Resources, LLC	1328 Texas	Navarro
42.349.16000.0012		Amy Gatlin Heard Caviness	O'Brien Resources, LLC	1327 Texas	Navarro
42.349.16000.0013		John G. Jackson et ux	O'Brien Resources, LLC	1326 Texas	Navarro
42.349.16000.0014 42.349.16000.0016		Robert L. Boyd et ux Donald Jock et ux	O'Brien Resources, LLC O'Brien Resources, LLC	1329 Texas 1315 Texas	Navarro
42.349.16000.0016		Geneva Penney	O'Brien Resources, LLC	1401 Texas	Navarro Navarro
42.349.16000.0022		Linda Boyd Sparkman	O'Brien Resources, LLC	1350 Texas	Navarro
42.349.16000.0039		Carol Boyd Stephenson	O'Brien Resources, LLC	1349 Texas	Navarro
42.349.16000.0040		Floyd Dell Nutt et ux	O'Brien Resources, LLC	1352 Texas	Navarro
42.349.16000.0043		Marsylene Jenkins	O'Brien Resources, LLC	1351 Texas	Navarro
42.349.16000.0044		Mary Sue Brown	O'Brien Resources, LLC	1347 Texas	Navarro
42.349.16000.0047		Wallis M. Boyd et ux	O'Brien Resources, LLC	1343 Texas	Navarro
42.349.16000.0048		Austin Walker Boyd	O'Brien Resources, LLC	1342 Texas	Navarro
42.349.16000.0050	1/5/2012	James Jarrell Boyd	O'Brien Resources, LLC	1289 Texas	Navarro
42.349.16000.0052		Betty Boyd Storms	O'Brien Resources, LLC	1339 Texas	Navarro
42.349.16000.0054		Ellen Nutt, Life Tennant, & Kristy Nutt Ind & Remainderman	O'Brien Resources, LLC	1338 Texas	Navarro
42.349.16000.0055		Linda Nutt Lee	O'Brien Resources, LLC	1337 Texas	Navarro
42.349.16000.0056		Ruby Lee Radney	O'Brien Resources, LLC	1336 Texas	Navarro
42.349.16000.0065	1/3/2012		O'Brien Resources, LLC	1298 Texas	Navarro
42.349.16000.0080	1/10/2012	Ivie Ray Nutt et ux	O'Brien Resources, LLC	2275 Texas	Navarro
42.349.16000.0094		Brice R. Bonner IV	O'Brien Resources, LLC	1322 Texas	Navarro
42.349.16000.0095	2/15/2012	Kristen Bonner Jennings	O'Brien Resources, LLC	1323 Texas	Navarro

Lease #	Date	Lessor	Lessee	Book-Page State	County
42.349.16000.0188	1/31/2011	ANN LIVELY GATLIN	Team Invest Inc.	2614 Texas	Navarro
42.349.16000.0268	2/16/2012	D'Ann Gatlin Carrington	O'Brien Resources, LLC	1266 Texas	Navarro
42.349.16000.0279	3/5/2012	Suzanne Slatkin	O'Brien Resources, LLC	2320 Texas	Navarro
42.349.16000.0286	3/5/2012	Larry R. Norman et ux Revocable Living Trust	O'Brien Resources, LLC	3494 Texas	Navarro
42.349.16000.0305	9/22/2011	Percy Travis Fullwood, II	O'Brien Resources, LLC	7874 Texas	Navarro
42.349.16000.0307	3/5/2012	Beth Norman Hyatt	O'Brien Resources, LLC	2322 Texas	Navarro
42.349.16000.0352	3/13/2012	John T. May et ux	O'Brien Resources, LLC	2337 Texas	Navarro
42.349.16000.0365	5/8/2012	William R. Dorsett	O'Brien Resources, LLC	5373 Texas	Navarro
42.349.16000.0370	3/30/2012	Doris Jean Eaton	O'Brien Resources, LLC	3905 Texas	Navarro
42.349.16000.0371	5/8/2012	Rosemary Moore	O'Brien Resources, LLC	5371 Texas	Navarro
42.349.16000.0439	6/7/2012	Nicky Dale Herndon	O'Brien Resources, LLC	5375 Texas	Navarro
42.349.16000.0451	6/7/2012	James A. Wood	O'Brien Resources, LLC	6756 Texas	Navarro
42.349.16000.0452	6/5/2012	Lori A. Dohanich	O'Brien Resources, LLC	6757 Texas	Navarro
42.349.16000.0478	7/10/2012	Susan Wood	O'Brien Resources, LLC	6782 Texas	Navarro
42.349.16000.0531	8/14/2012	Paul Wood	O'Brien Resources, LLC	8261 Texas	Navarro
42.349.16000.0562	8/20/2012	Jean Elizabeth Wilson	O'Brien Resources, LLC	8272 Texas	Navarro
42.349.16000.0571	7/29/2012	Chad Bingham and wife, Amy	O'Brien Resources, LLC	10147 Texas	Navarro
42.349.16000.0577	8/20/2012	William T. Patterson, Jean Wilson Trustee	O'Brien Resources, LLC	8271 Texas	Navarro
42.001.14520.0018	10/21/2006	William M. Williamson, Jr.	Burk Royalty Co., LTD.	2012-239 Texas	Anderson
42.001.14520.0136	7/15/2008	Julia Elizabeth Borch Bohl	O'Brien Energy Company	2116-502 Texas	Anderson
42.213.26000.0011	1/18/2017	J. Rodney Johnson and Raynell Johnson	O'Brien Resources, LLC	2017-00011737 Texas	Henderson
42.213.26000.0012	1/18/2017	Judy Crain	O'Brien Resources, LLC	2017-00011748 Texas	Henderson
42.213.26000.0013	1/18/2017	Leslie Simon, Jr.	O'Brien Resources, LLC	2017-00011749 Texas	Henderson
42.213.26000.0014	2/1/2017	Peggy J. Browning	O'Brien Resources, LLC	2017-00011750 Texas	Henderson
42.213.26000.0015	1/19/2017	Debra Allen Alford	O'Brien Resources, LLC	2017-00011736 Texas	Henderson
42.213.26000.0016	1/18/2017	Heleen Mange	O'Brien Resources, LLC	2017-00011746 Texas	Henderson
42.213.26000.0017	1/18/2017	Carol Ginsberg	O'Brien Resources, LLC	2017-00011747 Texas	Henderson
42.213.26000.0018	2/1/2017	Suzanne Lynn	O'Brien Resources, LLC	2017-00011741 Texas	Henderson
42.213.26000.0019	2/1/2017	Patrick McMenamy	O'Brien Resources, LLC	2017-00011742 Texas	Henderson
42.213.26000.0020	1/30/2017	Patricia Counts	O'Brien Resources, LLC	2017-00011743 Texas	Henderson
42.213.26000.0021	2/1/2017	Melanie Buchi	O'Brien Resources, LLC	2017-00011744 Texas	Henderson
42.213.26000.0022	2/10/2017	Kyle McMenamy	O'Brien Resources, LLC	2017-00011745 Texas	Henderson
42.213.26000.0023	2/1/2017	Lindsey Seale	O'Brien Resources, LLC	2017-00011739 Texas	Henderson
42.213.26000.0024	1/18/2017	Roberta Bates	O'Brien Resources, LLC	2017-00011738 Texas	Henderson
42.213.26000.0025	2/1/2017	William Marshall Moore III	O'Brien Resources, LLC	2017-00011740 Texas	Henderson
42.213.26000.0026	1/15/2017	Stella Floy Pace	O'Brien Resources, LLC	2017-00011706 Texas	Henderson
42.213.26000.0027	1/15/2017	Jerry Lynn Pugh	O'Brien Resources, LLC	2017-00011705 Texas	Henderson
42.213.26000.0028		Jon William Pugh	O'Brien Resources, LLC	2017-00011704 Texas	Henderson
42.213.26000.0031	2/1/2017	Hilda N. Carrigan	O'Brien Resources, LLC	2017-00013118 Texas	Henderson
42.213.26000.0032	2/1/2017	James W. Carrigan	O'Brien Resources, LLC	2017-00011735 Texas	Henderson
42.213.26000.0033	2/15/2017	WILLIAM LLOYD PACE	O'Brien Resources, LLC	2017-00011710 TEXAS	HENDERSO
42.213.26000.0034	2/15/2017	ROBIN CHRISTINE RIFE ENGLE	O'Brien Resources, LLC	2017-00011712 TEXAS	HENDERSO
42.213.26000.0035	1/15/2017	PEGGY HALL	O'Brien Resources, LLC	2017-00011709 TEXAS	HENDERSO
42.213.26000.0036		MICHAEL DAVID RIFE	O'Brien Resources, LLC	2017-00011711 TEXAS	HENDERSO
42.213.26000.0037		MARJORIE OWENS	O'Brien Resources, LLC	2017-00011708 TEXAS	HENDERSO
42.213.26000.0038		ELAINE ASMUSSEN	O'Brien Resources, LLC	2017-00011707 TEXAS	HENDERSO
42.213.26000.0039		Joan Catherine Kahl Fischer	O'Brien Resources, LLC	2017-00011727 Texas	Henderson
42.213.26000.0040		Kathleen Reese	O'Brien Resources, LLC	2017-00011724 Texas	Henderson
42.213.26000.0041		Mary Ann Groutt	O'Brien Resources, LLC	2017-00011726 Texas	Henderson
42.213.26000.0042		Nancy Hoyem	O'Brien Resources, LLC	2017-00011721 Texas	Henderson
42.213.26000.0043	2/1/2017		O'Brien Resources, LLC	2017-00011723 Texas	Henderson
42.213.26000.0044	2/1/2017		O'Brien Resources, LLC	2017-00011734 Texas	Henderson
42.213.26000.0045	2/16/2017		O'Brien Resources, LLC	2017-00011725 Texas	Henderson
42.213.26000.0046		JOY PUGH BURNETT	O'Brien Resources, LLC	2017-00011713 TEXAS	HENDERSO
42.213.26000.0048		Trisha Lanee Stockard	O'Brien Resources, LLC	2017-00011732 Texas	Henderson
42.213.26000.0049	2/2/2017	Tammy K. Allen	O'Brien Resources, LLC	2017-00011733 Texas	Henderson
42.213.26000.0050	2/15/2017		O'Brien Resources, LLC	2017-00011716 TEXAS	HENDERSO
42.213.26000.0051		RONALD THOMAS	O'Brien Resources, LLC	2017-00011714 TEXAS	HENDERSO
42.213.26000.0052		FRANK LEE THOMAS	O'Brien Resources, LLC	2017-00011715 TEXAS	HENDERSO
42.213.26000.0053		JOY L. TRAMMELL	O'Brien Resources, LLC	2017-00011717 TEXAS	HENDERSO
42.213.26000.0054		PAUL W. CROWSON	O'Brien Resources, LLC	2017-00011717 TEXAS	HENDERSO
42.213.26000.0054		PEGGY J. BROWNING	O'Brien Resources, LLC	2017-00011718 TEXAS	HENDERSO
42.213.26000.0056	4/5/2017	Clayton H. Riddlesperger	O'Brien Resources, LLC	2017-00011719 TEXAS	Henderson
42.213.26000.0057		Rosalyn R. Lane	O'Brien Resources, LLC	2017-00011730 Texas	Henderson
42.213.26000.0057		William Crutcher	O'Brien Resources, LLC	2017-00011731 Texas	Henderson
72.210.20000.0000	7/3/2017		S Brion (Cabulota, LLO	LUTI-UUTTIZƏ TEXAS	i iciluci sUII

Lease #	Date	Lessor	Lessee	Book-Page State	County
42.213.26000.0059	2/21/2017		O'Brien Resources, LLC	2017-00011722 Texas	Henderson
42.213.26000.0059	2/21/2017	Amie Eloise Selecman	O'Brien Resources, LLC	2017-00011722 Texas	Henderson
42.213.26000.0060	4/5/2017	Paul Crutcher	O'Brien Resources, LLC	2017-00011728 Texas	Henderson
42.213.26000.0061	2/2/2017	Barbara Etoch	O'Brien Resources, LLC	2017-00011701 Texas	Henderson
42.213.26000.0062	2/16/2017	Nancy Louise Kahl Sed (New Mineral Ownership	O'Brien Resources, LLC	2017-00011720 Texas	Henderson
42.213.26000.0063	7/21/2017	Alders Living Trust, Marital Trust and Family Trust dated July 30, 1990	O'Brien Resources, LLC	2017-00011703 Texas	Henderson
42.213.26000.0064	7/21/2017	Richard Warren Alders and Johnyce Alders	O'Brien Resources, LLC	2017-00011702 Texas	Henderson
42.213.26000.0065	6/19/2017	Lester Kinabrew Jr. Enterprises Inc.	O'Brien Resources, LLC	2017-00011700 Texas	Henderson
		Louise Boykin (O'Benco IV et al)	O'Brien Resources, LLC	Texas	
15400		Louise Boykin (O'Benco IV et al)	O'Brien Resources, LLC	Texas	
42.213.15400.0001	4/26/2016	Billy Joe Miller	O'Brien Resources, LLC	2515-718 Texas	Anderson
42.213.15400.0003	3/21/2011	John C. Cain, dealing herein with his sole and separate property	O'Brien Resources, LLC	2270-0055 Texas	Anderson
42.213.15400.0004	3/23/2011	Jerald Lambright and wife Ruth Lambright	O'Brien Resources, LLC	2270-036 Texas	Anderson
42.213.15400.0005	4/14/2016	Doris Marie Hillhouse et vir Roy Hillhouse	O'Brien Resources, LLC	2016-00006224 Texas	Henderson
42.213.15400.0006	4/6/2016	Ira Jene Jackson	O'Brien Resources, LLC	2011-00013984 Texas	Henderson
42.213.15400.0007	3/28/2011	Betty D. Watkins, dealing with her sole and separate property	O'Brien Resources, LLC	2011-00013985 Texas	Henderson
42.213.15400.0009	3/29/2011	Nolan A. Phillips	O'Brien Resources, LLC	2270-060 Texas	Anderson
42.213.15400.0012	4/7/2011	Bettye F. Mackie	O'Brien Resources, LLC	2270-0039 Texas	Anderson
42.213.15400.0013	4/12/2011	John M. Johnston and wife Marie R. Johnston	O'Brien Resources, LLC	2011-00013988 Texas	Henderson
42.213.15400.0014	4/20/2011	Cary S. Cowart et ux	O'Brien Resources, LLC	2270-0043 Texas	Anderson
42.213.15400.0015	4/15/2011	Charmaine Denius McGill	O'Brien Resources, LLC	2270-0064 Texas	Anderson
42.213.15400.0016	4/13/2011	Susan A. Soulsby	O'Brien Resources, LLC	2270-081 Texas	Anderson
42.213.15400.0017	4/13/2011	David R. Soulsby	O'Brien Resources, LLC	2270-088 Texas	Anderson
42.213.15400.0019	5/13/2011	Thomas Harold Grantham, Trustee of the Children of Sarah Helen Smith Grantham Trust	O'Brien Resources, LLC	2011-00013990 Texas	Henderson
42.213.15400.0020	5/12/2011	J. Rodney Johnson	O'Brien Resources, LLC	2011-00013991 Texas	Henderson
42.213.15400.0021	5/13/2011	Dane H. Smith AIF for Heather Royall Smith	O'Brien Resources, LLC	2011-00013992 Texas	Henderson
42.213.15400.0022	5/14/2011	Jared D. Jones	O'Brien Resources, LLC	2011-00013993 Texas	Henderson
42.213.15400.0023	5/4/2011	S. Scott MacDonald	O'Brien Resources, LLC	2011-00013994 Texas	Henderson
42.213.15400.0024	5/4/2011	Cary P. MacDonald	O'Brien Resources, LLC	2011-00013995 Texas	Henderson
42.213.15400.0025	5/27/2011	Freda Gail Stern Estate of Freda Gail Stern, Harley Soltes Ind Exec	O'Brien Resources, LLC	2011-00013996 Texas	Henderson
42.213.15400.0026	6/13/2011	Dewey Pierce	O'Brien Resources, LLC	2270-069 Texas	Anderson
42.213.15400.0028	7/1/2011	Buford R. Hayes et ux	O'Brien Resources, LLC	2270-073 Texas	Anderson
42.213.15400.0029	10/14/2016	Weldon J. Taylor (Petra Ziegerman)	O'Brien Resources, LLC	2017-00001404 Texas	Henderson
42.213.15400.0030	7/7/2011	Don N. MacDonald	O'Brien Resources, LLC	2011-00013998 Texas	Henderson
42.213.15400.0031	7/5/2011	Robert A. Haslam	O'Brien Resources, LLC	2011-00013999 Texas	Henderson
42.213.15400.0033	7/1/2011	Don E. Gordon et ux	O'Brien Resources, LLC	2270-077 Texas	Anderson
42.213.15400.0034	4/15/2011	Frank Wofford Denius	O'Brien Resources, LLC	2291-755 Texas	Anderson
42.213.15400.0035	5/31/2013	Doris Marie Hillhouse et vir Roy Hillhouse	O'Brien Resources, LLC	2013-00009773 Texas	Henderson
42.213.15400.0037	7/6/2012	Mike Angel	O'Brien Resources, LLC	2012-00012207 Texas	Henderson
42.213.15400.0038	7/6/2012	Christopher Angel	O'Brien Resources, LLC	2012-00012208 Texas	Henderson
42.213.15400.0039	7/17/2012	Harold Gregg Larimer, Jr.	O'Brien Resources, LLC	2012-00012206 Texas	Henderson
42.213.15400.0040	8/2/2012	Carol Hall Wood, Indiv and as Ind Executrix and Trustee u/w/o Billy Gene Wood, Deceased	O'Brien Resources, LLC	2314-0079 Texas	Anderson
42.213.15400.0041	7/26/2012	Cynthia D. Kiser	O'Brien Resources, LLC	2012-00012226 Texas	Henderson
42.213.15400.0042	8/1/2012	William P. Jones	O'Brien Resources, LLC	2012-00012219 Texas	Henderson
42.213.15400.0043	7/26/2012	Lloyd E. Giles	O'Brien Resources, LLC	2012-00012225 Texas	Henderson
42.213.15400.0044	7/26/2012	Evelyn Giles Sims	O'Brien Resources, LLC	2012-00012209 Texas	Henderson
42.213.15400.0045	7/25/2012	Debbie Lambright	O'Brien Resources, LLC	2318-549 Texas	Anderson
42.213.15400.0046	7/25/2012	Leslie Lambright	O'Brien Resources, LLC	2324-473 Texas	Anderson
42.213.15400.0047	8/17/2012	Larry Price	O'Brien Resources, LLC	2012-00012217 Texas	Henderson
42.213.15400.0048	8/17/2012	James Harvill, Jr.	O'Brien Resources, LLC	2012-00012211 Texas	Henderson
	4/4/0040	Lenell Lambright	O'Brien Resources, LLC	2016-00003607 Texas	Henderson
42.213.15400.0049	1/4/2016	Leneil Lambright	O Brioti (Goodi Goo, EEG		
42.213.15400.0049 42.213.15400.0050		L. B. Eaton	O'Brien Resources, LLC	2012-00012218 Texas	Henderson
	8/3/2012	·		2012-00012218 Texas 2012-00012212 Texas	Henderson Henderson
42.213.15400.0050	8/3/2012 8/17/2012	L. B. Eaton	O'Brien Resources, LLC		
42.213.15400.0050 42.213.15400.0051	8/3/2012 8/17/2012 8/14/2012	L. B. Eaton Jean L. Harvill	O'Brien Resources, LLC O'Brien Resources, LLC	2012-00012212 Texas	Henderson
42.213.15400.0050 42.213.15400.0051 42.213.15400.0052	8/3/2012 8/17/2012 8/14/2012 5/27/2011	L. B. Eaton Jean L. Harvill Diane Dickson	O'Brien Resources, LLC O'Brien Resources, LLC O'Brien Resources, LLC	2012-00012212 Texas 2314-0063 Texas	Henderson Anderson
42.213.15400.0050 42.213.15400.0051 42.213.15400.0052 42.213.15400.0053	8/3/2012 8/17/2012 8/14/2012 5/27/2011 5/19/2011	L. B. Eaton Jean L. Harvill Diane Dickson Aletta LeeVee Lambright et al	O'Brien Resources, LLC O'Brien Resources, LLC O'Brien Resources, LLC Energia Tejas, LLC	2012-00012212 Texas 2314-0063 Texas 2011-00012607 Texas	Henderson Anderson Henderson
42.213.15400.0050 42.213.15400.0051 42.213.15400.0052 42.213.15400.0053 42.213.15400.0054	8/3/2012 8/17/2012 8/14/2012 5/27/2011 5/19/2011 11/9/2011	L. B. Eaton Jean L. Harvill Diane Dickson Aletta LeeVee Lambright et al Aletta LeeVee Lambright et al	O'Brien Resources, LLC O'Brien Resources, LLC O'Brien Resources, LLC Energia Tejas, LLC Energia Tejas, LLC	2012-00012212 Texas 2314-0063 Texas 2011-00012607 Texas 2011-00010839 Texas	Henderson Anderson Henderson
42.213.15400.0050 42.213.15400.0051 42.213.15400.0052 42.213.15400.0053 42.213.15400.0054 42.213.15400.0055	8/3/2012 8/17/2012 8/14/2012 5/27/2011 5/19/2011 11/9/2011 5/26/2011	L. B. Eaton Jean L. Harvill Diane Dickson Aletta LeeVee Lambright et al Aletta LeeVee Lambright et al Anthony Dale Dehart	O'Brien Resources, LLC O'Brien Resources, LLC O'Brien Resources, LLC Energia Tejas, LLC Energia Tejas, LLC Energia Tejas, LLC	2012-00012212 Texas 2314-0063 Texas 2011-00012607 Texas 2011-00010839 Texas 2012-00007061 Texas	Henderson Anderson Henderson Henderson
42.213.15400.0050 42.213.15400.0051 42.213.15400.0052 42.213.15400.0053 42.213.15400.0054 42.213.15400.0055 42.213.15400.0056	8/3/2012 8/17/2012 8/14/2012 5/27/2011 5/19/2011 11/9/2011 5/26/2011 3/30/2011	L. B. Eaton Jean L. Harvill Diane Dickson Aletta LeeVee Lambright et al Aletta LeeVee Lambright et al Anthony Dale Dehart Anton E. Fraps, III	O'Brien Resources, LLC O'Brien Resources, LLC O'Brien Resources, LLC Energia Tejas, LLC Energia Tejas, LLC Energia Tejas, LLC Energia Tejas, LLC	2012-00012212 Texas 2314-0063 Texas 2011-00012607 Texas 2011-00010839 Texas 2012-00007061 Texas 2011-00010888 Texas	Henderson Anderson Henderson Henderson Henderson
42.213.15400.0050 42.213.15400.0051 42.213.15400.0052 42.213.15400.0053 42.213.15400.0054 42.213.15400.0056 42.213.15400.0056 42.213.15400.0057	8/3/2012 8/17/2012 8/14/2012 5/27/2011 5/19/2011 11/9/2011 5/26/2011 3/30/2011	L. B. Eaton Jean L. Harvill Diane Dickson Aletta LeeVee Lambright et al Aletta LeeVee Lambright et al Anthony Dale Dehart Anton E. Fraps, III Archie A. Dennis Jr. et ux June M. Dennis	O'Brien Resources, LLC O'Brien Resources, LLC O'Brien Resources, LLC Energia Tejas, LLC	2012-00012212 Texas 2314-0063 Texas 2011-00012607 Texas 2011-00010839 Texas 2012-00007061 Texas 2011-00010888 Texas 2011-00010827 Texas	Henderson Henderson Henderson Henderson Henderson Henderson Henderson
42.213.15400.0050 42.213.15400.0051 42.213.15400.0052 42.213.15400.0053 42.213.15400.0054 42.213.15400.0056 42.213.15400.0056 42.213.15400.0057 42.213.15400.0058	8/3/2012 8/17/2012 8/14/2012 5/27/2011 5/19/2011 11/9/2011 5/26/2011 3/30/2011	L. B. Eaton Jean L. Harvill Diane Dickson Aletta LeeVee Lambright et al Aletta LeeVee Lambright et al Anthony Dale Dehart Anton E. Fraps, Ill Archie A. Dennis Jr. et ux June M. Dennis Archie M. Dennis Betty Darsey Woods	O'Brien Resources, LLC O'Brien Resources, LLC O'Brien Resources, LLC Energia Tejas, LLC	2012-00012212 Texas 2314-0063 Texas 2011-00012607 Texas 2011-00010839 Texas 2012-00007061 Texas 2011-00010888 Texas 2011-00010827 Texas 2012-00010611 Texas	Henderson Anderson Henderson Henderson Henderson Henderson Henderson
42.213.15400.0050 42.213.15400.0051 42.213.15400.0052 42.213.15400.0054 42.213.15400.0056 42.213.15400.0056 42.213.15400.0057 42.213.15400.0058 42.213.15400.0058	8/3/2012 8/17/2012 8/14/2012 5/27/2011 5/19/2011 11/9/2011 5/26/2011 3/30/2011 8/16/2011	L. B. Eaton Jean L. Harvill Diane Dickson Aletta LeeVee Lambright et al Aletta LeeVee Lambright et al Anthony Dale Dehart Anton E. Fraps, Ill Archie A. Dennis Jr. et ux June M. Dennis Archie M. Dennis Betty Darsey Woods	O'Brien Resources, LLC O'Brien Resources, LLC O'Brien Resources, LLC Energia Tejas, LLC	2012-00012212 Texas 2314-0063 Texas 2011-00012607 Texas 2011-00010839 Texas 2012-00007061 Texas 2011-00010888 Texas 2011-00010827 Texas 2012-00010611 Texas 2011-00012602 Texas	Henderson Anderson Henderson Henderson Henderson Henderson Henderson Henderson
42.213.15400.0050 42.213.15400.0051 42.213.15400.0052 42.213.15400.0053 42.213.15400.0054 42.213.15400.0055 42.213.15400.0056 42.213.15400.0057 42.213.15400.0058 42.213.15400.0058 42.213.15400.0059 42.213.15400.0059	8/3/2012 8/17/2012 8/14/2012 5/27/2011 5/19/2011 11/9/2011 5/26/2011 3/30/2011 8/16/2011 3/30/2011	L. B. Eaton Jean L. Harvill Diane Dickson Aletta LeeVee Lambright et al Aletta LeeVee Lambright et al Anthony Dale Dehart Anton E. Fraps, III Archie A. Dennis Jr. et ux June M. Dennis Archie M. Dennis Betty Darsey Woods Bettye F. Mackle	O'Brien Resources, LLC O'Brien Resources, LLC O'Brien Resources, LLC Energia Tejas, LLC	2012-00012212 Texas 2314-0063 Texas 2011-00012607 Texas 2011-00010839 Texas 2012-00007061 Texas 2011-00010888 Texas 2011-00010827 Texas 2012-00010611 Texas 2011-00012602 Texas 2011-00012602 Texas 2011-00006509 Texas	Henderson Anderson Henderson Henderson Henderson Henderson Henderson Henderson Henderson

	Lease #	Date	Lessor	Lessee	Book-Page State	County
1931 1932					-	Henderson
CASES PRINCESSOR SQL STREAMSORM Control Control Control	42.213.15400.0065	6/21/2011		Energia Tejas, LLC	2011-00012631 Texas	Henderson
SATE STANDAM SATE STANDAM<	42.213.15400.0067	3/30/2011	Brad Lambright	Energia Tejas, LLC	2011-00006511 Texas	Henderson
42.31.31.60.0007 30.000 control from the process of th	42.213.15400.0070	5/18/2011	Cary S. "Pete" Cowart	Energia Tejas, LLC	2011-00012622 Texas	Henderson
4215100000000000000000000000000000000000	42.213.15400.0071	3/30/2011	Charles P. Lewis	Energia Tejas, LLC	2011-00006572 Texas	Henderson
431131660000 100000 Control Design Train, LLG 2015000000 201500000 201500000 201500000 201500000 201500000 201500000 2015000000 2015000000 2015000000 2015000000 2015000000 2015000000 2015000000 2015000000 2015000000 2015000000 2015000000 2015000000 2015000000 2015000000 2015000000 2015000000 20150000000 2015000000 2015000000 2015000000 20150000000 20150000000 20150000000 20150000000 20150000000 201500000000 201500000000 201500000000000 20150000000000 201500000000000000 201500000000000000000000000000000000000	42.213.15400.0072	7/26/2011	Charles P. Lewis	Energia Tejas, LLC	2011-00012611 Texas	Henderson
43.13.13.00.000 19.0000 months Comparation Log 20.10.0000 months	42.213.15400.0075	6/9/2011	Claudia J. Stromberg (and Russell S. Stromberg, a married couple)	Energia Tejas, LLC	2011-00010909 Texas	Henderson
23.15.00.0000 1907 Lower Institution Company Texas 1908 Texas 2011 Monthly Company 2011 Monthly C	42.213.15400.0076	6/1/2011	Clifford O. Baggett	Energia Tejas, LLC	2011-00010877 Texas	Henderson
SEAT STREAM STATE Investigation Comparison Comparis	42.213.15400.0078	3/30/2011	Dana Mitchell	Energia Tejas, LLC	2011-00006540 Texas	Henderson
4.3.1.3.1.3.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0	42.213.15400.0079	8/1/2011	Dana Mitchell	Energia Tejas, LLC	2011-00012610 Texas	Henderson
43.13 Modelle Min Modelle Mindelle	42.213.15400.0080	6/1/2011	David Baggett	Energia Tejas, LLC	2011-00010847 Texas	Henderson
23.13 Sealons 1 Sealons 1 Sealons 1 Sealons 2 Sealons	42.213.15400.0081	5/19/2016	David Tritter, Trustee	O'Brien Resources, LLC	2016-00012765 Texas	Henderson
23.13.15.00.00000 1.00.00000 1.00.00000 1.00.00000 1.00.00000 1.00.00000 1.00.00000 1.00.00000 1.00.00000 1.00.00000 1.00.000000 1.00.000000 1.00.000000 1.00.000000 1.00.000000 1.00.0000000 1.00.0000000 1.00.0000000 1.00.0000000 1.00.0000000 1.00.0000000 1.00.0000000 1.00.00000000 1.00.00000000 1.00.000000000 1.00.000000000 1.00.000000000000 1.00.000000000000000000000000000000000	42.213.15400.0082	7/9/2016	Deanie Miller	O'Brien Resources, LLC	2016-00012762 Texas	Henderson
433151600000 7 190211 March Lemburg Pengal Frail, Log 1310-000000 Frail Responsible for the American Control of th	42.213.15400.0083	8/24/2016	Debby L. Henderson	O'Brien Resources, LLC	2016-00016262 Texas	Henderson
CATS DASSIGNIONE Salight In Months Policy Company Man (1998) Michael Months Policy	42.213.15400.0084	5/27/2011	Deborah J. Schrader	Energia Tejas, LLC	2011-00010873 Texas	Henderson
SIZISISSOROMO School Poole Modes Priced School Pool School Pool Pool Pool Pool Pool Pool Pool	42.213.15400.0085	7/1/2011	Debra K. Lambright	Energia Tejas, LLC	2011-00010886 Texas	Henderson
2915 SIZUSINSONIOURS Separate Basebash Flackbook In the North Workshort Sontered Penagha Flack ILL 15000000000000000000000000000000000000	42.213.15400.0088	3/31/2011	Keith Lee Pate	Energia Tejas, LLC	2011-00006566 Texas	Henderson
4.3.11.5.6.0000000 2.00000000 Claypoon Controller Design Design Controller Design A Journal Design A J	42.213.15400.0090	5/7/2017	Robert Michael Pickel	O'Brien Resources, LLC	2017-00010902 Texas	Henderson
4.2315.00000000 2000000000000000000000000000000000000	42.213.15400.0092	5/4/2011	Elizabeth Staggs Huckabay, Tr. of the Mary Kathryn Grisso Revocable Trust	Energia Tejas, LLC	2011-00010832 Texas	Henderson
42213 SEROURDOR 1000001 ICPUE TIME LICE 2012 COUNTY TIME 41 (1000) 42213 SEROURDOR 670001 Composition of County of	42.213.15400.0093	2/23/2011		Energia Tejas, LLC	2012-00009535 Texas	Henderson
4273 154000000000000000000000000000000000000	42.213.15400.0095					Henderson
43.91.516.00.00000 60 ppc but Windows 60 ppc	42.213.15400.0096	8/30/2011	FDIC	Energia Tejas, LLC	2012-00007057 Texas	Henderson
42731 5600000 Mg 700011 Mg Vacco Withorses are solding of Service Microsoph Congress and Service Microsoph Congress Annalysis (August and Service Congress Annalysis August 1540) (Service Microsoph Congress Annalysis Annalys	42.213.15400.0097	2/25/2011	Gary Don Wilbanks		2011-00006551 Texas	Henderson
4211 5480011000 6000011 Cented Knoeppt and wife Contry/Reciption 6000000000000000000000000000000000000	42.213.15400.0098	6/7/2011			2011-00010880 Texas	Henderson
2211-156001736 202001 10 could 10 co	42.213.15400.0101	5/27/2011	Gerald C. Knepple and wife, Carolyn Knepple	Energia Tejas, LLC	2011-00010852 Texas	Henderson
42211 560001300 402001 Century Information of the ball Family Recorded Trust Emigra Figs. LLC 2011-0000035 Towal Record 4221 1560001300 4020011 Classify Bert Figs. LLC 2011-0000005 Towal Record 4221 1560001300 4020011 Classify Bert Figs. LLC 2011-0000008 Towal Record 4221 1560001300 4020011 Classify Bert Figs. LLC 2011-0000008 Towal Record 4221 1560001300 4020011 Classify Bert Figs. LLC 2011-0000008 Towal Record 4221 1560001301 402001 Clear Mark Clear Mark Record Record Record Record 4221 1560001310 402001 Health Final Mark Record Record Record Record 4221 156001310 102001 Health Final Mark Record Record Record Record 4221 156001310 10200131 Health Final Mark Record Record Record 4221 156001310 10200131 Health Final Mark Annual Mark Record Record Record Record 4221 156	42.213.15400.0102	5/3/2011		Energia Tejas, LLC	2011-00010854 Texas	Henderson
2211 154000716 2020 10 10 10 10 10 10	42.213.15400.0103	5/3/2011		Energia Tejas, LLC	2011-00010853 Texas	Henderson
2015 1500	42.213.15400.0104	2/23/2011			2011-00006553 Texas	Henderson
Page	42.213.15400.0105	6/7/2011			2011-00010868 Texas	Henderson
4213 15400101	42.213.15400.0106	4/6/2011			2011-00006561 Texas	Henderson
2211.5400.011	42.213.15400.0107	5/27/2011		Energia Tejas, LLC	2011-00010841 Texas	Henderson
2211 1400 1701 1702 1703	42.213.15400.0108	5/5/2011	Glenn Holley	Energia Tejas, LLC	2011-00010865 Texas	Henderson
42213.5460.0111 9882011 Heigh Herscheach, Trustee of the Herschbach Living Trust dated June 20,2002; Rivervood Energy, LLC O'Brien Resources, LLC 214-00102987 reas Herd 42213.5460.0114 11/2521 Herschbach Prolitioner Co., Ltd. 210-100010897 reas Herd 42213.5460.0117 6 0900011 Helder Demonstration of Section 1 Energia Tejas, LLC 211-00012804 Teas Herd 42213.5460.0118 7 826011 Increase Energy, Inc. 211-00012804 Teas Herd 42213.5460.0118 7 826011 Increase Energy, Inc. 211-00012804 Teas Herd 42213.5460.0118 7 826011 Jacobas Herd	42.213.15400.0110	8/27/2011	Gwen Murrell	Energia Tejas, LLC	2011-00140104 Texas	Henderson
2213.1540.0114 17.250.011	42.213.15400.0112	3/23/2011	Harvil Lester Lambright	Energia Tejas, LLC	2011-00010859 Texas	Henderson
	42.213.15400.0113	8/26/2014	Heidy Herschbach, Trustee of the Herschbach Living Trust dated June 20, 2002; RIverwood Energy, LLC	O'Brien Resources, LLC	2014-00015294 Texas	Henderson
4213.1500.0171 828.0211 Inversion Energy, Inc. Control Control Energy Englas, LCQ 2011.00012604 Teams Head 42.213.1500.0178 322011 Jackson Energy Englas, LCQ 2011.00010604 Teams Hend 42.213.1500.0122 7280.011 J.D. Black Energy Englas, LCQ 2011.0001064 Teams Hend 42.213.1500.0124 92.0211 J.D. Black Energy Englas, LCQ 2011.0001261 Team Hend 42.213.1500.0124 92.0211 Jack P. Rahm, Trustee of the Jack P. Rahm Revocable Trust Energy Englas, LCQ 2011.0001200 Team Hend 42.213.1500.0124 92.0211 Jack P. Rahm, Trustee of the Jack P. Rahm Revocable Trust Energy Englas, LCQ 2011.0001200 Team Hend 42.213.1500.0128 92.0211 Jack P. Rahm, Trustee of the Jack P. Rahm Revocable Trust Energy Englas, LCQ 2011.0001200 Team Hend 42.213.1500.0128 92.0211 Jack J. Santa Lym Shafer Enysten Hend	42.213.15400.0114	11/26/2014	Herschbach Petroleum Co., Ltd.	O'Brien Resources, LLC	2015-00000196 Texas	Henderson
42213.15400.0118 7282011 In Jenes Jackson Energia Tejas, LLC 2011-0001984 Tesas Hend 42213.15400.0118 232011 Online Jung to Lambright Central Tejas, LLC 2011-0000884 Tesas Hend 42213.15400.0128 782010 Jung Beaut Dennyia Tejas, LLC 2011-0000884 Tesas Hend 42213.15400.0128 923011 Jack P. Rahm, Trastee of the Jack P. Rahm Revocable Trast Jene Hend 42213.15400.0128 Energia Tejas, LLC 2011-0001280 Tesas Hend 42213.15400.0128 923011 Jack P. Rahm, Trastee of the Jack P. Rahm Revocable Trast Hend 42213.15400.0128 Energia Tejas, LLC 2011-0001280 Tesas Hend 42213.15400.0128 913012 James Robert Garrett James Robert Garrett Hend 42213.15400.0128 Pringia Tejas, LLC 2011-000180 Tesas Hend 42213.15400.0132 913021 Jennis Robert Garrett Jennis Robert Garrett Hend 42213.15400.0128 Pringia Tejas, LLC 2011-000180 Tesas Hend 42213.15400.0133 913021 Jenny Neith Michel Wibanis Hend 42213.15400.013 Hend	42.213.15400.0115	6/30/2011	Holley Lake Club, c/o Ramon Douglas	Energia Tejas, LLC	2011-00012593 Texas	Henderson
42213.15400.0118 7282011 In Jenes Jackson Energia Tejas, LLC 2011-0001984 Tesas Hend 42213.15400.0118 232011 Online Jung to Lambright Central Tejas, LLC 2011-0000884 Tesas Hend 42213.15400.0128 782010 Jung Beaut Dennyia Tejas, LLC 2011-0000884 Tesas Hend 42213.15400.0128 923011 Jack P. Rahm, Trastee of the Jack P. Rahm Revocable Trast Jene Hend 42213.15400.0128 Energia Tejas, LLC 2011-0001280 Tesas Hend 42213.15400.0128 923011 Jack P. Rahm, Trastee of the Jack P. Rahm Revocable Trast Hend 42213.15400.0128 Energia Tejas, LLC 2011-0001280 Tesas Hend 42213.15400.0128 913012 James Robert Garrett James Robert Garrett Hend 42213.15400.0128 Pringia Tejas, LLC 2011-000180 Tesas Hend 42213.15400.0132 913021 Jennis Robert Garrett Jennis Robert Garrett Hend 42213.15400.0128 Pringia Tejas, LLC 2011-000180 Tesas Hend 42213.15400.0133 913021 Jenny Neith Michel Wibanis Hend 42213.15400.013 Hend	42.213.15400.0117	8/26/2011	Inverness Energy, Inc.	Energia Tejas, LLC	2011-00012604 Texas	Henderson
42213.1540.0122 7/28/2011 3.0. Black 2.0. Control 2.0. Black 2.0. Control 2.0. Con	42.213.15400.0118			Energia Tejas, LLC	2011-00010904 Texas	Henderson
42213.15400.0122 782011 30.0 Black Energia Tejas, LLC 2011-0001281 Texas Hend 42213.15400.0124 923011 Jacque D. Hyde individually and as executivi for the Estate of Jack L. Coleman C916 resources, LLC 2015-00002020 Texas Hend 42213.15400.0125 823011 Jack P. Rahm, Trustee of the Jack P. Rahm Revocable Trust Energia Tejas, LLC 2011-00012800 Texas Hend 42213.15400.0126 833010 Jane Lynn Shafer Boyanton Energia Tejas, LLC 2011-0001241 Texas Hend 42213.15400.0130 6130211 Jane Lynn Shafer Boyanton Energia Tejas, LLC 2011-0001041 Texas Hend 42213.15400.0130 6130211 Jentry W. Half Mitchell Wilbanks Hend Penergia Tejas, LLC 2011-0001080 Texas Hend 42213.15400.0132 232011 Jeny Neil Mitchell Wilbanks Hend Penergia Tejas, LLC 2011-0001086 Texas Hend 42213.15400.0132 232011 Jeny Neil Mitchell Wilbanks Hend Penergia Tejas, LLC 2011-0001086 Texas Hend 42213.15400.0132 377021 Jeny Neil Mitchell Wilbanks Jeny Neil Mitchell Wilbanks Hend <td>42.213.15400.0119</td> <td>3/23/2011</td> <td>Shirley Joyce Lambright</td> <td>Energia Tejas, LLC</td> <td>2011-00006548 Texas</td> <td>Henderson</td>	42.213.15400.0119	3/23/2011	Shirley Joyce Lambright	Energia Tejas, LLC	2011-00006548 Texas	Henderson
4.2213.15400.0725 93.22011 Jack P, Rahm, Trustee of the Jack P, Rahm Revocable Trust Energia Tejas, LLC 2011-00012800 Texas Hend 4.2213.15400.0726 93.22011 Jack P, Rahm, Trustee of the Jack P, Rahm Revocable Trust Energia Tejas, LLC 2011-00012800 Texas Hend 4.2213.15400.0726 93.072015 James Robert Garrett 2016-0000289 Texas Hend 4.2213.15400.0732 93.072015 James Robert Garrett 2016-0000289 Texas Hend 4.2213.15400.0732 93.072011 James Robert Garrett 2011-00000818 Texas Hend 4.2213.15400.0732 93.072011 Jerny Mel Mitchell Wilbanks Energia Tejas, LLC 2011-0000685 Texas Hend 4.2213.15400.0732 23.072011 Jerny Nell Mitchell Wilbanks Energia Tejas, LLC 2011-0000685 Texas Hend 4.2213.15400.0732 23.072011 Jerny Nell Mitchell Wilbanks Hend 22.213.15400.0732 Energia Tejas, LLC 2011-0000685 Texas Hend 4.2213.15400.0733 73.072011 Jerny Nell Mitchell Wilbanks Jerny Nell Mitchell Wilbanks Hend 22.213.15400.0732 Energia Tejas, LLC 2011-0000685 Texas	42.213.15400.0122			Energia Tejas, LLC	2011-00012612 Texas	Henderson
42213.1540.0126 8/3/2011 Jack P. Rahm, Trustee of the Jack P. Rahm Revocable Trust Peregia Tejas, LLC 2011-000120 Texas Hend 42213.15400.0126 6/10/2011 James Robert Garrett 2011-00001410 Texas Hend 42213.15400.0130 6/10/2011 Jean J. Palmer 2011-00001410 Texas Hend 42213.15400.0131 5/13/201 Jean J. Palmer 2011-00001410 Texas Hend 42213.15400.0132 5/13/201 Jean J. Palmer 2011-00001490 Texas Hend 42213.15400.0132 5/13/201 Jean J. Palmer 2011-00001895 Texas Hend 42213.15400.0132 2/23/201 Jerny kell Michell Wibanks Energia Tejas, LLC 2011-00006895 Texas Hend 42213.15400.0133 3/7/201 Jerny kell Michell Wibanks Energia Tejas, LLC 2011-00006895 Texas Hend 42213.15400.0130 3/7/201 Jerny kell Michell Wibanks Energia Tejas, LLC 2011-00006895 Texas Hend 42213.15400.0130 3/7/201 Jerny kell Michell Wibanks Energia Tejas, LLC 2011-00006895 Texas Hend 42213.15400.0130 5/2/201	42.213.15400.0124	9/23/2014	Jacque D. Hyde individually and as executrix for the Estate of Jack L. Coleman	O'Brien Resources, LLC	2015-00000202 Texas	Henderson
42.213.15400.0127 3/31/2016 James Robert Garrett 2016-00006269 Texas Hend 42.213.15400.0128 6/10/2011 Janet Lynn Shafer Boyanton Energia Tejas, LLC 2011-00014110 Texas Hend 42.213.15400.0130 5/13/2011 Jeffery W. Ealton Energia Tejas, LLC 2011-00010800 Texas Hend 42.213.15400.0131 23/2011 Jerry Nell Mitchel Wilbanks Energia Tejas, LLC 2011-00010858 Texas Hend 42.213.15400.0133 4/28/2011 Jerry Wayne Rogers et ut Wanda Beth Rogers Hend 42.213.15400.0134 Energia Tejas, LLC 2011-00010885 Texas Hend 42.213.15400.0134 3/7/2011 John L. Brown et ux Wanda Beth Rogers Hend 42.213.15400.0134 Energia Tejas, LLC 2011-00010885 Texas Hend 42.213.15400.0136 6/17/2011 John K. Patton et ux Rose Hend 42.213.15400.0134 Energia Tejas, LLC 2011-00010885 Texas Hend 42.213.15400.0136 6/17/2011 John K. Patton et ux Boxe Hend 42.213.15400.0140 Energia Tejas, LLC 2011-00010885 Texas Hend 42.213.15400.0137 5/24/201 Jo	42.213.15400.0125	8/23/2011	Jack P. Rahm, Trustee of the Jack P. Rahm Revocable Trust	Energia Tejas, LLC	2011-00012600 Texas	Henderson
42213.154000128 6/10/2011 Janet Lynn Shafer Boyanton Energia Tejas, LLC 2011-00014110 Texas Hend 42213.15400.0130 5/13/2011 Jean J. Palmer Energia Tejas, LLC 2011-00010805 Texas Hend 42213.15400.0131 5/11/2011 Jerry Nell Mitchell Wilbanks Energia Tejas, LLC 2011-00006885 Texas Hend 42213.15400.0134 3/72011 Jerry Nell Mitchell Wilbanks Energia Tejas, LLC 2011-00006885 Texas Hend 42213.15400.0134 3/72011 Jerry Nell Mitchell Wilbanks Energia Tejas, LLC 2011-00006885 Texas Hend 42213.15400.0134 3/72011 Jerry Nell Mitchell Wilbanks Energia Tejas, LLC 2011-00006885 Texas Hend 42213.15400.0136 6/17/2011 John H. Brown et ux Judy aka Final Telley, Lld. Energia Tejas, LLC 2011-00006885 Texas Hend 42213.15400.0137 5/25/2011 John L. Schoellkopf Energia Tejas, LLC 2011-00006851 Texas Hend 42213.15400.0138 8/27/2015 Judy Nell Prichard, Tustee for Kriste Leigh Wakefield Energia Tejas, LLC 2011-00006831 Texas Hend 42213.15400.0149	42.213.15400.0126	8/23/2011	Jack P. Rahm, Trustee of the Jack P. Rahm Revocable Trust	Energia Tejas, LLC	2011-00012601 Texas	Henderson
4213.15400.0130 \$1/31/201 Jean J. Palmer 2011-00010900 Texas Hend 42.213.15400.0131 \$1/11/201 Jeffrey W. Eaton Energia Tejas, LLC 2011-00010856 Texas Hend 42.213.15400.0132 \$2/23/201 Jerry Nel Mitchell Wilbanks Energia Tejas, LLC 2011-00010858 Texas Hend 42.213.15400.0133 4/28/201 Jerry Wayer Rogers et ux Wanda Beth Rogers Hend 42.213.15400.0134 3/7/2011 Jerry Wayer Rogers et ux Wanda Beth Rogers Hend 42.213.15400.0137 5/25/201 John H. Brown et ux Judy aks Final Talley, Ltd. Energia Tejas, LLC 2011-00010885 Texas Hend 42.213.15400.0138 5/24/201 John L. Brown et ux Judy aks Final Talley, Ltd. Energia Tejas, LLC 2011-00010885 Texas Hend 42.213.15400.0138 5/24/201 John L. Schoelikopf Energia Tejas, LLC 2011-00010885 Texas Hend 42.213.15400.0138 5/24/201 John L. Schoelikopf Energia Tejas, LLC 2011-00010881 Texas Hend 42.213.15400.0149 3/19/201 July Nell Pritchard, Tustee for Kristle Leigh Wakefield Energia Tejas, LLC 2011-00006843 Texas	42.213.15400.0127	3/31/2016	James Robert Garrett	O'Brien Resources, LLC	2016-00006269 Texas	Henderson
42213.15400.0130 5131201 Jean J. Palmer 211-00010900 Texas Hend 42213.15400.0131 5/11/2011 Jeffrey W. Eaton Energia Tejas, LLC 2011-00010658 Texas Hend 42213.15400.0132 2/23/201 Jerry Nell Mitchell Wilbanks Energia Tejas, LLC 2011-00010658 Texas Hend 42213.15400.0133 4/28/201 Jerry Nell Mitchell Wilbanks Energia Tejas, LLC 2011-00010685 Texas Hend 42213.15400.0134 3/7/2011 Jerry Wayne Rogers et ux Wanda Beth Rogers Hend Hend 42213.15400.0134 Energia Tejas, LLC 2011-00010685 Texas Hend 42213.15400.0137 5/25/2011 John L. Brown et ux Judy aka Final Talley, Ltd. Energia Tejas, LLC 2011-00010685 Texas Hend 42213.15400.0137 5/24/201 John L. Schoelkopf John L. Schoelkopf Energia Tejas, LLC 2011-00010885 Texas Hend 42213.15400.0138 5/24/201 John L. Schoelkopf John L. Schoelkopf Energia Tejas, LLC 2011-00010881 Texas Hend 42213.15400.0149 3/19/201 John L. Schoelkopf John L. Schoelkopf Energia Tejas, LLC	42.213.15400.0128	6/10/2011	Janet Lynn Shafer Boyanton	Energia Tejas, LLC	2011-00014110 Texas	Henderson
42213.15400.0131 5/11/2011 Idfrey W. Eaton Energia Tejas, LLC 2011-00010856 Texas Hend 42.213.15400.0132 2/23/2011 Jerry Nell Mitchell Wilbanks Energia Tejas, LLC 2011-00010858 Texas Hend 42.213.15400.0133 4/28/2011 Jerry Nell Mitchell Wilbanks Energia Tejas, LLC 2011-00010858 Texas Hend 42.213.15400.0134 3/7/201 Jerry Wayne Rogers et ur Wanda Beth Rogers Energia Tejas, LLC 2011-00010851 Texas Hend 42.213.15400.0136 6/17/201 Jerry Hall Mitchell Wilbanks Energia Tejas, LLC 2011-00010851 Texas Hend 42.213.15400.0136 6/17/201 Jerry Hall Mitchell Vilbanks Energia Tejas, LLC 2011-0010851 Texas Hend 42.213.15400.0138 5/25/2011 John K. Patton et ux Rose Bergia Tejas, LLC 2011-0010851 Texas Hend 42.213.15400.0139 8/27/201 John K. Patton et ux Rose Bergia Tejas, LLC 2011-0001684 Texas Hend 42.213.15400.0139 8/27/201 July Vell Pritchard, Trustee for Kristle Leigh Wakefield Energia Tejas, LLC 2011-0001684 Texas Hend 42.213.15400.01	42.213.15400.0130				2011-00010900 Texas	Henderson
4213.15400.0132 2/3/2011 Jerry Nell Mitchell Wilbanks Energia Tejas, LLC 2011-00006536 Texas Hend 42.213.15400.0133 4/28/2011 Jerry Nell Mitchell Wilbanks Energia Tejas, LLC 2011-00016858 Texas Hend 42.213.15400.0134 3/7/2011 Jerry Wayne Rogers et ux Wanda Beth Rogers Hend 22.213.15400.0136 Energia Tejas, LLC 2011-0001685 Texas Hend 42.213.15400.0137 5/25/2011 John K, Patton et ux Rose Energia Tejas, LLC 2011-0001685 Texas Hend 42.213.15400.0138 5/24/2011 John K, Patton et ux Rose Energia Tejas, LLC 2011-0001685 Texas Hend 42.213.15400.0138 5/24/2011 John K, Patton et ux Rose Energia Tejas, LLC 2011-0001685 Texas Hend 42.213.15400.0139 5/24/2011 John K, Patton et ux Rose Pend 2011-0001685 Texas Hend 42.213.15400.0139 5/24/2011 Joyle Cok Energia Tejas, LLC 2011-00006634 Texas Hend 42.213.15400.0139 3/19/2011 Judy Nell'Intichard, Truste for Kristle Lejsh Wakefield Energia Tejas, LLC 2011-00006631 Texas Hend	42.213.15400.0131	5/11/2011	Jeffrey W. Eaton		2011-00010856 Texas	Henderson
42213.15400.0133 4/28/2011 Jerry Nell Mitchell Wilbanks Energia Tejas, LLC 2011-00016858 Texas Hend 42213.15400.0134 37/2011 John L. Brown et ux Judy aka Final Talley, Ltd. Energia Tejas, LLC 2011-0001685 Texas Hend 42213.15400.0136 6/17/2011 John K. Patton et ux Judy aka Final Talley, Ltd. Energia Tejas, LLC 2011-0001685 Texas Hend 42213.15400.0137 5/25/2011 John L. Schoellkopf Energia Tejas, LLC 2011-00016864 Texas Hend 42213.15400.0138 5/24/2011 John L. Schoellkopf Energia Tejas, LLC 2011-00016864 Texas Hend 42213.15400.0140 11/92011 Joycoc Ruetta Featherston Energia Tejas, LLC 2011-00006664 Texas Hend 42213.15400.0141 3/19/2011 Judy Nell Pritchard, Trustee for Kristie Leigh Wakefield Energia Tejas, LLC 2011-00006664 Texas Hend 42213.15400.0142 4/20/2011 Julia H. Mitchell Julia H. Mitchell 2011-00006664 Texas Hend 42213.15400.0143 6/10/2011 Julia H. Mitchell 2011-00006664 Texas Hend 42213.15400.0143 6/10/2011			·			Henderson
42 213 15400 0154 37/2011 Jerry Wayne Rogers et ux Wanda Beth Rogers Hend 42 213 15400 0156 6/17/2011 John H. Brown et ux Judy aka Final Talley, Ltd. 2011-0001686 Texas Hend 42 213 15400 0136 6/17/2011 John K. Patton et ux Rose Energia Tejas, LLC 2011-0001885 Texas Hend 42 213 15400 0138 5/24/201 John K. Patton et ux Rose Energia Tejas, LLC 2011-0001885 Texas Hend 42 213 15400 0139 8/27/2015 Joy Lock Energia Tejas, LLC 2011-0000664 Texas Hend 42 213 15400 0140 11/9/2011 Joye Ruetta Featherston Energia Tejas, LLC 2011-0000664 Texas Hend 42 213 15400 0140 3/19/201 Julia H. Mitchell Energia Tejas, LLC 2011-00006543 Texas Hend 42 213 15400 0143 6/10/2011 Julia H. Mitchell Energia Tejas, LLC 2011-00006543 Texas Hend 42 213 15400 0143 6/10/2011 Kalty Diane Halbert et vir Yra Wayne Halbert Energia Tejas, LLC 2011-0001869 Texas Hend 42 213 15400 0145 8/24/201 Kalty Diane Halbert et vir Yra Wayne Halbert Energia Tejas, LLC						Henderson
42.213.15400.0136 6/17/201 John H. Brown et ux Judy aka Final Talley, Ltd. Energia Tejas, LLC 2011-00010885 Texas Hend 42.213.15400.0137 5/25/2011 *John K. Patton et ux Rose Energia Tejas, LLC 2011-00010891 Texas Hend 42.213.15400.0138 5/24/2011 John L. Schoelikopf Energia Tejas, LLC 2011-00010894 Texas Hend 42.213.15400.0140 11/9/2011 Joyce Ruetta Featherston Energia Tejas, LLC 2011-00006543 Texas Hend 42.213.15400.0141 3/19/2011 July Neil Pritchard, Trustee for Kristie Leigh Wakefield Energia Tejas, LLC 2011-00006543 Texas Hend 42.213.15400.0142 4/20/2011 Julie Shafer Ciuro Energia Tejas, LLC 2011-00006543 Texas Hend 42.213.15400.0143 6/10/2011 Kithy Diane Halbert et vir Ira Wayne Halbert Energia Tejas, LLC 2011-00010869 Texas Hend 42.213.15400.0145 8/24/2016 Kathy Link O'Brien Resources, LLC 2011-00016861 Texas Hend 42.213.15400.0152 3/30/2011 Keetah Holley, widow Roger Dale Holley Energia Tejas, LLC 2011-0006513 Texas Hend	42.213.15400.0134				2011-00006565 Texas	Henderson
42.213.15400.0137 5/25/2011 *John K. Patton et ux Rose Energia Tejas, LLC 2011-00010851 Texas Hend 42.213.15400.0138 5/24/2011 John L. Schoellkopf Energia Tejas, LLC 2011-00010894 Texas Hend 42.213.15400.0139 8/27/2015 Joy Lock Energia Tejas, LLC 2011-00006564 Texas Hend 42.213.15400.0140 11/9/2011 Joy Regia Tejas, LLC 2011-00007660 Texas Hend 42.213.15400.0141 3/19/2011 Julia H. Mitchell Energia Tejas, LLC 2011-00006543 Texas Hend 42.213.15400.0142 4/20/2011 Julia Shafer Ciuro Energia Tejas, LLC 2011-00006531 Texas Hend 42.213.15400.0143 6/10/2011 Julia Shafer Ciuro Energia Tejas, LLC 2011-0001460 Texas Hend 42.213.15400.0144 6/7/2011 Kathy Diane Halbert et vir Ira Wayne Halbert Energia Tejas, LLC 2011-0001689 Texas Hend 42.213.15400.0145 8/24/2016 Kathy Link O'Brien Resources, LLC 2016-00016263 Texas Hend 42.213.15400.0152 3/30/2011 Keigh Liebly, widow Roger Dale Holley Energia Tejas, LLC 2011-00006514 Texas Hend 42.213.1540						Henderson
42.213.15400.0138 5/24/2011 John L. Schoellkopf Energia Tejas, LLC 2011-00016984 Texas Hend 42.213.15400.0140 11/9/2011 Joyce Ruetta Featherston Energia Tejas, LLC 2011-00006564 Texas Hend 42.213.15400.0141 3/19/2011 Judy Nell Pritchard, Trustee for Kristie Leigh Wakefield Energia Tejas, LLC 2011-00006531 Texas Hend 42.213.15400.0142 4/20/201 Julia H. Mitchell Energia Tejas, LLC 2011-00006531 Texas Hend 42.213.15400.0143 6/10/201 Julie Shafer Ciuro Energia Tejas, LLC 2011-0001689 Texas Hend 42.213.15400.0144 6/7/201 Kithy Diane Halbert et vir Ira Wayne Halbert Energia Tejas, LLC 2011-0001689 Texas Hend 42.213.15400.0145 8/24/206 Kathy Link O'Brien Resources, LLC 2016-00016263 Texas Hend 42.213.15400.0146 3/30/201 Keetah Holley, widow Roger Dale Holley Energia Tejas, LLC 2011-0006513 Texas Hend 42.213.15400.0152 3/30/201 Leel Lambright Energia Tejas, LLC 2011-0006514 Texas Hend 42.213.15400.0152 3/30/201 Leel Lambright Energia Tejas, LLC 2011-00016263 Texas <td></td> <td></td> <td></td> <td></td> <td></td> <td>Henderson</td>						Henderson
42.213.15400.0139 8/27/2015 Joy Lock Energia Tejas, LLC 2011-00006564 Texas Hend 42.213.15400.0140 11/9/2011 Joyce Ruetta Featherston Energia Tejas, LLC 2012-00007060 Texas Hend 42.213.15400.0141 3/19/2011 Judy Nell Pritchard, Trustee for Kristie Leigh Wakefield Energia Tejas, LLC 2011-00006543 Texas Hend 42.213.15400.0142 4/20/201 Julia H. Mitchell Energia Tejas, LLC 2011-00006531 Texas Hend 42.213.15400.0143 6/10/2011 Julie Shafer Ciuro Energia Tejas, LLC 2011-00014106 Texas Hend 42.213.15400.0144 6/7/201 Kathy Diane Halbert et vir Ira Wayne Halbert Energia Tejas, LLC 2011-00016869 Texas Hend 42.213.15400.0145 8/24/201 Kathy Link O'Brien Resources, LLC 2016-00016263 Texas Hend 42.213.15400.0152 3/30/201 Keetah Holley, widow Roger Dale Holley Energia Tejas, LLC 2011-0006513 Texas Hend 42.213.15400.0152 3/30/201 Lengli Lembright Energia Tejas, LLC 2011-0006514 Texas Hend 42.213.15400.0152 Hend						Henderson
42.213.15400.0140 11/9/2011 Joyce Ruetta Featherston Energia Tejas, LLC 2012-00007060 Texas Hend 42.213.15400.0141 3/19/2011 Judy Nell Pritchard, Trustee for Kristie Leigh Wakefield Energia Tejas, LLC 2011-00006543 Texas Hend 42.213.15400.0142 4/20/2011 Julia H. Mitchell Energia Tejas, LLC 2011-00006531 Texas Hend 42.213.15400.0143 6/10/2011 Julie Shafer Ciuro Energia Tejas, LLC 2011-00014106 Texas Hend 42.213.15400.0144 6/7/201 Kathy Diane Halbert et vir Ira Wayne Halbert Energia Tejas, LLC 2011-00010869 Texas Hend 42.213.15400.0145 8/24/2016 Kathy Link O'Brien Resources, LLC 2016-00016263 Texas Hend 42.213.15400.0146 3/30/201 Keetah Holley, widow Roger Dale Holley Energia Tejas, LLC 2011-00006513 Texas Hend 42.213.15400.0152 4/12/201 Liegh L. Williams Energia Tejas, LLC 2011-00006514 Texas Hend 42.213.15400.0152 3/30/201 Lenell Lambright Energia Tejas, LLC 2011-00006514 Texas Hend 42.213.15400.0152 4/12/201 Lenell Lambright Energia Tejas, LLC 2011-00006514 T			·			Henderson
42.213.15400.0141 3/19/2011 Judy Nell Pritchard, Trustee for Kristie Leigh Wakefield Energia Tejas, LLC 2011-00006543 Texas Hend 42.213.15400.0142 4/20/2011 Julia H. Mitchell Energia Tejas, LLC 2011-00006531 Texas Hend 42.213.15400.0143 6/10/2011 Julie Shafer Ciuro Energia Tejas, LLC 2011-00014106 Texas Hend 42.213.15400.0144 6/7/201 Kathy Diane Halbert et vir Ira Wayne Halbert Energia Tejas, LLC 2011-0001689 Texas Hend 42.213.15400.0145 8/24/2016 Kathy Link O'Brien Resources, LLC 2016-00016263 Texas Hend 42.213.15400.0146 3/30/2011 Keetah Holley, widow Roger Dale Holley Energia Tejas, LLC 2011-00006513 Texas Hend 42.213.15400.0152 4/12/201 Leigh L. Williams Energia Tejas, LLC 2011-00006514 Texas Hend 42.213.15400.0152 3/30/2011 Lenell Lambright Energia Tejas, LLC 2011-00006514 Texas Hend 42.213.15400.0152 3/30/2011 Lenell Lambright Energia Tejas, LLC 2011-00006514 Texas Hend 42.213.15400.0152 3/30/2011 Lenell Lambright Energia Tejas, LLC 2011-00006514 Texas </td <td></td> <td></td> <td></td> <td></td> <td></td> <td>Henderson</td>						Henderson
42.213.15400.0142 4/20/2011 Julia H. Mitchell Energia Tejas, LLC 2011-00006531 Texas Hend 42.213.15400.0143 6/10/2011 Julie Shafer Ciuro Energia Tejas, LLC 2011-00014106 Texas Hend 42.213.15400.0144 6/7/2011 Kathy Diane Halbert et vir Ira Wayne Halbert Hend 42.213.15400.0145 8/24/2016 Kathy Link 2011-00016263 Texas Hend 42.213.15400.0146 3/30/2011 Keetah Holley, widow Roger Dale Holley Energia Tejas, LLC 2011-00006513 Texas Hend 42.213.15400.0152 4/12/201 Leigh L. Williams Energia Tejas, LLC 2011-00006514 Texas Hend 42.213.15400.0152 3/30/2011 Lenell Lambright Energia Tejas, LLC 2011-00016268 Texas Hend 42.213.15400.0152 3/30/2011 Lenell Lambright Energia Tejas, LLC 2011-00016268 Texas Hend 42.213.15400.0152 3/30/2011 Lenell Lambright Energia Tejas, LLC 2011-00016268 Texas Hend 42.213.15400.0152 1/12/2011 Lenell Lambright Energia Tejas, LLC 2011-00016268 Texas Hend						Henderson
42.213.15400.0143 6/10/2011 Julie Shafer Ciuro Energia Tejas, LLC 2011-00014816 Texas Hend 42.213.15400.0144 6/7/2011 Kathy Diane Halbert et vir Ira Wayne Halbert Energia Tejas, LLC 2011-00010869 Texas Hend 42.213.15400.0145 8/24/2016 Kathy Link O'Brien Resources, LLC 2016-00016263 Texas Hend 42.213.15400.0146 3/30/201 Keetah Holley, widow Roger Dale Holley Energia Tejas, LLC 2011-00006513 Texas Hend 42.213.15400.0152 4/12/201 Leigh L. Williams Energia Tejas, LLC 2011-00006514 Texas Hend 42.213.15400.0152 3/30/201 Lenell Lambright Energia Tejas, LLC 2011-00016288 Texas Hend						Henderson
42.213.15400.0144 6/7/2011 Kathy Diane Halbert et vir Ira Wayne Halbert Energia Tejas, LLC 2011-00010869 Texas Hend 42.213.15400.0145 8/24/2016 Kathy Link O'Brien Resources, LLC 2016-00016263 Texas Hend 42.213.15400.0146 3/30/2011 Keetah Holley, widow Roger Dale Holley Energia Tejas, LLC 2011-00006513 Texas Hend 42.213.15400.0151 4/12/2011 Leigh L. Williams Energia Tejas, LLC 2011-00006514 Texas Hend 42.213.15400.0152 3/30/2011 Lenell Lambright Energia Tejas, LLC 2011-00016282 Texas Hend						Henderson
42.213.15400.0145 8/24/2016 Kathy Link O'Brien Resources, LLC 2016-00016263 Texas Hend 42.213.15400.0146 3/30/2011 Keetah Holley, widow Roger Dale Holley Energia Tejas, LLC 2011-00006513 Texas Hend 42.213.15400.0150 4/12/201 Leigh L. Williams Energia Tejas, LLC 2011-00006514 Texas Hend 42.213.15400.0152 3/30/2011 Lenell Lambright Energia Tejas, LLC 2011-00012628 Texas Hend						Henderson
42.213.15400.0146 3/30/2011 Keetah Holley, widow Roger Dale Holley Hend 42.213.15400.0151 4/12/2011 Leigh L. Williams Energia Tejas, LLC 2011-00006514 Texas Hend 42.213.15400.0152 3/30/2011 Lenell Lambright Energia Tejas, LLC 2011-00012628 Texas Hend						Henderson
42.213.15400.0151 4/12/2011 Leigh L. Williams Energia Tejas, LLC 2011-00006514 Texas Hend 42.213.15400.0152 3/30/2011 Lenell Lambright Energia Tejas, LLC 2011-00012628 Texas Hend						
42.213.15400.0152 3/30/2011 Lenell Lambright Energia Tejas, LLC 2011-00012628 Texas Hend						Henderson
						Henderson
42.213.1094UU.0103 4/14/2011 Leona Earl May Energia Tejas, LLC 2011-00006534 Texas Hend						Henderson
	42.213.15400.0153	4/14/2011	Leona Eari May	∟nergia rejas, LLC	2011-00006534 Texas	Henderson

	Date	Lessor	Lessee	Book-Page State	County
42.213.15400.0154	6/24/2011	Lillian L. Meriwether	Energia Tejas, LLC	2011-00010881 Texas	Henderson
42.213.15400.0156	3/17/2011	Lonnie Clint Vick, Sr. et ux Regina Anne Vick	Energia Tejas, LLC	2011-00006505 Texas	Henderson
42.213.15400.0158	4/3/2011	Jeanne D. Waters, by Lindsley Waters III, A-I-F	Energia Tejas, LLC	2011-00010828 Texas	Henderson
42.213.15400.0159	4/12/2011	Maggie Nell Youngblood	Energia Tejas, LLC	2011-00006562 Texas	Henderson
42.213.15400.0161	5/27/2011	*Margaret Avriette, Trustee of the Gene and Margaret Avriette Family Trust	Energia Tejas, LLC	2011-00010890 Texas	Henderson
42.213.15400.0162	4/14/2011	Margaret Ellen Robison (dec'd)	Energia Tejas, LLC	2011-00006515 Texas	Henderson
42.213.15400.0164	4/28/2011	Marvin Holcomb , Glenn Holcomb & Ronnie Holcomb	Energia Tejas, LLC	2011-00010870 Texas	Henderson
42.213.15400.0165	6/9/2011	Mary Ann Cohen	Energia Tejas, LLC	2011-00010906 Texas	Henderson
42.213.15400.0166	5/27/2011	Mary Lambright	Energia Tejas, LLC	2011-00010903 Texas	Henderson
42.213.15400.0167	5/12/2011	Mary Lambright	Energia Tejas, LLC	2011-00010838 Texas	Henderson
42.213.15400.0168	3/8/2011	*Mary Joanne Wilson (partially sold)	Energia Tejas, LLC	2011-00010834 Texas	Henderson
42.213.15400.0169	11/9/2011	Melvin Lynn Dehart	Energia Tejas, LLC	2012-00007059 Texas	Henderson
42.213.15400.0171	3/23/2011	Michael Lambright et al	Energia Tejas, LLC	2011-00012613 Texas	Henderson
42.213.15400.0172	5/12/2011	Mila Lambright Baker	Energia Tejas, LLC	2011-00012595 Texas	Henderson
42.213.15400.0173	7/14/2011	Nancy Burnett	Energia Tejas, LLC	2011-00012592 Texas	Henderson
42.213.15400.0176	5/4/2011	Annette Goolsby et al	Energia Tejas, LLC	2014-00006543 Texas	Henderson
42.213.15400.0177	3/30/2011	Norine Kelly (now Brenda Nikel, et al)	Energia Tejas, LLC	2011-00006570 Texas	Henderson
42.213.15400.0178	8/4/2011	Norine Kelly	Energia Tejas, LLC	2011-00014112 Texas	Henderson
42.213.15400.0179	5/27/2011	Obdiah Lloyd Lambright	Energia Tejas, LLC	2011-00010848 Texas	Henderson
42.213.15400.0180	5/3/2011	Obdiah Lloyd Lambright	Energia Tejas, LLC	2011-00010833 Texas	Henderson
42.213.15400.0181		Ozie Dean Till	Energia Tejas, LLC	2011-00012626 Texas	Henderson
42.213.15400.0183	4/14/2011	Patricia Ann Bragg	Energia Tejas, LLC	2011-00006571 Texas	Henderson
42.213.15400.0184	6/9/2011	Patricia Hillman Murrell	Energia Tejas, LLC	2011-00010902 Texas	Henderson
42.213.15400.0185	3/17/2011		Energia Tejas, LLC	2011-00006546 Texas	Henderson
42.213.15400.0186	5/26/2011	Patrick H. Wageman	Energia Tejas, LLC	2011-00010835 Texas	Henderson
42.213.15400.0187		Patsy Jo Lambright	Energia Tejas, LLC	2011-00010893 Texas	Henderson
42.213.15400.0188	5/19/2011	Patsy Lindsey Hardy et vir Nicky	Energia Tejas, LLC	2011-00010836 Texas	Henderson
42.213.15400.0189	5/4/2011	Peggy H. Kirkland, A-I-F for Norma Dell Lambright Hillhouse	Energia Tejas, LLC	2011-00010844 Texas	Henderson
42.213.15400.0190	5/27/2011	Peggy H. Kirkland, A-I-F for Norma Dell Lambright Hillhouse	Energia Tejas, LLC	2011-00010843 Texas	Henderson
42.213.15400.0191		Peter Wesley Williford	Energia Tejas, LLC	2011-00006516 Texas	Henderson
42.213.15400.0192	2/25/2011	Randy Allen Wilbanks	Energia Tejas, LLC	2011-00006552 Texas	Henderson
42.213.15400.0193	7/26/2016	Randy Miller and Elizabeth Miller	O'Brien Resources, LLC	2016-00016264 Texas	Henderson
42.213.15400.0195	8/25/2011	Rentier, Inc. (renewed #1242)	Energia Tejas, LLC	2011-00014105 Texas	Henderson
42.213.15400.0196	3/31/2011	Richard D. Couron	Energia Tejas, LLC	2011-00006517 Texas	Henderson
42.213.15400.0197	4/3/2011	Richard E. Dwelle et ux Peggy D. Dwelle	Energia Tejas, LLC	2011-00006569 Texas	Henderson
42.213.15400.0198	5/11/2011	Richard H. Eaton	Energia Tejas, LLC	2011-00010855 Texas	Henderson
42.213.15400.0199	8/5/2011	Rick G. Baldree et ux Denise Baldree	Energia Tejas, LLC	2012-00009534 Texas	Henderson
42.213.15400.0200	9/23/2011	The Robert Welch Foundation	Energia Tejas, LLC	2011-00014103 Texas	Henderson
42.213.15400.0201	5/23/2011	Robert Dehart	Energia Tejas, LLC	2011-00010891 Texas	Henderson
42.213.15400.0202	4/14/2011	Robert Edward Lee Williford, II	Energia Tejas, LLC	2011-00006533 Texas	Henderson
42.213.15400.0203	5/11/2011	Robert W. Percival	Energia Tejas, LLC	2011-00010862 Texas	Henderson
42.213.15400.0204	3/30/2011	Robert Avon Holley	Energia Tejas, LLC	2011-00006541 Texas	Henderson
42.213.15400.0205	3/21/2011	Ronald Charles Holcomb et ux Anne Holcomb	Energia Tejas, LLC	2011-00006556 Texas	Henderson
42.213.15400.0206	5/13/2011	Ronald C. Pate et ux Kimberly	Energia Tejas, LLC	2011-00010830 Texas	Henderson
42.213.15400.0207	3/7/2011	Ronald L. Williams	Energia Tejas, LLC	2017-00010907 Texas	Henderson
42.213.15400.0208	8/8/2011	Ronald L. Williams	Energia Tejas, LLC	2017-00010907 Texas	Henderson
42.213.15400.0209	7/6/2011	Ronnie O. Justice et ux Peggy	Energia Tejas, LLC	2011-00010874 Texas	Henderson
42.213.15400.0210		Roy Lee Lambright	Energia Tejas, LLC	2011-00010905 Texas	Henderson
42.213.15400.0211		Royce W. Benson et ux Anita D.	Energia Tejas, LLC	2011-00006532 Texas	Henderson
42.213.15400.0212		Ruthie F. Lambright	Energia Tejas, LLC	2011-00006522 Texas	Henderson
42.213.15400.0213		Sharon Lorraine Anderson	Energia Tejas, LLC	2011-00006510 Texas	Henderson
42.213.15400.0215		Shirley B. Hilton	Energia Tejas, LLC	2011-00010846 Texas	Henderson
42.213.15400.0216		Shirley B. Hilton	Energia Tejas, LLC	2011-00006506 Texas	Henderson
42.213.15400.0217		Sonja Lewis Simon	Energia Tejas, LLC	2011-00012623 Texas	Henderson
42.213.15400.0218		Sonja Lewis Simon	Energia Tejas, LLC	2011-00012606 Texas	Henderson
42.213.15400.0219		Sonya Havard Moon	Energia Tejas, LLC	2011-00012615 Texas	Henderson
42.213.15400.0220		Stephen Gwyn Landis et ux Dorothy Anne Landis	Energia Tejas, LLC	2011-00006567 Texas	Henderson
42.213.15400.0221	3/7/2011	Stephen Kelley Presswood	Energia Tejas, LLC	2011-00006502 Texas	Henderson
42.213.15400.0224		Susan P. Meriwether	Energia Tejas, LLC	2011-00010882 Texas	Henderson
42.213.15400.0226		Thomas C. Browning	Energia Tejas, LLC	2011-00010837 Texas	Henderson
42.213.15400.0226		Thomas L. Soutter	Energia Tejas, LLC	2011-00010657 Texas	Henderson
12.210.10400.0221		Thomas L. Soutter	Energia Tejas, LLC Energia Tejas, LLC	2011-00006547 Texas 2011-00012624 Texas	Henderson
42 213 15400 0228	5/31/2011				
42.213.15400.0228 42.213.15400.0229					
42.213.15400.0229	8/15/2011	Todd Havard	Energia Tejas, LLC	2011-00012617 Texas	Henderson
42.213.15400.0228 42.213.15400.0229 42.213.15400.0230 42.213.15400.0231	8/15/2011 5/12/2011				

Lease #	Date	Lessor	Lessee	Book-Page State	County
42.213.15400.0232	5/13/2011	Vera Odle, A-I-F for Helen Burns	Energia Tejas, LLC	2011-00010892 Texas	Henderson
42.213.15400.0235	3/7/2011	Virginia Gail Willis	Energia Tejas, LLC	2011-00012596 Texas	Henderson
42.213.15400.0236		VWP JR, Inc. (Victor W. Pryor, Jr.)	Energia Tejas, LLC	2011-00012603 Texas	Henderson
42.213.15400.0237		VWP JR, Inc. (Victor W. Pryor, Jr.)	Energia Tejas, LLC	2011-00012599 Texas	Henderson
42.213.15400.0238		*W. G. Darsey, III	Energia Tejas, LLC	2011-00010872 Texas	Henderson
42.213.15400.0239	9/23/2011	Walter C. Windsor (renewed #1230)	Energia Tejas, LLC	2012-00009533 Texas	Henderson
42.213.15400.0240		Wanda Lindsey Erekson et vir Robbiie	Energia Tejas, LLC	2011-00010849 Texas	Henderson
42.213.15400.0241		Wayne Mitchell	Energia Tejas, LLC	2011-00010866 Texas	Henderson
42.213.15400.0243		Weldon J. Taylor (Petra Zeigerman)	Energia Tejas, LLC	2011-00012594 Texas	Henderson
42.213.15400.0247	2/23/2016		O'Brien Resources, LLC	2016-00003043 Texas	Henderson
42.213.15400.0248	4/14/2011	William Richards Williford (dec'd)	Energia Tejas, LLC	2011-00006519 Texas	Henderson
42.213.15400.0249		William Thomas Meriwether	Energia Tejas, LLC	2011-00010845 Texas	Henderson
42.213.15400.0250		Windsor Interests, Ltd.	Energia Tejas, LLC	2011-00014111 Texas	Henderson
42.213.15400.0251		Yukon Energy, LLC	Energia Tejas, LLC	2011-00012605 Texas	Henderson
42.213.15400.0253		Charles P. Lewis	Energia Tejas, LLC	2272-002 Texas	Anderson
42.213.15400.0254	8/4/2011	Ira Jene Jackson	Energia Tejas, LLC	2271-797 Texas	Anderson
42.213.15400.0255		*J.D. Black	Energia Tejas, LLC	2271-0800 Texas	Anderson
42.213.15400.0256		James C. Hanks, Jr.	O'Brien Resources, LLC	2518-479 Texas	Anderson
42.213.15400.0257		Kenneth V. Malone as trustee of Mark Ray Malone Living Trust	O'Brien Resources, LLC	2518-487 Texas	Anderson
42.213.15400.0259		Lloyd E. Hanks et ux Barbara	Energia Tejas, LLC	2272-005 Texas	Anderson
42.213.15400.0260		Marlene Kay Hardy a/k/a Merlene Kay Common	O'Brien Resources, LLC	2011-00010887 Texas	Anderson
42.213.15400.0262	5/4/2016	Paula Barnes	O'Brien Resources, LLC	2518-467 Texas	Anderson
42.213.15400.0263		Ronald Dale Matney	O'Brien Resources, LLC	2515-670 Texas	Anderson
42.213.15400.0265		Sonja Hanks-Rigsby	Energia Tejas, LLC	2271-0799 Texas	Anderson
42.213.15400.0266	8/10/2011	Sonja Lewis Simon	Energia Tejas, LLC	2272-0001 Texas	Anderson
42.213.15400.0267		Timothy D. Eyssen	O'Brien Resources, LLC	2518-475 Texas	Anderson
42.213.15400.0268		Ben & Patsy Ruth Johnson Revocable Trust	O'Brien Resources, LLC	2318-562 Texas	Anderson
42.213.15400.0270		Teresa Heilig	O'Brien Resources, LLC	2015-00014861 Texas	Henderson
42.213.15400.0275		Bettye L. Newberry	O'Brien Resources, LLC	2012-00012213 Texas	Henderson
42.213.15400.0276		Gregory Paul Cox	O'Brien Resources, LLC	2012-00012231 Texas	Henderson
42.213.15400.0277		James G. Cox	O'Brien Resources, LLC	2012-00012214 Texas	Henderson
42.213.15400.0278		Barbara L. Sides	O'Brien Resources, LLC	2013-00006352 Texas	Henderson
42.213.15400.0281		Rhonda Mahoney	O'Brien Resources, LLC	2012-00015812 Texas	Henderson
42.213.15400.0283		Glen McWhorter	O'Brien Resources, LLC	2013-00009786 Texas	Henderson
42.213.15400.0284		*James McWhorter	O'Brien Resources, LLC	2013-00009785 Texas	Henderson
42.213.15400.0285		David McWhorter	O'Brien Resources, LLC	2013-00009787 Texas	Henderson
42.213.15400.0287		Elizabeth Gardner	O'Brien Resources, LLC	2318-559 Texas	Anderson
42.213.15400.0291	9/6/2012	Joe B. Gardner	O'Brien Resources, LLC	2318-556 Texas	Anderson
42.213.15400.0293		Lana G. Hart et vir	O'Brien Resources, LLC	2012-00015815 Texas	Henderson
42.213.15400.0294	8/9/2012	George G. Wood	O'Brien Resources, LLC	2012-00014511 Texas	Henderson
42.213.15400.0297	7/19/2012	Daphne Elizabeth Carpenter	O'Brien Resources, LLC	2012-00012227 Texas	Henderson
42.213.15400.0298	12/3/2015	Gary Don Wilbanks et ux	O'Brien Resources, LLC	2016-00001912 Texas	Henderson
42.213.15400.0300		James Bert Griffin Trust	O'Brien Resources, LLC	2012-00014509 Texas	Henderson
42.213.15400.0302		*Suzanne Comeilus Krause	O'Brien Resources, LLC	2318-564 Texas	Anderson
42.213.15400.0303		E. E. Cornelius, III	O'Brien Resources, LLC	2318-563 Texas	Anderson
42.213.15400.0305		Damon Douglas	O'Brien Resources, LLC	2012-00016072 Texas	Henderson
42.213.15400.0307		Flora Dell Price, Dorothy McGee AIF	O'Brien Resources, LLC	2324-476 Texas	Anderson
42.213.15400.0309		Linda Hurtado	O'Brien Resources, LLC	2332-491 Texas	Anderson
42.213.15400.0311		Eugene E. Soutter et ux (now deceased)	O'Brien Resources, LLC	2012-00016076 Texas	Henderson
42.213.15400.0312		Patricia Ward Bryant	O'Brien Resources, LLC	2013-00000482 Texas	Henderson
42.213.15400.0314	11/7/2012	Amy Parrish Wischnewsky	O'Brien Resources, LLC	2330-774 Texas	Anderson
42.213.15400.0315	2/16/2017	Kenneth North and Lydia J. North	O'Brien Resources, LLC	2574-0655 Texas	Anderson
42.213.15400.0318	2/1/2016	Eddie Don Price	O'Brien Resources, LLC	2016-00005131 Texas	Henderson
42.213.15400.0323	10/18/2012	Holly Avriette Fox	O'Brien Resources, LLC	2012-00015823 Texas	Henderson
42.213.15400.0325	2/1/2016	Stephen Wayne Woodall	O'Brien Resources, LLC	2016-00003606 Texas	Henderson
42.213.15400.0326	2/1/2016	Della Mae Woodall Turner	O'Brien Resources, LLC	2016-00003605 Texas	Henderson
42.213.15400.0327	11/7/2012	Stephanie Parrish	O'Brien Resources, LLC	2377-534 Texas	Anderson
42.213.15400.0328	8/13/2012	Julius W. Wylie	O'Brien Resources, LLC	2013-00001772 Texas	Henderson
42.213.15400.0329	11/12/2012	Alfred M. Yeatts, Jr.	O'Brien Resources, LLC	2013-00000468 Texas	Henderson
42.213.15400.0330	2/1/2016	Kenneth Goodrum	O'Brien Resources, LLC	2016-00005126 Texas	Henderson
42.213.15400.0331	11/17/2015	Clarine Kellam	O'Brien Resources, LLC	2016-00000589 Texas	Henderson
42.213.15400.0332	9/19/2012	Gilbert S. Hernandez	O'Brien Resources, LLC	2012-00016073 Texas	Henderson
42.213.15400.0333	11/6/2012	Joy Anderson	O'Brien Resources, LLC	2337-728 Texas	Anderson
42.213.15400.0335	12/12/2012	*J. Bradley Jeffreys	O'Brien Resources, LLC	2013-00000469 Texas	Henderson
42.213.15400.0336	4/4/2016	Lynda Lou Goodrum	O'Brien Resources, LLC	2016-00006209 Texas	Henderson

Lease #	Date Lessor	Lessee	Book-Page State	County
42.213.15400.0337	11/16/2012 William "Bain" Ward, III	O'Brien Resources, LLC	2013-00000481 Texas	Henderson
42.213.15400.0338	11/17/2015 Steve Harrington	O'Brien Resources, LLC	2015-00015673 Texas	Henderson
42.213.15400.0339	11/17/2015 John Harrington	O'Brien Resources, LLC	2015-00015668 Texas	Henderson
42.213.15400.0340	10/12/2015 Charlotte Heppel	O'Brien Resources, LLC	2015-00015672 Texas	Henderson
42.213.15400.0341	11/17/2015 Tom Harrington III	O'Brien Resources, LLC	2015-00015674 Texas	Henderson
42.213.15400.0342	2/1/2016 Roger Warren Price	O'Brien Resources, LLC	2016-00005125 Texas	Henderson
42.213.15400.0344	12/14/2012 Sydonia L. Sonnenburg (deceased)	O'Brien Resources, LLC	2013-00000483 Texas	Henderson
42.213.15400.0345	11/13/2012 Robin S. Ritchie	O'Brien Resources, LLC	2013-00000484 Texas	Henderson
42.213.15400.0346	12/10/2012 Chieftain Exploration Co	O'Brien Resources, LLC	2013-00000479 Texas	Henderson
42.213.15400.0347	2/1/2016 Rhonda S. Davidson	O'Brien Resources, LLC	2016-00006216 Texas	Henderson
42.213.15400.0349	12/16/2015 Gwendolyn Kay Michael	O'Brien Resources, LLC	2016-00001915 Texas	Henderson
42.213.15400.0351	9/28/2012 Tracey Lambright et al	O'Brien Resources, LLC	2013-00001773 Texas	Henderson
42.213.15400.0352	12/18/2012 Neill Oil, LLC	O'Brien Resources, LLC	2013-00001766 Texas	Henderson
42.213.15400.0354	1/14/2013 *Sunmore Energy Partners, LLC	O'Brien Resources, LLC	2013-00001771 Texas	Henderson
42.213.15400.0355	1/17/2013 Est of T. L. Huddleston by H. T. Hyde, Jr., Trustee	O'Brien Resources, LLC	2013-00006358 Texas	Henderson
42.213.15400.0356	5/9/2013 Hugh Craft	O'Brien Resources, LLC	2013-00006348 Texas	Henderson
42.213.15400.0357	1/24/2013 Libby Craft	O'Brien Resources, LLC	2013-00006362 Texas	Henderson
42.213.15400.0358	6/14/2016 Mary L. Fitch	O'Brien Resources, LLC	2016-00012779 Texas	Henderson
42.213.15400.0360	2/20/2013 Bandera Minerals, LLC	O'Brien Resources, LLC	2013-00009794 Texas	Henderson
42.213.15400.0362	1/9/2013 Diane Gilbert Clark	O'Brien Resources, LLC	2013-00006347 Texas	Henderson
42.213.15400.0363	2/4/2013 Schare, Ltd.	O'Brien Resources, LLC O'Brien Resources, LLC	2013-00006365 Texas	Henderson
42.213.15400.0364 42.213.15400.0365	2/4/2013 Panacea Energy, Inc.		2013-00006363 Texas	Henderson
42.213.15400.0365	2/4/2013 K. Phil Fleetwood 1/23/2013 Nathan Leslie James	O'Brien Resources, LLC O'Brien Resources, LLC	2013-00006364 Texas 2013-00006360 Texas	Henderson Henderson
42.213.15400.0367	1/23/2013 Nathan Lesile James 1/23/2013 Gleith James Mann	O'Brien Resources, LLC	2013-00006360 Texas	Henderson
42.213.15400.0367	2/21/2013 Alva June Daily	O'Brien Resources, LLC	2013-00006351 Texas	Henderson
42.213.15400.0368	1/23/2013 Eva Evans (minerals now to heirs)	O'Brien Resources, LLC	2013-00006354 Texas	Henderson
42.213.15400.0309	2/26/2013 James Scott, Jr.	O'Brien Resources, LLC	2013-00006357 Texas	Henderson
42.213.15400.0371	2/26/2013 James Scott, Sr.	O'Brien Resources, LLC	2013-00006366 Texas	Henderson
42.213.15400.0372	2/20/2013 Royce M. Benson	O'Brien Resources, LLC	2013-00006353 Texas	Henderson
42.213.15400.0375	3/4/2013 Rena S. Hemphill Bennett	O'Brien Resources, LLC	2377-509 Texas	Anderson
42.213.15400.0376	1/22/2013 Katrina Lynn Pound	O'Brien Resources, LLC	2013-00006350 Texas	Henderson
42.213.15400.0377	8/16/2011 Gary Wayne Dennis, Trustee	Energia Tejas, LLC	2012-00010613 Texas	Henderson
42.213.15400.0378	1/22/2013 Benjamin Aaron Cohen	O'Brien Resources, LLC	2013-00006356 Texas	Henderson
42.213.15400.0379	4/18/2013 Troy C. House	O'Brien Resources, LLC	2013-00009795 Texas	Henderson
42.213.15400.0380	2/28/2013 Mary Jo Mullins	O'Brien Resources, LLC	2016-00008872 Texas	Henderson
42.213.15400.0381	4/17/2013 Mack Powell	O'Brien Resources, LLC	2013-00009797 Texas	Henderson
42.213.15400.0382	4/18/2013 Geraldine Rounsavall	O'Brien Resources, LLC	2013-00009798 Texas	Henderson
42.213.15400.0384	2/26/2013 Owen Oil & Gas Properties, LLC	O'Brien Resources, LLC	2013-00017297 Texas	Henderson
42.213.15400.0385	2/18/2013 George S. Tallichet	O'Brien Resources, LLC	2013-00009804 Texas	Henderson
42.213.15400.0386	2/18/2013 Henri Lucien Tallichet	O'Brien Resources, LLC	2013-00009805 Texas	Henderson
42.213.15400.0387	2/18/2013 George S. Heyer, Jr.	O'Brien Resources, LLC	2013-00009803 Texas	Henderson
42.213.15400.0388	11/19/2012 Kenneth Haesly	O'Brien Resources, LLC	2013-00009802 Texas	Henderson
42.213.15400.0389	3/25/2013 Michael A. Maynard	O'Brien Resources, LLC	2013-00009806 Texas	Henderson
42.213.15400.0390	5/2/2013 Lucille Collins	O'Brien Resources, LLC	2013-00009800 Texas	Henderson
42.213.15400.0391	5/2/2013 Glenda Langley	O'Brien Resources, LLC	2013-00009799 Texas	Henderson
42.213.15400.0392	4/30/2013 Southwest Mineral Group, Ltd.	O'Brien Resources, LLC	2013-00009801 Texas	Henderson
42.213.15400.0393	4/9/2013 Margaret P. Tyson	O'Brien Resources, LLC	2013-00009788 Texas	Henderson
42.213.15400.0394	4/30/2013 Glenda K. Hayes	O'Brien Resources, LLC	2013-00009790 Texas	Henderson
42.213.15400.0395	5/23/2016 Faye P. Harris	O'Brien Resources, LLC	2016-00012776 Texas	Henderson
42.213.15400.0396	5/9/2013 Rosemary Williams	O'Brien Resources, LLC	2013-00009789 Texas	Henderson
42.213.15400.0397	4/16/2013 The Estate of Buck J. Wynne, Jr.	O'Brien Resources, LLC	2013-00009777 Texas	Henderson
42.213.15400.0398	4/16/2013 Adele Monning	O'Brien Resources, LLC	2013-00009780 Texas	Henderson
42.213.15400.0399	4/16/2013 Oscar Bruce Chamberlain	O'Brien Resources, LLC	2013-00009778 Texas	Henderson
42.213.15400.0400	4/16/2013 Karen Hunnicutt	O'Brien Resources, LLC	2013-00009781 Texas	Henderson
42.213.15400.0401	4/16/2013 Holly Chamberlain aka Holly Casey	O'Brien Resources, LLC	2013-00009779 Texas	Henderson
42.213.15400.0402	4/17/2013 Kirby Minerals	O'Brien Resources, LLC	2013-00009782 Texas	Henderson
42.213.15400.0403	7/28/2013 Ronald L. Williams	O'Brien Resources, LLC	2017-00010907 Texas	Henderson
42.213.15400.0404	1/14/2013 Lolita Anne Rothfuss, f/k/a Lolita A. Langham	O'Brien Resources, LLC	2013-00009776 Texas	Henderson
42.213.15400.0407	7/1/2013 Peggy Kirkland, Ind. & as AIF for Norma Dell Lambright Hillhouse et al	O'Brien Resources, LLC	2383-584 Texas	Anderson
42.213.15400.0408	6/18/2013 Bailey Smiley	O'Brien Resources, LLC	2013-00009775 Texas	Henderson
42.213.15400.0409	8/2/2013 Brian Utz et ux Janet	O'Brien Resources, LLC	2013-00014065 Texas	Henderson
42.213.15400.0410	7/24/2013 Richard C. Palmer et ux	O'Brien Resources, LLC	2013-00014064 Texas	Hendeson
42.213.15400.0411	7/24/2013 Pamala Solis et vir	O'Brien Resources, LLC	2013-00014063 Texas	Henderson
42.213.15400.0412	7/12/2013 Loughridge C. Murrell, Jr.	O'Brien Resources, LLC	2013-00014061 Texas	Henderson

Lease #	Date	Lessor	Lessee	Book-Page State	County
42.213.15400.0414		Dorothy Brooks Steele	O'Brien Resources, LLC	2389-507 Texas	Anderson
42.213.15400.0415		James M. Chatfield	O'Brien Resources, LLC	2013-00017299 Texas	Henderson
42.213.15400.0416		Helen Weeks-Chatfield	O'Brien Resources, LLC	2013-00017300 Texas	Henderson
42.213.15400.0417	8/5/2013	Betty Darsey Woods	O'Brien Resources, LLC	2377-526 Texas	Anderson
42.213.15400.0417	8/5/2013	Betty Darsey Woods	O'Brien Resources, LLC	2377-526 Texas	Anderson
42.213.15400.0418	8/5/2013	*W. G. Darsey, III	O'Brien Resources, LLC	2377-530 Texas	Anderson
42.213.15400.0419		Dolores A. Haytar	O'Brien Resources, LLC	2383-608 Texas	Anderson
42.213.15400.0420		Karen Sue Kale	O'Brien Resources, LLC	2383-600 Texas	Anderson
42.213.15400.0421		Sheila K. Marshall	O'Brien Resources, LLC	2383-604 Texas	Anderson
42.213.15400.0422		Robert W. Palmer	O'Brien Resources, LLC	2013-00016532 Texas	Henderson
42.213.15400.0425		Kirby Minerals	O'Brien Resources, LLC	2013-00016535 Texas	Henderson
42.213.15400.0426		Bandera Minerals, LLC	O'Brien Resources, LLC	2014-00001070 Texas	Henderson
42.213.15400.0429	11/7/2013		O'Brien Resources, LLC	2389-499 Texas	Anderson
42.213.15400.0430		Hugh R. Craft	O'Brien Resources, LLC	2377-517 Texas	Anderson
42.213.15400.0431		Richard H. Eaton	O'Brien Resources, LLC	2383-592 Texas	Anderson
42.213.15400.0432		Robert W. Percival	O'Brien Resources, LLC	2383-596 Texas	Anderson
42.213.15400.0433		I.P. LaRue Mineral Properties, LLC	O'Brien Resources, LLC	2013-00016541 Texas	Henderson
42.213.15400.0434		Owen Oil & Gas Properties, LLC	O'Brien Resources, LLC	2013-00017298 Texas	Henderson
42.213.15400.0436		Gleith James Mann	O'Brien Resources, LLC	2383-574 Texas	Anderson
42.213.15400.0437		Frederick Palmer	O'Brien Resources, LLC	2013-00016534 Texas	Henderson
42.213.15400.0438		John Palmer	O'Brien Resources, LLC	2013-00016533 Texas	Henderson
42.213.15400.0439		Nathan Leslie James	O'Brien Resources, LLC	2383-578 Texas	Anderson
42.213.15400.0440		Jeffrey W. Eaton	O'Brien Resources, LLC	2383-566 Texas	Anderson
42.213.15400.0441		Terry M. Thorn	O'Brien Resources, LLC	2383-583 Texas	Anderson
42.213.15400.0442		Earl Debbs Hillhouse	O'Brien Resources, LLC	2383-570 Texas	Anderson
42.213.15400.0443		James N. Parsons III	O'Brien Resources, LLC	2383-565 Texas	Anderson
42.213.15400.0444		Peggy Hillhouse Kirkland	O'Brien Resources, LLC	2383-588 Texas	Anderson
42.213.15400.0445		Jean Brooks AIF for Martha Brooks Adams	O'Brien Resources, LLC	2389-503 Texas	Anderson
42.213.15400.0446		Linda L. Kropp	O'Brien Resources, LLC	2013-00016540 Texas	Henderson
42.213.15400.0448		Jeana Walker	O'Brien Resources, LLC	2013-00016539 Texas	Henderson
42.213.15400.0449		Susan Prescott AIF for Carrold C. Dollar (Deceased)	O'Brien Resources, LLC	2013-00016538 Texas	Henderson
42.213.15400.0450		Herschbach Petroleum Company, Ltd.	O'Brien Resources, LLC	2383-557 Texas	Anderson
42.213.15400.0451		David Anglin	O'Brien Resources, LLC	2389-495 Texas	Anderson
42.213.15400.0452		Southwest Mineral Group, Ltd.	O'Brien Resources, LLC	2389-511 Texas	Anderson
42.213.15400.0453		Herschbach Living Trust and Riverwood Energy, LLC	O'Brien Resources, LLC	2389-618 Texas	Anderson
42.213.15400.0454		Sheila Hunt	O'Brien Resources, LLC	2014-00001082 Texas	Henderson
42.213.15400.0455		Jay N. Besett	O'Brien Resources, LLC	2014-00001002 Texas	Henderson
42.213.15400.0456		Katherine L. Edwards	O'Brien Resources, LLC	2014-00001072 Texas	Henderson
42.213.15400.0457		Margaret Jean Lambright	O'Brien Resources, LLC	2014-00001000 Texas	Henderson
42.213.15400.0458		Barbara Rhine	O'Brien Resources, LLC	2014-00001077 Texas	Henderson
42.213.15400.0459		Michael Dale Lambright	O'Brien Resources, LLC	2014-00001076 Texas	Henderson
			O'Brien Resources, LLC		
42.213.15400.0460 42.213.15400.0461		Audrey Neel David Lambright	O'Brien Resources, LLC	2014-00001080 Texas 2014-00001086 Texas	Henderson Henderson
42.213.15400.0462		William Nell Besett	O'Brien Resources, LLC	2014-00001069 Texas	Henderson
42.213.15400.0463		Pertha Sue Littlefield	O'Brien Resources, LLC	2014-00001009 Texas	Henderson
42.213.15400.0463		Betty Adcock	O'Brien Resources, LLC	2014-00001079 Texas 2014-00001084 Texas	Henderson
42.213.15400.0464		Hazel Elder	O'Brien Resources, LLC O'Brien Resources, LLC	2014-00001084 Texas 2014-00001078 Texas	Henderson
42.213.15400.0466		Shirley N. McKinney	O'Brien Resources, LLC	2014-00001076 Texas	Henderson
42.213.15400.0467			O'Brien Resources, LLC	2014-000016548 Texas	
42.213.15400.0467		Cindy Lambright Kenneth D. Clark	O'Brien Resources, LLC O'Brien Resources, LLC	2014-00006548 Texas 2014-00001074 Texas	Henderson Henderson
		Sherrie Battaqlia			
42.213.15400.0469 42.213.15400.0470		Timothy Lunceford	O'Brien Resources, LLC O'Brien Resources, LLC	2014-00001075 Texas	Henderson
		•	O'Brien Resources, LLC	2417-0300 Texas	Anderson
42.213.15400.0471 42.213.15400.0472		Estate of T. L. Huddleston by H. T. Hyde, Jr. Trustee	O'Brien Resources, LLC O'Brien Resources, LLC	2396-225 Texas	Anderson
		Eveline Lambright Baker		2014-00001068 Texas	Henderson
42.213.15400.0473		Harold Ray Lambright	O'Brien Resources, LLC	2014-00001067 Texas	Henderson
42.213.15400.0474		Carroll Springs Cemetery Assoc.	O'Brien Resources, LLC	2014-00003078 Texas	Henderson
42.213.15400.0475		Jack P. Rahm, Trustee of the Jack P. Rahm Revocable Trust	O'Brien Resources, LLC	2396-0200 Texas	Anderson
42.213.15400.0476		Karen Hunnicutt	O'Brien Resources, LLC	2396-0217 Texas	Anderson
42.213.15400.0477		Rentier Inc. by Randall Pogue President	O'Brien Resources, LLC	2396-0203 Texas	Anderson
42.213.15400.0478		Glenn Riley Brooks	O'Brien Resources, LLC	2396-0205 Texas	Anderson
42.213.15400.0479		VWP Jr, Inc., by Victor W. Pryor, Jr., President	O'Brien Resources, LLC	2396-221 Texas	Anderson
42.213.15400.0480		Estate of Buck J. Wynne, Jr. by Nancy Ann Chandler Executrix	O'Brien Resources, LLC	2396-0209 Texas	Anderson
42.213.15400.0481		Connie Ruth Brooks	O'Brien Resources, LLC	2014-00001073 Texas	Henderson
42.213.15400.0482		Oscar Bruce Chamberlain	O'Brien Resources, LLC	2466-485 Texas	Anderson
42.213.15400.0483	1/16/2014	Holly Chamberlain aka Holly Casey	O'Brien Resources, LLC	2396-213 Texas	Anderson

Lease #	Date	Laccor	Lessee	Book-Page State	County
Lease # 42.213.15400.0484	3/3/2014	Lessor *N. D. Bingham II	O'Brien Resources, LLC	2502-449 Texas	Anderson
42.213.15400.0485		Ann C. Maroe	O'Brien Resources, LLC	2417-303 Texas	Anderson
42.213.15400.0486		Ann C. Maroe	O'Brien Resources, LLC	2014-00003081 Texas	Henderson
42.213.15400.0487		Patsy C. Sorenson	O'Brien Resources, LLC	2014-00003082 Texas	Henderson
42.213.15400.0488	2/3/2014	Betty Darsey Woods	O'Brien Resources, LLC	2417-307 Texas	Anderson
42.213.15400.0489	2/21/2014	Michelle Avriette Ogden	O'Brien Resources, LLC	2014-00003083 Texas	Henderson
42.213.15400.0490	2/7/2014	*W. G. Darsey, III	O'Brien Resources, LLC	2417-311 Texas	Anderson
42.213.15400.0491	2/13/2014	Lynette L. Murrell Gammon	O'Brien Resources, LLC	2014-00003077 Texas	Henderson
42.213.15400.0492	2/12/2014	Nathan Murrell	O'Brien Resources, LLC	2014-00003079 Texas	Henderson
42.213.15400.0494	3/6/2014	Denise Scott Trautz	O'Brien Resources, LLC	2417-292 Texas	Anderson
42.213.15400.0495	2/14/2014	Diane Scott Jovanovich	O'Brien Resources, LLC	2417-296 Texas	Anderson
42.213.15400.0496		Ernest Edwin Moyers	O'Brien Resources, LLC	2014-00006559 Texas	Henderson
42.213.15400.0497		Martin R. Bennett, Receiver - Cause 2014C -00880 Unknown Heirs of Brenda Lambright; Prentice Berwick; LaVerne Davis; Prentice	O'Brien Resources, LLC	2014-00004666 Texas	Henderson
42.213.15400.0498		Kelly Milan Trahan	O'Brien Resources, LLC	2014-00006558 Texas	Henderson
42.213.15400.0500 42.213.15400.0501		Adele Monning	O'Brien Resources, LLC O'Brien Resources, LLC	2417-315 Texas 2014-00006556 Texas	Anderson Henderson
42.213.15400.0504		Father Flanagan's Boys' Home Walter C. Windsor	O'Brien Resources, LLC	2417-320 Texas	Anderson
42.213.15400.0505		Kerry Vance	O'Brien Resources, LLC	2014-00006553 Texas	Henderson
42.213.15400.0506		Chester Vance Chapel	O'Brien Resources, LLC	2014-00006551 Texas	Henderson
42.213.15400.0507		Charles Ashburn Davis	O'Brien Resources, LLC	2014-00006545 Texas	Henderson
42.213.15400.0508	4/23/2014	Veronica Lois Orr	O'Brien Resources, LLC	2014-00006554 Texas	Henderson
42.213.15400.0510	5/20/2014	Travis Vernon Batson, Jr.	O'Brien Resources, LLC	2014-00007929 Texas	Henderson
42.213.15400.0511	5/20/2014	Jack P. Rahm, Trustee of the Jack P. Rahm Revocable Trust	O'Brien Resources, LLC	2014-00013015 Texas	Henderson
42.213.15400.0512	5/21/2014	David R. Soulsby	O'Brien Resources, LLC	2014-00011626 Texas	Henderson
42.213.15400.0513	5/29/2014	Glenn Holley	O'Brien Resources, LLC	2014-00007928 Texas	Henderson
42.213.15400.0514	3/12/2014	Ed H. Clark	O'Brien Resources, LLC	2014-00006549 Texas	Henderson
42.213.15400.0515	5/27/2014	Carolyn Holley Knepple	O'Brien Resources, LLC	2014-00007931 Texas	Henderson
42.213.15400.0516		Ronald L. Holley	O'Brien Resources, LLC	2014-00009458 Texas	Henderson
42.213.15400.0517		Michael W. Holley (Deceased)	O'Brien Resources, LLC	2014-00009457 Texas	Henderson
42.213.15400.0518		Lisa Wiklund	O'Brien Resources, LLC	2014-00007932 Texas	Henderson
42.213.15400.0519 42.213.15400.0520		Leslie R. Landry, Jr. John C. Vance	O'Brien Resources, LLC O'Brien Resources, LLC	2014-00007927 Texas 2014-00007930 Texas	Henderson Henderson
42.213.15400.0521		Cary S. "Pete" Cowart	O'Brien Resources, LLC	2014-00007930 Texas	Henderson
42.213.15400.0523		Vicki Jo Chappell Morris	O'Brien Resources, LLC	2014-00007921 Texas	Henderson
42.213.15400.0524		Billy Joe Davis	O'Brien Resources, LLC	2014-00007937 Texas	Henderson
42.213.15400.0525	4/17/2014	Billy Joe Lambright	O'Brien Resources, LLC	2014-00007936 Texas	Henderson
42.213.15400.0526	4/26/2014	Gerthe Ann Tapin	O'Brien Resources, LLC	2014-00007934 Texas	Henderson
42.213.15400.0527	4/26/2014	Hazel Louise Koslowski	O'Brien Resources, LLC	2014-00007935 Texas	Henderson
42.213.15400.0529	4/23/2014	Jeremy L. Vance	O'Brien Resources, LLC	2014-00006550 Texas	Henderson
42.213.15400.0530	4/17/2014	Cordell Vance	O'Brien Resources, LLC	2014-00006552 Texas	Henderson
42.213.15400.0531		Debra V. Reece	O'Brien Resources, LLC	2014-00006555 Texas	Henderson
42.213.15400.0532		Betty Jo Guffey	O'Brien Resources, LLC	2014-00006546 Texas	Henderson
42.213.15400.0533		Annie Marie Grimes	O'Brien Resources, LLC	2014-00006547 Texas	Henderson
42.213.15400.0534		Helen Splawn (Deceased)	O'Brien Resources, LLC	2014-00009459 Texas	Henderson
42.213.15400.0535		Violet Landis Holley (deceased)	O'Brien Resources, LLC O'Brien Resources, LLC	2014-00009456 Texas	Henderson
42.213.15400.0539 42.213.15400.0540		Mary Ann Lambright Jerry Wayne Davis	O'Brien Resources, LLC	2014-00009446 Texas 2014-00009442 Texas	Henderson Henderson
42.213.15400.0541		Joan Marie Batson Bandsuch	O'Brien Resources, LLC	2014-00009442 Texas	Henderson
42.213.15400.0542		James T. Palmer	O'Brien Resources, LLC	2014-00009444 Texas	Henderson
42.213.15400.0543		Bobby Lee Davis	O'Brien Resources, LLC	2014-00009441 Texas	Henderson
42.213.15400.0544		Rentier, Inc.	O'Brien Resources, LLC	2014-00011625 Texas	Henderson
42.213.15400.0545	4/25/2014	Josh Vance	O'Brien Resources, LLC	2014-00009448 Texas	Henderson
42.213.15400.0546	6/10/2014	Joan Elizabeth Sitzlow	O'Brien Resources, LLC	2014-00009451 Texas	Henderson
42.213.15400.0547	6/10/2014	Tommy Gene Mott	O'Brien Resources, LLC	2014-00009452 Texas	Henderson
42.213.15400.0548	6/10/2014	Johanna Mott Thomason	O'Brien Resources, LLC	2014-00009450 Texas	Henderson
42.213.15400.0549	5/29/2014	Shirley B. Hilton	O'Brien Resources, LLC	2014-00009449 Texas	Henderson
42.213.15400.0550		Jordan L. Vance	O'Brien Resources, LLC	2014-00009447 Texas	Henderson
42.213.15400.0552		Susan A. Soulsby	O'Brien Resources, LLC	2014-00009438 Texas	Henderson
42.213.15400.0553		Margaret Ann Hannie	O'Brien Resources, LLC	2014-00009440 Texas	Henderson
42.213.15400.0554		VWP JR, Inc. (Victor W. Pryor, Jr.)	O'Brien Resources, LLC	2014-00007925 Texas	Henderson
42.213.15400.0555		Forrest Carl Till	O'Brien Resources, LLC	2014-00011618 Texas	Henderson
42.213.15400.0556		Sherill Lee Bruce	O'Brien Resources, LLC	2014-00011624 Texas	Henderson
42.213.15400.0557 42.213.15400.0558		Howard Donald Lambright Violet Landis Holley (deceased)	O'Brien Resources, LLC O'Brien Resources, LLC	2014-00011631 Texas 2014-00011632 Texas	Henderson Henderson
42.213.15400.0559		Mila Lambright Baker	O'Brien Resources, LLC	2014-00011633 Texas	Henderson
	5,2014	·			

Lease #	Date	Lessor	Lessee	Book-Page State	County
42.213.15400.0560	7/10/2014	Ronald L. Holley	O'Brien Resources, LLC	2014-00011639 Texas	Henderson
42.213.15400.0561	7/10/2014	Michael W. Holley (Deceased)	O'Brien Resources, LLC	2014-00011634 Texas	Henderson
42.213.15400.0562	7/10/2014	Keetah Holley	O'Brien Resources, LLC	2014-00011635 Texas	Henderson
42.213.15400.0563	7/10/2014	Bradley Scott Lambright	O'Brien Resources, LLC	2014-00011637 Texas	Henderson
42.213.15400.0564	7/10/2014	Dana Mitchell	O'Brien Resources, LLC	2014-00011636 Texas	Henderson
42.213.15400.0567	7/18/2014	Linda Sue Douglas	O'Brien Resources, LLC	2014-00011638 Texas	Henderson
42.213.15400.0568	4/18/2014	Katrina Lynn Pound	O'Brien Resources, LLC	2014-00011621 Texas	Henderson
42.213.15400.0569	7/28/2014	Chelsea Willis Caver	O'Brien Resources, LLC	2014-00011620 Texas	Henderson
42.213.15400.0570	7/14/2014	Katherine Stanley	O'Brien Resources, LLC	2014-00011640 Texas	Henderson
42.213.15400.0571	8/19/2014	Wynne West Anderson	O'Brien Resources, LLC	2014-00014706 Texas	Henderson
42.213.15400.0572	8/19/2014	Shawn West Webb	O'Brien Resources, LLC	2014-00014705 Texas	Henderson
42.213.15400.0573	8/19/2014	The Catherine Coleman Murchison Exmpt Tr by George Bedell, Trustee	O'Brien Resources, LLC	2014-00014707 Texas	Henderson
42.213.15400.0574	8/19/2014	The Mark V West Testamentary Tr c/o Texas Bank & Tr Company	O'Brien Resources, LLC	2014-00014709 Texas	Henderson
42.213.15400.0575	8/19/2014	Shelly West Bowman	O'Brien Resources, LLC	2014-00014710 Texas	Henderson
42.213.15400.0576	8/19/2014	Waverly West Burford aka Waverly West, aka Ann Waverly West; AIF Stephanie Shawn West, Deceased	O'Brien Resources, LLC	2014-00014708 Texas	Henderson
42.213.15400.0577	8/11/2014	Donald O. Bennett	O'Brien Resources, LLC	2434-733 Texas	Anderson
42.213.15400.0578	8/6/2014	Jan Gary and Ann Gary	O'Brien Resources, LLC	2014-00011622 Texas	Henderson
42.213.15400.0584	7/7/2014	Judy Ann Spindor	O'Brien Resources, LLC	2014-00013010 Texas	Henderson
42.213.15400.0585	8/20/2014	Jerry Ann Henderson	O'Brien Resources, LLC	2434-729 Texas	Anderson
42.213.15400.0586	8/20/2014	Molly Henderson Boyd	O'Brien Resources, LLC	2434-725 Texas	Anderson
42.213.15400.0587	8/20/2014	William D. Henderson	O'Brien Resources, LLC	2434-721 Texas	Anderson
42.213.15400.0588	7/23/2014	Irvin H. Lambright	O'Brien Resources, LLC	2014-00013009 Texas	Henderson
42.213.15400.0589	8/20/2014	J. E. & L. E. Mabee Foundation by Raymond L. Tullius Jr. Vice Chairman	O'Brien Resources, LLC	2434-734 Texas	Anderson
42.213.15400.0590	9/3/2014		O'Brien Resources, LLC	2014-00013015 Texas	Henderson
42.213.15400.0591	8/22/2014	Laurie Hart	O'Brien Resources, LLC	2014-00013018 Texas	Henderson
42.213.15400.0592	9/4/2014	Rentier, Inc., Shirley Pogue, President	O'Brien Resources, LLC	2014-00013016 Texas	Henderson
42.213.15400.0593	9/3/2014	VWP, Jr., Inc. by Victor W. Pryor Jr., Trustee	O'Brien Resources, LLC	2014-00013017 Texas	Henderson
42.213.15400.0594	9/3/2014	Leigh L. Williams	O'Brien Resources, LLC	2439-387 Texas	Anderson
42.213.15400.0595	8/19/2014	Wynne West Anderson	O'Brien Resources, LLC	2439-457 Texas	Anderson
42.213.15400.0596		Waverly West, Deceased	O'Brien Resources, LLC	2439-459 Texas	Anderson
42.213.15400.0597	8/19/2014	The Catherine Coleman Murchison Ex Tr	O'Brien Resources, LLC	2439-458 Texas	Anderson
42.213.15400.0598	8/19/2014	The Mark V West Testamentary Tr c/o Texas Bank & Tr Company	O'Brien Resources, LLC	2439-460 Texas	Anderson
42.213.15400.0599	8/19/2014	Shelly West Bowman	O'Brien Resources, LLC	2439-461 Texas	Anderson
42.213.15400.0600		Larry C. Henderson	O'Brien Resources, LLC	2434-717 Texas	Anderson
42.213.15400.0601	8/19/2014	Stephanie Shawn West Webb, AKA known as Shawn West Webb	O'Brien Resources, LLC	2439-456 Texas	Anderson
42.213.15400.0602	9/16/2014	Teena Mitchell Whitsell	O'Brien Resources, LLC	2014-00013019 Texas	Henderson
42.213.15400.0603	9/3/2014	Maggie Nell Youngblood	O'Brien Resources, LLC	2434-738 Texas	Anderson
42.213.15400.0608	9/18/2014	Allan Goldston King Residuary Tr, David Russell King Res Tr, Duane Hadley King Res Tr, Gwendolyn Carlisle King Res Tr & Susan	O'Brien Resources, LLC	2014-00015293 Texas	Henderson
42.213.15400.0609		Horace Reagon Webb (O'Benco IV)	O'Brien Resources, LLC	2014-00014701 Texas	Henderson
42.213.15400.0610		Ronald C. Pate	O'Brien Resources, LLC	2014-00014703 Texas	Henderson
42.213.15400.0611	9/3/2014	Sharon L. Pugh	O'Brien Resources, LLC	2439-453 Texas	Anderson
42.213.15400.0612	9/24/2014	Dorothy Anne Landis	O'Brien Resources, LLC	2014-00014702 Texas	Henderson
42.213.15400.0613	9/25/2014	Randy Lambright	O'Brien Resources, LLC	2014-00014711 Texas	Henderson
42.213.15400.0614		J. W. Lambright	O'Brien Resources, LLC	2014-00014713 Texas	Henderson
42.213.15400.0615		Nell Pharris Burnett	O'Brien Resources, LLC	2014-00014712 Texas	Henderson
42.213.15400.0617	8/11/2014	David R. Soulsby	O'Brien Resources, LLC	2014-00014704 Texas	Henderson
42.213.15400.0619	7/10/2014	Helen Splawn (Deceased)	O'Brien Resources, LLC	2014-00014717 Texas	Henderson
42.213.15400.0620		Homer Pate	O'Brien Resources, LLC	2014-00014716 Texas	Henderson
42.213.15400.0621		Keith Lee Pate	O'Brien Resources, LLC	2014-00014719 Texas	Henderson
42.213.15400.0622		Lenell Lambright	O'Brien Resources, LLC	2014-00014714 Texas	Henderson
42.213.15400.0624		Susan A. Soulsby	O'Brien Resources, LLC	2014-00014721 Texas	Henderson
42.213.15400.0625		Mary Louise Moore Wilbanks	O'Brien Resources, LLC	2014-00014718 Texas	Henderson
42.213.15400.0626		RVS Minerals, LLC , Margaret McDaniel, Manager	O'Brien Resources, LLC	2014-00014710 Texas	Henderson
42.213.15400.0627		J.T. & D.A. Robertson Living Tr	O'Brien Resources, LLC	2445-765 Texas	Anderson
42.213.15400.0631		Harvil Lester Lambright	O'Brien Resources, LLC	2014-00014726 Texas	Henderson
42.213.15400.0632		Anita Lee Robinson	O'Brien Resources, LLC	2014-00014722 Texas	Henderson
42.213.15400.0633		M.M.M. & S., Ltd.	O'Brien Resources, LLC	2439-462 Texas	Anderson
42.213.15400.0635		Janet E. Woodman	O'Brien Resources, LLC	2014-00014728 Texas	Henderson
42.213.15400.0636		Tommie Michael Lambright	O'Brien Resources, LLC	2014-00014726 Texas 2014-00014730 Texas	Henderson
42.213.15400.0638		Walter R. Neill, Jr.	O'Brien Resources, LLC	2015-0000799 Texas	Henderson
42.213.15400.0638		Waller K. Neill, Jr. Karleen Neill Raper	O'Brien Resources, LLC	2015-00000799 Texas 2015-00000803 Texas	Henderson
42.213.15400.0639			O'Brien Resources, LLC O'Brien Resources, LLC	2015-00000803 Texas 2014-00014732 Texas	
		Kathy Link	O'Brien Resources, LLC O'Brien Resources, LLC		Henderson
42.213.15400.0641		Debby L. Henderson		2014-00014733 Texas	Henderson
42.213.15400.0644		Jerry Nell Mitchell Wilbanks	O'Brien Resources, LLC	2014-00015305 Texas	Henderson
42.213.15400.0645	10/14/2014	Wayne Mitchell	O'Brien Resources, LLC	2014-00015306 Texas	Henderson

Lease #	Date Lessor	Lessee	Book-Page State	County
42.213.15400.0646	10/22/2014 Bettie Luttrall	O'Brien Resources, LLC	2014-00015307 Texas	Henderson
42.213.15400.0647	10/22/2014 Doris Jean Peevy George	O'Brien Resources, LLC	2014-00015309 Texas	Henderson
42.213.15400.0648	10/22/2014 Harlon Lambright	O'Brien Resources, LLC	2014-00015308 Texas	Henderson
42.213.15400.0650	9/16/2014 Ricky Campbell	O'Brien Resources, LLC	2014-00015301 Texas	Henderson
42.213.15400.0651	9/16/2014 Rayeanne Owen	O'Brien Resources, LLC	2014-00015297 Texas	Henderson
42.213.15400.0652	9/16/2014 Mark Green and Renee Green	O'Brien Resources, LLC	2015-00015298 Texas	Henderson
42.213.15400.0653	9/29/2014 Judith Ann Grimes	O'Brien Resources, LLC	2014-00015299 Texas	Henderson
42.213.15400.0655	9/16/2014 Diane Bennett	O'Brien Resources, LLC	2014-00015300 Texas	Henderson
42.213.15400.0656	8/28/2014 Brenda Lois McGee	O'Brien Resources, LLC	2014-00015296 Texas	Henderson
42.213.15400.0658	8/28/2014 Charlcie Reynolds	O'Brien Resources, LLC	2014-00015295 Texas	Henderson
42.213.15400.0659	10/21/2014 J. T. Peevy	O'Brien Resources, LLC	2014-00015303 Texas	Henderson
42.213.15400.0660	10/30/2014 E. Wayne Greer	O'Brien Resources, LLC	2015-00000199 Texas	Henderson
42.213.15400.0661	11/5/2014 E. B. LaRue, Jr.	O'Brien Resources, LLC	2015-00000218 Texas	Henderson
42.213.15400.0662	10/8/2014 Debbie R. Jenkins	O'Brien Resources, LLC	2015-00000217 Texas	Henderson
42.213.15400.0663	10/2/2014 Elizabeth W. Neill	O'Brien Resources, LLC	2014-00015304 Texas	Henderson
42.213.15400.0664	11/4/2014 Storey Brothers Investments, LLC by Harry Storey as President	O'Brien Resources, LLC	2442-771 Texas	Anderson
42.213.15400.0665	10/23/2014 Margaret Stevens Partners, Ltd. by Margaret Stevens Partner	O'Brien Resources, LLC	2015-00000200 Texas	Henderson
42.213.15400.0666	10/30/2014 Grubb Holdings, L.P. #3, by Greg Grubb as Partner	O'Brien Resources, LLC	2015-00000198 Texas	Henderson
42.213.15400.0667	10/13/2014 Mary H. Hodge, aka Mary M. Hodge	O'Brien Resources, LLC	2015-00000197 Texas	Henderson
42.213.15400.0668	2/16/2015 David M. Pritchard and Judy Pritchard	O'Brien Resources, LLC	2015-00003846 Texas	Henderson
42.213.15400.0669	5/20/2014 Windsor Interests, Ltd.	O'Brien Resources, LLC	2014-00014700 Texas	Henderson
42.213.15400.0670	11/4/2014 Nell Oden Jackson	O'Brien Resources, LLC	2453-636 Texas	Anderson
42.213.15400.0671	9/24/2014 Kathryn Spring Wilkins	O'Brien Resources, LLC	2015-00000216 Texas	Henderson
42.213.15400.0672	11/4/2014 The Estate of Ray Doss Peeler, Jr., Deceased by Bryan Peeler, Ind Executor	O'Brien Resources, LLC	2453-634 Texas	Anderson
42.213.15400.0673	10/22/2014 Deana Tillman Crockett	O'Brien Resources, LLC	2015-00000213 Texas	Henderson
42.213.15400.0674	11/17/2014 W. C. Perryman III	O'Brien Resources, LLC	2015-00000215 Texas	Henderson
42.213.15400.0675	11/25/2014 E. B. Larue, Jr.	O'Brien Resources, LLC	2453-641 Texas	Anderson
42.213.15400.0676	11/25/2014 E. B. Larue, Jr.	O'Brien Resources, LLC	2015-00000206 Texas	Henderson
42.213.15400.0677	11/17/2014 Kristin Larue Perryman	O'Brien Resources, LLC	2015-00000214 Texas	Henderson
42.213.15400.0678	12/1/2014 Richard Sanders, as 2014A-0869 Receiver for Lambright Heirs et al	O'Brien Resources, LLC	2014-00016169 Texas	Henderson
42.213.15400.0679	11/21/2014 R. Kirk Cobb c/o Robbins, Tapp Cobb & Assoc PLLC	O'Brien Resources, LLC	2453-0642 Texas	Anderson
42.213.15400.0680	10/27/2014 Gretchen Perryman Lahourcade	O'Brien Resources, LLC	2015-00000204 Texas	Henderson
42.213.15400.0681	11/17/2014 Gretchen Perryman Lahourcade	O'Brien Resources, LLC	2015-00000212 Texas	Henderson
42.213.15400.0682	11/17/2014 LaGrass Property Holdings by John W. Roberts	O'Brien Resources, LLC	2015-00000203 Texas	Henderson
42.213.15400.0683	11/4/2014 John Charles Storey	O'Brien Resources, LLC	2453-639 Texas	Anderson
42.213.15400.0685	11/16/2014 John Lingo Schoellkopf, Alan C. Schoellkopf, Wilson Schoellkopf, Jr.	O'Brien Resources, LLC	2015-00010670 Texas	Henderson
42.213.15400.0686	12/9/2014 Mark H. Cohen	O'Brien Resources, LLC	2015-00000201 Texas	Henderson
42.213.15400.0687	12/8/2014 Keetah Holley, widow of Roger Dale Holley	O'Brien Resources, LLC	2015-00000205 Texas	Henderson
42.213.15400.0688	10/8/2014 Leland D. Lambright	O'Brien Resources, LLC	2015-00000211 Texas	Henderson
42.213.15400.0689	11/4/2014 Frances Elise Storey Lincoln	O'Brien Resources, LLC	2453-638 Texas	Anderson
42.213.15400.0690	12/12/2014 Robert Avon Holley	O'Brien Resources, LLC	2015-00001741 Texas	Henderson
42.213.15400.0691	12/8/2014 Mary Louise Moore Wilbanks	O'Brien Resources, LLC	2014-00001742 Texas	Henderson
42.213.15400.0692	12/8/2014 Debra K. Vinyard	O'Brien Resources, LLC	2015-00001744 Texas	Henderson
42.213.15400.0693	12/8/2014 Vickie E. Hewitt	O'Brien Resources, LLC	2015-00001743 Texas	Henderson
42.213.15400.0694	11/4/2014 Robert Gerald Storey III	O'Brien Resources, LLC	2453-632 Texas	Anderson
42.213.15400.0695	12/9/2014 Timmie J. Coleman	O'Brien Resources, LLC	2015-00001745 Texas	Henderson
42.213.15400.0696	1/26/2015 Henderson-Elmcrest Trust by Lauren Butler, Trustee	O'Brien Resources, LLC	2015-00004946 Texas	Henderson
42.213.15400.0698	8/19/2014 Shawn West Webb	O'Brien Resources, LLC	2015-00000815 Texas	Henderson
42.213.15400.0699	8/19/2014 Waverly West Burford aka Waverly West, aka Ann Waverly West; AIF Stephanie Shawn West. Deceased	O'Brien Resources, LLC	2015-00000813 Texas	Henderson
42.213.15400.0700	8/19/2014 Wynne West Anderson	O'Brien Resources, LLC	2015-00000811 Texas	Henderson
42.213.15400.0701	8/19/2014 The Catherine Coleman Murchison Ex Tr, George Bedell Trustee	O'Brien Resources, LLC	2015-00000820 Texas	Henderson
42.213.15400.0702	8/19/2014 The Mark V West Testamentary Tr c/o Texas Bank & Tr Company	O'Brien Resources, LLC	2015-00000819 Texas	Henderson
42.213.15400.0703	8/19/2014 Shelly West Bowman	O'Brien Resources, LLC	2015-00000817 Texas	Henderson
42.213.15400.0704	2/6/2015 Tommy Lynn Jernigan	O'Brien Resources, LLC	2015-00003843 Texas	Henderson
42.213.15400.0705	1/30/2015 Karla Nelson	O'Brien Resources, LLC	2453-0644 Texas	Anderson
42.213.15400.0706	1/30/2015 Karen Belk	O'Brien Resources, LLC	2453-0647 Texas	Anderson
42.213.15400.0709	3/18/2015 Chad Arlen Gilbert	O'Brien Resources, LLC	2015-00004942 Texas	Henderson
42.213.15400.0710	4/1/2015 Eddith G. Reynolds Trust	O'Brien Resources, LLC	2015-00006851 Texas	Henderson
42.213.15400.0711	3/25/2015 Carroll Springs United Meth Church c/o Wayne Mitchell	O'Brien Resources, LLC	2015-00006849 Texas	Henderson
42.213.15400.0712	2/16/2015 Neill Oil, LLC by Virginia B. Price, Mgr	O'Brien Resources, LLC	2015-00004941 Texas	Henderson
42.213.15400.0713	3/1/2012 Terry Glenn Laymance	CenTex Oil & Gas, Inc.	2012-00008729 Texas	Henderson
42.213.15400.0715	3/31/2015 Owen Oil & Gas Properties, LLC	O'Brien Resources, LLC	2015-00006847 Texas	Henderson
42.213.15400.0728	4/7/2015 Richard Allen Lewis	O'Brien Resources, LLC	2015-00006855 Texas	Henderson
42.213.15400.0729	4/2/2015 Karleen Neill Raper	O'Brien Resources, LLC	2015-00006856 Texas	Henderson
42.213.15400.0729	4/14/2015 Neill Oil, LLC by Virginia B. Price, Manager	O'Brien Resources, LLC	2015-00006862 Texas	Henderson
.2.210.10400.0133		O Blieff Resources, ELC	ZUTU-UUUUUUZ TEXAS	i idiludi SUN

	-		1	David David	0
Lease #	Date	Lessor	Lessee	Book-Page State	County
42.213.15400.0734			O'Brien Resources, LLC	2015-00006861 Texas	Henderson
42.213.15400.0735		Patricia Ann Bragg	O'Brien Resources, LLC O'Brien Resources, LLC	2015-00000827 Texas	Henderson
42.213.15400.0736		The R. E. and P. D. Dwelle Living Trust	O'Brien Resources, LLC O'Brien Resources, LLC	2015-00000809 Texas 2015-00000808 Texas	Henderson
42.213.15400.0737 42.213.15400.0738		Betty Sue Dwelle Waters	O'Brien Resources, LLC		Henderson
42.213.15400.0739		Leona Earl May	O'Brien Resources, LLC	2015-00000828 Texas 2015-00000791 Texas	Henderson
42.213.15400.0739	9/11/2014	Jeanne Dwelle Waters by Lindsley Waters III, AIF Robert Clifford Louder	O'Brien Resources, LLC		Henderson Anderson
42.213.15400.0740		Leonard Ray Louder	O'Brien Resources, LLC	2466-482 Texas 2474-540 Texas	Anderson
42.213.15400.0741		James Strowd	O'Brien Resources, LLC	2015-00006865 Texas	Henderson
42.213.15400.0746		George Couch	O'Brien Resources, LLC	2015-00006864 Texas	Henderson
42.213.15400.0755		Unknown Heirs of W. H. Baker by Richard Sanders Receiver	O'Brien Resources, LLC	2015-00005482 Texas	Henderson
42.213.15400.0756		Unknown Heirs of Ola Mae Starkey Coleman et al by Richard Sanders Receiver CV 15-0033-173	O'Brien Resources, LLC	2015-00005483 Texas	Henderson
42.213.15400.0758		Richard Louder	O'Brien Resources, LLC	2466-473 Texas	Anderson
42.213.15400.0759		Linda Kay Morrison	O'Brien Resources, LLC	2466-476 Texas	Anderson
42.213.15400.0760		Walter R. Neill	O'Brien Resources, LLC	2015-00006839 Texas	Henderson
42.213.15400.0762		Ruby Elaine Everett	O'Brien Resources, LLC	2015-00009916 Texas	Henderson
42.213.15400.0768		David Haywood	O'Brien Resources, LLC	2467-735 Texas	Anderson
42.213.15400.0769		Karen Kale	O'Brien Resources, LLC	2474-532 Texas	Anderson
42.213.15400.0788		Elizabeth W. Neill	O'Brien Resources, LLC	2015-00009914 Texas	Henderson
42.213.15400.0789	5/20/2015	Sheila K. Marshall	O'Brien Resources, LLC	2474-580 Texas	Anderson
42.213.15400.0790		Natalia A. Lamas	O'Brien Resources, LLC	2474-576 Texas	Anderson
42.213.15400.0792		Trudy Fay Hawkins	O'Brien Resources, LLC	2015-00009904 Texas	Henderson
42.213.15400.0794		Freda Sue Gallatin	O'Brien Resources, LLC	2474-544 Texas	Anderson
42.213.15400.0795		Troy V. Smith	O'Brien Resources, LLC	2015-00009906 Texas	Henderson
42.213.15400.0797		Katherine Wynne, Executrix Buck J. Wynne III Estate	O'Brien Resources, LLC	2467-734 Texas	Anderson
42.213.15400.0801		William Luther Kale by AIF Elena Kale Glasscock	O'Brien Resources, LLC	2474-520 Texas	Anderson
42.213.15400.0802	5/15/2015	Kevin Lee Everett	O'Brien Resources, LLC	2015-00009900 Texas	Henderson
42.213.15400.0803		Levi Clakley	O'Brien Resources, LLC	2015-00009888 Texas	Henderson
42.213.15400.0805		Cova E. Galloway Smither by Murray Smither as AIF	O'Brien Resources, LLC	2474-549 Texas	Anderson
42.213.15400.0806		Carolyn Goldston	O'Brien Resources, LLC	2015-00009889 Texas	Henderson
42.213.15400.0807	5/28/2015	Elena Kale Glasscock	O'Brien Resources, LLC	2474-516 Texas	Anderson
42.213.15400.0808	6/29/2015	Don S. Newby, Indiv. & as Ind. Executor of the Estate of Floye M. Newby, deceased	O'Brien Resources, LLC	2474-552 Texas	Anderson
42.213.15400.0809	5/21/2015	Goldston Oil Corporation as Nominee for Susan Mayfield, 1997, Marital Tr; Jack H. Mayfield, III; Walter G. Mayfield; Walter G.	O'Brien Resources, LLC	2015-00014556 Texas	Henderson
42.213.15400.0811	5/28/2015	Southwest Mineral Group, Ltd.	O'Brien Resources, LLC	2015-00009909 Texas	Henderson
42.213.15400.0812	5/20/2015	Dolores A. Hayter	O'Brien Resources, LLC	2474-572 Texas	Anderson
42.213.15400.0813	5/22/2015	Teresa Galloway Brown	O'Brien Resources, LLC	2474-569 Texas	Anderson
42.213.15400.0815	5/19/2015	Mary E. Grisso Trust No. 1 by Harry J. Schafer III, Trustee	O'Brien Resources, LLC	2015-00012654 Texas	Henderson
42.213.15400.0816	3/18/2015	William Adcock	O'Brien Resources, LLC	2015-00009907 Texas	Henderson
42.213.15400.0817	5/20/2015	Beth C. Burton	O'Brien Resources, LLC	2474-527 Texas	Anderson
42.213.15400.0818	7/15/2015	Bob J. Chronister aka Bobby Chronister	O'Brien Resources, LLC	2484-258 Texas	Anderson
42.213.15400.0819	6/29/2015	Rick V. Newby	O'Brien Resources, LLC	2484-252 Texas	Anderson
42.213.15400.0820	7/15/2015	Coy Lee Kale, Jr.	O'Brien Resources, LLC	2484-0262 Texas	Anderson
42.213.15400.0821	6/29/2015	Michael D. Kale	O'Brien Resources, LLC	2486-122 Texas	Anderson
42.213.15400.0822	7/15/2015	J. Todd Kale	O'Brien Resources, LLC	2484-260 Texas	Anderson
42.213.15400.0829	5/20/2015	Mary Kathryn Kale Bolt	O'Brien Resources, LLC	2474-536 Texas	Anderson
42.213.15400.0830	7/23/2015	Edwin Cox Jr. Trust No. 12 by William L. Graham Trustee	O'Brien Resources, LLC	2015-00013129 Texas	Henderson
42.213.15400.0832	7/21/2015	Mary Kathryn Johnson	O'Brien Resources, LLC	2484-264 Texas	Anderson
42.213.15400.0835	7/20/2015	Kimberly Sinclair	O'Brien Resources, LLC	2015-00013093 Texas	Henderson
42.213.15400.0836	7/17/2015	Roger Dale Cross	O'Brien Resources, LLC	2484-272 Texas	Anderson
42.213.15400.0837	5/20/2015	Terrell Lee Selcer	O'Brien Resources, LLC	2484-267 Texas	Anderson
42.213.15400.0839	7/21/2015	Ashley Katherine Radford	O'Brien Resources, LLC	2484-270 Texas	Anderson
42.213.15400.0847	5/15/2015	Aaron Lee Everett	O'Brien Resources, LLC	2015-00013095 Texas	Henderson
42.213.15400.0849		Pine Ridge Ranch, LLC; William Barrett Travis Director	O'Brien Resources, LLC	2015-00013097 Texas	Henderson
42.213.15400.0850	8/5/2015	Mike Woodchek	O'Brien Resources, LLC	2015-00013094 Texas	Henderson
42.213.15400.0852		Charles R. Fagg	O'Brien Resources, LLC	2484-285 Texas	Anderson
42.213.15400.0855		Gold Struck Investments, L.P. Wells Fargo N.A. Agent	O'Brien Resources, LLC	2015-00013739 Texas	Henderson
42.213.15400.0856		Megan Goldston Family Trust IMA, Wells Fargo Bank N.A. Agent	O'Brien Resources, LLC	2015-00013738 Texas	Henderson
42.213.15400.0857		Bill Goldston Family Trust IMA, Wells Fargo Bank, N.A., Agent	O'Brien Resources, LLC	2015-00013737 Texas	Henderson
42.213.15400.0858		Bradley Lee Clakley	O'Brien Resources, LLC	2015-00013741 Texas	Henderson
42.213.15400.0859		Dempsey Lee Clakley	O'Brien Resources, LLC	2015-00013100 Texas	Henderson
42.213.15400.0860	8/27/2015	John W. Murchison Oil & Gas, Ltd.	O'Brien Resources, LLC	2015-00015105 Texas	Henderson
42.213.15400.0861		John W. Murchison Oil & Gas, Ltd.	O'Brien Resources, LLC	2015-00015101 Texas	Henderson
42.213.15400.0862		John W. Murchison Oil & Gas, Ltd.	O'Brien Resources, LLC	2015-00015100 Texas	Henderson
42.213.15400.0863		John W. Murchison Oil & Gas, Ltd.	O'Brien Resources, LLC	2015-00015102 Texas	Henderson
42.213.15400.0864	8/27/2015	John W. Murchison Oil & Gas, Ltd.	O'Brien Resources, LLC	2015-00015106 Texas	Henderson

Lease #	Date	1	Lessee	Book Bogo State	County
42.213.15400.0865	8/27/2015	Laura M. Woodali and Meredith Mallory, III	O'Brien Resources, LLC	Book-Page State 2015-00015109 Texas	County
42.213.15400.0866	8/27/2015		O'Brien Resources, LLC	2015-00015108 Texas	Henderson
42.213.15400.0867		John W. Murchison Oil & Gas, Ltd.	O'Brien Resources, LLC	2015-00015107 Texas	Henderson
42.213.15400.0890	8/27/2015	MI4 Minerals, Ltd	O'Brien Resources, LLC	2015-00015117 Texas	Henderson
42.213.15400.0891	8/27/2015	MI4 Minerals, Ltd	O'Brien Resources, LLC	2015-00015112 Texas	Henderson
42.213.15400.0892	8/27/2015	MI4 Minerals, Ltd	O'Brien Resources, LLC	2015-00015110 Texas	Henderson
42.213.15400.0893	8/27/2015	MI4 Minerals, Ltd	O'Brien Resources, LLC	2015-00015111 Texas	Henderson
42.213.15400.0894	8/27/2015	MI4 Minerals, Ltd	O'Brien Resources, LLC	2015-00015118 Texas	Henderson
42.213.15400.0895	8/27/2015	MI4 Minerals, Ltd	O'Brien Resources, LLC	2015-00015119 Texas	Henderson
42.213.15400.0904	9/11/2015	Catherine Trojacek Mohr	O'Brien Resources, LLC	2486-129 Texas	Anderson
42.213.15400.0905	9/22/2015	Kenneth Earl Watkins	O'Brien Resources, LLC	2015-00013742 Texas	Henderson
42.213.15400.0906		Theresa Elizabeth Trojacek	O'Brien Resources, LLC	2486-128 Texas	Anderson
42.213.15400.0909		PEC Minerals, LP	O'Brien Resources, LLC	2015-00015049 Texas	Henderson
42.213.15400.0910		The Estate of Kathleen Mincey by Melinda Hartnett as Successor Dependent Administrator	O'Brien Resources, LLC	2484-282 Texas	Anderson
42.213.15400.0911		J&A Acquisitions, Inc. by James T. McCain III, Mgr	O'Brien Resources, LLC	2491-239 Texas	Anderson
42.213.15400.0912		LDF-ETX, Inc. by: Lucas D. Freeman, Mgr	O'Brien Resources, LLC	2491-246 Texas	Anderson
42.213.15400.0913 42.213.15400.0914		Michael Ralph Jenkins, Jr. Arroyo Land & Minerals, LLC by: Bailey Fender, Mgr	O'Brien Resources, LLC O'Brien Resources, LLC	2491-248 Texas 2491-257 Texas	Anderson Anderson
42.213.15400.0915		Envision Land CO., LLC by: Charles C. McCain, Mgr.	O'Brien Resources, LLC	2491-0250 Texas	Anderson
42.213.15400.0916	9/3/2015		O'Brien Resources, LLC	2015-00015666 Texas	Henderson
42.213.15400.0917	9/3/2015	•	O'Brien Resources. LLC	2015-00015667 Texas	Henderson
42.213.15400.0918	2/9/2015	Richard Sanders, Receiver Cause #2014B-1181 Heirs of Charles Mills etal	O'Brien Resources, LLC	2015-00001905 Texas	Henderson
42.213.15400.0919	9/25/2014	Ernest C. Bryant III	O'Brien Resources, LLC	2015-00003844 Texas	Henderson
42.213.15400.0920	9/23/2015	PEC Minerals, LP	O'Brien Resources, LLC	2015-00015052 Texas	Henderson
42.213.15400.0921	9/23/2015	PEC Minerals, LP	O'Brien Resources, LLC	2015-00015048 Texas	Henderson
42.213.15400.0922	9/23/2015	PEC Minerals, LP	O'Brien Resources, LLC	2015-00015050 Texas	Henderson
42.213.15400.0923	3/23/2017	PEC Minerals, LP	O'Brien Resources, LLC	2015-00015051 Texas	Henderson
42.213.15400.0924	9/23/2015	PEC Minerals, LP	O'Brien Resources, LLC	2015-00015053 Texas	Henderson
42.213.15400.0935	8/15/2014	Jeffrey W. Eaton	O'Brien Resources, LLC	2015-00000842 Texas	Henderson
42.213.15400.0936	8/15/2014	Richard H. Eaton	O'Brien Resources, LLC	2015-00000834 Texas	Henderson
42.213.15400.0937	8/15/2014	Robert W. Percival	O'Brien Resources, LLC	2015-00000836 Texas	Henderson
42.213.15400.0938		Windsor Interests, Ltd.	O'Brien Resources, LLC	2389-616 Texas	Anderson
42.213.15400.0939		Charles Naubert	O'Brien Resources, LLC	2491-0410 Texas	Anderson
42.213.15400.0941		Christopher B. Naubert	O'Brien Resources, LLC	2491-0435 Texas	Anderson
42.213.15400.0946		Nancy Naubert	O'Brien Resources, LLC	2491-0414 Texas	Anderson
42.213.15400.0948 42.213.15400.0949		Cora Hanson Jill Peavy David	O'Brien Resources, LLC O'Brien Resources, LLC	2491-0406 Texas 2015-00000860 Texas	Anderson Henderson
42.213.15400.0949		Wesson E. Gaige	O'Brien Resources, LLC	2015-00000831 Texas	Henderson
42.213.15400.0955		John Naubert	O'Brien Resources, LLC	2491-0439 Texas	Anderson
42.213.15400.0968		5 Patricia Claire Lamb Individually and as Executrix for the Estate of Dolores A. Lamb	O'Brien Resources, LLC	2491-704 Texas	Anderson
42.213.15400.0972		5 Brenda Turbyfill	O'Brien Resources, LLC	2015-00015675 Texas	Henderson
42.213.15400.0977		5 Mary Priscilla Jenkins Meador	O'Brien Resources, LLC	2015-00015676 Texas	Henderson
42.213.15400.0990	11/12/2015	5 Peter Wesley Williford	O'Brien Resources, LLC	2499-561 Texas	Anderson
42.213.15400.0991	10/20/2015	5 Jerry Braddock, Jr.	O'Brien Resources, LLC	2015-00015677 Texas	Henderson
42.213.15400.0992	11/9/2015	Melba Ruth Davis	O'Brien Resources, LLC	2499-671 Texas	Anderson
42.213.15400.0993	11/4/2015	Jon Eric Scudder	O'Brien Resources, LLC	2499-603 Texas	Anderson
42.213.15400.0995	11/18/2015	5 Jimmy Robert Bearden	O'Brien Resources, LLC	2499-0716 Texas	Anderson
42.213.15400.0997	11/9/2015	Carl C. Tidgren	O'Brien Resources, LLC	2499-0720 Texas	Anderson
42.213.15400.0999	11/9/2015	William Edward Rushing	O'Brien Resources, LLC	2499-555 Texas	Anderson
42.213.15400.1003		Betty Nell Curtis*	O'Brien Resources, LLC	2499-552 Texas	Anderson
42.213.15400.1004		Harold Lewis Rushing*	O'Brien Resources, LLC	2499-558 Texas	Anderson
42.213.15400.1005		James L. Rushing aka J. L. Rushing	O'Brien Resources, LLC	2499-522 Texas	Anderson
42.213.15400.1007		Randy Jenkins	O'Brien Resources, LLC	2016-00000588 Texas	Henderson
42.213.15400.1012		Final Talley, Ltd	O'Brien Resources, LLC	2503-201 Texas	Anderson
42.213.15400.1019 42.213.15400.1020		5 Kerry T. Hurt Cheryl Blackwell-Rhea fka Cheryl Ann Jenkins Randy Jenkins	O'Brien Resources, LLC O'Brien Resources, LLC	2016-00000596 Texas 2016-00000590 Texas	Henderson Henderson
42.213.15400.1020 42.213.15400.1021		Stanley V. Prickett	O'Brien Resources, LLC O'Brien Resources, LLC	2016-00000590 Texas 2499-749 Texas	Anderson
42.213.15400.1021		5 Michael S. Barrett	O'Brien Resources, LLC	2499-697 Texas	Anderson
42.213.15400.1022		Julia Jenkins	O'Brien Resources, LLC	2016-00000600 Texas	Henderson
42.213.15400.1026		Billy Joe Miller	O'Brien Resources, LLC	2499-0578 Texas	Anderson
42.213.15400.1027		5 Genevieve White Deatherage	O'Brien Resources, LLC	2499-729 Texas	Andeson
42.213.15400.1028		5 Larry A. Kale	O'Brien Resources, LLC	2499-723 Texas	Anderson
42.213.15400.1033		Sarah Jenkins	O'Brien Resources, LLC	2016-00000591 Texas	Henderson
42.213.15400.1034		5 George E. Prickett	O'Brien Resources, LLC	2499-680 Texas	Anderson
42.213.15400.1039		5 Bobby Jim Stringer	O'Brien Resources, LLC	2499-732 Texas	Anderson

Lease #	Date	Lessor	Lessee	Book-Page State	County
42.213.15400.1040	11/19/2015	Dennis E. Marshall	O'Brien Resources, LLC	2499-735 Texas	Anderson
42.213.15400.1041	11/24/2015	Donna S. Ledbetter	O'Brien Resources, LLC	2499-726 Texas	Anderson
42.213.15400.1043	11/2/2015	Linda Ovino (Carson)	O'Brien Resources, LLC	2016-00000592 Texas	Henderson
42.213.15400.1045	11/2/2015	Julianna Bowling fka Julianna Jenkins	O'Brien Resources, LLC	2016-00000594 Texas	Henderson
42.213.15400.1048	11/24/2015	Norman Wayne Stringer	O'Brien Resources, LLC	2499-683 Texas	Anderson
42.213.15400.1049	12/3/2015	Ada Willene Ratliff	O'Brien Resources, LLC	2499-686 Texas	Anderson
42.213.15400.1050	11/19/2015	Larry Leon Everett	O'Brien Resources, LLC	2499-625 Texas	Anderson
42.213.15400.1051	11/19/2015	Larry Leon Everett	O'Brien Resources, LLC	2499-629 Texas	Anderson
42.213.15400.1056	12/8/2015	Mark H. Cohen	O'Brien Resources, LLC	2016-00001919 Texas	Henderson
42.213.15400.1061	11/27/2015	Stephen Dollar	O'Brien Resources, LLC	2016-00001916 Texas	Henderson
42.213.15400.1070	12/8/2015	Benjamin Aaron Cohen	O'Brien Resources, LLC	2016-00001917 Texas	Henderson
42.213.15400.1071	9/5/2015	Amanda Beth (Cohen)Godwin	O'Brien Resources, LLC	2016-00001918 Texas	Henderson
42.213.15400.1073	11/24/2015	Edwin J. Sheek, Jr.	O'Brien Resources, LLC	2503-330 Texas	Anderson
42.213.15400.1076	10/5/2015	Richard Naubert, Jr.	O'Brien Resources, LLC	2503-0279 Texas	Anderson
42.213.15400.1080	12/3/2015	Nellie Sue Webb	O'Brien Resources, LLC	2503-333 Texas	Anderson
42.213.15400.1084	11/9/2015	Charles E. Rushing	O'Brien Resources, LLC	2503-204 Texas	Anderson
42.213.15400.1091	12/3/2015	Glenda Winters Harkness	O'Brien Resources, LLC	2503-228 Texas	Anderson
42.213.15400.1094	11/30/2015	Michael George Watkins	O'Brien Resources, LLC	2016-00001922 Texas	Henderson
42.213.15400.1097	1/5/2016	Gary G. Holcomb, his separate property and Murray Holcomb, his separate property	O'Brien Resources, LLC	2506-240 Texas	Anderson
42.213.15400.1098	1/14/2016	Douglas Jay Crow	O'Brien Resources, LLC	2016-00003594 Texas	Henderson
42.213.15400.1100	12/3/2015	Dale Winters	O'Brien Resources, LLC	2503-257 Texas	Anderson
42.213.15400.1101	12/11/2015	Tiffany Lynn Morris	O'Brien Resources, LLC	2508-259 Texas	Anderson
42.213.15400.1102		Karon Winters Smith	O'Brien Resources, LLC	2503-263 Texas	Anderson
42.213.15400.1103		Bruce S. Kale, III	O'Brien Resources, LLC	2503-254 Texas	Anderson
42.213.15400.1104		Stephen T. Stafford	O'Brien Resources, LLC	2503-246 Texas	Anderson
42.213.15400.1105		Pamela Jane Shilling (Stafford)	O'Brien Resources, LLC	2503-236 Texas	Anderson
42.213.15400.1106		Pamela Jane Shilling (Stafford)	O'Brien Resources, LLC	2503-239 Texas	Anderson
42.213.15400.1107		Stephen T. Stafford	O'Brien Resources, LLC	2503-243 Texas	Anderson
42.213.15400.1108		Lester Fred Henry	O'Brien Resources, LLC	2508-270 Texas	Anderson
42.213.15400.1109		Mackilee Bennett Martin	O'Brien Resources, LLC	2508-265 Texas	Anderson
42.213.15400.1110		Sandra J. Pontes	O'Brien Resources, LLC O'Brien Resources, LLC	2016-00003596 Texas	Henderson
42.213.15400.1111 42.213.15400.1113		Stanley Max Everett-	O'Brien Resources, LLC	2016-00003595 Texas 2530-722 Texas	Henderson
42.213.15400.1113	1/22/2016	Albert Eugene Louder	O'Brien Resources, LLC	2530-722 Texas 2508-300 Texas	Anderson Anderson
42.213.15400.1115		Sidney Castleberry	O'Brien Resources, LLC	2515-780 Texas	Anderson
42.213.15400.1116		*Shirley Wade Thomas	O'Brien Resources, LLC	2508-295 Texas	Anderson
42.213.15400.1117		Myrna Beth Coffman	O'Brien Resources, LLC	2508-305 Texas	Anderson
42.213.15400.1120	2/1/2016	Melva Jo Shinpaugh FKA Melva Jo Henderson	O'Brien Resources, LLC	2508-291 Texas	Anderson
42.213.15400.1121		Susan G. Till	O'Brien Resources, LLC	2016-00003608 Texas	Henderson
42.213.15400.1124		Dorothy Holzworth	O'Brien Resources, LLC	2015-00000844 Texas	Henderson
42.213.15400.1125		Sandra Lee Davis	O'Brien Resources, LLC	2015-00000864 Texas	Henderson
42.213.15400.1126	7/15/2014	James Sweatt	O'Brien Resources, LLC	2015-00000850 Texas	Henderson
42.213.15400.1127	7/15/2014	Michael Lynn Easterwood	O'Brien Resources, LLC	2015-00000884 Texas	Henderson
42.213.15400.1128	7/17/2014	Robert Gray et u9thx	O'Brien Resources, LLC	2015-00000857 Texas	Henderson
42.213.15400.1129	7/15/2014	Donna Jean Colley	O'Brien Resources, LLC	2015-00000858 Texas	Henderson
42.213.15400.1130	7/24/2014	Norman Joseph Colley, Ind. & as Atty-in-Fact for Charlotte V. Bennett, FKA Charlotte V. Colley	O'Brien Resources, LLC	2015-00000874 Texas	Henderson
42.213.15400.1131	7/15/2015	Bettye June Chronister	O'Brien Resources, LLC	2508-310 Texas	Anderson
42.213.15400.1132	1/26/2016	Robert Coleman	O'Brien Resources, LLC	2016-00003613 Texas	Henderson
42.213.15400.1134	2/3/2016	Virginia Ruth Sabatini	O'Brien Resources, LLC	2514-777 Texas	Anderson
42.213.15400.1135	10/16/2015	King Oil & Gas of Texas, Ltd.	O'Brien Resources, LLC	2016-00003591 Texas	Henderson
42.213.15400.1136	2/3/2016	James Robert Brummett	O'Brien Resources, LLC	2508-318 Texas	Anderson
42.213.15400.1137	1/26/2016	Charlotte Arnold	O'Brien Resources, LLC	2016-00003611 Texas	Henderson
42.213.15400.1138	1/26/2016	Bettye Lelane Harvill Newberry	O'Brien Resources, LLC	2016-00003610 Texas	Henderson
42.213.15400.1139	2/1/2016	Troy M. Price	O'Brien Resources, LLC	2016-00005132 Texas	Henderson
42.213.15400.1140	2/3/2016	Sam Henderson	O'Brien Resources, LLC	2515-026 Texas	Anderson
42.213.15400.1141	2/3/2016	Roy Ratheal	O'Brien Resources, LLC	2515-0021 Texas	Anderson
42.213.15400.1142	2/2/2016	Amanda Beth (Cohen)Godwin	O'Brien Resources, LLC	2016-00005128 Texas	Henderson
42.213.15400.1143	2/3/2016	Rita Powell	O'Brien Resources, LLC	2514-782 Texas	Anderson
42.213.15400.1144	2/3/2016	Richard Basinger	O'Brien Resources, LLC	2515-0006 Texas	Anderson
42.213.15400.1145	2/9/2016	Mary Coleman Nestor	O'Brien Resources, LLC	2016-00005129 Texas	Henderson
42.213.15400.1146	2/3/2016	Alan B. Cawthon	O'Brien Resources, LLC	2515-031 Texas	Anderson
42.213.15400.1147	2/3/2016	Karen H. Lakey	O'Brien Resources, LLC	2515-011 Texas	Anderson
42.213.15400.1148	1/12/2016	Joyce A. Easley Pell	O'Brien Resources, LLC	2514-792 Texas	Anderson
42.213.15400.1150	2/3/2016	John Davidson Brummett	O'Brien Resources, LLC	2514-797 Texas	Anderson
42.213.15400.1151	5/23/2016	Barbara Tucker	O'Brien Resources, LLC	2522=675 Texas	Anderson

1999 1999	Lease #	Date	Lessor	Lessee	Book-Page State	County
1922-1923-1923-1923-1923-1923-1923-1923-	42.213.15400.1152	2/3/2016		O'Brien Resources, LLC	-	Anderson
23.5.5.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.	42.213.15400.1153	2/3/2016	Donald Martin Henderson	O'Brien Resources, LLC	2515-001 Texas	Anderson
1922 1922	42.213.15400.1154	11/12/2015	Robert D. Johnston, Receiver Cause No. DCCV15-177-87, Frances Kale Peterson & Unknown Heirs et al	O'Brien Resources, LLC	2492-0578 Texas	Anderson
	42.213.15400.1155	11/6/2015	Robert D. Johnston, Receiver Cause No. DCCV15-177-87, Frances Kale Peterson et al	O'Brien Resources, LLC	2492-0572 Texas	Anderson
\$2.00 10.0	42.213.15400.1156	11/6/2015	Robert D. Johnston Receiver for Cause NO. DCCV15-176-3 Unknown Heirs of Berniece H. Hughes Walker et al	O'Brien Resources, LLC	2502-0043 Texas	Anderson
1971 1971	42.213.15400.1162	12/11/2015	Kerby L. Kale	O'Brien Resources, LLC	2514-770 Texas	Anderson
1.25.00.00.00.00.00.00.00.00.00.00.00.00.00	42.213.15400.1163	1/20/2016	Don Wayne Kale	O'Brien Resources, LLC	2514-774 Texas	Anderson
Page	42.213.15400.1165	1/22/2016	Melissa J. Bond Barenthsen	O'Brien Resources, LLC	2515-036 Texas	Anderson
22.2.3.1.0.0.0.17.0 CORPTO Log College System France College Co	42.213.15400.1167	1/12/2016	Barbara Jean Wade Ratliff	O'Brien Resources, LLC	2515-723 Texas	Anderson
42.2.3.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.	42.213.15400.1168	2/23/2016	Malcolm Lee Watson Sr.	O'Brien Resources, LLC	2016-00003045 Texas	Henderson
US-21 St.00.01171 Office	42.213.15400.1170	1/26/2016	Jean Lynette Harvill	O'Brien Resources, LLC	2016-00006217 Texas	Henderson
\$2.00.000.000.000.000.000.000.000.000.00	42.213.15400.1172	1/22/2016	Joye M. Bond	O'Brien Resources, LLC	2016-00006268 Texas	Henderson
\$2,113.1.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.0.	42.213.15400.1173	1/26/2016	James Linwood Harvill, Jr.	O'Brien Resources, LLC	2016-00006220 Texas	Henderson
	42.213.15400.1174	2/3/2016	Carolyn Garvin Berry	O'Brien Resources, LLC	2016-00006265 Texas	Henderson
12.73.15.00.171 0.00.0000 1.00.0000 1.00.0000 1.00.000000 1.00.00000 1.00.00000 1.00.00000 1.00.00000 1.00.000000 1.00.000000 1.00.00000 1.00.000000 1.00.000000 1.00.000000 1.00.000000 1.00.000000 1.00.000000 1.00.000000 1.00.000000 1.00.000000 1.00.0000000 1.00.0000000000	42.213.15400.1175	2/3/2016	Barbara Bogart	O'Brien Resources, LLC	2515-750 Texas	Anderson
12.233 10.001 10.00000 10.0000 10.0000 10.0000 10.0000 10.0000 10.0000 10.0000 10.0000 10.0000 10.0000 10.0000 10.0000 10.0000 10.0000 10.0000 10.0000 10.0000 10.0000 10.00000 10.00000 10.00000 10.00000 10.00000 10.00000 10.00000 10.00000 10.00000 10.00000 10.00000 10.000000 10	42.213.15400.1177	2/5/2016	Erica Ann Steele Adams	O'Brien Resources, LLC	2515-734 Texas	Anderson
	42.213.15400.1179	1/29/2016	Lennie Susan Flippen	O'Brien Resources, LLC	2515-729 Texas	Anderson
AUTINITION AUTINITIA AUTINITION AUTINITION AUTINITION AUTINITION AUTINITIA AUTINITION AUTINITION AUTINITION AUTINITION AUTINITIA AUTINITION AUTINITION AUTINITIA AUTINITIA AUTINITIA AUTINITIA AUTINITIA AUTINITIA AUTINITIA AUTINITA	42.213.15400.1181	2/17/2016	Sandra Brown	O'Brien Resources, LLC	2515-739 Texas	Anderson
22.31.500.116 20.126 20.1201	42.213.15400.1183	4/1/2016	Jewell Ramirez	O'Brien Resources, LLC	2016-00007423 Texas	Henderson
23.13 10.01 10.01 10.00 10.0	42.213.15400.1184	4/1/2016	Billy James Hooker	O'Brien Resources, LLC	2016-00007416 Texas	Henderson
23.13.16.01.167 23.00.016 International Propose A Lambaget 23.13.16.00.116 23.00.016 23.00.00.016 23.00.00.016 23.00.00.016 23.00.00.016 23.00.00.016 23.00.00.016 23.00.00.016 23.00.00.016 23.00.00.016 23.00.00.016 23.00.00.016 23.00.00.016 23.00.00.016 23.00.00.016 23.00.00.016 23.00.00.016 23.00.00.016 23.00.00.016 23.00.00.00.016 23.00.00.016 23.00.00.016 23.00.00.016 23.00.00.00.016 23.00.00.00.016 23.00.00.00.016 23.00.00.00.016 23.00.00.00.00.00.00.00.00.00.00.00.00.00	42.213.15400.1185	2/17/2016	Ronald Allen Sikes, Jr.	O'Brien Resources, LLC	2515-755 Texas	Anderson
223115001199 3020000 Verwin (International Lambright and Policias A Lambright (Verwin (International Lambright and Policias A Lambright (Verwin (International Lambright) 201500000000000000000000000000000000000	42.213.15400.1186	2/1/2016	Lacey Nicole Terry	O'Brien Resources, LLC	2016-00006226 Texas	Henderson
2011-1560-159 2020	42.213.15400.1187	2/3/2016	Hayden P. Mitchell	O'Brien Resources, LLC	2515-770 Texas	Anderson
2011 5001 1019 7020 2015 7020	42.213.15400.1189	3/24/2016	Harvill Lester Lambright and Patricia A. Lambright	O'Brien Resources, LLC	2016-00006231 Texas	Henderson
22.11.15.00.10.177 71.00.117 Poggy (Saken, Earl Debel Hillower, and Dearny Melline Company LD 2510-0.00.10.10 Tosse Anderson Company LD 2510-0.00.10.10 Tosse Anderson Company LD 2510-0.00.10.10.10.10.10.10.10.10.10.10.10.10	42.213.15400.1193	3/29/2016	Wesley Richmon and Jodi Richmon	O'Brien Resources, LLC	2016-00006230 Texas	Henderson
2211 1500 1201 2010	42.213.15400.1196	2/5/2016	Joan Ann Armstrong Glass	O'Brien Resources, LLC	2515-775 Texas	Anderson
### ### ### ### ### ### ### ### ### ##	42.213.15400.1197	7/1/2013	Peggy Kirkland, Earl Debbs Hillhouse, and Deany Melton	O'Brien Resources, LLC	2513-0168 Texas	Anderson
22 21 31 5000 1220 4422010 B Revalor Graw Markery O'Bien Resources, LLC 25 5 680 Team More Markery More Resources, LLC 25 16 600000021 Teams Henders 42 21 31 5000 1220 4472010 B Librar Cox Cole Warren O'Bien Resources, LLC 2016 00000021 Teams Henders 42 31 31 5000 1220 4407010 B Librar Cox Cole Warren O'Bien Resources, LLC 25 16 60000021 Teams Henders 42 31 31 5000 1220 4007010 B Salvago Regare and Wardel Belie Regers O'Bien Resources, LLC 25 16 600 Team Henders 42 21 31 5000 1211 200200 B Order Charles Essalvy, Jr. O'Bien Resources, LLC 25 16 400 Team Henders 42 21 31 5000 1212 428001 B Kally Libr Wally Libr O'Bien Resources, LLC 25 16 400 Team Andread 42 21 31 5000 1212 14000 121 A Salvago Libr Wally Libr O'Bien Resources, LLC 25 16 400 Team Andread 42 21 31 5000 1212 14000 121 A Salvago Libr Wally	42.213.15400.1200	3/31/2016	Black Stone Minerals Company, LP	O'Brien Resources, LLC	2529-470 Texas	Anderson
42.211.156/0.1204 442016 John Wulf Simm Cox Cole Wurten Others Resources, LLC 2016-00006212 Tows Hersiter A 27.131.156/0.1206 470.10098 Simm Cox Cole Wurten Others Resources, LLC 2016-00006213 Tows Andreas A 27.131.156/0.1209 April Simm Cox Cole Wurten Others Resources, LLC 2016-00006213 Tows Andreas A 27.131.156/0.1201 April Simm Cox Cole Wurten Others Resources, LLC 2016-00006214 Tows Andreas A 27.131.156/0.1201 April Simm Cox Cole Wurten Others Resources, LLC 2016-00006214 Tows Andreas A 27.131.156/0.1212 A282016 Kalty Link Others Resources, LLC 2016-0000614 Tows Andreas A 27.131.156/0.1212 A282016 Kalty Link Others Resources, LLC 2016-0000614 Tows Andreas A 27.131.156/0.1212 A282016 Kalty Link Others Resources, LLC 2016-0000614 Tows Andreas A 27.131.156/0.1212 A282016 Kalty Link Others Resources, LLC 2016-00006414 Tows Andreas A 27.131.156/0.1213 April Simm Cole Berlink Andreas A 27.131.156/0.1214 A282016 Kalty Link Others Resources, LLC 2016-0000742 Tows Andreas A 27.131.156/0.1214 A282016 April Simm Cole Berlink Andreas A Address A Address A Address A Address A Ad	42.213.15400.1201	4/1/2016	Purvis Wayne Hooker	O'Brien Resources, LLC	2016-00007424 Texas	Henderson
A2213154001200 A210016 Elaine Cox Cole Warren Colle Warren Colle Monte College Col	42.213.15400.1202	4/22/2016	Randal Glen Matney	O'Brien Resources, LLC	2515-666 Texas	Anderson
42.213.1540.1207 480018 Charles L. Price L. Pric	42.213.15400.1204	4/4/2016	John Wulf	O'Brien Resources, LLC	2016-00006212 Texas	Henderson
Recipro	42.213.15400.1206	4/21/2016	Elaine Cox Cole Warren	O'Brien Resources, LLC	2016-00006213 Texas	Henderson
Age 1,5460,121 2,1201	42.213.15400.1207	4/6/2016	Charles L. Price	O'Brien Resources, LLC	2515-674 Texas	Anderson
42 231 15400 1212 4/280 16 Maily Laik Corona Cibrin Resources, LLC 2515-83 Teurs Andresou Cibrin Resources, LLC	42.213.15400.1208	3/31/2016	Jerry Wayne Rogers and Wanda Beth Rogers	O'Brien Resources, LLC	2016-00006214 Texas	Henderson
1422115 1442016 1122015 21440	42.213.15400.1211	2/3/2016	Orville Charles Easley, Jr.	O'Brien Resources, LLC	2515-016 Texas	Anderson
Registro 11/22015	42.213.15400.1212	4/28/2016	Kathy Link	O'Brien Resources, LLC	2515-683 Texas	Anderson
42 213 15400 1277 54/2016 Nodra Jean Skiles 0'Binen Resources, LLC 2518-463 Texas Anderson 42 213 15400 1278 41722016 Debbis Jenishins 0'Binen Resources, LLC 2016-00007427 Texas Henderson 42 213 15400 1220 41722016 Tormine Michael Lambright 0'Binen Resources, LLC 2016-00007427 Texas Henderson 42 213 15400 1220 41722016 Arita Robinson 0'Binen Resources, LLC 2016-00007427 Texas Henderson 42 213 15400 1222 41722016 Arita Robinson 0'Binen Resources, LLC 2016-00007427 Texas Henderson 42 213 15400 1222 41722016 Arita Robinson 0'Binen Resources, LLC 2016-00007420 Texas Henderson 42 213 15400 1222 4172016 Arita Robinson 0'Binen Resources, LLC 2016-00007420 Texas Henderson 42 213 15400 1222 4172016 Arita Robinson 0'Binen Resources, LLC 2016-00007420 Texas Henderson 42 213 15400 1222 4172016 Arita Robinson 0'Binen Resources, LLC 2016-00007420 Texas Henderson 42 213 15400 1227 4172016 Arita Robinson 0'Binen Resources, LLC 2016-00007420 Texas Henderson 42 213 15400 1227 4172016 Arita Robinson 0'Binen Resources, LLC 2016-00007420 Texas Henderson 42 213 15400 1227 4172016 Arita Robinson 0'Binen Resources, LLC 2016-00007420 Texas Henderson 42 213 15400 1227 4172016 Arita Robinson 0'Binen Resources, LLC 2016-00007420 Texas Henderson 42 213 15400 1227 4172016 Arita Robinson 0'Binen Resources, LLC 2016-00007420 Texas Henderson 42 213 15400 1227 4172016 Arita Robinson 0'Binen Resources, LLC 2016-000007420 Texas Anderson 42 213 15400 1227 4172016 Arita Robinson 0'Binen Resources, LLC 2016-00007420 Texas Anderson 42 213 15400 1221 5000 123 5000 10 Anocyte Arita Robinson 10 Arita Robinson 1	42.213.15400.1213	1/14/2016	Ronnie Gene Crow	O'Brien Resources, LLC	2518-406 Texas	Anderson
42 213 15400 1218 41/22018 Debbie Jenkins O'Brien Resources, LLC 2016-00007428 Texas Henders 42 213 15400 1229 41/122016 Leinral Lambright O'Brien Resources, LLC 2016-00007428 Texas Henders 42 213 15400 1220 41/122016 Anter Resources, LLC 2016-00007428 Texas Henders 42 213 15400 1221 41/122016 Anter Resources, LLC 2016-00007428 Texas Henders 42 213 15400 1221 41/122016 Anter Resources, LLC 2016-00007428 Texas Henders 42 213 15400 1222 44/2016 Anter Resources, LLC 2016-00007428 Texas Henders 42 213 15400 1223 44/2016 Burby Henderson O'Brien Resources, LLC 2016-00007428 Texas Anterson 42 213 15400 1223 44/2016 Cheby Henderson O'Brien Resources, LLC 2016-00000742 Texas Henders 42 213 15400 1223 44/2016 Cheby Henderson O'Brien Resources, LLC 2016-0000742 Texas Henders 42 213 15400 1224 41/2016 Cheby Henderson C'Brien Resources, LLC 2016-0000742 Texas Henders 42 213 15400 1	42.213.15400.1216	11/2/2015	Joshua Earl Jenkins	O'Brien Resources, LLC	2016-00007434 Texas	Henderson
42 2213 15400 1219 4122016 Celand Lambright O'Brien Resources, LLC 2016-00007427 Texas Henders 42 2213 15400 1221 4122016 Tommis Michael Lambright O'Brien Resources, LLC 2016-0000742 Texas Henders 42 2213 15400 1221 4122016 Antia Robinson O'Brien Resources, LLC 2016-0000742 Texas Henders 42 2213 15400 1221 4122016 Debty Henderson O'Brien Resources, LLC 2016-0000742 Texas Henders 42 2213 15400 1223 4162016 Debty Henderson O'Brien Resources, LLC 2016-0000742 Texas Henders 42 2213 15400 1223 4162016 Debty Henderson O'Brien Resources, LLC 2016-0000742 Texas Henders 42 2213 15400 1223 4162016 Debty Henderson O'Brien Resources, LLC 2016-0000742 Texas Henders 42 2213 15400 1227 4162016 Debty Henderson O'Brien Resources, LLC 2016-0000742 Texas Henders 42 2213 15400 1227 4162016 Debty Henderson O'Brien Resources, LLC 2016-0000742 Texas Henders 42 2213 15400 1227 4162016 Roburt E. Swill II Commission Henders 42 2213 15400 1227 4162016 Roburt E. Swill II Commission Henders 42 2213 15400 1227 4162016 Roburt E. Swill II Commission Henders 42 2213 15400 1227 4162016 Roburt E. Swill II Commission Henders 42 2213 15400 1228 4162016 Roburt E. Swill II Commission Henders 42 2213 15400 1229 4162016 Roburt E. Swill II Commission Henders 42 2213 15400 1229 4162016 Roburt E. Swill II Commission Henders 42 2213 15400 1229 4162016 Roburt I. Tidgren Commission Henders 42 2213 15400 1229 4162016 Roburt I. Tidgren Commission Henders 42 2213 15400 1229 4162016 Roburt I. Tidgren Commission Henders 42 2213 15400 1229 4162016 Roburt I. Tidgren Commission Henders 42 2213 15400 1229 4162016 Roburt I. Tidgren Commission Henders 42 2213 15400 1229 41620 416	42.213.15400.1217	5/4/2016	Nedra Jean Skiles	O'Brien Resources, LLC	2518-463 Texas	Anderson
42.213.15400.1220 4/12/2016 Tormine Michael Lambright O'Brien Resources, LLC 2016-00007428 Texas Henders 42.213.15400.1221 4/12/2016 Artila Sobianson O'Brien Resources, LLC 2016-00007425 Texas Henders 42.213.15400.1222 4/42/2016 Carrier Wulf Cates O'Brien Resources, LLC 2516-471 Texas Anderson 42.213.15400.1223 4/42/2016 Deby Henderson O'Brien Resources, LLC 2516-471 Texas Anderson 42.213.15400.1225 4/42/2018 Janet Wulf Martinez O'Brien Resources, LLC 2016-00007429 Texas Henders 42.213.15400.1227 10/70/201 John L. Powers and Debra O'Brien Resources, LLC 2016-00007429 Texas Henders 42.213.15400.1228 3/21/2016 Nell Year In by Victor W. Pryor, Jir., Trustee O'Brien Resources, LLC 2016-000007429 Texas Henders 42.213.15400.1223 5/8/2016 Narror Creedius McCall O'Brien Resources, LLC 2516-408 Texas Anderson 42.213.15400.1223 5/8/2016 Sharror Derec O'Brien Resources, LLC 2518-481 Texas Anderson 42.213.15400.1223 <td< td=""><td>42.213.15400.1218</td><td>4/12/2016</td><td>Debbie Jenkins</td><td>O'Brien Resources, LLC</td><td>2016-00007428 Texas</td><td>Henderson</td></td<>	42.213.15400.1218	4/12/2016	Debbie Jenkins	O'Brien Resources, LLC	2016-00007428 Texas	Henderson
A2 213 15400 1221 4/12/2016 Antia Robinson O'Brien Resources, LLC 2016-00007425 Texas Henderson C'Brien Resources, LLC 2016-00007425 Texas Henderson C'Brien Resources, LLC 2016-00007425 Texas Henderson C'Brien Resources, LLC 2016-00007425 Texas Anderson C'Brien Resources, LLC 2016-00007425 Texas Henderson C'Brien Resources, LLC 2016-00007425 Texas Anderson C'	42.213.15400.1219	4/12/2016	Leland Lambright	O'Brien Resources, LLC	2016-00007427 Texas	Henderson
	42.213.15400.1220	4/12/2016	Tommie Michael Lambright	O'Brien Resources, LLC	2016-00007426 Texas	Henderson
42.213.15400.1223 46/2016 Debby Henderson O'Brien Resources, LLC 2518-471 Texas Anderson 42.213.15400.1225 44/2016 John Lowers and Debra O'Brien Resources, LLC 2016-00007429 Texas Henders 42.213.15400.1226 37.21016 Richard E. Swift II O'Brien Resources, LLC 2016-0000787 Texas Henders 42.213.15400.1228 37.21016 Richard E. Swift II O'Brien Resources, LLC 2016-0000787 Texas Henders 42.213.15400.1229 46/2016 WP, Jr In by Victor W, Pryor, Jr., Trustee O'Brien Resources, LLC 2016-0000677 Texas Henders 42.213.15400.1229 46/2016 WP, Jr In by Victor W, Pryor, Jr., Trustee O'Brien Resources, LLC 2518-498 Texas Henders 42.213.15400.1229 56/2016 Robert J. Tidgren O'Brien Resources, LLC 2518-498 Texas Anderson 42.213.15400.1230 56/2016 Robert J. Tidgren O'Brien Resources, LLC 2518-494 Texas Anderson 42.213.15400.1235 56/2016 Robert J. Tidgren O'Brien Resources, LLC 2518-494 Texas Anderson 42.213.15400.1235 56/2016 Leonard Keith Baker O'Brien Resources, LLC 2518-517 Texas Anderson 42.213.15400.1235 47.2200 Jense Harland Price O'Brien Resources, LLC 2518-517 Texas Anderson 42.213.15400.1235 47.2200 Jense Harland Price O'Brien Resources, LLC 2518-517 Texas Anderson 42.213.15400.1236 47.2200 Jense Harland Price O'Brien Resources, LLC 2518-517 Texas Anderson 42.213.15400.1236 47.2200 Jense Harland Price O'Brien Resources, LLC 2518-517 Texas Anderson 42.213.15400.1236 56/2016 Charles L. Price O'Brien Resources, LLC 2518-517 Texas Anderson 42.213.15400.1239 56/2016 Charles L. Price O'Brien Resources, LLC 2518-517 Texas Anderson 42.213.15400.1239 56/2016 Charles L. Price O'Brien Resources, LLC 2518-517 Texas Anderson 42.213.15400.1239 56/2016 Charles L. Price O'Brien Resources, LLC 2518-517 Texas Anderson 42.213.15400.1239 56/2016 Charles L. Price O'Brien Resources, LLC 2518-517 Texas Anderson 42.213.15400.1230 56/2016 Charles L. Price O'Brien Resources, LLC 2518-517 Texas Anderson 42.213.15400.1230 56/2016 Charles L. Price O'Brien Resources, LLC 2518-518 Texas Anderson 42.213.15400.1230 56/2016 Charles L. Price O'Brien Resources,	42.213.15400.1221	4/12/2016	Anita Robinson	O'Brien Resources, LLC	2016-00007425 Texas	Henderson
42 213 15400 1225	42.213.15400.1222	4/4/2016	Karin Wulf Cates	O'Brien Resources, LLC	2016-00007430 Texas	Henderson
42.213.15400.1227 10/772014 John L. Powers and Debra O'Brien Resources, LLC 2015-00000798 Texas Henders 42.213.15400.1228 3/21/2016 Richard E. Swift II O'Brien Resources, LLC 2515-705 Texas Andeson 42.213.15400.1229 4/6/2016 VIP Jrin by Victor W. Pryor, Jr., Trustee O'Brien Resources, LLC 2016-00006211 Texas Andeson 42.213.15400.1231 5/3/2016 Nancy Ann Crecetius McCell O'Brien Resources, LLC 2518-498 Texas Anderson 42.213.15400.1232 5/6/2016 Robert J. Tidgren O'Brien Resources, LLC 2518-491 Texas Anderson 42.213.15400.1235 5/3/2016 Leonard Keith Baker O'Brien Resources, LLC 2518-521 Texas Anderson 42.213.15400.1236 4/28/2016 Kale Jeanette Price Pierce O'Brien Resources, LLC 2518-521 Texas Anderson 42.213.15400.1237 4/28/2016 Charles L. Price O'Brien Resources, LLC 2518-514 Texas Anderson 42.213.15400.1241 5/14/2016 Amanda DeShazo aka Amanda Grubbs O'Brien Resources, LLC <td>42.213.15400.1223</td> <td>4/6/2016</td> <td>Debby Henderson</td> <td>O'Brien Resources, LLC</td> <td>2518-471 Texas</td> <td>Anderson</td>	42.213.15400.1223	4/6/2016	Debby Henderson	O'Brien Resources, LLC	2518-471 Texas	Anderson
	42.213.15400.1225	4/4/2016	Janet Wulf Martinez	O'Brien Resources, LLC	2016-00007429 Texas	Henderson
4213.15400.1229 4/6/2016 VWP Jr Inc by Victor W. Pryor, Jr., Trustee O'Brien Resources, LLC 2016-00006211 Texas Henders Face State S	42.213.15400.1227			, ,	2015-00000798 Texas	Henderson
42.213.15400.1231 5/3/2016 Nancy Ann Crecelius McCall O'Brien Resources, LLC 2518-498 Texas Andersources, LLC 42.213.15400.1232 5/6/2016 Robert J. Tidgren O'Brien Resources, LLC 2518-491 Texas Andersources, LLC 42.213.15400.1234 5/6/2016 Sharron Perez O'Brien Resources, LLC 2518-521 Texas Andersources, LLC 42.213.15400.1235 5/3/2016 Leonard Keith Baker O'Brien Resources, LLC 2518-521 Texas Andersources, LLC 42.213.15400.1235 Arctic State Jeanette Price Pierce O'Brien Resources, LLC 42.213.15400.1236 A/28/2016 Mares Harfand Price Arctic State Jeanette Price Pierce O'Brien Resources, LLC 42.213.15400.1237 A/28/2016 James Harfand Price O'Brien Resources, LLC 42.213.15400.1238 A/28/2016 Armanda DeShazo aka Amanda Grubbs O'Brien Resources, LLC 42.213.15400.1239 S/4/2016 Dona McEwen O'Brien Resources, LLC 42.213.15400.1239 S/4/2016 Dona McEwen O'Brien Resources, LLC 42.213.15400.1234 S/18/2016 Suthwest Mineral Group by: George Henderson III, Manager O'Brien Resources, LLC 42.213.15400.1234 S/18/2016 Resources, LLC 42.213.15400.1234	42.213.15400.1228	3/21/2016	Richard E. Swift II	O'Brien Resources, LLC	2515-705 Texas	Andeson
42.213.15400.1232 516/2016 Robert J. Tidgren O'Brien Resources, LLC 2518-494 Texas Anderson 42.213.15400.1234 514/2016 Sharron Perez O'Brien Resources, LLC 2518-491 Texas Anderson 42.213.15400.1235 51/3/2016 Leonard Keith Baker O'Brien Resources, LLC 2518-521 Texas Anderson 42.213.15400.1237 4/28/2016 Katie Jeanette Price Pierce O'Brien Resources, LLC 2518-511 Texas Anderson 42.213.15400.1237 4/28/2016 Lames Harland Price O'Brien Resources, LLC 2518-511 Texas Anderson 42.213.15400.1238 4/28/2016 Charles L. Price O'Brien Resources, LLC 2518-511 Texas Anderson 42.213.15400.1239 5/4/2016 Amanda DeShazo aka Amanda Grubbs O'Brien Resources, LLC 2518-517 Texas Anderson 42.213.15400.1240 5/18/2016 Southwest Mineral Group by: George Henderson III, Manager O'Brien Resources, LLC 2016-00012753 Texas Henders 42.213.15400.1245 5/6/2016 Tommy Everett O'Brien Resources,	42.213.15400.1229	4/6/2016	VWP Jr Inc by Victor W. Pryor, Jr., Trustee		2016-00006211 Texas	Henderson
42.213.15400.1234 5/4/2016 Sharron Perez O'Brien Resources, LLC 2518-491 Texas Anderson Aderson 42.213.15400.1235 5/3/2016 Leonard Keith Baker O'Brien Resources, LLC 2518-521 Texas Anderson 42.213.15400.1236 4/28/2016 Katie Jeanette Price Pierce O'Brien Resources, LLC 2518-501 Texas Anderson 42.213.15400.1237 4/28/2016 James Harland Price O'Brien Resources, LLC 2518-511 Texas Anderson 42.213.15400.1238 4/28/2016 Charles L. Price O'Brien Resources, LLC 2518-514 Texas Anderson 42.213.15400.1239 5/4/2016 Amanda DeShazo aka Amanda Grubbs O'Brien Resources, LLC 2518-517 Texas Anderson 42.213.15400.1240 5/4/2016 Donna McEwen O'Brien Resources, LLC 2518-505 Texas Anderson 42.213.15400.1241 5/18/2016 Southwest Mineral Group by: George Henderson III, Manager O'Brien Resources, LLC 2016-00012/53 Texas Henders 42.213.15400.1242 5/16/2016 Rentier, Inc. by Shirley Pogue, President O'Br	42.213.15400.1231	5/3/2016	Nancy Ann Crecelius McCall	O'Brien Resources, LLC	2518-498 Texas	Anderson
42.213.15400.1235 5/3/2016 Leonard Keith Baker O'Brien Resources, LLC 2518-521 Texas Anderson 42.213.15400.1236 4/28/2016 Katie Jeanette Price Pierce O'Brien Resources, LLC 2518-502 Texas Anderson 42.213.15400.1237 4/28/2016 James Harland Price O'Brien Resources, LLC 2518-511 Texas Anderson 42.213.15400.1238 4/28/2016 Charles L. Price O'Brien Resources, LLC 2518-511 Texas Anderson 42.213.15400.1239 5/4/2016 Amanda DeShazo aka Amanda Grubbs O'Brien Resources, LLC 2518-517 Texas Anderson 42.213.15400.1240 5/4/2016 Donna McEwen O'Brien Resources, LLC 2518-505 Texas Anderson 42.213.15400.1241 5/18/2016 Southwest Mineral Group by: George Henderson III, Manager O'Brien Resources, LLC 2016-00012753 Texas Henders 42.213.15400.1242 5/16/2016 Rentier, Inc. by Shirley Pogue, President O'Brien Resources, LLC 2016-00007422 Texas Henders 42.213.15400.1244 5/6/2016 Tommy Everett O'Brien Resources, LLC 2016-00007420 Texas Henders 42.213.15400.1245 5/23/2016 Michael E. Hurt O'Brien Resources, LLC 2518-419 Texas Anderson 42.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-419 Texas Anderson 42.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-412 Texas Anderson 42.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-412 Texas Anderson 42.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-412 Texas Anderson 42.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-412 Texas Anderson 42.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-412 Texas Anderson 42.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-412 Texas Anderson 42.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-412 Texas Anderson 42.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-427 Texas Anderson 42.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-427 Texas Anderson 42.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-4	42.213.15400.1232	5/6/2016	Robert J. Tidgren	O'Brien Resources, LLC	2518-494 Texas	Anderson
42.213.15400.1236 4/28/2016 Katie Jeanette Price Pierce O'Brien Resources, LLC 2518-502 Texas Anderson At 2.213.15400.1237 4/28/2016 James Harland Price O'Brien Resources, LLC 2518-511 Texas Anderson At 2.213.15400.1238 4/28/2016 Charles L. Price O'Brien Resources, LLC 2518-514 Texas Anderson At 2.213.15400.1239 5/4/2016 Amanda DeShazo aka Amanda Grubbs O'Brien Resources, LLC 2518-517 Texas Anderson At 2.213.15400.1240 5/4/2016 Donna McEwen O'Brien Resources, LLC 2518-517 Texas Anderson At 2.213.15400.1240 5/18/2016 Southwest Mineral Group by: George Henderson III, Manager O'Brien Resources, LLC 2016-00012753 Texas Henders At 2.213.15400.1242 5/16/2016 Rentier, Inc. by Shirley Pogue, President O'Brien Resources, LLC 2016-00007422 Texas Henders At 2.213.15400.1244 5/6/2016 Tommy Everett O'Brien Resources, LLC 2518-416 Texas Anderson At 2.213.15400.1245 5/23/2016 Michael E. Hurt O'Brien Resources, LLC 2518-419 Texas Anderson At 2.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-412 Texas Anderson At 2.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-412 Texas Anderson At 2.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-412 Texas Anderson At 2.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-412 Texas Anderson At 2.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-412 Texas Anderson At 2.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-412 Texas Anderson At 2.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-412 Texas Anderson At 2.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-412 Texas Anderson At 2.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-422 Texas Anderson At 2.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-427 Texas Anderson At 2.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-427 Texas Anderson At 2.213.15400.1247 5/4/2016 Gary Wayne Cr	42.213.15400.1234	5/4/2016	Sharron Perez	O'Brien Resources, LLC	2518-491 Texas	Anderson
42.213.15400.1237 4/28/2016 Unames Harland Price O'Brien Resources, LLC 2518-511 Texas Anderson At 2.213.15400.1238 4/28/2016 Charles L. Price O'Brien Resources, LLC 2518-514 Texas Anderson At 2.213.15400.1239 5/4/2016 Amanda DeShazo aka Amanda Grubbs O'Brien Resources, LLC 2518-517 Texas Anderson At 2.213.15400.1240 5/4/2016 Donna McEwen O'Brien Resources, LLC 2518-505 Texas Anderson At 2.213.15400.1241 5/18/2016 Southwest Mineral Group by: George Henderson III, Manager O'Brien Resources, LLC 2016-00012753 Texas Henders At 2.213.15400.1242 5/16/2016 Rentier, Inc. by Shirley Pogue, President O'Brien Resources, LLC 2016-00007422 Texas Henders At 2.213.15400.1244 5/6/2016 Tommy Everett O'Brien Resources, LLC 2016-00007420 Texas Henders At 2.213.15400.1245 5/23/2016 Michael E. Hurt O'Brien Resources, LLC 2016-00007420 Texas Henders At 2.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-419 Texas Anderson At 2.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-422 Texas Anderson At 2.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-412 Texas Anderson At 2.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-412 Texas Anderson At 2.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-412 Texas Anderson At 2.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-412 Texas Anderson At 2.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-412 Texas Anderson At 2.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-412 Texas Anderson At 2.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-412 Texas Anderson At 2.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-412 Texas Anderson At 2.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-412 Texas Anderson At 2.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-412 Texas Anderson At 2.213.15400.1247 5/4/2016 Gary Wayne C	42.213.15400.1235	5/3/2016	Leonard Keith Baker	O'Brien Resources, LLC	2518-521 Texas	Anderson
42.213.15400.1238 4/28/2016 Charles L. Price O'Brien Resources, LLC 2518-514 Texas Anderson Addresson Addresson Anderson Addresson Anderson Addresson Anderson Addresson Anderson Addresson Addres	42.213.15400.1236	4/28/2016	Katie Jeanette Price Pierce	O'Brien Resources, LLC	2518-502 Texas	Anderson
42.213.15400.1239 5/4/2016 Amanda DeShazo aka Amanda Grubbs O'Brien Resources, LLC 2518-517 Texas Anderson 42.213.15400.1240 5/4/2016 Donna McEwen O'Brien Resources, LLC 2518-505 Texas Anderson 42.213.15400.1241 5/18/2016 Southwest Mineral Group by: George Henderson III, Manager O'Brien Resources, LLC 2016-00012753 Texas Henders 42.213.15400.1242 5/16/2016 Rentier, Inc. by Shirley Pogue, President O'Brien Resources, LLC 2016-00007422 Texas Henders 42.213.15400.1244 5/6/2016 Tommy Everett O'Brien Resources, LLC 2518-416 Texas Anderson 42.213.15400.1245 5/23/2016 Michael E. Hurt O'Brien Resources, LLC 2016-00007420 Texas Henders 42.213.15400.1245 5/23/2016 James Harland Price O'Brien Resources, LLC 2518-419 Texas Anderson 42.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-422 Texas Anderson 42.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-422 Texas Anderson 42.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-422 Texas Anderson 42.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-422 Texas Anderson 42.213.15400.1247 5/4/2016 Gary Wayne Crecelius	42.213.15400.1237	4/28/2016	James Harland Price	O'Brien Resources, LLC	2518-511 Texas	Anderson
42.213.15400.1240 5/4/2016 Donna McEwen O'Brien Resources, LLC 2518-505 Texas Anderson 42.213.15400.1241 5/18/2016 Southwest Mineral Group by: George Henderson III, Manager O'Brien Resources, LLC 2016-00012753 Texas Henders 42.213.15400.1242 5/16/2016 Rentier, Inc. by Shirley Pogue, President O'Brien Resources, LLC 2016-00007422 Texas Henders 42.213.15400.1244 5/6/2016 Tommy Everett O'Brien Resources, LLC 2518-416 Texas Anderson 42.213.15400.1245 5/23/2016 Michael E. Hurt O'Brien Resources, LLC 2016-00007420 Texas Henders 42.213.15400.1246 5/23/2016 James Harland Price O'Brien Resources, LLC 2518-419 Texas Anderson 42.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-422 Texas Anderson	42.213.15400.1238	4/28/2016	Charles L. Price	O'Brien Resources, LLC	2518-514 Texas	Anderson
42.213.15400.1241 5/18/2016 Southwest Mineral Group by: George Henderson III, Manager O'Brien Resources, LLC 2016-00012753 Texas Henders 42.213.15400.1242 5/16/2016 Rentier, Inc. by Shirley Pogue, President O'Brien Resources, LLC 2016-00007422 Texas Henders 42.213.15400.1244 5/6/2016 Tommy Everett O'Brien Resources, LLC 2518-416 Texas Anderson 42.213.15400.1245 5/23/2016 Michael E. Hurt O'Brien Resources, LLC 2016-00007420 Texas Henders 42.213.15400.1246 5/23/2016 James Harland Price O'Brien Resources, LLC 2518-419 Texas Anderson 42.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-422 Texas Anderson	42.213.15400.1239	5/4/2016	Amanda DeShazo aka Amanda Grubbs	O'Brien Resources, LLC	2518-517 Texas	Anderson
42.213.15400.1242 5/16/2016 Rentier, Inc. by Shirley Pogue, President O'Brien Resources, LLC 2016-00007422 Texas Henders 42.213.15400.1244 5/6/2016 Tommy Everett O'Brien Resources, LLC 2518-416 Texas Andersor 42.213.15400.1245 5/23/2016 Michael E. Hurt O'Brien Resources, LLC 2016-00007420 Texas Henders 42.213.15400.1246 5/23/2016 James Harland Price O'Brien Resources, LLC 2518-419 Texas Andersor 42.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-422 Texas Andersor	42.213.15400.1240	5/4/2016	Donna McEwen	O'Brien Resources, LLC	2518-505 Texas	Anderson
42.213.15400.1244 5/6/2016 Tommy Everett O'Brien Resources, LLC 2518-416 Texas Anderson Aderson Aderso	42.213.15400.1241	5/18/2016	Southwest Mineral Group by: George Henderson III, Manager	O'Brien Resources, LLC	2016-00012753 Texas	Henderson
42.213.15400.1245 5/23/2016 Michael E. Hurt O'Brien Resources, LLC 2016-00007420 Texas Henders 42.213.15400.1246 5/23/2016 James Harland Price O'Brien Resources, LLC 2518-419 Texas Anderson 42.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-422 Texas Anderson	42.213.15400.1242	5/16/2016	Rentier, Inc. by Shirley Pogue, President	O'Brien Resources, LLC	2016-00007422 Texas	Henderson
42.213.15400.1246 5/23/2016 James Harland Price O'Brien Resources, LLC 2518-419 Texas Anderson 42.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-422 Texas Anderson 42.213.15400.1247 5/4/2016 Gary Wayne Crecelius	42.213.15400.1244	5/6/2016	Tommy Everett	O'Brien Resources, LLC	2518-416 Texas	Anderson
42.213.15400.1247 5/4/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-422 Texas Anderson	42.213.15400.1245	5/23/2016	Michael E. Hurt	O'Brien Resources, LLC	2016-00007420 Texas	Henderson
	42.213.15400.1246	5/23/2016	James Harland Price	O'Brien Resources, LLC	2518-419 Texas	Anderson
42.213.15400.1248 5/23/2016 Gary Wayne Crecelius O'Brien Resources, LLC 2518-426 Texas Anderson	42.213.15400.1247	5/4/2016	Gary Wayne Crecellus	O'Brien Resources, LLC	2518-422 Texas	Anderson
	42.213.15400.1248	5/23/2016	Gary Wayne Crecelius	O'Brien Resources, LLC	2518-426 Texas	Anderson

Lease #	Date	Lessor	Lessee	Book-Page State	County
42.213.15400.1249	5/23/2016	Debby L. Henderson	O'Brien Resources, LLC	2518-4302 Texas	Anderson
42.213.15400.1250	5/23/2016	Charles L. Price	O'Brien Resources, LLC	2518-433 Texas	Anderson
42.213.15400.1251	5/23/2016	Brenda Nipper (Cole)	O'Brien Resources, LLC	2518-436 Texas	Anderson
42.213.15400.1252	5/19/2016	Billy Gene Fisher	O'Brien Resources, LLC	2016-00007419 Texas	Henderson
42.213.15400.1254	11/4/2015	Barbara Vick	O'Brien Resources, LLC	2518-452 Texas	Anderson
42.213.15400.1255	5/4/2016	Barbara Tucker	O'Brien Resources, LLC	2518-445 Texas	Anderson
42.213.15400.1256		Tommy Everett	O'Brien Resources, LLC	2518-412 Texas	Anderson
42.213.15400.1257		Judy Harwick	O'Brien Resources, LLC	2518-508 Texas	Anderson
42.213.15400.1258	5/6/2016	Mary Kathryn Pickle	O'Brien Resources, LLC	2522-0718 Texas	Anderson
42.213.15400.1259		W. C. "Chip" Perryman III	O'Brien Resources, LLC O'Brien Resources, LLC	2016-00008879 Texas	Henderson
42.213.15400.1260 42.213.15400.1261	5/4/2016	Judy Harwick Roy Eggleston	O'Brien Resources, LLC	2522-683 Texas 2522-690 Texas	Anderson Anderson
42.213.15400.1262		Roy Eggleston	O'Brien Resources, LLC	2522-687 Texas	Anderson
42.213.15400.1263		Rhonda G. Garcia	O'Brien Resources, LLC	2522-701 Texas	Anderson
42.213.15400.1264		Rhonda G. Garcia	O'Brien Resources, LLC	2522-705 Texas	Anderson
42.213.15400.1265	5/23/2016		O'Brien Resources, LLC	2522-0712 Texas	Anderson
42.213.15400.1266	5/23/2016	Nancy Ann Crecelius McCall	O'Brien Resources, LLC	2522-697 Texas	Anderson
42.213.15400.1267	5/23/2016	Kathy Link	O'Brien Resources, LLC	2522-693 Texas	Anderson
42.213.15400.1268	5/29/2016	Premier LM LP	O'Brien Resources, LLC	2016-00008876 Texas	Henderson
42.213.15400.1269	5/29/2016	Premier LM II, LP	O'Brien Resources, LLC	2017-00001394 Texas	Henderson
42.213.15400.1270	5/20/2016	Premier LM III, LP	O'Brien Resources, LLC	2522-678 Texas	Anderson
42.213.15400.1271	5/23/2016	Donna McEwen	O'Brien Resources, LLC	2522-708 Texas	Anderson
42.213.15400.1273	5/19/2016	Brenda Fisher Wheeler	O'Brien Resources, LLC	2016-00008867 Texas	Henderson
42.213.15400.1274	5/19/2016	Jan Lancaster Stanton	O'Brien Resources, LLC	2016-00008868 Texas	Henderson
42.213.15400.1275	5/4/2016	Jo Ann Meeks Delano aka Dean	O'Brien Resources, LLC	2522-676 Texas	Anderson
42.213.15400.1276	5/23/2016		O'Brien Resources, LLC	2522-671 Texas	Anderson
42.213.15400.1277		Patricia T. McGuffey	O'Brien Resources, LLC	2016-00008871 Texas	Henderson
42.213.15400.1278		Wanda Stafford Monk	O'Brien Resources, LLC	2522-0680 Texas	Anderson
42.213.15400.1279 42.213.15400.1282		James W. Hurt	O'Brien Resources, LLC O'Brien Resources, LLC	2016-00008865 Texas 2530-751 Texas	Henderson
42.213.15400.1282	4/4/2016	Katie Jeanette Price Pierce Robert Wulf (now deceased)	O'Brien Resources, LLC	2530-751 Texas 2016-00012778 Texas	Anderson Henderson
42.213.15400.1284		Marshall D. Brown, Jr.	O'Brien Resources, LLC	2530-754 Texas	Anderson
42.213.15400.1285	1/26/2016		O'Brien Resources, LLC	2016-00007436 Texas	Henderson
42.213.15400.1286		Robert J. Tidgren	O'Brien Resources, LLC	2530-744 Texas	Anderson
42.213.15400.1287		Leonard Keith Baker	O'Brien Resources, LLC	2530-741 Texas	Anderson
42.213.15400.1288	5/23/2016	Amanda DeShazo aka Amanda Grubbs	O'Brien Resources, LLC	2530-747 Texas	Anderson
42.213.15400.1291	7/1/2016	Stephen T. Stafford	O'Brien Resources, LLC	2530-737 Texas	Anderson
42.213.15400.1292	7/1/2016	Patricia Ann Bragg	O'Brien Resources, LLC	2016-00012773 Texas	Henderson
42.213.15400.1293	6/23/2016	Norma Smith	O'Brien Resources, LLC	2530-707 Texas	Anderson
42.213.15400.1294	7/15/2016	Leona Earl May	O'Brien Resources, LLC	2016-00012763 Texas	Henderson
42.213.15400.1296	5/4/2016	Brenda Kay Pickle Nipper (Cole)	O'Brien Resources, LLC	2530-705 Texas	Anderson
42.213.15400.1297	4/28/2016	Sherwin Hutchins	O'Brien Resources, LLC	2530-720 Texas	Anderson
42.213.15400.1298		Kelly Milan Trahan and Scott Trahan	O'Brien Resources, LLC	2016-00012510 Texas	Henderson
42.213.15400.1299		Sharron Perez	O'Brien Resources, LLC	2530-717 Texas	Anderson
42.213.15400.1300		Margie Louder Ireton Griffin	O'Brien Resources, LLC	2530-713 Texas	Anderson
42.213.15400.1301		S & L Energy Company, LLC	O'Brien Resources, LLC	2551-0260 Texas	Anderson
42.213.15400.1302 42.213.15400.1303		Freda Gail Stern Estate of Freda Gail Stern, Harley Soltes Ind Exec James Carl Louder	O'Brien Resources, LLC O'Brien Resources, LLC	2016-00012766 Texas 2530-726 Texas	Henderson Anderson
42.213.15400.1304		Premier LM III. LP	O'Brien Resources, LLC	2016-00016258 Texas	Anderson
42.213.15400.1305		Peggy Allbright	O'Brien Resources, LLC	2539-0494 Texas	Anderson
42.213.15400.1306		Betty Nix Aaron f/k/a Betty Cozette Carroll	O'Brien Resources, LLC	2530-707 Texas	Anderson
42.213.15400.1307		Johnny E. Walker and Jo Walker	O'Brien Resources, LLC	2530-730 Texas	Anderson
42.213.15400.1308	12/3/2015	Judy Ann Sherley Guido	O'Brien Resources, LLC	2539-0501 Texas	Anderson
42.213.15400.1309		Jane Russell Indiv and as Sole Beneficiary of the Est of Joan Frances Courtney	O'Brien Resources, LLC	2539-0498 Texas	Anderson
42.213.15400.1310	8/15/2016	J. W. Wylie, Jr. and Joe Ann Wylie by ahd thru Agt &AIF Karen Dublin	O'Brien Resources, LLC	2016-00016269 Texas	Henderson
42.213.15400.1311	5/19/2016	Robert D. Johnston County Judge, Anderson County, Texas acting in his capacity as Receiver for Mineral Interests appointed by the	O'Brien Resources, LLC	2016-00007978 Texas	Anderson
42.213.15400.1312	8/23/2016	Shirley Ann Stringfield Hooks	O'Brien Resources, LLC	2016-00016265 Texas	Henderson
42.213.15400.1313	7/26/2016	Randy Miller and Elizabeth Miller	O'Brien Resources, LLC	2539-0505 Texas	Anderson
42.213.15400.1315		Karen E. Clark	O'Brien Resources, LLC	2539-0512 Texas	Anderson
42.213.15400.1316		Stephanie Rusk Jefcoat	O'Brien Resources, LLC	2016-00016266 Texas	Henderson
42.213.15400.1318		Joe Ladd Faulk	O'Brien Resources, LLC	2539-0510 Texas	Anderson
42.213.15400.1319		Annabelle McGuffey Dixon	O'Brien Resources, LLC	2016-00016268 Texas	Henderson
42.213.15400.1320		Vickie Elaine Cooper aka Vickie Bullard	O'Brien Resources, LLC	2016-00016270 Texas	Henderson
42.213.15400.1321		Brenda Joyce Nikel	O'Brien Resources, LLC	2016-00016271 Texas	Henderson
42.213.15400.1323	9/14/2016	Laura Dianne Slate	O'Brien Resources, LLC	2016-00016273 Texas	Henderson

Lease #	Date	Lessor	Lessee	Book-Page State	County
42.213.15400.1324	5/23/2016	*Rodney Eggleston	O'Brien Resources, LLC	2539-0517 Texas	Anderson
42.213.15400.1325	5/4/2016	*Rodney Eggleston	O'Brien Resources, LLC	2539-0521 Texas	Anderson
42.213.15400.1326	5/19/2016	Ruth Anne Homan	O'Brien Resources, LLC	2016-00016274 Texas	Henderson
42.213.15400.1327	9/14/2016	Weldon Darwin Cooper	O'Brien Resources, LLC	2016-00016276 Texas	Henderson
42.213.15400.1328	8/31/2016	David R. Soulsby	O'Brien Resources, LLC	2017-00001398 Texas	Henderson
42.213.15400.1329	8/31/2016	Susan A. Soulsby	O'Brien Resources, LLC	2017-00001407 Texas	Henderson
42.213.15400.1330	9/14/2016	*Mary Cox	O'Brien Resources, LLC	2017-00001395 Texas	Henderson
42.213.15400.1331	9/19/2016	Milton Ferrell Holcomb*	O'Brien Resources, LLC	2551-0210 Texas	Anderson
42.213.15400.1332	8/23/2016	Donald L. Winters	O'Brien Resources, LLC	2551-0213 Texas	Anderson
42.213.15400.1333	9/26/2016	Wanda F. Marshall	O'Brien Resources, LLC	2017-00001397 Texas	Henderson
42.213.15400.1334	9/13/2016	Ronald L. Williams	O'Brien Resources, LLC	2017-00010907 Texas	Henderson
42.213.15400.1335	1/18/2017	Sonja Lewis Simon	O'Brien Resorces, LLC	2016-00001275 Texas	Henderson
42.213.15400.1336		LaMerle Boyd	OBrien Resources, LLC	2551-0219 Texas	Anderson
42.213.15400.1337	9/26/2016	LaMerle Boyd	OBrien Resources, LLC	2551-0217 Texas	Anderson
42.213.15400.1339	10/12/2016	The G. J. and C. J. Landis Living Tr by Gary J. Landis and Candice J. Landis, Co-Trustees	O'Brien Resources, LLC	2551-0240 Texas	Anderson
42.213.15400.1340	9/13/2016	County Line Cemetery Assoc	O'Brien Resources, LLC	2551-0222 Texas	Anderson
42.213.15400.1341	9/27/2016	Dorthea Eyssen Testamentary Tr by Christopher W. Burkhart	O'Brien Resources, LLC	2551-0226 Texas	Anderson
42.213.15400.1342		Rayford Fisher	O'Brien Resources, LLC	2017-00001399 Texas	Henderson
42.213.15400.1343	10/17/2016	Gerald W. Lasater	O'Brien Resources, LLC	2551-0228 Texas	Anderson
42.213.15400.1344		Stephen G. Landis	O'Brien Resources, LLC	2551-0230 Texas	Anderson
42.213.15400.1345		Robbie Mims aka Robert Mims	O'Brien Resources, LLC	2551-0240 Texas	Anderson
42.213.15400.1346	10/13/2016	Wanda F. Marshall	O'Brien Resources, LLC	2017-00001401 Texas	Henderson
42.213.15400.1347		Joseph Percy Coats	O'Brien Resources, LLC	2017-00001402 Texas	Henderson
42.213.15400.1348		William H. Faulk and Mary Lou Faulk, Individually and as Co-Trustees for Faulk Family Living Tr	O'Brien Resources, LLC	2561-0518 Texas	Anderson
42.213.15400.1349		William H. Faulk and Mary Lou Faulk, Individually and as Co-Trustees for Faulk Family Living Tr	O'Brien Resources, LLC	2561-0520 Texas	Anderson
42.213.15400.1350		Ashley Marie Rossi	O'Brien Resources, LLC	2551-0272 Texas	Anderson
42.213.15400.1351		Joseph Percy Coats	O'Brien Resources, LLC	2017-00001403 Texas	Henderson
42.213.15400.1352		Jeff Stringfield	O'Brien Resources, LLC	2551-0251 Texas	Anderson
42.213.15400.1353		Jason Stringfield	O'Brien Resources, LLC	2561-0516 Texas	Anderson
42.213.15400.1354		David Stringfield	O'Brien Resources, LLC	2551-0247 Texas	Anderson
42.213.15400.1355		Dana G. Mitchell	O'Brien Resources, LLC	2017-00001405 Texas	Henderson
42.213.15400.1356		Dana G. Mitchell	O'Brien Resources, LLC	2017-00001400 Texas	Henderson
42.213.15400.1357 42.213.15400.1358		Shirley Stringfield Hooks Jennifer Holcomb DiMaggio	O' Brien Resources, LLC O' Brien Resources, LLC	2017-00001406 Texas 2551-0266 Texas	Henderson Anderson
42.213.15400.1359		Jenniler Holcomb Dilwaggio Michael Travis Holcomb	O' Brien Resources, LLC	2551-0263 Texas	Anderson
42.213.15400.1361		Ronald Mark Hill	O'Brien Resources, LLC	2017-00004157 Texas	Henderson
42.213.15400.1361		Jackie Dwight Hill	O'Brien Resources, LLC	2017-00004157 Texas 2017-00004153 Texas	Henderson
42.213.15400.1363		Diane Hill Duncan	O'Brien Resources, LLC	2017-00004153 Texas	Henderson
42.213.15400.1364		Margaret Alice McDaniel	O'Brien Resources, LLC	2015-00004132 Texas	Henderson
42.213.15400.1365		Owen Oil & Gas Properties, LLC	O'Brien Resources, LLC	2015-00000849 Texas	Henderson
42.213.15400.1366		Mack Parmer Stringfield	O'Brien Resources, LLC	2017-00004156 Texas	Henderson
42.213.15400.1368		Janet Davis FKA Janet Jenkins	O'Brien Resources, LLC	2017-00004150 Texas	Henderson
42.213.15400.1369		Daphne Davis German	O'Brien Resources, LLC	2017-00004151 Texas	Henderson
42.213.15400.1370		Stephanie Rusk Jefcoat	O'Brien Resources, LLC	2017-00004150 Texas	Henderson
42.213.15400.1371		Mack Parmer Stringfield	O'Brien Resources, LLC	2017-00004148 Texas	Henderson
42.213.15400.1372		Glen Parker	O'Brien Resources, LLC	2017-00004147 Texas	Henderson
42.213.15400.1373		Bobby A. Huddleston and Edwina W. Huddleston	O'Brien Resources, LLC	2559-0541 Texas	Anderson
42.213.15400.1375		Harold Gene Royle	O' Brien Resources, LLC	2017-00004166 Texas	Henderson
42.213.15400.1376		Tommy Franklin Royle	O' Brien Resources, LLC	2017-00004165 Texas	Henderson
42.213.15400.1377	9/23/2015	Richard Sanders, Receivership Cause # CV15-0318-3	O'Brien Resources, LLC	2015-00013843 Texas	Henderson
			O'Brien Resources, LLC	2552-0217 Texas	Anderson
42.213.15400.1378	1/17/2017	Cole O'Neill Ltd. by: Connie Fain Vice PResidena and Trust Officer	O Brieff (Coouroco, ELO	2002*0211 TEXAS	
42.213.15400.1378 42.213.15400.1379				2017-00002184 Texas	
	1/17/2017	Cole O'Neil Ltd. by: Connie Fain Vice Presidena and Trust Officer Robert Michael Pickel Thomas M. Colley Residual Tr by and through its Trustee Citizens National Bank	O'Brien Resources, LLC O'Brien Resources, LLC		Henderson Anderson
42.213.15400.1379	1/17/2017 1/17/2017	Robert MIchael Pickel	O'Brien Resources, LLC	2017-00002184 Texas	Henderson
42.213.15400.1379 42.213.15400.1380	1/17/2017 1/17/2017 11/30/2016	Robert Michael Pickel Thomas M. Colley Residual Tr by and through its Trustee Citizens National Bank	O'Brien Resources, LLC O'Brien Resources, LLC	2017-00002184 Texas 2552-0215 Texas	Henderson Anderson
42.213.15400.1379 42.213.15400.1380 42.213.15400.1381	1/17/2017 1/17/2017 11/30/2016 1/20/2017	Robert MIchael Pickel Thomas M. Colley Residual Tr by and through its Trustee Citizens National Bank Krista Baldwin	O'Brien Resources, LLC O'Brien Resources, LLC O'Brien Resources, LLC	2017-00002184 Texas 2552-0215 Texas 2017-00004146 Texas	Henderson Anderson Henderson
42.213.15400.1379 42.213.15400.1380 42.213.15400.1381 42.213.15400.1390	1/17/2017 1/17/2017 11/30/2016 1/20/2017 2/2/2017	Robert MIchael Pickel Thomas M. Colley Residual Tr by and through its Trustee Citizens National Bank Krista Baldwin Larry Coker	O'Brien Resources, LLC O'Brien Resources, LLC O'Brien Resources, LLC O'Brien Resources, LLC	2017-00002184 Texas 2552-0215 Texas 2017-00004146 Texas 2559-0540 Texas	Henderson Anderson Henderson Anderson
42.213.15400.1379 42.213.15400.1380 42.213.15400.1381 42.213.15400.1390 42.213.15400.1391	1/17/2017 1/17/2017 11/30/2016 1/20/2017 2/2/2017 3/10/2015	Robert MIchael Pickel Thomas M. Colley Residual Tr by and through its Trustee Citizens National Bank Krista Baldwin Larry Coker Luther Wade Sides	O'Brien Resources, LLC	2017-00002184 Texas 2552-0215 Texas 2017-00004146 Texas 2559-0540 Texas 2017-00004171 Texas	Henderson Anderson Henderson Anderson Henderson
42.213.15400.1379 42.213.15400.1380 42.213.15400.1381 42.213.15400.1390 42.213.15400.1391 42.213.15400.1393	1/17/2017 1/17/2017 11/30/2016 1/20/2017 2/2/2017 3/10/2015 9/6/2016	Robert Michael Pickel Thomas M. Colley Residual Tr by and through its Trustee Citizens National Bank Krista Baldwin Larry Coker Luther Wade Sides Petra Zeigermann	O'Brien Resources, LLC McCutchin Genpar, LLC	2017-00002184 Texas 2552-0215 Texas 2017-00004146 Texas 2559-0540 Texas 2017-00004171 Texas 2015-00003648 Texas	Henderson Anderson Henderson Anderson Henderson
42.213.15400.1379 42.213.15400.1380 42.213.15400.1381 42.213.15400.1391 42.213.15400.1391 42.213.15400.1393 42.213.15400.1393	1/17/2017 1/17/2017 11/30/2016 1/20/2017 2/2/2017 3/10/2015 9/6/2016 2/15/2017	Robert MIchael Pickel Thomas M. Colley Residual Tr by and through its Trustee Citizens National Bank Krista Baldwin Larry Coker Luther Wade Sides Petra Zeigermann Patch Energy, LLC; Candlewood Resources, LLC; Preston Patry	O'Brien Resources, LLC McCutchin Genpar, LLC O'Brien Resources, LLC	2017-00002184 Texas 2552-0215 Texas 2017-00004146 Texas 2559-0540 Texas 2017-00004171 Texas 2015-00003648 Texas 2017-00003358 Texas	Henderson Anderson Henderson Anderson Henderson Henderson
42.213.15400.1379 42.213.15400.1380 42.213.15400.1381 42.213.15400.1391 42.213.15400.1391 42.213.15400.1393 42.213.15400.1393 42.213.15400.1394 42.213.15400.1395	1/17/2017 1/17/2017 11/30/2016 1/20/2017 2/2/2017 3/10/2015 9/6/2016 2/15/2017	Robert MIchael Pickel Thomas M. Colley Residual Tr by and through its Trustee Citizens National Bank Krista Baldwin Larry Coker Luther Wade Sides Petra Zeigermann Patch Energy, LLC; Candlewood Resources, LLC; Preston Patry Brenda Joyce Nikel	O'Brien Resources, LLC McCutchin Genpar, LLC O'Brien Resources, LLC	2017-00002184 Texas 2552-0215 Texas 2017-00004146 Texas 2559-0540 Texas 2017-00004171 Texas 2015-00003648 Texas 2017-00003358 Texas 2017-00004172 Texas	Henderson Anderson Henderson Henderson Henderson Henderson Henderson
42.213.15400.1379 42.213.15400.1380 42.213.15400.1381 42.213.15400.1390 42.213.15400.1391 42.213.15400.1393 42.213.15400.1394 42.213.15400.1395 42.213.15400.1396	1/17/2017 1/17/2017 11/30/2016 1/20/2017 2/2/2017 3/10/2015 9/6/2016 2/15/2017 12/14/2016 2/15/2017	Robert MIchael Pickel Thomas M. Colley Residual Tr by and through its Trustee Citizens National Bank Krista Baldwin Larry Coker Luther Wade Sides Petra Zeigermann Patch Energy, LLC; Candlewood Resources, LLC; Preston Patry Brenda Joyce Nikel Joshua German	O'Brien Resources, LLC McCutchin Genpar, LLC O'Brien Resources, LLC O'Brien Resources, LLC	2017-00002184 Texas 2552-0215 Texas 2017-00004146 Texas 2559-0540 Texas 2017-00004171 Texas 2015-00003648 Texas 2017-00003358 Texas 2017-00004172 Texas 2017-00004173 Texas	Henderson Anderson Henderson Henderson Henderson Henderson Henderson
42.213.15400.1389 42.213.15400.1380 42.213.15400.1381 42.213.15400.1391 42.213.15400.1391 42.213.15400.1393 42.213.15400.1393 42.213.15400.1395 42.213.15400.1396 42.213.15400.1397	1/17/2017 1/17/2017 11/30/2016 1/20/2017 2/2/2017 3/10/2015 9/6/2016 2/15/2017 12/14/2016 2/15/2017 2/15/2017	Robert Michael Pickel Thomas M. Colley Residual Tr by and through its Trustee Citizens National Bank Krista Baldwin Larry Coker Luther Wade Sides Petra Zeigermann Patch Energy, LLC; Candlewood Resources, LLC; Preston Patry Brenda Joyce Nikel Joshua German Laura Dianne Slate	O'Brien Resources, LLC McCutchin Genpar, LLC O'Brien Resources, LLC O'Brien Resources, LLC O'Brien Resources, LLC	2017-00002184 Texas 2552-0215 Texas 2017-00004146 Texas 2559-0540 Texas 2017-00004171 Texas 2015-00003648 Texas 2017-00003358 Texas 2017-00004172 Texas 2017-00004173 Texas 2017-00004174 Texas	Henderson Anderson Henderson Henderson Henderson Henderson Henderson Henderson
42.213.15400.1379 42.213.15400.1380 42.213.15400.1381 42.213.15400.1390 42.213.15400.1391 42.213.15400.1393 42.213.15400.1394 42.213.15400.1395 42.213.15400.1396 42.213.15400.1397 42.213.15400.1398	1/17/2017 1/17/2017 11/30/2016 1/20/2017 2/2/2017 3/10/2015 9/6/2016 2/15/2017 12/14/2016 2/15/2017 2/15/2017 2/15/2017	Robert Michael Pickel Thomas M. Colley Residual Tr by and through its Trustee Citizens National Bank Krista Baldwin Larry Coker Luther Wade Sides Petra Zeigermann Patch Energy, LLC; Candlewood Resources, LLC; Preston Patry Brenda Joyce Nikel Joshua German Laura Dianne Slate Vickie Elaine Cooper	O'Brien Resources, LLC McCutchin Genpar, LLC O'Brien Resources, LLC O'Brien Resources, LLC O'Brien Resources, LLC O'Brien Resources, LLC	2017-00002184 Texas 2552-0215 Texas 2017-00004146 Texas 2559-0540 Texas 2017-00004171 Texas 2015-00003648 Texas 2017-00003358 Texas 2017-00004172 Texas 2017-00004173 Texas 2017-00004174 Texas 2017-00004175 Texas	Henderson Anderson Henderson Henderson Henderson Henderson Henderson Henderson Henderson Henderson Henderson

Lease #	Date	Lessor	Lessee	Book-Page State	County
42.213.15400.1402	2/22/2017	Hollyne Clair Dublin	O'Brien Resources, LLC	2017-00006576 Texas	Henderson
42.213.15400.1403			O'Brien Resources, LLC	2017-00006570 Texas	Henderson
42.213.15400.1404			O'Brien Resources, LLC	2017-00006569 Texas	Henderson
42.213.15400.1406	3/6/2017		O'Brien Resources, LLC	2017-00006572 Texas	Henderson
42.213.15400.1407	4/7/2017	The Patricia Louise Clements Trust	O'Brien Resources, LLC	2017-00007227 Texas	Henderson
42.213.15400.1408		Tara Guzman	O'Brien Resources, LLC	2017-00006574 Texas	Hendeson
42.213.15400.1409 42.213.15400.1411		Betty Rhea Wade Fluegge Ronald Henderson	O'Brien Resources, LLC O'Brien Resources, LLC	2017-00006561 Texas 2017-00006562 Texas	Anderson Anderson
42.213.15400.1411		Charlotte J. West	O'Brien Resources, LLC	2574-0644 Texas	Anderson
42.213.15400.1413		Daniel Monroe Matthews	O'Brien Resources, LLC	2574-0651 Texas	Anderson
42.213.15400.1414		Charlotte J. West	O'Brien Resources, LLC	2574-0646 Texas	Anderson
42.213.15400.1415	9/27/2017	Daniel Monroe Matthews	O'Brien Resources, LLC	2574-0648 Texas	Anderson
42.213.15400.1416	9/26/2017	Edwina W. Huddleston	O'Brien Resources, LLC	2574-0641 Texas	Anderson
42.213.15400.1417	3/23/2017	PEC Minerals, LP (bad lease)	O'Brien Resources, LLC	2015-00015050 Texas	Henderson
42.213.15400.1418	3/23/2017	PEC Minerals, LP	O'Brien Resources, LLC	2017-00010903 Texas	Henderson
42.213.15400.1419	6/21/2017	Ronald L. Williams	O'Brien Resources, LLC	2017-00010907 Texas	Henderson
42.213.15400.1420	9/20/2016	Doyal Trey Henderson	O'Brien Resources, LLC	2017-00017563 Texas	Henderson
42.213.15400.1422	7/27/2017	Windsor Interests, Ltd.	O'Brien Resources, LLC	201-700017564 Texas	Henderson
42.213.15400.1423	7/27/2017	Windsor Interests, Ltd.	O'Brien Resources, LLC	2017-00017556 Texas	Henderson
42.213.15400.1424	3/22/2017	Sonja Lewis Simon	O'Brien Resorces, LLC	2017-00017559 Texas	Henderson
42.213.15400.1425	2/9/2017	Robert D. Johnston, County Judge, Receivership for J. W. Hanks et al Heirs Cause # DCCV16-611-349	O'Brien Resorces, LLC	2552-0693 Texas	Anderson
42.213.15400.1426	10/6/2016		O'Brien Resources, LLC	2536-0514 Texas	Anderson
42.213.15400.1428		Carl G. Hood, Carolyn F. Hood & State of Texas	O'Brien Resources, LLC	2018-00001285 Texas	Henderson
42.213.15400.1429 42.213.15400.1430		Carey L. Swinney, Virginia C. Swinney & State of Texas	O'Brien Resources, LLC	2018-00001286 Texas	Henderson
42.213.15400.1430		Daniel V. Heller & State of Texas Ricky B. Woolverton & State of Texas	O'Brien Resources, LLC O'Brien Resources, LLC	2018-00001287 Texas 2018-00001288 Texas	Henderson Henderson
42.213.15400.1431		Emmett W. Woolverton & State of Texas	O'Brien Resources, LLC	2018-00001289 Texas	Henderson
42.213.15400.1433		Teresa Woolverton & State of Texas	O'Brien Resources, LLC	2018-00001290 Texas	Henderson
42.213.15400.1434		Carl E. Baublet & State of Texas	O'Brien Resources, LLC	2018-00001291 Texas	Henderson
42.213.15400.1435		Tanya M. Blink & State of Texas	O'Brien Resources, LLC	2594-0746 Texas	Anderson
42.213.15400.1436	10/26/2017	Kasey Marie Reeves Garcia & State of Texas	O'Brien Resources, LLC	2594-0746 Texas	Henderson
42.213.15400.1437	1/12/2016	John Youngblood, Receiver Cause No. CV15 -0485-392, I.H. Lambright & Unknown Heirs et al	O'Brien Resources, LLC	2016-00000413 Texas	Henderson
42.213.15400.1438	11/30/2015	John Youngblood, Receiver Cause No. DCCV15-0322-392, E.A. Cummings & Unknown Heirs et al	O'Brien Resources, LLC	2015-0016263 Texas	Henderson
42.213.15400.1442	9/16/2016	Robert D. Johnston, Receiver for Mineral Interests Cause No. DCCV16-323-349. Unknown Heirs of Travis Williams, Kenneth	O'Brien Resources, LLC	2533-0492 Texas	Anderson
42.213.15400.1443	8/3/2016	Robert D. Johnston, Receiver for Mineral Interests Cause No. DCCV16-207-369. Unknown Heirs of Clifton Johnson and Ruth	O'Brien Resources, LLC	2527-0480 Texas	Anderson
42.213.15400.1444	5/19/2016	Robert D. Johnston, Receiver for Mineral Interests Cause No. DCCV16-092-369. Unknown Heirs of Angie Kale Hughes, Laura	O'Brien Resources, LLC	2519-0142 Texas	Anderson
42.213.15400.5000	4/1/2010	Donna J. Abernathy	Perryman Exploration	2010-00014050 Texas	Henderson
42.213.15400.5001	4/1/2010	David Abemathy	Perryman Exploration	2010-00005386 Texas	Henderson
42.213.15400.5002	4/1/2010	Herbert Abernathy (Deceased)	Perryman Exploration	2010-00014501 Texas	Henderson
42.213.15400.5003		P. Barbara G. Ackeret	Beaufort Operating, LLC	2357-0576 Texas	Anderson
42.213.15400.5004		Charles Leonard Adams	Beaufort Operating, LLC	2012-00014015 Texas	Henderson
42.213.15400.5005 42.213.15400.5006		Daisy Till Akin (dec'd) ! Ella Marie Albers	Beaufort Operating, LLC Beaufort Operating, LLC	2886-304 Texas 2357-0582 Texas	Henderson Anderson
42.213.15400.5010		Harold Randolph Bailey, Individually and as Executor of the Estate of Ruth Cotton Whiddon, Dec'd	Beaufort Operating, LLC	2012-00014044 Texas	Henderson
42.213.15400.5011		Richard T. Bailey (Staria J. Bailey)	Beaufort Operating, LLC	2012-00014073 Texas	Henderson
42.213.15400.5012		William Randolph Bailey	Beaufort Operating, LLC	2012-00014092 Texas	Henderson
42.213.15400.5013		Joan Marie Batson Bandsuch	Eiche, Mapes and Company,	2013-0003828 Texas	Henderson
42.213.15400.5014		Prittney M. Bann	Beaufort Operating, LLC	2357-0601 Texas	Anderson
42.213.15400.5015	8/22/2012	Linda Davis Barnes	Beaufort Operating, LLC	2338-0461 Texas	Anderson
42.213.15400.5016	11/1/2012	Sarah Louder Barnes	Beaufort Operating, LLC	2357-0573 Texas	Anderson
42.213.15400.5017	2/23/2005	Travis V. Batson, Jr.	Eiche, Mapes and Company,	2627-0195 Texas	Henderson
42.213.15400.5018	2/25/2015	Brook Baugh	O'Brien Resources, LLC	2015-00003847 Texas	Henderson
42.213.15400.5019	6/23/2008	David Wayne Bolt	Perryman Exploration	2886-299 Texas	Henderson
42.213.15400.5020	9/13/2012	Ernest C. Bryant, III	Beaufort Operating, LLC	2013-00012958 Texas	Henderson
42.213.15400.5021		Don Bullock and wife, Shirley Bullock	Perryman Exploration	2869-094 Texas	Henderson
42.213.15400.5022		Daniel Wilson Bywaters and wife, Leslie Dale Bywaters	Perryman Acquisitions 2010,	2010-00007590 Texas	Henderson
42.213.15400.5025		Mary Cameron, Heir of Loraine Carlson Sigler	Beaufort Operating, LLC	2012-00014062 Texas	Henderson
42.213.15400.5026		Johns E. Clark	Perryman Exploration	2010-00005384 Texas	Henderson
42.213.15400.5028		Cathy Cole	Beaufort Operating, LLC	2012-00014096 Texas	Henderson
42.213.15400.5029	8/23/2012		Beaufort Operating, LLC	2012-00014095 Texas	Henderson
42.213.15400.5031		Clay Cotton, Jr.	Beaufort Operating, LLC	2012-00014100 Texas	Henderson
42.213.15400.5032 42.213.15400.5033	10/2/2012 7/1/2008	Bernetta Curtis *W. G. Darsey, III	Beaufort Operating, LLC Perryman Exploration	2013-00012954 Texas 2894-772 Texas	Henderson Henderson
42.213.15400.5033		Charles Davis, Jr.	Beaufort Operating, LLC	2012-00014068 Texas	Henderson
42.213.15400.5035		Citizen National Bank, Successor Trustee of the Henry Ford Davis Trust, U/W/O Kathryn Milner Davis	Beaufort Operating, LLC	2338-0538 Texas	Anderson
		,	- F		

Lease #	Date	Lessor	Lessee	Book-Page State	County
42.213.15400.5036	8/22/2012	Kenneth D. Davis	Beaufort Operating, LLC	2012-00014067 Texas	Henderson
42.213.15400.5037	10/2/2012	Betty Demarest	Beaufort Operating, LLC	2013-00012952 Texas	Henderson
42.213.15400.5042	4/1/2010	Dereck G. Dodson	Perryman Exploration	2010-00005385 Texas	Henderson
42.213.15400.5043	11/11/2012	Damon Douglas	Beaufort Operating, LLC	2371-0429 Texas	Anderson
42.213.15400.5044	8/7/2012	Damon Douglas Trustee	Beaufort Operating, LLC	2012-00014019 Texas	Henderson
42.213.15400.5045	8/23/2012	June Dreyer	Beaufort Operating, LLC	2012-00014088 Texas	Henderson
42.213.15400.5046	1/1/2011	Easter Seals of Greater North Texas	Perryman Acquisitions 2010,	2011-00000911 Texas	Henderson
42.213.15400.5051	8/20/2012	Virginia Ewald	Beaufort Operating, LLC	2012-00014051 Texas	Henderson
42.213.15400.5052	8/23/2012	Hardy England	Beaufort Operating, LLC	2012-00014093 Texas	Henderson
42.213.15400.5053	11/11/2012	Jeanna L. France	Beaufort Operating, LLC	2357-0589 Texas	Anderson
42.213.15400.5054	10/8/2007	Stephen K. Friedrich and wife, Doree Friedrich	Perryman Exploration	2944-539 Texas	Henderson
42.213.15400.5055	9/21/2012	Brock Gibson	Beaufort Operating, LLC	2012-00014020 Texas	Henderson
42.213.15400.5056	10/15/2012	David Gibson	Beaufort Operating, LLC	2012-00014064 Texas	Henderson
42.213.15400.5057		Alan Giles (Duplicate Lease, see our 42.213.15400.0043 that covers this interest Lloyd Giles)	Eiche, Mapes and Company,	n/a Texas	
42.213.15400.5058	6/23/2004	Lowell Giles	Eiche, Mapes and Company,	2469-292 Texas	Henderson
42.213.15400.5059	9/12/2008	Gayle Hanna	Perryman Exploration	2010-00007589 Texas	Henderson
42.213.15400.5060	6/10/2012	John M. Harrison III and Mary S. Harrison Trustees of the Harrison Family Trust	Beaufort Operating, LLC	2338-0531 Texas	Anderson
42.213.15400.5061	4/30/2012	Robert Bryan Hash Family 2006 GST Exempt Trust	Beaufort Operating, LLC	2012-00014074 Texas	Henderson
42.213.15400.5062	4/30/2012	Steven Vincent Hash Family 2006 GST Exempt Trust	Beaufort Operating, LLC	2012-00014072 Texas	Henderson
42.213.15400.5063	11/11/2012	Brice E. Hedlund	Beaufort Operating, LLC	2357-0604 Texas	Anderson
42.213.15400.5064	5/8/2012	James Albert Hemminger	Beaufort Operating, LLC	2012-00014022 Texas	Henderson
42.213.15400.5065	8/23/2012	Ruth Ann Hendrick	Beaufort Operating, LLC	2012-00014091 Texas	Henderson
42.213.15400.5066	6/26/2012	Johanna Hermanson	Beaufort Operating, LLC	2012-00014061 Texas	Henderson
42.213.15400.5067	8/8/2012	Becky S. Jenkins Hill	Beaufort Operating, LLC	2012-00014023 Texas	Henderson
42.213.15400.5068	8/13/2012	Fayrene Hillhouse	Beaufort Operating, LLC	2371-0422 Texas	Anderson
42.213.15400.5069	11/14/2012	Fayrene Hillhouse	Beaufort Operating, LLC	2371-0420 Texas	Anderson
42.213.15400.5070	11/14/2012	Joleene Kale Hutchens	Beaufort Operating, LLC	2422-0513 Texas	Anderson
42.213.15400.5071	11/1/2012	Linda Louder Huckaby	Beaufort Operating, LLC	2357-0607 Texas	Anderson
42.213.15400.5073	5/8/2012	Annie Jenkins	Beaufort Operating, LLC	2012-00014024 Texas	Henderson
42.213.15400.5074	9/21/2012	Billy Ralph (Bill) Jenkins	Beaufort Operating, LLC	2012-00014024 Texas	Henderson
42.213.15400.5075	9/20/2012	Charles Ashley Jenkins	Beaufort Operating, LLC	2012-00014034 Texas	Henderson
42.213.15400.5076		Elfreida Jenkins, Trustee of the J.R. and E.R. Jenkins Living Trust	Beaufort Operating, LLC	2012-00014035 Texas	Henderson
42.213.15400.5077		Harry Albert Jenkins	Beaufort Operating, LLC	2012-00014038 Texas	Henderson
42.213.15400.5078	9/25/2012	Jason H. Jenkins	Beaufort Operating, LLC	2014-00009003 Texas	Henderson
42.213.15400.5079		Ottis Ray Jenkins	Beaufort Operating, LLC	2014-00009744 Texas	Henderson
42.213.15400.5080		Reagan E. Jenkins, Jr.	Beaufort Operating, LLC	2012-00014039 Texas	Henderson
42.213.15400.5081		Wyley Herbert Jenkins	Beaufort Operating, LLC	2012-00014037 Texas	Henderson
42.213.15400.5083		Debra Jordan	O'Brien Resources, LLC	2015-00006836 Texas	Henderson
42.213.15400.5084		Elsie Kale	Beaufort Operating, LLC	2357-0598 Texas	Anderson
42.213.15400.5085		Florence L. Kale	O'Brien Resources, LLC	2515-691 Texas	Anderson
42.213.15400.5089		Kathryn Louella Tipton Kelly (dec'd)	Perryman Exploration	2875-629 Texas	Henderson
42.213.15400.5090		Kathryn Louella Tipton Kelly (dec'd)	Perryman Exploration	2894-766 Texas	Henderson
42.213.15400.5091		Aline Lambright (Deceased) Life Estate	Perryman Exploration	2903-191 Texas	Henderson
42.213.15400.5092	8/20/2008		Perryman Exploration	2903-195 Texas	Henderson
42.213.15400.5093	7/6/2004	Debbie Lambright	Eiche, Mapes and Company,	2469-280 Texas	Henderson
42.213.15400.5094		Debbie Lambright	Eiche, Mapes and Company,	2568-758 Texas	Anderson
42.213.15400.5095	8/20/2008		Perryman Exploration	2903-196 Texas	Henderson
42.213.15400.5096		Lenell Lambright and Bradley Scott Lambright	Beaufort Operating, LLC	2338-0529 Texas	Anderson
42.213.15400.5097		Leslie Lambright	Eiche, Mapes and Company,	2469-283 Texas	Anderson
42.213.15400.5098		Lucille M. Jenkins Latronica (dec'd)	Beaufort Operating, LLC	2012-00014025 Texas	Henderson
42.213.15400.5099		B.G. Lewis	Beaufort Operating, LLC	2357-0570 Texas	Anderson
42.213.15400.5100		Richard L. Lewis	Beaufort Operating, LLC	2357-0567 Texas	Anderson
42.213.15400.5102		Floy E. Little (now deceased)	Perryman Exploration	2010-00014049 Texas	Henderson
42.213.15400.5103		Doris Mallory	Beaufort Operating, LLC	2371-0427 Texas	Anderson
42.213.15400.5105		Ronald Lewis McElroy	O'Brien Resources, LLC	2015-00006846 Texas	Henderson
42.213.15400.5108		David McNair	Beaufort Operating, LLC	2012-00014071 Texas	Henderson
42.213.15400.5111		Patricia Miller	Perryman Exploration	2938-520 Texas	Henderson
42.213.15400.5113		Sterling Miller	Beaufort Operating, LLC	2012-00014043 Texas	Henderson
42.213.15400.5114		Roy D. Mills and wife, Kandy Mills	Beaufort Operating, LLC	2013-00012948 Texas	Henderson
42.213.15400.5115		Dana Mitchell	O'Brien Resources, LLC	2016-00012768 Texas	Henderson
42.213.15400.5117		Wayne Mitchell	Beaufort Operating, LLC	2371-0383 Texas	Anderson
42.213.15400.5118		James E. Morris	Beaufort Operating, LLC	2357-0595 Texas	Anderson
42.213.15400.5119		Suzanne Murphy	Beaufort Operating, LLC	2012-00014060 Texas	Henderson
42.213.15400.5120		Sherrie Negrete	Perryman Exploration	2010-00005382 Texas	Henderson
42.213.15400.5121	8/8/2012	Elizabeth W. Neill	Beaufort Operating, LLC	2012-00014057 Texas	Henderson

Lease #	Date	Lessor	Lessee	Book-Page State	County
42.213.15400.5122	6/6/2012	Neill Oil, LLC	Beaufort Operating, LLC	2013-00012955 Texas	Henderson
42.213.15400.5124	8/8/2012	Walter R. Neill, Jr.	Beaufort Operating, LLC	2012-00014040 Texas	Henderson
42.213.15400.5125	11/11/2012	Retha J. Nelson	Beaufort Operating, LLC	2357-0585 Texas	Anderson
42.213.15400.5127	11/11/2012	Sharon K. Olson	Beaufort Operating, LLC	2357-0579 Texas	Anderson
42.213.15400.5128	7/31/2012	Owen Oil & Gas Properties, LLC	Beaufort Operating, LLC	2012-00014099 Texas	Henderson
42.213.15400.5129	8/8/2012	Marianne Parish	Beaufort Operating, LLC	2012-00014041 Texas	Henderson
42.213.15400.5130	8/7/2012	Robert W. Percival	Beaufort Operating, LLC	2371-393 Texas	Anderson
42.213.15400.5132	8/8/2012	Carol Maude Perkins	Beaufort Operating, LLC	2012-00014042 Texas	Henderson
42.213.15400.5133	8/8/2012	Sherill Jan Jenkins Poff	Beaufort Operating, LLC	2012-00014026 Texas	Henderson
42.213.15400.5134	1/31/2012	Charles Lesley Price	Beaufort Operating, LLC	2338-0535 Texas	Anderson
42.213.15400.5135	8/8/2012	Roy E. Pulis and wife, Carolyn S. Pulis	Beaufort Operating, LLC	2012-00014027 Texas	Henderson
42.213.15400.5137	8/8/2012	Karleen Neill Raper	Beaufort Operating, LLC	2012-00014063 Texas	Henderson
42.213.15400.5139		Eddith G. Reynolds Trust Childrens Medical Center Dallas; William C. Herndon, Trustee	Beaufort Operating, LLC	2012-00014103 Texas	Henderson
42.213.15400.5141		Milly Schaer	Beaufort Operating, LLC	2012-00014089 Texas	Henderson
42.213.15400.5142	8/25/2008		Beaufort Operating, LLC	2905-437 Texas	Henderson
42.213.15400.5143	9/10/2008		Beaufort Operating, LLC	2011-00000912 Texas	Henderson
42.213.15400.5144 42.213.15400.5145		Julie Jenkins Shaw Donna Gail Shelby	Beaufort Operating, LLC	2014-00009004 Texas	Henderson
42.213.15400.5146		Sloan Shuffler, et vir and Clifton Edward New	Perryman Exploration Beaufort Operating, LLC	2010-00003378 Texas 2357-0588 Texas	Henderson Anderson
42.213.15400.5147		Evelyn Giles Sims	Eiche, Mapes and Company,	2469-286 Texas	Anderson
42.213.15400.5147		Elise Eilenberger Smith	Beaufort Operating, LLC	2012-00014029 Texas	Henderson
42.213.15400.5150	6/11/2012		Beaufort Operating, LLC	2338-0540 Texas	Anderson
42.213.15400.5151		Irene Stafford	Beaufort Operating, LLC	2371-0405 Texas	Anderson
42.213.15400.5152		Irene Stafford	Beaufort Operating, LLC	2371-0410 Texas	Anderson
42.213.15400.5154		Ralph V. St. John	Beaufort Operating, LLC	2012-00014030 Texas	Henderson
42.213.15400.5156		Marilyn Hemminger Talley	Beaufort Operating, LLC	2012-00014036 Texas	Henderson
42.213.15400.5157		Dorothy Kale Terry	Beaufort Operating, LLC	2466-490 Texas	Anderson
42.213.15400.5158		Steven Terry, Ind. & as Exec. of the Estate of Shige Kale, Dec'd	Beaufort Operating, LLC	2371-0389 Texas	Anderson
42.213.15400.5159		Robert Tidgren	Beaufort Operating, LLC	2012-00014059 Texas	Henderson
42.213.15400.5160		Carl Tidgren	Beaufort Operating, LLC	2012-00014058 Texas	Henderson
42.213.15400.5162	6/23/2008	Shelia Marie Thornton	Perryman Exploration	2886-300 Texas	Henderson
42.213.15400.5163	10/4/2012	Jimmie L. Till, AlF Randall Till, Deceased	Beaufort Operating, LLC	2012-00014014 Texas	Henderson
42.213.15400.5166	10/2/2012	Elizabeth Everett Troyer	Beaufort Operating, LLC	2013-00012953 Texas	Henderson
42.213.15400.5168	9/26/2012	Jennifer Allison Jenkins Turner	Beaufort Operating, LLC	2014-00009002 Texas	Henderson
42.213.15400.5169	8/3/2012	Kyle S. Turner and wife, Kimberly Turner	Beaufort Operating, LLC	2371-0397 Texas	Anderson
42.213.15400.5170	4/1/2010	Jackie Watson	Perryman Exploration	2010-00005383 Texas	Henderson
42.213.15400.5171	8/23/2012	Sharon England Weinbeg	Beaufort Operating, LLC	2012-00014094 Texas	Henderson
42.213.15400.5172	5/24/2012	The Mark V West Testamentary Tr c/o Texas Bank & Tr Company	Beaufort Operating, LLC	2012-00014032 Texas	Henderson
42.213.15400.5173	3/19/2015	Rebecca D. Westbrook	O'Brien Resources, LLC	2015-00006854 Texas	Henderson
42.213.15400.5174	8/23/2012	Thomas Eugene Whiddon	Beaufort Operating, LLC	2012-00014090 Texas	Henderson
42.213.15400.5175		Jerry Mitchell Wilbanks	Beaufort Operating, LLC	2371-0401 Texas	Anderson
42.213.15400.5176	8/6/2012	Judee Bailey Williams (Deceased)	Beaufort Operating, LLC	2012-00014070 Texas	Henderson
42.213.15400.5178		Norma Williams	Beaufort Operating, LLC	2012-00014052 Texas	Henderson
42.213.15400.5179	7/1/2008	Betty Darsey Woods	Perryman Exploration	2894-773 Texas	Henderson
42.213.15400.5180	2/15/2010		Perryman Exploration	2010-00003379 Texas	Henderson
42.213.15400.5181	9/7/2012	Mary Jean Turner	Beaufort Operating, LLC	2012-00014047 Texas	Henderson
42.213.15400.5182	9/7/2012	Paulette Flannery	Beaufort Operating, LLC	2012-00014048 Texas	Henderson
42.213.15400.5183 42.213.15400.5184	9/7/2012 9/7/2012	Johanna Flannery Denson, Ind. & as Exec. of the Estate of Clara Malinda Flannery Clara Mart (decid)	Beaufort Operating, LLC Beaufort Operating, LLC	2012-00014050 Texas 2012-00014045 Texas	Henderson Henderson
42.213.15400.5185		Clara West (dec'd) Malinda F. Arrant	Beaufort Operating, LLC Beaufort Operating, LLC	2012-00014049 Texas	Henderson
42.213.15400.5186		Dan R. Murchison	Beaufort Operating, LLC	2357-0610 Texas	Anderson
42.213.15400.5187		James A. Murchison, Sr.	Beaufort Operating, LLC	2371-387 Texas	Anderson
42.067.28100.0001		The Barbara K. Converse Living Trust, et al	O'Brien Resources, LLC	2017004431 Texas	Cass
42.067.28100.0002		Steven R. Rainey	O'Brien Resources, LLC	2017004432 Texas	Cass
42.067.28100.0003		Pruitt Land & Minerals	O'Brien Resources, LLC	2018000104 Texas	Cass
42.067.28100.0004		Angela Pruitt Jones	O'Brien Resources, LLC	2017004899 Texas	Cass
42.067.28100.0005		Jennifer Jones McGee	O'Brien Resources, LLC	2017004901 Texas	Cass
42.067.28100.0006		Geoffrey Pruitt Jones	O'Brien Resources, LLC	2017004897 Texas	Cass
42.067.28100.0007		Charles Edmond Pruitt III	O'Brien Resources, LLC	2017004898 Texas	Cass
42.067.28100.0008		Charles E. Pruitt	O'Brien Resources, LLC	2017004900 Texas	Cass
42.067.28100.0009		Mary A. Darley	Petro-Land Group, Inc.	2014000898 Texas	Cass
42.067.28100.0010	8/28/2013		Petro-Land Group, Inc.	2014000734 Texas	Cass
42.067.28100.0011	9/6/2013	Elizabeth Morris Cohan	Petro-Land Group, Inc.	2014000899 Texas	Cass
42.067.28100.0012	8/28/2013	Roger Norwood Smith	Petro-Land Group, Inc.	2014000733 Texas	Cass
42.067.28100.0013	8/28/2013	Andrew H. Bemis	Petro-Land Group, Inc.	2014000731 Texas	Cass

Lease #	Date	Lessor	Lessee	Book-Page	State	County
42.067.28100.0014	9/3/2013	Elizabeth Carrington Campos	Petro-Land Group, Inc.	2014000737		Cass
42.067.28100.0015	9/3/2013	Virginia Jo Beaver	Petro-Land Group, Inc.	2014000736	Texas	Cass
42.067.28100.0016	10/2/2013	The Katherine B. Piper Trust	Petro-Land Group, Inc.	2013001665	Texas	Cass
42.067.28100.0017	10/1/2013	John D. Putnam	Petro-Land Group, Inc.	2014000902	Texas	Cass
42.067.28100.0018	9/5/2013	Michael B. Smith	Petro-Land Group, Inc.	2014000740	Texas	Cass
42.067.28100.0019	9/4/2013	Ann Smith Downs	Petro-Land Group, Inc.	2014000738	Texas	Cass
42.067.28100.0020	8/28/2013	Duncan David Smith	Petro-Land Group, Inc.	2014000732	Texas	Cass
42.067.28100.0021	8/28/2013	Harold S. Bemis	Petro-Land Group, Inc.	2014000730	Texas	Cass
42.067.28100.0022	10/29/2013	Mary Darsie Alexander	Petro-Land Group, Inc.	2014000931	Texas	Cass
42.067.28100.0023	4/4/2014	Big Sky Mineral Trust	Petro-Land Group, Inc.	2014002095		Cass
42.067.28100.0024	5/1/2014	Jessica Bemis Ward	Petro-Land Group, Inc.	2014002580		Cass
42.067.28100.0025	5/1/2014	James H. Bemis III	Petro-Land Group, Inc.	2014002583		Cass
42.067.28100.0026		Edward D. Bemis	Petro-Land Group, Inc.	2014000964		Cass
42.067.28100.0027 42.067.28100.0028	5/1/2014	Peter Morrell Hale	Petro-Land Group, Inc. Petro-Land Group, Inc.	2014002581 2014000844	Texas	Cass
42.067.28100.0029	5/2/2014	Ann Avery Kelley Kathryn Schneebeck Ottenjohn	Petro-Land Group, Inc.	2014000544		Cass
42.067.28100.0030			Petro-Land Group, Inc.	2014000715		Cass
42.067.28100.0031		Frederick D. Avery Jr	Petro-Land Group, Inc.	2014000845		Cass
42.067.28100.0032		Barbara Duke O&G	Petro-Land Group, Inc.	2014002585		Cass
42.067.28100.0033		Cynthia Darsie	Petro-Land Group, Inc.	2014002088		Cass
42.067.28100.0034	3/12/2014	TKO Royalty Partners	Petro-Land Group, Inc.	2014001600	Texas	Cass
42.067.28100.0035	5/1/2014	Edward E. Prewitt	Petro-Land Group, Inc.	2014002582	Texas	Cass
42.067.28100.0036	2/27/2014	Judith Pribor Revocable Trust	Petro-Land Group, Inc.	2014001777	Texas	Cass
42.067.28100.0037	5/1/2014	Huetta Norvelle Prewitt	Petro-Land Group, Inc.	2014002593	Texas	Cass
42.067.28100.0038	5/1/2014	Douglas Knox Bemis Jr	Petro-Land Group, Inc.	2014002584	Texas	Cass
42.067.28100.0039	5/1/2014	Elizabeth Bemis Prewitt	Petro-Land Group, Inc.	2014002595	Texas	Cass
42.067.28100.0040	5/1/2014	Thomas Rutherford Cravens	Petro-Land Group, Inc.	2014002591	Texas	Cass
42.067.28100.0041	5/1/2014	Charlene Koger	Petro-Land Group, Inc.	2014002921		Cass
42.067.28100.0042	4/4/2014	William Bennett Cravens	Petro-Land Group, Inc.	2014002085		Cass
42.067.28100.0043	5/1/2014	Meredith DeWoody	Petro-Land Group, Inc.	2014002917		Cass
42.067.28100.0044	5/1/2014	Wharton Sparks Cravens	Petro-Land Group, Inc.	2014002919		Cass
42.067.28100.0045 42.067.28100.0046	5/1/2014 5/1/2014	Charles Fadjo Cravens Robert Pledger	Petro-Land Group, Inc. Petro-Land Group, Inc.	2014002915 2014003189		Cass
42.067.28100.0047	5/1/2014	Katherine McRae Hale	Petro-Land Group, Inc.	2014003103		Cass
42.067.28100.0048	5/1/2014	David Anthony Cravens	Petro-Land Group, Inc.	2014002920		Cass
42.067.28100.0049		Frank Pledger	Petro-Land Group, Inc.	2014003185		Cass
42.067.28100.0050	8/20/2018	Tonya Mitchell	O'Brien Resources, LLC	2018005262	Texas	Cass
42.067.28100.0051	10/1/2013	Marita Jane Bemis Parrish	Petro-Land Group, Inc.	2014000765	Texas	Cass
42.067.28100.0052	11/28/2017	James D. Mohr	O'Brien Resources, LLC	2017004893	Texas	Cass
42.067.28100.0053	12/6/2017	Thomas Garard Smith	O'Brien Resources, LLC	2018000106	Texas	Cass
42.067.28100.0054	12/6/2017	Janis Lee Burkett	O'Brien Resources, LLC	2018000107	Texas	Cass
42.067.28100.0055	12/8/2017	Jana Daniels Davis	O'Brien Resources, LLC	2018000276	Texas	Cass
42.067.28100.0056	12/8/2017	Mary Jane Caver	O'Brien Resources, LLC	2018000381	Texas	Cass
42.067.28100.0057	12/8/2017	Romie I. Caver	O'Brien Resources, LLC	2017004903	Texas	Cass
42.067.28100.0058		Morris A. George	O'Brien Resources, LLC	2018000108		Cass
42.067.28100.0059		Sarah George Partain	O'Brien Resources, LLC	2018000096		Cass
42.067.28100.0060		Suzanne Ullom Johnson	O'Brien Resources, LLC	2018000275		Cass
42.067.28100.0061		Thomas Edward George	O'Brien Resources, LLC	2018000109		Cass
42.067.28100.0062		Peggy George Allen	O'Brien Resources, LLC O'Brien Resources, LLC	2018000101		Cass
42.067.28100.0063 42.067.28100.0064		Todd E. Fitts, et al Angela S. Hufhines, et al	O'Brien Resources, LLC O'Brien Resources, LLC	2018000102 2018000105		Cass
42.067.28100.0064 42.067.28100.0065		Angela S. Humines, et al Ken Andrus	O'Brien Resources, LLC O'Brien Resources, LLC	2018000105		Cass
42.067.28100.0067		Katie E. Nielsen	O'Brien Resources, LLC	2018000111		Cass
42.067.28100.0068		Faye G. Landham	O'Brien Resources, LLC	2018000111		Cass
42.067.28100.0069		Laurie Farrall	O'Brien Resources, LLC	2018000114		Cass
42.067.28100.0070		Roger Norwood Smith	O'Brien Resources, LLC	2018000098		Cass
42.067.28100.0071		Michael B. Smith	O'Brien Resources, LLC	2018000110		Cass
42.067.28100.0072		Ann Smith Downs	O'Brien Resources, LLC	2018000097		Cass
42.067.28100.0073		Duncan David Smith	O'Brien Resources, LLC	2018000094	Texas	Cass
42.067.28100.0074	11/28/2017	Bret Crutcher	O'Brien Resources, LLC	2018000095	Texas	Cass
42.067.28100.0075	11/28/2017	Jack R. Parker, Jr.	O'Brien Resources, LLC	2017004894	Texas	Cass
42.067.28100.0076	11/28/2017	Christopher B. Coats	O'Brien Resources, LLC	2018000099	Texas	Cass
42.067.28100.0077	12/8/2017	Noelle Newton Schrader	O'Brien Resources, LLC	2018000274	Texas	Cass
42.067.28100.0078	11/28/2017	Penni Parker Mitchell	O'Brien Resources, LLC	2018000100	Texas	Cass
42.067.28100.0079	12/13/2017	Brandon Morse	O'Brien Resources, LLC	2018000112	Texas	Cass

Lease #	Date	Lessor	Lessee	Book-Page	State	County
42.067.28100.0080		Brandon Crutcher	O'Brien Resources, LLC	2018000273		Cass
42.067.28100.0081		Elizabeth Morris Cohan	O'Brien Resources, LLC	2018000380	Texas	Cass
42.067.28100.0082	1/10/2018	Mary A. Darley	O'Brien Resources, LLC	2018000382	Texas	Cass
42.067.28100.0087	10/24/2018	Stephanie Clift	O'Brien Resources, LLC	2019000642	Texas	Cass
42.067.28100.0096	11/30/2018	Keith P. Smith	O'Brien Resources, LLC	2018006379	Texas	Cass
42.067.28100.0097	11/30/2018	Laurie S. Rhyne	O'Brien Resources, LLC	2018006387	Texas	Cass
42.067.28100.0098	11/30/2018	Julie D. Canty	O'Brien Resources, LLC	2018006386	Texas	Cass
42.067.28100.0102	11/30/2018	Robert M. Clift, Jr.	O'Brien Resources, LLC	2019000018	Texas	Cass
42.067.28100.0103		Mary Jane Cole	O'Brien Resources, LLC	2018006381		Cass
42.067.28100.0104		Barbara Francis	O'Brien Resources, LLC	2019000020		Cass
42.067.28100.0105		Lindsay Clift	O'Brien Resources, LLC	2018006378		Cass
42.067.28100.0107		Nancy N. Morris	O'Brien Resources, LLC	2018001404		Cass
42.067.28100.0108		Thomas K. Nichols, Jr.	O'Brien Resources, LLC	2018001406		Cass
42.067.28100.0109 42.067.28100.0111		Diane N. Rogers Carrie Lou Sheppard	O'Brien Resources, LLC O'Brien Resources, LLC	2018001405 2018001408		Cass
42.067.28100.0111			O'Brien Resources, LLC	2018001408		Cass
42.067.28100.0113	3/22/2018	James L. Hanks	O'Brien Resources, LLC	2018001409		Cass
42.067.28100.0114		Ruby F. Hanks Samuel	O'Brien Resources, LLC	2018001411		Cass
42.067.28100.0115		Floretta Hanks Jackson	O'Brien Resources, LLC	2018001412		Cass
42.067.28100.0116		Robert Lee Warren	O'Brien Resources, LLC	2018001971		Cass
42.067.28100.0117		Ronald Lynn Newland	O'Brien Resources, LLC	2019000404		Cass
42.067.28100.0120		Henry A. King, Jr.	O'Brien Resources, LLC	2018001403		Cass
42.067.28100.0141		Rebecca Thurber Byars	O'Brien Resources, LLC	2019001609	Texas	Cass
42.067.28100.0142	2/20/2018	Elizabeth S. Thurber	O'Brien Resources, LLC	2019001608	Texas	Cass
42.067.28100.0143	3/22/2018	Marie Louise Powell	O'Brien Resources, LLC	2018001407	Texas	Cass
42.067.28100.0144	4/13/2018	John A. Howe	O'Brien Resources, LLC	2018001973	Texas	Cass
42.067.28100.0145	4/12/2018	David Carlow Howe	O'Brien Resources, LLC	2018001974	Texas	Cass
42.067.28100.0146	5/2/2018	Donald Howe	O'Brien Resources, LLC	2018001972	Texas	Cass
42.067.28100.0147	1/11/2019	Deborah Newland Fair	O'Brien Resources, LLC	2019000405	Texas	Cass
42.067.28100.0148	3/27/2019	Annelle Newland Mulcahy	O'Brien Resources, LLC	2019002576	Texas	Cass
42.067.28100.0153	10/25/2018	Julie White Ray	O'Brien Resources, LLC	2018006143	Texas	Cass
42.067.28100.0154	10/29/2018	Sandra Kay Stewart Mann	O'Brien Resources, LLC	2018006147	Texas	Cass
42.067.28100.0168		Sarah Martha Puckett	O'Brien Resources, LLC	2019002578		Cass
42.067.28100.0169		Wesley O. Shelton	Petro-Land Group, Inc.	2014003855		Cass
42.067.28100.0170		Rose Ann Bleiler	Petro-Land Group, Inc.	2014003846		Cass
42.067.28100.0171		James Venoy Peeler	Petro-Land Group, Inc.	2014003849		Cass
42.067.28100.0172 42.067.28100.0173		Wesley O. Shelton	Petro-Land Group, Inc. Petro-Land Group, Inc.	2014003850 2014003851	Texas	Cass
42.067.28100.0174		David Cecil Keys Carol Keys Worsham F/K/A Carol Keys	Petro-Land Group, Inc.	2014003831		Cass
42.067.28100.0175		Kelly Keys Elkins F/k/A Kelly Keys	Petro-Land Group, Inc.	2014004260		Cass
42.067.28100.0176		Mark David Keys	Petro-Land Group, Inc.	2014004261		Cass
42.067.28100.0177		Molly Joan Thompson	Petro-Land Group, Inc.	2014005940		Cass
42.067.28100.0178		Georgia Martin	Petro-Land Group, Inc.	2014005937		Cass
42.067.28100.0179		Ramona Stratton	Petro-Land Group, Inc.	2014005938		Cass
42.067.28100.0180	9/23/2014	Billie Jean Hansen	Petro-Land Group, Inc.	2014005939	Texas	Cass
42.067.28100.0181	10/29/2014	Jason Scott Zervos F/K/A Jason Scott Fela	Petro-Land Group, Inc.	2014006453	Texas	Cass
42.067.28100.0182	2/25/2015	Beverly McReynolds Alexander	Petro-Land Group, Inc.	2015001440	Texas	Cass
42.067.28100.0183	3/4/1959	M.M. Turener, et al	George E. Jinkins	369/107	Texas	Cass
42.067.28100.0184	3/11/1959	James Woods Powers, et ux	George E. Jinkins	369/117	Texas	Cass
42.067.28100.0185	3/12/1959	R.S. Meadows, Jr., et al	George E. Jinkins	369/352	Texas	Cass
42.067.28100.0186	2/7/1959	Ella Mae Ann Probst Wilson, et al	George E. Jinkins	369/102	Texas	Cass
42.067.28100.0187	3/16/1959	Lewis Buick, Inc.	George E. Jinkins	369/45	Texas	Cass
42.067.28100.0188		Claire Marshall, et al	L.F. Brothers	253/174	Texas	Cass
42.067.28100.0189		Leila Lemmon, et vir	L.F. Brothers	253/182	Texas	Cass
42.067.28100.0190		H.G. Jaynes, et ux	L.F. Brothers	253/186	Texas	Cass
42.067.28100.0191		E.L. Jaynes, et ux	L.F. Brothers	253/258	Texas	Cass
42.067.28100.0192		Vernon Jaynes, et ux	L.F. Brothers	253/262	Texas	Cass
42.067.28100.0193		Mary Badgett, et vir	L.F. Brothers	253/264	Texas	Cass
42.067.28100.0194		Manly W. Jaynes, et ux	L.F. Brothers	253/270	Texas	Cass
42.067.28100.0195		Troy A. Jaynes, et ux	L.F. Brothers	253/279	Texas	Cass
42.067.28100.0196		Claire Marshall, et al	L.F. Brothers	253/443	Texas	Cass
42.067.28100.0197 42.067.28100.0198		Janie Carter Morrison, et al I.R. Lemmon, et al	Durbin Bond L.F. Brothers	254/217 254/220	Texas	Cass
42.067.28100.0198		W.S. Sylvest, et ux	L.F. Brothers	254/225	Texas	Cass
42.067.28100.0200		O.W. Lemmon, et ux	L.F. Brothers	254/237	Texas	Cass
00720100.0200	5,20,1348			237/231	. UNUS	0433

L	ease #	Date	Lessor	Lessee	Book-Page	State	County
4	2.067.28100.0201	5/28/1949	R.H. Lemmon, et al	L.F. Brothers	254/241	Texas	Cass
4	2.067.28100.0202	5/28/1949	J.M. Lemmon, et ux	L.F. Brothers	254/245	Texas	Cass
4	2.067.28100.0203	5/28/1949	A.S. Lemmon, et ux	L.F. Brothers	254/249	Texas	Cass
4	2.067.28100.0204	6/29/1951	Nettie Mae Lemmon	Sam Williams, et al	263/344	Texas	Cass
4	2.067.28100.0205	12/8/1955	Sam B. Hall	C.A. McLean	324/388	Texas	Cass
4	2.067.28100.0206	10/20/2018	Lynn Fouche Correa	O'Brien Resources, LLC	2018006377	Texas	Cass
4	2.067.28100.0207	10/25/2018	Jesse C. Deese & Carol Deese, h/w	O'Brien Resources, LLC	2019000270	Texas	Cass
4	2.067.28100.0209	11/30/2018	Faye Gibbons Landham	O'Brien Resources, LLC	2019000022	Texas	Cass
4	2.067.28100.0210	10/20/2018	Adeline Old Vaill Trust	O'Brien Resources, LLC	2019001805	Texas	Cass
4	2.067.28100.0211	11/11/2018	Dianne Harrist Whatley	O'Brien Resources, LLC	2019000019	Texas	Cass
4	2.067.28100.0212	10/20/2018	Patricia Ann Walker Burgoyne	O'Brien Resources, LLC	2019000271	Texas	Cass
4	2.067.28100.0213	9/20/2018	Charles R. Peacock	O'Brien Resources, LLC	2019000272	Texas	Cass
4	2.067.28100.0214	8/27/2018	Scotty C. Prince	O'Brien Resources, LLC	2019000274	Texas	Cass
4	2.067.28100.0215	12/5/2018	Todd Phillip Fouche	O'Brien Resources, LLC	2019000267	Texas	Cass
4	2.067.28100.0216	12/5/2018	James Andrew Fouche	O'Brien Resources, LLC	2019000269	Texas	Cass
4	2.067.28100.0217	10/15/2018	James D. Runnels & Regina Roberta Runnels, h/w	O'Brien Resources, LLC	2019000268	Texas	Cass
4	2.067.28100.0218	1/11/2019	Ronnie L. Jinkins	O'Brien Resources, LLC	2019000276	Texas	Cass
4	2.067.28100.0219	4/17/2019	Melinda Watkins Miller	O'Brien Resources, LLC	2019000275	Texas	Cass
4	2.067.28100.0220	1/11/2019	Rita Sue Kranson Williams	O'Brien Resources, LLC	2019000402	Texas	Cass
4	2.067.28100.0221	1/11/2019	Randall Proctor West	O'Brien Resources, LLC	2019000273	Texas	Cass
4	2.067.28100.0222	1/15/2019	Violet Marie Howe	O'Brien Resources, LLC	2019001078	Texas	Cass
4	2.067.28100.0223	1/11/2019	Wilma Jean Peacock Klaus	O'Brien Resources, LLC	2019000641	Texas	Cass
4	2.067.28100.0224	1/31/2019	Virgil Neal Peacock	O'Brien Resources, LLC	2019000643	Texas	Cass
4	2.067.28100.0225	1/11/2019	Cheryl Ann Perry	O'Brien Resources, LLC	2019000403	Texas	Cass
4	2.067.28100.0226	1/31/2019	Kenneth W. Peacock	O'Brien Resources, LLC	2019000644	Texas	Cass
4	2.067.28100.0227	1/11/2019	Wynette West Stone	O'Brien Resources, LLC	2019000645	Texas	Cass
4	2.067.28100.0228	1/11/2019	Estate of Amanda Morse Albright, Deceased	O'Brien Resources, LLC	2019001077	Texas	Cass
4	2.067.28100.0229	11/1/2018	Phyllis M. Sperling f/k/a Phyllis Mitchell	O'Brien Resources, LLC	2019001251	Texas	Cass
4	2.067.28100.0230	3/18/2019	D.W. and J.A. Stewart Living Trust dated November 19, 2018	O'Brien Resources, LLC	2019001627	Texas	Cass
4	2.067.28100.0231	3/27/2019	Christi R. Nutt	O'Brien Resources, LLC	2019001618	Texas	Cass
4	2.067.28100.0232	3/27/2019	John W. Nutt and wife, Christi Nutt	O'Brien Resources, LLC	2019001617	Texas	Cass
4	2.067.28100.0233	3/27/2019	Keith D. Kiles	O'Brien Resources, LLC	2019001621	Texas	Cass
4	2.067.28100.0234	3/27/2019	John Frederic Self	O'Brien Resources, LLC	2019001622	Texas	Cass
4	2.067.28100.0235	3/27/2019	Denise Harper	O'Brien Resources, LLC	2019001626	Texas	Cass
4	2.067.28100.0236	3/27/2019	Mark Harper	O'Brien Resources, LLC	2019001623	Texas	Cass
4	2.067.28100.0237	4/17/2019	Judith Ann Andrus Robbins	O'Brien Resources, LLC	2019001624	Texas	Cass
4	2.067.28100.0238	3/27/2019	Jean Wyrick	O'Brien Resources, LLC	2019001620	Texas	Cass
4	2.067.28100.0239	3/27/2019	Patricia Jerome	O'Brien Resources, LLC	2019001625	Texas	Cass
4	2.067.28100.0240	3/27/2019	Michael L. Schutz	O'Brien Resources, LLC	2019001616	Texas	Cass
4	2.067.28100.0241	3/27/2019	Larry E. Waters	O'Brien Resources, LLC	2019001812	Texas	Cass
4	2.067.28100.0242	3/27/2019	Ann Hooper Stacy	O'Brien Resources, LLC	2019001814	Texas	Cass
4	2.067.28100.0243	3/27/2019	Kathryn Jean Selgrath	O'Brien Resources, LLC	2019001813	Texas	Cass
4	2.067.28100.0244	3/27/2019	Linda Carol Hooper	O'Brien Resources, LLC	2019001811	Texas	Cass
4	2.067.28100.0245	3/27/2019	Frances Loris Lott	O'Brien Resources, LLC	2019001818	Texas	Cass
4	2.067.28100.0246	3/27/2019	Toby Lynn Williams	O'Brien Resources, LLC	2019001619	Texas	Cass
4	2.067.28100.0247	3/27/2019	Clayton Bataille	O'Brien Resources, LLC	2019001809	Texas	Cass
4	2.067.28100.0248	3/27/2019	Don R. Yates	O'Brien Resources, LLC	2019001810	Texas	Cass
4	2.067.28100.0249	3/27/2019	Kenneth E. Stewart	O'Brien Resources, LLC	2019001807	Texas	Cass
4	2.067.28100.0250	3/27/2019	Anna S. Parker	O'Brien Resources, LLC	2019001806	Texas	Cass
4	2.067.28100.0251	3/27/2019	James Michael Elrod	O'Brien Resources, LLC	2019001925	Texas	Cass
4	2.067.28100.0252	3/27/2019	Robert Cherry	O'Brien Resources, LLC	2019001915	Texas	Cass
4	2.067.28100.0253	3/27/2019	The Thomas Family Trust	O'Brien Resources, LLC	2019001914	Texas	Cass
4	2.067.28100.0254	3/27/2019	Elizabeth A. Rindt	O'Brien Resources, LLC	2019001911	Texas	Cass
4	2.067.28100.0255	3/27/2019	Mary Jane Bradshaw Beauchamp	O'Brien Resources, LLC	2019001921	Texas	Cass
4	2.067.28100.0256	3/27/2019	Jack H. Harper	O'Brien Resources, LLC	2019001924	Texas	Cass
4	2.067.28100.0257	3/27/2019	Ruth B. Cigainero,	O'Brien Resources, LLC	2019001922	Texas	Cass
4	2.067.28100.0258	3/27/2019	Elanore S. Boone	O'Brien Resources, LLC	2019001923	Texas	Cass
4	2.067.28100.0259	3/27/2019	Mary Grisham	O'Brien Resources, LLC	2019001913	Texas	Cass
4	2.067.28100.0260	3/27/2019	John Scott Hammonds	O'Brien Resources, LLC	2019001615	Texas	Cass
4	2.067.28100.0261	4/15/2019	John Pierce Ellis	O'Brien Resources, LLC	2019001910	Texas	Cass
4	2.067.28100.0262	3/27/2019	Deborah Koemer	O'Brien Resources, LLC	2019002052	Texas	Cass
4	2.067.28100.0263	3/27/2019	Yvette Bataille	O'Brien Resources, LLC	2019002051	Texas	Cass
4	2.067.28100.0264	4/25/2019	Hershel Randall Hanner Residuary Trust	O'Brien Resources, LLC	2019002271	Texas	Cass
4	2.067.28100.0265	3/27/2019	James Tural Kiles, Jr.	O'Brien Resources, LLC	2019002055	Texas	Cass
4	2.067.28100.0266	3/27/2019	Lynn Esmay a/k/a Evelyn Elizabeth Youngs Esmay f/k/a Evelyn Elizabeth Johnston	O'Brien Resources, LLC	2019002054	Texas	Cass

Lease #	Date	Lessor	Lessee	Book-Page	State	County
42.067.28100.0267	4/15/2019	David L. Dollins	O'Brien Resources, LLC	2019002053		Cass
42.067.28100.0268	3/27/2019	Michael Harper	O'Brien Resources, LLC	2019002050	Texas	Cass
42.067.28100.0269	3/27/2019	Guy C. Alloway	O'Brien Resources, LLC	2019002059	Texas	Cass
42.067.28100.0270	3/27/2019	Patricia Clement Cole	O'Brien Resources, LLC	2019002268	Texas	Cass
42.067.28100.0271	4/15/2019	Scotty C. Prince	O'Brien Resources, LLC	2019002056	Texas	Cass
42.067.28100.0272	3/27/2019	Linda Marie Waters	O'Brien Resources, LLC	2019002057	Texas	Cass
42.067.28100.0273	4/15/2019	George Oen Dollins	O'Brien Resources, LLC	2019002061	Texas	Cass
42.067.28100.0274	3/27/2019	Christine Dildy Fail	O'Brien Resources, LLC	2019002058	Texas	Cass
42.067.28100.0275	4/25/2019	David M. Oney	O'Brien Resources, LLC	2019002270	Texas	Cass
42.067.28100.0276	3/27/2019	Gerald W. Jeffery	O'Brien Resources, LLC	2019002269	Texas	Cass
42.067.28100.0277		Horseshoe Minerals, Inc.	O'Brien Resources, LLC	2019002049		Cass
42.067.28100.0278		Reba Diane Gordon	O'Brien Resources, LLC	2019002273		Cass
42.067.28100.0279		Estate of Akemi T. McGlothlin, Deceased	O'Brien Resources, LLC	2019002272		Cass
42.067.28100.0280		James Douglas Eastham and wife, Phillis Audrey Eastham	O'Brien Resources, LLC O'Brien Resources, LLC	2019002274		Cass
42.067.28100.0281 42.067.28100.0282		Dustin Lee Mason Puth Appetto Cook Mellor et al.	O'Brien Resources, LLC	2019002275 2019002305	Texas	Cass
42.067.28100.0283		Ruth Annette Cook Walker, et al Dewayne F. Yates, Jr.	O'Brien Resources, LLC	2019002303		Cass
42.067.28100.0284		Betty Lou Wall, Individually and as Executrix of the Estate of Boyd Wall, Deceased	O'Brien Resources, LLC	2019002270		Cass
42.067.28100.0285	3/27/2019		O'Brien Resources, LLC	2019002384	Texas	Cass
42.067.28100.0286		Ran-Man Limited Partnership, a Texas Limited Partnership dated October 11, 1999 with Bobbie L. Hardy as General Partner	O'Brien Resources, LLC		Texas	Cass
42.067.28100.0287		John Randall King	O'Brien Resources, LLC	2019002382		Cass
42.067.28100.0288		Mike L. Albers	O'Brien Resources, LLC			Cass
42.067.28100.0289	5/7/2019	Kevin Walker	O'Brien Resources, LLC	2019002580	Texas	Cass
42.067.28100.0290	5/14/2019	Ree Hale Almon	O'Brien Resources, LLC	2019002586	Texas	Cass
42.067.28100.0291	5/17/2019	Robert Bataille	O'Brien Resources, LLC	2019002581	Texas	Cass
42.067.28100.0292	5/14/2019	Jan Hanner Knowles	O'Brien Resources, LLC	2019002584	Texas	Cass
42.067.28100.0293	5/14/2019	Jan Hanner Knowles	O'Brien Resources, LLC	2019002585	Texas	Cass
42.067.28100.0295	5/28/2019	Methodist Children's Home	O'Brien Resources, LLC	2019002706	Texas	Cass
42.067.28100.0296	5/28/2019	Southwestern University	O'Brien Resources, LLC	2019002705	Texas	Cass
42.067.28100.0297	3/27/2019	Charles Daniel Baker	O'Brien Resources, LLC		Texas	Cass
42.067.28100.0298	3/27/2019	Margaret Ruth Torrans	O'Brien Resources, LLC	2019002577	Texas	Cass
42.067.28100.0299		Theresia M. Johns	O'Brien Resources, LLC	2019002583	Texas	Cass
42.067.28100.0300	5/7/2019	Linda L. Curry	O'Brien Resources, LLC	2019002579		Cass
42.067.28100.0301		Milton Thomas O'Rear, II	O'Brien Resources, LLC	2019002808		Cass
42.067.28100.0302		Robert Wall	O'Brien Resources, LLC O'Brien Resources, LLC	2019002810	Texas	Cass
42.067.28100.0303 42.067.28100.0304		Sarah Wall Trant L. Kidd	O'Brien Resources, LLC	2019002809	Texas	Cass
42.067.28100.0305		Owen B. Whatley	O'Brien Resources, LLC	2019002797	Texas	Cass
42.067.28100.0306			O'Brien Resources, LLC	2019002935	Texas	Cass
42.067.28100.0307	5/21/2019		O'Brien Resources, LLC	2019002936	Texas	Cass
42.067.28100.0308	5/21/2019	Sharon C. Thomason	O'Brien Resources, LLC	2019002932	Texas	Cass
42.067.28100.0309	5/21/2019	Cathy Lou Clements	O'Brien Resources, LLC	2019002933	Texas	Cass
42.067.28100.0310	5/21/2019	Jay S. Clements, Jr.	O'Brien Resources, LLC	2019002937	Texas	Cass
42.067.28100.0311	5/21/2019	Charlotte Clements Major	O'Brien Resources, LLC	2019002934	Texas	Cass
42.067.28100.0312	6/5/2019	Franklin Ray Lawrence, Jr.	O'Brien Resources, LLC	2019002795	Texas	Cass
42.067.28100.0313	6/4/2019	Charles Ray Peacock	O'Brien Resources, LLC	2019002796	Texas	Cass
42.067.28100.0431	5/15/2018	Pruitt Land & Minerals	O'Brien Resources, LLC	2018002610	Texas	Cass
42.067.28100.0433		Robert Lee Warren	O'Brien Resources, LLC	2018002603		Cass
42.067.28100.0435		The Lovelace Family Real Estate Trust, Joe B. Lovelace, Trustee	O'Brien Resources, LLC			Cass
42.067.28100.0436		Kenneth N. Brown	O'Brien Resources, LLC	2018002605		Cass
42.067.28100.0437		David Shields Hough	O'Brien Resources, LLC	2018003179		Cass
42.067.28100.0438		Glennie S. Allen	O'Brien Resources, LLC O'Brien Resources, LLC	2018003870		Cass
42.067.28100.0439		Daniel H. Florence	O'Brien Resources, LLC	2018002608 2018002609		Cass
42.067.28100.0440 42.067.28100.0441		Pollyanna Florence Southerland Bobbye J. Wendell	O'Brien Resources, LLC	2018002609		Cass
42.067.28100.0442		Noah A. Blizzard	O'Brien Resources, LLC			Cass
42.067.28100.0443		The Monelle J. Hardy Trust	O'Brien Resources, LLC	2018003176		Cass
42.067.28100.0444		Charles R. Peacock	O'Brien Resources, LLC	2018003882		Cass
42.067.28100.0445		Phillip Hayes	O'Brien Resources, LLC	2018004206		Cass
42.067.28100.0446		Billie M. Ridge	O'Brien Resources, LLC	2018003180		Cass
42.067.28100.0447	6/25/2018	Sowilla Ann Newland Mills	O'Brien Resources, LLC	2018004207	Texas	Cass
42.067.28100.0448	6/5/2018	Finis Mitchell, Jr. and Rosemary Mitchell, h/w	O'Brien Resources, LLC	2018003174	Texas	Cass
42.067.28100.0449	6/5/2018	Billy J. Ayers and Bonnie R. Ayers, husband and wife	O'Brien Resources, LLC	2018002604	Texas	Cass
42.067.28100.0450	6/5/2018	Bryan Ayers	O'Brien Resources, LLC	2018002614	Texas	Cass
42.067.28100.0452	5/25/2018	Jerry F. Daniel and Barbara Ann Daniel, h/w	O'Brien Resources, LLC	2018002611	Texas	Cass

Lease #	Date	Lessor	Lessee	Book-Page	State	County
42.067.28100.0453	6/5/2018	Murphy D. Mitchell and Shirley Mitchell, h/w, and Murphy D. Mitchell, Jr.	O'Brien Resources, LLC	2018003178		Cass
42.067.28100.0454	6/11/2018	Lottie Mitchell Creer	O'Brien Resources, LLC	2018003175	Texas	Cass
42.067.28100.0455	6/15/2018	Barbara S. McKittrick w/ life estate	O'Brien Resources, LLC	2018003869	Texas	Cass
42.067.28100.0456	6/15/2018	Barbara S. McKittrick	O'Brien Resources, LLC	2018003870	Texas	Cass
42.067.28100.0458	11/30/2018	Katie Gibbons Nielsen	O'Brien Resources, LLC	2019000023	Texas	Cass
42.067.28100.0460	7/18/2018	Patricia H. Florence & Estate of Buck Florence	O'Brien Resources, LLC	2018003880	Texas	Cass
42.067.28100.0461	6/25/2018	C.R. Newland, III	O'Brien Resources, LLC	2018003871	Texas	Cass
42.067.28100.0462	6/25/2018	Philip Keith Newland	O'Brien Resources, LLC	2018003872	Texas	Cass
42.067.28100.0463	6/4/2018	Lawrence Mitchell	O'Brien Resources, LLC	2018003873	Texas	Cass
42.067.28100.0464	6/4/2018	Joby Wallace	O'Brien Resources, LLC	2018003874	Texas	Cass
42.067.28100.0465	6/15/2018	Lewis D. Holcomb and Kathy T. Holcomb	O'Brien Resources, LLC	2018003875	Texas	Cass
42.067.28100.0466	6/20/2018	Norma Reynolds Allen & The Jack C. Allen Family Trust	O'Brien Resources, LLC	2018003876	Texas	Cass
42.067.28100.0467	7/18/2018	Carolyn Brooks Bonin	O'Brien Resources, LLC	2018003877	Texas	Cass
42.067.28100.0468	7/25/2018	Mabeleen Elizabeth Mitchell	O'Brien Resources, LLC	2018003878	Texas	Cass
42.067.28100.0469	7/25/2018	Lincoln L. Butts	O'Brien Resources, LLC	2018003879	Texas	Cass
42.067.28100.0476	7/18/2018	Barbara Brooks Goodwin	O'Brien Resources, LLC	2018003881	Texas	Cass
42.067.28100.0482		Benton Douglas Tolleson	O'Brien Resources, LLC	2018005264	Texas	Cass
42.067.28100.0483		Pruitt Land & Minerals	O'Brien Resources, LLC			Cass
42.067.28100.0484	9/20/2018	Kenneth Hayes	O'Brien Resources, LLC	2018005265	Texas	Cass
42.067.28100.0485	9/19/2018	David Carlow Howe	O'Brien Resources, LLC	2018005266	Texas	Cass
42.067.28100.0486	9/20/2018		O'Brien Resources, LLC	2018005267	Texas	Cass
42.067.28100.0487	9/20/2018	Kenneth W. Peacock	O'Brien Resources, LLC	2018005449		Cass
42.067.28100.0488		Virginia Diane Tolleson Leuba	O'Brien Resources, LLC	2018005268		Cass
42.067.28100.0489	9/1/2018	Brooks K. McCord	O'Brien Resources, LLC	2018004739		Cass
42.067.28100.0490	9/1/2018	Cynthia Brooks McCord	O'Brien Resources, LLC	2018004454		Cass
42.067.28100.0491	9/1/2018	Christen McCord Sutherland	O'Brien Resources, LLC	2018004455		Cass
42.067.28100.0492		Cheryl Kale	O'Brien Resources, LLC	2018005274		Cass
42.067.28100.0493		Charles Michael Cromer	O'Brien Resources, LLC	2018005273		Cass
42.067.28100.0494		John A. Howe	O'Brien Resources, LLC	2018005275		Cass
42.067.28100.0495		John Davis McCoy	O'Brien Resources, LLC	2018005272		Cass
42.067.28100.0496	10/9/2018		O'Brien Resources, LLC	2018005440		Cass
42.067.28100.0497 42.067.28100.0498		Carol Carney Nasits	O'Brien Resources, LLC O'Brien Resources, LLC	2018005439 2018005276		Cass
42.067.28100.0499		Penny Cromer Hudson Betty Lou Burton Adams	O'Brien Resources, LLC	2018005277		Cass
42.067.28100.0500		Hardy O. Johnson	O'Brien Resources, LLC	2018005269		Cass
42.067.28100.0501		Dianne Harrist Whatley	O'Brien Resources, LLC	2018005271		Cass
42.067.28100.0502		James Morgan McCoy, III	O'Brien Resources, LLC	2018005270		Cass
42.067.28100.0503		Thomas A. King, III	O'Brien Resources, LLC	2018005279		Cass
42.067.28100.0504		Odie Lauree Harris	O'Brien Resources, LLC	2018005278		Cass
42.067.28100.0505		Wynette J. Atkins a/k/a Wynette Holtman Atkins	O'Brien Resources, LLC	2018005280		Cass
42.067.28100.0506		Rebecca Wiggins Watson a/k/a Rebecca Sue Watson	O'Brien Resources, LLC	2018005281	Texas	Cass
42.067.28100.0507	9/20/2018	Donna Kay Smith	O'Brien Resources, LLC	2018005283	Texas	Cass
42.067.28100.0508	9/20/2018	Rebecca Dean Malone	O'Brien Resources, LLC	2018005282	Texas	Cass
42.067.28100.0509	10/4/2018	Carson Dwayne Whatley	O'Brien Resources, LLC	2018005451	Texas	Cass
42.067.28100.0510	10/4/2018	Mary Lynn McDonald	O'Brien Resources, LLC	2018005453	Texas	Cass
42.067.28100.0511	10/15/2018	Barbara Peacock Weems	O'Brien Resources, LLC	2018005434	Texas	Cass
42.067.28100.0512	10/15/2018	Eunice Peacock Travis	O'Brien Resources, LLC	2018005435	Texas	Cass
42.067.28100.0513	10/15/2018	Jerry William Peacock	O'Brien Resources, LLC	2018005433	Texas	Cass
42.067.28100.0514	10/15/2018	Jimmy Royce Peacock	O'Brien Resources, LLC	2018005437	Texas	Cass
42.067.28100.0515	10/15/2018	Mary Louise Peacock Goodnow	O'Brien Resources, LLC	2018005436	Texas	Cass
42.067.28100.0516	10/15/2018	Vicky Peacock Keith	O'Brien Resources, LLC	2018005438	Texas	Cass
42.067.28100.0517	10/5/2018	Melvin Lee Holtman	O'Brien Resources, LLC	2018005452	Texas	Cass
42.067.28100.0518	10/4/2018	Margaret Ann Whatley Hall	O'Brien Resources, LLC	2018005455	Texas	Cass
42.067.28100.0519	10/20/2018	Robert I. Allday, etal	O'Brien Resources, LLC	2018005634	Texas	Cass
42.067.28100.0520	4/17/2019	Kyra Joan Clift Bowman	O'Brien Resources, LLC	2018006146	Texas	Cass
42.067.28100.0521	10/20/2018	Loyd Wesley Judd, III	O'Brien Resources, LLC	2018006144	Texas	Cass
42.067.28100.0522	9/20/2018	Rebecca Diane Almand Fairl a/k/a Rebecca D. Fairl	O'Brien Resources, LLC	2018006145	Texas	Cass
42.067.28100.0529	10/20/2018	Thomas A. King, III	O'Brien Resources, LLC	2018005446	Texas	Cass
42.067.28100.0530	9/20/2018	Tommy Ray Smith	O'Brien Resources, LLC	2018005454	Texas	Cass
42.067.28100.0531	10/20/2018	Craig Bohuslav and Debra Bohuslav, h/w	O'Brien Resources, LLC	2018005442	Texas	Cass
42.067.28100.0532	10/20/2018	Craig Bohuslav and Debra Bohuslav, h/w	O'Brien Resources, LLC	2018005441	Texas	Cass
42.067.28100.0533	10/20/2018	The Jack M. "Fuzzy" Harmon Trust	O'Brien Resources, LLC	2018005432	Texas	Cass
42.067.28100.0534		Marrita Jan W. Hamilton	O'Brien Resources, LLC	2018005450		Cass
42.067.28100.0535		Jacquie Ann Judd Daugherty	O'Brien Resources, LLC	2018005635		Cass
42.067.28100.0536	10/20/2018	Thomas K. Nichols, Jr.	O'Brien Resources, LLC	2018005448	Texas	Cass

Lease #	Date	Lessor	Lessee	Book-Page	State	County
42.067.28100.0537	10/15/2018	Deanna D. OlsonNichols	O'Brien Resources, LLC	2018005447	Texas	Cass
42.067.28100.0538	10/24/2018	Virgil Neal Peacock	O'Brien Resources, LLC	2018005445	Texas	Cass
42.067.28100.0539	10/5/2018	Annette J. Vecchio	O'Brien Resources, LLC	2018005444	Texas	Cass
42.067.28100.0540	10/20/2018	Nancy N. Morris	O'Brien Resources, LLC	2018005641	Texas	Cass
42.067.28100.0541	10/10/2018	Gayla Settoon, aka Gayla Theresa Settoon	O'Brien Resources, LLC	2018005443	Texas	Cass
42.067.28100.0542	10/31/2018	Anthony Family Property Holdings, Inc.	O'Brien Resources, LLC	2018005636	Texas	Cass
42.067.28100.0543	10/29/2018	D.W. and J.A. Stewart Living Trust dated November 19, 2018	O'Brien Resources, LLC	2018005639	Texas	Cass
42.067.28100.0544	10/10/2018	Shayne M. King f/k/a Shayne M. Kay	O'Brien Resources, LLC	2018005638	Texas	Cass
42.067.28100.0545	10/20/2018	Diane N. Rogers	O'Brien Resources, LLC	2018005642	Texas	Cass
42.067.28100.0546	9/20/2018	Donna Smith Woody, f/k/a Donna Fern Smith	O'Brien Resources, LLC	2018005640	Texas	Cass
42.067.28100.0547		Donald Howe	O'Brien Resources, LLC	2018005637	Texas	Cass
42.067.28100.0548		Pamela S. Wassom	O'Brien Resources, LLC		Texas	Cass
42.067.28100.0549		Norman Michael Walker	O'Brien Resources, LLC	2018006382		Cass
42.067.28100.0550		Susan S. Bedier	O'Brien Resources, LLC		Texas	Cass
42.067.28100.0551		Georgiana S. Kremer	O'Brien Resources, LLC	2018006384		Cass
42.067.28100.0552		Julia S. Mays	O'Brien Resources, LLC	2018006383		Cass
42.067.28100.0553		Deanne D. Trodden	O'Brien Resources, LLC		Texas	Cass
42.067.28100.0554		David A. Deaton	O'Brien Resources, LLC	2018006141		Cass
42.067.28100.0557		David N. Walls a/k/a David Nathan Walls	O'Brien Resources, LLC	2019000021		Cass
42.067.28100.0558		Leonard E. Hough, III	O'Brien Resources, LLC	2018006388		Cass
42.067.28100.9000		Carolyn Brooks Bonin	Stroud Exploration Company,		Texas	Cass
42.067.28100.9001		Barbara Brooks Goodwin	Stroud Exploration Company,	2012004362		Cass
42.067.28100.9002 42.067.28100.9003		Patricia H. Florence and Thomas Turner, exec of Estate of Buck Florence	Stroud Exploration Company,		Texas	Cass
42.067.28100.9007		North American Mission Board of Southern Baptist Convention	Stroud Exploration Company, Stroud Exploration Company,	2012005927		Cass
42.067.28100.9007		Cynthia Brooks McCord Brooks K. McCord	Stroud Exploration Company,	2012005925		Cass
42.067.28100.9008		Christen McCord Sutherland	Stroud Exploration Company,		Texas	Cass
42.067.28100.9011		The Executive Board of the Louisiana Baptist Convention	Stroud Exploration Company,	2013000025		Cass
42.067.28100.9013		S & P Co a Louisiana Partnership represented by Fred Phillips	Stroud Exploration Company,	2013003683		Cass
42.067.28100.9014		Charles H. Johnson and Estelle McDuff Johnson	Samedan Oil Corporation		Texas	Cass
42.067.28100.9015	9/18/1985		Samedan Oil Corporation		Texas	Cass
42.067.28100.9016		Whitaker Petroleum, L.L.C.	Stroud Exploration Company,		Texas	Cass
42.067.28100.9017		Johnson Investments, LLC	Stroud Exploration Company,	2014001057		Cass
42.067.28100.9018	3/3/2014	Tyrette May Cowan	Stroud Exploration Company,	2014001442		Cass
42.067.28100.9019	3/3/2014	George R. May	Stroud Exploration Company,	2014001443	Texas	Cass
42.067.28100.9020	3/3/2014	Terri May Carlton	Stroud Exploration Company,	2014001444	Texas	Cass
42.067.28100.9021	3/3/2014	Robert R. May	Stroud Exploration Company,	2014001445	Texas	Cass
42.067.28100.9022	3/17/2014	Margaret Ann Ward	Stroud Exploration Company,	2014001867	Texas	Cass
42.067.28100.9023	4/7/2014	Beckyann White Cook	Stroud Exploration Company,	2014001986	Texas	Cass
42.067.28100.9024	3/6/2014	Edward Beauregard Parker	Stroud Exploration Company,	2014002163	Texas	Cass
42.067.28100.9025	3/6/2014	William Beauregard Parker, III	Stroud Exploration Company,	2014002164	Texas	Cass
42.067.28100.9026	3/6/2014	Lauren Elizabeth Parker Humphrey	Stroud Exploration Company,	2014002159	Texas	Cass
42.067.28100.9027	3/24/2014	James W. Caldwell	Stroud Exploration Company,	2014002160	Texas	Cass
42.067.28100.9028	3/24/2014	Elise Wells Caldwell	Stroud Exploration Company,	2014002161	Texas	Cass
42.067.28100.9029	3/6/2014	Sheila Michelle Parker McCathran	Stroud Exploration Company,	2014002162	Texas	Cass
42.067.28100.9030	4/7/2014	Daniel Davis White	Stroud Exploration Company,	2014002156	Texas	Cass
42.067.28100.9032	8/25/2014	Christopher Scott Simmons	Stroud Exploration Company,	2014004944	Texas	Cass
42.067.28100.9033	8/25/2014	Daniel Bradford Terral	Stroud Exploration Company,	2014004941	Texas	Cass
42.067.28100.9034	9/8/2014	Forrest Davis Terral	Stroud Exploration Company,	2014004943	Texas	Cass
42.067.28100.9035	8/26/2014	Thomas Morris Terral	Stroud Exploration Company,	2014004942	Texas	Cass
42.067.28100.9037	3/24/2015	Muslow Oil & Gas, Inc.	Stroud Exploration Company,	2015002056	Texas	Cass
42.067.28100.9038	3/23/2015	TIP, Inc.	Stroud Exploration Company,	2015002714	Texas	Cass
42.067.28100.9039	5/29/2015	Becky Wilbanks, Receiver for Andy Elrod or the unknown heirs of Andy Elrod, deceased in Cause No. 15C125, Fifth Judicial Dist Court of	Stroud Exploration Company,	2015003206	Texas	Cass
42.067.28100.9040	9/29/2015	Carruth Heirs, LLC	Stroud Exploration Company,	2015005925	Texas	Cass
42.067.28100.9041	4/8/1985	Juanita K. Keel Indivand as Ind Exec and Sole Devisee Under the Will of J. D. Keel, Deceased	TXP Operating Company	0745-0079	Texas	Cass
42.067.28100.9042	3/13/1989		TXP Operating Company		Texas	Cass
42.067.28100.9043		A & L Oil Company	TXP Operating Company		Texas	Cass
42.067.28100.9044		Idris Davis White	TXP Operating Company		Texas	Cass
42.067.28100.9045		Thomas W. Keith, Sr., Indiv & as Exec of the Est of Juanita Beasley, Deceased; Ronnie Dean Keith, Donnie Eugene Keith; Thomas	TXP Operating Company		Texas	Cass
42.067.28100.9046	4/8/1985	James Burton Steed aka J. B. Steed	TXP Operating Company		Texas	Cass
42.067.28100.9047		H. Blume Johnson	TXP Operating Company		Texas	Cass
42.067.28100.9048		David Crow	TXP Operating Company		Texas	Cass
42.067.28100.9049		Muslow Oil & Gas, Inc.	TXP Operating Company		Texas	Cass
42.067.28100.9050	3/13/1989		TXP Operating Company		Texas	Cass
42.067.28100.9051	3/13/1989	Kerr-McGee Corp.	TXP Operating Company	0852-0672	Texas	Cass

Lease #	Date	Lessor	Lessee	Book-Page \$	State	County
42.067.28100.9052	9/12/1989	Ruth Caldwell Self	Samedan Oil Corporation		Texas	Cass
42.067.28100.9053	10/10/1989	Ludeweka Butler McCleery	Samedan Oil Corporation	0864-0496	Texas	Cass
42.067.28100.9054	9/18/1989	Robert D. Caldwell	Samedan Oil Corporation	0864-0492	Texas	Cass
42.067.28100.9055	5/15/1985	First Baptist Church of Vivian, Louisiana by and through its duly appointed Trustees	TXP Operating Company	0746-0451	Texas	Cass
42.067.28100.9056	9/29/1989	Ruth Lurline Grubbs, Indiv and Ind Executrix Under the will of Claude L. Grubbs, Jr., deceased	Samedan Oil Corporation	0864-0496	Texas	Cass
42.067.28100.9057	9/26/1989	Allen Lee McDuff	Samedan Oil Corporation	0864-0498	Texas	Cass
42.067.28100.9058	10/10/1989	Fred Logan Reagor, Jr	Samedan Oil Corporation	0864-0502	Texas	Cass
42.067.28100.9059		O. Kenneth Hickman	TXP Operating Company		Texas	Cass
42.067.28100.9060	7/17/1990	Jane Flaitz Burks DeYoung	Samedan Oil Corporation		Texas	Cass
42.067.28100.9061	9/9/1964	Rosalie Price Nichols and T. K. Nichols	Geo. E. Jinkins		Texas	Cass
42.067.28100.9062 42.067.28100.9063		V. A. Richey, Mozelle Richey Smith a Widow; T. L. Richey	Geo. E. Jinkins		Texas	Cass
42.067.28100.9063		Verda M. Rawles, et al Tom N. Cope, Receiver, acting under order of the District Court of Cass County, Texas in Cause No. 18,529, styled George E. Jinkins vs.	Geo E. Jenkins Geo E. Jenkins		Texas Texas	Cass
42.067.28100.9065		T. L. Richey	Geo E. Jenkins		Texas	Cass
42.067.28100.9066		Courtney Stewart	Geo E. Jenkins		Texas	Cass
42.067.28100.9067	3/16/1964	Hartwell C. Elkins	Geo E. Jenkins		Texas	Cass
42.067.28100.9068		Alline E. Womack and husband, E. E. Womack	Geo E. Jenkins		Texas	Cass
42.067.28100.9069		Opal E. Childress, and husband L. C. Childress	Geo E. Jenkins		Texas	Cass
42.067.28100.9070	11/24/1964	Gladys Elkins King	Geo E. Jenkins	0465-0379	Texas	Cass
42.067.28100.9071	2/20/1965	Clyde Festervand and Curtis Festervand	Geo E. Jenkins	0466-0645	Texas	Cass
42.067.28100.9072	2/20/1965	Lula Festervan, widow of Howard N. Festervan Sr., deceased; Patricia Festervan; Jimmy Festervan; Jerry Festervan; Howard Festervan	Geo E. Jenkins	0473-0225	Texas	Cass
42.067.28100.9073	12/6/1965	Clara Furr Hamilton, surviving widow of W. L. Hamilton	Geo. E. Jinkins	0475-0227	Texas	Cass
42.067.28100.9074	2/25/1964	R. C. Hardy	Geo E. Jenkins	0451-0013	Texas	Cass
42.067.28100.9077	8/10/2006	Jim N. Heath	Steve Coleman	2006005822	Texas	Cass
42.067.28100.9079	8/10/2006	Donald Leon Smith	Steve Coleman	2006005824	Texas	Cass
42.067.28100.9080	7/10/2006	Philip A. Starcke, Trustee of Oliver Andrew Starcke Revocable Trust	Steve Coleman	2006005825	Texas	Cass
42.067.28100.9081		Anna June Starcke Trust, Irene A. Moore or John A. Starcke Trustee	Steve Coleman		Texas	Cass
42.067.28100.9082		Thomas Turner, Estate of Buck Florence, and Patricia H. Florence	Steve Coleman		Texas	Cass
42.067.28100.9083	4/8/1985	Earlieen Lloyd	TXP Operating Company		Texas	Cass
42.067.28100.9084		Marilyn Minnette Flaitz Knolle	Samedan Oil Corporation		Texas	Cass
42.067.28100.9085 42.067.28100.9086		William Burroughs curator of the Interdict, D. Thomason also known as Dorris Thomason Lewell C. Butler, Jr.	TXP Operating Company Samedan Oil Corporation		Texas Texas	Cass
42.067.28100.9087		Edward Craig Hickman	TXP Operating Company		Texas	Cass
42.067.28100.9088	5/8/1985	Woman's Missionary Union	TXP Operating Company		Texas	Cass
42.067.28100.9089		Margaret H. Guyton	TXP Operating Company		Texas	Cass
42.067.28100.9090	9/5/1989	Daniel Caldwell; Robert C. Caldwell, Michael R. Caldwell; Charles H. Caldwell; Gwendolyn C. Melancon and Nowella C. Caldwell, a widow	Samedan Oil Corporation	0864-0490	Texas	Cass
42.067.28100.9091	9/9/1964	Lillie Johnson Price, widow of R. H. Price, deceased	Geo. E. Jinkins	0459-0422	Texas	Cass
42.067.28100.9092	8/12/2006	Patricia H. Florence and Jacquelyn W. Goodson, Co-Trustees of the Vernon R. Hampton Trust, benefit of Judy Hampton Ennis,	Steve Coleman	2006005828	Texas	Cass
42.067.28100.9093	10/10/2012	Marion Katherine McCraw	Stroud Exploration Company,	2012005934	Texas	Cass
42.067.28100.9097	9/26/1989	Daniel Ryan McDuff	Samedan Oil Corporation	0864-0500	Texas	Cass
42.257.15500.0004	9/27/2011	M & I Jackson Properties, Ltd	O'Brien Resources, LLC	4034-050	Texas	Kaufman
42.257.15500.0005	8/30/2011	Judy Grace Bryant	O'Brien Resources, LLC	4034-035	Texas	Kaufman
42.257.15500.0006		Douglass Frederick Wygal, ssp	O'Brien Resources, LLC	4034-040	Texas	Kaufman
42.257.15500.0008	9/28/2011	Early Douglas Lane, Jr. et ux	O'Brien Resources, LLC		Texas	Kaufman
42.257.15500.0009		The Estate of Charles F. Myers	O'Brien Resources, LLC		Texas	Kaufman
42.257.15500.0010		Linda Jean Brunette	O'Brien Resources, LLC		Texas	Kaufman
42.257.15500.0012		Billie Jo Rash	O'Brien Resources, LLC		Texas	Kaufman
42.257.15500.0013 42.257.15500.0014	9/7/2011	Fred E. McDonald et ux The Est of Michle Akin et al	O'Brien Resources, LLC O'Brien Resources, LLC		Texas Texas	Kaufman Kaufman
42.257.15500.0014	9/28/2011	Clark M. Griffith Family Tr, Nancy L. Griffith Trustee	O'Brien Resources, LLC		Texas	Kaufman
42.257.15500.0025		Natural Area Preservation Association, Inc	O'Brien Resources, LLC		Texas	Kaufman
42.257.15500.0047	8/1/2013	Shannon Dee Akin	O'Brien Resources, LLC		Texas	Kaufman
42.257.15500.0048	8/1/2013	Brad C. Akin et ux	O'Brien Resources, LLC		Texas	Kaufman
42.257.15500.0049	8/1/2013	Craig Bryan Akin	O'Brien Resources, LLC	4429-066	Texas	Kaufman
42.203.16110.0002		Gloria M. Smith	JB Land Services, LLC	2015-000000414	Texas	Harrison
42.203.16110.0003	9/17/2014	Floyd A. Duncan	O'Brien Resources, LLC	2015-000000412	Texas	Harrison
42.203.16110.0008	10/1/2014	Shelia Gail Duncan Hodges	O'Brien Resources, LLC	2015-000000411	Texas	Harrison
42.203.16110.0015	10/1/2014	Jeannie Lee Duncan Crisman	O'Brien Resources, LLC	2015-000000406	Texas	Harrison
42.203.16110.0021	9/16/2014	Joni L. Hall Gegg	O'Brien Resources, LLC	2015-000000401	Texas	Harrison
42.001.16350.0002	4/1/2015	Clara LaRoe Edwards	O'Brien Resources, LLC	2464-0456	Texas	Anderson
42.001.16350.0003	4/1/2015	Dan M. LaRoe, Jr.	O'Brien Resources, LLC	2464-0444	Texas	Anderson
42.001.16350.0004	4/1/2015	Joel Miller Herrington	O'Brien Resources, LLC		Texas	Anderson
42.001.16350.0006	4/1/2015	Mary Lois Lacy	O'Brien Resources, LLC		Texas	Anderson
42.001.16350.0007	4/1/2015	Melissa Dawn Herrington Thomas	O'Brien Resources, LLC		Texas	Anderson
42.001.16350.0008	4/1/2015	Roy Allen Herrington, III	O'Brien Resources, LLC	2464-0489	Texas	Anderson

Lease #	Date	Lessor	Lessee	Book-Page	State	County
42.001.16350.0009	4/1/2015	Roy Allen Herrington, Jr., individually and as life estate owner	O'Brien Resources, LLC	2464-0482	Texas	Anderson
42.001.16350.0010	7/7/2015	John Buck Scarborough	O'Brien Resources, LLC	2476-321	Texas	Anderson
42.001.16350.0011	7/7/2015	Ezem G. Scarborough	O'Brien Resources, LLC	2476-0317	Texas	Anderson
42.001.16350.0012	8/10/2015	Charles E. Ward	O'Brien Resources, LLC	2491-194	Texas	Anderson
42.001.16350.0013	8/10/2015	Lou Dell Rhodes Ferguson	O'Brien Resources, LLC	2491-191	Texas	Anderson
42.001.16350.0014	8/10/2015	Samuel David Rhodes	O'Brien Resources, LLC	2491-188	Texas	Anderson
42.001.16350.0015	8/11/2015	Final Talley, Ltd.	O'Brien Resources, LLC	2589-0643	Texas	Anderson
42.001.16350.0016	8/12/2015	Alice Wallace Martin (Deceased)	O'Brien Resources, LLC	2589-0649	Texas	Anderson
42.001.16350.0017	8/10/2015	Cathy Bailey Burgess	O'Brien Resources, LLC	2491-185	Texas	Anderson
42.001.16350.0018	8/11/2015	Vickie Fitzgerald	O'Brien Resources, LLC	2497-033	Texas	Anderson
42.001.16350.0019	9/9/2015	Sherrill Lynn Magee Stidham	O'Brien Resources, LLC	2589-0658	Texas	Anderson
42.001.16350.0020	8/17/2015	Cariton Ray Allen	O'Brien Resources, LLC	2491-182	Texas	Anderson
42.001.16350.0022	8/6/2015	Billy Scarborough	O'Brien Resources, LLC	2491-176	Texas	Anderson
42.001.16350.0023	9/16/2015	Debbie Elaine Webb Vann	O'Brien Resources, LLC	2589-0652	Texas	Anderson
42.001.16350.0024	8/17/2015	Wayne Norris Cox	O'Brien Resources, LLC	2491-173	Texas	Anderson
42.001.16350.0025	8/17/2015	Charlene Miriam Ellsworth	O'Brien Resources, LLC	2491-170	Texas	Anderson
42.001.16350.0026	8/17/2015	Frank Lee Williams	O'Brien Resources, LLC	2491-164	Texas	Anderson
42.001.16350.0027	9/22/2015	Laura W. Queen Morris	O'Brien Resources, LLC	2491-167	Texas	Anderson
42.001.16350.0029	9/10/2015	Joyce Westmoreland Dunlap	O'Brien Resources, LLC	2491-161	Texas	Anderson
42.001.16350.0030	9/10/2015	Jeanette Westmoreland Caldwell	O'Brien Resources, LLC	2491-158	Texas	Anderson
42.001.16350.0031	10/14/2015	Joe Keith Welch	O'Brien Resources, LLC	2493-375	Texas	Anderson
42.001.16350.0032	10/14/2015	James Eddie Welch	O'Brien Resources, LLC	2493-379	Texas	Anderson
42.001.16350.0033	10/14/2015	Frances Farmer Westmoreland	O'Brien Resources, LLC	2491-155	Texas	Anderson
42.001.16350.0034	10/14/2015	Dari W. Johnson	O'Brien Resources, LLC	2491-152	Texas	Anderson
42.001.16350.0035	10/14/2015	Bill Baker	O'Brien Resources, LLC	2491-149	Texas	Anderson
42.001.16350.0036	8/17/2015	Joel and Misty Smith	O'Brien Resources, LLC	2491-387	Texas	Anderson
42.001.16350.0037	8/17/2015	Misty Smith	O'Brien Resources, LLC	2491-384	Texas	Anderson
42.001.16350.0038	8/17/2015	Donald R. Allen & Nora C. McClure	O'Brien Resources, LLC	2491-381	Texas	Anderson
42.001.16350.0039	10/14/2015	Carolyn A. Estes	O'Brien Resources, LLC	2491-200	Texas	Anderson
42.001.16350.0040	10/14/2015	Jack D. Brown	O'Brien Resources, LLC	2491-197	Texas	Anderson
42.001.16350.0041	11/15/2015	Francis J. Eaton	O'Brien Resources, LLC	2498-555	Texas	Anderson
42.001.16350.0042	11/15/2015	Barney Johnston	O'Brien Resources, LLC	2498-546	Texas	Anderson
42.001.16350.0043	11/15/2015	Thomas W. Johnston	O'Brien Resources, LLC	2498-558	Texas	Anderson
42.001.16350.0044	11/15/2015	James Johnston and Wanda Johnston	O'Brien Resources, LLC	2498-552	Texas	Anderson
42.001.16350.0045	11/15/2015	Billy M. Johnston and Brenda Ann Johnston	O'Brien Resources, LLC	2498-561	Texas	Anderson
42.001.16350.0046	11/15/2015	Matthew S. Richardson and Brenda Richardson	O'Brien Resources, LLC	2498-549	Texas	Anderson
42.001.16350.0050	8/11/2015	Virginia Fitzgerald	O'Brien Resources, LLC	2498-543	Texas	Anderson
42.001.16350.0051	10/14/2015	Demaris Westmoreland	O'Brien Resources, LLC	2498-567	Texas	Anderson
42.001.16350.0054	11/15/2015	Lewis Ballard and Glenda Ballard	O'Brien Resources, LLC	2498-564	Texas	Anderson
42.001.16350.0055	1/6/2016	Jackson Hanks as Trustee of the C. W. Hanks Children's Trust	O'Brien Resources, LLC	2505-355	Texas	Anderson
42.001.16350.0058	12/7/2015	Mary Richert	O'Brien Resources, LLC	2505-352	Texas	Anderson
42.001.16350.0071	5/5/2016	Julie Ann Beard Goodman	O'Brien Resources, LLC	2523-0629	Texas	Anderson
42.001.16350.0072	5/5/2016	Dayna Beard Isensee	O'Brien Resources, LLC	2523-0623	Texas	Anderson
42.001.16350.0073	5/5/2016	Denise Beard Hoeksema	O'Brien Resources, LLC	2523-0617	Texas	Anderson
42.001.16350.0075	5/5/2016	Robert Alan Beard	O'Brien Resources, LLC	2523-0605	Texas	Anderson
42.001.16350.0077	5/5/2016	Lou Ellen Mills	O'Brien Resources, LLC	2523-0644	Texas	Anderson
42.001.16350.0078		Richard Dwayne Henderson	O'Brien Resources, LLC	2524-0385	Texas	Anderson
42.001.16350.0079	5/24/2016	Julie Anne Miller	O'Brien Resources, LLC	2524-0382	Texas	Anderson
42.001.16350.0080	5/24/2016	J. W. Brown	O'Brien Resources, LLC	2524-0388	Texas	Anderson
42.001.16350.0081	5/24/2016	Dianne Henderson	O'Brien Resources, LLC	2524-0397	Texas	Anderson
42.001.16350.0082	5/24/2016	Toby Pierce	O'Brien Resources, LLC	2524-0394	Texas	Anderson
42.001.16350.0083		William Bryan Townsend, Jr.	O'Brien Resources, LLC	2524-0391	Texas	Anderson
42.001.16350.0084	5/24/2016	Betty Joyce Brotherton	O'Brien Resources, LLC	2426-609	Texas	Anderson
42.001.16350.0085	5/24/2016	Robert Wayne Henderson	O'Brien Resources, LLC	2526-606	Texas	Anderson
42.001.16350.0086	5/24/2016	Dorothy L. Cross	O'Brien Resources, LLC	2528-603	Texas	Anderson
42.001.16350.0087	7/7/2016	Grace L. Teer	O'Brien Resources, LLC	2556-0414	Texas	Anderson
42.001.16350.0088	10/27/2016	Charles E. Eikner, Jr., dealing herein with his sole and separate property	O'Brien Resources, LLC	2556-0430	Texas	Anderson
42.001.16350.0089		David A. Eikner, dealing herein with his sole and separate property	O'Brien Resources, LLC	2556=-426	Texas	Anderson
42.001.16350.0090	10/27/2016	Elaine Estridge, dealing herein with her sole and separate property	O'Brien Resources, LLC	256-0428	Texas	Anderson
42.001.16350.0091		Elvin D. White, dealing herein with his sole and separate property	O'Brien Resources, LLC	2560-0347	Texas	Anderson
42.001.16350.0100	11/21/2016	Mance Michael Park, dealing herein with his sole and separate property.	O'Brien Resources, LLC	2564-0001	Texas	Anderson
42.001.16350.0101		June L. Francis	O'Brien Resources, LLC	2547-0388	Texas	Anderson
42.001.16350.0102		Mary E. Stark	O'Brien Resources, LLC	2547-0383	Texas	Anderson
42.001.16350.0103		Judy Jordan Eckert dealing in her sole and separate property	O'Brien Resources, LLC	2547-0365	Texas	Anderson
42.001.16350.0104	11/22/2016	William D. Jordan	O'Brien Resources, LLC	2547-0401	Texas	Anderson

Lease #	Date	Lessor	Lessee	Book-Page	State	County
42.001.16350.0105	11/17/2016	Mary Ann Rogstad	O'Brien Resources, LLC	2547-375	Texas	Anderson
42.001.16350.0106	11/17/2016	Martha S. Davis	O'Brien Resources, LLC	2547-0371	Texas	Anderson
42.001.16350.0107	11/17/2016	Cynthia Gail Kenter	O'Brien Resources, LLC	2547-0379	Texas	Anderson
42.001.16350.0108	11/10/2016	represented by John Madison Clement Milam	O'Brien Resources, LLC	2549-0356	Texas	Anderson
42.001.16350.0109	11/17/2016	Peggy Sue Chauvin	O'Brien Resources, LLC	2547-0393	Texas	Anderson
42.001.16350.0110	11/17/2016	Jack C. Davis	O'Brien Resources, LLC	2547-0397	Texas	Anderson
42.001.16350.0111	1/18/2017	Cole O'Neill Ltd. by: Connie Fain, VP and Tr Officer	O'Brien Resources, LLC	2252-0213	Texas	Anderson
42.001.16350.0112	1/17/2017	HCB, Inc.	O'Brien Resources, LLC	2552-0209	Texas	Anderson
42.001.16350.0113	1/18/2017	Thomas M. Colley Residual Tr by and through its Trustee Citizens National Bank	O'Brien Resources, LLC	2552-0211	Texas	Anderson
42.001.16350.0114	11/21/2016	Virginia Etheredge Thompson	O'Brien Resources, LLC	2589-0655	Texas	Anderson
42.001.16350.0115	12/23/2016	David Lee McGee and Judith Hatfield McGee, Individually and as Co-Trustees of the McGee Family Trust	O'Brien Resources, LLC	2556-0435	Texas	Anderson
42.001.16350.0116	1/25/2017	Clifford E. Bomer, dealing herein with his sole and separate property	O'Brien Resources, LLC	2560-0334	Texas	Anderson
42.001.16350.0117	1/10/2017	New Midway Baptist Church	O'Brien Resources, LLC	2557-0180	Texas	Anderson
42.001.16350.0118	11/1/2014	Margaret Jane Dansby	O'Brien Resources, LLC	2449-125	Texas	Anderson
42.001.16350.0123	11/18/2014	Joy Beth Berryhill Strange	O'Brien Resources, LLC	2453-630	Texas	Anderson
42.001.16350.0125	11/10/2014	Ethel Lillian Berryhill	O'Brien Resources, LLC	2453-628	Texas	Anderson
42.001.16350.0132	4/28/2016	John E. Fairchild, dealing herein with his sole and separate property	O'Brien Resources, LLC	2523-0646	Texas	Anderson
42.001.16350.0133	4/28/2016	Robert R. Sellers, dealing herein with his sole and separate property	O'Brien Resources, LLC	2523-0635	Texas	Anderson
42.001.16350.0134	4/28/2016	Mark A. Fairchild, dealing herein with his sole and separate property	O'Brien Resources, LLC	2523-0642	Texas	Anderson
42.001.16350.0135	4/28/2016	Ray T. Fairchild, dealing herein with his sole and separate property	O'Brien Resources, LLC	2523-0638	Texas	Anderson
42.001.16350.0136	4/28/2016	Nancy Sellers Fairchild, dealing herein with her sole and separate property	O'Brien Resources, LLC	2523-0640	Texas	Anderson
42.001.16350.0136	4/28/2016	Nancy Sellers Fairchild, dealing herein with her sole and separate property	O'Brien Resources, LLC	2523-0640	Texas	Anderson
42.001.16350.0137	8/7/2016	James Dow Henderson Trust	O'Brien Resources, LLC	2562-0138	Texas	Anderson
42.001.16350.0139	3/23/2017	Jo Cynthia Monroe	O'Brien Resources, LLC	2568-0721	Texas	Anderson
42.001.16350.0140	3/29/2017	Elizabeth Marie Howard	O'Brien Resources, LLC	2571-255	Texas	Anderson
42.001.16350.0141	3/23/2017	James M. Bobbitt	O'Brien Resources, LLC	2568-0724	Texas	Anderson
42.001.16350.0142	3/23/2017	Whitney M. Bobbitt	O'Brien Resources, LLC	2568-0721	Texas	Anderson
42.001.16350.0143	3/30/2017	Megan M. Wortham et al	O'Brien Resources, LLC	2568-0730	Texas	Anderson
42.001.16350.0144	4/18/2017	John Park Howard	O'Brien Resources, LLC	2568-0733	Texas	Anderson
42.001.16350.0144	8/17/2015	Jack G. Williams, Jr.	O'Brien Resources, LLC	2491-179	Texas	Anderson
42.001.16350.0145	4/18/2017	Dennis Howard	O'Brien Resources, LLC	2568-0736	Texas	Anderson
42.001.16350.0146	4/18/2017	Steven Howard	O'Brien Resources, LLC	2571-246	Texas	Anderson
42.001.16350.0147		Terry Howard	O'Brien Resources, LLC	2573-0391	Texas	Anderson
42.001.16350.0148		Susan Howard York	O'Brien Resources, LLC	2571-243	Texas	Anderson
42.001.16350.0149	4/20/2017		O'Brien Resources, LLC	2571-249	Texas	Anderson
42.001.16350.0150		Brenda Lott Anderson aka Brenda Lott	O'Brien Resources, LLC	2571-252	Texas	Anderson
42.001.16350.0151		The Deobra Gist Rev Living Tr	O'Brien Resources, LLC	2575-583	Texas	Anderson
42.001.16350.0152		Jonell Lipsey Thompson	O'Brien Resources, LLC	Pending	Texas	Anderson
42.001.16350.0153		Joan Bruchmiller Johnston	O'Brien Resources, LLC	2587-0797	Texas	Anderson
42.001.16350.0154		Leon Bruchmiller	O'Brien Resources, LLC	2587-0788	Texas	Anderson
42.001.16350.0155		Chyleen L. Feagin	O'Brien Resources, LLC	2587-0791	Texas	Anderson
42.001.16350.0156		O.W. & E. M. Craft Living Trust	O'Brien Resources, LLC	2587-0794	Texas	Anderson
42.293.15200.0004		Sybil H. Brock, Peter Lozada as Agent & AIF	Texas Royalty Corp	1293-860	Texas	Limestone
42.293.15200.0007		Barbara Cantu	Texas Royalty Corp	1290-142	Texas	Limestone
42.293.15200.0008		Patricia M. Carr	Texas Royalty Corp	1303-718	Texas	Limestone
42.293.15200.0009		Frederick Ferguson	Texas Royalty Corp	1302-721	Texas	Limestone Limestone
42.293.15200.0010 42.293.15200.0013		Four Winns Partnership, LTD David W. Gann	Texas Royalty Corp Texas Royalty Corp	1342-486 1304-394	Texas	Limestone
42.293.15200.0019		Lua Hancock	Texas Royalty Corp	1297-663	Texas	Limestone
42.293.15200.0019		Glenda J. Pittillo	Texas Royalty Corp	1302-713	Texas	Limestone
42.293.15200.0027		Jack Reed		1295-877	Texas	Limestone
42.293.15200.0027		Luminant Mineral Development Company, LLC	Texas Royalty Corp Texas Royalty Corp	1299-471	Texas	Limestone
42.293.15200.0028					Texas	
42.293.15200.0029		Joseph M. Moore W. B. Orand	Texas Royalty Corp Texas Royalty Corp	1297-666 1300-492	Texas	Limestone
42.293.15200.0030				1293-863	Texas	Limestone
42.293.15200.0031	10/1/2008	Mary Ellen Richardson April Reed	Texas Royalty Corp Texas Royalty Corp	1293-669	Texas	Limestone
42.293.15200.0032	1/2/2009	Rosa D. Richardson	Texas Royalty Corp	1311-026	Texas	Limestone
42.293.15200.0033		Rosa D. Richardson Barbara Reed Rudolph	Texas Royalty Corp	1293-851	Texas	Limestone
42.293.15200.0034		Martin Rudolph	Texas Royalty Corp	1293-651	Texas	Limestone
42.293.15200.0035		Meranda L. Tillmon	Texas Royalty Corp	1299-408	Texas	Limestone
42.293.15200.0042		Ricky Boyd Tillmon	Texas Royalty Corp	1297-672	Texas	Limestone
42.293.15200.0047	8/19/2008	Joseph P. Williams, Jr.	Texas Royalty Corp	1295-884	Texas	Limestone
42.293.15200.0049	8/10/2008	Brady Family Trust D, Theresa L. Brady, Sole Trustee	Texas Royalty Corp	1290-139	Texas	Limestone
42.293.15200.0055		Jason Hoisager, et ux	Texas Royalty Corp	1290-145	Texas	Limestone
42.293.15200.0056		Richard E. Witt	Texas Royalty Corp	1290-793	Texas	Limestone
12.200.10200.0000	5, 15/2000			.200-100	. 0,20	

Lease #	Date	Lessor	Lessee	Book-Page State	County
42.293.15200.0057	8/10/2008	Charles R. Wiggins	Sendero Resources, Inc.	1290-148 Texas	Limestone
42.293.15200.0089	7/21/2009	Katherine Komenda	Texas Royalty Corp	1318-448 Texas	Limestone
42.293.15200.0101	2/23/2006	Mary Jean Burton	Samson Lone Star Limited	1203-805 Texas	Limestone
42.213.14100.0001	12/30/2015	Jackie Lou Phillips	BROGO Minerals, LLC	2016-00011431 Texas	Henderson
42.213.14100.0002	12/30/2015	Dickey Family Trust dated July 3, 2006, Bill O. Dickey and Lynn C. Dickey	BROGO Minerals, LLC	2016-00011538 Texas	Henderson
42.213.14100.0003	1/6/2016	Norma Smith	BROGO Minerals, LLC	2016-00011451 Texas	Henderson
42.213.14100.0005	1/7/2016	Melissa Jo Pirtle Hollingsworth	BROGO Minerals, LLC	2016-00011587 Texas	Henderson
42.213.14100.0006		Sylvia Smith Lewis	BROGO Minerals, LLC	2016-00011622 Texas	Henderson
42.213.14100.0007		Danny Lynn Gandy	BROGO Minerals, LLC	2016-00011553 Texas	Henderson
42.213.14100.0008		Randy Flynn Gandy	BROGO Minerals, LLC	2016-00011552 Texas	Henderson
42.213.14100.0009		Ronny Quinn Gandy	BROGO Minerals, LLC	2017-00007373 Texas	Henderson
42.213.14100.0012		Nell Walker Wilbanks	BROGO Minerals, LLC	2017-00000361 Texas	Henderson
42.213.14100.0013		Kaye Walker Weil	BROGO Minerals, LLC	2017-00000361 Texas	Henderson
42.213.14100.0014		Linda E. Walker, a widow of Charles G. Walker	BROGO Minerals, LLC	2017-00000361 Texas	Henderson
42.213.14100.0016		Dan M. Rasco	BROGO Minerals, LLC	2017-00007374 Texas	Henderson
42.213.14100.0017		Barrett Thomas	BROGO Minerals, LLC	2016-00011464 Texas	Henderson
42.213.14100.0018		Debra Kinser	BROGO Minerals, LLC	2016-00011610 Texas	Henderson
42.213.14100.0020		Robert Alden Washburn	BROGO Minerals, LLC BROGO Minerals. LLC	2016-00011475 Texas	Henderson
42.213.14100.0021		Russell T. Rudy Energy LLC,		2016-00011445 Texas	Henderson
42.213.14100.0023		Nancy Nell Cato Bullock Sulvia Smith Lewise	BROGO Minerals, LLC BROGO Minerals, LLC	2016-00011519 Texas	Henderson
42.213.14100.0027 42.213.14100.0028		Sylvia Smith Lewis James W. Jackson and Nelda Jackson, husband and wife	BROGO Minerals, LLC BROGO Minerals, LLC	2016-00011621 Texas 2016-00011595 Texas	Henderson
42.213.14100.0028		James W. Jackson and Neida Jackson, nusband and wife Billie Joyce Watson & Margie Claudine Chastain	BROGO Minerals, LLC BROGO Minerals, LLC	2016-00011595 Texas 2016-00012809 Texas	Henderson Henderson
42.213.14100.0029		Frank Hubbard Johns	BROGO Minerals, LLC	2016-00012609 Texas	Henderson
42.213.14100.0038		The Kennedy Family Trust, John Kennedy II	BROGO Minerals, LLC	2016-00011609 Texas	Henderson
42.213.14100.0039		Peggy Louise Buford Oldham	BROGO Minerals, LLC	2016-00011429 Texas	Henderson
42.213.14100.0041		Ross Earl Smith	BROGO Minerals, LLC	2016-00011457 Texas	Henderson
42.213.14100.0042		J. R. Coker	BROGO Minerals, LLC	2016-00011529 Texas	Henderson
42.213.14100.0044		Barry J. Boswell	BROGO Minerals, LLC	2017-00004199 Texas	Henderson
42.213.14100.0045		Tony J. Proctor	BROGO Minerals, LLC	2016-00011435 Texas	Henderson
42.213.14100.0046		Richard Earl Roby	BROGO Minerals, LLC	2017-00007376 Texas	Henderson
42.213.14100.0049	9/21/2016	Ben-Z Lawrence	BROGO Minerals, LLC	2016-00017840 Texas	Henderson
42.213.14100.0050	9/29/2016	Robert Scogin	BROGO Minerals, LLC	2016-00017836 Texas	Henderson
42.213.14100.0056	3/29/2015	Matt L. Mitchell and Cheryl Mitchell	BROGO Minerals, LLC	2016-00011637 Texas	Henderson
42.213.14100.0057	1/22/2016	Betty Boswell	BROGO Minerals, LLC	2016-00011512 Texas	Henderson
42.213.14100.0069	5/4/2016	Katherine Turner Cotter	BROGO Minerals, LLC	2016-00011530 Texas	Henderson
42.213.14100.0102	2/22/2016	Wayne West Browning	BROGO Minerals, LLC	2016-00011517 Texas	Henderson
42.213.14100.0110	5/19/2016	Gia Freeman Newell	BROGO Minerals, LLC	2016-00011427 Texas	Henderson
42.213.14100.0116	3/21/2016	Barry J. Boswell	BROGO Minerals, LLC	2016-00011510 Texas	Henderson
42.213.14100.0119	2/22/2016	Wayne West Browning	BROGO Minerals, LLC	2016-00011516 Texas	Henderson
42.213.14100.0120	3/31/2016	Allen B. Boswell	BROGO Minerals, LLC	2016-00011506 Texas	Henderson
42.213.14100.0121	3/21/2016	Bradley Wayne Godwin	BROGO Minerals, LLC	2016-00011555 Texas	Henderson
42.213.14100.0122	3/21/2016	John Bruce Godwin	BROGO Minerals, LLC	2016-00011562 Texas	Henderson
42.213.14100.0125	3/21/2016	Vaclay Jim Jezek	BROGO Minerals, LLC	2016-00011598 Texas	Henderson
42.213.14100.0126		Niki Diane Godwin Majors	BROGO Minerals, LLC	2016-00011626 Texas	Henderson
42.213.14100.0127		Gia Freeman Newell	BROGO Minerals, LLC	2016-00011426 Texas	Henderson
42.213.14100.0128		Susan B. Kelley	BROGO Minerals, LLC	2016-00011605 Texas	Henderson
42.213.14100.0135		Gia Freeman Newell	BROGO Minerals, LLC	2016-00011427 Texas	Henderson
42.213.14100.0136		William H. and Kathleen Gardner Revocable Tr by William H. Gardner and Kathleen R. Gardner Co-Trustees	BROGO Minerals, LLC	2016-00011662 Texas	Henderson
42.213.14100.0139		Frances Gwendolyn Lee	BROGO Minerals, LLC	2016-00011620 Texas	Henderson
42.213.14100.0148		Wayne West Browning	BROGO Minerals, LLC	2016-00011515 Texas	Henderson
42.213.14100.0157		William R. Wakefield	BROGO Minerals, LLC	2016-00011472 Texas	Henderson
42.213.14100.0158 42.213.14100.0160		Yel-Roc Family Tr by John Craig Corley Militan Louce Mills	BROGO Minerals, LLC BROGO Minerals, LLC	2016-00011642 Texas 2016-00011635 Texas	Henderson Henderson
42.213.14100.0160		Wilma Joyce Mills Troy Calhoun	BROGO Minerals, LLC	2016-00011635 Texas 2016-00011523 Texas	Henderson
42.213.14100.0162		Thomas Roy Jensen	BROGO Minerals, LLC BROGO Minerals, LLC	2016-00011523 Texas 2016-00011597 Texas	Henderson
		Jimmie H. Graham	BROGO Minerals, LLC	2016-00011597 Texas 2016-00011573 Texas	Henderson
42.213.14100.0164	2/2010	Dan M. Rasco	BROGO Minerals, LLC	2016-00011373 Texas	Henderson
42.213.14100.0164 42.213.14100.0165	7/12/2016	Dan III. Tadoo	5.1000 Millordio, EEO	_0.0 000.1442 10.00	Henderson
42.213.14100.0165		Rrenda Marilyn Rehkon by Lila I. Rehkon her AIF	BROGO Minerals LLC	2016-00011443 Texas	
42.213.14100.0165 42.213.14100.0166	7/18/2016	Brenda Marilyn Rehkop by Lila L. Rehkop her AIF	BROGO Minerals, LLC	2016-00011443 Texas 2017-00007378 Texas	
42.213.14100.0165 42.213.14100.0166 42.213.14100.0167	7/18/2016 7/19/2016	Donna Pahl	BROGO Minerals, LLC	2017-00007378 Texas	Henderson
42.213.14100.0165 42.213.14100.0166 42.213.14100.0167 42.213.14100.0168	7/18/2016 7/19/2016 7/19/2016	Donna Pahl Dianna R. Lee	BROGO Minerals, LLC BROGO Minerals, LLC	2017-00007378 Texas 2017-0007380 Texas	Henderson Henderson
42.213.14100.0165 42.213.14100.0166 42.213.14100.0167	7/18/2016 7/19/2016 7/19/2016 7/19/2016	Donna Pahl Dianna R. Lee Dianna R. Lee	BROGO Minerals, LLC	2017-00007378 Texas	Henderson
42.213.14100.0165 42.213.14100.0166 42.213.14100.0167 42.213.14100.0168 42.213.14100.0169	7/18/2016 7/19/2016 7/19/2016 7/19/2016 7/26/2016	Donna Pahl Dianna R. Lee	BROGO Minerals, LLC BROGO Minerals, LLC BROGO Minerals, LLC	2017-00007378 Texas 2017-0007380 Texas 2017-00007379 Texas	Henderson Henderson Henderson

Lease #	Date	Lessor	Lessee	Book-Page State	County
42.213.14100.0174	7/18/2016	Charles S. Bass	BROGO Minerals, LLC	2016-00011496 Texas	Henderson
42.213.14100.0175	7/17/2016	Clyde Corbin Humphries	BROGO Minerals, LLC	2016-00012811 Texas	Henderson
42.213.14100.0177	7/16/2016	Joe David Adams	BROGO Minerals, LLC	2016-00011484 Texas	Henderson
42.213.14100.0181	7/12/2016	Robert Hoover and Lynn Hoover	BROGO Minerals, LLC	2016-00011589 Texas	Henderson
42.213.14100.0182	7/7/2016	Dwain Johnson and Gracie Lylnn Johnson	BROGO Minerals, LLC	2016-00011601 Texas	Henderson
42.213.14100.0183	3/21/2016	Vallie Mae Gualdarama	BROGO Minerals, LLC	2016-00011579 Texas	Henderson
42.213.14100.0186	1/27/2016	Terry Frank Buford	BROGO Minerals, LLC	2017-00007375 Texas	Henderson
42.213.14100.0187	7/19/2016	Donna Pahl	BROGO Minerals, LLC	2017-00007377 Texas	Henderson
42.213.14100.0189	7/27/2016	Ewel Eugene Buford	BROGO Minerals, LLC	2016-00012808 Texas	Henderson
42.213.14100.0190	7/14/2016	Gary Joe Bass	BROGO Minerals, LLC	2016-00011660 Texas	Henderson
42.213.14100.0191	7/14/2016	Cheryl Kay Waterman	BROGO Minerals, LLC	2016-00011651 Texas	Henderson
42.213.14100.0192	7/26/2016	Michael M. Cade, Trustee of the Marital Trust and as Executor of the Est of Grace McCallum Mones Cade	BROGO Minerals, LLC	2016-00011658 Texas	Henderson
42.213.14100.0193	7/26/2016	Mattie Ann Andrews	BROGO Minerals, LLC	2016-00012807 Texas	Henderson
42.213.14100.0194	7/18/2016	Steven Wayne Waterman	BROGO Minerals, LLC	2016-00011652 Texas	Henderson
42.213.14100.0195	7/22/2016	William Earl Pruitt	BROGO Minerals, LLC	2016-00011645 Texas	Henderson
42.213.14100.0196	7/12/2016	Jerry Leon Adams	BROGO Minerals, LLC	2016-00012806 Texas	Henderson
42.213.14100.0197	7/8/2016	Milton Keith Adams	BROGO Minerals, LLC	2016-00011649 Texas	Henderson
42.213.14100.0198	7/20/2016	Victoria Place Phase II Ltd., Emanuel H. Glockzin, Jr., Trustee	BROGO Minerals, LLC	2016-00011650 Texas	Henderson
42.213.14100.0199	7/20/2016	Victoria Place, Ltd., Emanuel H. Glockzin, Jr. Trustee	BROGO Minerals, LLC	2016-00011653 Texas	Henderson
42.213.14100.0200	7/26/2016	Michael Marlan Cade and Billie Cade	BROGO Minerals, LLC	2016-00011659 Texas	Henderson
42.213.14100.0201	7/20/2016	Glockzin Family Partnership Ltd., Legally known as Glockzin Family Partners, Ltd., A Texas Limited Partnership	BROGO Minerals, LLC	2016-00011654 Texas	Henderson
42.213.14100.0202	7/18/2016	Merlon G. Avant	BROGO Minerals, LLC	2016-00012805 Texas	Henderson
42.213.14100.0203	8/4/2016	Michael Press and Holly Michelle Press	BROGO Minerals, LLC	2016-00012804 Texas	Henderson
42.213.14100.0204	8/4/2016	Thomas Roy Jensen	BROGO Minerals, LLC	2016-00012803 Texas	Henderson
42.213.14100.0205	7/25/2016	Donald Lawrence Ward	BROGO Minerals, LLC	2016-00012802 Texas	Henderson
42.213.14100.0206	7/25/2016	Ronny Alan Ward	BROGO Minerals, LLC	2016-00012800 Texas	Henderson
42.213.14100.0209	8/5/2016	Ivory Joe Givens et ux	BROGO Minerals, LLC	2016-00012793 Texas	Henderson
42.213.14100.0211	8/12/2016	Lee Ann Andrews Latimer and Larry Latimer	BROGO Minerals, LLC	2016-00014487 Texas	Henderson
42.213.14100.0212	7/13/2016	Carol Bardwell Landers aka Carol Christy Landers TDCJ ID No 02035033	BROGO Minerals, LLC	2016-00012799 Texas	Henderson
42.213.14100.0214	8/10/2016	Sara Alice Scott Cox	BROGO Minerals, LLC	2016-00012794 Texas	Henderson
42.213.14100.0215	8/4/2016	Lucy Hooker by Nancy Meredith Hooker AIF	BROGO Minerals, LLC	2016-00012795 Texas	Henderson
42.213.14100.0216	8/2/2016	J.M. & Lola McCulloch Ventures LLC, c/o Dan McCulloch	BROGO Minerals, LLC	2016-00012796 Texas	Henderson
42.213.14100.0217		Billy Dean Paris and Sharla G. Paris	BROGO Minerals, LLC	2016-00011648 Texas	Henderson
42.213.14100.0218	7/18/2016	James Bennett Givens and Doris ANn Givens	BROGO Minerals, LLC	2016-00012797 Texas	Henderson
42.213.14100.0219	7/18/2016	Robert L. Bass	BROGO Minerals, LLC	2016-00012798 Texas	Henderson
42.213.14100.0220		Mattie Andrews and Lee Ann Andrews Latimer	BROGO Minerals, LLC	2016-00014470 Texas	Henderson
42.213.14100.0221		Connie F. Lehowicz	BROGO Minerals, LLC	2016-00014478 Texas	Henderson
42.213.14100.0222		Paul A. Neill	BROGO Minerals, LLC	2016-00014479 Texas	Henderson
42.213.14100.0223		Randy Allen Coker	BROGO Minerals, LLC	2016-00014477 Texas	Henderson
42.213.14100.0224		Larry D. Yates	BROGO Minerals, LLC	2016-00014476 Texas	Henderson
42.213.14100.0225		Martha Kate Moser Anderson	BROGO Minerals, LLC	2016-00014474 Texas	Henderson
42.213.14100.0226		Chris Allen Moser	BROGO Minerals, LLC	2016-00014473 Texas	Henderson
42.213.14100.0227		Wayne West Browning	BROGO Minerals, LLC	2016-00014472 Texas	Henderson
42.213.14100.0228		Kenny Yee	BROGO Minerals, LLC	2016-00014475 Texas	Henderson
42.213.14100.0229		Rekay Bass	BROGO Minerals, LLC	2016-00014471 Texas	Henderson
42.213.14100.023		Douglas C. Koch	BROGO Minerals, LLC	2016-00017847 Texas	Henderson
42.213.14100.0230		Nelda Jean Ramsey Loper	BROGO Minerals, LLC BROGO Minerals, LLC	2017-00003161 Texas	Henderson
42.213.14100.0231		Jessie Nell Ramsey Dixon		2017-00003160 Texas	Henderson
42.213.14100.0232		Erin L. Alexander	BROGO Minerals, LLC	2016-00016237 Texas	Henderson
42.213.14100.0233		Charla Royston Lake	BROGO Minerals, LLC	2016-00014495 Texas	Henderson
42.213.14100.0234	8/9/2016	Richard C. Royston	BROGO Minerals, LLC	2016-00014494 Texas	Henderson
42.213.14100.0235		Paul S. Wall Trust No. 1 Dated 6-1-1986 by Paul Mae Wall Hughes Successor Trustee	BROGO Minerals, LLC	2016-00014493 Texas	Henderson
42.213.14100.0236		Randy Allen Coker	BROGO Minerals, LLC	2016-00014491 Texas	Henderson
42.213.14100.0237	8/9/2016	Ruahama Royston Tucker	BROGO Minerals, LLC	2016-00014490 Texas	Henderson
42.213.14100.0238		Billy Dean Paris et ux	BROGO Minerals, LLC	2016-00014492 Texas	Henderson
42.213.14100.0239		Gia Freeman Newell	BROGO Minerals, LLC	2016-00014483 Texas	Henderson
42.213.14100.0240		Gary McDaniel	BROGO Minerals, LLC BROGO Minerals, LLC	2016-00014484 Texas	Henderson
42.213.14100.0241		Larry G. Thomas		2016-00014485 Texas	Henderson
42.213.14100.0244		Lee Tillman	BROGO Minerals, LLC	2016-00014465 Texas	Henderson
42.213.14100.0245		James Robert Ward	BROGO Minerals, LLC	2016-00014489 Texas	Henderson
42.213.14100.0246		Jimmie H. Graham	BROGO Minerals, LLC	2016-00014464 Texas	Henderson
42.213.14100.0247		Gary Bowen	BROGO Minerals, LLC	2016-00014480 Texas	Henderson
42.213.14100.0248		Elise Hilton Haralson	BROGO Minerals, LLC BROGO Minerals, LLC	2016-00014481 Texas	Henderson
42.213.14100.0249		BBT Enterprises Ltd by Barbara B. Thomas Managing Partner		2016-00019413 Texas	Henderson
42.213.14100.0250	ช/15/2016	Madelaine Jane Mose Matteson	BROGO Minerals, LLC	2016-00014482 Texas	Henderson

Lease #	Date	Lessor	Lessee	Book-Page State	County
42.213.14100.0251	8/23/2016	Shirley A. McGlaun	BROGO Minerals, LLC	2016-00014463 Texas	Henderson
42.213.14100.0252	8/29/2016	RWM 1988 Trust by R. W. Moncrief Trustee	BROGO Minerals, LLC	2016-00014469 Texas	Henderson
42.213.14100.0253	8/29/2016	WAM3 Genpar LLC as General Partner of WAM3LP by C. B. Moncrief Manger	BROGO Minerals, LLC	2016-00014468 Texas	Henderson
42.213.14100.0254	8/29/2016	CBMoncrief O&G, LLC by C. B. Moncrief	BROGO Minerals, LLC	2016-00014467 Texas	Henderson
42.213.14100.0256	11/3/2016	Wayne A. Bissett	BROGO Minerals, LLC	2017-00004207 Texas	Henderson
42.213.14100.0259	3/10/2016	Archie A. Dennis Jr. et ux	BROGO Minerals, LLC	2016-00011684 Texas	Henderson
42.213.14100.0260	4/18/2017	Margaret Alice McDaniel as Manager for RVS Minerals, LLC	BROGO Minerals, LLC	2017-00007371 Texas	Henderson
42.213.14100.0262	9/16/2016	Betty Jane Coker	BROGO Minerals, LLC	2016-00016235 Texas	Henderson
42.213.14100.0263	8/24/2016	Brenda G. Tinney	BROGO Minerals, LLC	2016-00014449 Texas	Henderson
42.213.14100.0264	8/12/2016	Virginia P. Wyatt	BROGO Minerals, LLC	2016-00014450 Texas	Henderson
42.213.14100.0265	8/1/2016	Linda McAtee Secrest	BROGO Minerals, LLC	2016-00014452 Texas	Henderson
42.213.14100.0266	8/1/2016	Stephen Edward McAtee	BROGO Minerals, LLC	2016-00014453 Texas	Henderson
42.213.14100.0267	8/1/2016	Robert W. McAtee, Jr.	BROGO Minerals, LLC	2016-00014454 Texas	Henderson
42.213.14100.0268	9/8/2016	Marvin Ray Crawford, Jr.	BROGO Minerals, LLC	2016-00014455 Texas	Henderson
42.213.14100.0269	8/25/2016	Rural Shade, Inc. by Gretchen P. Lahourcade President	BROGO Minerals, LLC	2016-00014456 Texas	Henderson
42.213.14100.0270	8/25/2016	Gretchen Perryman Lahourcade	BROGO Minerals, LLC	2016-00014459 Texas	Henderson
42.213.14100.0271	8/10/2016	Meda L. Dickerson by and through her Agt & AIF Tina Beaird	BROGO Minerals, LLC	2016-00014458 Texas	Henderson
42.213.14100.0272	8/18/2016	Mary Jo Sparks	BROGO Minerals, LLC	2016-00014457 Texas	Henderson
42.213.14100.0273	8/16/2016	Jimmy Ray Coker	BROGO Minerals, LLC	2016-00014451 Texas	Henderson
42.213.14100.0274	9/1/2016	Ruth Barron Zumwalt	BROGO Minerals, LLC	2016-00014448 Texas	Henderson
42.213.14100.0278	3/21/2016	Ross Earl Smith and Andrea J. Smith	BROGO Minerals, LLC	2016-00011456 Texas	Henderson
42.213.14100.0280	3/29/2016	Paul Tiner	BROGO Minerals, LLC	2016-00011467 Texas	Henderson
42.213.14100.0281	3/23/2016	Paul Dustin Godwin, Dylan Cole Godwin and Debbie Godwin Life Tenant	BROGO Minerals, LLC	2016-00011566 Texas	Henderson
42.213.14100.0282	3/28/2016	James P. Leake by Lynda H. Leake Indiv & AIF	BROGO Minerals, LLC	2016-00011619 Texas	Henderson
42.213.14100.0284	4/19/2017	Keith William Wallace	BROGO Minerals, LLC	2017-00007370 Texas	Henderson
42.213.14100.0287	9/22/2016	Pamela Hudson Kullerd	BROGO Minerals, LLC	2016-00016239 Texas	Henderson
42.213.14100.0288		Mattie Ann Andrews	BROGO Minerals, LLC	2016-00016236 Texas	Henderson
42.213.14100.0289	9/13/2015	Harry N. Scott et al	BROGO Minerals, LLC	2016-00016238 Texas	Henderson
42.213.14100.0290		Triple B Company LLC by Paul J. Korte Its President	BROGO Minerals, LLC	2016-00014447 Texas	Henderson
42.213.14100.0291		LGB Holding Co LLC by Paul J. Korte Its President	BROGO Minerals, LLC	2016-00014446 Texas	Henderson
42.213.14100.0292		Cheryl Chipley Lambright	BROGO Minerals, LLC	2016-00016234 Texas	Henderson
42.213.14100.0293		Larry Glynn Chipley	BROGO Minerals, LLC	2016-00016233 Texas	Henderson
42.213.14100.0294		Guy S. Sovia and Linda M. Sovia	BROGO Minerals, LLC	2016-00016240 Texas	Henderson
42.213.14100.0295		Madeline Dinwiddie Hanrahan	BROGO Minerals, LLC	2016-00016249 Texas	Henderson
42.213.14100.0296		Betty B. Boswell	BROGO Minerals, LLC	2016-00016248 Texas	Henderson
42.213.14100.0297		Evelyn Barron Robinson	BROGO Minerals, LLC	2016-00016247 Texas	Henderson
42.213.14100.0298		Ruby L. Pirtle	BROGO Minerals, LLC	2016-00016246 Texas 2016-00016245 Texas	Henderson
42.213.14100.0299 42.213.14100.0300		Justin J. Alexander Mark Alan Sullivan	BROGO Minerals, LLC BROGO Minerals, LLC	2016-00016245 Texas 2016-00016244 Texas	Henderson Henderson
42.213.14100.0300		Reid A. Bond	BROGO Minerals, LLC	2016-00016244 Texas	Henderson
42.213.14100.0301		Sandra Chipley Kellogg	BROGO Minerals, LLC	2016-00016242 Texas	Henderson
42.213.14100.0302		Danny F. Jones	BROGO Minerals, LLC	2016-00016242 Texas	Henderson
42.213.14100.0303		Richard A. Dinwiddie	BROGO Minerals, LLC	2016-00016250 Texas	Henderson
42.213.14100.0305		Rebecca D. Kunneman	BROGO Minerals, LLC	2016-00017837 Texas	Henderson
42.213.14100.0306		Susan Kathleen Sullivan	BROGO Minerals, LLC	2016-00016251 Texas	Henderson
42.213.14100.0307		Lucy Yerly Walters by Lucy Yerly Walters, LLC	BROGO Minerals, LLC	2017-00005878 Texas	Henderson
42.213.14100.0308		Peter M. Tumball for Hat Oil & Gas, LP a Texas Limited Partnership	BROGO Minerals, LLC	2017-00000341 Texas	Henderson
42.213.14100.0309		Ted Fritz Royston	BROGO Minerals, LLC	2016-00017839 Texas	Henderson
42.213.14100.0310		Robert E. Gibson III, by Steven H. Gibson Agt & AIF	BROGO Minerals, LLC	2016-00017838 Texas	Henderson
42.213.14100.0311		Cindy J. Booth	BROGO Minerals, LLC	2017-00004229 Texas	Henderson
42.213.14100.0312		James W. Westfall 1993 Revocable Trust, dated 12/14/1993, Nancy Westfall as Trustee	BROGO Minerals, LLC	2017-00005853 Texas	Henderson
42.213.14100.0313		Virginia Dinwiddie Parker	BROGO Minerals, LLC	2016-00017841 Texas	Henderson
42.213.14100.0313					
42.213.14100.0314	10/12/2016	Karleen Neill Raper	BROGO Minerals, LLC	2016-00017844 Texas	Henderson
		Karleen Neill Raper Walter R. Neill, Jr.	BROGO Minerals, LLC BROGO Minerals, LLC	2016-00017844 Texas 2016-00017843 Texas	Henderson
42.213.14100.0314	10/12/2016				
42.213.14100.0314 42.213.14100.0315	10/12/2016 9/7/2016	Walter R. Neill, Jr.	BROGO Minerals, LLC	2016-00017843 Texas	Henderson
42.213.14100.0314 42.213.14100.0315 42.213.14100.0316	10/12/2016 9/7/2016 9/13/2016	Walter R. Neill, Jr. Mary Youmans Udouj	BROGO Minerals, LLC BROGO Minerals, LLC	2016-00017843 Texas 2016-00017842 Texas	Henderson Henderson
42.213.14100.0314 42.213.14100.0315 42.213.14100.0316 42.213.14100.0317	10/12/2016 9/7/2016 9/13/2016 9/13/2016	Walter R. Neill, Jr. Mary Youmans Udouj Lori Bardwell Halbert	BROGO Minerals, LLC BROGO Minerals, LLC BROGO Minerals, LLC	2016-00017843 Texas 2016-00017842 Texas 2016-00017851 Texas	Henderson Henderson Henderson
42.213.14100.0314 42.213.14100.0315 42.213.14100.0316 42.213.14100.0317 42.213.14100.0318	10/12/2016 9/7/2016 9/13/2016 9/13/2016 10/18/2016	Walter R. Neill, Jr. Mary Youmans Udouj Lori Bardwell Halbert Robert Carey Bardwell	BROGO Minerals, LLC BROGO Minerals, LLC BROGO Minerals, LLC BROGO Minerals, LLC	2016-00017843 Texas 2016-00017842 Texas 2016-00017851 Texas 2016-00017846 Texas	Henderson Henderson Henderson
42.213.14100.0314 42.213.14100.0315 42.213.14100.0316 42.213.14100.0317 42.213.14100.0318 42.213.14100.0319	10/12/2016 9/7/2016 9/13/2016 9/13/2016 10/18/2016 9/30/2016	Walter R. Neill, Jr. Mary Youmans Udouj Lori Bardwell Halbert Robert Carey Bardwell RVS Minerals LLC by Margaret A. McDaniel Its Manger	BROGO Minerals, LLC	2016-00017843 Texas 2016-00017842 Texas 2016-00017851 Texas 2016-00017846 Texas 2016-00017850 Texas	Henderson Henderson Henderson Henderson
42.213.14100.0314 42.213.14100.0315 42.213.14100.0316 42.213.14100.0317 42.213.14100.0318 42.213.14100.0319 42.213.14100.0320	10/12/2016 9/7/2016 9/13/2016 9/13/2016 10/18/2016 9/30/2016 10/24/2016	Walter R. Neill, Jr. Mary Youmans Udouj Lori Bardwell Halbert Robert Carey Bardwell RVS Minerals LLC by Margaret A. McDaniel Its Manger Neill Oil, LLC by Virginia B. Price its Manager	BROGO Minerals, LLC	2016-00017843 Texas 2016-00017842 Texas 2016-00017851 Texas 2016-00017846 Texas 2016-00017850 Texas 2016-00017849 Texas	Henderson Henderson Henderson Henderson Henderson
42.213.14100.0314 42.213.14100.0315 42.213.14100.0316 42.213.14100.0317 42.213.14100.0318 42.213.14100.0320 42.213.14100.0320 42.213.14100.0321	10/12/2016 9/7/2016 9/13/2016 9/13/2016 10/18/2016 9/30/2016 10/24/2016	Walter R. Neill, Jr. Mary Youmans Udouj Lori Bardwell Halbert Robert Carey Bardwell RVS Minerals LLC by Margaret A. McDaniel Its Manger Neill Oil, LLC by Virginia B. Price its Manager Ridgway Management, Inc. by Bill Ridgway its Manager	BROGO Minerals, LLC	2016-00017843 Texas 2016-00017842 Texas 2016-00017851 Texas 2016-00017846 Texas 2016-00017850 Texas 2016-00017849 Texas 2016-00017856 Texas	Henderson Henderson Henderson Henderson Henderson Henderson
42.213.14100.0314 42.213.14100.0315 42.213.14100.0316 42.213.14100.0317 42.213.14100.0318 42.213.14100.0320 42.213.14100.0320 42.213.14100.0321 42.213.14100.0322	10/12/2016 9/7/2016 9/13/2016 9/13/2016 10/18/2016 9/30/2016 10/24/2016 10/18/2016	Walter R. Neill, Jr. Mary Youmans Udouj Lori Bardwell Halbert Robert Carey Bardwell RVS Minerals LLC by Margaret A. McDaniel Its Manger Neill Oil, LLC by Virginia B. Price its Manager Ridgway Management, Inc. by Bill Ridgway its Manager Robert Tidgren	BROGO Minerals, LLC	2016-00017843 Texas 2016-00017842 Texas 2016-00017846 Texas 2016-00017846 Texas 2016-00017850 Texas 2016-00017849 Texas 2016-00017848 Texas 2016-00017848 Texas	Henderson Henderson Henderson Henderson Henderson Henderson Henderson
42.213.14100.0314 42.213.14100.0315 42.213.14100.0316 42.213.14100.0317 42.213.14100.0318 42.213.14100.0319 42.213.14100.0320 42.213.14100.0321 42.213.14100.0322 42.213.14100.0323	10/12/2016 9/7/2016 9/13/2016 9/13/2016 10/18/2016 9/30/2016 10/24/2016 10/18/2016 8/9/2016	Walter R. Neill, Jr. Mary Youmans Udouj Lori Bardwell Halbert Robert Carey Bardwell RVS Minerals LLC by Margaret A. McDaniel Its Manger Neill Oil, LLC by Virginia B. Price its Manager Ridgway Management, Inc. by Bill Ridgway its Manager Robert Tidgren Douglas C. Koch	BROGO Minerals, LLC	2016-00017843 Texas 2016-00017842 Texas 2016-00017846 Texas 2016-00017850 Texas 2016-00017849 Texas 2016-00017849 Texas 2016-00017848 Texas 2016-00017847 Texas	Henderson Henderson Henderson Henderson Henderson Henderson Henderson Henderson

Lease #	Date Lessor	Lessee	Book-Page State	County
42.213.14100.0327	10/25/2016 Horseshoe Minerals, Inc. by Harry S. Armstrong	BROGO Minerals, LLC	2016-00017853 Texas	Henderson
42.213.14100.0328	10/3/2016 Beatrice Nettle	BROGO Minerals, LLC	2016-00017854 Texas	Henderson
42.213.14100.0329	9/23/2016 James A. Dinwiddie	BROGO Minerals, LLC	2017-00000342 Texas	Henderson
42.213.14100.0330	8/29/2016 Gary Quin Ellis	BROGO Minerals, LLC	2017-00000344 Texas	Henderson
42.213.14100.0331	8/22/2016 William Edward Wall	BROGO Minerals, LLC	2017-00000343 Texas	Henderson
42.213.14100.0332	11/3/2016 Suzanne Hupfeld	BROGO Minerals, LLC	2016-00017835 Texas	Henderson
42.213.14100.0333	9/14/2016 The Norman Family Revocable Living Tr by: Kay Horton Its: Trustee	BROGO Minerals, LLC	2016-00017834 Texas	Henderson
42.213.14100.0334	9/30/2016 LaRue 1992 Trust by: Sharon Allen LaRue, Camille LaRue Olsen, Elaine LaRue Colley, Annette LaRue Morgan and Robert Colley Co-	BROGO Minerals, LLC	2016-00017487 Texas	Henderson
42.213.14100.0335	11/22/2016 Debra Lynn Slaughter	BROGO Minerals, LLC	2017-00000363 Texas	Henderson
42.213.14100.0336	9/23/2016 Elizabeth Dinwiddie Edwards	BROGO Minerals, LLC	2017-00000356 Texas	Henderson
42.213.14100.0337	11/14/2016 Wayne Fulton Brown	BROGO Minerals, LLC	2017-00000359 Texas	Henderson
42.213.14100.0338	11/22/2016 Cynthia Crawford Zajic	BROGO Minerals, LLC	2017-00000362 Texas	Henderson
42.213.14100.0339	1/30/2017 Karol B. Williamson Trustee for Karol B. Williamson Tr	BROGO Minerals, LLC	2017-00004209 Texas	Henderson
42.213.14100.0340	11/23/2016 Kelly Browning Henager, Jan Marie Browning and Sally Browning, as Life Tenant	BROGO Minerals, LLC	2017-00000354 Texas	Henderson
42.213.14100.0341	8/29/2016 Tom O. Moncrief 1967 Tr by Tom O. Moncrief, Trustee	BROGO Minerals, LLC	2017-00000360 Texas	Henderson
42.213.14100.0342	11/10/2016 William Grant Banchard III	BROGO Minerals, LLC	2017-00000364 Texas	Henderson
42.213.14100.0343	11/3/2016 Kelly Hupfeld Merkur	BROGO Minerals, LLC	2017-00000355 Texas	Henderson
42.213.14100.0344	10/3/2016 Mary Masterson McIntosh	BROGO Minerals, LLC	2017-00000357 Texas	Henderson
42.213.14100.0345	10/3/2016 Horace A. Masterson	BROGO Minerals, LLC	2017-00000358 Texas	Henderson
42.213.14100.0346	8/15/2016 Shawn West Webb	BROGO Minerals, LLC	2017-00000352 Texas	Henderson
42.213.14100.0347	8/15/2016 Shawn West Webb	BROGO Minerals, LLC	2017-00000353 Texas	Henderson
42.213.14100.0348	8/15/2016 The Mark V West Testamentary Tr c/o Texas Bank & Tr Company	BROGO Minerals, LLC	2017-00000348 Texas	Henderson
42.213.14100.0349	3/9/2017 Paul Adron Neill and Diane Neill	BROGO Minerals, LLC	2017-00004202 Texas	Henderson
42.213.14100.0350	3/7/2017 George Henry Neill	BROGO Minerals, LLC	2017-00004210 Texas	Henderson
42.213.14100.0351	8/15/2016 Shawn West Webb	BROGO Minerals, LLC	2017-00000351 Texas	Henderson
42.213.14100.0352	11/16/2016 Randall Scott Browning	BROGO Minerals, LLC	2017-00000371 Texas	Henderson
42.213.14100.0353	8/15/2016 Shelly West Bowman	BROGO Minerals, LLC	2017-00000345 Texas	Henderson
42.213.14100.0354	8/15/2016 Shelly West Bowman	BROGO Minerals, LLC	2017-00000347 Texas	Henderson
42.213.14100.0355	8/15/2016 Shelly West Bowman	BROGO Minerals, LLC	2017-00000346 Texas	Henderson
42.213.14100.0358	8/15/2016 The Mark V West Testamentary Tr c/o Texas Bank & Tr Company	BROGO Minerals, LLC	2017-00000350 Texas	Henderson
42.213.14100.0359 42.213.14100.0360	8/15/2016 The Mark V West Testamentary Tr c/o Texas Bank & Tr Company 11/7/2016 Randy Lynn Paris	BROGO Minerals, LLC BROGO Minerals, LLC	2017-00000359 Texas 2017-00000391 Texas	Henderson Henderson
42.213.14100.0361	11/29/2016 Jo Katherine Walker Well	BROGO Minerals, LLC	2017-00000031 Texas	Henderson
42.213.14100.0362	11/29/2016 Huey Ray Wilbanks and Neil Walker Wilbanks	BROGO Minerals, LLC	2017-00000361 Texas	Henderson
42.213.14100.0363	11/23/2016 Clark Family Partnership by Mildred C. Gronlund, Manager	BROGO Minerals, LLC	2017-00000380 Texas	Henderson
42.213.14100.0364	11/22/2016 Linda E. Walker	BROGO Minerals, LLC	2017-00000361 Texas	Henderson
42.213.14100.0365	11/7/2016 Billy Dean Paris	BROGO Minerals, LLC	2017-00000377 Texas	Henderson
42.213.14100.0366	11/7/2016 Margie Basher Ward	BROGO Minerals, LLC	2017-00000378 Texas	Henderson
42.213.14100.0367	11/14/2016 Jane Elizabeth Maxwell and Thomas M. Maxwell, Trustees for The Jane E. Maxwell Trust No. 1 dated April 1, 2013	BROGO Minerals, LLC	2017-00009680 Texas	Henderson
42.213.14100.0368	11/14/2016 Patricia Roxanne Young	BROGO Minerals, LLC	2017-00000385 Texas	Henderson
42.213.14100.0369	11/14/2016 June Suzette Snyder	BROGO Minerals, LLC	2017-00000386 Texas	Henderson
42.213.14100.0370	12/1/2016 Tony Ray Buford	BROGO Minerals, LLC	2017-00000387 Texas	Henderson
42.213.14100.0371	8/15/2016 The Catherine Coleman Murchison Ex Tr By: George R. Bedell Its Trustee	BROGO Minerals, LLC	2017-00000367 Texas	Henderson
42.213.14100.0372	8/15/2016 The Catherine Coleman Murchison Ex Tr By: George R. Bedell Its Trustee	BROGO Minerals, LLC	2017-00000372 Texas	Henderson
42.213.14100.0373	8/15/2016 The Catherine Coleman Murchison Ex Tr By: George R. Bedell Its Trustee	BROGO Minerals, LLC	2017-00000373 Texas	Henderson
42.213.14100.0374	8/15/2016 Wynne West Anderson	BROGO Minerals, LLC	2017-00000370 Texas	Henderson
42.213.14100.0375	8/15/2016 Wynne West Anderson	BROGO Minerals, LLC	2016-00011667 Texas	Henderson
42.213.14100.0376	8/15/2016 Wynne West Anderson	BROGO Minerals, LLC	2016-00011668 Texas	Henderson
42.213.14100.0377	11/17/2016 Buttram Energies, Inc. by Kristen Berry VP	BROGO Minerals, LLC	2017-00000325 Texas	Henderson
42.213.14100.0378	11/17/2016 Buttram Energies, Inc. by Kristen Berry VP	BROGO Minerals, LLC	2017 Texas	Henderson
42.213.14100.0379	12/2/2016 The Adele Blanchard Goodwin Revocable Living Tr by Susan Adele Goodwin Trustee	BROGO Minerals, LLC	2016-00000390 Texas	Henderson
42.213.14100.0380	12/6/2016 MCW Royalties, a Texas General Partnership by Frank Wheelock, Manager	BROGO Minerals, LLC	2017-00000388 Texas	Henderson
42.213.14100.0382	12/14/2016 Jo Anne Burnett Cooper	BROGO Minerals, LLC	2017-00000372 Texas	Henderson
42.213.14100.0383	12/12/2016 WCP Energy, Inc. by W. C. "Chip Perryman III Its President	BROGO Minerals, LLC	2017-00000383 Texas	Henderson
42.213.14100.0384	12/12/2016 W. C. "Chip" Perryman III	BROGO Minerals, LLC	2017-00000384 Texas	Henderson
42.213.14100.0385	11/3/2016 Emily Glenn McDonald	BROGO Minerals, LLC	2017-00000381 Texas	Henderson
42.213.14100.0386	11/3/2016 Paxson Glenn, Sr. and Marion Ellen Glenn	BROGO Minerals, LLC	2016-00014461 Texas	Henderson
42.213.14100.0388	11/7/2016 Ronnie Basher	BROGO Minerals, LLC	2017-00001452 Texas	Henderson
42.213.14100.0389	11/8/2016 Valeria Tucker Chachere	BROGO Minerals, LLC	2017-00001451 Texas	Henderson
42.213.14100.0390	11/29/2016 C&A Royalties L.P. by Willam R. Seals III Its General Partner	BROGO Minerals, LLC	2017-00001450 Texas	Henderson
42.213.14100.0391	12/2/2016 David P. Sides Indiv and as Admin for the Est of Rebecca Ann Sides	BROGO Minerals, LLC	2017-00001449 Texas	Henderson
42.213.14100.0392	12/13/2016 Sally Jacoby Veazey	BROGO Minerals, LLC	2017-00001448 Texas	Henderson
42.213.14100.0393	12/14/2016 Karleen Neill Raper	BROGO Minerals, LLC	2017-00001443 Texas	Henderson
42.213.14100.0394	12/15/2016 Robbye Bowman Harris	BROGO Minerals, LLC	2017-00001447 Texas	Henderson
42.213.14100.0395	12/15/2016 Frankie Jean Bowman Kirkwood	BROGO Minerals, LLC	2017-00001446 Texas	Henderson

		·			
Lease #	Date	Lessor	Lessee	Book-Page State	County
42.213.14100.0396		Dorothy Bowman Perry	BROGO Minerals, LLC	2017-00001445 Texas	Henderson
42.213.14100.0397		Mildred Benford Elam	BROGO Minerals, LLC	2017-00001444 Texas	Henderson
42.213.14100.0398		Thomas Nathan Frizzell, Jr.	BROGO Minerals, LLC	2017-00001442 Texas	Henderson
42.213.14100.0399		Fred D. Vasbinder and Kimberly Vasbinder	BROGO Minerals, LLC	2017-00001441 Texas	Henderson
42.213.14100.0400	11/6/2016	Wilma Terry Finley Fam Tr by Murry "Pap" Roark, Trustee	BROGO Minerals, LLC	2017-00001440 Texas	Henderson
42.213.14100.0405	1/11/2017	Mary Elizabeth Carberry	BROGO Minerals, LLC	2017-00001457 Texas	Henderson
42.213.14100.0406		Elizabeth W. Neill	BROGO Minerals, LLC	2017-00001466 Texas	Henderson
42.213.14100.0407		Jennifer M. Wise	BROGO Minerals, LLC	2017-00001467 Texas	Henderson
42.213.14100.0408		Wanda Bowman Lee	BROGO Minerals, LLC	2017-00001464 Texas	Henderson
42.213.14100.0409	12/15/2016	Shelley R. Berry et vir	BROGO Minerals, LLC	2017-00001465 Texas	Henderson
42.213.14100.0410	12/15/2016	Lynda Janell Hobbs	BROGO Minerals, LLC	2017-00001463 Texas	Henderson
42.213.14100.0411	12/15/2016	Margaret Hobbs	BROGO Minerals, LLC	2017-00001462 Texas	Henderson
42.213.14100.0412		Jeannine Pickle Kirkland	BROGO Minerals, LLC	2017-00001461 Texas	Henderson
42.213.14100.0413	10/3/2016	Margaret Masterson Testmentary Trust by Thomas W. Masterson III Trustee	BROGO Minerals, LLC	2017-00001460 Texas	Henderson
42.213.14100.0414	1/11/2017	Mary Elizabeth Carberry	BROGO Minerals, LLC	2017-00001458 Texas	Henderson
42.213.14100.0415		Don Ray Christopher and Jane Dillaplain Christopher	BROGO Minerals, LLC	2017-00001459 Texas	Henderson
42.213.14100.0416	1/30/2017	Kay L. Bass Trust	BROGO Minerals, LLC	2017-00004208 Texas	Henderson
42.213.14100.0417	12/15/2016	Ira Jean Royall Shaw	BROGO Minerals, LLC	2017-00004206 Texas	Henderson
42.213.14100.0418	3/9/2017	Paul Adron Neill and Diane Neill	BROGO Minerals, LLC	2017-00004201 Texas	Henderson
42.213.14100.0419	3/7/2017	Gerald Forester and Harriette Ann Forrester	BROGO Minerals, LLC	2017-00004203 Texas	Henderson
42.213.14100.0420	11/22/2016	Chad Walker	BROGO Minerals, LLC	2017-00000361 Texas	Henderson
42.213.14100.0421	1/18/2017	Citizens National Bank Trustee for Elizabeth Dunn Talley Revocable Tr	O'Brien Resources, LLC	2017-00002196 Texas	Henderson
42.213.14100.0422	1/18/2017	Citizens National Bank Trustee for Peyton Dunn Revocable Trust	O'Brien Resources, LLC	2017-00002786 Texas	Henderson
42.213.14100.0423	1/18/2017	Citizens National Bank Trustee for Susan Jane Wheelock Trust	O'Brien Resources, LLC	2017-00002194 Texas	Henderson
42.213.14100.0424	1/18/2017	Tucker B. Royall Trust	O'Brien Resources, LLC	2017-00002192 Texas	Henderson
42.213.14100.0425	1/18/2017	Frances Royall Alexander Tr	O'Brien Resources, LLC	2017-00002193 Texas	Henderson
42.213.14100.0426	1/18/2017	Rebecca Royall Tr	O'Brien Resources, LLC	2017-00002191 Texas	Henderson
42.213.14100.0427	1/17/2017	Brian K. Christopher	BROGO Minerals, LLC	2017-00003173 Texas	Henderson
42.213.14100.0428	1/17/2017	Kirk Russell Christopher	BROGO Minerals, LLC	2017-00003174 Texas	Henderson
42.213.14100.0429	12/12/2016	Nolan Ray Jensen	BROGO Minerals, LLC	2017-00003176 Texas	Henderson
42.213.14100.0430	12/14/2016	Walter R. Neill Jr	BROGO Minerals, LLC	2017-00000430 Texas	Henderson
42.213.14100.0431	12/12/2016	Mark C. Roberts	BROGO Minerals, LLC	2017-00001470 Texas	Henderson
42.213.14100.0432		Winnie Sjogren by Albert Lillie Agt & AIF	BROGO Minerals, LLC	2017-00001469 Texas	Henderson
42.213.14100.0434		Robert Pat Smith, Jr.	BROGO Minerals, LLC	2017-00003162 Texas	Henderson
42.213.14100.0435	1/16/2017	Ellen Rowan Justice	BROGO Minerals, LLC	2017-00003163 Texas	Henderson
42.213.14100.0436		Thomas Mack Bowman Jr	BROGO Minerals, LLC	2017-00003164 Texas	Henderson
42.213.14100.0437		Kevin Bernard Bowman	BROGO Minerals, LLC	2017-00003165 Texas	Henderson
42.213.14100.0438		Oweta Powell Bowman	BROGO Minerals, LLC	2017-00003166 Texas	Henderson
42.213.14100.0440	12/14/2016	Jackie Lou Phillips	BROGO Minerals, LLC	2017-00003168 Texas	Henderson
42.213.14100.0441	12/14/2016	Dickey Family Tr dated 7-3-2006 by Bill O. Dickey and Lynn C. Dicket, Its Co-Trustees	BROGO Minerals, LLC	2017-00003169 Texas	Henderson
42.213.14100.0442	12/14/2016	Tanya Ann Jackson Irrevocable Tr by Meredith Jackson Its Successor Trustee	BROGO Minerals, LLC	2017-00003170 Texas	Henderson
42.213.14100.0443	12/14/2016	Annette Marie Faulk	BROGO Minerals, LLC	2017-00003171 Texas	Henderson
42.213.14100.0444		Susan J. Pirtle	BROGO Minerals, LLC	2017-00000384 Texas	Henderson
42.213.14100.0445		Tracy Lance Edwards	BROGO Minerals, LLC	2017-00009678 Texas	Henderson
42.213.14100.0446	1/10/2017	Gary Christian and Beth Christian	BROGO Minerals, LLC	2017-00003175 Texas	Henderson
42.213.14100.0447	1/31/2017	Nancy Nell Cato Bullock	BROGO Minerals, LLC	2017-00003189 Texas	Henderson
42.213.14100.0448	1/17/2017	Don Ray Christopher and Jane Dillaplain Christopher	BROGO Minerals, LLC	2017-00003172 Texas	Henderson
42.213.14100.0449	2/1/2017	Jerry Ray Latimer	BROGO Minerals, LLC	2017-00001950 Texas	Henderson
42.213.14100.0450	1/30/2017	Leneata Morgan Hall	BROGO Minerals, LLC	2017-00003179 Texas	Henderson
42.213.14100.0451	2/3/2017	Jean Adams Marital Tr no. Two - GST Non- Exempt et al	BROGO Minerals, LLC	2017-00003180 Texas	Henderson
42.213.14100.0452	12/5/2016	Sherry Werneking	BROGO Minerals, LLC	2017-00003177 Texas	Henderson
42.213.14100.0453	2/7/2017	Melissa Walker Penix	BROGO Minerals, LLC	2017-00003178 Texas	Henderson
42.213.14100.0454	12/14/2016	Joan Miller Ganz, Indiv & Successor Trustee W/W/O Mary Duke Pearlbrook, Deceased et al and John W. Lyttle, Jr. Indiv and as Successor	BROGO Minerals, LLC	2017-00003191 Texas	Henderson
42.213.14100.0455		Connie Dee Thornton	BROGO Minerals, LLC	2017-00003190 Texas	Henderson
42.213.14100.0456	12/15/2016	Tawana Joy Royall	BROGO Minerals, LLC	2017-00003183 Texas	Henderson
42.213.14100.0457	12/15/2016	Myra Bowman Burns	BROGO Minerals, LLC	2017-00003182 Texas	Henderson
42.213.14100.0458	2/27/2017	FDIC for Franklin Bank SSB by David Stell AIF	BROGO Minerals, LLC	2017-00003184 Texas	Henderson
42.213.14100.0459	2/28/2017	FDIC for Franklin Bank SSB, by David Stell, AIF	BROGO Minerals, LLC	2017-00003185 Texas	Henderson
42.213.14100.0460	2/15/2017	Eugene Robinson III	BROGO Minerals, LLC	2017-00003188 Texas	Henderson
42.213.14100.0461	2/15/2017	Suzanne Robinson Shumpert	BROGO Minerals, LLC	2017-00003187 Texas	Henderson
42.213.14100.0462	2/15/2017	David E. Robinson	BROGO Minerals, LLC	2017-00003186 Texas	Henderson
42.213.14100.0465	2/15/2017	Jill Denise Robinson	BROGO Minerals, LLC	2017-00004197 Texas	Henderson
42.213.14100.0466	3/2/2017	Annette Marie Faulk	BROGO Minerals, LLC	2017-00004195 Texas	Henderson
42.213.14100.0467	2/28/2017	Richard Keith Bartlett	BROGO Minerals, LLC	2017-00004191 Texas	Henderson
42.213.14100.0468	2/28/2017	Susan B. Kelley	BROGO Minerals, LLC	2017-00004193 Texas	Henderson

Lease #	Date	Lessor	Lessee	Book-Page State	County
42.213.14100.0471	2/7/2017	Michael Walker	BROGO Minerals, LLC	2017-00004190 Texas	Henderson
42.213.14100.0472	2/20/2017	Bryan Wayne Bartlett	BROGO Minerals, LLC	2017-00004196 Texas	Henderson
42.213.14100.0473	6/14/2017	Kristin LaRue Perryman	BROGO Minerals, LLC	2017-00009679 Texas	Henderson
42.213.14100.0474	2/28/2017	Allen B. Boswell	BROGO Minerals, LLC	2017-00004194 Texas	Henderson
42.213.14100.0475	3/6/2017	Johnny Frank Allen and Besita Allen	BROGO Minerals, LLC	2017-00004198 Texas	Henderson
42.213.14100.0476	3/8/2017	FDIC Receiver for Franklin Bank SSB by David Stell, AIF	BROGO Minerals, LLC	2017-00004200 Texas	Henderson
42.213.14100.0477	3/1/2017	Jon De Montreville	BROGO Minerals, LLC	2017-00004204 Texas	Henderson
42.213.14100.0478	2/28/2017	Sherry Werneking	BROGO Minerals, LLC	2017-00004205 Texas	Henderson
42.213.14100.0479	3/13/2017	Sandra Van Wyk	BROGO Minerals, LLC	2017-00004603 Texas	Henderson
42.213.14100.0480	3/16/2017	Paul Adron Neill	BROGO Minerals, LLC	2017-00004605 Texas	Henderson
42.213.14100.0481	5/12/2017	Faye Edwards Harris	BROGO Minerals, LLC	2017-00009677 Texas	Henderson
42.213.14100.0482	3/9/2017	Anita Murray	BROGO Minerals, LLC	2017-00004602 Texas	Henderson
42.213.14100.0483	3/16/2017	Tina Fisher Ward	BROGO Minerals, LLC	2017-00004604 Texas	Henderson
42.213.14100.0484	12/15/2016	Delores Royall Foggle	BROGO Minerals, LLC	2017-00005050 Texas	Henderson
42.213.14100.0485	2/20/2017	Raymond Lewis Bartlett, III	BROGO Minerals, LLC	2017-00005049 Texas	Henderson
42.213.14100.0486	3/7/2017	Winfred E. Adams	BROGO Minerals, LLC	2017-00005048 Texas	Henderson
42.213.14100.0487	3/7/2017	Quinclyn L. Hillier, fka Quinclyn L. Roland	BROGO Minerals, LLC	2017-00005047 Texas	Henderson
42.213.14100.0488	3/17/2017	Don E. Watt, Jr.	BROGO Minerals, LLC	2017-00005046 Texas	Henderson
42.213.14100.0489	3/17/2017	Sally Ann McCreary	BROGO Minerals, LLC	2017-00005045 Texas	Henderson
42.213.14100.0490	3/17/2017	Richard Dennis Watt	BROGO Minerals, LLC	2017-00005044 Texas	Henderson
42.213.14100.0491	3/23/2017	Marjorie Perry McGannon	BROGO Minerals, LLC	2017-00005043 Texas	Henderson
42.213.14100.0492	3/23/2017	Linda J. Alexander	BROGO Minerals, LLC	2017-00005042 Texas	Henderson
42.213.14100.0493	3/23/2017	Charlotte Ann Morris	BROGO Minerals, LLC	2017-00005041 Texas	Henderson
42.213.14100.0494	3/23/2017	Barbara Barber	BROGO Minerals, LLC	2017-00005040 Texas	Henderson
42.213.14100.0495	3/24/2017	Polly Gavord Pena	BROGO Minerals, LLC	2017-00005039 Texas	Henderson
42.213.14100.0496	3/30/2017	Dimitri Mataragas as General Partner for Therma, LTD	BROGO Minerals, LLC	2017-00005877 Texas	Henderson
42.213.14100.0497	3/27/2017	Bruce Monning as Manager for Monning & Wynne, PLLC	BROGO Minerals, LLC	2017-00005038 Texas	Henderson
42.213.14100.0498	3/28/2017	Matthew Lynn Ottaberry	BROGO Minerals, LLC	2017-00005052 Texas	Henderson
42.213.14100.0499	3/28/2017	C. Michael Williams	BROGO Minerals, LLC	2017-00005053 Texas	Henderson
42.213.14100.0500	3/8/2017	Stephen Edward McAtee	BROGO Minerals, LLC	2017-00005036 Texas	Henderson
42.213.14100.0501	3/27/2017	Roy J. Linn	BROGO Minerals, LLC	2017-00005037 Texas	Henderson
42.213.14100.0502	3/15/2017	Marjorie Miller as Trustee for The Miller Family Trust	BROGO Minerals, LLC	2017-00005876 Texas	Henderson
42.213.14100.0503	3/27/2017	William L. Graham as Trustee for Edwin L. Cox, Jr. Trust #12	O'Brien Resources, LLC	2017-00005861 Texas	Henderson
42.213.14100.0504	3/29/2017	Cynthia Coldwell Phipps	BROGO Minerals, LLC	2017-00005034 Texas	Henderson
42.213.14100.0505	3/23/2017	Thomas Turner	BROGO Minerals, LLC	2017-00005035 Texas	Henderson
42.213.14100.0506	3/28/2017	Michael L. Ottaberry	BROGO Minerals, LLC	2017-00005051 Texas	Henderson
42.213.14100.0507	4/3/2017	Gail Delaine Parmer Morris	BROGO Minerals, LLC	2017-00005875 Texas	Henderson
42.213.14100.0508		William Turner	BROGO Minerals, LLC	2017-00005870 Texas	Henderson
42.213.14100.0509	3/22/2017	Melissa Jo Pirtle Hollingsworth	BROGO Minerals, LLC	2017-00005867 Texas	Henderson
42.213.14100.0510	3/23/2017	Felecia Turner Nemons	BROGO Minerals, LLC	2017-00005869 Texas	Henderson
42.213.14100.0511	4/3/2017	Suzanne Cornelius Krause	BROGO Minerals, LLC	2017-00005871 Texas	Henderson
42.213.14100.0512	7/20/2016	Martha Lou Mohler as Individually and as Trustee for Thoams C. Mohler, Benjamin Thomas Mohler, Rachael E. Mohler and	BROGO Minerals, LLC	2017-00005862 Texas	Henderson
42.213.14100.0513		James and Gail Saia, Trustees for James and Gail Saia Family Trust	BROGO Minerals, LLC	2017-00005863 Texas	Henderson
42.213.14100.0514		Kay Bateman	BROGO Minerals, LLC	2017-00005864 Texas	Henderson
42.213.14100.0515		Beverly G. Skelton	BROGO Minerals, LLC	2017-00005872 Texas	Henderson
42.213.14100.0516		Christopher Turner	BROGO Minerals, LLC	2017-00005868 Texas	Henderson
42.213.14100.0517		Cristi G. Joffrion	BROGO Minerals, LLC	2017-00005860 Texas	Henderson
42.213.14100.0518		Linda McAtee Secrest	BROGO Minerals, LLC	2017-00005865 Texas	Henderson
42.213.14100.0519		Tena L. Coldwell	BROGO Minerals, LLC	2017-00005874 Texas	Henderson
42.213.14100.0520		Frederick W. Geisendorff, Jr.	BROGO Minerals, LLC	2017-00005866 Texas	Henderson
42.213.14100.0521		James W. Westfall 1993 Revocable Trust, dated 12/14/1993, Nancy Westfall as Trustee	BROGO Minerals, LLC	2017-00005873 Texas	Henderson
42.213.14100.0522		Claudia Dickson Pruitt	BROGO Minerals, LLC	2017-00005858 Texas	Henderson
42.213.14100.0523		Deborah Dickson Russell	BROGO Minerals, LLC	2017-00005855 Texas	Henderson
42.213.14100.0524		Odis Delmon Parmer	BROGO Minerals, LLC	2017-00005854 Texas	Henderson
42.213.14100.0525		Jerry R. Vaught	BROGO Minerals, LLC	2017-00005857 Texas	Henderson
42.213.14100.0526		Fern B. Clark, President of DIL Corp., as Managing Partner for Hager Oil & Gas, a Texas General Partnership	BROGO Minerals, LLC	2017-00005856 Texas	Henderson
42.213.14100.0527		Melvarene Parmer Allison	BROGO Minerals, LLC	2017-00005859 Texas	Henderson
42.213.14100.0528		Lora Evelyn Fairris Pirtle	BROGO Minerals, LLC	2017-00006611 Texas	Henderson
42.213.14100.0529		Richard Wallace	BROGO Minerals, LLC	2017-00006614 Texas	Henderson
42.213.14100.0530		Phillip Hatton as Independent Administrator for Estate of Paul T. Pirtle, Deceased	BROGO Minerals, LLC	2017-00006613 Texas	Henderson
42.213.14100.0531		Phillip D. Hatton	BROGO Minerals, LLC	2017-00006612 Texas	Henderson
42.213.14100.0532		Joe Neal Graham	BROGO Minerals, LLC	2017-00007372 Texas	Henderson
42.213.14100.0533		Garland Pool	BROGO Minerals, LLC	2017-00006603 Texas	Henderson
42.213.14100.0534		Keith William Wallace	BROGO Minerals, LLC	2017-00006605 Texas	Henderson
42.213.14100.0535	4/24/2017	Amy Christine Cozby	BROGO Minerals, LLC	2017-00006604 Texas	Henderson

Lease #	Date	Lessor	Lessee	Book-Page State	County
42.213.14100.0536	5/2/2017	Dan B. Godwin	BROGO Minerals, LLC	2017-00007366 Texas	Henderson
42.213.14100.0537	4/24/2017	Richard Wallace	BROGO Minerals, LLC	2017-00006606 Texas	Henderson
42.213.14100.0538	4/19/2017	LaVerne Arnett	BROGO Minerals, LLC	2017-00006607 Texas	Henderson
42.213.14100.0539	3/21/2017	Lana C. Caperton	BROGO Minerals, LLC	2017-00006601 Texas	Henderson
42.213.14100.0540	4/26/2017	Walter David Stalnaker, Jr.	BROGO Minerals, LLC	2017-00006609 Texas	Henderson
42.213.14100.0541	4/26/2017	Walter David Stalnaker, Jr., Manager for Walton Springs LLC	BROGO Minerals, LLC	2017-00006610 Texas	Henderson
42.213.14100.0542	4/21/2017	Jennifer Eileen Hatton Smiley	BROGO Minerals, LLC	2017-00007367 Texas	Henderson
42.213.14100.0543	4/19/2017	Mel Jason Pirtle	BROGO Minerals, LLC	2017-00006602 Texas	Henderson
42.213.14100.0544	4/11/2017	David Royall	BROGO Minerals, LLC	2017-00006600 Texas	Henderson
42.213.14100.0545	4/21/2017	Andrew Thomas Hatton	BROGO Minerals, LLC	2017-00007511 Texas	Henderson
42.213.14100.0546	5/9/2017	Martha Jane Adair Ellis	BROGO Minerals, LLC	2017-00007505 Texas	Henderson
42.213.14100.0547	5/1/2017	Charles E. Davis	BROGO Minerals, LLC	2017-00007369 Texas	Henderson
42.213.14100.0548	5/1/2017	Virginia Baker	BROGO Minerals, LLC	2017-00007368 Texas	Henderson
42.213.14100.0549	5/15/2017	Lewie L. Travis	BROGO Minerals, LLC	2017-00007728 Texas	Henderson
42.213.14100.0550	5/15/2017	Bruce Morgan	BROGO Minerals, LLC	2017-00007727 Texas	Henderson
42.213.14100.0551	5/1/2017	John Mastin Travis, III	BROGO Minerals, LLC	2017-00009316 Texas	Henderson
42.213.14100.0552	5/1/2017	Pauline June Travis Westbrook	BROGO Minerals, LLC	2017-00009314 Texas	Henderson
42.213.14100.0553	4/21/2017	Stephen Hatton	BROGO Minerals, LLC	2017-00007726 Texas	Henderson
42.213.14100.0554	4/27/2017	Rex Cotter	BROGO Minerals, LLC	2017-00007725 Texas	Henderson
42.213.14100.0555	3/17/2017	The Termo Company	BROGO Minerals, LLC	2017-00009315 Texas	Henderson
42.213.14100.0556	5/10/2017	Jane Murchison Hamilton	BROGO Minerals, LLC	2017-0000-8127 Texas	Henderson
42.213.14100.0557		Candace Alfred Donnell	BROGO Minerals, LLC	2017-00008124 Texas	Henderson
42.213.14100.0558	5/8/2017	Jerry Burch	BROGO Minerals, LLC	2017-00008123 Texas	Henderson
42.213.14100.0559	5/12/2017	Gaylon Ozell Edwards	BROGO Minerals, LLC	2017-00008128 Texas	Henderson
42.213.14100.0560	5/17/2017	Cynthia Straus Bell	BROGO Minerals, LLC	2017-00008122 Texas	Henderson
42.213.14100.0561	5/1/2017	Judy Lynn Jones	BROGO Minerals, LLC	2017-00008126 Texas	Henderson
42.213.14100.0562	5/1/2017	Jimmy Carroll Buford	BROGO Minerals, LLC	2017-00008125 Texas	Henderson
42.213.14100.0563	5/15/2017	Betty Travis Wilkes	BROGO Minerals, LLC	2017-00009313 Texas	Henderson
42.213.14100.0564		James Dowdle	BROGO Minerals, LLC	2017-00009318 Texas	Henderson
42.213.14100.0565	6/1/2017	Ronald Paul Nelson and wife, Kaye Nelson	BROGO Minerals, LLC	2017-00009319 Texas	Henderson
42.213.14100.0566	5/1/2017	Charles L. Jones	BROGO Minerals, LLC	2017-00009322 Texas	Henderson
42.213.14100.0567		Clint M. Leake, Jr.	BROGO Minerals, LLC	2017-00009321 Texas	Henderson
42.213.14100.0568		Katherine Turner Cotter	BROGO Minerals, LLC	2017-00009320 Texas	Henderson
42.213.14100.0569		Clyde Dowdle	BROGO Minerals, LLC	2017-00009317 Texas	Henderson
42.213.14100.0570		Jean Dixon Bessent	BROGO Minerals, LLC	2017-00009310 Texas	Henderson
42.213.14100.0571		Aaron Alfred	BROGO Minerals, LLC	2017-00009674 Texas	Henderson
42.213.14100.0572		Laurette Ireland, POA for Jay Stanley Ireland	BROGO Minerals, LLC	2017-00009676 Texas	Henderson
42.213.14100.0573		W. Boyd Manning	BROGO Minerals, LLC	2017-00009309 Texas	Henderson
42.213.14100.0574		Edward De Loach Jr.	BROGO Minerals, LLC	2017-00009312 Texas	Henderson
42.213.14100.0575		Janet B. Pirtle	BROGO Minerals, LLC	2017-00009311 Texas	Henderson
42.213.14100.0576		Rebecca Jane Brown	BROGO Minerals, LLC	2017-00009675 Texas	Henderson
42.213.14100.0577	6/6/2017	Mt. Moriah Baptist Church	BROGO Minerals, LLC BROGO Minerals, LLC	2017-00010927 Texas	Henderson
42.213.14100.0578		Charles B. Davis, Jr.		2017-00010926 Texas	Henderson
42.213.14100.0579		Raymond Lewis Bartlett II	BROGO Minerals, LLC	2017-00004192 Texas	Henderson
42.213.14100.0580		Harriett Lee Davis Riggs, Trustee for The Riggs Family Trust	BROGO Minerals, LLC	2017-00010923 Texas	Henderson
42.213.14100.0581 42.213.14100.0582		Sue Madeod	BROGO Minerals, LLC	2017-00010924 Texas 2017-00012981 Texas	Henderson
42.213.14100.0582	5/8/2017	Cayce Rumsey, III Myra J. LaStrapes	BROGO Minerals, LLC BROGO Minerals, LLC	2017-00012981 Texas	Henderson Henderson
42.213.14100.0584			BROGO Minerals, LLC	2017-00010925 Texas 2017-00012982 Texas	Henderson
42.213.14100.0586		Elizabeth Ann Kight			
		Denise Peters Magaret Leake Gunn	BROGO Minerals, LLC	2017-00014399 Texas	Henderson
42.213.14100.0587 42.213.14100.3000		Magaret Leake Gunn Johnny Davis	BROGO Minerals, LLC Riverwood Exploration, LLC	2017-00014178 Texas	Henderson Henderson
				2016-00012957 Texas	
42.213.14100.3001 42.213.14100.3002		J.W. & B.A. Starr Living Trust	Riverwood Exploration, LLC Riverwood Exploration, LLC	2016-00012970 Texas 2016-00012969 Texas	Henderson
42.213.14100.3002		Ruth Sitton Clinton Davis	Riverwood Exploration, LLC Riverwood Exploration, LLC	2016-00012969 Texas 2016-00012960 Texas	Henderson Henderson
42.213.14100.3007		Gary Wayne Dennis Supan Dennis Moddock	Riverwood Exploration, LLC Riverwood Exploration, LLC	2016-00012974 Texas	Henderson
42.213.14100.3008 42.213.14100.3009		Susan Dennis Maddock Wesley Don Dennis	Riverwood Exploration, LLC Riverwood Exploration, LLC	2016-00012973 Texas 2016-00012972 Texas	Henderson Henderson
42.213.14100.3009			Riverwood Exploration, LLC Riverwood Exploration, LLC	2016-00012972 Texas 2016-00012977 Texas	Henderson
42.213.14100.3010		Gerald W. King Derrel Wall	Riverwood Exploration, LLC Riverwood Exploration, LLC	2016-00012977 Texas 2016-00012961 Texas	Henderson
42.213.14100.3015		Gerald Wall	Riverwood Exploration, LLC	2016-00012959 Texas	Henderson
42.213.14100.3016 42.213.14100.3017		Ronald Rosenthal Diago Villa intercept Lea unrecorded V	Riverwood Exploration, LLC Riverwood Exploration, LLC	2016-00012975 Texas Texas	Henderson Henderson
42.213.14100.3017		Diana Vitek (No interest - Ise unrecorded) Mary Louise Stegall	Riverwood Exploration, LLC Riverwood Exploration, LLC	2016-00013029 Texas	Henderson
2 10.14100.3010	U1 10/2010				Henderson
42.213.14100.3019	E/46/0040	Barbara H. Stegall	Riverwood Exploration, LLC	2016-00013028 Texas	

Lease #	Date	Lessor	Lessee	Book-Page State	County
42.213.14100.3020	5/17/2016	James Johnson	Riverwood Exploration, LLC	2016-00012956 Texas	Henderson
42.213.14100.3022	5/16/2016	John W. Howard, et ux	Riverwood Exploration, LLC	2016-00013425 Texas	Henderson
42.213.14100.3024	6/13/2016	Alvin C. Ford Trust	Riverwood Exploration, LLC	2016-00013042 Texas	Henderson
42.213.14100.3025	5/10/2016	Linda Wills Harris	Riverwood Exploration, LLC	2016-00013033 Texas	Henderson
42.213.14100.3033	5/13/2016	Suzanne Caudle McPherson	Riverwood Exploration, LLC	2016-00013020 Texas	Henderson
42.213.14100.3034	5/13/2016	Patricia Ann Little	Riverwood Exploration, LLC	2016-00012978 Texas	Henderson
42.213.14100.3035	5/18/2016	Sarah Katherine Menefee	Riverwood Exploration, LLC	2016-00013040 Texas	Henderson
42.213.14100.3037	6/27/2016	Gary Wayne McWilliams	Riverwood Exploration, LLC	2016-00013037 Texas	Henderson
42.213.14100.3038	6/7/2016	John Lynn McWilliams	Riverwood Exploration, LLC	2016-00012986 Texas	Henderson
42.213.14100.3039	6/7/2016	Nettie Jo Phillips	Riverwood Exploration, LLC	2016-00012987 Texas	Henderson
42.213.14100.3041	6/9/2016	Johnny Davis	Riverwood Exploration, LLC	2016-00013041 Texas	Henderson
42.213.14100.3043	6/8/2016	Patrick Carhart(sold minerals to SW Pet)	Riverwood Exploration, LLC	2016-00013030 Texas	Henderson
42.213.14100.3046	6/6/2016	Deborah LeDoux	Riverwood Exploration, LLC	2016-00013022 Texas	Henderson
42.213.14100.3047	7/1/2016	Estate of Paul Pirtle	Riverwood Exploration, LLC	2016-00013026 Texas	Henderson
42.213.14100.3048	7/1/2016	Phillip Douglas Hatton	Riverwood Exploration, LLC	2016-00013027 Texas	Henderson
42.213.14100.3050	7/1/2016	Carol Sue Sanders	Riverwood Exploration, LLC	2016-00012992 Texas	Henderson
42.213.14100.3051	6/22/2016	Gia Freeman Newell	Riverwood Exploration, LLC	2016-00013031 Texas	Henderson
42.213.14100.3053	7/8/2016	George A. Pitts	Riverwood Exploration, LLC	2016-00012981 Texas	Henderson
42.213.14100.3054	7/6/2016	Thelma E. Holt	Riverwood Exploration, LLC	2016-00015868 Texas	Henderson
42.213.14100.3055	7/1/2016	Mary E. Massey	Riverwood Exploration, LLC	2016-00013005 Texas	Henderson
42.213.14100.3056	7/1/2016	Kyle Eaton	Riverwood Exploration, LLC	2016-00012993 Texas	Henderson
42.213.14100.3057	7/1/2016	Lorenz T. Stover	Riverwood Exploration, LLC	2016-00015867 Texas	Henderson
42.213.14100.3058	7/1/2016	L.B. Eaton	Riverwood Exploration, LLC	2016-00012991 Texas	Henderson
42.213.14100.3059	6/20/2016	Carol Carmichael	Riverwood Exploration, LLC	2016-00013002 Texas	Henderson
42.213.14100.3060	7/11/2016	Dana R. Davidson	Riverwood Exploration, LLC	2016-00012996 Texas	Henderson
42.213.14100.3061	7/11/2016	Amy Sue Cox	Riverwood Exploration, LLC	2016-00012994 Texas	Henderson
42.213.14100.3062	7/11/2016	Sharon Randel Turner	Riverwood Exploration, LLC	2016-00012995 Texas	Henderson
42.213.14100.3063	7/7/2016	Rayne S. Barfield	Riverwood Exploration, LLC	2016-00013019 Texas	Henderson
42.213.14100.3064	7/20/2016	Clinton & Cassie McBride Koerth, h/w	Riverwood Exploration, LLC	2016-00013017 Texas	Henderson
42.213.14100.3065	7/25/2016	Virginia Kauffmann Head	Riverwood Exploration, LLC	2016-00012982 Texas	Henderson
42.213.14100.3067	7/20/2016	Jay and Carla Frazier, h/w	Riverwood Exploration, LLC	2016-00013018 Texas	Henderson
42.213.14100.3068	7/7/2016	Rocky S. Lindamood	Riverwood Exploration, LLC	2016-00012988 Texas	Henderson
42.213.14100.3069	6/20/2016	John R. Carmichael	Riverwood Exploration, LLC	2016-00013016 Texas	Henderson
42.213.14100.3070	7/20/2016	Sherry Carhart	Riverwood Exploration, LLC	2016-00013006 Texas	Henderson
42.213.14100.3071	7/15/2016	Peggy Jean Cochran and Gary Cochran	Riverwood Exploration, LLC	2016-00012999 Texas	Henderson
42.213.14100.3072	8/2/2016	Melissa J. Hollingsworth	Riverwood Exploration, LLC	2016-00012990 Texas	Henderson
42.213.14100.3073	8/5/2016	Mel Jason Pirtle	Riverwood Exploration, LLC	2016-00013023 Texas	Henderson
42.213.14100.3074	7/1/2016	Terry Sue Moralez	Riverwood Exploration, LLC	2016-00012998 Texas	Henderson
42.213.14100.3075	7/15/2016	Sharon Ruth Taylor, et vir	Riverwood Exploration, LLC	2016-00013000 Texas	Henderson
42.213.14100.3081	8/3/2016	Douglas Nichols Sholars	Riverwood Exploration, LLC	2016-00015862 Texas	Henderson
42.213.14100.3082	8/3/2016	Celia Juanita Smith	Riverwood Exploration, LLC	2016-00015864 Texas	Henderson
42.213.14100.3083	7/13/2016	Bobby Leon Wills	Riverwood Exploration, LLC	2016-00013004 Texas	Henderson
42.213.14100.3084	10/3/2016	BBT Enterprises, Ltd.	Riverwood Exploration, LLC	2016-00019413 Texas	Henderson
42.213.14100.3085	5/10/2016	William D. & Margaret J. Breedlove, h/w	Riverwood Exploration, LLC	2016-00012997 Texas	Henderson
42.213.14100.3086	12/5/2016	Noel Elaine Breaux	O'Brien Resources, LLC	2017 Texas	Henderson
42.213.14100.3087	11/15/2016	George Larry Howard	Riverwood Exploration, LLC	2016-00019422 Texas	Henderson
42.213.14100.3088	8/3/2016	J.W. & B.A. Starr Living Trust	Riverwood Exploration, LLC	2016-00015866 Texas	Henderson
42.213.14100.3089	8/8/2016	John Charles Cain	Riverwood Exploration, LLC	2016-00013009 Texas	Henderson
42.213.14100.3090	8/8/2016	Frank Wofford Denius	Riverwood Exploration, LLC	2016-00013007 Texas	Henderson
42.213.14100.3091	8/8/2016	Charmaine Denius McGill	Riverwood Exploration, LLC	2016-00015874 Texas	Henderson
42.213.14100.3092	7/30/2016	Diane M. Hatai	Riverwood Exploration, LLC	2016-00013010 Texas	Henderson
42.213.14100.3093	7/25/2016	Patrick Lee Pettiette	Riverwood Exploration, LLC	2016-00013013 Texas	Henderson
42.213.14100.3094	8/5/2016	Myra Pirtle LaStrapes	Riverwood Exploration, LLC	2016-00013008 Texas	Henderson
42.213.14100.3095	7/29/2016	Ronald G. McClain	Riverwood Exploration, LLC	2016-00013011 Texas	Henderson
42.213.14100.3096	8/31/2016	O'Brien 2007 Trust	Riverwood Exploration, LLC	2016-00017048 Texas	Henderson
42.213.14100.3097	8/19/2016	Donna Peters	Riverwood Exploration, LLC	2016-00013014 Texas	Henderson
42.213.14100.3098	8/19/2016	Sharon Davis	Riverwood Exploration, LLC	2016-00013015 Texas	Henderson
42.213.14100.3099	8/8/2016	William H. Brewer, Jr.	Riverwood Exploration, LLC	2016-00013012 Texas	Henderson
42.213.14100.3100	10/10/2016	Betty Boswell	Riverwood Exploration, LLC	2016-00019419 Texas	Henderson
42.213.14100.3101	8/1/2016	Fran Havard	Riverwood Exploration, LLC	2016-00015851 Texas	Henderson
42.213.14100.3102	8/15/2016	Frank Larry Stegall	Riverwood Exploration, LLC	2016-00015865 Texas	Henderson
42.213.14100.3103	8/17/2016	Larry M. Shelton	Riverwood Exploration, LLC	2016-00015860 Texas	Henderson
42.213.14100.3104	8/19/2016	Nancy Sheridan	Riverwood Exploration, LLC	2016-00015861 Texas	Henderson
42.213.14100.3105	8/19/2016	Susan B. Kelley	Riverwood Exploration, LLC	2016-00015855 Texas	Henderson
42.213.14100.3106	9/1/2016	Garland C. Pool, et ux	Riverwood Exploration, LLC	2016-00015872 Texas	Henderson

CRAINSTRUCKER CRAINSTR	Lease #	Date	Lessor	Lessee	Book-Page State	County
PAST MISSTAME PAST MISSTAME MISSTAME MARKED STATEMENT AND MISSTAME MISS	42.213.14100.3107	9/6/2016	Dolores Pool Herrington	Riverwood Exploration, LLC	2016-00015852 Texas	Henderson
25.3.1.0.1.0.1.0.0.0.0.0.0.0.0.0.0.0.0.0.0	42.213.14100.3108	9/6/2016	Garland C. Pool	Riverwood Exploration, LLC	2016-00015873 Texas	Henderson
ADDITION STATES AND STATES AND	42.213.14100.3109	8/25/2016	Dorothy Dodson	Riverwood Exploration, LLC	2016-00015849 Texas	Henderson
25.15.10.00.00.00 30.000.00 70.000.000.00 70.000.00 70.000.00 70.000.000.00	42.213.14100.3110	8/25/2016	Margaret Ryder	Riverwood Exploration, LLC	2016-00015859 Texas	Henderson
CORDITIONAL STATE OR AND PROSEDY Account primate III. C. 2000 (1980) To 10. Control C	42.213.14100.3111	8/17/2016	Kenneth C. Jones	Riverwood Exploration, LLC	2016-00015853 Texas	Henderson
42.3.3.1.1.0	42.213.14100.3112	8/23/2016	Peter W. Jones	Riverwood Exploration, LLC	2016-00015854 Texas	Henderson
2011 1000000000000000000000000000000000	42.213.14100.3113	9/1/2016	The ARJO Trust	Riverwood Exploration, LLC	2016-00015875 Texas	Henderson
EXESTIGNIONI Remons Operation IX. 2010 Months (August August Aug	42.213.14100.3114	9/8/2016	Harriette Ann Forrester	Riverwood Exploration, LLC	2016-00015850 Texas	Henderson
E3.51141003175 PRIVIDED GENERAL PLAN PRIVIDED GENERAL PLAN Common Co	42.213.14100.3115	8/29/2016	Deborah K. Church	Riverwood Exploration, LLC	2016-00015848 Texas	Henderson
GRASH (000) 1000 (0000 Front) Prior (00000 Front) Prior (0000 Fron	42.213.14100.3116	8/19/2016	John A. Peters	Riverwood Exploration, LLC	2016-00015858 Texas	Henderson
27.53 (1867) 300 Defended State (1867) Processed State (1867) Processed State (1867) Control (1867)	42.213.14100.3117	8/19/2016	James M. Peters	Riverwood Exploration, LLC	2016-00015857 Texas	Henderson
25.15.16.00.00.000 IRS.000	42.213.14100.3118	8/29/2016	George Randal Pitts	Riverwood Exploration, LLC	2016-00015869 Texas	Henderson
4.2.3.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1.1	42.213.14100.3119	8/17/2016	The John Kennedy II Family Trust	Riverwood Exploration, LLC	2016-00015871 Texas	Henderson
42.33 W. 19.00000000000000000000000000000000000	42.213.14100.3120	8/19/2016	Paula McPherson	Riverwood Exploration, LLC	2016-00015856 Texas	Henderson
\$12,000,000 \$12,000	42.213.14100.3121	8/29/2016	Rick D. Pitts	Riverwood Exploration, LLC	2016-00015870 Texas	Henderson
23.13.14.09.23.03 S00.000 for the flame fund Revoked Exposed, LD 20.00 600 817 Feed Revoked Exposed, LD Revoked Exposed, LD Revoked Exposed, LD Revoked	42.213.14100.3122	10/5/2016	The Josie Evans Penny Trust	Riverwood Exploration, LLC	2016-00018112 Texas	Henderson
23.13.14.00.00.00.00.00.00.00.00.00.00.00.00.00	42.213.14100.3123	10/1/2016	Capitol Street, LLC	Riverwood Exploration, LLC	2016-00017049 Texas	Henderson
23.23 May 20.230 May 20.200 May Control May 20.200 May	42.213.14100.3124	9/26/2016	Ewell Eugene Buford	Riverwood Exploration, LLC	2016-00018111 Texas	Henderson
42213140033207 7350000 Constitution Constroll, 10 2014-00000000 mm 100-00000000 mm 100-000000000 mm 100-0000000000000000000000000000000000	42.213.14100.3125			Riverwood Exploration, LLC	2016-00019378 Texas	Henderson
1972 1972	42.213.14100.3126	9/27/2016	Andrew Seale Cutbirth	Riverwood Exploration, LLC	2016-00019384 Texas	Henderson
SEZIONASSIONASIONASIONASIONASIONASIONASIONA	42.213.14100.3127	7/13/2016	Cecil Milton Crawford, Jr.	Riverwood Exploration, LLC	2016-00019390 Texas	Henderson
4221141003131 502018 Nome of Epithody Author Control (1978) Control	42.213.14100.3128	10/12/2016	George Henry Neill	Riverwood Exploration, LLC	2016-00019383 Texas	Henderson
4221341003132 10000000 Moyeau Lial Moyeau Ceptamore (Linguistic) 10 Hobitisty Trace Henderore 4221341003133 10 1010000 Patrole American Ceptamore (Linguistic) 20 Hobitistic Henderore 4221341003133 10 1010000 Patrole American Ceptamore (Linguistic) 20 Hobitistic Henderore 421341003133 10 1010000 Patrole American Ceptamore (Linguistic) 20 Hobitistic Hobitistic 421341003133 10 1010000 Patrole Market Market American Ceptamore (Linguistic) 20 Hobitistic Hobitistic 421341003133 10 1010000 Patrole Market Market Market Market Hobitistic Hobitistic <t< td=""><td>42.213.14100.3129</td><td>9/30/2016</td><td>The J.F. Sharp Revocable Trust</td><td>Riverwood Exploration, LLC</td><td>2016-00019392 Texas</td><td>Henderson</td></t<>	42.213.14100.3129	9/30/2016	The J.F. Sharp Revocable Trust	Riverwood Exploration, LLC	2016-00019392 Texas	Henderson
42211410031M 10112016 Predact Fact Cerebona Cigaristin, LD 201000087 Tame Inchestore 422114100331M 10112016 Predact Fact All State March 1 Cerebona Cigaristin, LD 20160001807 Tame Inchestore 422114100331M 10200018 Fracture Lee Fact JL All State March 1 Cerebona Cigaristin, LD 20160001807 Tame Inchestore 422134100331M 10200018 Fracture Lee Fact JL All State March 1 Cerebona Cigaristin, LD 20160001807 Tame Inchestore 422134100331M 10120018 Berkels Fact Galler Berkels Galler Cerebona Cigaristin, LD 20160001807 Tame Inchestore 422134100331M 10120018 Berkels Galler Berkels Galler Cerebona Cigaristin, LD 20160001807 Tame Inchestore 42213410031M 10120018 Berkels Galler All State Galler Cerebona Cigaristin, LD 20160001807 Tame Inchestore 42213410031M 10120018 Berkels Galler All State Galler All Stat	42.213.14100.3131	9/26/2016	Kevin Upchurch, et ux	Riverwood Exploration, LLC	2016-00018110 Texas	Henderson
42213 41003305 10102010 Amount Septiment, LIC 2015-00000000000 Herberter 42213 41003305 1020201 Facus Lee Ford, J. Herberter 42213 41003305 1020201 Facus Lee Ford, J. Herberter 42213 41003305 10210010 Merber Miller Amount of Particular, LIC 2016-00000000 Herberter 42213 41003305 10110010 Merber Miller Merber Miller Merber Miller Herberter 42213 41003305 10120010 Herber Miller Merber Miller Herberter Herberter 42213 41003301 10220010 Herber Miller Merber Miller Herber Miller 42213 41003101 10220010 Herber Miller Herberter Herberter 42213 41003101 10220010 Lee Miller Herberter Herberter 42213 41003101 1022001 Lee Miller Herberter Herberter 42213 4100310 1012001 Mark Conde Herberter 42213 4100310 1012001 Mark Conde Herberter Herberter 42213 4100310 10	42.213.14100.3132	10/3/2016	Mayna J. Hall	Riverwood Exploration, LLC	2016-00018114 Texas	Henderson
422131410033187 69201018 Tract Lacia Pent. Za. Processor Designation 415-belleting 415-belletin	42.213.14100.3134	10/11/2016	Freddie Fladd	Riverwood Exploration, LLC	2016-00019388 Texas	Henderson
4221341003337 1002000 The Lusin Peter Trad Celebrate Ce	42.213.14100.3135	10/11/2016	Nancy W. Eddy	Riverwood Exploration, LLC	2016-00019387 Texas	Henderson
42213 14003308 10110200 Reese Nay Oldes Halbard Revenued Explanation. 2010 0010380 Team 10-decident 42213 141003109 1014/2018 Reese May Oldes Halbard Revenued Explanation. 2010 0010380 Team 10-decident 42213 141003101 1014/2018 Researched State State 10-decident 10-decident 10-decident 42213 141003104 1012/2018 Researched State 10-decident 10-decident 20-decident 20-decident </td <td>42.213.14100.3136</td> <td>9/22/2016</td> <td>Francis Lee Ford, Jr.</td> <td>Riverwood Exploration, LLC</td> <td>2016-00019352 Texas</td> <td>Henderson</td>	42.213.14100.3136	9/22/2016	Francis Lee Ford, Jr.	Riverwood Exploration, LLC	2016-00019352 Texas	Henderson
4223.141003139 1042018 Beech Kay Gibb Holand Riversood Epistanion, L.C. 2016-0000812 Team Herderson 4223.141003140 10220191 Isra'f ard Smith Riversood Epistanion, L.C. 2017-0000081 Team Herderson 4223.141003140 10220181 Isray M. Cooker, et us Riversood Epistanion, L.C. 2016-0001873 Team Herderson 4223.141003140 10122018 Isray M. Cooker, et us Riversood Epistanion, L.C. 2016-0001873 Team Herderson 4223.141003140 10122018 Isray M. Cooker, et us Riversood Epistanion, L.C. 2016-0001814 Team Herderson 4223.141003140 10122018 Isray M. Cooker, et us Riversood Epistanion, L.C. 2016-0001814 Team Herderson 4223.141003140 10172018 Mark Keller Fask, et us Riversood Epistanion, L.C. 2016-0001814 Team Herderson 4223.141003140 1012018 Mark Keller Fask, et us Riversood Epistanion, L.C. 2016-0001818 Team Herderson 4223.141003150 1022019 Mark Keller Fask, et us Riversood Epistanion, L.C. 2016-0001818 Team Herderson 4223.141003151 1022019 Mark Keller Fask, et us Riversood Epistanion, L.C. 2016-0001818 Team Herderson	42.213.14100.3137	10/26/2016	The Lucia Parker Trust	O'Brien Resources, LLC	2016-00018113 Texas	Henderson
42213 14100.3144 1012/1016 Hard Smith Application (Lot 5010000887 Tools 14 Inchesion 42213 14100.3144 1012/2018 Rose Mark Smith Reviewed Epitoration, Lot 2011-00000887 Tools 14 Inchesion 42213 14100.3144 1015/0018 Ediples F, Rosethoro Reviewed Epitoration, Lot 2016-0001879 Tools 14 Inchesion 42213 14100.3144 1015/0018 Ediples F, Rosethoro Reviewed Epitoration, Lot 2016-0001879 Tools 14 Inchesion 42213 14100.3146 1017/2018 Mark C, Owen Reviewed Epitoration, Lot 2016-0001881 Tools 14 Inchesion 42213 14100.3146 1017/2018 Mark C, Owen Reviewed Epitoration, Lot 2016-0001881 Tools 14 Inchesion 42213 14100.3146 1017/2018 Mark C, Owen Reviewed Epitoration, Lot 2016-0001881 Tools 14 Inchesion 42213 14100.3146 1017/2018 Mark C, Owen Reviewed Epitoration, Lot 2016-0001881 Tools 14 Inchesion 42213 14100.3158 1017/2018 Mark C, Owen Reviewed Epitoration, Lot 2016-0001881 Tools 14 Inchesion 42213 14100.3159 1017/2018 Mark S, Alexander Lane Reviewed Epitoration, Lot 2016-0001881 Tools 14 Inchesion <tr< td=""><td>42.213.14100.3138</td><td>10/11/2016</td><td>Martha Williams</td><td>Riverwood Exploration, LLC</td><td>2016-00019389 Texas</td><td>Henderson</td></tr<>	42.213.14100.3138	10/11/2016	Martha Williams	Riverwood Exploration, LLC	2016-00019389 Texas	Henderson
42.213.1410.0314 100.21016 Read Mark Smith 201.15000000 Lany W. Cocker, et us. Anteniors April 10.0000000 Lany W. Cocker, et us. Anteniors April 10.000000 Lany W. Cocker, et us. Anteniors April 10.000000 Lany W. Cocker, et us. Anteniors April 10.000000000 Lany W. Cocker, et us. Anteniors	42.213.14100.3139	10/14/2016	Brenda Kay Gibbs Holland	Riverwood Exploration, LLC	2016-00019408 Texas	Henderson
42.213.14(0.03.14z) 10.95201b Eurly W. Cooler, et us 2.016.0010373 From 1.016.0010373 From 2.016.0010370 Trous 1.016.0010370	42.213.14100.3140	10/21/2016	Hal Ford Smith	Riverwood Exploration, LLC	2017-00000982 Texas	Henderson
42.213.1410.03344 0.915.2216 February Fe Robertson Robertson Cold-coloration Coloration Hondreson 42.213.1410.03344 0.912.0216 Allen B. Bowell, et al. 2016.0019141 Towas Hondreson 42.213.1410.0346 0.117.0216 Ministry Coloration Reviewood Epichanish LLC 2016.0019141 Towas Hondreson 42.213.1410.0346 0.117.0216 Ministry Reviewood Epichanish LLC 2016.0019181 Towas Hondreson 42.213.1410.0347 0.117.0216 Value Keller Flauk, et us Professor Hondreson Hondreson Hondreson Hondreson 42.213.1410.0340 0.002018 Publication Publication Reversond Epichanish LLC 2016.0019387 Towa Hondreson 42.213.1410.0350 0.002018 Value Turner Publication Reversond Epichanish LLC 2016.0019387 Towa Hondreson 42.213.1410.0351 1.002018 Jona Dial Turner Publication Reversond Epichanish LLC 2016.0019387 Towa Hondreson 42.213.1410.0352 1.017.0252 Terms Language Publication Reviewood Epichanish LLC	42.213.14100.3141	10/21/2016	Ross Mark Smith	Riverwood Exploration, LLC	2017-00000981 Texas	Henderson
42.213.1410.3144 91.22016 Mem B. Bossek et ux Polemonal Springston, LLC 2016.001941 Tools Memorand Exploration, LLC 2016.001941 Tools	42.213.14100.3142	10/26/2016	Larry W. Cooker, et ux	Riverwood Exploration, LLC	2016-00019379 Texas	Henderson
4213.1410.3145 11/12018 Mark C. Owen Reversood Exploration, LLC 2016.0019417 Tools Herinden 4213.1410.3146 11/12018 Samuel Loveless, Jr. Olifo.0019381 Tools Herinden 4213.1410.3140 11/12018 Barrier Rolls, et al. Herinden Reversood Exploration, LLC 2016.0019387 Tools Herinden 4213.1410.3140 11/12018 Barly Well Feller Falls, et al. Colorosiste Falls Herinden 42213.1410.3140 10/20218 Palls Tation Meteral Reversood Exploration, LLC 2016.0019387 Tools Herinden 42213.1410.3151 10/20218 Barly Tation Reversood Exploration, LLC 2016.0019387 Tools Herinden 42213.1410.3152 10/20218 Barly Savenil Herinden Reversood Exploration, LLC 2016.0019387 Tools Herinden 42213.1410.3152 10/20218 Barly Savenil Herinden Reversood Exploration, LLC 2016.0019387 Tools Herinden 42213.1410.3152 11/152016 Borby N. Harris, Jr., et u. Herinden Reversood Exploration, LLC 2016.0019387 Tools Herinden 42213.1410.3153	42.213.14100.3143	10/15/2016	Earlyne F. Robertson	Riverwood Exploration, LLC	2016-00017050 Texas	Henderson
42213.4100.3146 1017/2016 Semout Levelets, Jr. Class Annual Control Reversood Exploration, LLC 2016.001881 Tous Hendreson 42213.4100.3147 11/12016 Water Kellet Faulk, et ux 1016.001941 Tous Hendreson 42213.4100.3149 6 1002016 Beylew, Godorf Reversood Exploration, LLC 2016.0019389 Tous Hendreson 42213.4100.3150 10202016 Vely Turne Reversood Exploration, LLC 2016.0019389 Tous Hendreson 42213.4100.3150 1052016 Author Abexander Lane Reversood Exploration, LLC 2016.0019389 Tous Hendreson 42213.4100.3152 1052016 Author Abexander Lane Reversood Exploration, LLC 2016.0019387 Tous Hendreson 42213.4100.3152 10152016 Author Abexander Lane Reversood Exploration, LLC 2016.0019387 Tous Hendreson 42213.4100.3152 11/152016 Bellevi Maria, J., et ux Author Reversood Exploration, LLC 2016.0019387 Tous Hendreson 42213.4100.3152 11/152016 Alexis Maria Author Reversood Exploration, LLC 2016.0019387 Tous Hendreson	42.213.14100.3144	9/12/2016	Allen B. Boswell, et ux	Riverwood Exploration, LLC	2016-00019414 Texas	Henderson
4.2.13.4100.3147 11/12/016 Water Kellet Fault, et uX 61/00.0148 Texas Henderson 4.2.13.14100.3148 51/02/016 Bettyse W. Gorfort Riverwood Epiperison, LLC 2016.0001937 Texas Henderson 4.2.13.14100.3150 10/02/02/01 Publia Texon Mentalf Riverwood Epiperison, LLC 2016.0001937 Texas Henderson 4.2.21.3.14100.3150 10/02/02/01 Selty Boswell Riverwood Epiperison, LLC 2016.0001938 Texas Henderson 4.2.21.3.14100.3150 10/02/01 Audrey Alexander Line Henderson Henderson 4.2.21.3.14100.3150 10/02/01 Audrey Alexander Line Henderson Henderson 4.2.21.3.14100.3150 10/02/01 Audrey Alexander Line Henderson Henderson 4.2.21.3.14100.3150 10/12/016 Belby N. Harris, J., et ux Henderson Henderson 4.2.21.3.14100.3150 10/12/016 Belby N. Harris, J., et ux Henderson Henderson 4.2.21.3.14100.3150 10/12/016 Bobby N. Harris, J., et ux Henderson Henderson 4.2.21.3.14100.3150 10/12/016 David Mellanderson <td< td=""><td>42.213.14100.3145</td><td>11/1/2016</td><td>Mark C. Owen</td><td>Riverwood Exploration, LLC</td><td>2016-00019417 Texas</td><td>Henderson</td></td<>	42.213.14100.3145	11/1/2016	Mark C. Owen	Riverwood Exploration, LLC	2016-00019417 Texas	Henderson
4.213.4100.3148 510'00 Betty W. Geforth Cold-coll place faller Cold-coll pl	42.213.14100.3146	10/17/2016	Samuel Loveless, Jr.	Riverwood Exploration, LLC	2016-00019381 Texas	Henderson
42213.4100.3160 10/20/2016 Paula Tatum Meteral Riverwood Exploration, LLC 2016-0001988 Texas Henderson 42213.4100.3151 8/19/2018 8/19/2018 10/202016 Victy Turre 2016-0001988 Texas Henderson 42213.4100.3152 10/52/2018 40/52/2018 Auther Meterson Riverwood Exploration, LLC 2016-0001988 Texas Henderson 42213.4100.3152 10/52/2018 Auther Meterson Riverwood Exploration, LLC 2016-0001988 Texas Henderson 42213.4100.3153 10/32/2018 Bebby N. Harris, Jr., et ux 2016-0001987 Texas Henderson 42213.4100.3156 11/10/2018 Bobby N. Harris, Jr., et ux Henderson 42213.4100.3157 11/10/2018 Bobby N. Harris, Jr., et ux Henderson 42213.4100.3158 10/11/2019 David Westherford Riverwood Exploration, LLC 2016-0001947 Texas Henderson 42213.4100.3159 11/15/2018 Don'll, Coker Riverwood Exploration, LLC 2016-0001947 Texas Henderson 42213.4100.3159 11/15/2018 Don'll, Coker Don'll, Coker Henderson 42213	42.213.14100.3147	11/1/2016	Walter Kellett Faulk, et ux	Riverwood Exploration, LLC	2016-00019418 Texas	Henderson
4.2.13.1410.3150 10/20/21 Vicky Turner Riverwood Exploration, LLC 2016-0019891 Texas Henderson 4.2.21.3.14100.3152 10/32016 Belty Bossell 2016-0019887 Texas Henderson 4.2.21.3.14100.3152 10/32016 Texas Legyant 2016-0019887 Texas Henderson 4.2.21.3.14100.3152 10/32016 Texas Legyant Riverwood Exploration, LLC 2016-0019887 Texas Henderson 4.2.21.3.14100.3152 11/15/2016 Bobby N. Harris, Jr., et ux Henderson Henderson 4.2.21.3.14100.3152 11/15/2016 Bobby N. Harris, Jr., et ux 10/16/2019 Henderson 4.2.21.3.14100.3152 11/11/2019 John School Harris, Jr., et ux 10/16/2019 Henderson 4.2.21.3.14100.3152 11/11/2019 John School Harris, Jr., et ux 10/16/2019 Henderson 4.2.21.3.14100.3152 11/11/2019 Henderson Riverwood Exploration, LLC 2016-00019317 Texas Henderson 4.2.21.3.14100.3152 11/15/2019 Donk Coker Donk Coker Periodical Coker 2016-00019317 Texas Henderson 4.2.21.3.14100.3162 11/15/2019	42.213.14100.3148	5/10/2016	Bettye W. Goforth	Riverwood Exploration, LLC	2016-00019375 Texas	Henderson
42.213.4100.3150 819 2016 Betty Boswell Riverwood Exploration, LLC 2016-00018947 Texas Henderson 42.213.4100.3152 105/2016 Audrey Alexander Lane Riverwood Exploration, LLC 2016-00019836 Texas Henderson 42.213.4100.3152 105/2016 Audrey Alexander Lane Riverwood Exploration, LLC 2016-00019837 Texas Henderson 42.213.4100.3152 11/10/2016 Bobby N. Harris, J., et ux OB file no. Resources, LLC 2016-00019415 Texas Henderson 42.213.4100.3152 11/10/2016 Bobby N. Harris, J., et ux Henderson Riverwood Exploration, LLC 2016-00019415 Texas Henderson 42.213.4100.3152 11/10/2016 Intil 12/2016 Den M. Coker Cate Cate Company, LLC 2016-00019407 Texas Henderson 42.213.4100.3152 11/15/2016 Den M. Coker Waters Cate Cate Company, LLC 2016-0001947 Texas Henderson 42.213.4100.3162 11/15/2016 Den M. Coker Waters Riverwood Exploration, LLC 2016-0001947 Texas Henderson 42.213.4100.3162 11/15/2016 Riverwood Exploration, LLC 2016-0001947 Texas Henderson	42.213.14100.3149	10/20/2016	Paula Tatom Metcalf	Riverwood Exploration, LLC	2016-00019386 Texas	Henderson
42.213.14100.3152 10/5/2016 Audrey Abzander Lane Riverwood Exploration, LLC 2016-00019835 Texas Henderson 42.213.14100.3153 10/3/2016 Texas, Le Bryant 2016-00019837 Texas Henderson 42.213.14100.3154 11/5/2016 Bobby N. Harris, Jr., et ux O'Brien Resources, LLC 2016-00019876 Texas Henderson 42.213.14100.3156 11/1/2016 Bobby N. Harris, Jr., et ux C'Brien Resources, LLC 2016-00019410 Texas Henderson 42.213.14100.3156 10/1/2016 Bord W. Washerford Riverwood Exploration, LLC 2016-0001940 Texas Henderson 42.213.14100.3156 10/1/2016 No. Oker O'Brien Resources, LLC 2016-0001940 Texas Henderson 42.213.14100.3169 11/1/2016 Cond. Oker O'Brien Resources, LLC 2016-0001940 Texas Henderson 42.213.14100.3161 11/1/2016 Cond. Oker Waters O'Brien Resources, LLC 2016-0001947 Texas Henderson 42.213.14100.3162 11/1/2016 Koverwood Exploration, LLC 2016-0001947 Texas Henderson 42.213.14100.3162 11/1/2016 Koverwood Exploration, LLC <t< td=""><td>42.213.14100.3150</td><td>10/20/2016</td><td>Vicky Turner</td><td>Riverwood Exploration, LLC</td><td>2016-00019391 Texas</td><td>Henderson</td></t<>	42.213.14100.3150	10/20/2016	Vicky Turner	Riverwood Exploration, LLC	2016-00019391 Texas	Henderson
42.213.14100.3150 10/32016 Teena L Byynn1 10/32016 Teena L Byynn1 2016-00019387 Texas Henderson 42.213.14100.3150 11/10/2016 Bobby N. Harris, Jr., et ux 2016-00019378 Texas Henderson 42.213.14100.3150 11/10/2016 Alien Boswell, et ux 2016-00019410 Texas Henderson 42.213.14100.3150 10/11/2016 David Weatherford Rivervood Exploration, LLC 2016-00019410 Texas Henderson 42.213.14100.3150 10/11/2016 David Weatherford Rivervood Exploration, LLC 2016-00019410 Texas Henderson 42.213.14100.3150 11/15/2016 Don M. Coker Cole Company, LLC O'Brien Resources, LLC 2016-0001937 Texas Henderson 42.213.14100.3160 11/15/2016 Don M. Coker Cole Cole Cole Cole Cole Cole Cole Cole	42.213.14100.3151	8/19/2016	Betty Boswell	Riverwood Exploration, LLC	2016-00015847 Texas	Henderson
42:13:14100.3154 11/15/2016 Bobby N. Harris, Jr., et ux Clife Good 1937 6 Texas Henderson 42:213:14100.3155 11/10/2016 Bobby N. Harris, Jr., et ux 2016-00019376 Texas Henderson 42:213:14100.3156 11/10/2016 Double Welsterford Riverwood Exploration, LLC 2016-00019415 Texas Henderson 42:213:14100.3156 10/19/2016 Tree E. Cattle Company, LLC 2016-0001947 Texas Henderson 42:213:14100.3159 11/15/2016 Don M. Coker OBrien Resources, LLC 2016-0001947 Texas Henderson 42:213:14100.3159 11/15/2016 Double S. Coker OBrien Resources, LLC 2016-00019347 Texas Henderson 42:213:14100.3150 11/15/2016 Coker Waters All Fill Fill Fill Fill Fill Fill Fill F	42.213.14100.3152	10/5/2016	Audrey Alexander Lane	Riverwood Exploration, LLC	2016-00019385 Texas	Henderson
42213.14100.3155 11/10/2016 Allen B. Boswell, et tx Henderson 42213.14100.3156 10/11/2016 David Weatherford Riverwood Exploration, LLC 2016-00019409 Texas Henderson 42213.14100.3157 10/19/2016 The E. Cattle Company, LLC 2016-0001940 Texas Henderson 42213.14100.3158 11/15/2016 Down M. Ocker O'Brien Resources, LLC 2016-00019347 Texas Henderson 42213.14100.3159 11/15/2016 Douglas B. Coker O'Brien Resources, LLC 2016-00019347 Texas Henderson 42213.14100.3160 11/15/2016 Linda Coker Waters D'Brien Resources, LLC 2016-00019347 Texas Henderson 42213.14100.3161 11/17/2016 Roya Albright Fitzgerald Riverwood Exploration, LLC 2016-0001942 Texas Henderson 42213.14100.3162 11/17/2016 Raymond H. Stanford, II, et ux 2016-0001942 Texas Henderson 42213.14100.3164 11/15/2016 Raymond H. Stanford, II, et ux 2016-0001947 Texas Henderson 42213.14100.3169 11/15/2016 Adam Bryant Adam Syant All enderson Pill enderson	42.213.14100.3153	10/3/2016	Teena L. Bryant	Riverwood Exploration, LLC	2016-00019382 Texas	Henderson
42.21.3.14100.3156 10/11/2016 David Weatherford Riverwood Exploration, LLC 2016-00019409 Texas Henderson 42.21.3.14100.3157 10/19/2016 True E. Cattle Company, LLC 2016-00019410 Texas Henderson 42.213.14100.3158 11/15/2016 Douglas B. Coker 2016-00019377 Texas Henderson 42.213.14100.3159 11/15/2016 Douglas B. Coker 2016-00019347 Texas Henderson 42.213.14100.3161 11/15/2016 Linda Coker Waters 2016-00019347 Texas Henderson 42.213.14100.3162 11/15/2016 Kupe Albright Etzgerald 2016-00019347 Texas Henderson 42.213.14100.3162 11/15/2016 Rose Albright Barrows Riverwood Exploration, LLC 2016-00019347 Texas Henderson 42.213.14100.3162 11/15/2016 Rose Albright Barrows Riverwood Exploration, LLC 2016-00019347 Texas Henderson 42.213.14100.3163 11/25/2016 Adam Bryant Albright Barrows Riverwood Exploration, LLC 2016-0001937 Texas Henderson 42.213.14100.3163 11/15/2016 July Barrows Albright Barrows Albright Barrows 2016-	42.213.14100.3154	11/15/2016	Bobby N. Harris, Jr., et ux	O'Brien Resources, LLC	2016-00019376 Texas	Henderson
42.21.3.14.00.3157 10/19/2016 Three E. Cattle Company, LLC 2016-00019410 Texas Henderson 42.21.3.14.100.3158 11/15/2016 Don M. Coker 2016-00019377 Texas Henderson 42.21.3.14.100.3159 11/15/2016 Douglas B. Coker 2016-00019347 Texas Henderson 42.21.3.14.100.3160 11/15/2016 Linda Coker Waters 2016-00019347 Texas Henderson 42.21.3.14.100.3161 11/15/2016 Kaye Albright Fitzgerald Riverwood Exploration, LLC 2016-00019424 Texas Henderson 42.21.3.14.100.3162 11/15/2016 Rose Albright Barrows Riverwood Exploration, LLC 2016-00019421 Texas Henderson 42.21.3.14.100.3162 11/16/2016 Raymond H. Stanford, II, et ux Riverwood Exploration, LLC 2016-00019421 Texas Henderson 42.21.3.14.100.3163 11/15/2016 Algorith G. Luster O'Brien Resources, LLC 2016-00019373 Texas Henderson 42.21.3.14.100.3167 11/15/2016 Lengthing G. Luster Pill-00019373 Texas Henderson 42.21.3.14.100.3167 10/26/2016 Lengthing G. Luster 2016-00019373 Texas Henderson	42.213.14100.3155	11/10/2016	Allen B. Boswell, et ux	Riverwood Exploration, LLC	2016-00019415 Texas	Henderson
42.213.14100.3157 10/19/2016 Three E. Cattle Company, LLC 2016-00019410 Texas Henderson 42.213.14100.3158 11/15/2016 Do M. Coker 2016-00019377 Texas Henderson 42.213.14100.3159 11/15/2016 Douglas B. Coker 2016-00019347 Texas Henderson 42.213.14100.3160 11/15/2016 Linda Coker Waters 2016-00019347 Texas Henderson 42.213.14100.3161 11/15/2016 Kaye Albright Etizgerald Riverwood Exploration, LLC 2016-00019424 Texas Henderson 42.213.14100.3162 11/15/2016 Raymond H. Stanford, II, et ux Riverwood Exploration, LLC 2016-00019421 Texas Henderson 42.213.14100.3163 11/25/2016 Raymond H. Stanford, II, et ux Riverwood Exploration, LLC 2016-00019421 Texas Henderson 42.213.14100.3165 11/15/2016 Larger G. Luster 2016-00019373 Texas Henderson 42.213.14100.3165 11/15/2016 Larger G. Luster 2016-00019373 Texas Henderson 42.213.14100.3167 10/26/2016 Larger G. Luster 2016-00019373 Texas Henderson 42.213.14100.3167 10/26/2016<	42.213.14100.3156	10/11/2016	David Weatherford	Riverwood Exploration, LLC	2016-00019409 Texas	Henderson
42.21.3.14100.3158 11/15/2016 Dom M. Coker 2016-00019377 Texas Henderson 42.21.3.14100.3159 11/15/2016 Douglas B. Coker 2016-00019346 Texas Henderson 42.21.3.14100.3160 11/15/2016 Linda Coker Waters 2016-00019347 Texas Henderson 42.213.14100.3161 11/17/2016 Skye Abright Fitzgerald Riverwood Exploration, LLC 2016-00019424 Texas Henderson 42.213.14100.3162 11/17/2016 Raymond H. Starford, II, et ux 2016-00019424 Texas Henderson 42.213.14100.3163 11/25/2016 Raymond H. Starford, II, et ux 2016-00019411 Texas Henderson 42.213.14100.3164 11/15/2016 Adam Bryant 2016-00019411 Texas Henderson 42.213.14100.3165 11/15/2016 Luster 2016-0001937 Texas Henderson 42.213.14100.3165 11/15/2016 Luster 2016-0001937 Texas Henderson 42.213.14100.3167 11/15/2016 Luster 2016-0001937 Texas Henderson 42.213.14100.3168 11/15/2016 Luster 2016-0001937 Texas Henderson 42.213.14100	42.213.14100.3157			Riverwood Exploration, LLC	2016-00019410 Texas	Henderson
42 21 3.14 100.3160 11/15/2016 Linda Coker Waters 126 1-00019347 Texas Henderson 42 21 3.14 100.3161 11/17/2016 Kaye Albright Fitzgerald Rivenwood Exploration, LLC 2016-00019424 Texas Henderson 42 21 3.14 100.3162 11/17/2016 Rose Albright Barrows Rivenwood Exploration, LLC 2016-00019423 Texas Henderson 42 21 3.14 100.3163 11/25/2016 Raymond H. Stanford, II, et ux Rivenwood Exploration, LLC 2016-00019412 Texas Henderson 42 21 3.14 100.3163 11/25/2016 Julia Stanford, II, et ux 2016-00019412 Texas Henderson 42 21 3.14 100.3165 11/15/2016 Julia Stanford, III, et ux 2016-00019412 Texas Henderson 42 21 3.14 100.3166 11/15/2016 Julia Stanford, III, et ux 2016-00019437 Texas Henderson 42 21 3.14 100.3167 11/15/2016 Laury W. Cooker, et ux 2016-00019437 Texas Henderson 42 21 3.14 100.3168 11/15/2016 Dan M. Rasco O'Brien Resources, LLC 2016-00019432 Texas Henderson 42 21 3.14 100.3176 11/15/2016 Betty G. Fowler O'Brien Resources, LLC 2016-0001942 T	42.213.14100.3158	11/15/2016	Don M. Coker	O'Brien Resources, LLC	2016-00019377 Texas	Henderson
42.21.3.14100.3160 11/15/2016 Linda Coker Waters 2016-00019347 Texas Henderson 42.21.3.14100.3161 11/17/2016 Kaye Albright Fitzgerald Rivenwood Exploration, LLC 2016-00019424 Texas Henderson 42.21.3.14100.3162 11/17/2016 Rose Albright Barrows Rivenwood Exploration, LLC 2016-00019423 Texas Henderson 42.21.3.14100.3163 11/25/2016 Raymond H. Stanford, II, et ux Rivenwood Exploration, LLC 2016-00019412 Texas Henderson 42.21.3.14100.3164 11/15/2016 Jerri G. Luster O'Brien Resources, LLC 2016-00019374 Texas Henderson 42.21.3.14100.3165 11/15/2016 Jerri G. Luster O'Brien Resources, LLC 2016-00019374 Texas Henderson 42.21.3.14100.3166 11/15/2016 Laquinta G. Palmer O'Brien Resources, LLC 2016-00019373 Texas Henderson 42.21.3.14100.3167 11/15/2016 Laquinta G. Palmer D'Brien Resources, LLC 2016-00019380 Texas Henderson 42.21.3.14100.3168 11/15/2016 Dan M. Rasco O'Brien Resources, LLC 2016-00019342 Texas Henderson 42.21.3.14100.3170 11/15/2016 Betty G. Fowler O'Brien Resources, LLC 2016-00019342 Texas Henderson <t< td=""><td></td><td></td><td></td><td></td><td></td><td></td></t<>						
42.213.14100.3161 11/17/2016 Kaye Albright Fitzgerald Rivenwood Exploration, LLC 2016-00019424 Texas Henderson 42.213.14100.3162 11/17/2016 Rose Albright Barrows Rivenwood Exploration, LLC 2016-00019413 Texas Henderson 42.213.14100.3163 11/25/2016 Raymond H. Stanford, II, et ux Rivenwood Exploration, LLC 2016-00019411 Texas Henderson 42.213.14100.3164 11/15/2016 Adam Bryant 2016-00019412 Texas Henderson 42.213.14100.3165 11/15/2016 Jerri G. Luster O'Brien Resources, LLC 2016-0001937 Texas Henderson 42.213.14100.3166 11/15/2016 Larry W. Cooker, et ux Riverwood Exploration, LLC 2016-0001937 Texas Henderson 42.213.14100.3168 11/15/2016 Larry W. Cooker, et ux 2016-0001937 Texas Henderson 42.213.14100.3169 11/15/2016 Dan M. Rasco O'Brien Resources, LLC 2016-0001938 Texas Henderson 42.213.14100.3169 11/15/2016 Betty G. Fowler C'Brien Resources, LLC 2016-00019342 Texas Henderson 42.213.14100.3170 11/17/2016 Gil Cook Morgan						
42.21.3.14100.3162 11/17/2016 Rose Albright Barrows Riverwood Exploration, LLC 2016-00019423 Texas Henderson 42.21.3.14100.3163 11/25/2016 Raymond H. Stanford, II, et ux Riverwood Exploration, LLC 2016-00019412 Texas Henderson 42.21.3.14100.3164 11/28/2016 Adam Bryant Riverwood Exploration, LLC 2016-00019374 Texas Henderson 42.21.3.14100.3165 11/15/2016 Lequitad G. Palmer O'Brien Resources, LLC 2016-00019373 Texas Henderson 42.21.3.14100.3167 10/26/2016 Larry W. Cooker, et ux Riverwood Exploration, LLC 2016-00019380 Texas Henderson 42.21.3.14100.3168 11/15/2016 Dan M. Rasco O'Brien Resources, LLC 2017-0000552 Texas Henderson 42.21.3.14100.3169 11/15/2016 Betty G. Fowler O'Brien Resources, LLC 2016-00019342 Texas Henderson 42.21.3.14100.3170 11/17/2016 Gail Cook Morgan Riverwood Exploration, LLC 2016-00019421 Texas Henderson 42.21.3.14100.3172 11/15/2016 Cora Annell G. Doris Cora Annell G. Doris Henderson	42.213.14100.3161			Riverwood Exploration, LLC	2016-00019424 Texas	Henderson
42.213.14100.3163 11/25/2016 Raymond H. Stanford, II, et ux Rivenvood Exploration, LLC 2016-00019411 Texas Henderson 42.213.14100.3164 11/28/2016 Adam Bryant Rivervood Exploration, LLC 2016-00019374 Texas Henderson 42.213.14100.3165 11/15/2016 Lequinta G. Palmer O'Brien Resources, LLC 2016-00019374 Texas Henderson 42.213.14100.3166 11/15/2016 Larry W. Cooker, et ux Rivervood Exploration, LLC 2016-00019373 Texas Henderson 42.213.14100.3168 11/15/2016 Larry W. Cooker, et ux 2016-00019372 Texas Henderson 42.213.14100.3168 11/15/2016 Dan M. Rasco O'Brien Resources, LLC 2016-00019382 Texas Henderson 42.213.14100.3169 11/15/2016 Betty G. Fowler 2016-00019342 Texas Henderson 42.213.14100.3170 11/17/201 Gil Cook Morgan Rivervood Exploration, LLC 2016-00019421 Texas Henderson 42.213.14100.3171 11/15/2016 Cora Annell G. Doris Pinderson Pinderson Pinderson Pinderson Pinderson Pinderson Pinderson Pinderson						
42.213.14100.3164 11/28/2016 Adam Bryant Riverwood Exploration, LLC 2016-00019412 Texas Henderson 42.213.14100.3165 11/15/2016 Jerri G. Luster O'Brien Resources, LLC 2016-00019373 Texas Henderson 42.213.14100.3166 11/15/2016 Laquinta G. Palmer O'Brien Resources, LLC 2016-00019373 Texas Henderson 42.213.14100.3167 10/26/2016 Larry W. Cooker, et ux Riverwood Exploration, LLC 2016-00019380 Texas Henderson 42.213.14100.3168 11/15/2016 Dan M. Rasco O'Brien Resources, LLC 2017-00000552 Texas Henderson 42.213.14100.3169 11/15/2016 Betty G. Fowler O'Brien Resources, LLC 2016-00019342 Texas Henderson 42.213.14100.3170 11/17/2016 Gail Cook Morgan Riverwood Exploration, LLC 2016-00019421 Texas Henderson 42.213.14100.3171 11/17/2016 Tony Glenn Cook Riverwood Exploration, LLC 2016-00019421 Texas Henderson 42.213.14100.3172 11/15/2016 Cora Annell G. Doris O'Brien Resources, LLC 2016-00019421 Texas Henderson	42.213.14100.3163					Henderson
42.213.14100.3165 11/15/2016 Jarri G. Luster O'Brien Resources, LLC 2016-00019374 Texas Henderson 42.213.14100.3166 11/15/2016 Laquinta G. Palmer O'Brien Resources, LLC 2016-00019373 Texas Henderson 42.213.14100.3167 10/26/2016 Larry W. Cooker, et ux Riverwood Exploration, LLC 2016-00019380 Texas Henderson 42.213.14100.3169 11/15/2016 Betty G. Fowler O'Brien Resources, LLC 2016-00019342 Texas Henderson 42.213.14100.3170 11/17/2016 Gall Cook Morgan Riverwood Exploration, LLC 2016-00019421 Texas Henderson 42.213.14100.3171 11/17/2016 Tony Glenn Cook Riverwood Exploration, LLC 2016-00019342 Texas Henderson 42.213.14100.3172 11/15/2016 Cora Annell G. Doris Portion Cook Riverwood Exploration, LLC 2016-00019342 Texas Henderson						
42.213.14100.3166 11/15/2016 Laquinta G. Palmer O'Brien Resources, LLC 2016-00019373 Texas Henderson 42.213.14100.3167 10/26/2016 Larry W. Cooker, et ux Riverwood Exploration, LLC 2016-00019380 Texas Henderson 42.213.14100.3168 11/15/2016 Dan M. Rasco O'Brien Resources, LLC 2017-00000552 Texas Henderson 42.213.14100.3169 11/15/2016 Betty G. Fowler O'Brien Resources, LLC 2016-00019342 Texas Henderson 42.213.14100.3170 11/17/201 Gill Cook Morgan Riverwood Exploration, LLC 2016-00019421 Texas Henderson 42.213.14100.3172 11/15/2016 Cora Annell G. Doris O'Brien Resources, LLC 2016-00019345 Texas Henderson						
42.213.14100.3167 10/26/2016 Larry W. Cooker, et ux Riverwood Exploration, LLC 2016-00019380 Texas Henderson 42.213.14100.3168 11/15/2016 Dan M. Rasco O'Brien Resources, LLC 2017-00000552 Texas Henderson 42.213.14100.3169 11/15/2016 Betty G. Fowler O'Brien Resources, LLC 2016-00019342 Texas Henderson 42.213.14100.3170 11/17/2016 Gail Cook Morgan Riverwood Exploration, LLC 2016-00019421 Texas Henderson 42.213.14100.3172 11/15/2016 Tony Glenn Cook Riverwood Exploration, LLC 2016-00019420 Texas Henderson 42.213.14100.3172 11/15/2016 Cora Annell G. Doris O'Brien Resources, LLC 2016-00019345 Texas Henderson						
42.213.14100.3168 11/15/2016 Dan M. Rasco O'Brien Resources, LLC 2017-00000552 Texas Henderson 42.213.14100.3169 11/15/2016 Betty G. Fowler O'Brien Resources, LLC 2016-00019342 Texas Henderson 42.213.14100.3170 11/17/2016 Gail Cook Morgan Riverwood Exploration, LLC 2016-00019421 Texas Henderson 42.213.14100.3171 11/17/2016 Tony Glenn Cook Riverwood Exploration, LLC 2016-00019420 Texas Henderson 42.213.14100.3172 11/15/2016 Cora Annell G. Doris O'Brien Resources, LLC 2016-00019345 Texas Henderson						
42.213.14100.3169 11/15/2016 Betty G. Fowler O'Brien Resources, LLC 2016-00019342 Texas Henderson 42.213.14100.3170 11/17/2016 Gail Cook Morgan Riverwood Exploration, LLC 2016-00019421 Texas Henderson 42.213.14100.3171 11/17/2016 Tony Glenn Cook Riverwood Exploration, LLC 2016-00019420 Texas Henderson 42.213.14100.3172 11/15/2016 Cora Annell G. Doris O'Brien Resources, LLC 2016-00019342 Texas Henderson						
42.213.14100.3170 11/17/2016 Gail Cook Morgan Riverwood Exploration, LLC 2016-00019421 Texas Henderson 42.213.14100.3171 11/17/2016 Tony Glenn Cook Riverwood Exploration, LLC 2016-00019420 Texas Henderson 42.213.14100.3172 11/15/2016 Cora Annell G. Doris O'Brien Resources, LLC 2016-00019345 Texas Henderson						
42.213.14100.3171 11/17/2016 Tony Glenn Cook Riverwood Exploration, LLC 2016-00019420 Texas Henderson 42.213.14100.3172 11/15/2016 Cora Annell G. Doris Cora Annell G. Doris Resources, LLC 2016-00019345 Texas Henderson Henderson 42.213.14100.3172 11/15/2016 Cora Annell G. Doris Resources, LLC 2016-00019345 Texas Henderson 42.213.14100.3172 11/15/2016 Cora Annell G. Doris Resources, LLC 2016-00019345 Texas Henderson 42.213.14100.3172 11/15/2016 Cora Annell G. Doris Resources, LLC 2016-00019345 Texas Henderson 42.213.14100.3172 11/15/2016 Cora Annell G. Doris Resources, LLC 2016-00019345 Texas Henderson 42.213.14100.3172 11/15/2016 Cora Annell G. Doris Resources, LLC 2016-00019345 Texas Henderson 42.213.14100.3172 11/15/2016 Cora Annell G. Doris Resources, LLC 2016-00019345 Texas Henderson 42.213.14100.3172 11/15/2016 Cora Annell G. Doris Resources, LLC 2016-00019345 Texas Henderson 42.213.14100.3172 11/15/2016 Cora Annell G. Doris Resources, LLC 2016-00019345 Texas Henderson 42.213.14100.3172 11/15/2016 Cora Annell G. Doris Resources, LLC 2016-00019345 Texas Henderson 42.213.14100.3172 11/15/2016 Cora Annell G. Doris Resources, LLC 2016-00019345 Texas Henderson 42.213.14100.3172 11/15/2016 Cora Annell G. Doris Resources, LLC 2016-00019345 Texas Henderson 42.213.14100.3172 11/15/2016 Cora Annell G. Doris Resources, LLC 2016-00019345 Texas Henderson 42.213.14100.3172 11/15/2016 Texas Henderson 42.						
42.213.14100.3172 11/15/2016 Cora Annell G. Doris O'Brien Resources, LLC 2016-00019345 Texas Henderson						
	42.213.14100.3173			O'Brien Resources, LLC	2016-00019343 Texas	Henderson

Lease #	Date Lessor	Lessee	Book-Page State	County
42.213.14100.3174	11/15/2016 Tracie L. Hanna	O'Brien Resources, LLC	2016-00019344 Texas	Henderson
42.213.14100.3175	11/15/2016 Dianna R. Lee	O'Brien Resources, LLC	2016-00019398 Texas	Henderson
42.213.14100.3176	11/15/2016 Donna R. Pahl	O'Brien Resources, LLC	2016-00019399 Texas	Henderson
42.213.14100.3177	12/12/2016 Jerry Lee Foster	O'Brien Resources, LLC	2016-00019400 Texas	Henderson
42.213.14100.3178	12/10/2016 Betty Katherine Browning Whitehead	Riverwood Exploration, LLC	2016-00019407 Texas	Henderson
42.213.14100.3179	11/30/2016 Cameron Gee	Riverwood Exploration, LLC	2016-00019403 Texas	Henderson
42.213.14100.3180	12/10/2016 Gia Freeman Newell	Riverwood Exploration, LLC	2016-00019404 Texas	Henderson
42.213.14100.3181	10/10/2016 Marion Cutbirth Mitchell	Riverwood Exploration, LLC	2016-00019402 Texas	Henderson
42.213.14100.3182	11/17/2016 Michael Arnez Allbright	Riverwood Exploration, LLC	2016-00019405 Texas	Henderson
42.213.14100.3183	11/22/2016 The Richard Reimann and Marie Reimann Revocable Living Trust	Riverwood Exploration, LLC	2017-00000133 Texas	Henderson
42.213.14100.3184	12/10/2016 Wayne West Browning	Riverwood Exploration, LLC	2016-00019406 Texas	Henderson
42.213.14100.3185	12/12/2016 Teresa Collier	O'Brien Resources, LLC	2016-00019401 Texas	Henderson
42.213.14100.3186	9/12/2016 Allen B. Boswell	Riverwood Exploration, LLC	2016-00019416 Texas	Henderson
42.213.14100.3187	11/17/2016 Jerry Leon Adams	Riverwood Exploration, LLC	2016-00019350 Texas	Henderson
42.213.14100.3188	11/17/2016 Joe David Adams	Riverwood Exploration, LLC	2016-00019353 Texas	Henderson
42.213.14100.3189	12/15/2016 Dean Loper	Riverwood Exploration, LLC	2016-00019351 Texas	Henderson
42.213.14100.3190	10/20/2016 Jasper Brown Cutbirth, III	Riverwood Exploration, LLC	2016-00019354 Texas	Henderson
42.213.14100.3191	12/29/2016 Jimmy Foster	O'Brien Resources, LLC	2017-00000076 Texas	Henderson
42.213.14100.3192	12/15/2016 Becky Ann Emerson	Riverwood Exploration, LLC	2017-00000551 Texas	Henderson
42.213.14100.3193	12/5/2016 Joe Buck Coker	Riverwood Exploration, LLC	2017-00001908 Texas	Henderson
42.213.14100.3194	12/5/2016 Allan Wayne Coker	Riverwood Exploration, LLC	2017-00001908 Texas	Henderson
42.213.14100.3195	8/5/2016 Mel Jason Pirtle	Riverwood Exploration, LLC	2017-00000984 Texas	Henderson
42.213.14100.3196	11/17/2016 Milton Keith Adams	Riverwood Exploration, LLC	2017-00000983 Texas	Henderson
42.213.14100.3197	1/11/2017 Terry R. Foster	O'Brien Resources, LLC	2017-00000986 Texas	Henderson
42.213.14100.3198	12/5/2016 Gwenda Billups	Riverwood Exploration, LLC	2017-00001317 Texas	Henderson
42.213.14100.3199	12/5/2016 Charlotte Slaton	Riverwood Exploration, LLC	2017-00001316 Texas	Henderson
42.213.14100.3200	1/12/2017 Debbie Lynn Loosier	O'Brien Resources, LLC	2017-00001318 Texas	Henderson
42.213.14100.3201	7/1/2016 Phillip Douglas Hatton	Riverwood Exploration, LLC	2017-00001319 Texas	Henderson
42.213.14100.3202	7/1/2016 Estate of Paul Thomas Pirtle	Riverwood Exploration, LLC	2017-00001320 Texas	Henderson
42.213.14100.3203	12/29/2016 Linda R. Williams	Riverwood Exploration, LLC	2017-00001713 Texas	Henderson
42.213.14100.3204	12/19/2016 Terry L. Cook	Riverwood Exploration, LLC	2017-00001712 Texas	Henderson
42.213.14100.3205	11/30/2016 Edward Leslie Gee	Riverwood Exploration, LLC	2017-00002577 Texas	Henderson
42.213.14100.3206 42.213.14100.3207	11/30/2016 Martin Randolph Gee 10/5/2016 Thomas I. Alexander, IV	Riverwood Exploration, LLC Riverwood Exploration, LLC	2017-00002578 Texas 2017-00003227 Texas	Henderson Henderson
42.213.14100.3207	2/9/2017 Robert E. Luna, Trustee	Riverwood Exploration, LLC	2017-00003227 Texas 2017-00004795 Texas	Henderson
42.213.14100.3209	12/5/2016 Randall Slaton and Hollie Slaton	Riverwood Exploration, LLC	2017-00004795 Texas	Henderson
42.213.14100.3210	8/19/2016 Barbara Bolin	Riverwood Exploration, LLC	2017-00004790 Texas	Henderson
42.213.14100.3210	6/12/2012 Elizabeth McPherson Brown Living Trust, Elizabeth M. Brown	Eiche, Mapes and Company,	2013-00004797 Texas	Henderson
42.213.14000.0009	6/12/2012 Carol Stirman Barton	Eiche, Mapes and Company,	2013-00001493 Texas	Henderson
42.213.14000.0030	8/25/2012 Archie A. Dennis, Jr.	Eiche, Mapes and Company,	2013-00001442 Texas	Henderson
42.213.14000.0031	8/25/2012 Archie A. Dennis, Jr.	Eiche, Mapes and Company,	2013-00001441 Texas	Henderson
42.213.14000.0040	9/1/2012 Carol Bardwell Landers	Eiche, Mapes and Company,	2013-00001474 Texas	Henderson
42.213.14000.0041	9/1/2012 Lori Bardwell Hallbert	Eiche, Mapes and Company,	2013-00001473 Texas	Henderson
42.213.14000.0049	9/1/2012 Robert Carey Bardwell	Eiche, Mapes and Company,	2013-00001480 Texas	Henderson
42.213.14000.0050	9/19/2012 Cheryl Chipley Lambright	Eiche, Mapes and Company,	2013-00001482 Texas	Henderson
42.213.14000.0051	9/19/2012 Larry Glynn Chipley	Eiche, Mapes and Company,	2013-00001483 Texas	Henderson
42.213.14000.0052	9/27/2012 Declaration of Trust, dated 11/19/1985, William J. Beecher & Jean Beecher	Eiche, Mapes and Company,	2013-00001486 Texas	Henderson
42.213.14000.0054				
	9/19/2012 Sandra Chipley Kellogg	Eiche, Mapes and Company,	2013-00001484 Texas	Henderson
42.213.14000.0059	9/19/2012 Sandra Chipley Kellogg 10/17/2012 James E. Mitchell	Eiche, Mapes and Company, Eiche, Mapes and Company,	2013-00001484 Texas 2013-00001501 Texas	Henderson Henderson
42.213.14000.0059 42.213.14000.0068				
	10/17/2012 James E. Mitchell	Eiche, Mapes and Company,	2013-00001501 Texas	Henderson
42.213.14000.0068	10/17/2012 James E. Mitchell 10/6/2012 Barbara Van Horn Cox Revocable Trust, dated 10/26/1999, Barbara Van Horn Cox	Eiche, Mapes and Company,	2013-00001501 Texas 2013-00001524 Texas	Henderson Henderson
42.213.14000.0068 42.213.14000.0077	10/17/2012 James E. Mitchell 10/6/2012 Barbara Van Horn Cox Revocable Trust, dated 10/26/1999, Barbara Van Horn Cox 11/24/2012 Kenneth Trigg Dealey	Eiche, Mapes and Company, Eiche, Mapes and Company, Eiche, Mapes and Company,	2013-00001501 Texas 2013-00001524 Texas 2013-00001517 Texas	Henderson Henderson Henderson
42.213.14000.0068 42.213.14000.0077 42.213.14000.0078	10/17/2012 James E. Mitchell 10/6/2012 Barbara Van Horn Cox Revocable Trust, dated 10/26/1999, Barbara Van Horn Cox 11/24/2012 Kenneth Trigg Dealey 11/30/2012 Corley Family Trust, John C. Corley	Eiche, Mapes and Company, Eiche, Mapes and Company, Eiche, Mapes and Company, Eiche, Mapes and Company,	2013-00001501 Texas 2013-00001524 Texas 2013-00001517 Texas 2013-00001518 Texas	Henderson Henderson Henderson
42.213.14000.0068 42.213.14000.0077 42.213.14000.0078 42.213.14000.0080	10/17/2012 James E. Mitchell 10/6/2012 Barbara Van Horn Cox Revocable Trust, dated 10/26/1999, Barbara Van Horn Cox 11/24/2012 Kenneth Trigg Dealey 11/30/2012 Corley Family Trust, John C. Corley 12/4/2012 Josephine Dealey	Eiche, Mapes and Company, Eiche, Mapes and Company, Eiche, Mapes and Company, Eiche, Mapes and Company, Eiche, Mapes and Company,	2013-00001501 Texas 2013-00001524 Texas 2013-00001517 Texas 2013-00001518 Texas 2013-00001520 Texas	Henderson Henderson Henderson Henderson
42.213.14000.0068 42.213.14000.0077 42.213.14000.0078 42.213.14000.0080 42.213.14000.0137	10/17/2012 James E. Mitchell 10/6/2012 Barbara Van Horn Cox Revocable Trust, dated 10/26/1999, Barbara Van Horn Cox 11/24/2012 Kenneth Trigg Dealey 11/30/2012 Corley Family Trust, John C. Corley 12/4/2012 Josephine Dealey 7/12/2016 MCW Royalties, a Texas General Partnership, Betty W. Kennaugh and Frank S. Wheelock	Eiche, Mapes and Company, BROGO Minerals, LLC	2013-00001501 Texas 2013-00001524 Texas 2013-00001517 Texas 2013-00001518 Texas 2013-00001520 Texas 2016-00011630 Texas	Henderson Henderson Henderson Henderson Henderson
42.213.14000.0068 42.213.14000.0077 42.213.14000.0078 42.213.14000.0080 42.213.14000.0137 42.213.14000.0148	10/17/2012 James E. Mitchell 10/6/2012 Barbara Van Horn Cox Revocable Trust, dated 10/26/1999, Barbara Van Horn Cox 11/24/2012 Kenneth Trigg Dealey 11/30/2012 Corley Family Trust, John C. Corley 12/4/2012 Josephine Dealey 7/12/2016 MCW Royalties, a Texas General Partnership, Betty W. Kennaugh and Frank S. Wheelock 9/27/2014 Martha Ann Holliday	Eiche, Mapes and Company, BROGO Minerals, LLC Eiche, Mapes and Company,	2013-00001501 Texas 2013-00001524 Texas 2013-00001517 Texas 2013-00001518 Texas 2013-00001507 Texas 2016-00011630 Texas 2014-00014346 Texas	Henderson Henderson Henderson Henderson Henderson Henderson
42.213.14000.0068 42.213.14000.0077 42.213.14000.0078 42.213.14000.0080 42.213.14000.0148 42.213.14000.0148 42.213.14000.0149	10/17/2012 James E. Mitchell 10/6/2012 Barbara Van Horn Cox Revocable Trust, dated 10/26/1999, Barbara Van Horn Cox 11/24/2012 Kenneth Trigg Dealey 11/30/2012 Corley Family Trust, John C. Corley 12/4/2012 Josephine Dealey 7/12/2016 MCW Royalties, a Texas General Partnership, Betty W. Kennaugh and Frank S. Wheelock 9/27/2014 Martha Ann Holliday 9/30/2014 Dwain Graham Oliver	Eiche, Mapes and Company, BROGO Minerals, LLC Eiche, Mapes and Company, Eiche, Mapes and Company,	2013-00001501 Texas 2013-00001524 Texas 2013-00001517 Texas 2013-00001518 Texas 2013-00001520 Texas 2016-00011630 Texas 2014-00014346 Texas 2014-00014347 Texas	Henderson Henderson Henderson Henderson Henderson Henderson Henderson
42.213.14000.0068 42.213.14000.0077 42.213.14000.0078 42.213.14000.0080 42.213.14000.0148 42.213.14000.0149 42.213.14000.0150	10/17/2012 James E. Mitchell 10/6/2012 Barbara Van Horn Cox Revocable Trust, dated 10/26/1999, Barbara Van Horn Cox 11/24/2012 Kenneth Trigg Dealey 11/30/2012 Corley Family Trust, John C. Corley 12/4/2012 Josephine Dealey 7/12/2016 MCW Royalties, a Texas General Partnership, Betty W. Kennaugh and Frank S. Wheelock 9/27/2014 Martha Ann Holliday 9/30/2014 Dwain Graham Oliver 10/2/2014 Leah Jo Loper Brown	Eiche, Mapes and Company, BROGO Minerals, LLC Eiche, Mapes and Company, Eiche, Mapes and Company, Eiche, Mapes and Company,	2013-00001501 Texas 2013-00001524 Texas 2013-00001517 Texas 2013-00001518 Texas 2013-00001520 Texas 2016-00011630 Texas 2014-00014346 Texas 2014-00014347 Texas 2014-00014342 Texas	Henderson Henderson Henderson Henderson Henderson Henderson Henderson
42.213.14000.0068 42.213.14000.0077 42.213.14000.0078 42.213.14000.0137 42.213.14000.0148 42.213.14000.0149 42.213.14000.0150 42.213.14000.0151	10/17/2012 James E. Mitchell 10/6/2012 Barbara Van Horn Cox Revocable Trust, dated 10/26/1999, Barbara Van Horn Cox 11/24/2012 Kenneth Trigg Dealey 11/30/2012 Corley Family Trust, John C. Corley 12/4/2012 Josephine Dealey 7/12/2016 MCW Royalties, a Texas General Partnership, Betty W. Kennaugh and Frank S. Wheelock 9/27/2014 Martha Ann Holliday 9/30/2014 Dwain Graham Oliver 10/2/2014 Leah Jo Loper Brown 9/30/2014 Norma Graham and Joe Mack Graham	Eiche, Mapes and Company, BROGO Minerals, LLC Eiche, Mapes and Company,	2013-00001501 Texas 2013-00001524 Texas 2013-00001517 Texas 2013-00001518 Texas 2013-00001520 Texas 2016-00011630 Texas 2014-00014346 Texas 2014-00014347 Texas 2014-00014345 Texas 2014-00014345 Texas	Henderson Henderson Henderson Henderson Henderson Henderson Henderson Henderson Henderson
42.213.14000.0068 42.213.14000.0077 42.213.14000.0078 42.213.14000.0137 42.213.14000.0148 42.213.14000.0149 42.213.14000.0150 42.213.14000.0151 42.213.14000.0152	10/17/2012 James E. Mitchell 10/6/2012 Barbara Van Horn Cox Revocable Trust, dated 10/26/1999, Barbara Van Horn Cox 11/24/2012 Kenneth Trigg Dealey 11/30/2012 Corley Family Trust, John C. Corley 12/4/2012 Josephine Dealey 7/12/2016 MCW Royalties, a Texas General Partnership, Betty W. Kennaugh and Frank S. Wheelock 9/27/2014 Martha Ann Holliday 9/30/2014 Dwain Graham Oliver 10/2/2014 Leah Jo Loper Brown 9/30/2014 Belinda J. Loper Johnson	Eiche, Mapes and Company, BROGO Minerals, LLC Eiche, Mapes and Company,	2013-00001501 Texas 2013-00001517 Texas 2013-00001518 Texas 2013-00001518 Texas 2013-00001520 Texas 2014-00014346 Texas 2014-00014347 Texas 2014-00014345 Texas 2014-00014345 Texas 2014-00014345 Texas 2014-00014344 Texas	Henderson
42.213.14000.0068 42.213.14000.0077 42.213.14000.0078 42.213.14000.0137 42.213.14000.0148 42.213.14000.0149 42.213.14000.0150 42.213.14000.0151 42.213.14000.0152 42.213.14000.0153	10/17/2012 James E. Mitchell 10/6/2012 Barbara Van Horn Cox Revocable Trust, dated 10/26/1999, Barbara Van Horn Cox 11/24/2012 Kenneth Trigg Dealey 11/30/2012 Corley Family Trust, John C. Corley 12/4/2012 Josephine Dealey 7/12/2016 MCW Royalties, a Texas General Partnership, Betty W. Kennaugh and Frank S. Wheelock 9/27/2014 Martha Ann Holliday 9/30/2014 Dwain Graham Oliver 10/2/2014 Leah Jo Loper Brown 9/30/2014 Belinda J. Loper Johnson 10/2/2014 Barry N Loper	Eiche, Mapes and Company, BROGO Minerals, LLC Eiche, Mapes and Company,	2013-00001524 Texas 2013-00001517 Texas 2013-00001518 Texas 2013-00001518 Texas 2013-00001520 Texas 2014-00011630 Texas 2014-00014346 Texas 2014-00014347 Texas 2014-00014345 Texas 2014-00014344 Texas 2014-00014344 Texas 2014-00014343 Texas 2014-00014343 Texas	Henderson
42.213.14000.0068 42.213.14000.0077 42.213.14000.0078 42.213.14000.0137 42.213.14000.0148 42.213.14000.0149 42.213.14000.0150 42.213.14000.0151 42.213.14000.0152 42.213.14000.0153 42.213.14000.0154	10/17/2012 James E. Mitchell 10/6/2012 Barbara Van Horn Cox Revocable Trust, dated 10/26/1999, Barbara Van Horn Cox 11/24/2012 Kenneth Trigg Dealey 11/30/2012 Corley Family Trust, John C. Corley 12/4/2012 Josephine Dealey 7/12/2016 MCW Royalties, a Texas General Partnership, Betty W. Kennaugh and Frank S. Wheelock 9/27/2014 Martha Ann Holliday 9/30/2014 Dwain Graham Oliver 10/2/2014 Leah Jo Loper Brown 9/30/2014 Norma Graham and Joe Mack Graham 9/30/2014 Belinda J. Loper Johnson 10/2/2014 Mary Ida Graham Morgan	Eiche, Mapes and Company, BROGO Minerals, LLC Eiche, Mapes and Company,	2013-00001501 Texas 2013-00001524 Texas 2013-00001517 Texas 2013-00001518 Texas 2013-00001520 Texas 2014-00014346 Texas 2014-00014347 Texas 2014-00014342 Texas 2014-00014343 Texas 2014-00014343 Texas 2014-00014343 Texas 2014-00014343 Texas 2014-00014343 Texas 2014-00014344 Texas 2014-00014341 Texas	Henderson
42.213.14000.0068 42.213.14000.0077 42.213.14000.0078 42.213.14000.0137 42.213.14000.0148 42.213.14000.0149 42.213.14000.0150 42.213.14000.0151 42.213.14000.0152 42.213.14000.0153 42.213.14000.0154 42.213.14000.0155	10/17/2012 James E. Mitchell 10/6/2012 Barbara Van Horn Cox Revocable Trust, dated 10/26/1999, Barbara Van Horn Cox 11/24/2012 Kenneth Trigg Dealey 11/30/2012 Corley Family Trust, John C. Corley 12/4/2012 Josephine Dealey 7/12/2016 MCW Royalties, a Texas General Partnership, Betty W. Kennaugh and Frank S. Wheelock 9/27/2014 Martha Ann Holliday 9/30/2014 Dwain Graham Oliver 10/2/2014 Leah Jo Loper Brown 9/30/2014 Norma Graham and Joe Mack Graham 9/30/2014 Belinda J. Loper Johnson 10/2/2014 Mary Ida Graham Morgan 9/30/2014 Hollis M. Browning	Eiche, Mapes and Company, BROGO Minerals, LLC Eiche, Mapes and Company,	2013-00001524 Texas 2013-00001524 Texas 2013-00001517 Texas 2013-00001518 Texas 2013-00001520 Texas 2014-00014346 Texas 2014-00014347 Texas 2014-00014342 Texas 2014-00014345 Texas 2014-00014343 Texas 2014-00014343 Texas 2014-00014341 Texas 2014-00014341 Texas 2014-00014341 Texas 2014-00014340 Texas 2014-00014340 Texas	Henderson

Lease #	Date	Lessor	Lessee	Book-Page State	County
42.213.14000.0159	9/30/2014	Bill T. Browning	Eiche, Mapes and Company,	2014-00015024 Texas	Henderson
42.213.14000.0160	10/2/2014	Phillip Oneal Hanna	Eiche, Mapes and Company,	2014-00015025 Texas	Henderson
42.213.14000.0161	9/30/2014	Mary Helen Barnett Israell	Eiche, Mapes and Company,	2014-00015026 Texas	Henderson
42.213.14000.0162	10/2/2014	Betty Graham Fowler	Eiche, Mapes and Company,	2014-00015022 Texas	Henderson
42.213.14000.0163	10/2/2014	Tracie Hanna	Eiche, Mapes and Company,	2014-00015023 Texas	Henderson
42.213.14000.0164	10/2/2014	Jerri Graham Luster	Eiche, Mapes and Company,	2014-00015021 Texas	Henderson
42.213.14000.0165	10/2/2014	Laquinta Graham Palmer	Eiche, Mapes and Company,	2014-00016038 Texas	Henderson
42.213.14000.0166	10/23/2014	Melinda Wynne West Anderson	TDX Energy, LLC	2014-00016033 Texas	Henderson
42.213.14000.0167	10/23/2014	The Catherine Coleman Murchison Exempt Trust, c/o Meadown, Collier, Reed, Cousins, Crouch and Ungerman, LLP, George R. Bedell	TDX Energy, LLC	2014-00016032 Texas	Henderson
42.213.14000.0168	10/23/2014	The Mark V West Testamentary Tr c/o Texas Bank & Tr Company	TDX Energy, LLC	2014-00016034 Texas	Henderson
42.213.14000.0169	10/23/2014	Stephanie Shawn West	TDX Energy, LLC	2014-00016035 Texas	Henderson
42.213.14000.0170	10/23/2014	Ann Waverly West Burford, Deceased	TDX Energy, LLC	2014-00016036 Texas	Henderson
42.213.14000.0171	10/23/2014	Shelly West Bowman	TDX Energy, LLC	2014-00016037 Texas	Henderson
42.213.14000.0172	12/10/2014	Lynda Kay Brown	Eiche, Mapes and Company,	2014-00016998 Texas	Henderson
42.213.14000.0173	12/1/2014	Roy H. Laird Memorial Hospital Trust, J.P. Morgan Chase Bank, N.A.	Eiche, Mapes and Company,	2015-00000234 Texas	Henderson
42.213.14000.0174	12/16/2014	Methodist Childrens Home, Farmers National Company	Eiche, Mapes and Company,	2014-00016997 Texas	Henderson
42.213.14000.0175	10/30/2014	The Bishop Quin Foundation, Bank of America, N. A.	Eiche, Mapes and Company,	2015-00000783 Texas	Henderson
42.213.14000.0176	10/30/2014	The Episcopal Theology Seminar of The Southwest in Austin, Texas, Bank of America, N. A.	Eiche, Mapes and Company,	2015-00000782 Texas	Henderson
42.213.14000.0177	12/16/2014	Revocable Trust Management of Oil, Gas and Mineral Interests for Benefit of Ransome Chance Thomason, Andrew Kenneth	Eiche, Mapes and Company,	2015-00002310 Texas	Henderson
42.213.14000.0180	12/30/2015	Clifford J. Barton	BROGO Minerals, LLC	2017-00007362 Texas	Henderson
42.213.14000.0181	1/5/2016	J. R. Coker	BROGO Minerals, LLC	2016-00011528 Texas	Henderson
42.213.14000.0186	2/25/2016	Thomas Roy Jensen	BROGO Minerals, LLC	2016-00011596 Texas	Henderson
42.213.14000.0188	3/29/2016	Bonny F. McBride and William A. McBride	BROGO Minerals, LLC	2016-00011629 Texas	Henderson
42.213.14000.0191	2/28/2016	Allen Boswell	BROGO Minerals, LLC	2016-00011507 Texas	Henderson
42.213.14000.0196	3/29/2016	DHB Partnership	BROGO Minerals, LLC	2016-00011535 Texas	Henderson
42.213.14000.0197	3/29/2016	DHB Partnership	BROGO Minerals, LLC	2016-00011536 Texas	Henderson
42.213.14000.0198	5/4/2016	Rex Cotter	BROGO Minerals, LLC	2017-00004181 Texas	Henderson
42.213.14000.0199		Archie D. Graham	BROGO Minerals, LLC	2016-00011572 Texas	Henderson
42.213.14000.0200		Gala G. Pryor	BROGO Minerals, LLC	2016-00011436 Texas	Henderson
42.213.14000.0201		Michael Barnett	BROGO Minerals, LLC	2016-00011489 Texas	Henderson
42.213.14000.0203		Lisa Denise Barnett	BROGO Minerals, LLC	2016-00011488 Texas	Henderson
42.213.14000.0205		Vaclay Jim Jezek	BROGO Minerals, LLC	2016-00011599 Texas	Henderson
42.213.14000.0206		Susan B. Kelley	BROGO Minerals, LLC	2016-00011606 Texas	Henderson
42.213.14000.0207		Niki Diane Godwin Majors	BROGO Minerals, LLC	2016-00011627 Texas	Henderson
42.213.14000.0208		Gia Freeman Newell	BROGO Minerals, LLC	2016-00011639 Texas	Henderson
42.213.14000.0209		Vallie Mae Gualdarama	BROGO Minerals, LLC	2016-00011580 Texas	Henderson
42.213.14000.0210		John Bruce Godwin	BROGO Minerals, LLC	2016-00011563 Texas	Henderson
42.213.14000.0211		Wayne West Browning	BROGO Minerals, LLC	2016-00011515 Texas	Henderson
42.213.14000.0212		Bradley Wayne Godwin	BROGO Minerals, LLC	2016-00011556 Texas	Henderson
42.213.14000.0215		Barry J. Boswell	BROGO Minerals, LLC BROGO Minerals, LLC	2016-00011511 Texas	Henderson Henderson
42.213.14000.0217		Deborah Ledoux Porret Fision Fullsright		2016-00016279 Texas 2016-00011550 Texas	
42.213.14000.0218 42.213.14000.0219		Darryl Finis Fulbright	BROGO Minerals, LLC BROGO Minerals, LLC	2016-00011530 Texas 2016-00011549 Texas	Henderson Henderson
42.213.14000.0219		Betty Jo Fulbright Robert Tidgren	BROGO Minerals, LLC	2016-00011349 Texas 2016-00011466 Texas	Henderson
42.213.14000.0222		Douglas Gregory and Joyce Gregory	BROGO Minerals, LLC	2016-00011400 Texas	Henderson
42.213.14000.0225		Christopher K. Storm	BROGO Minerals, LLC	2016-00011459 Texas	Henderson
42.213.14000.0226		Donna R. Pahl	BROGO Minerals, LLC	2017-00007364 Texas	Henderson
42.213.14000.0227		The Elizabeth Andrews Testamentary Tr by Christopher K. Storm	BROGO Minerals, LLC	2016-00011544 Texas	Henderson
42.213.14000.0229		Debra Fulbright Kitchens	BROGO Minerals, LLC	2016-00011611 Texas	Henderson
42.213.14000.0230		David Edward Fulbright	BROGO Minerals, LLC	2016-00011551 Texas	Henderson
42.213.14000.0231		Gary Joe Bass	BROGO Minerals, LLC	2016-00011497 Texas	Henderson
42.213.14000.0232		Sue B. Russell	BROGO Minerals, LLC	2016-00011446 Texas	Henderson
42.213.14000.0233		Barbara Barnett Rainey	BROGO Minerals, LLC	2016-00011437 Texas	Henderson
42.213.14000.0234		Scott Mills	BROGO Minerals, LLC	2016-00011634 Texas	Henderson
42.213.14000.0236		Dan M. Rasco	BROGO Minerals, LLC	2016-00011441 Texas	Henderson
42.213.14000.0237		Michael Cade & Grace Cade Marital Tr, Michael Marlan Cade Trustee	BROGO Minerals, LLC	2016-00011631 Texas	Henderson
42.213.14000.0238		Daniel Gardner	BROGO Minerals, LLC	2016-00011554 Texas	Henderson
42.213.14000.0239		LeRoy Robert Barnett	BROGO Minerals, LLC	2016-00011490 Texas	Henderson
42.213.14000.0241		Nancy D. Turner	BROGO Minerals, LLC	2016-00011470 Texas	Henderson
42.213.14000.0242		Gay Tiner Burrows	BROGO Minerals, LLC	2016-00011521 Texas	Henderson
42.213.14000.0243		Betty Kathryn Browning Whitehead	BROGO Minerals, LLC	2016-00011478 Texas	Henderson
42.213.14000.0244		Melissa Jo Pirtle Hollingsworth	BROGO Minerals, LLC	2016-00011588 Texas	Henderson
42.213.14000.0266		Martha Ann Holiday	BROGO Minerals, LLC	2016-00012790 Texas	Henderson
42.213.14000.0267		Polly Gavord Pena	BROGO Minerals, LLC	2016-00012791 Texas	Henderson
42.213.14000.0268		The Westfall Family Trust by Leslie Stewart Westfall, Trustee	BROGO Minerals, LLC	2017-00018390 Texas	Henderson

Lease #	Date	Lessor	Lessee	Book-Page State	County
42.213.14000.0269		DHC Farms, L.P. by Douglas Hugh Cordell	BROGO Minerals, LLC	2016-00012789 Texas	Henderson
42.213.14000.0270		Dianna R. Lee	BROGO Minerals, LLC	2017-00007363 Texas	Henderson
42.213.14000.0271		Martha Kate Moser Anderson	BROGO Minerals, LLC	2016-00014441 Texas	Henderson
42.213.14000.0272		Chris Allen Moser	BROGO Minerals, LLC	2016-00014440 Texas	Henderson
42.213.14000.0273		Wayne West Browning	BROGO Minerals, LLC	2016-00014442 Texas	Henderson
42.213.14000.0274		John Thomas Trammell	BROGO Minerals, LLC	2016-00014445 Texas	Henderson
42.213.14000.0275		Madelaine Jane Moser Matteson	BROGO Minerals, LLC	2016-00014443 Texas	Henderson
42.213.14000.0276		Gia Freeman Newell	BROGO Minerals, LLC	2016-00014444 Texas	Henderson
42.213.14000.0281		Carl Tidgren	BROGO Minerals, LLC	2016-00014431 Texas	Henderson
42.213.14000.0282		Norma D. Walker Anderson	BROGO Minerals, LLC	2016-00014436 Texas	Henderson
42.213.14000.0283		Evelyn E. Walker Mason	BROGO Minerals, LLC	2016-00014435 Texas	Henderson
42.213.14000.0284		Linda Virginia Reese Keys	BROGO Minerals, LLC	2016-00014434 Texas	Henderson
42.213.14000.0285		Carolyn June Bray	BROGO Minerals, LLC	2016-00014433 Texas	Henderson
42.213.14000.0286		Daniel Pincus	BROGO Minerals, LLC	2016-00014432 Texas	Henderson
42.213.14000.0287		Gretchen Perryman Lahourcade	BROGO Minerals, LLC	2016-00014460 Texas	Henderson
42.213.14000.0288	9/9/2016	Stephanie Young Owens	BROGO Minerals, LLC	2016-00016281 Texas	Henderson
42.213.14000.0289		Robert D. Walker	BROGO Minerals, LLC	2016-00014430 Texas	Henderson
42.213.14000.0290		Vicki L. Walker Wiggin	BROGO Minerals, LLC	2016-00016280 Texas	Henderson
42.213.14000.0291		Deborah Lynn May	BROGO Minerals, LLC	2016-00016277 Texas	Henderson
42.213.14000.0291		William O. Gregg	BROGO Minerals, LLC	2016-00016277 Texas	Henderson
42.213.14000.0292		Betty B. Boswell	BROGO Minerals, LLC	2016-00016282 Texas	Henderson
42.213.14000.0293		Lee Tillman	BROGO Minerals, LLC	2016-00010282 Texas	Henderson
42.213.14000.0295		Windle Lynn Reese	BROGO Minerals, LLC	2016-00017822 Texas	Henderson
42.213.14000.0297		Robert Aaron Reese and Eva Joan Reese	BROGO Minerals, LLC	2016-00017823 Texas	Henderson
42.213.14000.0298		Neill Oil LLC by Virginia B. Price, its Manager	BROGO Minerals, LLC	2016-00016283 Texas	Henderson
42.213.14000.0299		James W. Westfall 1993 Revocable Trust dated 12-14-1993	BROGO Minerals, LLC	2016-00016284 Texas	Henderson
42.213.14000.0299		Robert Scogin	BROGO Minerals, LLC	2016-00010204 Texas	Henderson
42.213.14000.0300		David R. Soulsby	BROGO Minerals, LLC	2016-00017821 Texas	Henderson
42.213.14000.0301		Susan A. Soulsby	BROGO Minerals, LLC	2016-00017824 Texas	Henderson
42.213.14000.0303		Marilyn Sue West	BROGO Minerals, LLC	2016-00011647 Texas	Henderson
42.213.14000.0304		Maxie Nell Hubbard , Gayla Jan Shiilling as Agt & AlF	BROGO Minerals, LLC	2016-00017826 Texas	Henderson
42.213.14000.0304		Nancy McBride and Charles McBride	BROGO Minerals, LLC	2016-00017831 Texas	Henderson
42.213.14000.0303		Mary W. Watson	BROGO Minerals, LLC	2016-00017830 Texas	Henderson
42.213.14000.0307		Horseshoe Minerals, Inc. by Harry S. Armstrong	BROGO Minerals, LLC	2016-00017829 Texas	Henderson
42.213.14000.0308		William J. Watkins	BROGO Minerals, LLC	2016-00017827 Texas	Henderson
42.213.14000.0309		Michael Moose Watson and Katrina A. Watson	BROGO Minerals, LLC	2016-00017828 Texas	Henderson
42.213.14000.0309		Deborah Lynn Hambelton	BROGO Minerals, LLC	2017-000077020 Texas	Henderson
42.213.14000.0310		William Franklin Bell	BROGO Minerals, LLC	2017-00000308 Texas	Henderson
42.213.14000.0311		Charles G. Head	BROGO Minerals, LLC	2016-00017820 Texas	Henderson
42.213.14000.0312		LaRue 1992 Trust by: Sharon Allen LaRue, Camille LaRue Allen, Elaine LaRue Colley, Annette LaRue Morgan and Robert Colley, Co-	BROGO Minerals, LLC	2016-00017486 Texas	Henderson
42.213.14000.0313		Jack Eugene Mounts, Jr.	BROGO Minerals, LLC	2017-000077400 Texas	
42.213.14000.0314		William Grant Blanchard III	BROGO Minerals, LLC	2017-00000313 Texas	Henderson Henderson
42.213.14000.0315			BROGO Minerals, LLC	2017-00000311 Texas	Henderson
42.213.14000.0317		Boyce Paul Head Barbara West Farley	BROGO Minerals, LLC	2017-00000310 Texas	Henderson
42.213.14000.0317		Shawn West Webb	BROGO Minerals, LLC	2017-00000314 Texas 2017-00000315 Texas	
					Henderson
42.213.14000.0319		The Mark V West Testamentary Tr c/o Texas Bank & Tr Company	BROGO Minerals, LLC BROGO Minerals, LLC	2017-00000319 Texas	Henderson
42.213.14000.0320		Robert Milliken Mounts	BROGO Minerals, LLC	2017-00000312 Texas 2017-00000316 Texas	Henderson
42.213.14000.0321		Shawn West Webb			Henderson
42.213.14000.0322		Shelly West Bowman	BROGO Minerals, LLC	2017-00000317 Texas	Henderson
42.213.14000.0323		Shelly West Bowman	BROGO Minerals, LLC	2017-00000318 Texas	Henderson
42.213.14000.0324		The Mark V West Testamentary Tr c/o Texas Bank & Tr Company	BROGO Minerals, LLC	2017-00000320 Texas	Henderson
42.213.14000.0325		Barry Morris Head	BROGO Minerals, LLC	2017-00000331 Texas	Henderson
42.213.14000.0326		William Clough Richardson	BROGO Minerals, LLC	2017-00000330 Texas	Henderson
42.213.14000.0327		The Catherine Coleman Murchison Ex Tr by George R. Bedell, Trustee	BROGO Minerals, LLC	2017-00000322 Texas	Henderson
42.213.14000.0328		Wynne West Anderson	BROGO Minerals, LLC	2017-00000324 Texas	Henderson
42.213.14000.0329		The Catherine Coleman Murchison ExTr by George R. Bedell, Trustee	BROGO Minerals, LLC	2017-00000321 Texas	Henderson
42.213.14000.0330		Wynne West Anderson	BROGO Minerals, LLC	2017-00000323 Texas	Henderson
42.213.14000.0331		Buttram Energies, Inc. by Kristen Berry, Vice President	BROGO Minerals, LLC	2017-00000325 Texas	Henderson
42.213.14000.0332		Buttram Energies, Inc. by Kristen Berry, Vice President	BROGO Minerals, LLC	2017-00000329 Texas	Henderson
42.213.14000.0333		Annette T. Carrancejie	BROGO Minerals, LLC	2017-00000326 Texas	Henderson
42.213.14000.0334		Robert L. Walker and Shelia L. Walker	BROGO Minerals, LLC	2017-00000334 Texas	Henderson
42.213.14000.0335	11/30/2016	David Lee	BROGO Minerals, LLC	2017-00000333 Texas	Henderson
42.213.14000.0336	12/9/2016	Tommy Snowden and Patsy Snowden	BROGO Minerals, LLC	2017-00000328 Texas	Henderson
42.213.14000.0337	12/2/2016	The Adele Blanchard Goodwin Rev Liv Tr by Susan Adele Goodwin Trustee	BROGO Minerals, LLC	2.017E+11 Texas	Henderson
42.213.14000.0338	12/19/2016	Douglas Martin Stockard and Trisha L. Stockard	BROGO Minerals, LLC	2017-00000307 Texas	Henderson

Lease #	Date	Lessor	Lessee	Book-Page State	County
42.213.14000.0339		W. C. "Chip" Perryman III	BROGO Minerals, LLC	2017-00000332 Texas	Henderson
42.213.14000.0340	3/10/2016	G. Wayne Dennis Individually and as Trustee herein in his sole and separate property	BROGO Minerals, LLC	2016-00011687 Texas	Henderson
42.213.14000.0343	11/29/2016	C & A Royalties, L.P. by William R. Seals III Its General Partner	BROGO Minerals, LLC	2017-00001424 Texas	Henderson
42.213.14000.0344	10/4/2016	Samuel Alan Young	BROGO Minerals, LLC	2017-00003150 Texas	Henderson
42.213.14000.0345	12/13/2016	Sally Jacoby Veazey	BROGO Minerals, LLC	2017-00001423 Texas	Henderson
42.213.14000.0346	11/8/2016	Wilma Terry Finley Fam Tr, Murry "Pap" Roark, Trustee	BROGO Minerals, LLC	2017-00001422 Texas	Henderson
42.213.14000.0350	1/19/2017	Wheelock Energy, LP	O'Brien Resources, LLC	2017-00002185 Texas	Henderson
42.213.14000.0351	1/17/2017	Frances Royall Alexander Indivi and as Tr of the Frances Royall Tr under the Will of Frances K. Royall deceased by and through Citizens	O'Brien Resources, LLC	2017-00002188 Texas	Henderson
42.213.14000.0352	1/17/2017	Tucker B. Royall Indiv and as Trustee of the Tucker B. Royall Tr under the will of Frances K. Royall, Deceased by and through Citizens	O'Brien Resources, LLC	2017-00002190 Texas	Henderson
42.213.14000.0353	1/17/2017	Rebecca Royall Indiv and as Rebecca Royall Tr under the will of Frances K.Royall Deceased by and through Citizens National Bank, Agt	O'Brien Resources, LLC	2017-00002189 Texas	Henderson
42.213.14000.0355	1/23/2017	Bandera Minerals, LLC by M.G. Whitmire Managing Manager	BROGO Minerals, LLC	2017-00003146 Texas	Henderson
42.213.14000.0356	2/3/2017	Joni Gail Richardson Neeper	BROGO Minerals, LLC	2017-00003147 Texas	Henderson
42.213.14000.0357	2/3/2017	Jeanne Lynette Richardson Rogers	BROGO Minerals, LLC	2017-0000348 Texas	Henderson
42.213.14000.0358	1/31/2017	Dennis Claude Richardson	BROGO Minerals, LLC	2017-00003149 Texas	Henderson
42.213.14000.0359	1/24/2017	Unknown Heirs of J.D. Brown, Receivership Lease	BROGO Minerals, LLC	2017-00001144 Texas	Henderson
42.213.14000.0360	11/9/2016	David Richardson	BROGO Minerals, LLC	2017-00001429 Texas	Henderson
42.213.14000.0361	1/26/2017	Stephen Anthony Halbert	BROGO Minerals, LLC	2017-00003152 Texas	Henderson
42.213.14000.0362	1/26/2017	Michael Alan Halbert	BROGO Minerals, LLC	2017-00003153 Texas	Henderson
42.213.14000.0363	12/14/2016	Joan Miller Ganz, Indiv & Successor Trusee U/W/O Mary Duke Pearlbrook, Deceased et al and John W. Lyttle, Jr., Indiv and as	BROGO Minerals, LLC	2017-00003155 Texas	Henderson
42.213.14000.0364	2/9/2017	Phyllis Key (Haley)	BROGO Minerals, LLC	2017-00003154 Texas	Henderson
42.213.14000.0365	2/20/2017	Tyler Keith Barnett	BROGO Minerals, LLC	2017-00003156 Texas	Henderson
42.213.14000.0366	8/31/2016	Phillip David Trammell	BROGO Minerals, LLC	2017-00004182 Texas	Henderson
42.213.14000.0368	2/28/2017	Allen B. Boswell	BROGO Minerals, LLC	2017-00004186 Texas	Henderson
42.213.14000.0369	2/15/2017	Nelda Wilbanks	BROGO Minerals, LLC	2017-00004184 Texas	Henderson
42.213.14000.0370	2/28/2017	Susan B. Kelley	BROGO Minerals, LLC	2017-00004185 Texas	Henderson
42.213.14000.0371	2/15/2017	Kenneth M. Wilbanks	BROGO Minerals, LLC	2017-00004183 Texas	Henderson
42.213.14000.0372	1/25/2017	James Leeper Hawley	BROGO Minerals, LLC	2017-00004187 Texas	Henderson
42.213.14000.0373		Joe Rosser Hawley	BROGO Minerals, LLC	2017-00004188 Texas	Henderson
42.213.14000.0374	12/12/2016	Nolan Ray Jensen	BROGO Minerals, LLC	2017-00003151 Texas	Henderson
42.213.14000.0375	3/1/2017	Jeane L. Browning	BROGO Minerals, LLC	2017-00004180 Texas	Henderson
42.213.14000.0376	2/28/2017	Barry J. Boswell	BROGO Minerals, LLC	2017-00004189 Texas	Henderson
42.213.14000.0377		Charles Kittles	BROGO Minerals, LLC	2017-00004601 Texas	Henderson
42.213.14000.0378		Ridgway Management, Inc.	BROGO Minerals, LLC	2017-00005032 Texas	Henderson
42.213.14000.0380			BROGO Minerals, LLC	2017-00004600 Texas	Henderson
42.213.14000.0381		Misty Haley Nettles	BROGO Minerals, LLC	2017-00005031 Texas	Henderson
42.213.14000.0382		Melissa Jo Pirtle Hollingsworth	BROGO Minerals, LLC	2017-00005851 Texas	Henderson
42.213.14000.0383	4/3/2017	Suzanne Comelius Krause	BROGO Minerals, LLC	2017-00005852 Texas	Henderson
42.213.14000.0384	3/28/2017		BROGO Minerals, LLC	2017-00005853 Texas	Henderson
42.213.14000.0386		Receivership Lease - Unknown Heirs of Elinor Ann Kamen Edwards - Case No. CV16-0479-3	BROGO Minerals, LLC	2017-00002757 Texas	Henderson
42.213.14000.0387	4/6/2017	Fern B. Clark, President of DIL Corp., as Managing Partner for Hager Oil & Gas, a Texas General Partnership	BROGO Minerals, LLC	2017-00005850 Texas	Henderson
42.213.14000.0388		Buck Hodge, Jr.	BROGO Minerals, LLC BROGO Minerals, LLC	2017-00005849 Texas	Henderson
42.213.14000.0389		Phillip D. Hatton	BROGO Minerals, LLC	2017-00006624 Texas	Henderson Henderson
42.213.14000.0390 42.213.14000.0391		Sam Tillman Harlan E. Highfield	BROGO Minerals, LLC	2017-00008997 Texas 2017-00006619 Texas	Henderson
42.213.14000.0391		Warren C. Highfield	BROGO Minerals, LLC	2017-00006622 Texas	Henderson
42.213.14000.0392		Warren C. Filgmield Barbara Sue Mayfield Copeland	BROGO Minerals, LLC	2017-00006623 Texas	Henderson
42.213.14000.0393		Mel Jason Pirtle	BROGO Minerals, LLC	2017-00006623 Texas	Henderson
42.213.14000.0394		Jennifer Eileen Hatton Smiley	BROGO Minerals, LLC	2017-00000021 Texas 2017-00007360 Texas	Henderson
42.213.14000.0397		Vickey Lynn Thorpe	BROGO Minerals, LLC	2017-00007359 Texas	Henderson
42.213.14000.0397		RVS Minerals, LLC	BROGO Minerals, LLC	2017-00007359 Texas	Henderson
42.213.14000.0399		Andrew Thomas Hatton	BROGO Minerals, LLC	2017-00007510 Texas	Henderson
42.213.14000.0400		Stephen Hatton	BROGO Minerals, LLC	2017-000077724 Texas	Henderson
42.213.14000.0400		Susan Finegan Malmquist	BROGO Minerals, LLC	2017-00007724 Texas	Henderson
42.213.14000.0401		Marilou B. Burnett	BROGO Minerals, LLC	2017-00008491 Texas	Henderson
42.213.14000.0402		Susan B. Smalheer	BROGO Minerals, LLC	2017-00008491 Texas	Henderson
42.213.14000.0404		Elizabeth Shipley-Moses	BROGO Minerals, LLC	2017-00008494 Texas	Henderson
42.213.14000.0404	6/8/2017		BROGO Minerals, LLC	2017-00000494 Texas	Henderson
42.213.14000.0405		Laurette Ireland, POA for Jay Stanley Ireland	BROGO Minerals, LLC	2017-00009300 Texas	Henderson
42.213.14000.0407		Laurette ireiano, POA for Jay Stanley ireiano Luann Wilkinson	BROGO Minerals, LLC	2017-00009062 Texas 2017-00010930 Texas	Henderson
42.213.14000.0407		Kristin LaRue Perryman	BROGO Minerals, LLC	2017-00010930 Texas	Henderson
42.213.14000.0409		Donald A. Shipley	BROGO Minerals, LLC	2017-0009003 Texas 2017-00012999 Texas	Henderson
42.213.14000.0409		Elizabeth Ann Kight	BROGO Minerals, LLC	2017-00012999 Texas 2017-00012998 Texas	Henderson
42.213.14000.0410		Elizabeth Ann Night Sue Macleod	BROGO Minerals, LLC	2017-00012996 Texas 2017-00010928 Texas	Henderson
42.213.14000.0411		Myra J. LaStrapes	BROGO Minerals, LLC	2017-00010920 Texas	Henderson
42.213.14000.0412		The M.R. Westfall Family Revocable Trust dated November 1, 2010	O'Benco IV, LP	2017-00010329 Texas	Henderson
72.2 IO. 14UUU.U4 I3	0/ 14/2016	The Wart. We statill almity (Cevocable Trust dated Novelliber 1, 2010	O Belico IV, LP	ZUIT-UUUTOJET TEXAS	rienuerson

Lease #	Date	Lessor	Lessee	Book-Page State	County
42.213.14000.0414	3/23/2016	Paul Dustin Godwin, Dylan Cole Godwin and Debbie Godwin Life Tenant	BROGO Minerals, LLC	2016-00011566 Texas	Henderson
42.213.14000.0416	3/22/2018	J. L. Collins Educational Fund	O'Brien Resources, LLC	2018-00008913 Texas	Henderson
42.213.14000.0417	3/22/2018	Wayne West Browning	O'Brien Resources, LLC	2018-00006570 Texas	Henderson
42.213.14000.0418	3/22/2018	Richard Boyd Hardin	O'Brien Resources, LLC	2018-00006571 Texas	Henderson
42.213.14000.0419	3/22/2018	Don D. Johnson et ux	O'Brien Resources, LLC	2018-00006572 Texas	Henderson
42.213.14000.0420	3/22/2018	Paul Dustin Godwin	O'Brien Resources, LLC	2018-00006573 Texas	Henderson
42.213.14000.0421	3/22/2018	Deborah Ann Godwin	O'Brien Resources, LLC	2018-00006574 Texas	Henderson
42.213.14000.0422	3/22/2018	Dylan Cole Godwin	O'Brien Resources, LLC	2018-00006575 Texas	Henderson
42.213.14000.0423	3/22/2018	Thelma Tate Walthall	O'Brien Resources, LLC	2018-00006576 Texas	Henderson
42.213.14000.0424	3/22/2018	The Catherine Coleman Murchison Exempt Tr, George R. Bedell, Trustee	O'Brien Resources, LLC	2018-00006577 Texas	Henderson
42.213.14000.0425	3/22/2018	Shawn West Webb aka Stehpanie Shawn West Webb	O'Brien Resources, LLC	2018-00006578 Texas	Henderson
42.213.14000.0426	3/22/2018	Wynne West Anderson	O'Brien Resources, LLC	2018-00006579 Texas	Henderson
42.213.14000.0427	3/22/2018	Gia Freeman Newell	O'Brien Resources, LLC	2018-00006580 Texas	Henderson
42.213.14000.0428	3/22/2018	William M. Parrish Sr. Estate	O'Brien Resources, LLC	2018-00006581 Texas	Henderson
42.213.14000.0429	3/22/2018	Lois Tate Roady	O'Brien Resources, LLC	2018-00006582 Texas	Henderson
42.213.14000.0430	3/22/2018	J. F. Sharp Revocable Trust dated 2-9-1998	O'Brien Resources, LLC	2018-00006583 Texas	Henderson
42.213.14000.0431	3/22/2018	The Mark V West Testamentary Tr c/o Texas Bank & Tr Company	O'Brien Resources, LLC	2018-00006584 Texas	Henderson
42.213.14000.0432	3/22/2018	Shelly West Bowman	O'Brien Resources, LLC	2018-00006585 Texas	Henderson
42.213.14000.0433	3/22/2018	Paul Thomas Pirtle Estate	O'Brien Resources, LLC	2018-00006586 Texas	Henderson
42.213.14000.0434	3/22/2018	Karen Lee Chamberlain Hunnicutt	O'Brien Resources, LLC	2018-00006587 Texas	Henderson
42.213.14000.0435	3/22/2018	Gary Lee Hardin	O'Brien Resources, LLC	2018-00006588 Texas	Henderson
42.213.14000.0436	3/22/2018	Betty Kathryn Browning Whitehead	O'Brien Resources, LLC	2018-00006589 Texas	Henderson
42.213.14000.0437	3/22/2018	Jennifer Eileen Hatton Smiley	O'Brien Resources, LLC	2018-00006590 Texas	Henderson
42.213.14000.0438	3/22/2018	Phil and Elizabeth Roulac Trust dated 04-18 -1990, and Elizabeth Roulac	O'Brien Resources, LLC	2018-00006591 Texas	Henderson
42.213.14000.0439	3/22/2018	Myra J. LaStrapes	O'Brien Resources, LLC	2018-00006592 Texas	Henderson
42.213.14000.0440	3/22/2018	Susan B. Kelley	O'Brien Resources, LLC	2018-00008915 Texas	Henderson
			O'Brien Resources, LLC	2018-00008914 Texas	Henderson

SCHEDULE 2.2(h)

EXCLUDED CONTRACTS

SCHEDULE 2.2(h) EXCLUDED CONTRACTS

Contract	Party 1	Party 2	Signed	Effective	Stat	e Counties	Prospect	Contract Numb
			3					
1 Participation Agreement	O'BENCO III	Energia Tejas	29-Jun-2012	31-Jul-2012	TX	Anderson & Henderson	Hogan's Goat	
Participation Agreement	O'BENCO IV	BF Oil, LP.	23-Jul-2014	7/23/2014, as	s TX	Anderson & Henderson	Hogan's Goat Prospect	
2				amended December 1, 2014			-	
3 Operation & Maintenance Agreement	O'BENCO IV	BF Oil, LP.		27-Aug-2014	TX	Anderson & Henderson	Hogan's Goat	
Participation Agreement	O'BENCO IV	TDX Resources	15-Jan-2016	1-Nov-2015	TX	Henderson	Thomas Lake	
4								
5 Operation & Maintenance Agreement	ORLLC	O'Benco IV et al	24-Mar-2016		TX	Henderson	Thomas Lake	
6 PSA	O'BENCO IV, LP	Eagle Oil & Gas	29-Aug-2017	1-Oct-2017	TX		Thomas Lake / South LaRue	
7 Participation Agreement	O'BENCO IV, LP	Stroud Exploration Company, L.L.C.	25-Oct-2017	1-Aug-2017	TX	Cass	Hotel California	
8 Farmout Agreement	O'BENCO IV, LP	Stroud Exploration et al	25-Oct-2017		TX	Cass	Hotel California	
9 Farmout Agreement	O'BENCO IV, LP	Stroud Exploration et al	25-Oct-2017		TX		Hotel California	
10 Joint Exploration Agreement	Anadarko E&P Onshore	Calston Exploration USA Inc.	15-Apr-2014	1-Sep-2013	TX		Hotel California	
11 Participation Agreement	O'BENCO IV, LP	Tri-C Resources, LLC	24-May-2016	1-Apr-2016	TX		Blackfoot	
12 Purchase, Sale and Development Agreement	O'BENCO II, LP	Enduring Resources, LLC	31-Dec-2006		TX		Robertson / Bald Prairie	
13 Confidentiality Agreement	O'BENCO IV, LP	5P Investment Partners LLC (Pending 5P execution)	27-Feb-2019					
14 Confidentiality Agreement	O'BENCO IV, LP	Anaco Energy, LLC	21-Jun-2019		+			
15 Confidentiality Agreement	O'BENCO IV, LP	Arcadius Capital Partners, Inc.	1-Feb-2019		1	1		
16 Confidentiality Agreement	O'BENCO IV, LP	Bachtell Oil & Gas Ltd.	12-Mar-2019		+			
17 Confidentiality Agreement	O'BENCO IV, LP	Bain Capital Credit, LP	30-Jan-2019		+			
18 Confidentiality Agreement	O'BENCO IV, LP	BlackGold Capital Management LP	23-Jan-2019		+			
19 Confidentiality Agreement	O'BENCO IV, LP	Bluescape Energy Partners, LLC	14-Mar-2019		+			
20 Confidentiality Agreement	O'BENCO IV, LP	Calston Exploration Inc.	26-Mar-2019		+			
					+			
21 Confidentiality Agreement	O'BENCO IV, LP	Camterra Resources, Inc.	5-Apr-2019		+			
22 Confidentiality Agreement	O'BENCO IV, LP	Carlyle Global Credit Investment Management L.L.C.	31-Jan-2019		+			
23 Confidentiality Agreement	O'BENCO IV, LP	Chambers Energy Management, LP	5-Feb-2019		-			
24 Confidentiality Agreement	O'BENCO IV, LP	Colony HB2 Energy Acquisitions, LLC	11-Jul-2019		-			
25 Confidentiality Agreement	O'BENCO IV, LP	CRD Oil & Gas, LLC	14-Jun-2019		-			
26 Confidentiality Agreement	O'BENCO IV, LP	Denham Capital Management LP	29-May-2019		-			
27 Confidentiality Agreement	O'BENCO IV, LP	Denham Capital Management LP	26-Jun-2019		-			
28 Confidentiality Agreement	O'BENCO IV, LP	Eagle Oil & Gas Co.	19-Jun-2019					
29 Confidentiality Agreement	O'BENCO IV, LP	Edge Natural Resources LLC	7-Jun-2019		_			
30 Confidentiality Agreement	O'BENCO IV, LP	Energia Tejas LLC	2-Apr-2019					
31 Confidentiality Agreement	O'BENCO IV, LP	Exponent Energy, LLC	20-Jun-2019					
32 Confidentiality Agreement	O'BENCO IV, LP	FESCO, Ltd.	15-Mar-2019					
33 Confidentiality Agreement	O'BENCO IV, LP	FPCC USA, Inc.	7-Jul-2019					
34 Confidentiality Agreement	O'BENCO IV, LP	GSO Capital Partners LP	17-Jan-2019					
35 Confidentiality Agreement	O'BENCO IV, LP	Hillside Petroleum LLC	16-Jun-2019					
36 Confidentiality Agreement	O'BENCO IV, LP	HPS Long Creek, LP	16-Jul-2019					
37 Confidentiality Agreement	O'BENCO IV, LP	Juniper Resources, LLC	2-Apr-2019					
38 Confidentiality Agreement	O'BENCO IV, LP	K3 Oil & Gas Operating Company	27-Mar-2019					
39 Confidentiality Agreement	O'BENCO IV, LP	Marabou Midstream Services, LP	11-Jul-2019					
40 Confidentiality Agreement	O'BENCO IV, LP	Merced Capital, L.P.	28-Apr-2019					
41 Confidentiality Agreement	O'BENCO IV, LP	Mission Creek Resources, LLC	19-Jun-2019					
42 Confidentiality Agreement	O'BENCO IV, LP	Mountain Capital Management, LLC	24-Jan-2019					
43 Confidentiality Agreement	O'BENCO IV, LP	MS Capital Partners Adviser Inc.	25-Jan-2019					
44 Confidentiality Agreement	O'BENCO IV, LP	NewCo Oil and Gas, LLC	12-Jul-2019					
45 Confidentiality Agreement	O'BENCO IV, LP	NexGeo Resources LLC	28-Jun-2019					
46 Confidentiality Agreement	O'BENCO IV, LP	North Hudson Resource Partners LP	5-Jun-2019					
47 Confidentiality Agreement	O'BENCO IV, LP	Olympia Energy, LLC	10-Apr-2019					
48 Confidentiality Agreement	O'BENCO IV, LP	OnyxPoint Global Management LP	9-Jul-2019		T			
49 Confidentiality Agreement	O'BENCO IV, LP	Own Resources LLC	18-Jul-2019					
50 Confidentiality Agreement	O'BENCO IV, LP	Petralis Energy Resources, LLC	3-Jul-2019			1		
51 Confidentiality Agreement	O'BENCO IV, LP	Pickrell Acquisitions, Inc.	11-Jul-2019					
52 Confidentiality Agreement	O'BENCO IV, LP	Polaris Production Partners LLC	11-Jul-2019		1	1		
53 Confidentiality Agreement	O'BENCO IV, LP	RedBud Energy Partners, LLC	18-Jun-2019		+	+	+	

SCHEDULE 2.2(h) EXCLUDED CONTRACTS

Contract	Party 1	Party 2	Signed	Effective	State Counties	Prospect	Contract Number
54 Confidentiality Agreement	O'BENCO IV, LP	Redman Resources, LLC	20-Mar-2019				
55 Confidentiality Agreement	O'BENCO IV, LP	RHI Energy LLC	7-Mar-2019				
56 Confidentiality Agreement	O'BENCO IV, LP	Ritchie Exploration, Inc.	19-Jul-2019				
57 Confidentiality Agreement	O'BENCO IV, LP	Sentinel Petroleum, LLC	17-Jun-2019				
58 Confidentiality Agreement	O'BENCO IV, LP	Stellus Capital Management, LLC	29-Jan-2019				
59 Confidentiality Agreement	O'BENCO IV, LP	Summit Natural Resources, LLC	28-Mar-2019				
60 Confidentiality Agreement	O'BENCO IV, LP	Summit Partners Credit Advisors, L.P.	10-Apr-2019				
61 Confidentiality Agreement	O'BENCO IV. LP	Sydri Energy, Inc.	11-Jun-2019				
62 Confidentiality Agreement	O'BENCO IV, LP	Tanglewood Energy Partners LLC	13-Jun-2019				
63 Confidentiality Agreement	O'BENCO IV, LP	Taser, LLC	20-Mar-2019				
64 Confidentiality Agreement	O'BENCO IV, LP	Teton, Ltd	20-Jun-2019				
65 Confidentiality Agreement	O'BENCO IV, LP	Texas Energy Midstream, LLC (Joinder to Sydri CA)	24-Jun-2019				
66 Confidentiality Agreement	O'BENCO IV, LP	Titan Rock Exploration & Production, LLC	6-Mar-2019				
67 Confidentiality Agreement	O'BENCO IV, LP	TPG GLOBAL, LLC	17-Jan-2019				
68 Confidentiality Agreement	O'BENCO IV, LP	TPG Sixth Street Partners, LLC	4-Mar-2019				
69 Confidentiality Agreement	O'BENCO IV. LP	TXOFAC, LLC	18-Jun-2019				
70 Confidentiality Agreement	O'BENCO IV, LP	Valence Operating Company	18-Mar-2019				1
71 Confidentiality Agreement	O'BENCO IV, LP	Valence Operating Company Valence Operating Company	10-Jun-2019				1
72 Confidentiality Agreement	O'BENCO IV, LP	Wagner Oil Company	24-May-2019				1
73 Confidentiality Agreement	O'BENCO IV, LP	Will Energy Management, LLC	13-Mar-2019		+ +		1
74 Confidentiality Agreement	O'BENCO IV, LP	W K Energy, LLC	14-Jun-2019				
75 Master License Agreement for Multiclient Seismic Data, as	ORLLC	WesternGeco, LLC	July 23, 2013		TX	Blackfoot Prospect, Henderson County, TX	
76 Master Geophysical Data-Use License, as amended and	ORLLC	Seismic Exchange, Inc.	June 17, 2005		TX	Anderson, Henderson County, TX	
77 Data License Agreement, as amended and supplemented	ORLLC	HIS Global Inc.	July 15, 2003		1.0	Aliderson, Henderson County, 1X	
	ORLLC	Pro-Gas Services, LLC	September 9, 2014				
78 Master Services Agreement, as amended 79 MSA Exhibit A NGL Tank, as amended	ORLLC	Pro-Gas Services, LLC	October 15, 2014		TX	CSGS Holcomb Gathering	
80 MSA Exhibit A - Clay Filter	ORLLC	Pro-Gas Services, LLC	July 23, 2015		TX	CSGS Holcomb Gathering CSGS Holcomb Gathering	
81 MSA Exhibit A - Clay Filter 81 MSA Exhibit A - Model 1550C FGC	ORLLC		November 10, 2016		TX		
	ORLLC	Pro-Gas Services, LLC			TX	CSGS Holcomb Gathering	
82 MSA Exhibit A - Model 1550C FGC	ORLLC	Pro-Gas Services, LLC	November 26, 2018		TX	CSGS Holcomb Gathering	
83 Gas Treating Equipment Lease Agreement		Gas Processors, Inc.	May 21, 2015		1X	Holcomb/ETC SP Carroll Springs, Henderson	
84 Master Compression Services Agreement	ORLLC	CSI Compressco Operating LLC	July 22, 2015	August 1, 2015			
85 Compressor or Equipment Rental Amendment	ORLLC	CSI - Compressco LP				Larue Shared GL (Formerly Mitchell)	1263
86 Compressor or Equipment Rental Amendment	ORLLC	CSI - Compressco LP		January 4, 2016		Scapegoat #1H Gas Lift	1270
87 Compressor or Equipment Rental Amendment	ORLLC	CSI - Compressco LP		February 15, 2016		Pharris #1H Booster	1273
88 Compressor or Equipment Rental Amendment	ORLLC	CSI - Compressco LP		April 1, 2017		Glad Oaks #1H Gas Lift	1293
89 Compressor or Equipment Rental Amendment	ORLLC	CSI - Compressco LP		October 5, 2015		J A Newland #1H Booster (FORMERLY HILLHOUSE #1 BOOSTER)	1265
90 Compressor or Equipment Rental Amendment	ORLLC	CSI - Compressco LP		September 1, 2016		CSGS - Holcomb Residue #3	1283
91 Compressor or Equipment Rental Amendment	ORLLC	CSI - Compressco LP		October 27, 2016		CSGS - Holcomb VRU #1	129-13
92 Compressor or Equipment Rental Amendment	ORLLC	CSI - Compressco LP		November 18, 2016		CSGS - Holcomb VRU #2	129-13
93 Compressor or Equipment Rental Amendment	ORLLC	CSI - Compressco LP		April 21, 2017		CSGS - Holcomb Overhead Bullet Tank	1293
94 Compressor or Equipment Rental Amendment	ORLLC	CSI - Compressco LP		June 20, 2017		CSGS - Mitchell Booster #2	1297
95 Compressor or Equipment Rental Amendment	ORLLC	CSI - Compressco LP		November 30, 2015		CSGS - Palmer Station #2 (ON 1/2 RATE STANDBY AS OF 6/26/19)	1267
96 Compressor or Equipment Rental Amendment	ORLLC	CSI - Compressco LP		August 18, 2016		CSGS - Palmer Station #3	1282
97 Compressor or Equipment Rental Amendment	ORLLC	CSI - Compressco LP		April 25, 2019		Robertson Gathering #2 (INSTALLED 7/19)	N.
98 Compressor or Equipment Rental Amendment	ORLLC	CSI - Compressco LP		November 10, 2017		BPGS - Robertson Gathering #3	SO-0045
99 Compressor or Equipment Rental Amendment	ORLLC	CSI - Compressco LP		December 18, 2015		BPGS - Devon Fee 5-1 Booster	1269
100 Compressor or Equipment Rental Amendment	ORLLC	CSI - Compressco LP		November 5, 2015		BPGS - Reynolds Booster	1266
101 Compressor or Equipment Rental Amendment	ORLLC	CSI - Compressco LP		November 6, 2017		BPGS - Devon Fee 2-1 Booster	SO-0045
102 Compressor or Equipment Rental Amendment	ORLLC	CSI - Compressco LP		June 13, 2018		Moore-Starcke 5H - Gas Lift	SO-0057
103 Participation Agreement	ORLLC	Klassic Energy Partners, LLC		·	TX Anderson & Henderson	Hogan's Goat	
104 Participation Agreement	ORLLC	MTS Investments, LLC			TX Anderson & Henderson	Hogan's Goat	
05 Letter Agreement	ORLLC	Texas Royalty	29-Jul-2009	22-Jul-2009	TX Limestone & Robertson	North Bald Prairie Extension	
106 Participation Agreement	ORLLC	Nunley Investments, LLC			TX Limestone & Robertson	North Bald Prairie Extension	
107 Participation Agreement	ORLLC	SEPCO II, LLC			TX Limestone & Robertson	North Bald Prairie Extension	
108 Participation Agreement	ORLLC	Crow Horizons Company			TX Limestone & Robertson	North Bald Prairie Extension	
109 Participation Agreement	ORLLC	Minglewood Company			TX Limestone & Robertson	North Bald Prairie Extension	

SCHEDULE 2.2(h) EXCLUDED CONTRACTS

Contract	Party 1	Party 2	Signed	Effective	State	Counties	Prospect	Contract Number
110 Participation Agreement	ORLLC	DOMCO II, LLC			TX	Limestone & Robertson		T
111 Carroll Springs Gathering System Operation	on and Maintenance Agreement by and amon	O'Brien Resources, LLC, as operator and O'BENCO IV, LP and BF OIL,	LP, dated August 27, 201	14. Evergreen				
112 Gas Gathering Agreement, April 29, 2013,	, by and between Marabou Midstream Service	s, LP and O'BENCO IV, LP and the owners of the Robertson Gathering	System, represented by	the operator of such system, O'Brien Re	sources,	LLC. Term – month to month	•	
113 Crude Oil Purchase Agreement, June 6, 20	018, Lion Oil Trading and Transportation and C	BENCO, INC., LOTTOEC0718 ETX. Term June 30, 2019.						
114 Raw Make Purchase Agreement, June 1, 2	2017, ARM Energy and O'Brien Energy Compa	y. Term – month to month.						
115 Plains Crude Oil Purchase Contract, Contr	act No. 3669-1002, September 4, 2008, as am	ended, Plains Marketing, L.P. and O'BENCO INC. Term – month to mo	nth					
116 Orion Pipeline, Ltd. Gas Marketing Agreer	ment between O'BENCO, INC., May 1, 2018. T	rm – month to month.				Cass County		
117 Gas Gathering Agreement, dated June 1,	2015, Enbridge G&P (East Texas) L.P. and O'BE	NCO IV, LP, as amended. Agreement number 2006562. Term – June 3	, 2024					
118 Agency Agreement, June 1, 2015, by and	between O'BENCO IV, LP and Marabou Midstr	eam Services, LP. Term – Month to month.						
119 Agreement for Section 311 Transportatio	n Service, June 1, 2015, by and between Enbri	ige Pipelines (East Texas) LP and O'BENCO IV, LP, as amended. Agree	ment number 2006563. T	erm – May 31, 2024.				
120 Agreement for Intrastate Transportation	Service, June 1, 2015, by and between Enbridg	e Pipelines (East Texas) LP and O'BENCO IV, LP, as amended. Agreem	ent number 2006633. Te	rm – month to month.				
121 Gas Gathering Agreement, May 11, 2018,	by and between Stroud Petroleum, Inc. and C	BENCO IV, LP. Term – Month to month.						
122 Produced Fluids Disposal Agreement, Apr	il 30, 2018, by and between Cass Services, LLC	and O'BENCO IV, LP						
123 Farmout Agreement, dated July 30, 2018,	, by and between Goldco Energy Corporation,	s Farmor, and O'BENCO IV, LP, as Farmee, covering certain lands in 0	ass County, TX					
124 Base Contract for Sale and Purchase of Na	atural Gas, October 28, 2013, O'BENCO IV, LP	nd Marabou Midstream Services, LP. Term – month to month.						
125 Amendment to Special Provisions to NASI	EB Base Contract, May 31, 2016, O'BENCO IV,	P and Marabou Midstream Services, LP. Term – month to month.						
126 Gas Gathering Agreement, April 29, 2013,	, by and between Marabou Midstream Service	s, LP and O'BENCO IV, LP and the owners of the Robertson Gathering	System, represented by	the operator of such system, O'Brien Re	sources,	LLC. Term – month to month.		
127 Engagement Letter, November 1, 2018,	Houlihan Lokey Capital, Inc. and O'BENCO I	/, LP. Term – month to month.						
128 Engagement Letter, March 5, 2018, Sixp	point Partners LLC and O'BENCO IV, LP. Tern	- primary term expired in tail period.						
129 Multiple Confidentiality Agreements be	tween O'BENCO IV, LP, O'BRIEN RESOURCE	, LLC and unaffiliated third parties in connection with the Sixpoint	Partners financing Proc	ess. Dated from April 2018 through Se	ptembe	r 2018.		
130 Confidentiality Agreement between O'E	BENCO IV, LP and an unaffiliated third party	egarding a possible Gas Gathering Agreement between the partie	s. Dated March 5, 2019	. Term is one year				
131 Term Life Insurance Policy issued by Pru	udential on the life of William J. O'Brien III, \$	10,000,000 death benefit, O'BENCO IV, LP – Beneficiary, no cash v	alue.					
132 Term Life Insurance Policy issued by Pru	udential on the life of William J. O'Brien IV, S	10,000,000 death benefit, O'BENCO IV, LP – Beneficiary, no cash v	ralue.					
133 Amended and Restated Agreement of L	imited Partnership, April 28, 2016							
134 Credit Agreement dated as of January 1	0 2017 as amended, by and among O'REN	O IV I P and Associated Bank N.A. as administrative agent for Le	nders					

SCHEDULE 2.2(i)

EXCLUDED SURFACE INTERESTS

				Henderson Co.	Anderson Co.	Anderson Co.
ORLLC No.	Grantor	<u>Grantee</u>	<u>Date</u>	Instrument No.	Volume No.	Page No.
1	Gerald C. Knepple, et ux	O'Brien Resources, LLC	22-May-2013	2015-00007797		
2	James H. Reynolds, Jr., et ux	O'Brien Resources, LLC	22-Oct-2014	2015-00004203		
3	Jerry Nell Mitchell Wilbanks	O'Brien Resources, LLC	31-Oct-2014	2015-00004197		
4	Wayne Mitchell, et ux	O'Brien Resources, LLC	31-Oct-2014	2015-00004198		
5	Wayne Mitchell, et ux	O'Brien Resources, LLC	26-Jan-2016	2016-00001009		
6	Jerry Nell Mitchell Wilbanks	O'Brien Resources, LLC	26-Jan-2016	2016-00001010		
7	Ryan Bell, et ux	O'Brien Resources, LLC	4-Nov-2014	2015-00004199		
8	Gerald C. Knepple, et ux	O'Brien Resources, LLC	11-Nov-2014	2015-00004200		
9	Gerald Hilton, et ux	O'Brien Resources, LLC	11-Nov-2014	2015-00004201		
10	Glenn Holley	O'Brien Resources, LLC	12-Nov-2014	2015-00004202		
11	Roy Schwake	O'Brien Resources, LLC	8-Oct-2014	2014-00014737		
12	Bailey Smiley, et ux	O'Brien Resources, LLC	13-Mar-2014	2014-00012909		
13	William L Brown, et ux	O'Brien Resources, LLC	8-Apr-2014	2014-00012908		
14	Roy Wayne Hillhouse, et ux	O'Brien Resources, LLC	1-Oct-2014	2015-00015963		
15	Derrell Franklin Norman	O'Brien Resources, LLC	7-Mar-2014	2014-00012906		
16	Triton Land Company	O'Brien Resources, LLC	15-Apr-2015	2015-00008739		
17	Gerald C. Knepple, et ux	O'Brien Resources, LLC	24-Feb-2014	2015-00015962		
18	Gerald C. Knepple, et ux	O'Brien Resources, LLC	15-Apr-2014	2013-00013302	1	
19	Gerald C. Knepple, et ux	O'Brien Resources, LLC	28-Jan-2014	2015-00011436		
20	Gerald C. Knepple, et ux	O'Brien Resources, LLC	9-Apr-2014	2015-00011437		
21	Gary Reeves, et ux	O'Brien Resources, LLC	25-Mar-2014	2014-00012905		
22	Dewey Pierce	O'Brien Resources, LLC	10-Apr-2014	2014-00012903		
22-A	Dewey Pierce	O'Brien Resources, LLC	8-Dec-2014	2015-00012904		
	,	·				
23	David H. Till, et al	O'Brien Resources, LLC	22-Apr-2014	2014-00013029		
24	Jacqueline Lansford Capps	O'Brien Resources, LLC	7-May-2014	2014-00013030		
25	Diana S. Collins	O'Brien Resources, LLC	6-May-2014	2014-00013031		
26	Josephine Terry	O'Brien Resources, LLC	26-Jun-2015	2015-00011725		
27	Beverly Henry	O'Brien Resources, LLC	19-Aug-2014	2014-00013033		
28	J.T. & D.A. Robertson Living Trust Dated 5/9/09	O'Brien Resources, LLC	10-Jul-2014	2014-00013034	2436	708
28-A	J.T. & D.A. Robertson Living Trust Dated 5/9/09	O'Brien Resources, LLC	10-Apr-2014	2015-00014561	2508	166
29	Lacey M. Ward, et ux	O'Brien Resources, LLC	7-Aug-2014		2434	711
30	Bobby Joe Kubin, Sr., et ux	O'Brien Resources, LLC	19-Aug-2014		2434	715
31	James M. Fraizer, et ux	O'Brien Resources, LLC	24-Sep-2014		2434	713
32	Richard C. Palmer, et ux	O'Brien Resources, LLC	10-Jul-2015	2015-00011396	2131	713
33	James Robert Garrett	O'Brien Resources, LLC	23-Jul-2014	2014-00014736	2437	497
34	Kasey Marie Reeves Garcia	O'Brien Resources, LLC	1-Sep-2016	2016-00014025	2437	437
35	Ronald C. Holcomb, et al	O'Brien Resources, LLC	6-May-2014	2014-00013040		
36-A	Meazell Partnership	O'Brien Resources, LLC	24-Apr-2015	2015-00013040		
36-B	Meazell Partnership	O'Brien Resources, LLC	24-Apr-2015	2015-00006391		
37	Lonnie C. Vick, Sr., et al	O'Brien Resources, LLC	18-Jul-2014	2014-00013042		
37		O Brieff Resources, LLC	10-Jul-2014	2014-00013042		
38	The Marvin & Rosemary Holcomb Living Trust	O'Brien Resources, LLC	1-Sep-2016	2016-00013371		
39	Roger N. Reed, et ux	O'Brien Resources, LLC	21-Aug-2014	2014-00013245	SPECIAL FORM	
40	Merced Halyard Ventures, LLC	O'Brien Resources, LLC	31-Jul-2014	2014-00013315		
41	Jerald Lambright, et ux	O'Brien Resources, LLC	26-Sep-2014		2469	182
42	Paul W. Mackie, et al	O'Brien Resources, LLC	21-Sep-2015	2015-00016491	2511	106
43	Leslie Lambright	O'Brien Resources, LLC	4-Nov-2015	2016-00001656	2502	352
44	Don E. Gordon, et ux	O'Brien Resources, LLC	8-Dec-2014		2469	180
45	Evelyn Giles Sims	O'Brien Resources, LLC	8-Jun-2015	2015-00007792		
46	Lana G. Hart, et al	O'Brien Resources, LLC	16-Jun-2015	2015-00008014		
47	Lloyd Giles	O'Brien Resources, LLC	12-Jan-2015	2015-00007793		
48	Stephen Friedrich, et al	O'Brien Resources, LLC	15-Jul-2015	2015-00011398		
49	Tommy Lynn Jernigan	O'Brien Resources, LLC	31-Jul-2015	2015-00011399		
50	Roy D. Mills, et al	O'Brien Resources, LLC	16-Oct-2015	2015-00015964		
		,				
51	Carroll Springs Cemetery Association	O'Brien Resources, LLC	9-Feb-2015	2015-00008013		
52	Phillip E. Crombie, et ux	O'Brien Resources, LLC	10-Jan-2017	2017-00000758		

	Carroll Springs Gathering System Schedule											
ORLLC No.	Grantor	<u>Grantee</u>	<u>Date</u>	Henderson Co. Instrument No.	Anderson Co. Volume No.	Anderson Co. Page No.						
53	Kenneth L. Stone, et ux	O'Brien Resources, LLC	8-Jan-2015	2015-00007798								
54	Kenneth M. Farmer	O'Brien Resources, LLC	1-Aug-2014	2015-00007801								
55	Kenneth Jordon, et ux	O'Brien Resources, LLC	30-Jul-2014	2015-00007800								
56	Larry A. Kale	O'Brien Resources, LLC	21-Oct-2014	2015-00007799								
57	Wanda Lindsey Erekson	O'Brien Resources, LLC	9-May-2014	2014-00013035								
58	Patsy Hardy	O'Brien Resources, LLC	7-May-2014	2014-00013036								
59-A	Daniel Bywaters, et ux	O'Brien Resources, LLC	19-Nov-2014	2015-00007794								
59B	Daniel Bywaters, et ux	O'Brien Resources, LLC	14-Sep-2015	2015-00016492								
60	Tommy Lynn Jernigan	O'Brien Resources, LLC	31-Jul-2015	2015-00016494								
61	Scott Glenn Gorman, et al	O'Brien Resources, LLC	20-Aug-2015	2015-00012770								
62	Ezem G. Scarborough, Jr., et al	O'Brien Resources, LLC	15-Dec-2015		2502	359						
63	Prescott Interests, LTD	O'Brien Resources, LLC	8-Feb-2016	2016-00003276								
64	Ruben J. Avalos, et ux	O'Brien Resources, LLC	12-Feb-2016	2016-00002240								
65	Gary and Gail Hartstein Family Trust	O'Brien Resources, LLC	9-May-2016		2517	372						
66	Ezem G. Scarborough, Jr.	O'Brien Resources, LLC	11-May-2016		2518	314						
67	John Buck Scarborough	O'Brien Resources, LLC	11-May-2016		2518	312						
68	The Marvin & Rosemary Holcomb Living Trust	O'Brien Resources, LLC	3-Jan-2017	2017-00000751								
69	Gerald Knepple, et ux	O'Brien Resources, LLC	16-Aug-2016	2017-00000752								
70	Gary Reeves, et ux	O'Brien Resources, LLC	2-Mar-2017	2017-00003287								
71	Texan Real Property, Ltd.	O'Brien Resources, LLC	29-Jul-2016		2532	81						
72	Texan Real Property, Ltd.	O'Brien Resources, LLC	19-Oct-2016		2558	537						
73	Joseph F. Henry, Jr.	O'Brien Resources, LLC	30-Jun-2017		2572	479						
74	James Robert Garrett	O'Brien Resources, LLC	7-Feb-2018	2018-00003305	2599	228						

	Robertson County	Gathering Syste	em Sched	lule	I	
ORLLC No.	Grantor	<u>Grantee</u>	<u>Date</u>	<u>Book</u>	<u>Page</u>	File No.
	H. A. Mentzel, et ux	Enduring Resources, LLC	29-Mar-2006	1093	340	20095849
	Larry W. Morgan, et ux	Enduring Resources, LLC	7-Apr-2006	1093	330	20095847
14705 - 0003	Annie Atamanczuk, Ind and Exe Janie	Enduring Resources, LLC	28-Mar-2006	1093	335	20095848
1 4705 0004	Trojacek, Dec	5 l : D !!C	2.4. 2006	1100	472	20402676
	Edward Trojacek, et ux	Enduring Resources, LLC	2-Apr-2006	1108	472	20102676
14705 - 0005		O'Brien Resources, LLC O'Brien Resources, LLC	22-Feb-2012	1171	442	20122797 20122799
14705 - 0006	Shirley Trojacek Bielamowicz Oak Grove Mining Company LLC	O'Brien Resources, LLC	22-Feb-2012 19-Mar-2013	1171 1199	449 335	20122799
14705 - 0007	<u> </u>	O'Brien Energy Company	19-iviar-2013 14-Aug-2007	999	415	20131570
14705 - 0008	 	O'Brien Energy Company	23-Nov-2007	1017	634	20074767
14705 - 0010		O'Brien Resources, LLC	30-Apr-2008	1017	765	20080193
14705 - 0010		O'Brien Energy Company	15-May-2007	988	735	20083424
14703 0011	Liv Trust	o brieff Effergy company	15 Way 2007	300	733	20072701
14705 - 0012	Oak Grove Management Company LLC	O'Brien Resources, LLC	20-Oct-2010	1120	330	20104926
	Joe Ted Tullos, Sr., et ux	O'Brien Resources, LLC	30-Dec-2010	1124	309	20110105
	Joe Ted Tullos, Sr., et ux	O'Brien Resources, LLC	11-Jan-2008	1019	425	20080488
	Jim A. Bailey, et ux	O'Brien Resources, LLC	11-Jan-2008	1019	429	20080490
	Joe A. Bailey, et ux	O'Brien Resources, LLC	11-Jan-2008	1019	431	20080491
14705 - 0017		O'Brien Resources, LLC	11-Jan-2008	1019	427	20080489
	Oak Grove Management Company LLC	O'Brien Resources, LLC	9-Nov-2009	1093	214	20095826
14705 - 0019	Laura N. Gray	O'Brien Resources, LLC	12-Feb-2009	1066	368	20090857
14705 - 0020	Oak Grove Mining Company LLC/Oak Grove	O'Brien Resources, LLC	9-May-2012	1170	538	20122654
	Management Company LLC					
14705 - 0021	0 , ,	O'Brien Resources, LLC	6-Jun-2011	1139	435	20112767
14705 - 0022		O'Brien Resources, LLC	9-Mar-2009	1070	664	20091727
14705 - 0023	Oak Grove Management Company LLC	O'Brien Resources, LLC	9-Nov-2009	1093	214	20095826
14705 - 0024	. , ,	O'Brien Resources, LLC	3-Jun-2011	1139	439	20112768
	Management Company LLC					
14705 - 0025	· · ·	O'Brien Resources, LLC	3-Jun-2011	1139	444	20112769
14705 - 0026	<u> </u>	O'Brien Resources, LLC	16-May-2012	1171	452	20122800
14705 - 0027	Glenn E. Herzog	O'Brien Resources, LLC	12-Oct-2011	1169	406	20122445
14705 - 0028	3 ,	O'Brien Resources, LLC	16-May-2012	1171	445	20122798
14705 - 0029	3 . ,	O'Brien Resources, LLC	27-Oct-2008	1061	93	20088631
14705 - 0030	Oak Grove Mining Company LLC	O'Brien Resources, LLC	9-Jun-2009	1079	712	20093427
	Oak Grove Mining Company LLC	O'Brien Resources, LLC	16-Mar-2009	1070	677	20091731
	Oak Grove Mining Company LLC	O'Brien Resources, LLC	9-Jun-2009	1079	715	20093428
	Oak Grove Mining Company LLC	O'Brien Resources, LLC	16-Jun-2009	1078	156	20093125
	Jim A. Bailey, et ux	O'Brien Resources, LLC	30-Sep-2010	1146	3	20113928
	Oak Grove Mining Company LLC	O'Brien Resources, LLC	16-Aug-2010	1114	710	20103930
	Oak Grove Mining Company LLC	O'Brien Resources, LLC	16-Mar-2009	1070	674	20091730
14705 - 0037		O'Brien Resources, LLC	30-Aug-2010	1115	700	20104097
14705 - 0038	9	O'Brien Resources, LLC	16-Jun-2009	1078	150	20093123
14705 - 0039	<u> </u>	O'Brien Resources, LLC	9-Nov-2009	1093	218	20095827
14705 - 0040	• • • • • • • • • • • • • • • • • • • •	O'Brien Resources, LLC	7-Aug-2008	1047	365	20085865
14705 - 0041	9	O'Brien Resources, LLC	16-Jun-2009	1078	147	20093122
	Annie Atamanczuk	O'Brien Resources, LLC	11-Jun-2010	1113	14	20103527
	Janie Czajkowski	O'Brien Resources, LLC	27-Sep-2010	1117	406	20104379
	Trojacek Properties LLC	O'Brien Resources, LLC	14-Jun-2010	1113	20	20103529
14705 - 0045		O'Brien Resources, LLC	14-Jun-2010	1113	17	20103528
	Ronnie Trojacek	O'Brien Resources, LLC	10-Jun-2010	1113	8	20103525
	Joan Reynolds	O'Brien Resources, LLC	11-Jun-2010	1113	11	20103526
	Mable L. London	O'Brien Resources, LLC	30-Jul-2008	1047	362 667	20085864 20086158
14705 - 0049 14705 - 0050		O'Brien Resources, LLC O'Brien Resources, LLC	15-Aug-2008 19-Aug-2008	1048 1053	667 481	20086138
	Janie Czajowski	O'Brien Resources, LLC	5-Sep-2008	1053	531	20087205
14/02-0031	Janie Czajowski	O Brieff Resources, LLC	3-36h-5009	1030	221	20000374

	Robertson County Gathering System Schedule											
ORLLC No.	<u>Grantor</u>	<u>Grantee</u>	Date	Book	Page	File No.						
14705 - 0052	Trojacek Properties LLC	O'Brien Resources, LLC	15-Aug-2008	1048	664	20086157						
14705 - 0053	Debbie Molinor	O'Brien Resources, LLC	19-Aug-2008	1048	670	20086159						
14705 - 0054	Darrell Trojacek, et ux	O'Brien Resources, LLC	30-Jul-2008	1047	367	20085866						
14705 - 0055	Koehler Family Trust	O'Brien Resources, LLC	2-Aug-2011	1142	513	20113277						
14705 - 0056	Rene Trevino, Jr.	O'Brien Resources, LLC	19-Jul-2011	1142	216	20113194						
14705 - 0057	Kathryn Trevino Steury	O'Brien Resources, LLC	19-Jul-2011	1142	218	20113195						
14705 - 0058	Eloy Trevino	O'Brien Resources, LLC	19-Jul-2011	1142	214	20113193						
14705 - 0059	Mary Jo Reynolds	O'Brien Resources, LLC	16-Apr-2010	1110	408	20103000						
14705 - 0060	Mary Jo Reynolds	O'Brien Resources, LLC	4-Jun-2008	1042	697	20084947						
14705 - 0061	Maurine Lynan Thomas	O'Brien Resources, LLC	12-Jun-2008	1042	699	20084948						
14705 - 0062	Koehler Family Trust	O'Brien Resources, LLC	10-Jun-2008	1042	694	20084946						
14705 - 0063	Laura N. Gray	O'Brien Resources, LLC	7-Jul-2008	1045	674	20085563						
14705 - 0064	Gerald Heggie	O'Brien Resources, LLC	21-Jul-2008	1045	671	20085562						
14705 - 0065	Caroll Ray Commander, et ux	O'Brien Resources, LLC	19-May-2010	1110	400	20102996						
14705 - 0066	Jacqueline Hailey Vann, et vir	O'Brien Resources, LLC	19-May-2010	1110	402	20102997						
14705 - 0067	Walter Commander, et ux	O'Brien Resources, LLC	19-May-2010	1110	404	20102998						
14705 - 0068	Jerry Wayne Kocian	O'Brien Resources, LLC	21-May-2010	1110	406	20102999						
14705 - 0069	Oak Grove Mining Company LLC	O'Brien Resources, LLC	28-May-2010	1108	28	20102578						
14705 - 0070	Ronald L. White, et ux	O'Brien Resources, LLC	25-Mar-2011	1133	22	20111618						
14705 - 0071	Bernadine Zajac Kubiak	O'Brien Resources, LLC	6-Apr-2011	1133	19	20111617						
14705 - 0072	Rex Sidney Jackson	O'Brien Resources, LLC	6-Apr-2011	1133	13	20111615						
14705 - 0073	Barry Vann, et ux	O'Brien Resources, LLC	6-Apr-2011	1133	16	20111616						
14705 - 0074	Oak Grove Mining Company LLC	O'Brien Resources, LLC	3-Jun-2011	1139	448	20112770						
14705 - 0075	Oak Grove Mining Company LLC	O'Brien Resources, LLC	25-Mar-2009	1070	671	20091729						
14705 - 0076	Candace Elizabeth Wall	O'Brien Resources, LLC	20-Aug-2009	1083	102	20094010						
14705 - 0077	Oak Grove Mining Company LLC	O'Brien Resources, LLC	25-Nov-2009	1096	211	20100333						
14705 - 0078	Ruby Oldham	O'Brien Resources, LLC	14-Aug-2009	1083	104	20094011						
14705 - 0079	Oak Grove Mining Company LLC/Oak Grove	O'Brien Energy Company	30-Nov-2007	1018	85	20080239						
	Power Company LLC											
14705 - 0080	Oak Grove Mining Company LLC	O'Brien Resources, LLC	16-Jun-2009	1078	153	20093124						
14705 - 0081	Johnny Morris Dunn, et ux	O'Brien Resources, LLC	30-Nov-2011	1162	336	20121143						
14705 - 0082	Mary Jo Reynolds (Surface Lease-	O'Brien Energy Company	29-Nov-2007	1017	644	20080196						
	Compressor Site)											
14705 - 0083	Laura N. Gray	O'Brien Resources, LLC	26-Mar-2014	1248	115	20143395						
14705 - 0084	Oak Grove Management Company LLC	O'Brien Resources, LLC	3-Nov-2014	1254	476	20150155						
14705 - 0085	Murlene G. Swingler Trust	O'Brien Resources, LLC	18-Jan-2017	1315	327	20170760						

Excluded Surface Interests

Henderson County, Texas

- All that certain 10.02 acres, more or less in the J.A. Newland Survey, Abstract No. 576.
 General Warranty Deed dated March 20, 2015 between Roger Dean Teselle (Grantor) and
 O'Brien Resources, LLC (Grantee). Instrument number 2015-00003862, Henderson
 County, Texas, more fully described in the Warranty Deed to the Veterans Land Board
 dated May 4, 2000 recorded in Volume 1993, Page 865 of the Real Property Records of
 Henderson County, Texas; and further described in the Contract of Sale and Purchase dated
 May 5, 2000, recorded in Volume 1993, Page 869 of the Real Property Records of
 Henderson County, Texas.
- 2. All that certain 6.159 acres, more or less in the Francis F. Williams Survey, Abstract No. A-793. General Warranty Deed dated October 6, 2015 between The Mackie Family Trust (Grantor) and O'Brien Resources, LLC (Grantee). Instrument number 2015-00015148, Henderson County, Texas, being a part of that certain called 63 acre tract, designated "Second Tract" and a part of that certain 1.88 acre tract, designated as "Fourth Tract", described in deed from Paul W. Mackie, et ux, to The Mackie Family Trust, dated February 19, 2003, and recorded in Volume 2269, Page 866 of the Henderson County Real Property Records.

Robertson County, Texas

All that certain 10.00 acres, more or less in the Benjamin Cage Survey, Abstract No. 87.
 Warranty Deed dated June 21, 2010 between Loui M. Jaksik et ux (Grantor) and O'Brien Resources, LLC (Grantee). Document number 20102742, Book 1109, Page 13 of the Official Records in Robertson County, Texas.

SCHEDULE 2.3(h)

ASSUMED ACCOUNTS PAYABLES

None.

SCHEDULE 2.5(b)

EXECUTORY CONTRACTS AND CURE COSTS

Schedule 2.5(b) - Executory Contracts and Cure	Costs							
Contract	Party 1	Party 2	Signed	Effective	State	Counties	Prospect	Cure Cost
Participation Agreement	O'BENCO IV, LP	Marlin Exploration, LLC	29-Jul-2015	20-Jul-2015	AR	Columbia	Pisgah	\$0
JOA	Getty Oil Company	Mosbacher et al	15-Aug-1968	15-Aug-1968	MS	Clarke	East Nancy	\$0
Compressor or Equipment Rental Amendment	ORLLC	CSI - Compressco LP	15 / tag 1500	August 30, 2018		SO-006167	West Stockman	\$0
Participation Agreement	ORLLC	Team Invest	27-Dec-2011	3-Jan-2012	TX	Navarro	Cheneyboro	\$0
Participation Agreement	ORLLC	Klassic Energy Partners, LLC	22-May-2012		TX	Navarro	Cheneyboro	\$0
Participation Agreement	ORLLC	MTS Investments, LLC	22-May-2012		TX	Navarro	Cheneyboro	+
		(Danny Newell & DSK, Ltd.)						\$0
Participation Agreement	ORLLC	Figure 3 Working Interest, LLC	13-Sep-2011	6-Sep-2011	TX	Garza, Kent, Borden & Scurry	Permian Basin	\$0
Participation Agreement	ORLLC	Klassic Energy Partners, LLC		·	TX	Garza, Kent, Borden & Scurry	Permian Basin	\$0
Participation Agreement	ORLLC	MTS Investments, LLC			TX	Garza, Kent, Borden & Scurry	Permian Basin	\$0
Participation Agreement	O'Brien Energy Company	Pinkston Energy Corporation	27-Dec-2001		TX	Shelby	South Stockman Prospect	\$0
Farmout Agreement	O'Brien Energy Company	Devon Louisiana Corporation	5-Mar-2004		TX	Shelby	South Stockman Prospect	\$0
Participation Agreement	O'Brien Energy Company	Pinkston Energy Corporation	23-Dec-2004		TX	Shelby	West Stockman Prospect	\$0
Participation Agreement	ORLLC	Crow Horizons Company	4-Sep-2008		TX	Nacogdoches	West Stockman Extension	\$0
Participation Agreement	ORLLC	Kyro, Inc.	4-Sep-2008		TX	Nacogdoches	West Stockman Extension	\$0
Participation Agreement	ORLLC	O'BENCO II, LP	4-Sep-2008		TX	Nacogdoches	West Stockman Extension	\$0
Farmout Agreement	ORLLC	Devon Energy Production Company	10-Aug-2009		TX	Nacogdoches	West Stockman Extension	1
							Prospect	\$0
Participation Agreement	ORLLC	Nor-West Kansas Oil, LLC &	20-Jan-2011	1-Jan-2011	KS	Gove	West Gove	\$0
Participation Agreement	ORLLC	Red Oak Energy, Inc. Klassic Energy Partners, LLC	22-Feb-2011	1-Jan-2011	KS	Gove	West Gove	\$0
a dicipation Agreement	ONLLEC	(Dare Johnson)	22 1 CD 2011	1 3811 2011	KS	Gove	West dove	\$0
Participation Agreement	ORLLC	MTS Investments, LLC	22-Feb-2011	1-Jan-2011	KS	Gove	West Gove	1
		(Danny Newell & DSK, Ltd.)						\$0
Oil & Gas Project Agreement	O'Brien Energy Company	TGXploration,	31-Jan-2006	31-Jan-2006	KS	Various	Borealis Project	
		West 183rd, &						60
D-sti-iti A	ORLLC	Virco, Inc. KANEX, LLC & Harvest Energy LLC	1-Oct-2008	1-Oct-2008	I/C	Various	Prairie Wind Project	\$0 \$0
Participation Agreement Crude Oil Purchase Agreement, June 1, 2018, L					KS	various	Prairie WING Project	50
Plains Crude Oil Purchase Contract, Contract No							+	\$0 \$0 \$0
Plains Crude Oil Purchase Contract, Contract No Plains Crude Oil Purchase Contract, Contract No								\$L
Plains Crude Oil Purchase Contract, Contract No					-		-	\$(

SCHEDULE 7.2

OPERATIONS PRIOR TO THE CLOSING DATE

None.

SCHEDULE 8.2(a)

Allocated Values

SCHEDULE 8.2(a)

erated vs Nonop	perated Operator	Prop Num	Property Name	Well Status		County/ Parish		Joint Interest Billing Decimal Interest (Expense)	Total Revenue + minerals (Roy + ORRI + REV WI)	Allocated Value (\$ in 000s)
ed	O'BENCO, Inc.	15709	GAYLAN JANE COLLIER SWD	SWD	42-033-31623	Borden	TX	1.00000000	-	\$0
i	O'BENCO, Inc.	15702	JONES 519#1	producing	42-033-32301	Borden	TX	0.89623921	0.69707957	\$52
d	O'BENCO, Inc.	15705	JONES 519-2H	producing	42-033-32340	Borden	TX	0.89623921	0.69707957	\$900
d	O'BENCO, Inc.	14281	LITTLE #1	Shut-in	42-347-32992	Nacogdoches	TX	0.81062922	0.63097312	\$0
ed	O'BENCO, Inc.	14282	LITTLE #2 (West Gas Unit)	Shut-in	42-347-33043	Nacogdoches	TX	0.80933637	0.55405591	\$0
d	O'BENCO, Inc.	14283	LITTLE #3 (East Gas Unit)	producing	42-347-33055	Nacogdoches	TX	0.80933637	0.62996678	\$0
d	O'BENCO, Inc.	16001	BONNER #1-H	producing	42-349-34987	Navarro	TX	0.97975978	0.76225468	\$0
ed	O'BENCO, Inc.	16003	FINLEY #1	Shut-in	42-349-34998	Navarro	TX	0.99496315	0.75693022	\$0
ed	O'BENCO, Inc.	16011	BRICE BONNER	producing	42-349-35005	Navarro	TX	0.97590080	0.75388337	\$0
ed	O'BENCO, Inc.	14201	SAMFORD #1	Shut-in	42-419-30822	Shelby	TX	0.27562500	0.20576808	\$0
d	O'BENCO, Inc.	14202	SAMFORD #2	Shut-in	42-419-30848	Shelby	TX	0.28436041	0.21228131	\$0
d	O'BENCO, Inc.	14221	GRIFFIN#1	Shut-in	42-419-30997	Shelby	TX	0.58706783	0.43935690	\$0
d	O'BENCO, Inc.	14203	SAMFORD #3	Shut-in	42-419-31056	Shelby	TX	0.27250284	0.20448827	\$0
ed	O'BENCO, Inc.	14222	GRIFFIN #2	Shut-in	42-419-31068	Shelby	TX	0.56855883	0.42641912	\$0
d	O'BENCO, Inc.	14223	GRIFFIN #3	producing	42-419-31091	Shelby	TX	0.58855883	0.44201912	\$0
d	O'BENCO, Inc.	14224	GRIFFIN #4	producing	42-419-31124	Shelby	TX	0.56855883	0.42641912	\$0
d	O'BENCO, Inc.	14204	SAMFORD #4	producing	42-419-31125	Shelby	TX	0.27250284	0.20448827	\$0
d	O'BENCO, Inc.	14225	GRIFFIN #5	producing	42-419-31211	Shelby	TX	0.65758583	0.49318937	\$0
d	O'BENCO, Inc.	14226	GRIFFIN #6	Shut-in	42-419-31457	Shelby	TX	0.82775285	0.61343771	\$0
ed	O'Brien Resources LLC	18803	SWART 21-1	producing	15-063-22107	Gove	KS	0.92110169	0.73515428	\$216
ed	O'Brien Resources LLC	18802	PRATHER FARMS 22-1	producing	15-063-22108	Gove	KS	0.87131298	0.70903094	\$51
d	O'Brien Resources LLC	18806	MSJ-PENNER 28 #1	producing	15-063-22184	Gove	KS	0.94601212	0.75503593	\$468
d	O'Brien Resources LLC	18141	JENNISON 1-1	Shut-in	15-101-22099	Lane	KS	0.63888888	0.49993055	\$0
ed	O'Brien Resources LLC	18143	COLT 45 FBC #2-1	producing	15-101-22117	Lane	KS	0.63888889	0.49993055	\$0
d	O'Brien Resources LLC	18116	BOOMHOWER #36-1	Shut-in	15-101-22151	Lane	KS	0.85185185	0.66657407	\$0
1	O'Brien Resources LLC	18146	JENNISON 1-2 SWD	SWD	15-101-22203	Lane	KS	0.8888889	0.6955555	\$0
d	O'Brien Resources LLC	18148	SNIDER 35-1	Shut-in	15-101-22289	Lane	KS	0.92000000	0.71990000	\$0
d	O'Brien Resources LLC	18228	JENNISON 36 #1	producing	15-101-22311	Lane	KS	0.92000000	0.71990000	\$258
ed	O'Brien Resources LLC	18229	SHARP SEED 2 #1	Shut-in	15-101-22314	Lane	KS	0.80500000	0.62965000	\$0
d	O'Brien Resources LLC	18117	BOOMHOWER 36 #2	Shut-in	15-101-22343	Lane	KS	0.85185185	0.66657407	\$0
1	O'Brien Resources LLC	18230	SNIDER-SHARP SEED 35#1	producing	15-101-22445	Lane	KS	0.86250000	0.67652344	\$195
1	O'Brien Resources LLC	18173	BOOMHOWER 36#3	producing	15-101-22464	Lane	KS	0.85185185	0.66657407	\$0
ed	O'Brien Resources LLC	18174	REDBURN 1-1	producing	15-101-22468	Lane	KS	0.73250000	0.57318125	\$41
d	O'Brien Resources LLC	18234	COLT 45 FBCE 2#4	producing	15-101-22478	Lane	KS	0.66758081	0.52233665	\$110
ed	O'Brien Resources LLC	18175	BOOMHOWER 36 #4	producing	15-101-22481	Lane	KS	0.85185185	0.66657407	\$210
ed	O'Brien Resources LLC	18235	SHARP SEED 2 #2	producing	15-101-22493	Lane	KS	0.80500000	0.62965000	\$213
d	O'Brien Resources LLC	18156	HARPER 35#1	producing	15-101-22516	Lane	KS	0.92000000	0.71990000	\$213
d	O'Brien Resources LLC	18703	VONDRACEK 4#1	producing	15-165-22088	Rush	KS	0.88000000	0.68860000	\$210
d	O'Brien Resources LLC	18707	MOEDER TRUST 3 No 1	producing	15-165-22130	Rush	KS	0.88000000	0.71610000	\$344
d	O'Brien Resources LLC	18714	VONDRACEK ET AL 4 #2	producing	15-165-22133	Rush	KS	0.8800000	0.71610000	
ed.	O'Brien Resources LLC	18715	VONDRACEK ET AL 34 #1	producing	15-165-22136	Rush	KS	0.8800000	0.71610000	\$119 \$5
d	O'Brien Resources LLC	18717	BESPERAT 33 #1	producing	15-165-22139	Rush	KS	0.88000000	0.71610000	
d	O'Brien Resources LLC	18709	MOEDER 3 #2	Shut-in?	15-165-22143	Rush	KS	0.8800000	0.71610000	\$0
d	O'Brien Resources LLC	36207	CLOYD 29 #1	producing	17-013-21108	Rienville	LA	0.21639593	0.15193833	\$0
1	O'Brien Energy Company	42105	MASONITE 17-12	Shut-in	23-023-20108	Clarke	MS	0.97592500	0.1010000	\$0 \$0
1	O'Brien Energy Company O'Brien Energy Company	42105	MASONITE 17-13	Shut-in	23-023-20108	Clarke	MS	0.97592500		
1	O'Brien Energy Company O'Brien Energy Company	42100	REDDOCK 17-15	SMD SWD	23-023-20038	Clarke	MS	0.62938120		\$0
d	O'Brien Energy Company	42110	JESSIE ALLEN 20-3	Shut-in	23-023-20076	Clarke	MS	1.0000000	_	\$0
d	O'Brien Energy Company O'Brien Energy Company	42110	JESSIE ALLEN 20-3 JESSIE ALLEN 17 14 #1	producing	23-023-20049	Clarke	MS	0.63826036	0.49676301	\$0
	Smith Operating and Management	36206	McCULLIN FAMILY TRUST etal #1	producing	17-013-21528	Bienville	LA	0.04860010	0.03532769	\$0
	Smith Operating and Management Franks/Brammer	36206	ALLEN EST NO. 1 ALT		17-013-21528	Bienville Bossier	LA	0.04860010	0.03532769	\$0
	Franks/Brammer Franks/Brammer	36101	ALLEN EST. NO. 1 ALT	producing Shut-in	17-015-	Bossier	LA	0.09389424	0.07302473	\$0
		36102 11308	SPIVEY#1	Shut-in Shut-in	17-015-	Union	LA	0.81455920 0.06185350	0.06339613	\$0
	Comstock Oil & Gas, Lp		SPIVEY#1 WILL DRILL NORA GLIONES		17-027-21567			U.U6185350	0.04408805	\$0
	Indigo Minerals LLC	11340		ORLLC ORRI		Claiborne	LA			\$0
	PetroChem Operating Co, Inc	11311	SPIVEY 17-1	producing	17-027-22334	Claiborne	LA	0.05332189	0.03465923	\$0
	PetroChem Operating Co, Inc	11312	HATTER 17-1	producing	17-027-22344	Claiborne	LA	0.02857038	0.01857075	\$0
	PetroChem Operating Co, Inc	11313	BAILEY 17-1	producing	17-027-22469	Claiborne	LA	0.05332189	0.03465923	\$0
	TMR Exploration	11481	MAX DREYFUS ETAL 106 #1	500% non-consent penalty	17-077-20616	Pointe Coupee	LA	0.12500000	0.09375000	\$0
	Stroud Petroleum Inc.	11271	PAYNE #1	producing	17-111-25591	Union	LA	0.02995000	0.02405983	\$0
	Signal LLC	43301	MASONITE 15 15 #1	Shut-in	23-023-20276	Clarke	MS	0.00343600		\$0
	Signal LLC	43330	NANCY RILEY	Shut-in	23-077-20114	Lawrence	MS	0.00029297	0.00021094	\$0
	Tellus Operating Group LLC	43104	LANE SHANKS UNIT #1	producing	23-129-20039	Smith	MS		0.04204637	\$0

SCHEDULE 8.2(a)

Operated vs Nonoperate	ed Operator	Prop Num	Property Name	Well Status	API	County/ Parish	St	Joint Interest Billing Decimal Interest (Expense)	Total Revenue + minerals (Roy + ORRI + REV WI)	Allocated Value (\$ in 000s)
Nonop	Ranken Energy	19184	BRAUM, W. H. #1-7	producing	35-087-22013	McClain	OK			\$0

SCHEDULE 8.2(b)

Tax Allocation

Schedule 8.2(b)

Tax Allocation

Allocation of Purchase Price under IRC Sec. 1060

	Allocation of
	Purchase Price
Class I	\$0
Class II	\$0
Class III	\$0
Class IV	\$0
Class V	\$3,400,000
Class VI	\$0
Class VII	\$0
Total Purchase Price	\$3,400,000

SCHEDULE 8.5(c)

SECURITY ARRANGEMENTS

SCHEDULE 8.5(c) Security Arrangements

edule of Bonds Updated July 18, 201								
Bond No.	State Recognized Operator	NEXT Renewal Date	Term	Obligee	Bond Description	Current Bor Amount		
800015705	O'Brien Resources, LLC	8/1/2019 (In process of renewal)	12 Month	State of MS Oil and Gas Board	Surety Performance Bond for Plugging of Class II or Oil and Gas Wells	\$100,000		
800015700	O'Brien Resources, LLC	12/11/2019	12 Month	MS State Oil and Gas Board	Mississippi Plugging of Class II or Oil and Gas Wells Bond	\$30,000		
800015704	O'BENCO, Inc.	4/18/2020	12 Month	Navarro County Commission	License and/or Permit Continuous Bond FOR use of Navarro County Roads	\$75,000		
800015706	O'Brien Energy Company	5/18/2020	12 Month	LA Office of Conservation, State of LA	Permit to Drill for Minerals for wells and facilities	\$50,000		

DISCLOSURE SCHEDULE 5.2 REQUIRED CONSENTS

Disclosure Schedule 5.2 Required Consents

DISCLOSURE SCHEDULE 5.2									
COUNTY	STATE	DATE	GRANTOR	GRANTEE	VOLUME	PAGE	DOC NO.	DATE FILED	ASSIGNABILITY
Nacogdoches	TX	2/9/2010	Devon Energy Production	O'Brien Resources, LLC	3603	208	201110031	9/21/2011	Consent
Nacogdoches	TX	2/9/2010	Devon Energy Production	O'Brien Resources, LLC	3603	222	201110032	9/21/2011	Consent

DISCLOSURE SCHEDULE 5.4 MATERIAL ASSIGNED CONTRACTS

	DISCLOSURE SCHEDULE 5.4 - MATERIAL ASSIGN	ED CONTRACTS						
	Contract	Party 1	Party 2	Signed	Effective	State	Counties	Prospect
							- 1 1	
	0	O'BENCO IV, LP	Marlin Exploration, LLC	29-Jul-2015	20-Jul-2015	AR	Columbia	Pisgah
-		Getty Oil Company	Mosbacher et al	15-Aug-1968	15-Aug-1968	MS	Clarke	East Nancy
	compressor or Equipment nentary internament	ORLLC	CSI - Compressco LP		August 30, 2018		SO-006167	West Stockman
4		ORLLC	Team Invest	27-Dec-2011	3-Jan-2012	TX	Navarro	Cheneyboro
5	Participation Agreement	ORLLC	Klassic Energy Partners, LLC	22-May-2012		TX	Navarro	Cheneyboro
6	Participation Agreement	ORLLC	MTS Investments, LLC (Danny Newell & DSK, Ltd.)	22-May-2012		TX	Navarro	Cheneyboro
7	Participation Agreement	ORLLC	Figure 3 Working Interest, LLC	13-Sep-2011	6-Sep-2011	TX	Garza, Kent, Borden & Scurry	Permian Basin
8	Participation Agreement	ORLLC	Klassic Energy Partners, LLC		·	TX	Garza, Kent, Borden & Scurry	Permian Basin
9	Participation Agreement	ORLLC	MTS Investments, LLC			TX	Garza, Kent, Borden & Scurry	Permian Basin
10	Participation Agreement	O'Brien Energy Company	Pinkston Energy Corporation	27-Dec-2001		TX	Shelby	South Stockman Prospec
11	Farmout Agreement	O'Brien Energy Company	Devon Louisiana Corporation	5-Mar-2004		TX	Shelby	South Stockman Prospec
12	Participation Agreement	O'Brien Energy Company	Pinkston Energy Corporation	23-Dec-2004		TX	Shelby	West Stockman Prospect
13	Participation Agreement	ORLLC	Crow Horizons Company	4-Sep-2008		TX	Nacogdoches	West Stockman Extension
14	Participation Agreement	ORLLC	Kyro, Inc.	4-Sep-2008		TX	Nacogdoches	West Stockman Extensio
15	Participation Agreement	ORLLC	O'BENCO II, LP	4-Sep-2008		TX	Nacogdoches	West Stockman Extension
16	Farmout Agreement	ORLLC	Devon Energy Production Company	10-Aug-2009		TX	Nacogdoches	West Stockman Extension Prospect
17	Participation Agreement	ORLLC	Nor-West Kansas Oil, LLC & Red Oak Energy, Inc.	20-Jan-2011	1-Jan-2011	KS	Gove	West Gove
18	Participation Agreement	ORLLC	Klassic Energy Partners, LLC (Dare Johnson)	22-Feb-2011	1-Jan-2011	KS	Gove	West Gove
19	Participation Agreement	ORLLC	MTS Investments, LLC (Danny Newell & DSK, Ltd.)	22-Feb-2011	1-Jan-2011	KS	Gove	West Gove
	, ,	O'Brien Energy Company	TGXploration, West 183rd, &	31-Jan-2006	31-Jan-2006	KS	Various	Borealis Project
20			Virco Inc					
	i articipation Agreement	ORLLC	KANEX, LLC & Harvest Energy LLC	1-Oct-2008	1-Oct-2008	KS	Various	Prairie Wind Project
	Crude Oil Purchase Agreement, June 1, 2018, Li						-	
	Plains Crude Oil Purchase Contract, Contract No Plains Crude Oil Purchase Contract. Contract No						ļ	
	Plains Crude Oil Purchase Contract, Contract No Plains Crude Oil Purchase Contract, Contract No							

DISCLOSURE SCHEDULE 5.6

WELLS

SCHEDULE 5.6 Wells

Well Name	Location County/Parish	State / Lease Block	Play / Field / Trend	Land / Inland Waters / Offshore	Operator	Status of well - (Prod./SI/Inj/SWD /TA/P&A)	HBP / Not- HBP	Multi-Well Pad	Wells / Pad	Total Vertical Depth (TVD)	Total Measured Depth (TMD)	P&A Cost - Net equip and tubulars
JENNISON 1 No. 1,	Lane	KS	Wildcat	Land	O'Brien Resources, LLC	SI	Expired		1	4,613	4,613	\$10,000
JESSIE ALLEN 20-3 No. 1,	Clarke	MS	East Nancy	Land	O'Brien Resources, LLC	SI	Expired		1	14,400	14,400	\$40,000
Masonite 17-12 No. 1,	Clarke	MS	East Nancy	Land	O'Brien Resources, LLC	SI	Expired		1	14,400	14,400	\$40,000
Masonite 17-13 No. 1,	Clarke	MS	East Nancy	Land	O'Brien Resources, LLC	SI	Expired		1	14,750	14,750	\$40,000
LITTLE LEASE No. 2,	Nacogdoches	TX	Stockman (Cotton Valley)	Land	O'Benco, Inc.	SI	Expired		1	12,510	12,510	\$35,000
FINLEY No. 1H,	Navarro	TX	Cheneyboro Cotton Valley	Land	O'Benco, Inc.	SI	Expired		1	9,625	14,214	\$35,000

\$200,000

DISCLOSURE SCHEDULE 5.7 IMBALANCES

DISCLOSURE SCHEDULE 5.8

AFEs

SCHEDULE 5.8 AFEs

Non Stalking Horse Properties

						Tota	l Billed to
AFE#	Property#	Туре	Name	AFE (Gross Total	Date	
196009	15702	WORKOVER	JONES #1H-TBG LEAK REPAIR	\$	90,000.00	\$	86,701.83
194003	15705	PUMPING UNIT	JONES #2H-ESP TO PUMPING UNIT	\$	178,180.00	\$	177,750.14

DISCLOSURE SCHEDULE 5.9 NON-CONSENT OPERATIONS

Disclosure Schedule 5.9 - Non-Consent Operations

	<u>Prop</u>								WI Revenue	<u>NRI</u>
<u>Operator</u>	Num	Property Name	Well Status	<u>API</u>	County/ Parish	St	Name	Decimal Interest	Decimal Interest	(Revenue WI / Expense WI)
TMR Exploration	11481 MAX	DREYFUS ETAL 106 #1	500% NC penalty	17-077-20616	Pointe Coupee	LA		.12500000	.09375000	75.00%

DISCLOSURE SCHEDULE 5.11 PREFERENTIAL PURCHASE RIGHTS

DISCLOSURE SCHEDULE 5.12 INTELLECTUAL PROPERTY

DISCLOSURE SCHEDULE 5.14 LEGAL PROCEEDINGS

DISCLOSURE SCHEDULE 5.15 SUFFICIENCY OF THE OIL AND GAS ASSETS

DISCLOSURE SCHEDULE 5.16 COMPLIANCE WITH LAWS

DISCLOSURE SCHEDULE 5.19 TITLE MATTERS